

Comparative Effectiveness Review Number 260

Nutrition as Prevention for Improved Cancer Health Outcomes



Comparative Effectiveness Review

Number 260

Nutrition as Prevention for Improved Cancer Health Outcomes

Prepared for:

Agency for Healthcare Research and Quality U.S. Department of Health and Human Services 5600 Fishers Lane Rockville, MD 20857 www.ahrq.gov

Contract No. 75Q80120D00008

Prepared by:

Minnesota Evidence-based Practice Center Minneapolis, MN

Investigators:

Helen M. Parsons, Ph.D., M.P.H.
Mary L. Forte, Ph.D., D.C.
Hamdi I. Abdi, M.P.H.
Sallee Brandt, M.P.H.
Amy M. Claussen, M.L.I.S.
Timothy J. Wilt, M.D., M.P.H.
Mark Klein, M.D.
Elizabeth Ester, M.D.
Adrienne Landsteiner, Ph.D., M.P.H.
Aasma Shaukut, M.D., M.P.H.
Shalamar D. Sibley, M.D., M.P.H.
Joanne Slavin, Ph.D., R.D.N.
Catherine Sowerby, B.A.
Weiwen Ng, M.P.H.
Mary Butler, Ph.D., M.B.A.

AHRQ Publication No. 23-EHC004 May 2023 This report is based on research conducted by the Minnesota Evidence-based Practice Center (EPC) under contract to the Agency for Healthcare Research and Quality (AHRQ), Rockville, MD (Contract No. 75Q80120D00008). The report was commissioned and funded by the National Institutes of Health (NIH) Office of Disease Prevention to inform a Pathways to Prevention Workshop. The findings and conclusions in this document are those of the authors, who are responsible for its contents; the findings and conclusions do not necessarily represent the views of AHRQ or NIH. Therefore, no statement in this report should be construed as an official position of NIH, AHRQ, or the U.S. Department of Health and Human Services.

None of the investigators have any affiliations or financial involvement that conflicts with the material presented in this report.

The information in this report is intended to help healthcare decision makers—patients and clinicians, health system leaders, and policymakers, among others—make well-informed decisions and thereby improve the quality of healthcare services. This report is not intended to be a substitute for the application of clinical judgment. Anyone who makes decisions concerning the provision of clinical care should consider this report in the same way as any medical reference and in conjunction with all other pertinent information, i.e., in the context of available resources and circumstances presented by individual patients.

This report is made available to the public under the terms of a licensing agreement between the author and the Agency for Healthcare Research and Quality. Most AHRQ documents are publicly available to use for noncommercial purposes (research, clinical or patient education, quality improvement projects) in the United States, and do not need specific permission to be reprinted and used unless they contain material that is copyrighted by others. Specific written permission is needed for commercial use (reprinting for sale, incorporation into software, incorporation into for-profit training courses) or for use outside of the United States. If organizational policies require permission to adapt or use these materials, AHRQ will provide such permission in writing.

AHRQ or U.S. Department of Health and Human Services endorsement of any derivative products that may be developed from this report, such as clinical practice guidelines, other quality enhancement tools, or reimbursement or coverage policies, may not be stated or implied.

A representative from AHRQ served as a Contracting Officer's Representative and reviewed the contract deliverables for adherence to contract requirements and quality. AHRQ did not directly participate in the literature search, determination of study eligibility criteria, data analysis, interpretation of data, or preparation or drafting of this report.

AHRQ appreciates appropriate acknowledgment and citation of its work. Suggested language for acknowledgment: This work was based on an evidence report, Nutrition as Prevention for Improved Cancer Health Outcomes, by the Evidence-based Practice Center Program at the Agency for Healthcare Research and Quality (AHRQ).

Suggested citation: Parsons HM, Forte ML, Abdi H, Brandt S, Claussen AM, Wilt TJ, Klein M, Ester E, Landsteiner A, Shaukut A, Sibley SD, Slavin J, Sowerby C, Ng W, Butler M. Nutrition as Prevention for Improved Cancer Health Outcomes. Comparative Effectiveness Review No. 260. (Prepared by the Minnesota Evidence-based Practice Center under Contract No. 75Q80120D00008.) AHRQ Publication No. 23-EHC004. Rockville, MD: Agency for Healthcare Research and Quality; May 2023. DOI: <u>https://doi.org/10.23970/AHRQEPCCER260</u>. Posted final reports are located on the Effective Health Care Program search page.

Preface

The Agency for Healthcare Research and Quality (AHRQ), through its Evidence-based Practice Centers (EPCs), sponsors the development of evidence reports and technology assessments to assist public- and private-sector organizations in their efforts to improve the quality of healthcare in the United States.

The National Institutes of Health Office of Disease Prevention requested this report from the EPC Program at AHRQ to inform a Pathways to Prevention Workshop. The NIH Office of Disease Prevention provided the funding for this report through an Inter-Agency Agreement with AHRQ. AHRQ assigned this report to the following EPC: Minnesota Evidence-based Practice Center (Contract Number: 75Q80120D00008).

The reports and assessments provide organizations with comprehensive, evidence-based information on common medical conditions and new healthcare technologies and strategies. They also identify research gaps in the selected scientific area, identify methodological and scientific weaknesses, suggest research needs, and move the field forward through an unbiased, evidence-based assessment of the available literature. The EPCs systematically review the relevant scientific literature on topics assigned to them by AHRQ and conduct additional analyses when appropriate prior to developing their reports and assessments.

To bring the broadest range of experts into the development of evidence reports and health technology assessments, AHRQ encourages the EPCs to form partnerships and enter into collaborations with other medical and research organizations. The EPCs work with these partner organizations to ensure that the evidence reports and technology assessments they produce will become building blocks for healthcare quality improvement projects throughout the Nation. The reports undergo peer review and public comment prior to their release as a final report.

AHRQ expects that the EPC evidence reports and technology assessments, when appropriate, will inform individual health plans, providers, and purchasers as well as the healthcare system as a whole by providing important information to help improve healthcare quality.

If you have comments on this evidence report, they may be sent by mail to the Task Order Officer named below at: Agency for Healthcare Research and Quality, 5600 Fishers Lane, Rockville, MD 20857, or by email to epc@ahrq.hhs.gov.

Robert Otto Valdez, Ph.D., M.H.S.A. Director Agency for Healthcare Research and Quality	Therese Miller, Dr.P.H. Acting Director Center for Evidence and Practice Improvement Agency for Healthcare Research and Quality
Craig A. Umscheid, M.D., M.S.	Suchitra Iyer, Ph.D.
Director	Task Order Officer
Evidence-based Practice Center Program	Center for Evidence and Practice
Center for Evidence and Practice Improvement	Improvement
Agency for Healthcare Research and Quality	Agency for Healthcare Research and Quality

Acknowledgments

We thank many individuals for their contributions to this project: Suchitra Iyer (Task Order Officer) and Nora Mueller from the Agency for Healthcare Research and Quality (AHRQ); Carrie Klabunde, Keisha Shropshire, and Elizabeth Vogt from the National Institutes of Health (NIH) Office of Disease Prevention (ODP); Christopher Lynch and Roberto Flores from the Office of Nutrition Research (ONR); Karen Regan from the Office of Dietary Supplements (ODS); Elaine Trujillo, Sharon Ross, Joanne Elena, and Linda Nebeling from the National Cancer Institute (NCI); Ashley Vargas and Kimberlea Gibbs from the National Institute of Child Health and Human Development (NICHD); and Marcel Salive and Yih-Woei Fridell from the National Institute on Aging (NIA). We also thank Jeannine Ouellette for her exceptional editing, Bessie Peterson for her professional report preparation, Eric Linskens and Lauren McKenzie for their assistance with abstract reviews, and Carla Kahle for her technical skills that kept all our computers functional.

Technical Expert Panel

In designing the study questions and methodology at the outset of this report, the EPC consulted several technical and content experts. Broad expertise and perspectives were sought. Divergent and conflicting opinions are common and perceived as healthy scientific discourse that results in a thoughtful, relevant systematic review. Therefore, in the end, study questions, design, methodologic approaches, and/or conclusions do not necessarily represent the views of individual technical and content experts.

Technical Experts must disclose any financial conflicts of interest greater than \$5,000 and any other relevant business or professional conflicts of interest. Because of their unique clinical or content expertise, individuals with potential conflicts may be retained. The TOO and the EPC work to balance, manage, or mitigate any potential conflicts of interest identified.

The list of Technical Experts who provided input to this report follows:

Mridul Datta, Ph.D., R.D., L.D., FAND* Clinical Associate Professor Iowa State University Ames, IA

Charles Loprinzi, M.D. Medical Oncologist Mayo Clinic Rochester, MN

Electra D. Paskett, Ph.D.* Professor The Ohio State University Columbus, OH Kimberly Robien, Ph.D., R.D., L.D., CSO, FAND* Associate Professor George Washington University Washington, DC

Eric Roeland, M.D., FAAHPM Professor and Physician Harvard Medical School and Massachusetts General Hospital Cancer Center Boston, MA

Jeffrey White, M.D.* Associate Director National Cancer Institute Rockville, MD

*Provided input on Draft Report.

Peer Reviewers

Prior to publication of the final evidence report, EPCs sought input from independent Peer Reviewers without financial conflicts of interest. However, the conclusions and synthesis of the scientific literature presented in this report do not necessarily represent the views of individual reviewers. AHRQ may also seek comments from other Federal agencies when appropriate.

Peer Reviewers must disclose any financial conflicts of interest greater than \$5,000 and any other relevant business or professional conflicts of interest. Because of their unique clinical or content expertise, individuals with potential nonfinancial conflicts may be retained. The TOO and the EPC work to balance, manage, or mitigate any potential nonfinancial conflicts of interest identified.

The list of Peer Reviewers follows:

Aminah Jatoi, M.D. Medical Oncologist Mayo Clinic Rochester, MN

Harold Lau, M.D. Professor Department of Oncology University of Calgary Calgary, Alberta, Canada

Nutrition as Prevention for Improved Cancer Health Outcomes

Structured Abstract

Objective. To understand the evidence base for nutrition interventions delivered prior to or during cancer treatment for preventing and treating negative cancer and cancer treatment–related outcomes among individuals with or at risk for malnutrition. The primary purpose was to inform the National Institutes of Health (NIH) Pathways to Prevention workshop *Nutrition as Prevention for Improved Cancer Health Outcomes*, held July 26–28, 2022.

Data sources. We searched Ovid Medline[®], Ovid Embase[®], and Cochrane Central Register of Controlled Trials to identify studies from 2000 through July 2022. We conducted grey literature searches to identify additional resources relevant to the associated costs or value (e.g., cost-effectiveness, cost-benefit) of nutrition interventions.

Review methods. The review was guided by a set of Key Questions established by the NIH planning committee for the *Nutrition as Prevention for Improved Cancer Health Outcomes* workshop. We searched for studies that evaluated a broad range of nutrition interventions (e.g., dietary supplements, nutrition support, nutrition counseling) for preventing and treating negative outcomes of cancer and cancer-related treatment. Eligible studies included randomized controlled trials (RCTs) with enrollment \geq 50 participants. We extracted basic study information from all eligible studies, then grouped studies by broad intervention and cancer types. We provide a detailed evidence map for all included studies, but conducted risk of bias and additional qualitative descriptions of outcomes for only those intervention and cancer types with a larger volume of literature.

Results. We identified 9,798 unique references, with 206 studies from 219 publications reporting RCTs of nutrition interventions to potentially improve negative outcomes of cancer and cancerrelated treatment. Two decades of randomized trial evidence on nutrition interventions for adults prior to and/or during cancer treatment primarily focused on dietary supplements, nutrition support (including oral nutrition supplements), and the route or timing of nutrition interventions for gastrointestinal and head and neck cancers in the inpatient setting. Most studies evaluated changes in body weight/composition, adverse events, length of hospital stay, and quality of life. Few studies were conducted within the U.S. setting. Among intervention and cancer types with a high volume of literature (n=114), which predominantly included studies in dietary supplements and nutrition support in gastrointestinal and head and neck cancers, 11 percent (n=12) were rated as low risk of bias (higher quality), 40 percent (n=46) medium risk of bias, and 49 percent (n=56) high risk of bias (lower quality). Low and medium risk-of-bias studies reported mixed results on the effect of nutrition interventions across cancer and treatment-related outcomes. Although the evidence map shows a large volume of studies evaluating nutrition interventions and outcomes, these studies showed high heterogeneity across study populations, interventions, and outcomes (measure definitions, timing of measurements), even within nutrition intervention categories; as a result, we could not aggregate results. While studies enrolled individuals from multiple cancer types, treatments, and stages, across the lifespan, with varying degrees of muscle

wasting, and in those with a range of comorbid conditions, no eligible studies specifically evaluated whether the effects of nutrition interventions on preventing negative outcomes varied across these characteristics.

Among studies included in our Key Questions, we found that few (4%, n=8) published cost or value (e.g., cost-effectiveness, cost-benefit) information related to the intervention. In our grey literature search of additional studies examining cost or value of nutrition interventions, we found few studies that conducted cost-effectiveness or cost-benefit analyses; among those that did, we found the studies were conducted in non-U.S. health systems and demonstrated mixed results on the value of nutrition interventions.

Conclusions. Although overall RCT evidence focused on a wide range of nutrition interventions, studies were concentrated in use of dietary supplements, nutrition support, and the route or timing of nutrition interventions within gastrointestinal and head and neck cancers in inpatient settings. Among interventions with the highest volume of literature, the majority of studies were rated as high risk of bias. Our findings point to the need for rigorous new research to bolster the evidence base. Specifically, the field needs a more detailed future evaluation of a subset of nutrition interventions contained in this evidence map that focuses on priorities most relevant to specific stakeholders (e.g., oncologists, patients, dietitians, researchers, policymakers). Further, studies should be specifically designed to evaluate the main outcomes of interest for clinical practice. Future research would also benefit from creation of standardized taxonomies for interventions and outcomes as well as more rigorous design and reporting of nutrition interventions. As mentioned, heterogeneity of populations, interventions, comparators, and outcomes precluded aggregation. Currently, the quality and heterogeneity of the studies limit translation of findings into clinical practice or guidelines. In order to inform development of these guidelines, coordinated efforts are required to develop detailed conceptual frameworks for mechanisms of nutrition interventions most relevant to clinical care providers and patients. Such frameworks would help inform priorities for future research as well as guide practice and policy.

Contents

Executive Summary	. 1
Chapter 1. Introduction	. 1
Background and Objective for Systematic Review	. 1
Purpose and Scope	. 2
Report Organization	. 2
Chapter 2. Methods	, 4
Review Approach	. 4
Research (Key) Questions	. 4
Analytic Framework	. 9
Search Strategy and Study Selection	11
Assessing Methodological Risk of Bias of Individual Studies	12
Data Abstraction and Data Management	12
Data Synthesis	12
Chapter 3. Search Results	14
Chapter 4. Overview of Nutrition Interventions	16
Key Points	10
Chapter 5. Nutrition Interventions Prior to Concern Treatment	
Chapter 5. Nutrition Interventions Prior to Cancer Treatment	24 24
Charmient (24 77
Dietary Supplements	24
	25
Nutrition Support Including Oral Nutrition Supplements	26
Risk of Bias and Outcome Assessment	28
Chapter 6. Nutrition Interventions Prior to and Including the Initiation of Cancer	
Treatment	30
Key Points	30
Overview	30
Dietary Supplements	31
Route or Timing of Nutrition Interventions	33
Nutrition Support Including Oral Nutrition Supplements	35
Multi-Component Interventions	37
Risk of Bias and Outcome Assessment	37
Chapter 7. Nutrition Interventions After Cancer Treatment Began	39
Key Points	39
Overview	39
Nutrition Counseling	40
Dietary Supplements	42
Special Diets	46
Route or Timing of Nutrition Interventions	47

Multi-Component Interventions 52 Risk of Bias and Outcome Assessment 54 Chapter 8. Effect of Nutrition Interventions on Symptoms 56 Key Points 56 Overview 56 Nutrition Counseling 57 Dictary Supplements 59 Special Dicts 61 Route or Timing of Nutrition Interventions 62 Nutrition Support Including Oral Nutrition Supplements 64 Multi-Component Interventions 67 Risk of Bias and Outcome Assessment 68 Chapter 9. Effect of Nutrition Interventions Intended for Body Weight Loss 71 Overview 71 Special Diets 73 Chapter 10. Cost-Effectiveness of Nutrition Interventions 73 Key Points 73 Cost-Effectiveness 74 Special Diets 73 Chapter 11. Discussion 77 Outcomes<	Nutrition Support Including Oral Nutrition Supplements	49
Risk of Bias and Outcome Assessment 54 Chapter 8. Effect of Nutrition Interventions on Symptoms. 56 Key Points 56 Overview 56 Nutrition Counseling. 57 Dietary Supplements 59 Special Diets 61 Route or Timing of Nutrition Interventions. 61 Nutrition Support Including Oral Nutrition Supplements 64 Multi-Component Interventions 67 Risk of Bias and Outcome Assessment 68 Chapter 9. Effect of Nutrition Interventions Intended for Body Weight Loss 71 Overview. 71 Overview. 71 Overview. 71 Overview. 71 Special Diets 71 Overview. 73 Cost-Effectiveness of Nutrition Interventions. 73 Cost-Effectiveness of Nutrition Interventions. 73 Chapter 10. Cost-Effectiveness of Nutrition Interventions. 74 Overview. 73 Cost-Effectiveness. 73 Outcomes 78 Muthodological Rigor 79 Populations.	Multi-Component Interventions	52
Chapter 8. Effect of Nutrition Interventions on Symptoms. 56 Key Points 56 Overview 56 Nutrition Counseling. 57 Dietary Supplements 59 Special Diets 61 Route or Timing of Nutrition Interventions. 62 Nutrition Support Including Oral Nutrition Supplements 64 Multi-Component Interventions. 67 Risk of Bias and Outcome Assessment 68 Chapter 9. Effect of Nutrition Interventions Intended for Body Weight Loss 71 Overview. 71 Overview. 71 Special Diets 71 Chapter 10. Cost-Effectiveness of Nutrition Interventions. 73 Cost-Effectiveness 73 Cost-Effectiveness 73 Cost-Effectiveness 73 Overview. 77 Overview. 77 Populations. 80 Intervention 81 Outcomes 82 Implementation and Systems Complexity 83 Research Approaches. 83 Strengths and Limitations of the Review 84	Risk of Bias and Outcome Assessment	54
Key Points56Overview56Nutrition Counseling57Dictary Supplements59Special Diets61Route or Timing of Nutrition Interventions62Nutrition Support Including Oral Nutrition Supplements64Multi-Component Interventions67Risk of Bias and Outcome Assessment68Chapter 9. Effect of Nutrition Interventions Intended for Body Weight Loss71Key Points71Overview71Special Diets71Chapter 10. Cost-Effectiveness of Nutrition Interventions73Chapter 11. Discussion77Overview77Broader Context of Available Interventions and Strategies78Future Research78Methodological Rigor79Populations80Intervention81Outcomes82Implementation and Systems Complexity83Research Approaches83Strengths and Limitations of the Review84References86Abbreviations and Acronyms106	Chapter 8. Effect of Nutrition Interventions on Symptoms	56
Overview.56Nutrition Counseling.57Dictary Supplements.59Special Diets61Route or Timing of Nutrition Interventions.62Nutrition Support Including Oral Nutrition Supplements64Multi-Component Interventions.67Risk of Bias and Outcome Assessment68Chapter 9. Effect of Nutrition Interventions Intended for Body Weight Loss71Verview.71Overview.71Special Diets71Chapter 10. Cost-Effectiveness of Nutrition Interventions.73Chapter 11. Discussion77Overview.77Broader Context of Available Interventions and Strategies78Future Research78Methodological Rigor79Populations.80Intervention81Outcomes82Implementation and Systems Complexity83Research Approaches83Strengths and Limitations of the Review.84References86Abbreviations and Acronyms106	Key Points	. 56
Nutrition Counsening.57Dietary Supplements.59Special Diets61Route or Timing of Nutrition Interventions.62Nutrition Support Including Oral Nutrition Supplements64Multi-Component Interventions.67Risk of Bias and Outcome Assessment68Chapter 9. Effect of Nutrition Interventions Intended for Body Weight Loss71Key Points.71Overview.71Special Diets71Chapter 10. Cost-Effectiveness of Nutrition Interventions.73Cost-Effectiveness73Cost-Effectiveness73Cost-Effectiveness74Broader Context of Available Interventions and Strategies78Future Research79Populations.80Intervention81Outcomes82Implementation and Systems Complexity83Research Approaches83Strengths and Limitations of the Review.84References86Abbreviations and Acronyms106	Overview	56
Dictary Supplements59Special Dicts61Route or Timing of Nutrition Interventions62Nutrition Support Including Oral Nutrition Supplements64Multi-Component Interventions67Risk of Bias and Outcome Assessment68Chapter 9. Effect of Nutrition Interventions Intended for Body Weight Loss71Key Points71Overview71Special Dicts71Chapter 10. Cost-Effectiveness of Nutrition Interventions73Cost-Effectiveness73Cost-Effectiveness73Cost-Effectiveness74Broader Context of Available Interventions and Strategies78Future Research78Methodological Rigor79Populations80Intervention81Outcomes82Implementation and Systems Complexity83Research Approaches83Strengths and Limitations of the Review84References86Abbreviations and Acronyms106	Nutrition Counseling	
Special Diets61Route or Timing of Nutrition Interventions62Nutrition Support Including Oral Nutrition Supplements64Multi-Component Interventions67Risk of Bias and Outcome Assessment68Chapter 9. Effect of Nutrition Interventions Intended for Body Weight Loss71Key Points71Overview71Special Diets71Chapter 10. Cost-Effectiveness of Nutrition Interventions73Cost-Effectiveness73Cost-Effectiveness73Chapter 11. Discussion77Overview77Broader Context of Available Interventions and Strategies78Future Research78Methodological Rigor79Populations80Intervention81Outcomes82Implementation and Systems Complexity83Strengths and Limitations of the Review84Conclusions84Abbreviations and Acronyms106	Dietary Supplements	59
Route or Timing of Nutrition Interventions62Nutrition Support Including Oral Nutrition Supplements64Multi-Component Interventions67Risk of Bias and Outcome Assessment68Chapter 9. Effect of Nutrition Interventions Intended for Body Weight Loss71Overview71Overview71Special Diets71Chapter 10. Cost-Effectiveness of Nutrition Interventions73Key Points73Cost-Effectiveness73Cost-Effectiveness73Chapter 11. Discussion77Overview77Broader Context of Available Interventions and Strategies78Future Research78Methodological Rigor79Populations80Intervention81Outcomes82Implementation and Systems Complexity83Strengths and Limitations of the Review84Conclusions84Abbreviations and Acronyms106	Special Diets	. 61
Nutrition Support Including Oral Nutrition Supplements 64 Multi-Component Interventions 67 Risk of Bias and Outcome Assessment 68 Chapter 9. Effect of Nutrition Interventions Intended for Body Weight Loss 71 Key Points 71 Overview 71 Special Diets 73 Cost-Effectiveness of Nutrition Interventions 73 Cost-Effectiveness 73 Cost-Effectiveness 73 Cost-Effectiveness 73 Cost-Effectiveness 73 Cost-Effectiveness 73 Cost-Effectiveness 73 Points 77 Overview 77 Broader Context of Available Interventions and Strategies 78 Future Research 78 Methodological Rigor 79 Populations 80 Intervention 81 Outcomes 82 Implementation and Systems Complexity 83 Strengths and Limitations of the Review 84 Conclusions 84 References 86 Abbreviations and Acronyms<	Route or Timing of Nutrition Interventions	. 62
Multi-Component Interventions. 67 Risk of Bias and Outcome Assessment 68 Chapter 9. Effect of Nutrition Interventions Intended for Body Weight Loss 71 Key Points. 71 Overview. 71 Special Diets 71 Chapter 10. Cost-Effectiveness of Nutrition Interventions. 73 Key Points 73 Cost-Effectiveness 73 Cost-Effectiveness 73 Cost-Effectiveness 73 Cost-Effectiveness 73 Cost-Effectiveness 73 Cost-Effectiveness 73 Powerview. 77 Overview. 77 Broader Context of Available Interventions and Strategies 78 Methodological Rigor 79 Populations. 80 Intervention 81 Outcomes 82 Implementation and Systems Complexity 83 Research Approaches 83 Strengths and Limitations of the Review. 84 Conclusions 84 References 86 Abbreviations and Acronyms 10	Nutrition Support Including Oral Nutrition Supplements	. 64
Risk of Bias and Outcome Assessment68Chapter 9. Effect of Nutrition Interventions Intended for Body Weight Loss71Key Points71Overview71Special Diets71Chapter 10. Cost-Effectiveness of Nutrition Interventions73Key Points73Cost-Effectiveness73Cost-Effectiveness73Cost-Effectiveness73Cost-Effectiveness73Chapter 11. Discussion77Overview77Broader Context of Available Interventions and Strategies78Future Research78Methodological Rigor79Populations80Intervention81Outcomes82Implementation and Systems Complexity83Strengths and Limitations of the Review84References86Abbreviations and Acronyms106	Multi-Component Interventions	. 67
Chapter 9. Effect of Nutrition Interventions Intended for Body Weight Loss 71 Key Points 71 Overview 71 Special Diets 71 Chapter 10. Cost-Effectiveness of Nutrition Interventions 73 Key Points 73 Cost-Effectiveness 71 Poverview 77 Discussion 77 Poverview 77 Broader Context of Available Interventions and Strategies 78 Future Research 79 Populations 80 Intervention 81<	Risk of Bias and Outcome Assessment	68
Key Points71Overview71Special Diets71Chapter 10. Cost-Effectiveness of Nutrition Interventions73Key Points73Cost-Effectiveness73Cost-Effectiveness73Cost-Effectiveness73Cost-Effectiveness73Cost-Effectiveness73Cost-Effectiveness73Cost-Effectiveness73Cost-Effectiveness73Cost-Effectiveness73Cost-Effectiveness73Cost-Effectiveness77Poreview77Broader Context of Available Interventions and Strategies78Future Research78Methodological Rigor79Populations80Intervention81Outcomes82Implementation and Systems Complexity83Strengths and Limitations of the Review84References86Abbreviations and Acronyms106	Chapter 9. Effect of Nutrition Interventions Intended for Body Weight Loss	. 71
Overview.71Special Diets71Chapter 10. Cost-Effectiveness of Nutrition Interventions.73Key Points.73Cost-Effectiveness73Chapter 11. Discussion77Overview.77Broader Context of Available Interventions and Strategies78Future Research78Methodological Rigor79Populations.80Intervention81Outcomes82Implementation and Systems Complexity83Research Approaches83Strengths and Limitations of the Review84References86Abbreviations and Acronyms106	Key Points	. 71
Special Diets 71 Chapter 10. Cost-Effectiveness of Nutrition Interventions 73 Key Points 73 Cost-Effectiveness 73 Chapter 11. Discussion 77 Overview 77 Broader Context of Available Interventions and Strategies 78 Future Research 78 Methodological Rigor 79 Populations 80 Intervention 81 Outcomes 82 Implementation and Systems Complexity 83 Strengths and Limitations of the Review 84 Conclusions 84 References 86 Abbreviations and Acronyms 106	Overview	. 71
Chapter 10. Cost-Effectiveness of Nutrition Interventions73Key Points73Cost-Effectiveness73Chapter 11. Discussion77Overview77Broader Context of Available Interventions and Strategies78Future Research78Methodological Rigor79Populations80Intervention81Outcomes82Implementation and Systems Complexity83Research Approaches83Strengths and Limitations of the Review84References86Abbreviations and Acronyms106	Special Diets	. /1
Key Points73Cost-Effectiveness73Chapter 11. Discussion77Overview77Broader Context of Available Interventions and Strategies78Future Research78Methodological Rigor79Populations80Intervention81Outcomes82Implementation and Systems Complexity83Research Approaches83Strengths and Limitations of the Review84Conclusions84Abbreviations and Acronyms106	Chapter 10. Cost-Effectiveness of Nutrition Interventions	73
Chapter 11. Discussion 77 Overview 77 Broader Context of Available Interventions and Strategies 78 Future Research 78 Methodological Rigor 79 Populations 80 Intervention 81 Outcomes 82 Implementation and Systems Complexity 83 Research Approaches 83 Strengths and Limitations of the Review 84 References 86 Abbreviations and Acronyms 106	Key Points Cost-Effectiveness	. /3
Overview.77Broader Context of Available Interventions and Strategies78Future Research78Methodological Rigor79Populations.80Intervention81Outcomes82Implementation and Systems Complexity83Research Approaches83Strengths and Limitations of the Review.84Conclusions.84References86Abbreviations and Acronyms106	Chapter 11. Discussion	77
Broader Context of Available Interventions and Strategies78Future Research78Methodological Rigor79Populations80Intervention81Outcomes82Implementation and Systems Complexity83Research Approaches83Strengths and Limitations of the Review84Conclusions84References86Abbreviations and Acronyms106	Overview	77
Future Research78Methodological Rigor79Populations80Intervention81Outcomes82Implementation and Systems Complexity83Research Approaches83Strengths and Limitations of the Review84Conclusions84References86Abbreviations and Acronyms106	Broader Context of Available Interventions and Strategies	. 78
Methodological Rigor79Populations80Intervention81Outcomes82Implementation and Systems Complexity83Research Approaches83Strengths and Limitations of the Review84Conclusions84References86Abbreviations and Acronyms106	Future Research	78
Populations80Intervention81Outcomes82Implementation and Systems Complexity83Research Approaches83Strengths and Limitations of the Review84Conclusions84References86Abbreviations and Acronyms106	Methodological Rigor	. 79
Intervention81Outcomes82Implementation and Systems Complexity83Research Approaches83Strengths and Limitations of the Review84Conclusions84References86Abbreviations and Acronyms106	Populations	. 80
Outcomes82Implementation and Systems Complexity83Research Approaches83Strengths and Limitations of the Review84Conclusions84References86Abbreviations and Acronyms106	Intervention	. 81
Implementation and Systems Complexity 83 Research Approaches 83 Strengths and Limitations of the Review 84 Conclusions 84 References 86 Abbreviations and Acronyms 106	Outcomes	. 82
Research Approaches 83 Strengths and Limitations of the Review 84 Conclusions 84 References 86 Abbreviations and Acronyms 106	Implementation and Systems Complexity	. 83
Strengths and Limitations of the Review	Research Approaches	83
Conclusions 84 References 86 Abbreviations and Acronyms 106	Strengths and Limitations of the Review	84
References	Conclusions	. 84
Abbreviations and Acronyms 106	References	. 86
	Abbreviations and Acronyms	106

Tables

Table 1.1. Key terms	2
Table 2.1. Population, intervention, comparator, outcome, timing, and setting (PIC	COTS)5
Table 2.2. Intervention categories and descriptions	
Table 3.1. Identified unique included studies by intervention and Key Question	13

Table 4.1. Identified unique included studies by intervention and cancer type across Key
Questions
Table 4.2. Number of studies evaluating outcomes by Key Question
Table 4.3. Overview of risk of bias assessment
Table 5.1. Studies examining use of nutrition interventions prior to cancer treatment, stratified by
intervention type and cancer type (KQ1)
Table 5.2. Basic characteristics of studies for nutrition interventions prior to cancer treatment:
dietary supplements
Table 5.3. Basic characteristics of studies for nutrition interventions prior to cancer treatment:
nutrition support including oral nutrition supplements
Table 5.4 Risk of bias assessment for nutrition interventions prior to cancer treatment
Table 6.1. Studies for nutrition interventions prior to and including initiating cancer treatment,
stratified by intervention type and cancer type
Table 6.2. Basic characteristics of studies for nutrition interventions prior to and including
initiating cancer treatment: dietary supplements
Table 6.3. Basic characteristics of studies for nutrition interventions prior to and including
initiating cancer treatment: route or timing of nutrition interventions
Table 6.4. Basic characteristics of studies for nutrition interventions prior to and including
initiating cancer treatment: nutrition support including oral nutrition supplements
Table 6.5. Basic characteristics of studies for nutrition interventions prior to and including
initiating cancer treatment: multi-component interventions
Table 6.6 Risk of bias assess for nutrition interventions prior to and including initiating cancer
treatment
Table 7.1 Studies examining use of nutrition interventions after treatment began, stratified by
intervention type and cancer type
Table 7.2. Basic characteristics of studies for nutrition interventions after treatment began:
nutrition counseling
Table 7.3 Basic characteristics of studies for nutrition interventions after treatment began: dietary
supplements41
Table 7.4. Basic characteristics of studies for nutrition interventions after treatment began:
special diets
Table 7.5. Basic characteristics of studies for nutrition interventions after treatment began: route
or timing of nutrition interventions
Table 7.6. Basic characteristics of studies for nutrition interventions after treatment began:
nutrition support including oral nutrition supplements
Table 7.7 Basic characteristics of studies for nutrition interventions after treatment began: multi-
component interventions
Table 7.8 Risk of bias assess for nutrition interventions after treatment began
Table 8.1 Studies examining effect of nutrition interventions on symptoms, stratified by
intervention type and cancer type
Table 8.2. Basic characteristics of studies for effect of nutrition interventions on symptoms:
nutrition counseling
Table 8.3. Basic characteristics of studies for effect of nutrition interventions on symptoms:
dietary supplements
Table 8.4. Basic characteristics of studies for effect of nutrition interventions on symptoms:
special diets

Figures

Figure 2.1	Analytic Frame	work	 	8
Figure 3.1	. Literature flow	diagram	 	12

Appendixes

Appendix A. Methods
Appendix B. Studies Excluded at Full Text
Appendix C. Evidence Tables for Chapter 5
Appendix D. Evidence Tables for Chapter 6
Appendix E. Evidence Tables for Chapter 7
Appendix F. Evidence Tables for Chapter 8
Appendix G. Evidence Tables for Chapter 9
Appendix H. References for Appendixes C-G

Executive Summary

Main Points

- Two decades of randomized trial evidence from 206 studies of nutrition interventions in adults prior to and/or during cancer treatment focused on use of dietary supplements (not including vitamins and minerals), nutrition support (including oral nutrition supplements), and the route or timing of nutrition interventions. Studies were predominately conducted in individuals with gastrointestinal and head and neck cancers, with few studies conducted within the U.S. setting.
- Studies focused on evaluating changes in body weight/composition, adverse events, length of hospital stay, and quality of life.
- Among studies with a high volume of literature (n=114), which predominately examined dietary supplements and nutrition support in gastrointestinal and head and neck cancers, 11 percent (n=12) were rated as low risk of bias (higher quality), 40 percent (n=46) medium risk of bias and 49 percent (n=56) as high risk of bias (lower quality).
- Low- and medium-risk-of-bias studies reported mixed results on the effect of nutrition interventions across outcomes for cancer and cancer treatment (detailed in the evidence summary results below).
- Among studies included in our Key Questions, few (4%, n=8) studies reported a formal evaluation of the value of the nutrition interventions (e.g., cost-effectiveness, costbenefit) or provided costs detailed by intervention component; generally, these studies reported only overall costs from inpatient non-U.S. settings. In our grey literature search, we found few studies that conducted cost-effectiveness or cost-benefit analyses, and, among those, we found mixed results on the value of nutrition interventions in non-U.S. health systems.
- Future research would benefit from studies focused on priorities and interventions most relevant to specific stakeholders (e.g., oncologists, patients, dietitians, researchers, policymakers). Future studies could then be specifically designed to evaluate the main outcomes of interest relevant for clinical practice.
- Future research would also benefit from a creation of standardized taxonomies for interventions and outcomes as well as more rigorous design and reporting of nutrition interventions.

Background and Purpose

Among adults with cancer, malnutrition is associated with decreased treatment completion, more use of healthcare, and worse survival.¹⁻⁴ Prevalence of malnutrition is high among adults with cancer,⁵⁻⁷ but only 30 to 50 percent of cancer patients at risk for malnutrition receive nutrition support or intervention.^{8, 9} No high-quality guidelines exist with recommendations for preventing or treating malnutrition in adults with cancer, potentially due to several factors, including: (1) the broad range of criteria that define malnutrition, (2) variability of nutrition interventions (from medical nutrition therapy to optimize body weight or muscle mass to dietary supplements intended to replace specific micronutrient deficiencies), and (3) the lack of cohesive evidence-based approaches to malnutrition in this population.

This systematic review sought to examine the current evidence for the effectiveness of providing nutrition interventions before or during cancer therapy to improve outcomes for cancer

and cancer treatment, with the goal of informing stakeholders about relevant research gaps and challenges. Findings from the review informed discussions among experts and stakeholders at the National Institutes of Health (NIH) Pathways to Prevention (P2P) workshop, *Nutrition as Prevention for Improved Cancer Health Outcomes*, which took place July 26-28, 2022. In addition, our review aims to contribute to the development of a research agenda for evaluating nutrition interventions in inpatient and outpatient cancer care. Results may help inform clinical guidelines on prevention and treatment of malnutrition in cancer care by providing a summary and synthesis of the available evidence for clinical and policy stakeholders to use in the development of such guidelines.

Methods

The methods for this systematic review follow the Agency for Healthcare Research and Quality Methods Guide for Effectiveness and Comparative Effectiveness Reviews and the PRISMA reporting guidelines.¹⁰ See the review protocol (https://effectivehealthcare.ahrq.gov/ products/improved-cancer -outcomes/protocol) and the methods appendix for additional details. The review was guided by a set of Key Questions, which were established by the NIH planning committee for the *Nutrition as Prevention for Improved Cancer Health Outcomes* workshop. Briefly, we searched Ovid Medline®, Ovid Embase®, and the Cochrane Central Register of Controlled Trials to identify randomized controlled trials published and indexed in bibliographic databases from 2000 through July 2022. We also conducted grey literature searches to identify additional resources relevant to cost-effectiveness.

Results

We identified 9798 unique references, with 206 studies from 219 publications reporting findings from randomized controlled trials (RCTs) of nutrition interventions to improve negative outcomes from cancer treatment. Overall, we found two decades of randomized trial evidence on nutrition interventions for adults prior to and/or during cancer treatment. This evidence focused on use of dietary supplements, nutrition support (including oral nutrition supplements), and the route or timing of nutrition interventions. Studies were predominately conducted in populations with gastrointestinal and head and neck cancers, and included both inpatient surgical and outpatient settings. Most of the studies were conducted outside of the United States. Studies focused on evaluating changes in body weight/composition, adverse events, length of hospital stay, and quality of life. Few studies were conducted within the U.S. setting. Among studies with a high volume of literature, which predominately included studies in dietary supplements and nutrition support in gastrointestinal and head and neck cancers, 11 percent (n=12) were rated as low risk of bias (higher quality), 40 percent (n=46) medium risk of bias and 49 percent (n=56) high risk of bias (lower quality).

Low- and medium-risk-of-bias studies, reported mixed results on the effect of nutrition interventions outcomes for cancer and cancer treatment. Among eight low- or medium-risk-of-bias studies in nutrition support prior to cancer treatment, studies reported mixed results on development of complications, improvements in weight loss, and length of hospital stay.¹¹⁻¹⁸

Among studies initiated prior to and continuing through cancer treatment, four medium-riskof-bias studies across five publications showed mixed results for the effect of dietary supplements on weight changes, readmissions, length of hospital stay, development of complications, and survival.¹⁹⁻²³ Two low- and four medium-risk-of-bias study of nutrition support reported mixed results, with some studies reporting a benefit and some reporting no difference in reducing adverse events, readmissions, length of hospital stay, and survival.²⁴⁻³⁰

Among four low- and 14 medium-risk-of-bias studies of dietary supplements conducted during cancer treatment, results were mixed, with most studies reporting no benefit of added dietary supplements on body weight, adverse events, length of hospital stay, or survival.³¹⁻⁴⁵ Two low-risk-of-bias studies reported fewer adverse events and decreased length of hospital stay with soybean and fish oil.^{34, 45} One medium-risk-of-bias study of glutamine reported improved body weight, reduction in adverse events, and improved treatment tolerance,³⁸ while another medium-risk-of-bias study reported fewer adverse events among those using branch chain amino acid (BCAA)-enriched total parenteral nutrition (TPN).⁴⁰ Another medium-risk-of-bias study found improvement in postoperative complications with enteral and parenteral nutrition supplemented with omega-3 fatty acids.⁴⁴ Three studies reported improvements in length of hospital stay across diverse supplements.^{32, 43, 46}

One low- and 10 medium-risk-of-bias studies that examined route and timing of nutrition interventions conducted during cancer treatment demonstrated mixed results.⁴⁷⁻⁵⁷ The majority reported no difference for body weight, adverse events, readmissions, or death, but half reported reduced length of hospital stay. One low-risk-of-bias study reported that postoperative enteral nutrition reduced adverse events length of hospital stay.⁴⁸ Among three low- and eight medium-risk-of-bias studies that examined nutrition support (including oral nutrition) during cancer treatment,⁵⁸⁻⁷² results were mixed for body weight, nutrition status, and adverse events. Two low-^{66, 71} and three medium-risk-of-bias studies^{67, 69, 72} reported improvements in body weight or composition with postoperative nutrition support. Four out of ten studies^{58, 59, 65, 70} reported improvements in adverse events and three reported improvements in length of hospital stay^{58, 59, 70} across diverse enteral and oral nutrition support interventions.

Finally, among studies of the effects of nutrition interventions on symptoms, three mediumrisk-of-bias studies in dietary supplements reported mixed results for patient-reported symptoms.^{33 23, 43} One study of probiotics and omega-3 fatty acids reported improved quality of life,³³ while another reported no benefit.²³ Two studies reported mixed outcomes for patientreported symptoms with one reporting a benefit³³ and one no difference.⁴³ Among five low- and four medium-risk-of-bias study of nutrition support, reported results were mixed.^{13, 15, 16, 18, 28, 66, ^{68, 69, 71, 72} Two studies showed mixed results on functional status, with one showing a benefit and one no difference.^{28, 66} Two low-risk-of-bias studies reported improvement in nausea for individuals receiving preoperative oral carbohydrate drinks.^{28, 66} A third low-risk-of-bias study reported improvement in treatment tolerance and symptoms after use of oral nutrition supplements and dietary advice.^{66, 68}}

While studies addressing Key Questions (KQs) 1-3 enrolled individuals from multiple cancer types, treatments, and stages (KQ 1-3a) across the lifespan (KQ 1-3b), with varying degrees of muscle wasting (KQ 1-3c), and in those with a range of comorbid conditions (KQ 1-3d), no eligible studies specifically evaluated whether the effects of nutrition interventions on preventing negative outcomes varied across these characteristics.

Only four studies reported the effects of nutrition interventions intended for weight loss using special diets among individuals with breast cancer and assessed body weight and composition changes. One study reported on each of the following symptoms: quality of life, symptoms, and treatment tolerance.

Among studies included in our KQ we found that few (4 percent, n=8) published cost or value (e.g., cost-effectiveness, cost-benefit) information related to the intervention. In our grey

literature search of additional studies examining cost or value of nutrition interventions, we found few studies that conducted cost-effectiveness or cost-benefit analyses, and, among those, found mixed results on the value of nutrition interventions; most of these studies were conducted in non-U.S. health systems.

Limitations

The methods we used for this systematic review provided a detailed evidence map of the current state of literature on nutrition interventions, highlighting not only concentrations of literature but also gaps in intervention types. We purposefully chose broad definitions of nutrition interventions, thereby increasing the scope, breadth, and heterogeneity of the included literature in order to better assess the range and depth of available evidence. This decision allowed for demonstration of the diffuse literature set on the topic and highlighted the predominantly low quality of studies where there were concentrations of similar intervention types. However, this required focusing on high-level directionality of intervention effects across a broader range of nutrition interventions rather than looking for more detailed, precise estimates of intervention effects. Overall, this approach allowed for high level mapping of the evidence across Key Questions by patient, intervention, comparator, and outcome categories. It also revealed evidence gaps for future research.

Implications and Conclusions

Overall, the RCT evidence focused on a wide range of nutrition interventions, but studies were concentrated in use of dietary supplements (not including vitamins and minerals), nutrition support, and the route or timing of nutrition interventions within gastrointestinal and head and neck cancers in the inpatient setting of hospitals outside of the United States. Among interventions with the highest volume of literature, a majority of studies were evaluated as having high risk of bias. Our findings point to the need for rigorous new U.S.-based research to bolster the evidence base. Specifically, the field needs a more detailed future evaluation of a subset of nutrition interventions contained in this evidence map that focus on priorities most relevant to specific stakeholders (e.g., oncologists, patients, dietitians, researchers, policymakers). Further, studies should be specifically designed to evaluate the main outcomes of interest for clinical practice. Future research would also benefit from creation of standardized taxonomies for interventions and outcomes as well as more rigorous design and reporting of nutrition interventions. As mentioned, heterogeneity of populations, interventions, comparators and outcomes precluded aggregation. Currently, the quality and heterogeneity of the studies limit the ability to translate findings into clinical practice or guidelines. In order to inform development of guidelines, coordinated efforts are required to establish detailed conceptual frameworks for mechanisms of nutrition interventions most relevant to clinical care providers and patients. Such frameworks would help inform priorities for future research as well as guide practice and policy.

References

- Aaldriks AA, Maartense E, Nortier HJ, et al. Prognostic factors for the feasibility of chemotherapy and the Geriatric Prognostic Index (GPI) as risk profile for mortality before chemotherapy in the elderly. Acta Oncol. 2016 Jan;55(1):15-23. doi: 10.3109/0284186x.2015.1068446. PMID: 26271800.
- van Deudekom FJ, van der Velden LA, Zijl WH, et al. Geriatric assessment and 1-year mortality in older patients with cancer in the head and neck region: A cohort study. Head Neck. 2019 Aug;41(8):2477-83. doi: 10.1002/hed.25714. PMID: 30816619.
- Aparicio T, Bouché O, Francois E, et al. Geriatric analysis from PRODIGE 20 randomized phase II trial evaluating bevacizumab + chemotherapy versus chemotherapy alone in older patients with untreated metastatic colorectal cancer. Eur J Cancer. 2018 Jul;97:16-24. doi: 10.1016/j.ejca.2018.03.030. PMID: 29777975.
- Guner A, Kim SY, Yu JE, et al. Parameters for Predicting Surgical Outcomes for Gastric Cancer Patients: Simple Is Better Than Complex. Ann Surg Oncol. 2018 Oct;25(11):3239-47. doi: 10.1245/s10434-018-6684-2. PMID: 30069658.
- Ryan AM, Power DG, Daly L, et al. Cancerassociated malnutrition, cachexia and sarcopenia: the skeleton in the hospital closet 40 years later. Proc Nutr Soc. 2016 May;75(2):199-211. doi: 10.1017/s002966511500419x. PMID: 26786393.
- 6. Caillet P, Liuu E, Raynaud Simon A, et al. Association between cachexia, chemotherapy and outcomes in older cancer patients: A systematic review. Clin Nutr. 2017 Dec;36(6):1473-82. doi: 10.1016/j.clnu.2016.12.003. PMID: 28017447.
- Muscaritoli M, Lucia S, Farcomeni A, et al. Prevalence of malnutrition in patients at first medical oncology visit: the PreMiO study. Oncotarget. 2017 Oct 3;8(45):79884-96. doi: 10.18632/oncotarget.20168. PMID: 29108370.

- Planas M, Álvarez-Hernández J, León-Sanz M, et al. Prevalence of hospital malnutrition in cancer patients: a sub-analysis of the PREDyCES® study. Support Care Cancer. 2016 Jan;24(1):429-35. doi: 10.1007/s00520-015-2813-7. PMID: 26099900.
- 9. Hébuterne X, Lemarié E, Michallet M, et al. Prevalence of malnutrition and current use of nutrition support in patients with cancer. JPEN J Parenter Enteral Nutr. 2014 Feb;38(2):196-204. doi: 10.1177/0148607113502674. PMID: 24748626.
- Moher D, Liberati A, Tetzlaff J, et al. Preferred reporting items for systematic reviews and meta-analyses: the PRISMA statement. PLoS Med. 2009 Jul 21;6(7):e1000097. doi: 10.1371/journal.pmed.1000097. PMID: 19621072.
- Burden ST, Gibson DJ, Lal S, et al. Pre-operative oral nutritional supplementation with dietary advice versus dietary advice alone in weight-losing patients with colorectal cancer: single-blind randomized controlled trial. J Cachexia Sarcopenia Muscle. 2017 Jun;8(3):437-46. doi: 10.1002/jcsm.12170. PMID: 28052576.
- Burden ST, Hill J, Shaffer JL, et al. An unblinded randomised controlled trial of preoperative oral supplements in colorectal cancer patients. J Hum Nutr Diet. 2011 Oct;24(5):441-8. doi: 10.1111/j.1365-277X.2011.01188.x. PMID: 21699587.
- Rizvanovic N, Nesek Adam V, Causevic S, et al. A randomised controlled study of preoperative oral carbohydrate loading versus fasting in patients undergoing colorectal surgery. Int J Colorectal Dis. 2019 Sep;34(9):1551-61. doi: 10.1007/s00384-019-03349-4. PMID: 31309323.
- 14. Xu J, Zhong Y, Jing D, et al. Preoperative enteral immunonutrition improves postoperative outcome in patients with gastrointestinal cancer. World J Surg. 2006 Jul;30(7):1284-9. doi: 10.1007/s00268-005-0756-8. PMID: 16830214.

- Chen X, Li K, Yang K, et al. Effects of preoperative oral single-dose and doubledose carbohydrates on insulin resistance in patients undergoing gastrectomy:a prospective randomized controlled trial. Clin Nutr. 2021 Apr;40(4):1596-603. doi: 10.1016/j.clnu.2021.03.002. PMID: 33752148.
- 16. He FJ, Wang MJ, Yang K, et al. Effects of Preoperative Oral Nutritional Supplements on Improving Postoperative Early Enteral Feeding Intolerance and Short-Term Prognosis for Gastric Cancer: A Prospective, Single-Center, Single-Blind, Randomized Controlled Trial. Nutrients. 2022 Apr 1;14(7):01. doi: 10.3390/nu14071472. PMID: 35406085.
- 17. Lee SY, Lee J, Park HM, et al. Impact of Preoperative Immunonutrition on the Outcomes of Colon Cancer Surgery: Results from a Randomized Controlled Trial. Ann Surg. 2021 Aug 4;04:04. doi: 10.1097/SLA.000000000005140. PMID: 34353994.
- Tesar M, Kozusnikova V, Martinek L, et al. Preoperative nutritional support for patients undergoing elective colorectal cancer surgery - does it really work? Biomed Pap Med Fac Univ Palacky Olomouc Czech Repub. 2022 Mar 1;01:01. doi: 10.5507/bp.2022.009. PMID: 35258042.
- Jo S, Choi SH, Heo JS, et al. Missing effect of glutamine supplementation on the surgical outcome after pancreaticoduodenectomy for periampullary tumors: a prospective, randomized, double-blind, controlled clinical trial. World J Surg. 2006 Nov;30(11):1974-82; discussion 83-4. doi: 10.1007/s00268-005-0678-5. PMID: 16927064.
- Sorensen LS, Rasmussen SL, Calder PC, et al. Long-term outcomes after perioperative treatment with omega-3 fatty acid supplements in colorectal cancer. BJS Open. 2020 Aug;4(4):678-84. doi: 10.1002/bjs5.50295. PMID: 32391656.
- Sorensen LS, Thorlacius-Ussing O, Schmidt EB, et al. Randomized clinical trial of perioperative omega-3 fatty acid supplements in elective colorectal cancer surgery. Br J Surg. 2014 Jan;101(2):33-42. doi: 10.1002/bjs.9361. PMID: 24281905.

- 22. Sultan J, Griffin SM, Di Franco F, et al. Randomized clinical trial of omega-3 fatty acid-supplemented enteral nutrition versus standard enteral nutrition in patients undergoing oesophagogastric cancer surgery. Br J Surg. 2012 Mar;99(3):346-55. doi: 10.1002/bjs.7799. PMID: 22237467.
- 23. Serrano PE, Parpia S, Simunovic M, et al. Perioperative optimization with nutritional supplements in patients undergoing gastrointestinal surgery for cancer: A randomized, placebo-controlled feasibility clinical trial. Surgery. 2022 May 20;20:20. doi: 10.1016/j.surg.2022.04.001. PMID: 35606184.
- 24. Lidder P, Thomas S, Fleming S, et al. A randomized placebo controlled trial of preoperative carbohydrate drinks and early postoperative nutritional supplement drinks in colorectal surgery. Colorectal Dis. 2013 Jun;15(6):737-45. doi: 10.1111/codi.12130. PMID: 23406311.
- Bozzetti F, Gavazzi C, Miceli R, et al. Perioperative total parenteral nutrition in malnourished, gastrointestinal cancer patients: a randomized, clinical trial. JPEN J Parenter Enteral Nutr. 2000 Jan-Feb;24(1):7-14. doi: 10.1177/014860710002400107. PMID: 10638466.
- Chen H, Pan D, Li L. The effects of multi-oil fat emulsion on older patients with gastric cancer. Biomedical Research (India). 2017;28(1):4270-6. PMID: 616782903.
- 27. Moya P, Miranda E, Soriano-Irigaray L, et al. Perioperative immunonutrition in normonourished patients undergoing laparoscopic colorectal resection. Surg Endosc. 2016 Nov;30(11):4946-53. doi: 10.1007/s00464-016-4836-7. PMID: 26936601.
- 28. Feng J, Xu R, Li K, et al. Effects of preoperative oral carbohydrate administration combined with postoperative early oral intake in elderly patients undergoing hepatectomy with acute-phase inflammation and subjective symptom burden: A prospective randomized controlled study. Asian J Surg. 2022 Jan;45(1):386-95. doi: 10.1016/j.asjsur.2021.06.042. PMID: 34362624.

- 29. Sanchez-Guillen L, Soriano-Irigaray L, Lopez-Rodriguez-Arias F, et al. Effect of Early Peripheral Parenteral Nutrition Support in an Enhanced Recovery Program for Colorectal Cancer Surgery: A Randomized Open Trial. J Clin Med. 2021 Aug 18;10(16):18. doi: 10.3390/jcm10163647. PMID: 34441942.
- Lopez-Rodriguez-Arias F, Sanchez-Guillen L, Lillo-Garcia C, et al. Assessment of Body Composition as an Indicator of Early Peripheral Parenteral Nutrition Therapy in Patients Undergoing Colorectal Cancer Surgery in an Enhanced Recovery Program. Nutrients. 2021 Sep 18;13(9):18. doi: 10.3390/nu13093245. PMID: 34579122.
- 31. De Luis DA, Izaola O, Cuellar L, et al. A randomized double-blind clinical trial with two different doses of arginine enhanced enteral nutrition in postsurgical cancer patients. Eur Rev Med Pharmacol Sci. 2010 Nov;14(11):941-5. PMID: 21284343.
- 32. Farreras N, Artigas V, Cardona D, et al. Effect of early postoperative enteral immunonutrition on wound healing in patients undergoing surgery for gastric cancer. Clin Nutr. 2005 Feb;24(1):55-65. doi: 10.1016/j.clnu.2004.07.002. PMID: 15681102.
- 33. Golkhalkhali B, Rajandram R, Paliany AS, et al. Strain-specific probiotic (microbial cell preparation) and omega-3 fatty acid in modulating quality of life and inflammatory markers in colorectal cancer patients: a randomized controlled trial. Asia Pac J Clin Oncol. 2018 Jun;14(3):179-91. doi: 10.1111/ajco.12758. PMID: 28857425.
- 34. Jiang ZM, Wilmore DW, Wang XR, et al. Randomized clinical trial of intravenous soybean oil alone versus soybean oil plus fish oil emulsion after gastrointestinal cancer surgery. Br J Surg. 2010 Jun;97(6):804-9. doi: 10.1002/bjs.6999. PMID: 20473991.
- 35. Lobo DN, Williams RN, Welch NT, et al. Early postoperative jejunostomy feeding with an immune modulating diet in patients undergoing resectional surgery for upper gastrointestinal cancer: a prospective, randomized, controlled, double-blind study. Clin Nutr. 2006 Oct;25(5):716-26. doi: 10.1016/j.clnu.2006.04.007. PMID: 16777271.

- 36. Lu CY, Shih YL, Sun LC, et al. The inflammatory modulation effect of glutamine-enriched total parenteral nutrition in postoperative gastrointestinal cancer patients. Am Surg. 2011 Jan;77(1):59-64. PMID: 21396307.
- 37. Miyata H, Yano M, Yasuda T, et al. Randomized study of the clinical effects of omega-3 fatty acid-containing enteral nutrition support during neoadjuvant chemotherapy on chemotherapy-related toxicity in patients with esophageal cancer. Nutrition. 2017 Jan;33:204-10. doi: 10.1016/j.nut.2016.07.004. PMID: 27644137.
- 38. Pathak S, Soni TP, Sharma LM, et al. A Randomized Controlled Trial to Evaluate the Role and Efficacy of Oral Glutamine in the Treatment of Chemo-radiotherapyinduced Oral Mucositis and Dysphagia in Patients with Oropharynx and Larynx Carcinoma. Cureus. 2019 Jun 7;11(6):e4855. doi: 10.7759/cureus.4855. PMID: 31410338.
- Pottel L, Lycke M, Boterberg T, et al. Echium oil is not protective against weight loss in head and neck cancer patients undergoing curative radio(chemo)therapy: a randomised-controlled trial. BMC Complement Altern Med. 2014 Oct 7;14:382. doi: 10.1186/1472-6882-14-382. PMID: 25293388.
- 40. Sun LC, Shih YL, Lu CY, et al. Randomized, controlled study of branched chain amino acid-enriched total parenteral nutrition in malnourished patients with gastrointestinal cancer undergoing surgery. Am Surg. 2008 Mar;74(3):237-42. PMID: 18376691.
- 41. Takeshita S, Ichikawa T, Nakao K, et al. A snack enriched with oral branched-chain amino acids prevents a fall in albumin in patients with liver cirrhosis undergoing chemoembolization for hepatocellular carcinoma. Nutr Res. 2009 Feb;29(2):89-93. doi: 10.1016/j.nutres.2008.12.005. PMID: 19285598.
- 42. Wang WP, Yan XL, Ni YF, et al. Effects of lipid emulsions in parenteral nutrition of esophageal cancer surgical patients receiving enteral nutrition: a comparative analysis. Nutrients. 2013 Dec 27;6(1):111-23. doi: 10.3390/nu6010111. PMID: 24379010.

- Wang X, Pan L, Zhang P, et al. Enteral nutrition improves clinical outcome and shortens hospital stay after cancer surgery. J Invest Surg. 2010 Dec;23(6):309-13. doi: 10.3109/08941939.2010.519428. PMID: 21208095.
- 44. Yang J, Zhang X, Li K, et al. Effects of EN combined with PN enriched with n-3 polyunsaturated fatty acids on immune related indicators and early rehabilitation of patients with gastric cancer: A randomized controlled trial. Clin Nutr. 2022 Apr 6;41(6):1163-70. doi: 10.1016/j.clnu.2022.03.018. PMID: 35500316.
- 45. Zhu MW, Tang DN, Hou J, et al. Impact of fish oil enriched total parenteral nutrition on elderly patients after colorectal cancer surgery. Chin Med J (Engl). 2012 Jan;125(2):178-81. doi: 10.3760/cma.j.issn.0366-6999.2012.02.003. PMID: 22340541.
- 46. De Luis DA, Izaola O, Cuellar L, et al. High dose of arginine enhanced enteral nutrition in postsurgical head and neck cancer patients. A randomized clinical trial. Eur Rev Med Pharmacol Sci. 2009 Jul-Aug;13(4):279-83. PMID: 19694342.
- 47. Berkelmans GHK, Fransen LFC, Dolmans-Zwartjes ACP, et al. Direct Oral Feeding Following Minimally Invasive Esophagectomy (NUTRIENT II trial): An International, Multicenter, Open-label Randomized Controlled Trial. Ann Surg. 2020 Jan;271(1):41-7. doi: 10.1097/SLA.000000000003278. PMID: 31090563.
- Bozzetti F, Braga M, Gianotti L, et al. Postoperative enteral versus parenteral nutrition in malnourished patients with gastrointestinal cancer: a randomised multicentre trial. The Lancet. 2001;358(9292):1487-92.
- 49. Braga M, Gianotti L, Gentilini O, et al. Early postoperative enteral nutrition improves gut oxygenation and reduces costs compared with total parenteral nutrition. Critical care medicine. 2001;29(2):242-8.

- Feo CV, Romanini B, Sortini D, et al. Early oral feeding after colorectal resection: a randomized controlled study. ANZ J Surg. 2004 May;74(5):298-301. doi: 10.1111/j.1445-1433.2004.02985.x. PMID: 15144242.
- Kurbanalievich SD, Vladimirovich DV, Kabildina NA. Nutritional Support for Patients with Diseases of Hepatopancreotoduodenal Zone in the Early After the Operational Period. Open Access Macedonian Journal of Medical Sciences. 2020;8(B):769-74. doi: 10.3889/oamjms.2020.4717. PMID: 2005564409.
- Ma BQ, Chen SY, Jiang ZB, et al. Effect of postoperative early enteral nutrition on clinical outcomes and immune function of cholangiocarcinoma patients with malignant obstructive jaundice. World J Gastroenterol. 2020 Dec 14;26(46):7405-15. doi: 10.3748/wjg.v26.i46.7405. PMID: 33362392.
- 53. Sun HB, Li Y, Liu XB, et al. Early Oral Feeding Following McKeown Minimally Invasive Esophagectomy: An Open-label, Randomized, Controlled, Noninferiority Trial. Ann Surg. 2018 Mar;267(3):435-42. doi: 10.1097/SLA.00000000002304. PMID: 28549015.
- 54. Tao Z, Zhang Y, Zhu S, et al. A Prospective Randomized Trial Comparing Jejunostomy and Nasogastric Feeding in Minimally Invasive McKeown Esophagectomy. J Gastrointest Surg. 2020 Oct;24(10):2187-96. doi: 10.1007/s11605-019-04390-y. PMID: 31512101.
- 55. Wang J, Zhao J, Zhang Y, et al. Early enteral nutrition and total parenteral nutrition on the nutritional status and blood glucose in patients with gastric cancer complicated with diabetes mellitus after radical gastrectomy. Exp Ther Med. 2018 Jul;16(1):321-7. doi: 10.3892/etm.2018.6168. PMID: 29896256.
- 56. Wang Q, Yang KL, Guo BY, et al. Safety of early oral feeding after total laparoscopic radical gastrectomy for gastric cancer (SOFTLY-1): a single-center randomized controlled trial. Cancer Manag Res. 2019;11:4839-46. doi: 10.2147/CMAR.S199552. PMID: 31239762.

- 57. Xiao-Bo Y, Qiang L, Xiong Q, et al. Efficacy of early postoperative enteral nutrition in supporting patients after esophagectomy. Minerva Chir. 2014 Feb;69(1):37-46. PMID: 24504222.
- Barlow R, Price P, Reid TD, et al. Prospective multicentre randomised controlled trial of early enteral nutrition for patients undergoing major upper gastrointestinal surgical resection. Clin Nutr. 2011 Oct;30(5):560-6. doi: 10.1016/j.clnu.2011.02.006. PMID: 21601319.
- 59. Chu L, Ren Y, Zhang L, et al. Evaluation of effects of nutritional risk assessment and enteral and parenteral nutritional interventions after esophageal cancer surgery. International journal of clinical and experimental medicine. 2018;11(5):5110-6. PMID: 622367981.
- Klek S, Kulig J, Sierzega M, et al. Standard and immunomodulating enteral nutrition in patients after extended gastrointestinal surgery--a prospective, randomized, controlled clinical trial. Clin Nutr. 2008 Aug;27(4):504-12. doi: 10.1016/j.clnu.2008.04.010. PMID: 18571296.
- 61. Klek S, Scislo L, Walewska E, et al. Enriched enteral nutrition may improve short-term survival in stage IV gastric cancer patients: A randomized, controlled trial. Nutrition. 2017 Apr;36:46-53. doi: 10.1016/j.nut.2016.03.016. PMID: 28336107.
- Klek S, Sierzega M, Szybinski P, et al. Perioperative nutrition in malnourished surgical cancer patients - a prospective, randomized, controlled clinical trial. Clin Nutr. 2011 Dec;30(6):708-13. doi: 10.1016/j.clnu.2011.07.007. PMID: 21820770.
- 63. Klek S, Sierzega M, Szybinski P, et al. The immunomodulating enteral nutrition in malnourished surgical patients - a prospective, randomized, double-blind clinical trial. Clin Nutr. 2011 Jun;30(3):282-8. doi: 10.1016/j.clnu.2010.10.001. PMID: 21074910.

- 64. Klek S, Szybinski P, Szczepanek K. Perioperative immunonutrition in surgical cancer patients: a summary of a decade of research. World J Surg. 2014 Apr;38(4):803-12. doi: 10.1007/s00268-013-2323-z. PMID: 24178185.
- 65. Li C, Ni L, Liu C. Early enteral immunonutrition support protects the cellular and humoral immune functions of patients with pancreatic cancer after chemotherapy. International journal of clinical and experimental medicine. 2020;13(2):700-8. PMID: 2003914110.
- 66. Meng Q, Tan S, Jiang Y, et al. Post-discharge oral nutritional supplements with dietary advice in patients at nutritional risk after surgery for gastric cancer: A randomized clinical trial. Clin Nutr. 2021 Jan;40(1):40-6. doi: 10.1016/j.clnu.2020.04.043. PMID: 32563598.
- Miyazaki Y, Omori T, Fujitani K, et al. Oral nutritional supplements versus a regular diet alone for body weight loss after gastrectomy: a phase 3, multicenter, openlabel randomized controlled trial. Gastric Cancer. 2021 Sep;24(5):1150-9. doi: 10.1007/s10120-021-01188-3. PMID: 33835329.
- Tan S, Meng Q, Jiang Y, et al. Impact of oral nutritional supplements in post-discharge patients at nutritional risk following colorectal cancer surgery: A randomised clinical trial. Clin Nutr. 2021 Jan;40(1):47-53. doi: 10.1016/j.clnu.2020.05.038. PMID: 32563599.
- Wu W, Zhong M, Zhu DM, et al. Effect of Early Full-Calorie Nutrition Support Following Esophagectomy: A Randomized Controlled Trial. JPEN J Parenter Enteral Nutr. 2017 Sep;41(7):1146-54. doi: 10.1177/0148607116651509. PMID: 27208039.
- 70. Yao R, Zhang T, Zhang J, et al. Effects of postoperative enteral nutrition combined with adjuvant radiotherapy on inflammatory response, nutrition, healing and prognosis in patients receiving radical surgery for esophageal carcinoma. J BUON. 2019 Jul-Aug;24(4):1673-8. PMID: 31646824.

- 71. Zhu MW, Yang X, Xiu DR, et al. Effect of oral nutritional supplementation on the postdischarge nutritional status and quality of life of gastrointestinal cancer patients after surgery: a multi-center study. Asia Pac J Clin Nutr. 2019;28(3):450-6. doi: 10.6133/apjcn.201909_28(3).0004. PMID: 31464391.
- 72. Xie H, Chen X, Xu L, et al. A randomized controlled trial of oral nutritional supplementation versus standard diet following McKeown minimally invasive esophagectomy in patients with esophageal malignancy: a pilot study. Ann Transl Med. 2021 Nov;9(22):1674. doi: 10.21037/atm-21-5422. PMID: 34988183.

Chapter 1. Introduction

Background and Objective for Systematic Review

Among adults with cancer, malnutrition is associated with decreased treatment completion, greater healthcare use, and worse survival.¹⁻⁴ Cancer-related malnutrition⁵ results from inadequate nutrition intake due to the systemic effects of the disease, psychological effects or adverse effects of treatment, and other factors.⁶ Malnutrition can result in micronutrient deficiencies, deplete body fat and/or lean mass, create alterations in metabolism, and lead to reduced physical function and poor health outcomes (both cancer-related and other).⁷ Adults with cancer commonly experience malnutrition, with estimates ranging between 25 to 80 percent across patient populations.⁸⁻¹⁰ However, malnutrition varies substantially by patient characteristics such as age at diagnosis, tumor type, stage of disease, type of cancer treatment, and pre-existing conditions (e.g., diabetes), among other factors.^{9, 11} Further, many factors may increase risk or severity of malnutrition including cancer symptoms (e.g., anorexia, early satiety, and fatigue), treatment complications (e.g., mucositis, nausea, taste changes), and psychological distress.¹² In individuals with cancer, malnutrition often goes unrecognized —not only by providers during clinical assessment, but also by patients and caregivers.¹³ Even when it is recognized, malnutrition may not be adequately addressed. Only 30 to 50 percent of cancer patients at risk for malnutrition receive nutrition intervention.^{14, 15} Considering that an estimated 1.9 million individuals were diagnosed with cancer in 2021, between 570,000 and 950,000 individuals may be at risk for malnutrition.¹⁶

Both the American Society for Parenteral and Enteral Nutrition and the European Society for Clinical Nutrition and Metabolism recommend initial malnutrition screening and subsequent periodic reassessment during the course of cancer treatment and survivorship.^{7, 12} However, no guidelines based on comprehensive, high-quality evidence exist to screen or treat malnutrition in adults with cancer. Guideline development may be challenged by the broad range of criteria defining malnutrition, variability of nutrition interventions (from medical nutrition therapy to optimize body weight or muscle mass to dietary supplements intended to replace specific micronutrient deficiencies) and by the lack of cohesive evidence-based approaches to identify and address malnutrition in this population. This leaves cancer patients and their providers with decisional dilemmas. Furthermore, we do not know how treatment benefits and harms may be affected by patient characteristics (e.g., age, race/ethnicity, family/other support, social economic status including food security, pre-diagnosis obesity or underweight), cancer-related factors (e.g., cancer type, stage), treatment type (chemotherapy, radiation, surgery), treatment timing (before or after treatment) and provider/hospital/geographic characteristics (e.g., presence of integrated nutrition programs; specialist type, availability, and insurance coverage; and rural/urban location). Considering the poor access to outpatient nutrition care for cancer patients across the United States,¹⁷ understanding the most effective interventions for nutrition in this population is critical.

Medical nutrition therapy seeks to address current cancer issues and minimize side effects related to cancer treatment. Nutrition therapy may encompass a broad range of interventions such as nutrition and behavioral counseling, nutrition support (including oral nutrition supplements) or use of dietary supplements that vary in terms of timing and resources required to implement. Because nutrition therapies vary widely, and because settings differ greatly in their capacity to administer and support these interventions (particularly in outpatient settings), we need to better

understand their effectiveness at improving outcomes of cancer and cancer treatment among adults at risk for cancer-associated malnutrition. More clarity around which interventions or components of interventions work best and in which settings and situations will help patients, caregivers, and providers make more informed decisions and guide future research to close evidence gaps.

Purpose and Scope

The purpose of this systematic review is to examine the current evidence for how nutrition interventions before or during cancer therapy affect outcomes of cancer and cancer treatment, with a focus on research gaps and opportunities. Findings from this review informed discussion of experts and stakeholders at the National Institutes of Health Pathways to Prevention (P2P) workshop, *Nutrition as Prevention for Improved Cancer Health Outcomes*, held July 26-28, 2022 (https://prevention.nih.gov/research-priorities/research-needs-and-gaps/pathways-prevention/nutrition-prevention-improved-cancer-health-outcomes). Because our results describe the current body of scientific evidence, they may also be useful for shaping clinical guidelines on prevention and treatment of malnutrition in cancer care, and provide a summary and synthesis of the available evidence for clinical and policy stakeholders to apply in the development of such guidelines.

Report Organization

Chapter 2 outlines the methods used to conduct this systematic review. Chapter 3 presents the overall results of the search for the review's eligible studies. Chapter 4 summarizes the evidence map across Key Questions (KQs). Chapter 5 presents results from KQ1 that examined nutrition interventions that were only delivered prior to the initiation of cancer treatment. Chapter 6 separately presents studies of interventions that were initiated prior to and continued after cancer treatment (KQs 1 and 2). Chapter 7 presents results from KQ2 that were delivered only after treatment began. Chapters 8 and 9 address KQ3, and KQ4, respectively. Chapter 10 presents results for the Contextual Question. Chapter 11 discusses research gaps and future research considerations. An overview of key terms for the report is provided in Table 1.1. A complete list of acronyms can be found at the end of this report.

Term	Description
Eligible study	An eligible study is one that meets the initial study criteria that were defined in advance regarding the type of study that would be included in the systematic or comparative effectiveness review.
Evidence Map	A systematic search of a broad field to identify gaps in knowledge and/or future research needs. ¹⁸
Nutrition Interventions	Given that there is no agreed-upon classification system for grouping or describing non- pharmacologic nutrition interventions, we classified interventions based on the content and intent of the intervention and the intended audience, using taxonomies and definitions where available. Nutrition intervention categories include: nutrition counseling, dietary supplements (not including vitamins and minerals), special diets, route or timing of nutrition interventions, nutrition support (including oral nutrition supplements), and multi- component interventions. Detailed descriptions of studies falling within each category can be found in Chapter 2.
Pre-treatment nutrition	Pre-treatment nutrition interventions includes any intervention delivered from the date
interventions	of diagnosis through the initiation of cancer-directed therapy.

Table 1.1 Key terms

Term	Description
Nutrition Interventions	Nutrition interventions during cancer therapy include interventions delivered
during cancer therapy	simultaneously (at least in part) with cancer therapy (e.g., systemic therapy, radiation,
-	surgery, endocrine therapy), regardless of treatment intent (e.g., curative vs. palliative).
Cancer Type	Studies were classified by cancer type of the enrolled participants. Cancer type was based on site at diagnosis and grouped into broad categories based on related organ systems and where the literature was concentrated. Categories included gastrointestinal (e.g., stomach, colon, rectal), head and neck (e.g., laryngeal), and multiple cancers (i.e., enrolled patients with multiple distinct cancer types such as breast, lung and prostate cancer combined). The remaining studies enrolled individuals from a broad range of single cancer types (e.g., lymphoma) that did not represent a large volume of literature and were grouped into an 'Other Cancer Type' for analysis.
Risk of bias	Risk of bias is the extent to which the design and conduct of a study are likely to have prevented bias in the results.
At-risk for malnutrition	Given there are a wide variety of approaches to determining an individual's risk for cancer-associated malnutrition, ranging from use of validated screening tools to considering any individual with cancer as at-risk, we used author definitions of 'at-risk for malnutrition' in our reporting. However, we do indicate when specific tools or criteria were used to evaluate malnutrition within the results.

Chapter 2. Methods

Review Approach

The methods for this systematic review followed the Agency for Healthcare Research and Quality (AHRQ) Methods Guide for Effectiveness and Comparative Effectiveness Reviews (available at <u>https://effectivehealthcare.ahrq.gov/topics/cer-methods-guide/overview</u>). This systematic review also reports in accordance with the Preferred Items for Reporting in Systematic Reviews and Meta-Analyses (PRISMA).¹⁹ The final protocol was posted online on October 8, 2021 (<u>https://effectivehealthcare.ahrq.gov/products/improved-cancer-outcomes/protocol</u>). We registered the protocol on PROSPERO (CRD42021282881).

Research (Key) Questions

This review addressed four Key Questions (KQs) to evaluate the effects of nutrition interventions on outcomes of cancer and cancer treatment. Nutrition interventions can be very heterogeneous which may affect the cost and resources needed to deliver the interventions. Therefore, the review also included a Contextual Question to provide information on the cost-effectiveness of nutrition interventions.

Key Questions for Systematic Review

- KQ1: In adults diagnosed with cancer who have or are at risk for cancer-associated malnutrition, what is the effect of nutrition interventions prior to cancer treatment in preventing negative treatment outcomes such as effects on dose tolerance, hospital utilizations, adverse events, and survival?
 - **KQ1a:** Do the effects of nutrition interventions on preventing the negative outcomes associated with cancer treatment vary by cancer type, treatment type (chemotherapy, radiation, surgery), and stage of disease?
 - KQ1b: Do the effects of nutrition interventions vary across the lifespan (e.g., adults aged ≥65 years vs. <65 years)?
 - **KQ1c:** Compared to adults without muscle wasting, do nutrition interventions prevent the negative outcomes associated with cancer treatment in adults with muscle wasting?
 - **KQ1d:** Do the effects of nutrition interventions on preventing the negative outcomes associated with cancer treatment vary across special populations (e.g., individuals with multiple comorbid conditions)?
- KQ2: In adults diagnosed with cancer who have or are at risk for cancer-associated malnutrition, what is the effect of nutrition interventions *during* cancer treatment in preventing negative treatment outcomes such as effects on dose tolerance, hospital utilizations, adverse events, and survival?
 - **KQ2a:** Do the effects of nutrition interventions on preventing the negative outcomes associated with cancer treatment vary by cancer type, treatment type (chemotherapy, radiation, surgery), and stage of disease?

- KQ2b: Do the effects of nutrition interventions vary across the lifespan (e.g., adults aged ≥65 years vs. <65 years)?
- **KQ2c:** Compared to adults without muscle wasting, do nutrition interventions prevent the negative outcomes associated with cancer treatment in adults with muscle wasting?
- **KQ2d:** Do the effects of nutrition interventions on preventing the negative outcomes associated with cancer treatment vary across special populations (e.g., individuals with multiple comorbid conditions)?
- KQ3: In adults diagnosed with cancer who have or are at risk for cancer-associated malnutrition, what is the effect of nutrition interventions *prior to* or *during* cancer treatment on associated symptoms such as fatigue, nausea and vomiting, appetite, physical and functional status (e.g., frailty), and quality of life?
 - **KQ3a**: Do the effects of nutrition interventions on symptoms associated with cancer treatment vary by cancer type, treatment type (chemotherapy, radiation, surgery), and stage of disease?
 - KQ3b: Do the effects of nutrition interventions vary across the lifespan (e.g., adults aged ≥65 years vs. <65 years)?
 - KQ3c: Compared to adults without muscle wasting, do nutrition interventions differentially affect symptoms associated with cancer treatment in adults with muscle wasting?
 - **KQ3d:** Do the effects of nutrition interventions on symptoms associated with cancer treatment vary across special populations (e.g., individuals with multiple comorbid conditions)?
- KQ4: In adults with cancer who are overweight or obese, what is the effect of nutrition interventions intended for weight loss *prior to* or *during* cancer treatment in preventing negative treatment outcomes such as effects on dose, hospital utilizations, adverse events, and survival?

Contextual Question

1. What evidence is available on the cost-effectiveness of nutrition interventions for preventing negative outcomes associated with cancer treatment?

Table 2.1 provides details on the population, interventions, comparators, outcomes, timing, and setting for the research questions.

Element	Nutrition Interventions)	KQ2: Nutrition Interventions During Treatment	KQ3: Pre- or During Treatment Nutrition Interventions and Patient-Centered Outcomes	KQ4: Weight Loss in Overweight/Obese Adults With Cancer
Population	Adults diagnosed with cancer at or after age 18 who have or are at risk for cancer- associated malnutrition	Adults diagnosed with cancer at or after age 18 who have or are at risk for cancer- associated malnutrition	Adults diagnosed with cancer at or after age 18 who have or are at risk for cancer- associated malnutrition	Overweight (BMI 25- <30)/obese (BMI ≥30) adults ≥18y of age diagnosed with cancer
	Subgroups: • Cancer and treatment characteristics (cancer type, treatment type (systemic therapy, radiation, surgery), stage of disease) • Adults ≥65y vs younger • Muscle wasting (e.g., sarcopenia, cachexia, pre- cachexia) vs. no muscle wasting Special populations (individuals with multiple comorbid	Subgroups: • Cancer and treatment characteristics (cancer type, treatment type (systemic therapy, radiation, surgery), stage of disease) • Adults ≥65y vs younger • Muscle wasting (e.g., sarcopenia, cachexia, pre- cachexia) vs. no muscle wasting Special populations (individuals with multiple comorbid	Subgroups: • Cancer and treatment characteristics (cancer type, treatment type (systemic therapy, radiation, surgery), stage of disease) • Adults ≥65y vs younger • Muscle wasting (e.g., sarcopenia, cachexia, pre- cachexia) vs. no muscle wasting Special populations (individuals with multiple comorbid	

Table 2.1. Population, intervention, comparator, outcome, timing, and setting (PICOTS)

Element	KQ1: Pre-Treatment Nutrition Interventions)	KQ2: Nutrition Interventions During Treatment	KQ3: Pre- or During Treatment Nutrition Interventions and Patient-Centered Outcomes	KQ4: Weight Loss in Overweight/Obese Adults With Cancer
Intervention	Nutrition interventions under the supervision of a nutrition professional (e.g., dietitian, nutritionist, or other licensed clinicians) • Diet or nutrition therapy (via oral or enteral (e.g., nasogastric, gastrostomy, jejunostomy) feeding • Special diets (e.g., fasting (intermittent or short-term), calorie restriction, ketogenic, Mediterranean diet, high calorie, high protein) • Supplements • Total parenteral therapy • Nutrition counseling Combined nutrition interventions (e.g., nutrition counseling	Nutrition interventions under the supervision of a nutrition professional (e.g., dietitian, nutritionist, or other licensed clinicians) • Diet or nutrition therapy (via oral or enteral (e.g. nasogastric, gastrostomy, jejunostomy) feeding • Special diets (e.g., fasting (intermittent or short-term), calorie restriction, ketogenic, Mediterranean diet, high calorie, high protein) • Supplements • Total parenteral therapy • Nutrition counseling Combined nutrition interventions (e.g., nutrition counseling	Nutrition interventions under the supervision of a nutrition professional (e.g., dietitian, nutritionist, or other licensed clinicians) • Diet or nutrition therapy (via oral or enteral (e.g. nasogastric, gastrostomy, jejunostomy) feeding • Special diets (e.g., fasting (intermittent or short-term), calorie restriction, ketogenic, Mediterranean diet, high calorie, high protein) • Supplements • Total parenteral therapy • Nutrition counseling Combined nutrition interventions (e.g., nutrition counseling	Nutrition interventions intended for weight loss (includes both PNIs and NIDTs)
Comparators	Standard of care vs. PNIs or PNIs vs. PNIs	Standard of care vs. NIDTs, NIDT vs. NIDT, or PNIs vs. NIDTs	Standard of care vs. PNIs or NIDTs, NIDTs vs. NIDTs, PNIs vs. PNIs, PNIs vs. NIDTs	Standard of care vs. PNIs or NIDTs, NIDTs vs. NIDTs, PNIs vs. PNIs, PNIs vs. NIDTs

Element	KQ1: Pre-Treatment Nutrition Interventions)	KQ2: Nutrition Interventions During Treatment	KQ3: Pre- or During Treatment Nutrition Interventions and Patient-Centered Outcomes	KQ4: Weight Loss in Overweight/Obese Adults With Cancer
Outcomes	Intermediate Outcomes BMI, Body composition, Weight (loss, gain) Final Outcomes Cancer treatment tolerance: treatment interruptions, reductions, or delays Hospital utilizations: ER visits, Admissions, Length of hospital stay Adverse events: • Chemotherapy /radiation therapy limiting toxicity • Post-op complication	Intermediate Outcomes BMI, Body composition, Weight (loss, gain) Final Outcomes Cancer treatment tolerance: treatment interruptions, reductions, or delays Hospital utilizations: ER visits, Admissions, Length of hospital stay Adverse events: • Chemotherapy /radiation therapy limiting toxicity • Post-op complication	Fatigue, nausea and vomiting, appetite, physical/functional status (e.g., frailty) Quality of life	Intermediate Outcomes BMI, body composition, weight (loss, gain) Final Outcomes Cancer treatment tolerance: treatment interruptions, reductions, or delays Hospital utilizations: ER visits Admissions, Length of hospital stay Adverse events: • Chemotherapy/radiation therapy limiting toxicity • Post-op complication
	 NI-related AEs Unintended harms Survival: Nutrition status Malnutrition (underweight, wasting, overweight) 	 NI-related AEs Unintended harms Survival: Nutrition status Malnutrition (underweight, wasting, overweight) 		 NI-related AEs Unintended harms Survival: Nutrition Status Malnutrition (underweight, wasting, overweight)
Timing	Nutrition interventions delivered pre-cancer treatment (KQ1, KQ3, KQ4) and during cancer treatment (KQ2, KQ3, KQ4)	Nutrition interventions delivered pre-cancer treatment (KQ1, KQ3, KQ4) and during cancer treatment (KQ2, KQ3, KQ4)	Nutrition interventions delivered pre-cancer treatment (KQ1, KQ3, KQ4) and during cancer treatment (KQ2, KQ3, KQ4)	Nutrition interventions delivered pre-cancer treatment (KQ1, KQ3, KQ4) and during cancer treatment (KQ2, KQ3, KQ4)
Setting	Outpatient oncology care, ambulatory care, cancer treatment centers, inpatient, home- based, hospice, telemedicine	Outpatient oncology care, ambulatory care, cancer treatment centers, inpatient, home- based, hospice, telemedicine	Outpatient oncology care, ambulatory care, cancer treatment centers, inpatient, home- based, hospice, telemedicine	Outpatient oncology care, ambulatory care, cancer treatment centers, inpatient, home-based, hospice, telemedicine

Abbreviations: AE=adverse event; KQ=Key Question; BMI=body mass index; ER=emergency room; NI=nutrition intervention; NIDT=nutrition intervention during treatment; PICOTS=population, intervention, comparator, outcomes, timing, setting; PNI=pre-treatment nutrition intervention.

Analytic Framework

Figure 2.1 shows a visual representation of the analytic framework for the KQs, illustrating the relationship of interventions and outcomes.

Figure 2.1. Analytic framework



Abbreviations: BMI=body mass index; CQ = Contextual Question; KQ=Key Question; NIDT: Nutrition intervention during treatment: PNI: Pre-treatment nutrition intervention.

Search Strategy and Study Selection

We selected studies based on the population, intervention, comparator, outcome, timing, and setting (PICOTS) framework outlined in Table 2.1 if they were published in English in a peer-reviewed journal. We evaluated any randomized controlled trial that examined nutrition interventions delivered to adults prior to or during cancer treatment for relevance to either the Key Questions or the Contextual Question (CQ). To identify the literature with the highest likelihood of having statistical power to detect an effect from a nutrition intervention, we further limited included studies to those randomizing at least 50 participants (i.e., approximately 25 individuals per arm). Studies that described the cost or value (e.g., cost-effectiveness, cost-benefit) of nutrition interventions were considered eligible for the CQ as well.

The methods for this systematic review followed the AHRQ Methods Guide for Effectiveness and Comparative Effectiveness Reviews (available at

https://effectivehealthcare.ahrq.gov/topics/cer-methods-guide/overview) and the PRISMA¹⁹ and *PRISMA-Searching*²⁰ reporting guidelines. We conducted a comprehensive literature search in July 2022, searching MEDLINE (Ovid), Embase (Ovid), and Cochrane Central Register of Controlled Trials (Wiley). The search included literature published from 2000 through July 2022 to encompass contemporary cancer treatments (e.g., intensity modulated radiation therapy that allowed for better sparing of normal tissues and reduction in toxicity emerged in the late 1990s and rapidly advanced in use in early 2000s). See Appendix A for full details.

We reviewed bibliographic database search results for studies relevant to our PICOTS framework and study-specific criteria. Search results were downloaded to PICO PortalTM,²¹ an online systematic review platform, for screening. Two trained, independent investigators reviewed titles and abstracts for identified studies meeting PICOTS framework and study selection criteria. Two reviewers independently performed full text screening to determine whether studies met inclusion criteria. Differences in screening decisions were resolved by consultation between reviewers, and, if necessary, consultation with a third investigator. All citations deemed appropriate for inclusion through title and abstract review by both reviewers were then examined at full text. We documented inclusion and exclusion status of citations, noting reasons for exclusion (see Appendix B for list of excluded studies). Throughout the screening process, members of the review team met regularly to discuss training material and issues as they arose to ensure that inclusion criteria were consistently applied. For our contextual question, we evaluated all studies included in KQ 1-4 that included discussions of the cost and effectiveness of the intervention. We supplemented these studies with a grey literature search (see Appendix A for search strategy) of systematic reviews of the cost or value of nutrition interventions in oncology and published studies from national nutrition (e.g., American Society for Parenteral and Enteral Nutrition) and oncology groups (e.g., Oncology Nutrition Dietetic Practice Group of the Academy of Nutrition and Dietetics). We defined cost-effectiveness as the cost of intervention relative to another intervention (or status quo) to gain a unit of health outcome such as a life year gained or death prevented.²² Cost-benefit analyses evaluated the net benefits (minus the costs) of an intervention.²³

We conducted additional grey literature searches using the Google search engine to identify relevant completed and ongoing studies, outcomes, and analyses not reported in the published literature and to inform future research needs.

Assessing Methodological Risk of Bias of Individual Studies

Based on AHRQ guidance,²⁴ two independent reviewers assessed risk of bias on a subset of eligible studies. This subset of studies was those that had a relatively large number of studies within an intervention/cancer type. Our threshold of 10 studies within a specific intervention/cancer type resulted in dietary supplements and nutrition support for gastrointestinal cancers being a frequent category for risk of bias assessment. (See Data Synthesis section below for descriptions of intervention categories.) We then extended risk of bias to dietary supplements and nutrition support for gastrointestinal cancers across all KQs to ensure the majority of studies of nutrition interventions in the most common cancer type were assessed for risk of bias (See Table 4.1 for an overview of studies where risk of bias was assessed). Any discrepancies in overall risk of bias for each study as low, moderate, or high, based on the collective risk of bias inherent in each domain, and confidence that the results are believable given the study's limitations. Types of potential bias we evaluated for each eligible study included:

- Selection bias: adequacy of randomization method
- <u>Attrition bias:</u> loss to follow up, both overall and differentially between treatment groups
- Detection bias: outcome assessor masking, outcome measurement quality
- <u>Performance bias:</u> intention to treat analysis, adjustment for potential confounding variables, participant masking to treatment assignment
- <u>Reporting bias:</u> selective reporting of outcomes

Data Abstraction and Data Management

Studies meeting inclusion criteria were distributed among investigators for data extraction. For all study designs, these data fields included author, year of publication, PubMed Identification Number, study design, population (including patient characteristics of interest noted in Table 2.1), intervention(s), study follow up, and setting. As outlined above, we assessed a subset of eligible studies addressing KQ 1–4 for risk of bias. For studies with low- or mediumrisk-of-bias, we abstracted information on intervention duration, comparisons, outcomes, and funder. We provide a detailed evidence map/summary of studies deemed eligible but judged high risk of bias in Appendixes C–G. References for Appendixes C–G are provided in Appendix H.

Data Synthesis

Due to the significant heterogeneity of intervention types, comparators, outcomes, and timing evaluated within KQs, meta-analysis was generally not feasible or appropriate. Further, the number of eligible studies was much larger than anticipated yet distributed across a wide range of interventions and cancer types. We therefore determined that an evidence map synthesis would be most feasible and valuable. We organized the results by KQ, then broadly by type of nutrition intervention and type of cancer. As the mechanisms for the effect of nutrition interventions for preventing and treating the negative outcomes of cancer and cancer treatment may vary by cancer stage and the treatment itself, we also report these characteristics as well. As outlined above in the Report Organization, some interventions were not exclusively delivered prior to cancer treatment (KQ1, Chapter 5) or after cancer treatment began (KQ2, Chapter 7).

Therefore, chapter 6 separately presents studies of interventions that were initiated prior to and continued after cancer treatment began (i.e., spans KQs 1 and 2).

We grouped nutrition interventions into six broad categories (Table 2.2). Given the lack of a recognized classification system for grouping or describing non-pharmacologic nutrition interventions, we grouped studies into single intervention types based on the content and intent of the intervention and the intended audience, using taxonomies and definitions where available. We acknowledge that our categorization scheme represents broad definitions and that our assignments may not be precise. Classification, particularly within an intervention type, presented challenges due to the extreme heterogeneity of interventions and comparison populations. Therefore, we grouped studies best aligned with other studies in the same intervention type that assessed, to the best extent possible, similar intervention components (e.g., a single dietary supplement). Given the volume and heterogeneity of evidence, this strategy helped us to adequately synthesize and interpret results.

Intervention Category	Description
Nutrition Counseling	Nutrition counseling involves an individualized nutrition assessment followed by personalized care of nutrition and diet-related needs with the goal of achieving and maintaining optimal nutrition status across the continuum of care.
Dietary Supplements	Dietary Supplements include products (e.g., added arginine, glutamine, fish oil) containing one or more ingredients meant to supplement the diet for improved nutrition status, but were not meant to replace calories. Vitamins, minerals and antioxidants were not included.
Special Diets	Special diets include the use of defined nutrition plans or approaches such as fasting (intermittent or short term), calorie restriction, ketogenic, Mediterranean, high calorie, or high protein diets to support cancer care.
Route or Timing of Nutrition Interventions	Route or timing interventions involve testing only the route (e.g., PEG tube, enteral feeding) or timing (e.g., initiation or duration) of nutrition interventions with similar nutrition contents.
Nutrition Support Including Oral Nutrition Supplements	Nutrition Support interventions involve the use of total parenteral nutrition, enteral nutrition (tube feeding), and oral nutrition supplements (e.g., Ensure®, Boost®, immunonutrition (oral nutrition supplements that include a set of nutrients meant to have an effect of the immune system ²⁵)) to maintain or improve nutrition status.
Multi-Component Interventions	Multi-component interventions involve multiple strategies for nutrition interventions, such as counseling plus use of dietary supplements.

Tab	ble	2.2.	Intervention	on categor	ies and c	lescriptions
-----	-----	------	--------------	------------	-----------	--------------

For each intervention type, we present results in two general sections: Eligible Studies and Description of Interventions. To provide a deeper description of the literature, we give two further results sections for those intervention/cancer type groups with a relatively larger number of studies. The first additional section describes the assessed risk of bias for eligible studies. For studies with low and medium risk of bias, we then provide an additional section that qualitatively describes the reported directionality of outcomes. We use arrows for directionality to indicate, for each outcome comparison, whether an intervention shows a benefit, a harm, or neither benefit nor harm.
Chapter 3. Search Results

We list all studies excluded at full text screening, by exclusion category, in Appendix B. See Figure 3.1 for details of the screening process.





Based on a comprehensive literature search conducted July 2022, we identified 9,798 unique references. Based on title and abstract review, we excluded 8,644 references not relevant to Key Questions 1– 4. Of the remaining 1,154 references, 206 studies from 219 publications were considered eligible after full text review.

We organized the results by Key Question, then broadly by type of nutrition intervention and type of cancer. Table 3.1 provides an overview of the eligible literature set by nutrition intervention and Key Question. We describe studies for which risk of bias was assessed in Chapter 4, Table 4.1.

Location	Intervention	lotal
Chapter 5	Nutrition Counseling	0
Nutrition Interventions	Dietary Supplements	5
Prior to Cancer	Special Diets	0
Treatment	Route or Timing of Nutrition Interventions	0
(KQ1)	Nutrition Support Including Oral Nutrition Supplements	15
	Multi-Component Interventions	0
	Total	20
Chapter 6	Nutrition Counseling	0
Nutrition Interventions	Dietary Supplements	15
Prior to and Including	Special Diets	0
the Initiation of	Route or Timing of Nutrition Interventions	9
Cancer Treatment	Nutrition Support Including Oral Nutrition Supplements	13
(Spans KQ 1 and 2)	Multi-Component Interventions	1
	Total	38
Chapter 7	Nutrition Counseling	15
Nutrition Interventions	Dietary Supplements	34
after Treatment Began	Special Diets	8
(KQ2)	Route or Timing of Nutrition Interventions	31
	Nutrition Support Including Oral Nutrition Supplements	43
	Multi-Component Interventions	10
	Total	141
Chapter 8	Nutrition Counseling	12
Effect of Nutrition	Dietary Supplements	10
Interventions on	Special Diets	10
Symptoms	Route or Timing of Nutrition Interventions	8
(KQ3)	Nutrition Support Including Oral Nutrition Supplements	32
	Multi-Component Interventions	7
	Total	79
Chapter 9	Nutrition Counseling	0
Effect of Nutrition	Dietary Supplements	0
Interventions on	Special Diets	4
Weight Loss	Route or Timing of Nutrition Interventions	0
(NQ4)	Nutrition Support Including Oral Nutrition Supplements	0
	Multi-Component Interventions	0
	Total	4

Table 3.1. Identified unique included studies by intervention and Key Question

Note: Numbers of totals and numbers of included studies are not mutually exclusive. Studies addressing KQ3 could also address KQ1, KQ2 or span KQ1 and 2. includes a total of 206 unique studies across 219 publications. **Abbreviation:** KQ=Key Question.

Chapter 4. Overview of Nutrition Interventions

Key Points

- Two decades of randomized trial evidence across 206 studies of nutrition interventions for adults prior to and/or during cancer treatment focused on use of dietary supplements, nutrition support (including oral nutrition supplements), and the route or timing of nutrition interventions within gastrointestinal and head and neck cancers.
- Studies focused on evaluating changes in body weight/composition, adverse events, length of hospital stay, and quality of life.
- Among studies with a high volume of literature (n=114), which predominately examined dietary supplements and nutrition support in patients with gastrointestinal and head and neck cancers, 11 percent (n=12) were rated as low risk of bias (higher quality), 40 percent (n=46) medium risk of bias and 49 percent (n=56) high risk of bias (lower quality).
- Despite a large volume of studies evaluating nutrition interventions and outcomes of interest across Key Questions (KQs), heterogeneity was high across study populations, interventions, and outcomes (measure definitions, timing of measurements), even within nutrition intervention categories, making meta-analysis infeasible.

Intervention Type, Outcomes, and Risk of Bias

This chapter provides an evidence map of the KQ 1–4 included studies. Here we present interventions across KQs in aggregate to provide a broad evidence map of the types of interventions studied by KQ and cancer type (Table 4.1). Among this literature, the highest volume of studies focused on gastrointestinal cancers; head and neck cancers were second most common. Interventions focused mainly on use of dietary supplements, nutrition support (including oral nutrition supplements), and the route or timing of nutrition interventions. Few studies evaluated nutrition interventions aimed at body weight loss (KQ 4, n=4) among cancer patients who were overweight or obese. Within each KQ and nutrition intervention type, we found significant heterogeneity of study populations, timing, and duration of the interventions, as well as the interventions themselves (details are provided by KQ in Chapters 5-9).

Questions						
Location	Intervention	Head and Neck Cancer	Gastrointestinal Cancer	Multiple Cancers	Other Cancer Types [*]	Total [†]
Chapter 5	Nutrition Counseling	0	0	0	0	0
Nutrition	Dietary Supplements	1	2*	0	2	5
Interventions	Special Diets	0	0	0	0	0
Prior to Cancer	Route or Timing of Nutrition Interventions	0	0	0	0	0
Treatment (KQ1)	Nutrition Support Including Oral Nutrition Supplements	0	15*	0	0	15
	Multi-Component Interventions	0	0	0	0	0
	Total	-	-	-	-	20

 Table 4.1. Identified unique included studies by nutrition intervention and cancer type across Key

 Questions

Location	Intervention	Head and Neck	Gastrointestinal Cancer	Multiple Cancers	Other Cancer Types*	Total [†]
		Cancer			.,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	
Chapter 6	Nutrition Counseling	0	0	0	0	0
Nutrition	Dietary Supplements	2	10*	2	1	15
Interventions	Special Diets	0	0	0	0	0
Prior to and	Route or Timing of	2	6	1	0	9
Including the	Nutrition Interventions					
Initiation of	Nutrition Support	1	10*	0	2	13
Cancer	Including Oral Nutrition					
Ireatment	Supplements					
(Spans KQ 1	Multi-Component	0	0	1	0	1
anu z)	Interventions					
	Total	-	-	-	-	38
Chapter 7	Nutrition Counseling	3	4	6	2	15
Nutrition	Dietary Supplements	10*	16*	3	5	34
Interventions	Special Diets	0	1	2	5	8
anter	Route or Timing of	4	24*	1	2	31
Bogan	Nutrition Interventions					
(KO2)	Nutrition Support	4	27*	4	8	43
(1(32))	Including Oral Nutrition					
		0	0	0	4	10
		2	2	2	4	10
						4.4.4
Chanter 9	Nutrition Counceling	-	-	-	-	141
Effect of	Diotony Supplements	1	4 5*	4	3	10
Nutrition	Special Dieta	0		3	7	10
Interventions	Bouto or Timing of	0	1	2	1	0
on	Nutrition Interventions	2	4	1	1	0
Symptoms	Nutrition Support	5	20*	3	1	30
(KQ3)	Including Oral Nutrition	5	20	5	4	52
. ,	Supplements					
	Multi-Component	2	0	3	2	7
	Interventions	_	, i i i i i i i i i i i i i i i i i i i	C C	-	
	Total	-	-	-	-	79
Chapter 9	Nutrition Counseling	0	0	0	0	0
Effect of	Dietary Supplements	0	0	0	0	0
Nutrition	Special Diets	0	0	0	4	4
Interventions	Route or Timing of	0	0	0	0	0
on Weight	Nutrition Interventions					
Loss	Nutrition Support	0	0	0	0	0
(KQ4)	Including Oral Nutrition					
	Supplements					
	Multi-Component	0	0	0	0	0
	Interventions					
	Total	-	-	-	-	4

Note: Numbers of totals and numbers of included studies are not mutually exclusive. Studies addressing KQ3 could also address KQ1, KQ2, or span KQ 1 and 2. Includes a total of 206 unique studies across 219 publications.

Abbreviations: KQ= Key Question; - = Not Applicable.

*Other cancer types include studies evaluating all remaining cancer types not included in previous categories (e.g., breast, prostate, lung cancer;

[†]Indicates studies in which risk of bias was assessed.

Outcomes evaluated also varied significantly across KQs. Table 4.2 provides a summary of the outcomes evaluated. For nutrition interventions prior to cancer treatment (KQ1), studies most commonly evaluated intermediate outcomes such as changes in body weight/composition changes or nutrition status, and final outcomes such as adverse events and length of hospital stay.

Studies examining nutrition interventions prior to and including the initiation of cancer treatment (spans KQs 1 and 2) most commonly evaluated body weight/composition changes as intermediate outcomes and adverse events, survival, and length of hospital stay as final outcomes. Studies evaluating nutrition interventions during cancer treatment (KQ2) looked at a wide range of outcomes, most commonly evaluating adverse events, followed by body weight/composition changes and length of hospital stay. Evaluation of the effects of nutrition interventions on cancer and treatment-related symptoms (KQ3) most commonly evaluated quality of life outcomes followed by changes in reported symptoms. Few studies evaluated the impact of nutrition interventions on readmissions/emergency room visits or changes in treatment tolerance. Although a large volume of studies evaluated outcomes of interest across KQs, significant heterogeneity among measure definitions, timing of measurements, and populations evaluated made aggregation or meta-analyses less feasible.

Location	Intervention	Body Weight/Com position	Changes in Nutrition	AEs	Readmissions/ Emergency Room Visits	Survival	LOS	Treatment Tolerance	QOL	Symptoms*	Functional Status
			Status								
Chapter 5 Nutrition	Nutrition Counseling	0	0	0	0	0	0	-	-	-	-
Interventio	Dietary	2	1	2	0	2	1	-	-	-	-
Cancer	Special Diets	0	0	0	0	0	0		_		
Treatment	Route or	0	0	0	0	0	0	-	_	-	-
(KQ1)	Timing of Nutrition	Ŭ	0	0	Ū	Ŭ	Ū				
	Intervention		-		-						
	Nutrition	7	4	12	2	1	10	-	-	-	-
	Support										
	Including Oral										
	Nutrition										
	Supplement	0	0	0	0	0	0				
	Mulli-	0	0	0	0	U	0	-	-	-	-
	Total (20	0	5	1/	2	3	11	_	_	_	_
	studies)	9	5	14	2	5		-	-	-	-
Chanter 6	Nutrition	0	0	0	0	0	0		_	_	
Nutrition	Counseling	0	Ū	Ŭ	Ŭ	Ū	Ŭ				
Interventio	Dietary	9	3	13	2	9	7	_	-	_	_
ns Prior to	Supplements	Ū	Ū	10	-	Ū					
and	Special Diets	0	0	0	0	0	0	_	-	-	_
Including	Route or	4	2	6	1	1	5	_	-	-	_
the	Timing of			-			_				
Initiation	Nutrition										
of Cancer	Interventions										
Treatment	Nutrition	4	0	12	3	6	9	-	-	-	-
(Spans KQ	Support										
1 and 2)	Including Oral										
	Nutrition										
	Supplements										
	Multi-	0	0	0	0	1	0	-	-	-	-
	Component Interventions										

Table 4.2. Number of studies evaluating outcomes by Key Question*

Location	Intervention	Body Weight/Com position	Changes in Nutrition	AEs	Readmissions/ Emergency Room Visits	Survival	LOS	Treatment Tolerance	QOL	Symptoms [*]	Functional Status
		Promon	Status								
	Total (38 studies)	17	5	31	6	17	21	-	-	-	-
Chapter 7 Nutrition	Nutrition Counseling	12	9	6	1	6	1	-	-	-	-
Interventio ns after	Dietary Supplements	20	5	26	3	12	18	-	-	-	-
Treatment	Special Diets	4	0	3	0	2	1	-	-	-	-
Began (KQ2)	Route or Timing of Nutrition Interventions	12	6	28	2	10	22	-	-	-	-
	Nutrition Support Including Oral Nutrition Supplements	26	14	32	4	12	13	-	-	-	-
	Multi- Component Interventions	7	5	3	1	3	2	-	-	-	-
	Total (141 studies)	81	39	98	11	45	57	-	-	-	-
Chapter 8 Effect of	Nutrition Counseling	-	-	-	-	-	-	3	9	6	6
Nutrition Interventio	Dietary Supplements	-	-	-	-	-	-	2	7	5	2
ns on	Special Diets	-	-	-	-	-	-	0	7	7	3
Symptoms	Route or Timing of Nutrition Interventions	-	-	-	-	-	-	2	6	1	1
	Nutrition Support Including Oral Nutrition Supplements	-	-	-	-	-	-	7	20	13	9
	Multi- Component Interventions	-	-	-	-	-	-	0	6	3	2

Location	Intervention	Body Weight/Com position	Changes in Nutrition	AEs	Readmissions/ Emergency Room Visits	Survival	LOS	Treatment Tolerance	QOL	Symptoms*	Functional Status
			Status								
	Total (79	-	-	-	-	-	-	14	55	35	23
	studies)										
Chapter 9	Nutrition	0	0	0	0	0	-	-	-	-	
Effect of	Counseling										
Nutrition	Dietary	0	0	0	0	0	-	-	-	-	
Interventio	Supplements										
ns on	Special Diets	4	0	0	0	0	-	-	-	-	
Weight	Route or	0	0	0	0	0	-	-	-	-	
Loss	Timing of										
(KQ4)	Nutrition										
	Interventions										
	Nutrition	0	0	0	0	0	-	-	-	-	
	Support										
	Including Oral										
	Nutrition										
	Supplements										
	Multi-	0	0	0	0	0	-	-	-	-	
	Component										
	Interventions										
	Total (4	4	0	0	0	0	-	-	-	-	
	studies)										

Note: Studies may evaluate more than one outcome.

Abbreviations: AE= adverse events; LOS=Length of hospital stay; QOL=Quality of Life; - = Not Applicable. *Symptoms may include cancer or cancer treatment related symptoms such as fatigue, nausea and vomiting, and appetite.

As detailed in our methods, we assessed risk of bias for studies within a higher volume of nutrition intervention literature (Table 4.3). These interventions included studies predominantly evaluating dietary supplements, nutrition support (including oral nutrition supplements), and route or timing of nutrition interventions. Among these studies, we assessed the literature as having mostly moderate to high risk of bias, with 49 (n=56) percent of studies rated as high. Only 11 percent (n=12) of studies were rated as low risk of bias.

Location	Intervention	Head and Neck Cancer	Gastrointestinal Cancer	Multiple Cancers	Other Cancer Types [*]
Chapter 5	Nutrition Counseling	-	-	-	-
Nutrition Interventions	Dietary Supplements	-	2 High (N=2)	-	-
Prior to			(
Cancer	Special Diets	-	-	-	-
Treatment	Route or Timing of Nutrition Interventions	-	-	-	-
(KQ1)	Nutrition Support Including Oral Nutrition	-	2 low	-	-
	Supplements		6 Medium 7 High (N=15)		
	Multi-Component Interventions	-	-	-	-
Chapter 6	Nutrition Counseling	-	-	-	-
Nutrition Interventions Prior to and	Dietary Supplements	-	4 Medium 6 High (N=10)	-	-
Including the	Special Diets	-	-	-	-
Initiation of	Route or Timing of Nutrition Interventions	-	-	-	-
Treatment (Spans KQ 1 and 2)	Nutrition Support Including Oral Nutrition Supplements	-	2 Low 4 Medium 4 High (N=10)	-	-
	Multi-Component Interventions	-	-	-	-
Chapter 7	Nutrition Counseling	-	-	-	-
Nutrition Interventions after Treatment Began (KQ2)	Dietary Supplements	6 Medium 4 High (N=10)	4 Low 8 Medium 4 High (N=16)	-	-
	Special Diets	-	-	-	-
	Route or Timing of Nutrition Interventions	-	1 Low 10 Medium 13 High (N=24)	-	-
	Nutrition Support Including Oral Nutrition Supplements	-	3 Low 8 Medium 16 High (N=27)	-	-
	Multi-Component Interventions	-	-	-	-
Chapter 8	Nutrition Counseling	-	-	-	-
Effect of Nutrition Interventions	Dietary Supplements	-	3 Medium 2 High (N=5)	-	-
on	Special Diets	-	-	-	-

Table 4.3. Overview of risk of bias assessment

Location	Intervention	Head and Neck Cancer	Gastrointestinal Cancer	Multiple Cancers	Other Cancer Types [*]
Symptoms	Route or Timing of Nutrition Interventions	-	-	-	-
(KQ3)#	Nutrition (nutrition support)	-	5 Low 4 Medium 11 High (N=20)	-	-
	Multi-Component Interventions	-	-	-	-
Chapter 9	Nutrition Counseling	-	-	-	-
Effect of	Dietary Supplements	-	-	-	-
Nutrition	Special Diets	-	-	-	-
Interventions	Route or Timing of Nutrition Interventions	-	-	-	-
on Weight Loss	Nutrition Support Including Oral Nutrition Supplements	-	-	-	-
(KQ4)	Multi-Component Interventions	-	_	-	-

Note: Numbers of totals and numbers of included studies are not mutually exclusive. Studies addressing KQ3 could also address KQ1, KQ2 or span KQ1 and 2.

Abbreviations: - = Not Applicable; KQ=Key Question; N=number.

* Other cancer types include studies evaluating all remaining cancer types not included in previous categories (e.g., breast, prostate, lung cancer).

Chapter 5. Nutrition Interventions Prior to Cancer Treatment

Key Points

- Studies of nutrition interventions prior to cancer treatment included use of dietary supplements and nutrition support mostly among gastrointestinal cancer populations in the inpatient surgical setting and outpatient settings.
- Studies focused on evaluating changes in body weight/composition, changes in nutrition status, adverse events, survival, and length of hospital stay.
- Of studies assessed for risk of bias, all dietary supplement studies and 47 (7/15) percent of nutrition support studies were high risk of bias (poor quality).
- Among eight low- or medium-risk-of-bias studies in nutrition support, studies reported mixed results on development of complications, improvements in body weight loss, and length of hospital stay
- Studies showed high heterogeneity across study populations, interventions, and outcomes so meta-analysis was not feasible.

Overview

This chapter includes studies that addressed Key Question (KQ) 1 and examined nutrition interventions that were only delivered prior to the initiation of cancer treatment. Chapter 6 separately presents studies of interventions that were initiated prior to and continued during cancer treatment (KQs 1 and 2). In order to briefly discuss what has been examined for each intervention type, we present results in two general sections: Eligible Studies and Description of Interventions. We assessed risk of bias for nutrition intervention/cancer types with a higher volume of studies. This allowed us to identify the greatest amount of lower-risk of bias evidence concentrated by specific nutrition interventions by cancer type. The number of studies for which we assessed risk of bias is indicated in Table 5.1. For interventions with low- to medium-risk-of-bias studies, we note reported outcome directionality the end of the chapter. Detailed information on all eligible studies can be found in Appendix C.

For KQ1, we identified 20 unique studies that examined nutrition interventions prior to the initiation of cancer treatment. Studies examined the use of dietary supplements (n=5) and nutrition support (n=15) (Table 5.1). No studies examined the use of nutrition counseling alone, or the route or timing of nutrition interventions, special diets, or multicomponent interventions. Table 5.1 summarizes the characteristics of the literature set.

Table 5.1. Studies exam	nining use of nu	utrition i	ntervent	ions prior to	cancer t	reatment	, stratified by
intervention and cancer	r type* (KQ1)						

Intervention	Head and	Gastrointestinal	Multiple	Other Cancer	Total
	Neck		Cancers	Types [†]	
Nutrition Counseling	0	0	0	0	0
Dietary Supplements	1	2*	0	2	5
Special Diets	0	0	0	0	0
Route or Timing of	0	0	0	0	0
Nutrition Interventions					
Nutrition Support	0	15*	0	0	15
Including Oral					
Nutrition Supplements					

Intervention	Head and Neck	Gastrointestinal	Multiple Cancers	Other Cancer Types [†]	Total
Multi-Component Interventions	0	0	0	0	0
Total	-	-	-	-	20

Abbreviations: KQ= Key Question; - = Not applicable.

*Risk of bias was performed.

[†]Other cancer types include studies evaluating all remaining cancer types not included in previous categories (e.g., breast, prostate, lung cancer).

Dietary Supplements

Eligible Studies

Table 5.2 summarizes the characteristics of the literature set. We identified five unique small studies examining dietary supplements used prior to cancer treatment.²⁶⁻³⁰ All enrolled 100 or fewer individuals, and primarily assessed the impact of dietary supplements on clinical laboratory outcomes such as immune response,^{27, 28} inflammatory profiles,²⁸ or components of a metabolic panel.^{29, 30} The primary outcome of the study incorporating head and neck cancer patients was prevention of severe oral mucositis.²⁶ Dietary supplements were prescribed by a dietitian/nutritionist in two studies, where the primary cancer treatment was chemotherapy alone.^{26, 28} Appendix C provides detailed evidence tables.

Table 5.2. Basic characteristics of stud	ies of nutrition interventions	prior to cancer	treatment:
dietary supplements			

Characteristics	Information		
Total Included Studies	5 Studies		
Study Sample Size	3 50-75		
	2 76-100		
Cancer Type	1 Head & Neck		
	2 Gastrointestinal		
	2 Other cancer types [*]		
Intervention Delivery Setting	2 Inpatient		
	2 Outpatient		
	1 Not reported		
Dominant Cancer Treatment Type of	3 Surgery alone		
Participants	2 Chemotherapy alone		
Limited to Malnourished Patients	5 No		
Malnourishment Screening Tool Used	1 Multiple tools/metrics		
	2 Other tool/metric		
	2 Not reported		
Provider Prescribing or Delivering the	2 Dietitian/nutritionist		
Intervention	1 Other		
	1 Physician		
	1 Not reported		
Route of Administration	3 Oral		
	1 Parenteral		
	1 Not reported		
Geographic Region of Intervention	1 Asia		
	2 Europe		
	2 Other		
% of Participants with Stage IV Disease	2 <10%		
	1 >50%		
	2 Not reported		
% Female Participants	2 0-25%		
	2 26-50%		
	1 76-100%		

Characteristics	Information	
Mean Age of Participants	4 50-64	
	1 Not reported	
Outcomes Evaluated†	2 Body weight or composition changes	
	1 Changes in nutrition status	
	2 Adverse events	
	2 Survival	
	1 Length of stay	

*Other cancer types include studies evaluating all remaining cancer types not included in previous categories (e.g., breast, prostate, lung cancer).

† Studies may evaluate multiple outcomes.

Description of Interventions

One study compared parenteral fish oil lipid emulsion versus lipid emulsions rich in triglycerides prior to gastrointestinal cancer surgery.²⁷ Another study compared an oral nutrition formula enriched with omega-3 fatty acids versus a standard oral nutrition formula for gastric cancer.²⁸ A third study compared an immune modulating formula enriched with arginine, omega-3 fatty acids, and nucleotides versus a normal diet for lung cancer.²⁹ Another study compared an oral immunonutrition formula with fatty acids, arginine, fiber, and nucleotides versus a standard enteral formula in head and neck cancer patients.²⁶ A final study compared preoperative oral carbohydrate loading versus standard fasting prior to breast cancer surgery.³⁰ These studies were not U.S.-based, with two conducted in Brazil,^{27, 28} one in Norway,³⁰ one in Thailand,²⁶ and one in Turkey.²⁹ Overall, studies differed widely in type of dietary supplements administered, route of administration, and control populations.

Nutrition Support Including Oral Nutrition Supplements

Eligible Studies

Table 5.3 summarizes the characteristics of the literature set. We identified 15 unique studies examining nutrition support interventions prior to cancer treatment.³¹⁻⁴⁵ All interventions were delivered among patients with gastrointestinal cancer. Eight studies had sample sizes over 100. The primary aims of these studies varied and include adverse events or postoperative complications,^{31, 32, 38} body weight or composition changes,³⁷ changes in nutrition status,⁴²⁻⁴⁵ immune function,^{43, 44} intraoperative core temperature during surgery,³⁴ quality of life,³⁷ feeding intolerance rate,³⁵ physical status,⁴² and self-sufficiency.⁴² Five studies did not a report a specific primary aim. ^{33, 36, 39-41} Most studies investigated the effect of nutrition supplementation before cancer treatment on length of stay^{31-34, 36-42, 44, 45} and adverse events or complications.^{31-35, 38, 39, 41-44} Appendix C provides detailed evidence tables.

natition support including ordinatition supplements	
Characteristics	Information
Total Included Studies	15 Studies
Study Sample Size	6 50-75
	1 76-100
	8 >100
Cancer Type	15 Gastrointestinal

 Table 5.3. Basic characteristics of studies of nutrition interventions prior to cancer treatment:

 nutrition support including oral nutrition supplements

Characteristics	Information
Intervention Delivery Setting	10 Inpatient
	2 Outpatient
	1 Multiple settings
	1 Other
	1 Not reported
Dominant Cancer Treatment Type of	13 Surgery alone
Participants	2 Multiple therapies
Limited to Malnourished Patients	1 Yes
	14 No
Malnourishment Screening Tool Used	5 Nutrition Risk Screening NRS-2002
	5 Other tool/metric
	5 Not reported
Provider Prescribing or Delivering the	3 Dietitian/nutritionist
Intervention	1 Multiple
	11 Not reported
Route of Administration	12 Oral
	1 Parenteral
	2 Enteral
Geographic Region of Intervention	8 Asia
	6 Europe
	1 North America
% of Participants with Stage IV Disease	6 < 10%
	1 20-29%
	8 Not reported
% Female Participants	2 ≤ 25%
	11 26-50%
Manual Annual Deutlicius anto	251-75%
Mean Age of Participants	8 50-64
	0 00+
Outcomes Fuelueted*	7 Deduweight er composition changes
Outcomes Evaluated	A Changes in putrition status
	4 Changes in numition status
	2 Poodmissions or omorganou room visits
	2 Neaumissions of emergency room visits
	10 Longth of ctoy
	ן וט בפווענוו טו אמץ

*Studies may evaluate multiple outcomes.

Description of Interventions

Most interventions were delivered preoperatively in a surgical setting. One study compared the use of daily oral supplementation with Fortisip® oral nutrition support and dietary advice for a minimum of 10 days prior to surgery for colorectal cancer versus dietary advice alone.³² The same group then conducted a larger trial comparing daily oral supplementation with Fortsip Compact® along with dietary advice versus dietary advice alone before surgery in patients with colorectal cancer.³¹

Three studies specifically looked at immunonutrition. One compared oral whole immunonutrition (IMPACT Advanced Recovery) versus no additional supplementation in adults with locally advanced pancreatic adenocarcinoma before irreversible electroporation.³⁹ The second compared enteral immunonutrition (IMPACT) versus standard enteral nutrition before elective curative surgery for patients with colorectal or gastric carcinoma.⁴⁴ The third study investigated preoperative immunonutrient-enriched supplementation (Newcare Omega®) versus normal diet in patients receiving elective colon cancer resection.³⁸

Other studies were more diverse in the type of oral nutrition support across different types of gastrointestinal cancers. One study of gastrointestinal patients compared Nutricia® starting 1

week preoperatively and a short peptide-based elemental diet for nine days postoperatively via a nasogastric tube versus no preoperative use of oral nutrition supplements.⁴³ Another study compared preoperative, oral, hypercaloric supplementation (Nutridrink Protein®) versus a regular diet for individuals with gastrointestinal cancer.³⁶ A third examined peripheral intravenous nutrition on typically fasting days for patients undergoing workup for biliopancreatic masses.³⁷ Another study compared preoperative oral supplementation versus preoperative dietary advice for individuals undergoing surgery for gastrointestinal cancer.³⁵ One study compared oral nutrition combined with microbial preparations versus traditional intestinal preparation in elderly patients before radial resection of colorectal cancer.⁴¹ A sixth study examined preoperative oral nutrition compared to no oral supplementation in patients undergoing elective resection of colorectal carcinoma.⁴² The remaining study compared Nutrison Fiber® versus a routine preoperative diet in patients with Siewert II and III adenocarcinomas of esophagogastric junction after neoadjuvant chemotherapy.⁴⁵

The final three studies examined carbohydrate-rich beverages. One compared oral supplementation with a carbohydrate-rich beverage (Arginaid Water®) versus drinking clear water before undergoing laparoscopic colon cancer surgery.³⁴ The second compared a carbohydrate rich beverage in the 2 days prior to surgery for colorectal cancer versus preoperative fasting.⁴⁰ Another study investigated the effects of a single-dose versus a double-dose of a 10% glucose solution prior to radical gastrectomy.³³ Overall, studies varied widely in the type of nutrition support administered, route of administration, and control populations.

Risk of Bias and Outcome Assessment

Table 5.4 summarizes risk of bias assessment. Appendix C provides risk of bias assessments by study and qualitatively describes the reported directionality of reported outcomes.

Intervention	Head and Neck	Gastrointestinal	Multiple Cancers	Other Cancer Types [*]
Nutrition Counseling	-	-	-	-
Dietary Supplements	-	2 High (N=2)	-	-
Special Diets	-	-	-	-
Route or Timing of Nutrition Interventions	-	-	-	-
Nutrition Support Including Oral Nutrition Supplements	-	2 Low 6 Medium 7 High (N=15)	-	-
Multi-Component Interventions	-	-	-	-

 Table 5.4. Risk of bias assessment for nutrition interventions prior to cancer treatment

Note: - = risk of bias not performed.

* Other cancer types include studies evaluating all remaining cancer types not included in previous categories (e.g., breast, prostate, lung cancer).

Outcomes for Dietary Supplements

Both studies were evaluated as high risk of bias (low quality) and outcomes were not evaluated.

Outcomes for Nutrition Support Including Oral Nutrition Supplements

Six of eight studies of low to medium risk of bias examined surgically treated gastrointestinal cancer patients receiving nutrition interventions in the hospital.^{31, 32, 38, 40, 44, 42} Study sample sizes ranged from 50 to 161. One to two studies reported on each of the following outcomes: readmissions or emergency room visits and changes in nutritional status. Seven studies reported on adverse events.^{31-33, 35, 38, 42, 44} However, the studies were distributed across interventions that could not be aggregated. Studies showed mixed results for use of preoperative nutrition support. One medium-risk-of-bias study reported a benefit of preoperative oral nutrition support on reducing body weight loss and surgical site infections;³¹ however, it showed no benefit on changes in nutrition status, chest infections or other complications, length of hospital stay, or survival. Another medium-risk-of-bias study showed no benefit of a preoperative oral supplement drink on complications.³² A third study of preoperative nutrition support reported a benefit on noninfectious complications and length of hospital stay.⁴⁴ One medium-risk-of-bias study found no benefit for patients receiving a double-dose of a preoperative carbohydrate drink in adverse events, readmission or reoperation, and length of hospital stay.³³ Another mediumrisk-of-bias study found that oral nutrition supplementation during the seven days before surgery reported a benefit on body weight recovery after discharge, but no changes in body weight or composition change overall, adverse events, or length of stay.³⁸ A low-risk-of-bias study found that preoperative oral nutrition supplementation had no effect on adverse events or hospital readmission compared to dietary advice.³⁵ Another medium-risk-of-bias study reported no benefit in body weight or composition, complications, or hospital stay for those that received oral nutritional supplements before surgery.⁴² Finally, one low-risk-of-bias study reported that a preoperative carbohydrate drink reduced length of hospital stav.⁴⁰

Variation in the Effects of Nutrition Interventions on Preventing Negative Outcomes

While studies enrolled individuals from multiple cancer types, treatments, and stages (KQ1a), across the lifespan (KQ1b), with varying degrees of muscle wasting (KQ1c) and in those with a range of comorbid conditions (KQ1d), no eligible studies specifically evaluated whether the effects of nutrition interventions on preventing negative outcomes varied across these characteristics.

Chapter 6. Nutrition Interventions Prior to and Including the Initiation of Cancer Treatment

Key Points

- Studies of nutrition interventions prior to and continuing after the start of cancer treatment included use of nutrition support, dietary supplements, and evaluation of the route or timing of nutrition interventions, mostly among gastrointestinal cancer populations in the inpatient surgical setting.
- Studies most commonly focused on evaluating changes in body weight/composition, adverse events, length of hospital stay, and survival, with five studies evaluating changes in nutrition status, readmissions, or emergency room visits.
- Of studies assessed for risk of bias, 60 percent (n=6/10) of dietary supplement studies and 40 percent (n=4/10) of nutrition support studies were high risk of bias (poor quality), with only two nutrition support studies assessed as low risk of bias.
- Among four medium-risk-of-bias studies in dietary supplements, studies reported mixed results on body weight changes, readmissions, length of hospital stay, development of complications, and survival.
- Two low- and four medium-risk-of-bias study of nutrition support reported mixed results, with some studies reporting a benefit and some reporting no difference in reducing adverse events, readmissions, length of hospital stay, and survival.
- Studies showed high heterogeneity across study populations, interventions, and outcomes, making meta-analysis infeasible.

Overview

This chapter includes nutrition interventions initiated prior to and continued after cancer treatment began, encompassing both Key Questions (KQs) 1 and 2. This stands in contrast to interventions delivered only prior to the initiation of cancer treatment (Chapter 5) and interventions delivered only after cancer treatment began (Chapter 7). For each intervention type, we present results in two general sections: Eligible Studies and Description of Interventions for a brief discussion of what has been examined. We assessed risk of bias for intervention/cancer types with a higher volume of studies. This allowed us to identify the greatest amount of lower risk of bias evidence concentrated according to specific nutrition interventions by cancer type. The number of studies by intervention type and cancer type for which we assessed risk of bias is indicated in Table 6.1. For interventions with low- to medium-risk-of-bias studies, we note outcome directionality at the end of the chapter. Detailed information on all eligible studies can be found in Appendix D.

We identified 38 unique studies across 42 publications that examined nutrition interventions prior to and after initiation of cancer treatment. Studies examined the use of dietary supplements (n=15), the route or timing of nutrition interventions (n=9), the use of nutrition support (n=13), and multicomponent interventions (n=1) (Table 6.1). No studies examined nutrition counseling or the use of special diets.

Table 6.1. Studies of nutrition interventions prior to and including initiation of cancer treatment, stratified by intervention type and cancer type

Intervention	Head and	Gastrointestinal	Multiple	Other Cancer	Total
	Neck		Cancers	Types ^a	
Nutrition Counseling	0	0	0	0	0
Dietary Supplements	2	10*	2	1	15
Special Diets	0	0	0	0	0
Route or Timing of	2	6	1	0	9
Nutrition Interventions					
Nutrition Support	1	10*	0	2	13
Including Oral					
Nutrition Supplements					
Multi-Component	0	0	1	0	1
Interventions					
Total					38

*Risk of bias was performed.

^aOther cancer types include studies evaluating all remaining cancer types not included in previous categories (e.g., breast, prostate, lung cancer).

Dietary Supplements

Eligible Studies

Table 6.2 summarizes the characteristics of the literature set. We identified 15 unique studies across 16 publications examining the use of dietary supplements prior to and continued after the initiation of cancer treatment.⁴⁶⁻⁶¹ Most studies enrolled fewer than 100 individuals and primarily assessed the impact of the nutrition intervention on clinical laboratory outcomes such as components of a metabolic panel^{46, 51} and biomarkers;^{47, 49, 58, 59} however, a number of studies also assessed clinical outcomes such as complications.^{48, 56-60} Others examined multiple outcomes but did not clearly state the primary intent.^{50, 52-54, 61} The majority of studies were in gastrointestinal cancers. Dietary supplements were prescribed by a dietitian/nutritionist in only two studies, and the majority study populations were treated predominantly with surgery. Appendix D provides detailed evidence tables.

Characteristics	Information	
Total Included Studies	15 Studies	
Study Sample Size	8 50-75	
	2 76-100	
	5 >100	
Cancer Type	2 Head & neck	
	10 Gastrointestinal	
	2 Multiple cancers	
	1 Other cancer types⁺	
Intervention Delivery Setting	6 Inpatient	
	5 Outpatient	
	1 Multiple settings	
	1 Other	
	2 Not reported	
Dominant Cancer Treatment Type of	11 Surgery alone	
Participants	2 Chemotherapy alone	
	1 Radiation alone	
	1 Multiple therapies	
Limited to Malnourished Patients	2 Yes	
	13 No	

Table 6.2. Basic characteristics of studies of nutrition interventions prior to and including initiating cancer treatment: dietary supplements

Characteristics	Information			
Malnourishment Screening Tool Used	2 Malnourishment Screening Tool			
	1 Nutrition Risk Screening-2002			
	7 Other tool/metric			
	1 Multiple tools			
	4 Not reported			
Provider Prescribing or Delivering the	2 Dietitian/nutritionist			
Intervention	3 Multiple providers			
	1 Nurse			
	2 Other			
	7 Not reported			
Route of Administration	8 Oral			
	1 Parenteral			
	6 Enteral			
Geographic Region of Intervention	9 Europe			
	4 Asia			
	1 North America			
	1 Other			
% of Participants with Stage IV Disease	4 <10%			
	1 10-25%			
	1 25-50%			
	2 >50%			
	7 Not reported			
% Female Participants	3 0-25%			
	9 26-50%			
	1 51-75%			
	1 76-100%			
	1 Not reported			
Mean Age of Participants	11 50-64			
	4 ≥65			
Outcomes Evaluated*	9 Body weight or composition changes			
	3 Changes in nutrition status			
	13 Adverse events			
	2 Readmissions or emergency room visits			
	9 Survival			
	7 Length of stay			

*Studies may evaluate multiple outcomes. ⁺Other cancer types include studies evaluating all remaining cancer types not included in previous categories (e.g., breast, prostate, lung cancer).

Description of Interventions

One group of studies examined the use of a single supplement for immunonutrition (eicosapentaenoic - [EPA], omega-3 fatty acid or glutamine-enriched nutrition), but interventions varied by population and had a wide range of intervention durations around the time of cancer surgery. One study examined eicosapentaoenoic (EPA-enriched) nutrition in head and neck cancer versus standard nutrition delivered at least 7 days prior and up to 14 days postsurgery.⁵¹ A second study from two publications examined short- and long-term outcomes of treatment with omega-3 fatty acid-enriched oral nutrition supplements versus standard oral nutrition supplements for 7 days before and after colorectal cancer surgery.^{56, 57} Another study compared EPA-enriched enteral nutrition 5 days prior to surgery versus iso-caloric enteral nutrition but for 30 days postoperatively.⁵⁰ A single study compared amino acids 1 week prior to and for 5 weeks during chemotherapy for gastrointestinal cancer versus standard care.⁵⁹ Finally, three studies examined glutamine for immunonutrition. One of these compared parenteral glutamine supplements with enteral nutrition versus enteral nutrition alone for 5 days pre- and postoperatively in colorectal cancer.⁵³ The second compared

parenteral glutamine solution versus an isonitrogenous control 2 days preoperatively to 5 days postoperatively for pancreatic cancer.⁵² The third, outside of the surgical setting, compared glutamine versus placebo for 3 days prior to the initiation of radiation therapy through completion.⁶⁰

The second group of studies examined a combination of supplements for immunonutrition. One study compared immunonutrition-enhanced enteral nutrition with added L-arginine, omega-3 fatty acids, and nucleotides versus standard enteral nutrition, both delivered 2 days preoperatively through 7 days postoperatively for gynecologic cancer.⁴⁶ A second study compared immune-enhancing enteral nutrition (with added arginine, nucleotides, and fatty acids) delivered 5 days pre- and 7 days postsurgery versus an iso-caloric nutrition supplement for advanced head and neck cancer.⁴⁸ A third study compared immunomodulating nutrition (Oral Impact®) versus Resource 2.0 fiber in patients with gastrointestinal cancer before and after surgery.⁶¹

The remaining studies evaluated a variety of other supplement combinations. One compared nutrition counseling only versus nutrition counseling with oral whey protein supplementation for 3 months in malnourished advanced cancer patients.⁴⁷ Another, a four-arm study, compared omega-3 fatty acid and vitamin D in combination or alone versus a placebo in individuals eligible for chemotherapy for colorectal cancer for 8 weeks.⁴⁹ A third study investigated perioperative oral protein supplementation rich in arginine and omega-6 versus placebo in patients with gastrointestinal cancer undergoing surgery.⁵⁵ Studies were not U.S.-based, and were primarily conducted in Europe and Asia. Overall, studies differed widely in type of dietary supplements administered, route of administration, and control populations.

Route or Timing of Nutrition Interventions

Eligible Studies

Table 6.3 summarizes the characteristics of the literature set. We identified nine unique studies that examined the route or timing of nutrition interventions initiated prior to and continued after cancer treatment.⁶²⁻⁷⁰ Many studies enrolled 100 or more individuals. The primary outcomes of these studies varied considerably, with some seeking to assess the impact of the nutrition intervention on clinical laboratory outcomes such as components of a metabolic panel^{64, 69} and nutrition indicators,⁷⁰ as well as body weight or composition changes.^{67, 69, 71} Others had primary endpoints that assessed reductions in postoperative infections^{62, 65, 66, 68} or length of hospital stay.^{62, 66} Studies rarely indicated that the intervention was delivered by a dietitian/nutritionist (n=1), and almost all populations were treated with surgery alone. Appendix D provides detailed evidence tables.

Characteristics	Information
Total Included Studies	9 Studies
Study Sample Size	3 76-100
	6 >100
Cancer Type	2 Head & neck
	6 Gastrointestinal
	1 Multiple cancers
Intervention Delivery Setting	5 Inpatient
	3 Multiple settings
	1 Not reported

 Table 6.3. Basic characteristics of studies of nutrition interventions prior to and including initiating cancer treatment: route or timing of nutrition interventions

Characteristics	Information		
Dominant Cancer Treatment Type of	7 Surgery alone		
Participants	2 Multiple therapies		
Limited to Malnourished Patients	3 Yes		
	5 No		
	1 Not reported		
Malnourishment Screening Tool Used	3 Other tool/metric		
	1 Multiple tools/metrics		
	5 Not reported		
Provider Prescribing or Delivering the	1 Dietitian/nutritionist		
Intervention	1 Physician		
	7 Not reported		
Route of Administration	3 Oral		
	3 Enteral		
	3 Other		
Geographic Region of Intervention	3 Europe		
	4 Asia		
	2 Other		
% of Participants with Stage IV Disease	2 25-50%		
	7 Not reported		
% Female Participants	4 0-25%		
	4 26-50%		
	1 51-75%		
Mean Age of Participants	7 50-64		
	1 ≥65		
	1 Not reported		
Outcomes Evaluated	4 Body weight or composition changes		
	2 Changes in nutrition status		
	6 Adverse events		
	1 Readmissions or emergency room visits		
	1 Survival		
	5 Length of stay		

Description of Interventions

Two studies enrolled individuals with head and neck cancer. One study compared enteral nutrition initiated immediately after placement of a prophylactic gastrostomy tube and continued for the duration of treatment versus standard care, which included initiation of enteral nutrition when indicated (e.g., oral intake <60% of estimated energy requirements).⁷¹ A second study conducted a three-armed trial in head and neck cancer and compared 1) perioperative whole oral immunonutrition 7 days preoperatively through 7 to 15 days postoperatively, 2) perioperative use of whole oral nutrition without immunonutrients, and 3) preoperative whole oral immunonutrition with a standard diet 4 days postoperatively.⁶⁵ 67, 68 Six studies examined the timing of nutrition support in individuals with gastrointestinal cancer. One study compared preoperative enteral nutrition administered 1 week prior to gastric cancer surgery and continued 2 days postoperatively versus initiating enteral nutrition 2 days postoperatively.⁶⁴ A second study conducted a three-arm trial in gastrointestinal cancer patients comparing the use of either 1) an oral nutrition support five days prior to surgery coupled with intravenous glucose postsurgery, 2) the same preoperative treatment coupled with continued use of same nutrition support postoperatively until oral food resumed, or 3) no preoperative treatment coupled with intravenous glucose postsurgery.⁶⁶ Another study compared, in a three-arm trial, preoperative and postoperative whole immunonutrition versus preoperative only, and no use, in individuals undergoing gastrointestinal surgery.⁶² A fourth study compared enteral versus parenteral nutrition 3 days prior to the start of chemotherapy through 7 days post completion of

chemotherapy in individuals with esophageal cancer.⁶⁷ Another study was a four-arm study examining combinations of whole immunonutrition 7 days pre- and postoperatively versus standard nutrition in individuals with esophageal cancer.⁶⁸ Finally, one study conducted in patients with rectal cancer randomized individuals to enteral nutrition support 7 days preoperatively and a fiber-enriched nutrition support for 5 days postoperatively versus parenteral infusion of amino acids and glucose solutions, with similar energy consumption between groups.⁷⁰

One three-arm study included patients undergoing elective surgery for breast and colorectal cancers and compared consumption of a milk-based oral nutrition supplement preoperatively and ending supplementation at either discharge or 90 days post-discharge.⁶⁹ These non-U.S. studies took place primarily in Europe and Asia. Overall, studies varied in the type of nutrition administered, populations studied, timing and route of administration, and comparisons.

Nutrition Support Including Oral Nutrition Supplements

Eligible Studies

Table 6.4 summarizes the characteristics of the literature set. We identified 13 unique studies across 16 publications examining nutrition support interventions initiated prior to and continued after the start of cancer treatment.⁷²⁻⁸⁷ Most studies enrolled more than 100 individuals. The primary focus of these studies varied considerably, including changes in postoperative complications,^{78, 83-85} length of hospital stay,^{80, 87} hospital-free days,⁸² clinical laboratory markers of nutrition status,⁷³ changes in inflammatory mediators,⁷⁴ body weight or composition changes,⁷⁵ insulin resistance,⁷⁹ and decreased morbidity and mortality.^{72, 80, 81, 86} Most studies were in gastrointestinal patients (n=10). Studies rarely indicated that the intervention was delivered by a dietitian/nutritionist (n=2) and most populations were treated with surgery alone (n=12). Appendix D provides detailed evidence tables.

Characteristics	Information	
Total Included Studies	13 Studies	
Study Sample Size	1 50-75	
	2 76-100	
	10 >100	
Cancer Type	1 Head & Neck	
	10 Gastrointestinal	
	2 Other cancer types [*]	
Intervention Delivery Setting	9 Inpatient	
	2 Multiple settings	
	2 Not reported	
Dominant Cancer Treatment Type of	12 Surgery alone	
Participants	1 Chemotherapy alone	
Limited to Malnourished Patients	4 Yes	
	9 No	
Malnourishment Screening Tool Used	2 Nutrition Risk Screening NRS-2002	
	4 Other tools/metrics	
	7 Not reported	
Provider Prescribing or Delivering the	2 Dietitian/nutritionist	
Intervention	11 Not reported	

 Table 6.4. Basic characteristics of studies of nutrition interventions prior to and including initiating cancer treatment: nutrition support including oral nutrition supplements

Characteristics	Information
Route of Administration	8 Oral
	1 Enteral
	3 Parenteral
	1 Not reported
Geographic Region of Intervention	4 Europe
	8 Asia
	1 North America
% of Participants with Stage IV Disease	5 <10%
	1 >50%
	7 Not reported
% Female Participants	3 0-25%
	9 26-50%
	1 51-75%
Mean Age of Participants	4 50-64
	7 ≥65
	2 Not reported
Outcomes Evaluated†	4 Body weight or composition changes
	12 Adverse events
	3 Readmissions or emergency room visits
	6 Survival
	9 Length of stay

*Other cancer types include studies evaluating all remaining cancer types not included in previous categories (e.g., breast, prostate, lung cancer).

[†]Studies may evaluate multiple outcomes.

Description of Interventions

Ten studies examined nutrition support in gastrointestinal cancer. One study examined use of an oral nutrition supplement for 2 weeks before gastrectomy and for 4 weeks postoperatively versus unspecified standard care.⁷⁸ Another study compared whole immunonutrition for 7 days preoperatively to 5 days postoperatively versus dietary advice alone in colorectal surgery patients.⁸⁰ A third study compared oral immunonutrition with an EPA-enriched supplement for 7 days prior to surgery through 21 days after surgery for gastric cancer versus a standard diet.^{75, 76} Another three-arm trial among gastric cancer patients randomized individuals to 1) multi-oil fat emulsion 7 days preoperatively and intralipid nutrition 2 to 7 days postoperatively, 2) intralipid nutrition 7 days preoperatively and 2 to 7 days postoperatively, or 3) intralipid nutrition 2 to 7 days postoperatively.⁷³ A fourth study of individuals with liver cancer compared whole enteral nutrition 3 days preoperatively to 7 days postoperatively versus regular food intake with a targeted energy requirement.⁸⁷ A single study compared either parenteral or enteral nutrition delivered 7 days prior to 7 days postoperatively versus standard oral diet preoperatively and parenteral nutrition postoperatively in gastric and colorectal cancer patients.⁸⁶ One four-arm study compared a preoperative carbohydrate drink versus placebo and a polymeric formula supplement versus placebo postoperatively for colorectal cancer.⁷⁹ Another study investigated pre- and postoperative early oral feeding versus conventional fasting in elderly patients with hepatocellular carcinoma.⁷⁴ One study compared peripheral parenteral nutrition in the day before surgery and 3 days after surgery versus conventional fluid therapy in colorectal patients.^{83, 84} Finally, a single study compared total parenteral nutrition 10 days preoperatively to 9 days postoperatively in gastrointestinal cancer patients versus a standard preoperative oral diet with postoperative hypocaloric parenteral nutrition.⁷²

Three studies examined nutrition support outside of gastrointestinal cancer. One evaluated patients with resectable bladder cancer, comparing an omega-3 fatty acid-enriched oral nutrition supplement versus a multivitamin, with both administered 4 weeks prior through 4 weeks

postsurgery.⁸² Another compared amino acid-enriched oral nutrition support 1 week prior to the start of chemoembolization for up to 1 year versus usual diet among individuals with hepatocellular carcinoma.⁸¹ A final study investigated an immune-enhancing diet versus a hospital-prepared blenderized diet 7 days before the operation and 14 days after in head and neck cancer patients. ⁸⁵ One study was U.S.-based⁸² while a majority of the remaining studies were based in Asia. Although all studies examined the use of nutrition support, studies varied in the type and duration of nutrition support, cancer type, and control groups.

Multi-Component Interventions

Eligible Studies

Table 6.5 summarizes the characteristics of the literature set. Only one study examined the use of multi-component interventions initiated prior to and continued after the start of cancer treatment.⁸⁸ Appendix D provides detailed evidence tables.

Characteristics	Information
Total Included Studies	1 Study
Study Sample Size	1 >100
Cancer Type	1 Multiple cancers
Intervention Delivery Setting	1 Outpatient
Dominant Cancer Treatment Type of Participants	1 Chemotherapy alone
Limited to Malnourished Patients	1 Yes
Malnourishment Screening Tool Used	1 Not reported
Provider Prescribing or Delivering the	1 Dietitian/nutritionist
Intervention	
Route of Administration	1 Oral
Geographic Region of Intervention	1 Europe
% of Participants with Stage IV Disease	1 Not reported
% Female Participants	1 26-50%
Mean Age of Participants	1 ≥65
Outcomes Evaluated*	1 Survival

Table 6.5. Basic characteristics of studies of nutrition interventions prior to and including initiating cancer treatment: multi-component interventions

*Studies may evaluate multiple outcomes.

Description of Interventions

Only one study examined a multi-component intervention to improve nutrition. This large four-arm study enrolled individuals willing to undergo palliative chemotherapy for gastrointestinal cancer, lung cancer, or mesothelioma, and randomized participants to 1) no dietary advice, 2) dietary advice to increase food intake to a level needed to achieve body weight gain, 3) daily nutrition support supplements with daily multivitamins, and 4) dietary advice with nutrition support supplements, each for 6 weeks.⁸⁸

Risk of Bias and Outcome Assessment

Table 6.6 summarizes risk of bias. Appendix D provides risk of bias assessments and outcome details.

Table 6.6. Risk of bias assessment for nutrition interventions prior to and including initiating cancer treatment

Intervention	Head and Neck	Gastrointestinal	Multiple Cancers	Other Cancer Types [*]
Nutrition Counseling	-	-	-	-
Dietary Supplements	-	4 Medium 6 High (N=10)	-	-
Special Diets	-	-	-	-
Route or Timing of Nutrition Interventions	-	-	-	-
Nutrition Support Including Oral Nutrition Supplements	-	2 Low 4 Medium 4 High (N=10)	-	-
Multi-component Interventions	-	-	-	-

Abbreviations: - = risk of bias not performed.

*Other cancer types include studies evaluating all remaining cancer types not included in previous categories (e.g., breast, prostate, lung cancer).

Outcomes for Dietary Supplements

Four medium-risk-of-bias studies across five publications were conducted in the surgical inpatient setting for gastrointestinal cancer patients.^{52, 55-58} Study sample sizes ranged from 60 to 195 patients. Body weight changes, adverse events, readmissions, survival, and length of hospital stay were reported by one to four studies for each outcome, but the studies were distributed across interventions that could not be aggregated. All interventions reported no difference in measured outcome except for mid-arm circumference in a study of glutamine supplementation.⁵²

Outcomes for Nutrition Support Including Oral Nutrition Supplements

Two low- and four medium-risk-of-bias studies across seven publications examined surgical inpatients for gastrointestinal cancer.^{72-74, 79, 80, 83, 84} Study sample sizes ranged from 120 to 317. Length of hospital stay, adverse events, readmission and emergency room visits, and survival were reported in one to six of the studies, but these studies were distributed across interventions that could not be aggregated. Reported results were mixed, with some studies reporting a benefit, and some reporting no difference in improving development of adverse events, length of hospital stay, readmission and emergency room visits, and survival

Variation in the Effects of Nutrition Interventions on Preventing Negative Outcomes

While studies enrolled individuals from multiple cancer types, treatments, and stages (KQ 1&2a), across the lifespan (KQ 1&2b), with varying degrees of muscle wasting (KQ 1&2c) and in those with a range of comorbid conditions (KQ 1&2d), no eligible studies specifically evaluated whether the effects of nutrition interventions on preventing negative outcomes varied across these characteristics.

Chapter 7. Nutrition Interventions After Cancer Treatment Began

Key Points

- Studies of nutrition interventions after cancer treatment began examined use of nutrition counseling, dietary supplements, special diets, route or timing of nutrition interventions, nutrition support, and multi-component interventions.
- Studies were predominately conducted among gastrointestinal and head and neck cancer populations, both in inpatient surgical and outpatient settings.
- Of studies assessed for risk of bias, 31 percent (n=8/26) of dietary supplement studies, 54 percent (n=13/24) of nutrition support studies, and 59 percent (n=16/27) of studies evaluating the route or timing of nutrition interventions were high risk of bias (poor quality), with only 10 percent (n=8/77) assessed as low risk of bias.
- Among four low- and 14 medium-risk-of-bias studies in dietary supplements, results were mixed, with most studies reporting no benefit of added dietary supplements on body weight, adverse events, length of hospital stay, or survival. Two low-risk-of-bias studies reported fewer adverse events and decreased length of hospital stay with soybean and fish oil. One medium-risk-of-bias study of glutamine reported improved body weight, reduction in adverse events, and improved treatment tolerance, while another medium-risk-of-bias study reported fewer adverse events branched-chain amino acid (BCAA)-enriched total parenteral nutrition (TPN). One medium-risk-of-bias study found improvement in postoperative complications with enteral and parenteral nutrition supplemented with omega-3 fatty acids. Three studies reported reduced length of hospital stay across diverse supplements.
- Among one low- and 10 medium-risk-of-bias studies that examined route and timing of nutrition interventions, results were mixed. The majority reported no difference for body weight, adverse events, readmissions, or death, but half reported reduced length of hospital stay. One low-risk-of-bias study reported that postoperative enteral nutrition reduced adverse events.
- Among three low- and eight medium-risk-of-bias studies that examined nutrition support (including oral nutrition), results were mixed for body weight, nutrition status, and adverse events. Two low- and three medium-risk-of-bias studies reported improved body weight or composition with postoperative nutrition support. Four of ten studies reported improvements in adverse events, and three reported improvements in length of hospital stay across diverse enteral and oral nutrition support interventions.
- Studies showed high heterogeneity across study populations, interventions, and outcomes, making meta-analysis infeasible.

Overview

This chapter includes studies that addressed Key Question (KQ) 2. Such studies examined nutrition interventions delivered simultaneously (at least in part) with cancer therapy (e.g., systemic therapy, radiation, surgery), regardless of treatment intent (e.g., curative vs. palliative). In contrast, some interventions were delivered prior to the initiation of cancer treatment and

continued after cancer treatment began (addressing KQs 1 and 2), and we present these separately in Chapter 6. For each intervention type, in order to briefly discuss what has been examined, we present results in two general sections: Eligible Studies and Description of Interventions. We assessed risk of bias for intervention/cancer types with a higher volume of studies. This allowed us to identify the greatest amount of lower risk of bias evidence concentrated according to specific nutrition interventions by cancer type. The number of studies by intervention type and cancer type for which we assessed risk of bias are bolded in Table 7.1. For interventions with low- to medium-risk-of-bias studies, we note the reported outcome directionality at the end of the chapter. Detailed information on all eligible studies can be found in Appendix E.

For KQ2, we identified 141 studies across 150 publications that examined nutrition interventions after cancer treatment began. Studies examined nutrition counseling (n=15), dietary supplements (n=34), special diets (n=8), route or timing of nutrition interventions (n=31), nutrition support including oral nutrition supplements (n=43), and multi-component interventions (n=10). Table 7.1 summarizes the characteristics of the KQ2 literature set.

Table 7.1. Studies examining use of nutrition interventions after treatment began, stratified by intervention type and cancer type

Intervention	Head and Neck	Gastrointestinal	Multiple Cancers	Other Cancer	Total
Nutrition Counseling	3	4	6	2	15
Dietary Supplements	10*	16*	3	5	34
Special Diets	0	1	2	5	8
Route or Timing of	4	24*	1	2	31
Nutrition Interventions					
Nutrition Support	4	27*	4	8	43
Including Oral					
Nutrition Supplements					
Multi-Component	2	2	2	4	10
Interventions					
Total	-	-	-	-	141

Abbreviations: - = not applicable

*Indicates risk of bias was performed.

[†]Other cancer types include studies evaluating all remaining cancer types not included in previous categories (e.g., breast, prostate, lung cancer).

Nutrition Counseling

Eligible Studies

Table 7.2 summarizes the characteristics of the literature set. We identified 15 unique studies that examined nutrition counseling during cancer treatment.⁸⁹⁻¹⁰³ Four studies enrolled individuals with gastrointestinal cancer, three enrolled those with head and neck cancers, six enrolled people from multiple cancer types, and two addressed other cancer types. Over half of the studies enrolled more than 100 participants. Ten studies reported that the intervention was administered by a dietitian/nutritionist, while four used multiple or other types of providers. Three studies did not report the provider delivering the intervention. While studies varied in their primary intent, almost all assessed the impact of the intervention on body weight or composition, and more than half assessed changes in nutrition status. The primary outcome of these studies, when reported, varied and included mortality,⁸⁹ nutrition status,^{90, 93-95, 103} symptoms,⁹¹ and body composition.¹⁰² Appendix E provides detailed evidence tables.

Table 7.2. Basic characteristics of studies of nutrition interventions after treatment began: nutrition counseling

Characteristics	Information
Total Included Studies	15 Studies
Study Sample Size	3 50-75
	4 76-100
	8 >100
Cancer Type	4 Gastrointestinal
	3 Head and neck
	6 Multiple
	2 Other cancer types [*]
Intervention Delivery Setting	3 Inpatient
	11 Outpatient
	1 Not reported
Dominant Cancer Treatment Type of	3 Chemotherapy alone
Participants	6 Radiation alone
	5 Multiple therapies
	1 Not reported
Limited to Malnourished Patients	3 Yes
	12 No
Malnourishment Screening Tool Used	1 MNA-SF
	7 Other tool/metric
	4 Multiple tools
	3 Not reported
Provider Prescribing or Delivering the	10 Dietitian/nutritionist
Intervention	1 Nurse
	2 Multiple
	1 Other
	1 Not reported
Route of Administration	Not applicable
Geographic Region of Intervention	7 Europe
	4 Asia
	3 Other
	1 Not reported
% of Participants with Stage IV Disease	3 0-25%
	2 26-50%
	251-75%
	7 Net reported
% Formale Dortisinanta	
% Female Participants	4 0-25%
	9 20-50 %
Mean Age of Participante	
Mean Age of Participants	7 50-04
	$1 \ge 0.0$
Outcomes Evaluated	12 Rody weight or composition changes
	Changes in putrition status
	a Grianyes in numum sialus 6 Adverse events
	1 Readmission or emergency room visite
	6 Survival
	1 Length of stay

*Other cancer types include studies evaluating all remaining cancer types not included in previous categories (e.g., breast, prostate, lung cancer). †Studies may evaluate multiple outcomes.

Description of Interventions

Four studies evaluated standardized intensive nutrition counseling. One compared intensive nutrition counseling versus on-demand counseling in head and neck cancer patients undergoing chemoradiotherapy, with the groups differing in the number of nutrition consultations during treatment.⁹⁵ One study compared intensive nutrition counseling following the standard nutrition protocol of the Academy of Nutrition and Dietetics (AND) versus patients who received one routine educational session across multiple cancer types during and after radiotherapy.¹⁰¹ One study also compared intensive nutrition counseling following a predetermined standard nutrition protocol (medical nutrition therapy protocol of AND) versus usual care in patients with various cancers of the gastrointestinal tract and head and neck who were undergoing radiotherapy.⁹² Finally, one study examined individual sessions with a dietitian (aimed at increasing soluble fiber and decreasing lactose in the diet) through 8 weeks of radiation therapy, versus standard care, in individuals with prostate cancer.⁹¹

A second group of studies evaluated individualized nutrition counseling. One included patients with head and neck cancer undergoing chemoradiotherapy.⁹³ A second included patients with colorectal cancer undergoing radiotherapy,⁹⁸ and the third included colorectal cancer patients undergoing chemotherapy.¹⁰² Another study compared guideline-based nutrition education counseling based on the patient's Patient-Generated Subjective Global Assessment score versus no nutrition education in patients undergoing radiation therapy across multiple cancer types.¹⁰³

A third group of studies examined face-to-face interviewing. One multicenter trial compared counseling based on face-to-face interviewing and dietary advice cards in patients across multiple cancer types undergoing chemotherapy versus usual care.⁸⁹ A second study also compared dietary advice consisting of face-to-face interviews conducted during chemotherapy in patients across multiple cancer types.⁹⁹

The remaining studies were more diverse. One compared a standardized dietary advice approach for foods to avoid and consume in patients with prostate cancer undergoing radiotherapy.⁹⁶ Another study included patients across multiple cancer types who were given nutrition consulting and meal planning during the first week postdischarge.¹⁰⁰ A whole-course nutrition management model was used in one study for patients with gastrointestinal cancer undergoing chemoradiotherapy versus a general nutrition method.⁹⁷ One study compared medical nutrition therapy involving individualized dietary planning, nutrition education, and pharmacotherapy versus general nutrition advice in patients with gastrointestinal cancer undergoing chemoradiation.⁹⁴ A final study evaluated the Eating as Treatment (EAT) program, which is based on motivational interviewing and cognitive behavioral therapy. Participants in this study were undergoing radiation for head and neck cancer.⁹⁰ Most of these non-U.S. studies were conducted in Europe and Asia.

Dietary Supplements

Eligible Studies

Table 7.3 summarizes the characteristics of the literature set. We identified 34 unique studies across 35 publications that examined dietary supplements administered after the start of cancer treatment.¹⁰⁴⁻¹³⁸ Half of the studies enrolled between 50 and 75 participants. Studies varied in their intent, but many focused on improving biomarkers of nutrition status^{107, 109-112, 122, 127, 128, 135} or immune function^{118, 131, 132, 135} rather than clinical outcomes. Some studies, however, did focus

on clinical endpoints such as changes in side effects, adverse events, length of hospital stay, and survival.^{106, 117, 119, 125, 135} Appendix E provides detailed evidence tables.

Characteristics	Information
Total Included Studies	34 Studies
Study Sample Size	17 50-75
	7 76-100
	10 ≥100
Cancer Type	10 Head and neck
	16 Gastrointestinal
	3 Multiple cancers
	5 Other cancer types*
Intervention Delivery Setting	17 Inpatient
	9 Outpatient
	1 Other
	7 Not reported
Dominant Cancer Treatment Type of	20 Surgery alone
Participants	5 Chemotherapy alone
	1 Radiation alone
	7 Multiple therapies
	1 Not reported
Limited to Malnourished Patients	5 Yes
	29 No
Malnourishment Screening Tool Used	4 Nutrition Risk Screening NRS-2002
	1 Multiple tools/metrics
	6 Other tool/metric
	23 Not reported
Provider Prescribing or Delivering the	1 Dietitian/nutritionist
Intervention	2 Nurse
	3 Physician
	1 Other provider
	4 Multiple providers
	23 Not reported
Route of Administration	11 Oral
	13 Enteral
	9 Parenteral
	1 Other
Geographic Region of Intervention	14 Europe
	19 Asia
	1 Other
% of Participants with Stage IV Disease	6 ≤10%
	5 11-25%
	1 26-50%
	0 51-75%
	376-100%
0/ Formala Dortiainanta	
% Female Participants	
	15 20-30%
	3 76 100%
	1 Net reported
Moon Ago of Portiginants	
Mean Age of Participants	2 1 50 64
	7 >65
	1 Not reported

Table 7.3. Basic characteristics of studies of nutrition interventions after treatment began: dietary supplements

Characteristics	Information
Outcomes Evaluated†	20 Body weight or composition changes
	5 Changes in nutrition status
	26 Adverse events
	3 Readmissions or emergency room visits
	12 Survival
	18 Length of stay

*Other cancer types include studies evaluating all remaining cancer types not included in previous categories (e.g., breast, prostate, lung cancer).

†Studies may evaluate multiple outcomes.

Description of Interventions

Overall, studies fell into two categories: single and multi-component supplements.

Single Supplements

Several studies examined the effects of a single supplement for immunonutrition (eicosapentaoenoic acid (EPA), omega-3 fatty acids, or glutamine or arginine-enriched nutrition), but interventions varied by population and ranged widely for intervention durations around the time of cancer surgery. In the surgically treated populations, six studies examined arginine-enhanced oral or enteral nutrition support, but these studies varied in the cancers studied, length of follow up, and comparators (e.g., standard nutrition support, different concentrations of the dietary supplement). One study compared an arginine-enhanced nutrition support supplement compared versus a standard formula after head and neck cancer surgery approximately 4 days postoperatively,¹⁰⁷ while another two studies examined higher doses¹¹¹ or longer durations.¹⁰⁹ Another two studies compared higher versus lower ratios of arginine-enhanced enteral feeding, but varied in duration (e.g., 10 versus 15 days postsurgery) and in supplement concentration.^{112, 113}

A second group of studies examined omega-3 fatty acids, fish oil, or amino acids in surgically treated patients. One study compared intravenous soybean oil plus fish oil emulsion versus soybean oil alone for 7 days postsurgery for gastrointestinal cancer.¹¹⁸ Two studies compared omega-3-fatty-acid-enriched TPN 5 days¹³⁴ and 7 days¹³⁸ postsurgery in different cancers, while a third compared omega-3-fatty-acid-enhanced oral nutrition support versus arginine-enhanced formula for 12 weeks postsurgery in head and neck cancer.¹¹³ Two additional studies compared amino or fatty acid-enriched TPN versus standard TPN for approximately 1 week after gastrointestinal cancer surgery, but with distinct primary outcomes.^{127, 137} One study compared enteral nutrition combined with parenteral nutrition enriched with omega-3 polyunsaturated fatty acids versus enteral nutrition combined with standard parenteral nutrition among gastric cancer patients after elective radical gastrectomy.¹³⁵ Another study examined the use of an oral omega-3 fatty acid supplement for 12 weeks among lung cancer patients undergoing chemotherapy.¹⁰⁴ Another compared olive oil-based lipid emulsion versus medium/long chain triglycerides for 7 days postsurgery among individuals with esophageal cancer.¹³²

In surgical patients, a final set of studies examined the use of glutamine. One study compared glutamine-enhanced total parenteral nutrition versus standard TPN for 7 days postsurgery for gastrointestinal cancer,¹²² while the other three-arm trial examined glutamine and omega-3 fatty acids for 1 week in surgically treated gastric cancer patients versus standard TPN.¹¹⁹

In individuals undergoing chemotherapy or radiation, five studies examined omega-3 fatty acids, fish oil, or amino acids in different forms. One study compared an amino acid-enriched

snack versus no snack for 2 weeks during chemoembolization for hepatocellular carcinoma,¹²⁸ while another examined the use of orally administered amino acid jelly, Inner Power®, for 21 days among breast cancer patients undergoing chemotherapy.¹¹⁷ Three additional studies examined omega-3 fatty acid supplements in varying doses, but the interventions ranged from 12 days¹²⁴ to 12 weeks¹¹⁰ in head and neck cancer, and 8 weeks in colorectal cancer.¹¹⁶

Three studies examined glutamine among individuals undergoing chemotherapy or radiation. One compared glutamine injection plus TPN versus TPN alone for individuals with advanced gastric cancer undergoing chemotherapy,¹³¹ while another examined glutamine for 7 weeks in head and neck cancer.¹²⁵ The remaining study compared glutamine-enhanced TPN versus standard TPN for 6 days after allogeneic stem cell transplantation in individuals with leukemia.¹⁰⁶

Of the two remaining studies, one five-arm study evaluated the combinations of EPA, lcartinine, thalidomide, and medroxyprogesterone acetate/megestrol acetate (MA) for up to 4 months in individuals with cancer cachexia across different tumor types;¹²⁹ and the other compared twice daily Echium oil versus placebo for 7 weeks during chemoradiation for head and neck cancer.¹²⁶

Multi-Component Supplements

Five studies examined multiple supplements postoperatively to improve nutrition and clinical outcomes. One compared EPA and gamma-linolenic acid (GLA) enriched enteral nutrition, Oxepa®, versus standard enteral nutrition for up to 21 days postsurgery among individuals with esophageal cancer.¹²³ Another evaluated medium chain triglycerides and protein-enriched enteral nutrition, Nutrison® MCT, versus standard enteral nutrition for 5 days after surgery for gastrointestinal cancer.¹³³ A third compared an arginine, omega-3 fatty acids, and ribonucleic acid (RNA) enriched whole enteral nutrition versus standard whole enteral nutrition for up to 7 days postoperatively in individuals undergoing surgery for gastric cancer.¹¹⁴ Similarly, a single study compared an arginine, glutamine, and cysteine-enhanced whole enteral nutrition, Stresson-Nutricia®, versus standard whole enteral nutrition for 10 to 15 days postoperatively in gastrointestinal cancer patients.¹²⁰ Finally, one study examined immunonutrition, Cubitan® Nutricia, twice daily for 5 days versus standard nutrition for 5 days after pancreatic cancer surgery.¹³⁰

Three studies examined the use of multiple supplements during chemotherapy or radiation therapy. One study compared arginine, glutamine, and fish oil supplementation twice a day during radiation therapy for head and neck, esophageal, and cervical cancer versus a regular diet for approximately 6 weeks.¹⁰⁵ A second study evaluated EPA and docosahexaenoic acid (DHA) enhanced nutrition support versus standard nutrition support via a percutaneous endoscopic gastrostomy (PEG) tube among head and neck cancer patients receiving chemoradiation for a maximum of 14 weeks.¹¹⁵ A third study compared omega-3 fatty acid, glutamine, probiotic, and vitamin-enriched oral nutrition support versus standard oral nutrition support for 3 months in cachexic head and neck cancer patients undergoing chemoradiation.¹³⁶

Studies in dietary supplements were predominately conducted within Europe and Asia. Overall, studies varied in the type of supplements administered, route of administration, and comparison group during the delivery of cancer treatment.

Special Diets

Eligible Studies

Table 7.4 summarizes the characteristics of the literature set. We identified eight unique studies across nine publications that examined special diets after the initiation of cancer treatment.¹³⁹⁻¹⁴⁷ Most studies had sample sizes greater than 100. The primary focus of the studies varied significantly and included changes in symptoms,¹⁴⁰ quality of life,¹⁴⁴ adverse events,^{139,145} treatment toxicity,¹⁴⁷ survival,¹⁴⁶ and multiple primary outcomes.^{142, 143} Most studies were in cancers other than gastrointestinal. Appendix E provides detailed evidence tables.

 Table 7.4. Basic characteristics of studies of nutrition interventions after treatment began: special diets

Characteristics	Information
Total Included Studies	8 Studies
Study Sample Size	1 50-75
	3 76-100
	4 >100
Cancer Type	1 Gastrointestinal
	2 Multiple cancers
	5 Other cancer types [*]
Intervention Delivery Setting	2 Inpatient
	5 Outpatient
	1 Not reported
Dominant Cancer Treatment Type of	1 Surgery alone
Participants	4 Chemotherapy alone
	2 Radiation alone
	1 Multiple therapies
Limited to Malnourished Patients	8 No
Malnourishment Screening Tool Used	8 Not reported
Provider Prescribing or Delivering the	2 Dietitian/nutritionist
Intervention	1 Physician
	5 Not reported
Route of Administration	7 Oral
	1 Other
Geographic Region of Intervention	3 North America
	3 Europe
	1 Asia
	1 Other
% of Participants with Stage IV Disease	1 26-50%
	1 76-100%
	6 Not reported
% Female Participants	1 26-50%
	2 51-75%
	3 76-100%
	2 Not reported
Mean Age of Participants	4 50-64
	2 ≥ 65
	2 Not reported
Outcomes Evaluated†	4 Body weight or composition changes
	3 Adverse events
	2 Survival
	1 Length of stay

*Other cancer types include studies evaluating all remaining cancer types not included in previous categories (e.g., breast, prostate, lung cancer).

†Studies may evaluate multiple outcomes.

Description of Interventions

Two studies evaluated the use of specific drinks. One compared the use of a glass of wine before eating along with additional oral nutrition support, such as Boost® or Ensure®, for 3 to 4 weeks during treatment for advanced cancer versus oral nutrition support supplementation alone.¹⁴¹ Another compared grape juice versus placebo prior to meals for one week following each of four cycles of chemotherapy.¹⁴⁰

Another three studies examined fasting or calorie restriction. One study compared a ketogenic diet for 3 months versus standard diet in women with metastatic breast cancer undergoing chemotherapy,^{142, 143} while the second compared a fasting mimicking diet versus regular diet at the initiation of chemotherapy for breast cancer.¹⁴⁴ The third study compared a calorie-restricted ketogenic diet and fasting versus a calorie-unrestricted diet for recurrent gliomas during radiation therapy.¹⁴⁶

The remaining studies examined modified diets by consistency or contents. One study examined early initiation of a solid versus liquid diet after surgery for gastrointestinal cancer.¹⁴⁵ A second study compared a diet containing no raw fruits or vegetables (cooked diet) versus a diet containing fresh fruits and vegetables for individuals undergoing induction therapy for acute myeloid leukemia.¹³⁹ Finally, one study compared diets that were either low-fat, modified-fat, or normal fat diet in individuals undergoing radiation therapy for pelvic malignancies.¹⁴⁷ Studies were predominately conducted in North America and Europe. Overall, studies evaluating special diets varied in approach, populations, cancer treatments, and comparison groups.

Route or Timing of Nutrition Interventions

Eligible Studies

Table 7.5 summarizes the characteristics of the literature set. We identified 31 unique studies that examined the route or timing of nutrition interventions delivered, at least in part, during cancer treatment.¹⁴⁸⁻¹⁷⁸ Samples sizes were large, with most studies enrolling over 100 individuals. Most studies examined adverse events^{62, 137, 148, 149, 154, 157, 160-162, 165-169, 171-173, 175, 176, 179} and length of hospital stay.^{62, 149, 159, 160, 162, 165-167, 169, 171-173, 175, 176, 178, 179} Twenty-four studies enrolled individuals with gastrointestinal cancers and four enrolled those with head and neck cancers. In seven studies, the intervention was delivered by a physician, one by a dietitian/nutritionist, and three by multiple providers. The primary cancer treatment was surgery alone. Appendix E provides detailed evidence tables.

Characteristics	Information
Total Included Studies	31 Studies
Study Sample Size	4 50-75
	8 76-100
	19 >100
Cancer Type	24 Gastrointestinal
	4 Head and neck
	1 Multiple cancers
	2 Other cancer types*
Intervention Delivery Setting	27 Inpatient
	1 Multiple settings
	3 Not reported

Table 7.5. Basic characteristics of studies of nutrition interventions after treatment began: route or timing of nutrition interventions

Characteristics	Information
Dominant Cancer Treatment Type of	25 Surgery alone
Participants	1 Chemotherapy alone
	4 Multiple therapies
	1 Not reported
Limited to Malnourished Patients	3 Yes
	27 No
	1 Not reported
Malnourishment Screening Tool Used	3 Nutrition Risk Screening NRS-2002
	1 Malnutrition Universal Screening Tool (MUST)
	1 MNA
	6 Other tool/metric
	2 Multiple tools
	19 Not reported
Provider Prescribing or Delivering the	1 Dietitian/nutritionist
Intervention	7 Physician
	3 Multiple
	20 Not reported
Route of Administration	12 Oral
	2 Parenteral
	12 Enteral
	5 Other
	1 Not reported
Geographic Region of Intervention	20 Asia
	9 Europe
	1 North America
	1 Other
% of Participants with Stage IV Disease	8 <10%
	1 10-24%
	2 25-49%
	2 50-74%
	1 75-100%
	17 Not reported
% Female Participants	6 0-24%
	20 25-49%
	1 50-74%
	2 75-100%
	2 Not reported
Mean Age of Participants	2 40-49
	16 50-64
	7 65+
	6 Not reported
Outcomes Evaluated†	12 Body weight or composition changes
	6 Changes in nutrition status
	28 Adverse events
	2 Readmission or emergency room visits
	10 Survival
	22 Length of stay

*Other cancer types include studies evaluating all remaining cancer types not included in previous categories (e.g., breast, prostate, lung cancer).

[†]Studies may evaluate multiple outcomes.

Description of Interventions

Sixteen studies compared enteral nutrition versus parenteral nutrition. Of those, 14 compared enteral nutrition versus parenteral nutrition in postoperative individuals; most of the 14 enrolled individuals with gastrointestinal cancer,^{62, 155, 158-162, 165, 173, 175-177, 179} with only one study investigating those with laryngeal and pharyngeal cancer.¹⁶⁷ A portion of these studies evaluated more specific populations, including gastric cancer patients with diabetes mellitus,¹⁷⁵ patients

with gastrointestinal cancer who were malnourished,¹⁷⁹ and cholangiocarcinoma patients with malignant obstructive jaundice.¹⁶² Another study compared early enteral nutrition versus early parenteral nutrition in rectal carcinoma patients who had undergone after major rectal surgery and were recovering from neoadjuvant chemoradiation.¹⁴⁹ The final study, which compared enteral versus parenteral nutrition methods, enrolled esophageal cancer patients undergoing neoadjuvant chemotherapy.¹⁵⁷

Six studies compared oral feeding versus enteral or parenteral methods. Five of these compared early oral feeding versus enteral nutrition. Two of these studies evaluated esophageal cancer patients after minimally invasive esophagectomy and the use of early oral feeding versus a jejunostomy tube¹⁴⁸ and a nasoenteral feeding tube.¹⁷¹ Two studies compared early oral feeding versus a nasogastric tube; one enrolled postoperative gastric cancer patients¹⁵⁴ and the other enrolled patients who underwent elective colorectal resection.¹⁵³ Another study that investigated early oral feeding and enteral nutrition was conducted in patients who underwent total laryngectomy and were randomized to receive early oral nutrition or a tracheosophageal puncture.¹⁶⁹ Finally, one study compared total parenteral nutrition versus an oral diet in breast cancer patients undergoing high-dose chemotherapy and hematopoietic cell transplantation.¹⁶⁶

Three studies evaluated the effects of different enteral or parenteral methods. One compared a percutaneous endoscopic gastrostomy tube versus a nasogastric tube in head and neck cancer patients undergoing curative treatment including radical surgery with adjuvant radiotherapy, chemoradiotherapy, or concurrent chemo- and radiation therapy.¹⁶⁸ Another examined the use of jejunostomy feeding or nasogastric feeding in patients after a minimally invasive McKeown esophagectomy.¹⁷² A third compared a peripherally inserted central catheter versus a central venous catheter in gastric cancer patients after a radical gastrectomy.¹⁷⁸

Finally, five studies examined the timing of oral feeding after cancer surgery. One compared early (postoperative day 1) versus delayed (postoperative day 4) oral feeding after gastrectomy,¹⁷⁴ while two others examined early versus delayed feeding but used resolution of ileus to initiate delayed feeding in gastrointestinal¹⁶³ and gynecologic cancers.¹⁸⁰ Another study compared early feeding on postoperative day 0 compared versus standard delayed feeding until first stool.¹⁵² The last study compared early (postoperative day 1) versus delayed (postoperative day 7) in colorectal cancer surgery.¹⁷⁰ Studies were predominately conducted in Asia and Europe. Overall, studies varied in terms of duration of intervention, cancer therapy received, comparisons, and primary outcomes.

Nutrition Support Including Oral Nutrition Supplements

Eligible Studies

Table 7.6 summarizes the characteristics of the literature set. We identified 43 unique studies across 49 publications that examined nutrition support interventions during cancer treatment.¹⁸¹⁻²²⁹ Studies varied in sample size, with just under half enrolling more than 100 participants. The primary focus of these studies varied considerably; including changes in postoperative complications,^{199-204, 212, 217} biomarkers of nutrition status,^{186, 188, 216, 225} body weight changes,^{190, 206, 222} quality of life,^{191, 220} and length of hospital stay,^{204, 218} among others. Most studies enrolled gastrointestinal patients. Only in some studies was the intervention delivered by a dietitian/nutritionist (n=6), and across studies cancer treatment varied substantially. Appendix E provides detailed evidence tables.
Characteristics	Information
Total Included Studies	
Otarinciuded Studies	
Study Sample Size	9 50-75
	14 76-100
	20 >100
Cancer Type	4 Head and neck
	27 Gastrointestinal
	4 Multiple cancers
	8 Other cancer types
Intervention Delivery Setting	21 Inpatient
	14 Outpatient
	4 Other
	4 Not reported
Dominant Cancer Treatment Type of	20 Surgery alone
Participants	6 Chemotherapy alone
	3 Radiation alone
	14 Multiple therapies
Limited to Malnourished Patients	8 Yes
	35 No
Malnourishment Screening Tool Used	8 Nutrition Risk Screening NRS-2002
	1 MNA
	3 Multiple Tools
	9 Other tools/metrics
	22 Not reported
Provider Prescribing or Delivering the	5 Dietitian/nutritionist
Intervention	4 Physician
	11 Multiple providers
	23 Not reported
Route of Administration	26 Oral
	12 Enteral
	4 Parenteral
	1 Other
Geographic Region of Intervention	12 Europe
	27 Asia
	1 North America
	3 Other
% of Participants with Stage IV Disease	22 ≤10%
	4 11-50%
	5 51-75%
	12 Not reported
% Female Participants	11 0-25%
	24 26-50%
	6 51-75%
	1 76-100%
	1 Not reported
Mean Age of Participants	2 <50
5	20 50-64
	11 ≥65
	10 Not reported
Outcomes Evaluated+	26 Body weight or composition changes
	14 Changes in nutrition status
	32 Adverse events
	4 Readmissions or emergency room visits
	12 Survival
	13 Length of hospital stay

 Table 7.6. Basic characteristics of studies of nutrition interventions after treatment began:

 nutrition support including oral nutrition supplements

*Other cancer types include studies evaluating all remaining cancer types not included in previous categories (e.g., breast, prostate, lung cancer).

[†]Studies may evaluate multiple outcomes.

Description of Interventions

Nutrition support was delivered in three general intervention types: oral or enteral nutrition after surgery, oral or enteral nutrition during chemotherapy and/or radiation, and total parenteral nutrition across cancer treatments.

Oral or Enteral Nutrition After Surgery

Twenty-two studies examined oral or enteral nutrition support after cancer surgery, and they varied in the route and contents of nutrition support delivered as well as the timing of the comparisons. Four studies examined immunonutrition after surgery. One compared two different oral nutrition supplements, enhanced with EPA and DHA, MEIN® versus a low osmotic load oligomeric formula, HINE E-GEL®, for 6 days after surgery for esophageal cancer.²¹² One compared immunonutrition, Reconvan, enhanced with glutamine, arginine, and omega-3 fatty acids versus standard enteral nutrition for up to 7 days postsurgery for gastrointestinal cancer, 199-²⁰¹ while another examined a similar formula for 6 days postoperatively.²¹⁷ The fourth study compared immunonutrition, Stresson®, versus standard nutrition support or TPN for 7 days postoperatively across multiple cancers.^{202, 203} Another study compared early enteral nutrition versus parenteral nutrition in colon carcinoma patients.²¹¹ Among the remaining studies, the nutrition support varied in contents and quantity, including nutrition support supplements such as Elental[®], ^{193, 194} Jevity[®], ²⁰⁴ Racol[®], ²¹⁰ Nutren[®], ^{208, 209} and Nutrison Fibre[®]. ²²⁵ Additionally, studies examined standard nutrition support for widely ranging durations, from 3 days,²⁰⁴ to 2 weeks,²²⁶ to 1,^{187, 223} 2,^{186, 191, 193, 194} 3,^{209, 210, 228, 230} and 6²¹³ months postsurgery; one assessed administration until oral diet commenced.¹⁸³ Comparison populations varied substantially as well; some studies compared interventions with standard of care, which was often described as dietary advice,^{208, 209} while others compared to TPN alone.¹⁸⁸ While most interventions were delivered to head and neck and gastrointestinal cancers patients, two enrolled patients with urologic¹⁸⁹ and gynecologic¹⁸² cancers. Finally, one distinct study evaluated use of iEAT®, food disintegrated on the tongue, versus standard care after surgery for gastrointestinal cancer.²¹⁸

Oral or Enteral Nutrition During Chemotherapy and/or Radiation

Eighteen studies examined oral or enteral nutrition support during chemotherapy or radiation, and they varied in the route and contents of nutrition support delivered as well as the timing of comparisons. Three studies examined EPA enriched oral nutrition support, but varied by type of cancer, including pancreatic,¹⁸¹ lung,²¹⁶ and multiple types of advanced cancer;¹⁹⁷ these studies also varied substantially in duration from 5 weeks¹⁸¹ to 3 months.^{197,216} Another three studies examined hyper-protein-enhanced nutrition support,^{190, 229} Immax®, and omega-3 fatty acid-enriched oral nutrition support¹⁸⁵ for diverse populations and durations.

The remaining studies evaluated standard nutrition support. Four examined nutrition support for the duration of radiation in head and neck,^{192, 214} colorectal,²¹⁵ and lower gynecological and gastrointestinal cancers.²⁰⁷ Another six examined oral nutrition support during chemotherapy for leukemia,²²⁷ esophageal,¹⁹⁸ gastrointestinal,^{219, 229} pancreatic,²⁰⁵ and head and neck cancer,⁶³ and these varied in duration due to the standard of care for chemotherapy regimens across cancer types. Additionally, three studies examined nutrition support during combined chemoradiation. One study compared oral nutrition support, Niufutai®, for the duration of chemoradiation versus regular diet in individuals with advanced head and neck cancer.¹⁹⁵ The other two studies examined use of oral nutrition support versus dietary advice in individuals undergoing

chemoradiation.^{206, 224} A final study compared early enteral nutrition support with chemotherapy versus only chemotherapy in gastric cancer patients after a radical gastrectomy.²²¹

Total Parenteral Nutrition

Four studies examined TPN across cancer and treatment types. One compared oral feeding supplemented by TPN versus oral feeding alone in individuals with advanced cancer cachexia who were undergoing or had recently completed cancer treatment.¹⁸⁴ A second compared TPN, Olimel®, for 4 to 8 days postsurgery versus isotonic fluids in individuals with gastric cancer.¹⁹⁶A third compared TPN, Nutriflex®, versus standard diet for 5 days postoperatively in individuals undergoing bladder cancer surgery.²²⁰ A final study compared TPN plus enteral nutrition support, Nutrison®, for approximately 7 days postsurgery for esophageal cancer versus enteral nutrition support alone.²²²

Multi-Component Interventions

Eligible Studies

Table 7.7 summarizes the characteristics of the literature set. Ten studies across 11 publications examined multi-component interventions initiated after cancer treatment began.²³¹⁻²⁴¹ The primary focus was to assess laboratory markers of immune and nutrition status,^{232, 233, 235, 239, 241} body weight changes,²³⁶⁻²³⁸ and symptoms.²³¹ Half of the studies enrolled over 100 participants, and interventions were most often delivered or prescribed by a dietitian or nutritionist. Most studies were conducted outside of the United States. Appendix E provides detailed evidence tables.

Characteristics	Information		
Total Included Studies	10 Studies		
Study Sample Size	3 50-74		
	2 75-100		
	5 >100		
Cancer Type	2 Head and neck		
	2 Gastrointestinal		
	2 Multiple cancers		
	4 Other cancer types*		
Intervention Delivery Setting	5 Outpatient		
	4 Multiple settings		
	1 Not reported		
Dominant Cancer Treatment Type of	1 Surgery Alone		
Participants	5 Chemotherapy alone		
	3 Multiple therapies		
	1 Not reported		
Limited to Malnourished Patients	3 Yes		
	7 No		
Malnourishment Screening Tool Used	3 Nutrition Risk Screening NRS-2002		
	2 Other tool		
	5 Not assessed		
Provider Prescribing or Delivering the	6 Dietitian/nutritionist		
Intervention	1 Nurse		
	2 Multiple		
	1 Not reported		

 Table 7.7. Basic characteristics of studies of nutrition interventions after treatment began: multicomponent interventions

Characteristics	Information			
Route of Administration	5 Oral			
	4 Other			
	1 Enteral			
Geographic Region of Intervention	4 Asia			
	4 Europe			
	1 North America			
	1 Other			
% of Participants with Stage IV Disease	2 ≤10%			
	2 11-50%			
	1 51-75%			
	5 Not reported			
% Female Participants	1 0-25%			
	5 26-50%			
	1 51-75%			
	3 76-100%			
Mean Age of Participants	3 <50			
	4 50-64			
	2 ≥65			
	1 Not reported			
Outcomes Evaluated†	7 Body weight or composition changes			
	5 Changes in nutrition status			
	3 Adverse outcomes			
	1 Readmissions or emergency room visits			
	3 Survival			
	2 Length of hospital stay			

*Other cancer types include studies evaluating all remaining cancer types not included in previous categories (e.g., breast, prostate, lung cancer).

†Studies may evaluate multiple outcomes.

Description of Interventions

Multi-component interventions varied in the interventions administered, cancer type, comparators, and cancer treatments. Three studies examined individualized diet or nutrition plans combined with additional interventions such as education or counseling. One study compared a one-time face-to-face nutrition education session with a trained dietitian along with an individualized diet versus standard care in women undergoing chemotherapy for breast cancer.²³¹ One study compared an intensive individualized nutrition and education advice during hospitalization for gastric cancer.²⁴¹ Another study compared nutrition education combined with oral nutrition supplementation (Ensure®) versus nutrition education alone in ovarian cancer patients undergoing chemotherapy.²³⁵ Finally, one study compared an individualized nutrition support or protein rich snacks for 3 months versus standard care in individuals with cancer.²⁴⁰

The remaining studies varied widely in interventions administered. One compared individual recipes developed by patients, caregivers, and clinical specialists along with supplemental enteral or parenteral nutrition to meet energy requirements versus nutrition guidance alone in individuals undergoing chemotherapy for colorectal cancer.²³³ One three-arm study examined 1) a calcium rich diet, 2) a calcium rich diet plus exercise, or 3) a calcium rich, low-fat and high-fruit and vegetable diet plus exercise in women with breast cancer undergoing chemotherapy.²³² Another study compared a prophylactic percutaneous endoscopic gastrostomy (PEG) tube for early enteral feeding along with nutrition advice versus standard care in patients undergoing treatment for head and neck cancer.^{236, 237} A fourth compared oral, enteral, or TPN nutrition support during hospitalization followed by nutrition education support and provision of oral nutrition support

postdischarge versus standard care among individuals hospitalized for hematopoietic stem cell transplant.²³⁸ Finally, one study examined in-hospital nutrition education and early enteral nutrition support (24 hours after surgery) that continued 6 weeks postoperatively versus regular enteral nutrition support (48 hours after surgery) in individuals treated for laryngeal cancer.²³⁹

Risk of Bias and Outcome Assessment

Table 7.8 summarizes risk of bias assessments. Appendix E provides risk of bias assessments and outcomes.

Intervention	Head and	Gastrointestinal	Multiple Cancers	Other Cancer
	Neck			Types
Nutrition Counseling	-	-	-	-
Dietary Supplements	6 Medium	4 Low	-	-
	4 High	8 Medium		
	(N=10)	4 High		
		(N=16)		
Special Diets	-	-	-	-
Route or Timing of Nutrition	-	1 Low	-	-
Interventions		10 Medium		
		13 High		
		(N=24)		
Nutrition Support Including Oral	-	3 Low	-	-
Nutrition Supplements		8 Medium		
		16 High		
		(N=27)		
Multi-Component Interventions	-	-	-	-

Table 7.8. Risk of bias assessment for nutrition interventions after treatment began

Abbreviations: - = risk of bias not performed.

*Other cancer types include studies evaluating all remaining cancer types not included in previous categories (e.g., breast, prostate, lung cancer).

Outcomes for Dietary Supplements

Four low- and 14 medium-risk-of-bias studies evaluated use of dietary supplements during cancer treatment.^{109-112, 114, 116, 118, 120, 122, 124-128, 132, 133, 135, 138} Studies were one-third head and neck cancers (n=6/18) and two-thirds gastrointestinal cancers (n=12/18). Four of the six head and neck cancer studies were surgical only populations, while nine of the 11 gastrointestinal cancer populations were surgical. Study sample sizes were small to large, from 50 to 229 patients, and12 studies were under 100 patients. All studies of head and neck cancers reported body weight and composition changes but less than one-half (5/12) of the gastrointestinal cancer studies did so. Gastrointestinal cancer studies focused more on length of hospital stay as an outcome (8/12). Additionally, studies examined a wide range of interventions and outcome timing with little overlap, creating concerns about whether grouping or aggregation would be appropriate. Outcomes were mixed, with most studies reporting no benefit of added dietary supplements on body weight, adverse events, length of hospital stay, or survival. Two low-risk-of-bias studies reported benefits with soybean and fish oil for fewer adverse events and decreased length of hospital stay.^{118, 138} One medium-risk-of-bias study of glutamine reported improved body weight, reduction in adverse events, reduction in length of hospital stay and

improved treatment tolerance,¹²⁵ while another medium-risk-of-bias study reported a benefit of BCAA-enriched TPN for reducing adverse events.¹²⁷ Another medium-risk-of-bias study found improvement in postoperative complications with enteral and parenteral nutrition supplemented with omega-3 fatty acids.¹³⁵ Three studies reported improvements in length of hospital stay across diverse supplements.^{111, 114, 133}

Outcomes for Route or Timing of Nutrition Interventions

One low- and 10 medium-risk-of-bias studies evaluated the route or timing of nutrition interventions during cancer treatment.^{148, 150, 151, 153, 158, 162, 171, 172, 174-176} Populations were overwhelmingly surgical inpatients (10 of 11 studies). Sample sizes were midrange, from 60 to 317 patients. The most common outcome categories were body weight (4 studies), adverse events (9 studies), length of hospital stay (8 studies), and death (4 studies). Studies examined a wide range of populations, interventions, and outcome timing, with little overlap, creating concerns about whether grouping or aggregation would be appropriate. Studies reported mixed results, with the majority finding no difference based on route or timing of nutrition interventions for changes in body weight, adverse events, readmissions, or death, while half showed an improvement in length of hospital stay. One low-risk-of-bias study reported that postoperative enteral nutrition reduced adverse events and length of hospital stay.¹⁵⁰

Outcomes for Nutrition Support (Including Oral Nutrition Supplements)

Three low- and eight medium-risk-of-bias studies evaluated use of nutrition support during cancer treatment.^{183, 188, 199-203, 205, 208-210, 222, 223, 225, 228} Studies were overwhelmingly inpatients receiving oral or enteral nutrition after surgery (10 of 11 studies). Study sample sizes were midrange, from 78 to 1003. The most common outcome categories were body body weight/composition changes (5 studies), adverse events (10 studies), and length of hospital stay (6 studies). Studies were distributed across a wide range of populations, interventions, and outcome timing, with little overlap, creating concerns about whether grouping or aggregation would be appropriate. Two low-^{208, 228} and three medium-risk-of-bias studies^{210, 222, 223} reported improvements in body weight or composition with postoperative nutrition support. Four out of ten studies^{183, 188, 205, 225} reported improvements in adverse events and three reported improvements in length of hospital stay^{183, 188, 225} across diverse enteral and oral nutrition support interventions.

Variation in the Effects of Nutrition Interventions on Preventing Negative Outcomes

While studies enrolled individuals from multiple cancer types, treatments, and stages (KQ2a), across the lifespan (KQ2b), with varying degrees of muscle wasting (KQ2c) and in those with a range of comorbid conditions (KQ2d), no eligible studies specifically evaluated whether the effects of nutrition interventions on preventing negative outcomes varied across these characteristics.

Chapter 8. Effect of Nutrition Interventions on Symptoms

Key Points

- Studies of the effects of nutrition interventions on symptoms included nutrition counseling, use of dietary supplements, special diets, route or timing of nutrition interventions, and nutrition support (including oral nutrition supplements).
- Studies were predominately conducted among gastrointestinal and head neck cancer populations, both in inpatient surgical and outpatient settings.
- Of studies assessed for risk of bias, 40 percent (n=2/5) of dietary supplement studies and 55 percent (n=11/20) of nutrition support studies were high risk of bias (poor quality), with only 20 percent (n=5/20) low risk of bias.
- Among Five low- and four medium-risk-of-bias studies of nutrition support, results were mixed. Two studies showed mixed results for functional status, with one finding a benefit and one no difference. Two low-risk-of-bias studies reported improvement in nausea for individuals receiving preoperative oral carbohydrate drinks. A third low-risk-of-bias study reported improvement in treatment tolerance and symptoms after use of oral nutrition supplements and dietary advice.
- Studies showed high heterogeneity across study populations, interventions, and outcomes, making meta-analysis infeasible.

Overview

This chapter includes studies that addressed Key Question (KQ) 3. Such studies examined the effect of nutrition interventions delivered prior to or during cancer treatment on associated cancer and treatment-related symptoms such as fatigue, nausea and vomiting, appetite, physical and functional status, and quality of life. Because studies often examined the impact of nutrition interventions on preventing and treating negative treatment outcomes either prior to (KQ1) or during (KQ2) cancer treatment as well as the impact on improving symptoms (KQ3), many studies in this chapter have also been presented in Chapters 5 - 7.

For each intervention type, in order to briefly discuss what has been examined, we present results in two general sections: Eligible Studies and Description of Interventions. We assessed risk of bias for intervention/cancer types with a higher volume of studies. This allowed us to identify the greatest amount of lower risk of bias evidence concentrated according to specific nutrition interventions by cancer types. The number of studies by intervention type and cancer type for which we assessed risk of bias is indicated in Table 8.1. For interventions with low- to medium-risk-of-bias studies, we note the reported outcome directionality at the end of the

chapter. For all studies, we present a brief discussion of what has been examined. Detailed information on all eligible studies can be found in Appendix F.

For KQ3, we identified 79 studies across 83 publications that examined nutrition interventions after cancer treatment began. Studies examined nutrition counseling (n=12), dietary supplements (n=10), special diets (n=10), route or timing of nutrition interventions (n=8), nutrition support (including oral nutrition supplements) (n=32), and multi-component interventions (n=7). Table 8.1 summarizes the characteristics of the KQ3 literature set.

 Table 8.1. Studies examining effect of nutrition interventions on symptoms, stratified by intervention type and cancer type

Intervention	Head and Neck	Gastrointestinal	Multiple Cancers	Other Cancer Types†	Total
Nutrition Counseling	1	4	4	3	12
Dietary Supplements	1	5*	3	1	10
Special Diets	0	1	2	7	10
Route or Timing of Nutrition Interventions	2	4	1	1	8
Nutrition Support Including Oral Nutrition Supplements	5	20*	3	4	32
Multi-Component Interventions	2	0	3	2	7
Total	-	-	-	-	79

Abbreviations: - = Not applicable.

*Risk of bias was performed.

[†]Other cancer types include studies evaluating all remaining cancer types not included in previous categories (e.g., breast, prostate, lung cancer).

Nutrition Counseling

Eligible Studies

Table 8.2 summarizes the characteristics of the literature set. We identified 12 unique studies examining nutrition counseling on improving cancer symptoms prior to or during cancer treatment.^{90-92, 94, 96-99, 101-103, 242} Four of the studies were in populations with gastrointestinal cancer,^{94, 97, 98, 102} while the remaining eight studies were conducted in populations with head and neck cancer, multiple cancer types, or those that fell into the defined 'other cancer types' group. Most studies had study populations greater than 100. Most studies reported that the intervention was administered by a dietitian/nutritionist. While the primary focus of the studies varied, almost all assessed the impact of the intervention on quality of life; about half evaluated symptoms or functional status. Appendix F provides detailed evidence tables.

 Table 8.2. Basic characteristics of studies of effect of nutrition interventions on symptoms:

 nutrition counseling

Characteristics	Information
Total Included Studies	12 Studies
Study Sample Size	5 76-100
	7 >100
Cancer Type	4 Gastrointestinal
	1 Head and neck
	4 Multiple
	3 Other cancer types [*]

Characteristics	Information	
Intervention Delivery Setting	2 Inpatient	
	7 Outpatient	
	3 Not reported	
Dominant Cancer Treatment Type of	3 Chemotherapy alone	
Participants	6 Radiation alone	
	3 Multiple therapies	
Limited to Malnourished Patients	2 Yes	
	10 No	
Screening Tool Used	6 Other tool	
	2 Multiple tools	
	4 Not reported	
Provider Prescribing or Delivering the	8 Dietitian/nutritionist	
Intervention	1 Nurse	
	1 Multiple	
	1 Other	
	1 Not reported	
Route of Administration	Not applicable	
Geographic Region of Intervention	4 Europe	
	4 Asia	
	3 Other	
	1 Not reported	
% of Participants with Stage IV Disease	1 0-25%	
	1 26-50%	
	1 51-75%	
	1 76-100%	
	8 Not reported	
% Female Participants	4 0-25%	
	7 26-50%	
	1 76-100%	
Mean Age of Participants	6 <65	
	6 ≥65	
KQ3 Outcomes Evaluated [†]	3 Treatment tolerance	
	9 Quality of life	
	6 Symptoms	
	6 Functional status	

Abbreviation: KQ=Key Question.

*Other cancer types include studies evaluating all remaining cancer types not included in previous categories (e.g., breast, prostate, lung cancer).

†Studies may evaluate multiple outcomes.

Description of Interventions

Three studies evaluated standardized intensive nutrition counseling. One compared intensive nutrition counseling following the standard nutrition protocol of the Academy of Nutrition and Dietetics (AND) versus patients who received routine education across multiple cancer types during and after radiotherapy.¹⁰¹ One study compared intensive nutrition counseling following a predetermined standard nutrition protocol (medical nutrition therapy protocol of AND) versus usual care in patients with cancers of the gastrointestinal tract and head and neck undergoing radiotherapy.⁹² Finally, one study examined individual sessions with a dietitian (aimed at increasing soluble fiber and decreasing lactose in the diet) through 8 weeks of radiation therapy, versus standard care, in individuals with prostate cancer.⁹¹

Another group of studies evaluated individualized nutrition counseling or face-to-face interviewing. One included patients with colorectal cancer undergoing radiotherapy,⁹⁸ while the second also included colorectal cancer patients, but they were undergoing chemotherapy.¹⁰² A third study compared dietary advice consisting of face-to-face interviews conducted during

chemotherapy in patients across multiple cancer types.⁹⁹ Another study compared guidelinebased nutrition education counseling based on the patient's Patient-Generated Subjective Global Assessment score versus no nutrition education in patients undergoing radiation therapy across multiple cancer types.¹⁰³ Finally, one study evaluated the effect of face-to-face dietary counseling before chemotherapy sessions among individuals with breast cancer.²⁴²

The remaining studies were more diverse. One compared a standardized dietary advice approach for foods to avoid or consume in patients with prostate cancer undergoing radiotherapy.⁹⁶ One study compared a whole-course nutrition management model versus a general nutrition method for patients with gastrointestinal cancer undergoing chemoradiotherapy.⁹⁷ One study compared medical nutrition therapy that consisted of individualized dietary planning, nutrition education, as well as pharmacotherapy versus general nutrition advice in patients with gastrointestinal cancer undergoing chemoradiation.⁹⁴ A final study evaluated the Eating as Treatment (EAT) program which is based on motivational interviewing and cognitive behavioral therapy. Participants in this study were undergoing radiation for head and neck cancer.⁹⁰ Studies were not U.S.-based; most were conducted in Europe and Asia.

Dietary Supplements

Eligible Studies

Table 8.3 summarizes the characteristics of the literature set. We identified ten unique studies that examined dietary supplements initiated prior to or during cancer treatment and evaluated the impact of the intervention on symptoms.^{26, 47, 55, 59, 115-117, 129, 131, 133} Over half of the studies enrolled 100 or more individuals, and while most studies assessed the impact of the nutrition intervention on clinical outcomes, other outcomes, such as symptoms (nausea and fatigue), body weight loss, quality of life, and treatment tolerance were also included. Other studies examined multiple outcomes but were less clear on the primary intent. Most studies (n=5) enrolled individuals with gastrointestinal cancers. Dietary supplements were prescribed by a dietitian/nutritionist in only two study, and less than half of the study populations were treated predominantly with chemotherapy. Appendix F provides detailed evidence tables.

supplements	
Characteristics	Information
Total Included Studies	10 Studies
Study Sample Size	3 50-75
	1 76-100
	6 >100
Cancer Type	1 Head & Neck
	5 Gastrointestinal
	3 Multiple cancers
	1 Other cancer types*
Intervention Delivery Setting	3 Inpatient
	4 Outpatient
	2 Other
	1 Not reported
Dominant Cancer Treatment Type of	2 Surgery alone
Participants	4 Chemotherapy alone
	3 Multiple therapies
	1 Not reported

 Table 8.3. Basic characteristics of studies of effect of nutrition interventions on symptoms: dietary supplements

Characteristics	Information	
Limited to Malnourished Patients	2 Yes	
	8 No	
Malnourishment Screening Tool Used	1 MUST	
	3 Other tool/metric	
	1 Multiple tools	
	5 Not reported	
Provider Prescribing or Delivering the	2 Dietitian/nutritionist	
Intervention	1 Nurse	
	1 Physician	
	1 Multiple providers	
	5 Not reported	
Route of Administration	7 Oral	
	2 Enteral	
	1 Parenteral	
Geographic Region of Intervention	3 Europe	
	6 Asia	
	1 North America	
% of Participants with Stage IV Disease	1 10-25%	
	5 75-100%	
	4 Not reported	
% Female Participants	2 0-25%	
	6 26-50%	
	1 76-100%	
	1 Not reported	
Mean Age of Participants	6 <65	
	2 ≥65	
	2 Not reported	
KQ3 Outcomes Evaluated*	2 Treatment tolerance	
	7 Quality of life	
	5 Symptoms	
	2 Functional status	

Abbreviation: KQ=Key Question.

*Other cancer types include studies evaluating all remaining cancer types not included in previous categories (e.g., breast, prostate, lung).

Description of Interventions

Studies examined various types of supplements administered, timing, and route of administration, and comparators. Almost all of the studies included a measure of quality of life,^{47, 115-117, 129, 131} while fewer investigated cancer treatment symptom relief,^{117, 129} treatment tolerance,^{47, 59} or functional status.^{59, 115}

Seven studies examined a single supplement. Interventions varied by population with a wide range of durations and treatment contexts. One study compared supplementation with omega-3-fatty-acid-enriched nutrition versus a placebo while participants received chemotherapy for gastrointestinal cancer.¹¹⁶ Another study compared an oral nutrition supplement with an eicosapentaenoic acid (EPA)-enriched supplement versus a standard diet for a population with multiple cancers.¹¹⁵ A third examined an orally administered amino acid jelly, Inner Power®, for 21 days among breast cancer patients undergoing chemotherapy.¹¹⁷ Another compared a glutamine injection plus total parenteral nutrition (TPN) versus TPN alone for individuals with advanced gastric cancer undergoing chemotherapy.¹³¹ One study compared nutrition counseling only versus nutrition counseling with oral whey protein supplementation for 3 months in malnourished advanced cancer patients.⁴⁷ A single study compared amino acid supplements 1 week prior and for 5 weeks during chemotherapy versus standard care for gastrointestinal

cancer.⁵⁹ Finally, a five-arm study evaluated the combinations of EPA, 1-carnitine, thalidomide, and medroxyprogesterone acetate/megestrol acetate (MA) for up to 4 months in individuals with cancer cachexia across different tumor types.¹²⁹

Three studies examined multiple supplements to improve cancer treatment-related symptoms. One study investigated perioperative oral protein supplementation rich in arginine and omega-6 versus placebo in patients with gastrointestinal cancer undergoing surgery.⁵⁵ Another study compared and oral immunonutrition formula with fatty acids, arginine, fiber, and nucleotides versus a standard enteral formula in head and neck cancer patients undergoing chemotherapy.²⁶ The final study compared medium chain triglycerides and protein-enriched enteral nutrition, Nutrison® MCT, versus standard enteral nutrition for 5 days after surgery for gastrointestinal cancer.¹³³

Special Diets

Eligible Studies

Table 8.4 summarizes the characteristics of the literature set. We identified 10 unique studies across 11 publications that examined the effects of special diets prior to or during cancer treatment on treatment-related symptoms.^{139-147, 243, 244} The majority of studies enrolled patients with cancers other than gastrointestinal cancer, such as breast cancer^{142, 143} and gliomas.¹⁴⁶ 60 percent of the studies enrolled more than 100 participants. Only two studies reported that the intervention had been administered by a dietitian/nutritionist, and one was administered by a physician. Almost all studies assessed quality of life and symptom status, with fewer assessing the impact of the intervention on functional status. Appendix F provides detailed evidence tables.

Characteristics	Information
Total Included Studies	10 Studies
Study Sample Size	1 50-75
	3 76-100
	6>100
Cancer Type	1 Gastrointestinal
	2 Multiple
	7 Other cancer types*
Intervention Delivery Setting	2 Inpatient
	7 Outpatient
	1 Not reported
Dominant Cancer Treatment Type of	5 Chemotherapy alone
Participants	2 Radiation alone
	1 Surgery alone
	2 Multiple therapies
Limited to Malnourished Patients	10 No
Screening Tool Used	10 Not reported
Provider Prescribing or Delivering the	2 Dietitian/nutritionist
Intervention	1 Physician
	1 Other
	6 Not reported
Route of Administration	9 Oral
	1 Other
Geographic Region of Intervention	3 North America
	3 Europe
	3 Asia
	1 Other

Table 8.4. Basic characteristics of studie	ies of effect of nutrition interventions on symptoms
special diets	

Characteristics	Information
% of Participants with Stage IV Disease	1 0-10%
	1 26-50%
	1 100%
	7 Not reported
% Female Participants	1 26-50%
	2 51-75%
	5 76-100%
	2 Not reported
Mean Age of Participants	6 <65
	2 ≥65
	2 Not reported
KQ3 Outcomes Evaluated†	7 Quality of life
	7 Symptoms
	3 Functional status

Abbreviation: KQ=Key Question.

*Other cancer types include studies evaluating all remaining cancer types not included in previous categories (e.g., breast, prostate, lung cancer).

†Studies may evaluate multiple outcomes.

Description of Interventions

Two studies evaluated the use of specific drinks. One compared the use of a glass of wine before eating along with additional oral nutrition support, such as Boost® or Ensure®, for 3 to 4 weeks during treatment for advanced cancer, versus oral nutrition support supplementation alone.¹⁴¹ Another compared grape juice versus placebo prior to meals for one week following each of four cycles of chemotherapy.¹⁴⁰

Another four studies examined fasting or calorie restriction. One study compared a ketogenic diet for 3 months versus standard diet in women with metastatic breast cancer undergoing chemotherapy,^{142, 143} while the second compared a fasting mimicking diet versus standard diet at the initiation of chemotherapy for breast cancer.¹⁴⁴ The third study compared a calorie restricted ketogenic diet and fasting versus a calorie unrestricted diet for recurrent gliomas during radiation therapy.¹⁴⁶A final study compared calorie restriction with synbiotics versus placebo to evaluate the impact on quality of life in women with breast cancer.²⁴⁴

The remaining studies examined modified diets by consistency or contents. One study compared early initiation of a solid versus liquid diet after surgery for gastrointestinal cancer.¹⁴⁵ A second study compared a diet containing no raw fruits or vegetables (cooked diet) versus a diet containing fresh fruits and vegetables for individuals undergoing induction therapy for acute myeloid leukemia.¹³⁹ A third study evaluated use of ginger on chemotherapy-induced nausea and vomiting in women with breast cancer.²⁴³ Finally, one study compared a low-fat, modified-fat, and a normal-fat diet in individuals undergoing radiation therapy for pelvic malignancies.¹⁴⁷ Most of these studies were conducted in North America, Europe, and Asia. Overall, studies evaluating special diets varied in approach, populations, cancer treatments, and comparators.

Route or Timing of Nutrition Interventions

Eligible Studies

Table 8.5 summarizes the characteristics of the literature set. We identified eight studies that examined whether the route or timing of nutrition interventions delivered, at least in part, during cancer treatment, affected cancer or cancer treatment-related symptoms.^{63, 68, 69, 153, 166, 168, 171, 172}

Half of the studies enrolled over 100 individuals. The majority of these studies examined adverse events^{166, 168, 171, 172} and length of hospital stay.^{166, 171, 172} Almost all studies enrolled individuals with gastrointestinal or head and neck cancers. Nutrition interventions were delivered by a dietitian/nutritionist in one study. The primary cancer treatment was surgery alone. Appendix F provides detailed evidence tables.

Characteristics	Information
Total Included Studies	8 Studies
Study Sample Size	1 50-75
	3 76-100
	4 >100
Cancer Type	4 Gastrointestinal
	2 Head and neck
	1 Multiple cancer types
	1 Other cancer types*
Intervention Delivery Setting	5 Inpatient
	2 Multiple settings
	1 Not reported
Dominant Cancer Treatment Type of	5 Surgery alone
Participants	3 Multiple therapies
Limited to Malnourished Patients	1 Yes
	6 No
	1 Not reported
Malnourishment Screening Tool Used	1 Nutrition Risk Screening NRS-2002
	1 Other tool/metric
	1 Multiple tools
	5 Not reported
Provider Prescribing or Delivering the	1 Dietitian/nutritionist
Intervention	1 Physician
	2 Multiple
	4 Not reported
Route of Administration	3 Oral
	1 Parenteral
	3 Enteral
	1 Other
Geographic Region of Intervention	4 Asia
	1 Europe
	1 North America
0/ of Doutinin outprovide Otomo IV/ Discours	
% of Participants with Stage IV Disease	
	2 Not reported
% Fomolo Barticipanta	
	1 25 40%
	1 50 74%
	2 75-100%
	2 Not reported
Mean Age of Participants	6 < 65
	1 65+
	1 Not reported
KQ3 Outcomes Evaluated+	2 Treatment tolerance
	6 Quality of life
	1 Functional status
	1 Symptoms

 Table 8.5. Basic characteristics of studies of effect of nutrition interventions on symptoms: route or timing of nutrition interventions

Abbreviation: KQ=Key Question.

*Other cancer types include studies evaluating all remaining cancer types not included in previous categories (e.g., breast, prostate, lung). †Studies may evaluate multiple outcomes.

Description of Interventions

The eight studies identified varied in duration of intervention, comparator, cancer therapy received, and primary outcomes. Of the eight studies, four compared early oral feeding versus enteral nutrition. One study evaluated esophageal cancer patients after minimally invasive esophagectomy and compared early oral feeding compared to a nasogenteral feeding tube.¹⁷¹ Another investigated early oral feeding compared to a nasogastric tube in patients who underwent elective colorectal resection.¹⁵³ A third study compared immediate initiation of enteral nutrition, after placement of a prophylactic gastrostomy tube, that continued through the completion of treatment versus standard care, which included initiation of enteral nutrition when indicated (e.g., oral intake <60% of estimated energy requirements).⁷¹ Finally, a four-arm study compared combinations of predominately oral immunonutrition 7 days pre- and postoperatively versus standard nutrition support (Nestle Health Science) in individuals with esophageal cancer.⁶⁸Another study examining timing of nutrition supplementation consisted of three arms and included patients undergoing elective surgery for breast and colorectal cancers. This study compared consumption of a milk-based oral nutrition supplement preoperatively and ending supplementation at either discharge or 90 days post-discharge.⁶⁹

The remaining studies were more varied but focused on evaluating the mode of nutrition interventions on symptoms of cancer treatment. One study compared a percutaneous endoscopic gastrostomy tube versus a nasogastric tube in head and neck cancer patients undergoing curative treatment including radical surgery with adjuvant radiotherapy, chemoradiotherapy, or concurrent chemo- and radiation therapy.¹⁶⁸ Another study examined the use of jejunostomy feeding or nasogastric feeding in patients after a minimally invasive McKeown esophagectomy.¹⁷² Finally, one study compared total parenteral nutrition versus an oral diet in breast cancer patients undergoing high-dose chemotherapy and hematopoietic cell transplantation.¹⁶⁶

Nutrition Support Including Oral Nutrition Supplements

Eligible Studies

Table 8.6 summarizes the characteristics of the literature set. We identified 32 studies across 34 publications that examined the effect of nutrition support interventions during cancer treatment on cancer treatment-related symptoms.^{33, 35, 37, 40, 42, 74, 78, 81, 85, 182, 184, 185, 187, 190-192, 195-197, 206, 208, 209, 213-216, 219-224, 228, 229 Studies varied in sample size, with just over half enrolling more than 100 participants. The primary intent of these studies also varied considerably, including biomarkers of nutrition status,^{42, 216} inflammatory mediators,⁷⁴ feeding tolerance rate,³⁵ body weight changes,^{190, 206, 222, 223} and quality of life,^{191, 220} among others. While not the primary intent of most studies, almost all evaluated quality of life. Most studies were in gastrointestinal patients (n=20). Studies sometimes indicated that the intervention was delivered by a dietitian/nutritionist (n=7), and primary cancer treatments varied substantially across the studies. Appendix F provides evidence tables.}

Characteristics	Information
Total Included Studies	32 Studies
Study Sample Size	6 50-75
Olddy Gample Olze	11 76-100
	15 >100
Cancor Typo	5 Head and neck
Calleer Type	20 Castrointestinal
	20 Gastrointestinai
	A Other cancers
Intervention Delivery Setting	4 Other calleer types
Intervention Derivery Setting	13 Inpatient
	12 Outpatient
	A Other
	2 Not reported
Dominant Cancor Treatment Type of	15 Surgery along
Dominant Cancer Treatment Type of	A Chamatharapy along
Farticipants	2 Radiation alone
	11 Multiple therapies
Limited to Malnourished Patients	
	25 No
Malnourishment Screening Tool Used	10 Nutrition Risk Screening NRS-2002
manicarionniciti concoming roon cood	2 Multiple tools
	7 Other tools/metrics
	13 Not reported
Provider Prescribing or Delivering the	7 Dietitian/nutritionist
Intervention	1 Physician
	10 Multiple providers
	14 Not reported
Route of Administration	23 Oral
	4 Enteral
	5 Parenteral
Geographic Region of Intervention	10 Europe
	19 Asia
	3 Other
% of Participants with Stage IV Disease	14 ≤10%
	2 11-50%
	5 51-75%
	1 76-100%
	10 Not reported
% Female Participants	7 0-25%
	18 26-50%
	5 51-75%
	1 76-100%
	1 Not reported
Mean Age of Participants	19 <65
	7 ≥65
	6 Not reported
KQ 3 Outcomes Evaluated†	7 Treatment tolerance
	20 Quality of life
	13 Symptoms
	9 Functional status

 Table 8.6. Basic characteristics of studies of effect of nutrition interventions on symptoms:

 nutrition support including oral nutrition supplements

Abbreviation: KQ=Key Question.

*Other cancer types include studies evaluating all remaining cancer types not included in previous categories (e.g., breast, prostate, lung).

[†]Studies may evaluate multiple outcomes.

Description of Interventions

Although all studies examined nutrition support, studies varied in the type and duration of such support as well as in comparators and cancers studied. Overall, studies fell into three general categories: oral or enteral nutrition before or after surgery, oral or enteral nutrition during chemotherapy and/or radiation, and total parenteral nutrition across cancer treatments.

Oral or Enteral Nutrition Before or After Surgery

Fifteen studies examined oral or enteral nutrition support before or after cancer surgery; the studies varied in the route and contents of nutrition support delivered, and the timing of the intervention and comparison groups. Eight studies delivered oral or enteral nutrition preoperatively. One study compared a carbohydrate-rich beverage in the 2 days prior to surgery for colorectal cancer versus preoperative fasting.⁴⁰ Another examined the use of peripheral intravenous nutrition on typically fasting days for patients undergoing in-hospital workup for biliopancreatic mass lesions.³⁷ Additionally, one study compared a single-dose versus a doubledose of a 10-percent glucose solution prior to radical gastrectomy.³³ Another study compared preoperative oral supplementation versus preoperative dietary advice for individuals undergoing surgery for gastrointestinal cancer.³⁵ A fifth study compared preoperative oral nutrition versus no oral supplementation in patients undergoing elective resection of colorectal carcinoma.⁴² Another study investigated both pre- and postoperative early oral feeding versus conventional fasting in elderly patients with hepatocellular carcinoma.⁷⁴ Additionally, one study investigated an immune-enhancing diet versus a hospital-prepared blenderized diet 7 days before the operation and 14 days after in head and neck cancer patients.⁸⁵ Finally, one examined a whole oral nutrition supplement not enriched with immunonutrients 2 weeks prior to surgery and from preoperative days 2 to 14 among individuals undergoing gastrectomy versus unspecified standard care.⁷⁸ The remaining studies were delivered after cancer treatment began. While most interventions were delivered to head and neck and gastrointestinal cancer patients, one was assessed in women with gynecologic cancer.¹⁸² Additionally, standard nutrition support was evaluated for a wide range of intervention lengths from 2 days prior⁴⁰ to 1,^{187, 223} 2,¹⁹¹ 3,^{209, 228, 230} and 6^{213} months postsurgery.

Oral or Enteral Nutrition During Chemotherapy and/or Radiation

Thirteen studies examined oral or enteral nutrition support before or after chemotherapy and/or radition; however, they varied in the route and contents of nutrition support delivered and the timing of the intervention and comparison groups. One study compared amino acid-enriched oral nutrition support 1 week prior to the start of chemoembolization for up to 1 year versus usual diet among individuals with hepatocellular carcinoma.⁸¹ Two studies examined EPA-enriched oral nutrition support, but varied in populations, including lung²¹⁶ and multiple types of advanced cancer.¹⁹⁷ Another three studies evaluated hyperprotein-enhanced oral nutrition support, ^{190, 229} Immax®, and omega-3 fatty acid-enriched nutrition oral support for diverse populations and durations.¹⁸⁵ One study investigated standard nutrition support for the duration of chemotherapy in patients with gastrointestinal cancer.²¹⁹ One study examined standard nutrition support for the duration of radiation for patients with head and neck ¹⁹² and colorectal cancer.⁹⁸ Tree studies examined nutrition support, Niufutai®, for the duration of chemoradiation. Of these, one compared oral nutrition support, Niufutai®, for the duration of chemoradiation versus regular diet in individuals with advanced head and neck cancer.¹⁹⁵ The other two studies examined nutrition support versus dietary advice in individuals undergoing chemoradiation.²⁰⁶

²²⁴ A final study compared early enteral support with chemotherapy versus only chemotherapy in gastric cancer patients after a radical gastrectomy.²²¹ Total Parenteral Nutrition

Four studies examined TPN across cancer and treatment types. One study compared oral feeding supplemented by TPN versus oral feeding alone in individuals with advanced cancer cachexia who were undergoing or had recently completed cancer treatment.¹⁸⁴ A second study compared TPN, Olimel®, for 4 to 8 days postsurgery versus isotonic fluids in individuals with gastric cancer.¹⁹⁶ A third study compared TPN, Nutriflex®, versus standard diet for 5 days postoperatively in individuals undergoing bladder cancer surgery.²²⁰ A final study examined the use of TPN plus enteral nutrition support, Nutrison®, for approximately 7 days postsurgery for esophageal cancer versus enteral nutrition support alone.²²²

Multi-Component Interventions

Eligible Studies

Table 8.7 summarizes the characteristics of the literature set. Seven studies across eight publications examined multi-component nutrition interventions administered before or during cancer treatment.^{88, 232, 234, 236, 237, 239, 240, 245} While the primary intent of the studies was to assess biomarkers of immune and nutrition status,^{232, 239, 240} body weight changes,^{234, 236, 237} symptoms,²⁴⁵ and survival,⁸⁸ most also evaluated quality of life. Three studies enrolled over 100 participants. Studies varied in cancer populations, and interventions were most often delivered or prescribed by a dietitian or nutritionist. Most studies were conducted outside of the United States. Appendix F provides detailed evidence tables.

Characteristics	Information
Total Included Studies	7 Studies
Study Sample Size	2 50-74
	2 75-100
	3 >100
Cancer Type	2 Head and neck
	3 Multiple cancers
	2 Other cancer types [*]
Intervention Delivery Setting	6 Outpatient
	1 Multiple settings
Dominant Cancer Treatment Type of	1 Surgery alone
Participants	3 Chemotherapy alone
	1 Radiation therapy
	1 Multiple therapies
	1 Not reported
Limited to Malnourished Patients	3 Yes
	4 No
Malnourishment Screening Tool Used	2 Nutrition Risk Screening NRS-2002
	2 Other tool
	3 Not reported
Provider Prescribing or Delivering the	6 Dietitian/nutritionist
Intervention	1 Multiple
Route of Administration	5 Oral
	1 Other
	1 Enteral
Geographic Region of Intervention	1 Asia
	5 Europe
	1 North America

Table 8.7. Basic characteristics of studies of effect of nutrition interventions on symptoms: multicomponent interventions

Characteristics	Information
% of Participants with Stage IV Disease	1 <10%
	1 51-75%
	5 Not reported
% Female Participants	2 0-25%
	3 26-50%
	1 51-75%
	1 76-100%
Mean Age of Participants	3 <65
	4 ≥65
KQ3 Outcomes Evaluated [†]	6 Quality of life
	3 Symptoms
	2 Functional status

Abbreviation: KQ=Key Question.

*Other cancer types include studies evaluating all remaining cancer types not included in previous categories (e.g., breast, prostate, lung cancer).

†Studies may evaluate multiple outcomes.

Description of Interventions

Multi-component intervention studies varied in the type of interventions, comparators, cancer type, and cancer treatments delivered. Two studies examined individualized diet or nutrition plans combined with additional interventions such as education or counseling. One study examined nutrition counseling with individualized nutrition plans in individuals undergoing chemotherapy for multiple cancers.²³⁴ The other study compared an individualized nutrition plan including nutrition counseling and dietary interventions such as oral nutrition support or protein rich snacks for 3 months versus standard care in individuals with cancer.²⁴⁰ The remaining studies varied widely in the types of interventions delivered. One three-arm study compared 1) a calcium-rich diet, 2) a calcium-rich diet plus exercise, and 3) a calcium-rich, low-fat and high-fruit-and-vegetable diet plus exercise in women with breast cancer undergoing chemotherapy versus dietary advice or no intervention in individuals with lung cancer.⁸⁸ A third study compared in-hospital nutrition education, early enteral nutrition support (24 hours after surgery), that continued 6 weeks postoperatively versus regular enteral nutrition support (48 hours after surgery) in individuals treated for laryngeal cancer.²³⁹ Finally, one study compared a

prophylactic percutaneous endoscopic gastrostomy (PEG) tube for early enteral feeding along with nutrition advice versus standard care in patients undergoing treatment for head and neck cancer.^{236, 237}

Risk of Bias and Outcome Assessment

Table 8.8 summarizes risk of bias assessments. Appendix F provides risk of bias assessments and outcomes.

Intervention	Head and Neck	Gastrointestinal	Multiple Cancers	Other Cancer Types [*]
Nutrition Counseling	-	-	-	-
Dietary Supplements	-	3 Medium 2 High (N=5)	-	-
Special Diets	-	-	-	-

Table 8.8. Risk of bias assessment for nutrition interventions on symptoms

Intervention	Head and Neck	Gastrointestinal	Multiple Cancers	Other Cancer Types [*]
Route or Timing of Nutrition Interventions	-	-	-	-
Nutrition Support Including Oral Nutrition Supplements	-	5 Low 4 Medium 11 High (N=20)	-	-
Multi-Component Interventions	-	-	-	-

Abbreviation: - = risk of bias not performed.

*Other cancer types include studies evaluating all remaining cancer types not included in previous categories (e.g., breast, prostate, lung).

Outcomes for Dietary Supplements

Two of the three medium-risk-of-bias studies examined inpatient dietary supplementation, one with probiotics plus omega-3-fatty-acid-enriched nutrition in gastrointestinal cancer versus a placebo while participants received chemotherapy,¹¹⁶ while another examined the use of a triglyceride and protein-enriched enteral nutrition in surgically treated patients with gastrointestinal cancer.¹³³ The third medium-risk-of-bias study examined protein-rich dietary supplementation at home and postoperatively for gastrointestinal surgery,⁵⁵ Study sample sizes ranged from 71 to 229 patients. Two studies reported on quality of life and symptoms, but the studies were distributed across interventions that could not be aggregated. One study of probiotics and omega-3 fatty acids reported improvement in quality of life.¹¹⁶ The other study examining quality of life reported no benefit.⁵⁵ The two studies reported mixed outcomes for patient-reported symptoms with one reporting a benefit¹¹⁶ and one no difference.¹³³

Outcomes for Nutrition Support (Including Oral Nutrition Supplements)

Five low- and four medium-risk-of-bias study examined nutrition support interventions in the surgical inpatient setting.^{33, 35, 40, 42, 74, 208, 209, 222, 223, 228} Study sample sizes were midrange, from 50 to 353. Treatment tolerance, quality of life, symptoms and functional status were reported in one to five studies, but these studies were distributed across interventions that could not be aggregated. The studies reported mixed results. Two low-risk-of-bias studies reported improvement in nausea^{74, 208} and one reported improvement of functional status for individuals receiving preoperative oral carbohydrate drinks.⁴⁰ A second low-risk-of-bias study reported improvement in treatment tolerance and symptoms after use of oral nutrition supplements and dietary advice.^{208, 209} One low- and one medium-risk-of-bias study reported no improvement in symptoms after increased preoperative oral supplementation.^{33, 35} Another medium-risk-of-bias study reported no difference in quality of life or functional status after use of oral nutrition supplements after hospital discharge.²²⁸

Variation in the Effects of Nutrition Interventions on Preventing Negative Outcomes

While studies enrolled individuals from multiple cancer types, treatments, and stages (KQ3a), across the lifespan (KQ3b), with varying degrees of muscle wasting (KQ3c) and in those with a range of comorbid conditions (KQ3d), no eligible studies specifically evaluated

whether the effects of nutrition interventions on preventing negative outcomes varied across these characteristics.

Chapter 9. Effect of Nutrition Interventions Intended for Body Weight Loss

Key Points

- Four studies reported the effects of nutrition interventions intended for body weight loss using special diets among individuals with breast cancer.
- All studies assessed body weight and composition changes. One study reported on each of the following symptoms: quality of life, symptoms, and treatment tolerance.
- Studies showed high heterogeneity across study populations, interventions, and outcomes, making meta-analysis infeasible.

Overview

This chapter includes studies that addressed Key Question (KQ) 4 and examined the effect of nutrition interventions aimed at body weight loss for preventing and treating negative outcomes of cancer and cancer treatment in patients who are overweight or obese. We identified only four studies evaluating KQ4 (Table 9.1) and present a brief description of this limited literature set. Due to the small number of studies, we did not assess risk of bias. Detailed information on both studies can be found in Appendix G.

Table 9.1. Studies of effect of nutrition interventions intended for body weight loss, strat	tified by
intervention type and cancer type	-

Intervention	Head and Neck	Gastrointestinal	Multiple Cancers	Other Cancer Types [*]	Total
Nutrition Counseling	0	0	0	0	0
Dietary Supplements	0	0	0	0	0
Special Diets	0	0	0	4	4
Route or Timing of	0	0	0	0	0
Nutrition Interventions					
Nutrition Support	0	0	0	0	0
Including Oral					
Nutrition Supplements					
(Nutrition support)					
Multi-Component	0	0	0	0	0
Interventions					
Total	-	-	-	-	4

Abbreviations: - = Not applicable.

*Other cancer types include studies evaluating all remaining cancer types not included in previous categories (e.g., breast, prostate, lung cancer).

Special Diets

Eligible Studies

Table 9.2 summarizes the characteristics of the literature set. We identified four unique studies examining the use of special diets prior to or during cancer treatment intended for weight loss.^{244, 246-248}

Table 9.2. Basic characteristics of studies of effect of nutrition interventions intended for body weight loss: special diets

Characteristics	Information
Total Included Studies	4 Studies
Study Sample Size	2 76-100
	2 >100
Cancor Typo	1 Other
Intervention Delivery Setting	3 Outpatient
Intervention Derivery Setting	1 Other
Dominant Cancer Treatment Type of Participants	2 Chemotherapy
	2 Multiple therapies
Limited to Malnourished Patients	2 No
Screening Tool Used	4 Not reported
Provider Prescribing or Delivering the Intervention	2 Dietitian/nutritionist
	1 Multiple
	1 Not reported
Route of Administration	4 Oral
Geographic Region of Intervention	2 Europe
	2 Asia
% of Participants with Stage IV Disease	4 <10%
% Female Participants	4 76-100%
Mean Age of Participants	4 <65
Outcomes Evaluated	4 Body weight/composition changes

Description of Interventions

One study compared the effect of calorie restriction with synbiotics on quality of life and edema reduction in overweight and obese (body mass index between 25 and 35 kg/m²) individuals with breast cancer-related lymphedema versus no intervention.²⁴⁴ A second study compared the use of a diet designed to prevent weight gain versus control in women undergoing adjuvant chemotherapy for breast cancer who were close to being classified as overweight (average body mass index of 24.7 kg/m2) and were potentially at risk for body weight gain, particularly among women treated with cyclophosphamide, methotrexate, and fluorouracil (CMF).²⁴⁶ Another study investigated the effect of a Mediterranean diet and dietary advice compared to only dietary advice on patients undergoing chemotherapy for breast cancer.²⁴⁸ The final study compared intermittent 25-percent energy restriction versus continuous 25-percent energy restriction in breast cancer patients undergoing chemotherapy.²⁴⁷ One study was conducted in Iran, one in Italy, one in South Korea, and one in England.

Chapter 10. Cost-Effectiveness of Nutrition Interventions

Key Points

- Among studies evaluating effectiveness of nutrition interventions, less than 4 percent published any cost information related to the intervention. These studies were predominantly conducted in inpatient settings in non-U.S. health systems.
- In the grey literature, few studies conducted cost-effectiveness or cost-benefit analyses, but those that did demonstrated mix results on the value (cost-benefit, cost-effectiveness) of nutrition interventions in non-U.S. health systems.

Cost-Effectiveness

This chapter provides an overview of the literature set addressing the Contextual Question of cost-effectiveness of nutrition interventions for cancer. As noted in the methods, we took a broad approach to evaluate this contextual question and studies that evaluated the cost or value (e.g., cost-effectiveness, cost-benefit) of nutrition interventions for cancer. Among the literature set included in the Key Questions (n=206), fewer than 4 percent (n=8/206) of studies published any information on the cost of the nutrition intervention. Many times, the cost-related information was focused on the total cost of care for one group versus a comparison group, making it challenging to identify the exact cost of the intervention or its components. When cost information was provided, studies were often performed outside of the United States, including in Europe and Asia (Table 10.1). Within each intervention type where cost information was provided, study populations, timing, and duration of the interventions as well as the interventions themselves were very heterogeneous. All but one study evaluated costs in the inpatient setting.

Study	Intervention/ Control	Cost of Nutrition Intervention	Costs Broken Down by Intervention Components?	Formal Cost- Effectiveness Done?	Setting and Location of Intervention
Braga 2001 ¹⁵¹ Key Question 2	TPN / EEN	Cost per day TPN: \$90.60 vs EEN: \$25; (p < .001) (Denomination: US Dollars)	Yes Prescriptions: 73.2% of TPN costs vs. 22% of EEN costs; Monitoring and personnel costs similar between two groups	No	Inpatient Italy

Table 10.1. Summary of literature evaluating Contextual Question on intervention cost or costeffectiveness

Study	Intervention/ Control	Cost of Nutrition Intervention	Costs Broken Down by Intervention Components?	Formal Cost- Effectiveness Done?	Setting and Location of Intervention
Gianotti 2000 ²⁴⁹	Perioperative enteral immunonutrition/ Standard enteral diet	Cost per group: Perioperative: 35,437 Standard: 10,768 ; (Denomination: Euro)	No	Yes Cost- effectiveness: Net savings of 2,386 euros in the intervention per complication free patient	Inpatient Italy
Klek 2005 ¹¹⁹ Key Questions 1&2	Intervention Group 1: PN+ omega 3 fatty acids; Intervention Group 2: PN+ glutamine; Control: standard PN	Mean cost of overall hospital stay with cost of PN removed: PN + omega 3 fatty acids: 3,360 PN + glutamine: 3,240 Standard PN: 3,720 No p-value reported (Denomination: Euros)	Νο	Νο	Inpatient Poland
Li 2015 ²⁰⁴ Key Question 2	EN/ conventional perioperative treatment after gastric cancer surgery	Total hospital costs EN: 32 Conventional perioperative treatment: 35 P<0.01 (Denomination: US Dollars)	No	No	Inpatient China
Ryu 2009 ¹⁶⁷ Key Question 2	TPN/ (NGTN)	Per patient daily cost TPN group: 31,410 (\$34.41) versus NGTN group 19,600 (\$19.60) (Denomination: 2007 KRW)	Yes Nutrition: 28,320 (TPN) vs. 18,400 (NGTN); Personnel: 1,410 (TPN) vs. 1,200 (NGTN)	No	Inpatient Korea
Song 2017 ²³⁹ Key Question 2,3	Hospital to home nutrition management model /routine nutrition management	Overall hospitalization costs: Hospital to home: 2.73 vs. Routine: 2.65, P=0.778 (Denomination: RMB)	No	No	Inpatient, Home China
Wang 2019 ²⁵⁰ Key Question 2	EOF/ DOF after surgery	Overall hospitalization costs: EOF: 8.24 vs. DOF: 8.44, (p=0.492) (Denomination:10 ⁴ yuan)	No	No	Inpatient China

Study	Intervention/ Control	Cost of Nutrition Intervention	Costs Broken Down by Intervention Components?	Formal Cost- Effectiveness Done?	Setting and Location of Intervention
Xiao-Bo 2014 ¹⁷⁶ Key Question 2	TPN/EN	TPN: 123 vs EN: 66, (p<0.05), unclear of included cost components (Denomination: Unknown)	No	No	Inpatient China

Abbreviations: DOF= delayed oral feeding; EEN= early enteral nutrition; EN=enteral nutrition; EOF= early oral feeding;.KRW= Korean Wan; NGTN=nasogastric tube nutrition; RMB: renminbi; PN=parenteral immunonutrition; TPN= total parenteral nutrition;

In a grey literature search (see Appendix A for methods), including search of the Tufts Cost Effectiveness Analysis Registry, we identified a few additional studies that performed formal cost analyses of nutrition interventions. Two studies conducted a cost-benefit analysis of preoperative immunonutrition from a randomized controlled trial included in our Key Ouestions. One cost-effectiveness analysis²⁵¹ demonstrated an economic advantage for those receiving preoperative immunonutrition relative to the conventional approach based on a randomized controlled trial of individuals undergoing surgery for gastrointestinal cancer.⁶⁶ Another study evaluated the cost-effectiveness²⁵² of results from a randomized controlled trial¹⁸⁵ examining oral nutrition supplements with nutrition counseling among those undergoing radiation therapy for head and neck cancer. This study showed that the intervention was less costly and more effective than nutrition counseling alone. However, there was high uncertainty around this estimate. One study examined the cost-effectiveness of perioperative enteral nutrition versus standard of care for colorectal cancer surgery, finding that the intervention resulted in an incremental costeffectiveness ratio (ICER) of -6,276 Euros, with a marginal gain in quality adjusted life years.²⁵³ An additional recent study evaluated the use of supplemental parenteral nutrition versus enteral nutrition alone and found that for patients with Stage IV inoperable pancreatic cancer the incremental cost-effectiveness ratio was 41,350 or 91,501 British Pounds depending on whether nursing or home delivery costs were combined or provided separately.²⁵⁴ Finally, in a narrative review and economic analysis, one study examined the value of nutrition support in patients with gastrointestinal malignancies. The study identified published literature with cost and clinical outcomes and conducted a Value Analysis by developing models using Medicare claims from gastrointestinal cancer patients. The study concluded that application of nutrition interventions provides positive clinical and economic value to the health system for this population.²⁵⁵

In mixed populations (i.e., in studies that evaluate critically ill or hospitalized patients who possibly have cancer), a few additional resources might inform discussions on cost-effectiveness. One systematic review evaluated the cost effectiveness of using standard oral nutrition supplements in the hospital setting. The study identified eleven cost analyses, primarily comprising secondary analyses of randomized controlled trial data, that suggest standard oral nutrition supplements in the hospital setting produce a cost saving and are cost-effective.²⁵⁶ Another study examined the cost savings associated with nutrition support in medical inpatients based on a systematic review of randomized trials, finding that for medical inpatients who are malnourished or at nutrition risk, in-hospital nutrition support offers a cost-effective way to reduce readmissions.²⁵⁷ A third study evaluated the cost-effectiveness of omega-3-fatty-acid-enhanced parenteral nutrition versus standard parenteral nutrition in hospitalized patients, and showed that the intervention demonstrated superior clinical efficacy and decreased in mean

treatment costs.²⁵⁸ Finally, one recent study examined the value of nutrition support therapy (including a sub analyses of gastrointestinal cancers), modeling the impact of adoption in the Medicare population, and concluded that optimization of nutrition support therapy for specific patient populations is estimated to reduce Medicare spending by millions of dollars per year.²⁵⁹

Chapter 11. Discussion

Overview

This evidence map sought to provide a broad overview of randomized trial evidence for the effectiveness of nutrition interventions delivered prior to or during cancer treatment for preventing and treating negative outcomes of cancer and cancer treatment. Strengths of our report include: a comprehensive search and inclusion of randomized control trials relevant to the topic; high-level mapping of the evidence across Key Questions by patient, intervention, comparator, and outcome categories; identification of evidence gaps; creation of an active evidence repository to facilitate future, more detailed, analyses of the included studies; development of an analytic framework to guide our review questions; categorizations of populations, interventions, cancer types, treatments, and outcomes of interest to support a fresh, rigorous, and independent organizational framing to an extremely broad field; targeted assessment of study methodological quality focused on intervention and cancer areas with the highest volume of identified evidence; and suggestions for new research based on our findings.

Our review identified, from a broad range of nutrition interventions (e.g., dietary supplements, nutrition support, nutrition counseling), the most commonly reported strategies or components and their outcomes relevant across populations, cancers and cancer treatments. Our results should help clinical experts, policymakers, and funding agencies to prioritize a research agenda related to the use of nutrition interventions in cancer treatment.

Overall, we found two decades of randomized trial evidence across more than 206 studies on nutrition interventions for adults prior to and/or during cancer treatment focused on the use of dietary supplements, nutrition support (including oral nutrition supplements), and the route or timing of nutrition interventions within gastrointestinal and head and neck cancers. Most studies examined changes in body weight/composition, adverse events, length of hospital stay, and quality of life. Among interventions with a high volume of literature, made up mostly of studies of dietary supplements and nutrition support in patients with gastrointestinal and head and neck cancers, 11 percent (n=12) were rated as low risk of bias (higher quality), 40 percent (n=46) medium risk of bias and 49 percent (n=56) high risk of bias (lower quality). Low- and medium-risk-of-bias studies reported mixed results on the effect of nutrition interventions across outcomes of cancer and cancer treatment.

Our review process aimed to identify nutrition interventions supported by evidence that met a minimum threshold of quality. Due to the diversity and volume of the literature, we intentionally focused risk of bias assessment within intervention and cancer types where we found a high volume of studies, as we note in our methods. We used this strategy in order to identify the greatest amount of lower risk of bias evidence for specific nutrition interventions by cancer types. These mainly focused on use of dietary supplements, nutrition support, and the route or timing of interventions. Even within this subset of 100 studies, heterogeneity of populations, interventions (e.g., type, amount, duration), and outcomes (not only outcomes, but when and how measures were assessed) mostly precluded our ability to aggregate findings. Further, this literature comprised mostly high risk of bias studies. Therefore, our findings point to the need for rigorous new research to bolster the evidence base. Our findings also suggest the need for more a detailed future assessment of studies contained in this evidence map. Such a future assessment could focus on priorities and interventions most relevant to specific stakeholders (e.g., oncologists, patients, dietitians, researchers, policymakers), and therefore lead to studies

specifically designed to evaluate the main outcomes of interest relevant to the identified Key Questions. These results were presented to inform discussion of experts and stakeholders at the National Institutes of Health Pathways to Prevention Program (P2P) conference, held July 26th-28th, 2022. Based on this evidence, an independent panel developed a draft report to summarize the workshop discussions and identify specific future priorities, which can be found on the P2P Workshop website (https://prevention.nih.gov/research-priorities/research-needs-and-gaps/pathways-prevention/nutrition-prevention-improved-cancer-health-outcomes).

Broader Context of Available Interventions and Strategies

The contextual question addressed by our review examined the cost-effectiveness of nutrition interventions delivered prior to or during cancer treatment. However, few studies provided information on the cost of the interventions delivered or performed a formal cost-effectiveness analysis. The lack of cost-effectiveness literature in this topic area is not surprising. Research must first establish effectiveness of an intervention before cost-effectiveness can be addressed. However, as the literature matures to establish an evidence base for effectiveness of nutrition interventions, studies must provide sufficient information on the implementation of the interventions to allow for comprehensive cost-effectiveness analyses. Specifically, studies frequently reported only sparse details about added medical (e.g., supplies, technology, drugs) and personnel investments necessary to implement a given intervention. Additionally, when studies did report costs, they mostly did so in a summative way that lacked sufficient detail needed to identify which component(s) contributed to the overall intervention. Further, most cost information came from trials conducted outside of the United States, where costs of care and treatment contexts differ significantly from U.S. settings and therefore reduce applicability. Future research should emphasize not only effectiveness of interventions, but also the components of cost in a detailed manner applicable to the United States context.

Future Research

The question of if nutrition interventions (or components) can prevent negative health outcomes, and, for whom and under what circumstances is of vital importance. The number of individuals experiencing or at risk for cancer-related malnutrition remains substantial, with estimates ranging from 25 to 80 percent across patient populations.⁸⁻¹⁰ Because cancer risk increases with age, the rapidly growing older population in the United States will increase demand for cancer care and, by extension, nutrition therapy, over the coming decades.²⁶⁰ However, the topic of nutrition interventions to prevent adverse effects of treatments for cancer patients at risk for malnutrition is extremely broad. This breadth increases complexity for adequately addressing needed evidence. Although our report identified a great many studies, they differed widely by population, intervention type, intent, timing, and mode of delivery, as well as comparators, cancer types, treatments, and outcomes. Even within intervention categories, we found substantial heterogeneity, making aggregation infeasible. Importantly, many studies were not specifically designed to evaluate the main outcomes of interest relevant to the identified Key Questions. However, these findings should not serve to discourage, but rather bolster, the rigor and content of future research informing clinical practice on nutrition interventions for cancer, including expansion of investigations of the role of malnutrition screenings, which was not addressed by this review.

In the near term, research could evaluate evidence from existing methodologically rigorous observational studies. Our evidence map did not include such studies, but they could provide valuable information about the association between nutrition interventions and negative outcomes of cancer and cancer treatment. Further, observational studies could provide additional context on potential effectiveness when interventions are unethical to randomize or to describe the natural course of nutrition status and body composition changes over the course of treatment to identify individuals at risk for malnutrition. Ultimately, evidence maps scan the horizon of existing information, while providing a repository to facilitate a deeper assessment of the direction, magnitude, and certainty of intervention effects in high priority/high literature volume areas. Therefore, our map may provide the basis for a future quantitative assessment or more detailed stratification of our findings. Such an assessment could focus on the specific nutrition interventions, cancers, and cancer treatments most relevant to patients, clinicians, and guideline groups. Longer-term funding efforts, which should align with the priorities of the National Institutes of Health Precision Nutrition Initiative,²⁶¹ should also focus on 1) standardizing definitions and taxonomies for populations, interventions, and outcomes (including identifying those 'at-risk' for malnutrition), and 2) improving rigor in the design and reporting of nutrition interventions. To further expand on specific areas that could benefit from standardized definitions and improved rigor, we outline below several future research areas that could inform practice and policy.

Methodological Rigor

The literature included in this review represented an extremely broad and diffuse array of nutrition interventions delivered. Interventions ranged from short-term use of dietary supplements, many of which had minimum caloric value, to multi-month use of oral nutrition support (e.g., Ensure®). Such wide-ranging interventions may differ significantly in ability to address underlying malnutrition. Oftentimes, the literature lacked a clear conceptual model justifying how the intervention would be expected to improve outcomes. As a result, the field could benefit from development of a comprehensive and clear conceptual framework addressing how these diverse nutrition interventions fit within clinical pathways, as well as how they could improve key outcomes of most importance to stakeholders. Further, a detailed conceptual framework could outline potential mechanisms for the effect of nutrition interventions across patient nutrition risk categories, cancers, and cancer treatments. Such a framework could therefore help determine which areas are most likely to answer clinical questions and outcomes (rather than intermediate measures) that are of highest priority to clinicians and patients. Ultimately, this would help researchers prioritize, from the set of diverse interventions, those that conceptually might best address malnutrition in this population.

Additionally, although we limited our studies to include only randomized controlled trials with a minimum enrollment of \geq 50 participants, we were struck by the lack of adherence to basic reporting standards for randomized controlled trials within these publications. Substandard reporting ranged from missing information on the randomization process, blinding of participants and assessors, and populations analyzed, among others. Therefore, the current literature base was largely categorized as having a high risk of bias due to conduct and reporting, most often due to lack of clarity around the performance (e.g., analyzing individuals as randomized, blinding of participants and assessors), and detection (e.g., ability to assess how outcomes were defined and evaluated). Funding and publication of future research on nutrition interventions should emphasize consistent use of criteria laid out in the CONSORT statement.²⁶² CONSORT

comprises a 25-item checklist and flow diagram focused on accurate and consistent reporting of how a trial was designed, analyzed and interpreted. Further, clear and complete reporting of research sponsorship is critical as several included studies that were funded (in part or whole) by organizations with the potential to influence findings. More widespread adoption of this tool may enhance the methodological rigor of the conduct and reporting of future published nutrition intervention studies, increasing the overall quality of the literature. Finally, we recognize that limiting to evaluation to only randomized controlled trials may not address the underlying issue of bias in access to and enrollment in clinical trials, with prior research demonstrating underrepresentation among key populations such as those from communities of color, the uninsured and those experiencing adverse social determinants of health.²⁶³ Coupling strong methodological practices and reporting with coordinated efforts to inform research priorities could help to better focus resources on high priority areas with the greatest potential to improve outcomes and practice.

Populations

During topic refinement, stakeholders were eager to understand the available literature within specific populations of individuals receiving nutrition interventions (e.g., adults ≥ 65 years old, those with muscle wasting, individuals with comorbid conditions). Although the included literature may have enrolled individuals from these important subpopulations, studies rarely reported results according to these characteristics. Rather, studies tended to focus on very specific populations, such as individuals undergoing gastrectomy for early-stage gastric cancer. Results from such narrowly specific populations may not be broadly applicable or transferrable to interventions in other cancer populations or treatments. Additionally, many studies often used broad or relatively vague definitions for "at risk for malnutrition", ranging from defining all individuals with cancer diagnosis as having risk for malnutrition to using specific malnutrition screening tools (e.g., Nutrition Risk Screening (NRS)-2002), which were rarely reported in the literature. Other studies relied on body weight, weight loss, and body mass index as indicators of malnutrition. Such definitions cannot capture the comprehensiveness or complexity of malnutrition from a clinical, epidemiological, public health, or sociodemographic perspective including racial/ethnic disparities and food insecurity. These vague definitions of malnutrition made it difficult to assess and compare across studies the baseline need for nutrition interventions or opportunities for improvement in nutrition status, which should be consistently evaluated at baseline and post-intervention. Prior research examining malnutrition screening in well-defined populations has demonstrated it is feasible to identify and quantify individuals with cancer who are at risk for malnutrition using defined tools.²⁶⁴ Future research should seek to more clearly define malnutrition and risk for malnutrition. Better definitions would make it easier to identify malnutrition and its root cause(s) as well the groups most likely to respond to specific interventions.

All but a few studies within this literature set focused on head and neck and gastrointestinal cancers. The focus of nutrition interventions within head and neck and gastrointestinal populations is not unexpected due to the higher risk of malnutrition in these populations due to a combination of changes to the digestive tract as a result of the cancer as well as functional changes resulting from the cancer treatments themselves.²⁶⁵ This presents opportunities to evaluate within other cancer types to ensure that tested interventions are implemented in populations with the greatest need (i.e., highest risk for malnutrition) and burden (i.e., most common cancers). Additionally, as is common in randomized controlled trials, studies tended to

enroll younger populations (with mean ages at enrollment of <65 years) relative to the overall cancer population (with average age of diagnosis >65 years). Studies rarely reported race and ethnicity. Finally, almost all studies were conducted outside of the United States—mainly in Asia and Europe—where nutrition norms, costs of care and available resources differ greatly from the United States. While several ongoing studies registered on Clinicaltrials.gov hold potential to further inform the discussion on the effect of nutrition interventions and cancer such as DIANA-5 (NCT05019989), DEDiCa (NCT02786875), and MEDEA (NCT04304924), the majority of these studies are sponsored and conducted by non-U.S. institutions. To better inform future implementation, future research should examine the effects of interventions across populations and ensure they are applicable to United States settings to inform future implementation.

Intervention

Overall, we note significant diversity in the nutrition interventions tested to reduce the negative effects of cancer treatment and associated symptoms. Interventions were extremely heterogeneous, even within intervention types. For example, studies in dietary supplements included omega-3 fatty acids, glutamine or arginine, among others. Even when the studies used the same interventions, they varied dramatically in duration (e.g., 2 days versus 3 months), dosage, purity, route of administration, and type of cancer treatment received. Many of the interventions studied have little standardization even within agreed upon nutrition intervention categories. Further, we were often unable to ascertain specific features of the conduct of the trial such as detailed information on contents and modality of nutrition received in the interventions groups, the context of the settings (inpatient versus outpatient), and adherence. Additionally, few (<10%) studies specifically identified which nutrition professional prescribed or administered the intervention.

Comparison groups for studies also reflected heterogeneity, and varied in their use of comparisons to test different doses and types of the intervention against an all-encompassing "standard of care" that was often incompletely defined, if at all. Further, we often found it difficult to understand exactly what the comparison group received in terms of dose, duration or quantity of the nutrition intervention. In several cases, the nutrition interventions tested more than one variable at a time relative to the comparison group. For example, a study may have implemented a nutrition intervention later in the comparison group relative to the intervention and also calorie restricted the control group. These combined differences in the comparison group make it difficult to determine which factor(s) account for outcomes across groups. Study designs could be enhanced by complete reporting of total daily caloric intake across study groups. Finally, control group interventions for at-risk or malnourished adults should reflect current professional practice standards. Several non-US surgical studies used prolonged nonfeeding postoperatively as a control, which was deemed unethical by several authors^{62, 66, 150, 263} given recommendations of several professional societies more than two decades ago. Overall, heterogeneity in intervention type and reporting, and the resulting absence of key information, limited our ability to aggregate or conduct meta-analyses (beyond qualitative summaries) or accurately quantify risk of bias. Beyond the broad interventions themselves, studies rarely aimed nor were they statistically powered to examine variation in cancers studied or in types of healthcare providers. This made it difficult to determine the broader relevancy and applicability of findings from a given intervention for patients with a specific cancer (type/location/stage), treated in a selected setting (outpatient vs. inpatient) by different clinicians (surgeon, radiation

oncologist, oncologist). This may be particularly relevant among studies of individuals with advanced cancer, where body weight loss may no longer be a nutritional issue, but rather a metabolic issue that may be better treated pharmacologically.

The current evidence lacks overarching coordination based on a strong clinical foundation for effectiveness in preventing cancer treatment harms—a problem made worse by wide variation in cancer types, stages, symptoms, and treatment approaches. Comorbidities further complicate the picture, as does the relatively short duration of cancer treatments. The field needs to identify more comprehensive interventions and assess whether research should focus on specific cancer types/stages/treatments or use a more general approach across these areas. In other words, the field must determine if nutrition interventions are only effective in certain clinical situations or are more broadly applicable across cancer populations. Therefore, future research needs well-designed studies with detailed reporting on the interventions in settings and subgroups well aligned with priorities within U.S. settings and practice.

Outcomes

Outcomes collected to assess the impact of nutrition interventions were diverse and ranged from intermediate outcomes, such as changes in body weight, to longer-term outcomes such as changes in overall survival. We treated outcomes as a single category encompassing both outcomes of cancer and cancer treatment. We based this choice on the fact that the studies themselves did not differentiate between these outcome categories making it impossible for us to disaggregate them and assign to one category or the other.

Additionally, the fact that relevant outcomes were reported does not mean that these outcomes were the primary intent of the studies or that the studies were adequately powered to detect differences in specific outcomes. Qualitatively, in our data abstraction, we noted that many studies were focused on examining the impact of nutrition interventions on clinical laboratory markers of nutrition status or immune function; yet they also reported differences in mortality and length of hospitalization even though these were not stated as a primary, or even secondary, outcome. In addition, outcomes were often poorly defined, with few details provided on the timing or definitions of outcomes. While validated tools were sometimes used to quantify outcomes of interest, such as the Common Terminology Criteria for Adverse Events (CTCAE), more often studies only reported that an adverse event occurred. Studies reported few details on outcome evaluation (how, by whom, over what time frame and using what criteria the outcome was evaluated). Specifically, studies rarely reported differences in attrition or length of hospital stay/followup between the intervention and comparison groups. This presents particular challenges for outcomes such as adverse events, where the longer someone is observed, the more likely it is that outcomes of interest will be identified. Even when outcomes were clearly evaluated using validated tools, the tools selected (e.g., quality of life measures) varied across studies, making aggregation and comparison impossible.

One crucial way to improve outcome assessment in future nutrition intervention studies will be to use standardized assessments with common, validated tools coupled with clearly defined assessment time periods that are consistent between the intervention and comparison group. This may include routine data collection of dietary intake and other measure of nutrition status. Additionally, studies need to report how outcomes were assessed, by whom, and according to a standard *a-priori* criteria. Many of these components are encompassed more broadly within the CONSORT criteria. However, our review suggests opportunities for professional societies representing nutrition and oncology professionals to advance a minimum reporting set of validated outcomes for nutrition intervention studies. However, we also recognize that collection of standardized, validated tools for assessment of the most clinically relevant outcomes must be balanced with the feasibility of collection (e.g., imaging, invasive biopsies) that promote recruitment of adequate sample sizes from diverse populations for these nutrition interventions.

Implementation and Systems Complexity

In addition to intervention related considerations, nutrition and oncology professionals, policymakers and payers should remain aware of potential implementation considerations. Even when high-quality evidence is advanced to support the use of specific nutrition interventions, implementation may be difficult for many reasons. First, most nutrition intervention studies have been conducted in Asia and Europe, raising questions about feasibility or practicality of implementing these interventions in U.S. settings, where practice considerations and norms differ. Second, the United States continues to under-resource nutrition professionals in both inpatient and outpatient settings,^{266, 267} leading to poor overall access to comprehensive nutrition care for cancer patients across the United States¹⁷ Introducing new or expanded nutrition interventions without also enhancing capacity to implement these interventions will be problematic. Reimbursement for nutrition interventions, particularly outside of the inpatient setting, remains sparse. Finally, because studies were conducted in research settings with selected volunteers, we do not know whether our findings are applicable to non-research settings. Our report did not address how social determinants of health affect malnutrition risk, or barriers and facilitators for reducing malnutrition or adverse effects of cancer treatment outcomes, all of which needs attention from future research. As a result, implementation of nutrition interventions will require comprehensive considerations of equity across populations with different underlying social determinants of health.

Once evidence is established for the effectiveness of nutrition interventions, additional resources and administrative support will be required to integrate these interventions into complex care systems. Some approaches for facilitating needed resources and support include value-based care models or medical homes that support innovative approaches to addressing and improving outcomes more broadly among individuals with cancer. These models may encourage use of effective supportive care interventions (including use of dedicated nutrition professionals, such as registered dietitians) to improve outcomes in this population. As with studies evaluating the effectiveness of the nutrition interventions themselves, these approaches must be coupled with rigorous research designs in future studies to ensure evidence-based implementation.

Research Approaches

Our evidence report identified areas that have implications for future research and by extension practice and policy. This review illustrates the breadth of the topic and exposes the complexity of obtaining high-quality needed evidence. One important step may be to develop a clear and detailed conceptual framework that addresses the clinical pathways involved in the care process. Such a framework would focus and clarify definitions of individuals "at risk for malnutrition." It would encompass an extensive field of nutrition interventions that vary widely in delivery, composition, dose, and purity and standardization even within agreed upon nutrition intervention categories. The framework may address variation in cancers studied and cancer care providers. Importantly, this could help clarify the broader relevancy of findings from a given intervention administered in selected patients with a specific cancer (type/location/stage), treated

in a selected setting (outpatient vs. inpatient) by different clinicians (surgeon, radiation oncologist, oncologist). It could also serve to reinforce the value of patient-important and clinical outcomes over intermediate outcomes such as body weight or biochemical values. The framework would also place into context the role of social determinants of health (e.g., lack of access to insurance, economic and social stability, the built environment, social support) on the effects of nutrition interventions, particularly due to the lack of data on these factors in included studies.

Programmatic goals could target research toward support of large well designed randomized trials in high priority well-defined populations, using standardized nutrition interventions and delivery modes, and addressing cancers and cancer clinicians/treatments of interest and clinically relevant outcomes. Decisional dilemmas often revolve around whether to use broader statements of "what are the effects of nutrition interventions in cancer patients" or narrower "does a specific nutrition intervention delivered in a given way to a specific group of cancer patients treated in a certain setting by a given clinician ameliorate a selected negative outcome." And if the latter, what is the generalizability of the findings to other clinical situations?

Strengths and Limitations of the Review

We selected methods for this review that provided a detailed evidence map of the current state of literature on nutrition interventions, highlighting not only concentrations of literature but also gaps in intervention types. We used purposefully broad definitions of nutrition interventions, thereby increasing the scope, breadth and heterogeneity of the included literature to better assess the range and depth of available evidence. This decision allowed for demonstration of the diffuse literature set on the topic and highlighted the predominantly low quality of literature where there were concentrations of similar intervention types. However, this required focusing on high level directionality of intervention effects across a broader range of nutrition interventions rather than detailed, precise estimates of intervention effects. Overall, this approach allowed for high level mapping of the evidence across Key Questions by patient, intervention, comparator and outcome categories, which helped to identify evidence gaps for future research.

Conclusions

Overall, this evidence map of randomized controlled trials found great heterogeneity across populations, nutrition interventions, cancer types and treatments and outcomes. While this report identified a large number of studies, these studies examined a wide variety of populations, nutrition intervention types, intent, timing, and mode of delivery as well as comparators, cancer types and treatments, and outcomes. Additionally, heterogeneity within each of these factors precluded aggregation of results. The highest concentration of literature comprised mainly high-risk of bias studies. Importantly, few studies were specifically designed to evaluate the main outcomes of interest relevant to the identified Key Questions. In the relatively near term, future research could evaluate evidence from existing methodologically rigorous observational studies. We did not include such studies in our evidence map, but they might provide relatively rapid and valuable information of the association between nutrition interventions and cancer treatment outcomes. Additionally, mapping overviews serve mainly as a horizon scan of the existing information, while providing an evidence repository to facilitate a deeper assessment of the direction, magnitude, and certainty of intervention effects in high priority/high literature volume areas. Future research could involve a quantitative assessment or more detailed stratification of

findings for specific nutrition interventions, cancers, and cancer treatments (or across these categories) deemed most relevant to patients, clinicians, and guideline groups. Parallel efforts in the development of longer-term funding efforts should include a focus on creating standardized definitions and taxonomies for populations, interventions, and outcomes, as well as improved rigor in the design and reporting of nutrition interventions. Specifically, the field needs a more detailed future evaluation of a subset of nutrition interventions contained in this evidence map that focus on priorities most relevant to specific stakeholders (e.g., oncologists, patients, dietitians, researchers, policymakers). Currently, the quality and heterogeneity of the studies limit the ability to translate findings into clinical practice or guidelines. The field would benefit from coordinated efforts to develop detailed conceptual frameworks for mechanisms of nutrition interventions effects across patient nutrition risk categories, cancers, and cancer treatments. Such a framework would help determine those areas likely to answer clinical questions and outcomes (rather than intermediate measures) of highest priority to clinicians and patients. These frameworks can also help to prioritize research agendas as well as to inform study designs, and, ultimately clinical practice and health policy. Finally, this report did not address the relationship between social determinants of health and malnutrition risk, or relevant barriers and facilitators relevant to reducing malnutrition through nutrition interventions. Future research should examine the evidence relevant to these issues and effective interventions.
References

- Aaldriks AA, Maartense E, Nortier HJ, et al. Prognostic factors for the feasibility of chemotherapy and the Geriatric Prognostic Index (GPI) as risk profile for mortality before chemotherapy in the elderly. Acta Oncol. 2016 Jan;55(1):15-23. doi: 10.3109/0284186x.2015.1068446. PMID: 26271800.
- van Deudekom FJ, van der Velden LA, Zijl WH, et al. Geriatric assessment and 1-year mortality in older patients with cancer in the head and neck region: A cohort study. Head Neck. 2019 Aug;41(8):2477-83. doi: 10.1002/hed.25714. PMID: 30816619.
- Aparicio T, Bouché O, Francois E, et al. Geriatric analysis from PRODIGE 20 randomized phase II trial evaluating bevacizumab + chemotherapy versus chemotherapy alone in older patients with untreated metastatic colorectal cancer. Eur J Cancer. 2018 Jul;97:16-24. doi: 10.1016/j.ejca.2018.03.030. PMID: 29777975.
- Guner A, Kim SY, Yu JE, et al. Parameters for Predicting Surgical Outcomes for Gastric Cancer Patients: Simple Is Better Than Complex. Ann Surg Oncol. 2018 Oct;25(11):3239-47. doi: 10.1245/s10434-018-6684-2. PMID: 30069658.
- 5. World Health Organization. Malnutrition. 2022. <u>https://www.who.int/news-room/fact-sheets/detail/malnutrition#:~:text=Malnutrition%20refers%20to%20deficiencies%2C%2</u> <u>0excesses,low%20weight%2Dfor%2Dage</u>) <u>%3B</u>.
- Van Cutsem E, Arends J. The causes and consequences of cancer-associated malnutrition. European journal of oncology nursing. 2005;9:S51-S63.
- Arends J, Baracos V, Bertz H, et al. ESPEN expert group recommendations for action against cancer-related malnutrition. Clin Nutr. 2017 Oct;36(5):1187-96. doi: 10.1016/j.clnu.2017.06.017. PMID: 28689670.

- Ryan AM, Power DG, Daly L, et al. Cancerassociated malnutrition, cachexia and sarcopenia: the skeleton in the hospital closet 40 years later. Proc Nutr Soc. 2016 May;75(2):199-211. doi: 10.1017/s002966511500419x. PMID: 26786393.
- Caillet P, Liuu E, Raynaud Simon A, et al. Association between cachexia, chemotherapy and outcomes in older cancer patients: A systematic review. Clin Nutr. 2017 Dec;36(6):1473-82. doi: 10.1016/j.clnu.2016.12.003. PMID: 28017447.
- Muscaritoli M, Lucia S, Farcomeni A, et al. Prevalence of malnutrition in patients at first medical oncology visit: the PreMiO study. Oncotarget. 2017 Oct 3;8(45):79884-96. doi: 10.18632/oncotarget.20168. PMID: 29108370.
- Marshall KM, Loeliger J, Nolte L, et al. Prevalence of malnutrition and impact on clinical outcomes in cancer services: A comparison of two time points. Clin Nutr. 2019 Apr;38(2):644-51. doi: 10.1016/j.clnu.2018.04.007. PMID: 29789167.
- Walsh D, Szafranski M, Aktas A, et al. Malnutrition in Cancer Care: Time to Address the Elephant in the Room. J Oncol Pract. 2019 Jul;15(7):357-9. doi: 10.1200/jop.19.00165. PMID: 31188710.
- Gyan E, Raynard B, Durand JP, et al. Malnutrition in Patients With Cancer: Comparison of Perceptions by Patients, Relatives, and Physicians-Results of the NutriCancer2012 Study. JPEN J Parenter Enteral Nutr. 2018 Jan;42(1):255-60. doi: 10.1177/0148607116688881. PMID: 29505137.
- 14. Planas M, Álvarez-Hernández J, León-Sanz M, et al. Prevalence of hospital malnutrition in cancer patients: a sub-analysis of the PREDyCES® study. Support Care Cancer. 2016 Jan;24(1):429-35. doi: 10.1007/s00520-015-2813-7. PMID: 26099900.

- 15. Hébuterne X, Lemarié E, Michallet M, et al. Prevalence of malnutrition and current use of nutrition support in patients with cancer. JPEN J Parenter Enteral Nutr. 2014 Feb;38(2):196-204. doi: 10.1177/0148607113502674. PMID: 24748626.
- 16. American Cancer Society. Cancer facts & figures 2021. 2022. <u>https://www.cancer.org/research/cancer-facts-statistics/all-cancer-facts-figures-2021.html</u>. <u>Accessed on April 25 2022.</u>
- Trujillo EB, Claghorn K, Dixon SW, et al. Inadequate Nutrition Coverage in Outpatient Cancer Centers: Results of a National Survey. J Oncol. 2019;2019:7462940. doi: 10.1155/2019/7462940. PMID: 31885583.
- Miake-Lye IM, Hempel S, Shanman R, et al. What is an evidence map? A systematic review of published evidence maps and their definitions, methods, and products. Systematic Reviews. 2016 2016/02/10;5(1):28. doi: 10.1186/s13643-016-0204-x.
- Moher D, Liberati A, Tetzlaff J, et al. Preferred reporting items for systematic reviews and meta-analyses: the PRISMA statement. PLoS Med. 2009 Jul 21;6(7):e1000097. doi: 10.1371/journal.pmed.1000097. PMID: 19621072.
- Rethlefsen ML, Kirtley S, Waffenschmidt S, et al. PRISMA-S: an extension to the PRISMA Statement for Reporting Literature Searches in Systematic Reviews. Systematic Reviews. 2021 2021/01/26;10(1):39. doi: 10.1186/s13643-020-01542-z.
- 21. PICO Portal. 2022. picoportal.org2022.
- 22. Cost-Effectiveness Analysis. CDC; 2021. <u>https://www.cdc.gov/policy/polaris/economi</u> <u>cs/cost-effectiveness/index.html</u>. Accessed on October 12 2022.
- 23. Cost-Benefit Analysis. CDC; 2021. <u>https://www.cdc.gov/policy/polaris/economi</u> <u>cs/cost-effectiveness/index.html</u>. Accessed on October 12 2022.

- 24. Viswanathan M, Ansari M, Berkman N. Assessing the Risk of Bias of Individual Studies in Systematic Reviews of Health Care Interventions. Rockville, MD: Quality AfHRa; 2019. <u>https://effectivehealthcare.ahrq.gov/products</u> /methods-guidance-bias-individualstudies/methods
- Grimble RF. Basics in clinical nutrition: Immunonutrition–Nutrients which influence immunity: Effect and mechanism of action. e-SPEN, the European e-Journal of Clinical Nutrition and Metabolism. 2009;1(4):e10e3.
- Dechaphunkul T, Arundon T, Raungkhajon P, et al. Benefits of immunonutrition in patients with head and neck cancer receiving chemoradiation: A phase II randomized, double-blind study. Clin Nutr. 2022 Feb;41(2):433-40. doi: 10.1016/j.clnu.2021.12.035. PMID: 35007812.
- 27. de Miranda Torrinhas RS, Santana R, Garcia T, et al. Parenteral fish oil as a pharmacological agent to modulate post-operative immune response: a randomized, double-blind, and controlled clinical trial in patients with gastrointestinal cancer. Clin Nutr. 2013 Aug;32(4):503-10. doi: 10.1016/j.clnu.2012.12.008. PMID: 23398953.
- Feijo PM, Rodrigues VD, Viana MS, et al. Effects of omega-3 supplementation on the nutritional status, immune, and inflammatory profiles of gastric cancer patients: A randomized controlled trial. Nutrition. 2019 May;61:125-31. doi: 10.1016/j.nut.2018.11.014. PMID: 30710885.
- Kaya SO, Akcam TI, Ceylan KC, et al. Is preoperative protein-rich nutrition effective on postoperative outcome in non-small cell lung cancer surgery? A prospective randomized study. J Cardiothorac Surg. 2016 Jan 19;11:14. doi: 10.1186/s13019-016-0407-1. PMID: 26782276.

- Lende TH, Austdal M, Varhaugvik AE, et al. Influence of pre-operative oral carbohydrate loading vs. standard fasting on tumor proliferation and clinical outcome in breast cancer patients horizontal line a randomized trial. BMC Cancer. 2019 Nov 8;19(1):1076. doi: 10.1186/s12885-019-6275-z. PMID: 31703648.
- 31. Burden ST, Gibson DJ, Lal S, et al. Pre-operative oral nutritional supplementation with dietary advice versus dietary advice alone in weight-losing patients with colorectal cancer: single-blind randomized controlled trial. J Cachexia Sarcopenia Muscle. 2017 Jun;8(3):437-46. doi: 10.1002/jcsm.12170. PMID: 28052576.
- Burden ST, Hill J, Shaffer JL, et al. An unblinded randomised controlled trial of preoperative oral supplements in colorectal cancer patients. J Hum Nutr Diet. 2011 Oct;24(5):441-8. doi: 10.1111/j.1365-277X.2011.01188.x. PMID: 21699587.
- Chen X, Li K, Yang K, et al. Effects of preoperative oral single-dose and doubledose carbohydrates on insulin resistance in patients undergoing gastrectomy:a prospective randomized controlled trial. Clin Nutr. 2021 Apr;40(4):1596-603. doi: 10.1016/j.clnu.2021.03.002. PMID: 33752148.
- 34. Hamamoto H, Yamamoto M, Masubuchi S, et al. The impact of preoperative carbohydrate loading on intraoperative body temperature: a randomized controlled clinical trial. Surg Endosc. 2018 Nov;32(11):4393-401. doi: 10.1007/s00464-018-6273-2. PMID: 29915986.
- 35. He FJ, Wang MJ, Yang K, et al. Effects of Preoperative Oral Nutritional Supplements on Improving Postoperative Early Enteral Feeding Intolerance and Short-Term Prognosis for Gastric Cancer: A Prospective, Single-Center, Single-Blind, Randomized Controlled Trial. Nutrients. 2022 Apr 1;14(7):01. doi: 10.3390/nu14071472. PMID: 35406085.

- 36. Kabata P, Jastrzebski T, Kakol M, et al. Preoperative nutritional support in cancer patients with no clinical signs of malnutrition--prospective randomized controlled trial. Support Care Cancer. 2015 Feb;23(2):365-70. doi: 10.1007/s00520-014-2363-4. PMID: 25091056.
- 37. Kruger J, Meffert PJ, Vogt LJ, et al. Early Parenteral Nutrition in Patients with Biliopancreatic Mass Lesions, a Prospective, Randomized Intervention Trial. PLoS One. 2016;11(11):e0166513. doi: 10.1371/journal.pone.0166513. PMID: 27861546.
- Lee SY, Lee J, Park HM, et al. Impact of Preoperative Immunonutrition on the Outcomes of Colon Cancer Surgery: Results from a Randomized Controlled Trial. Ann Surg. 2021 Aug 4;04:04. doi: 10.1097/SLA.000000000005140. PMID: 34353994.
- Martin RC, 2nd, Agle S, Schlegel M, et al. Efficacy of preoperative immunonutrition in locally advanced pancreatic cancer undergoing irreversible electroporation (IRE). Eur J Surg Oncol. 2017 Apr;43(4):772-9. doi: 10.1016/j.ejso.2017.01.002. PMID: 28162818.
- Rizvanovic N, Nesek Adam V, Causevic S, et al. A randomised controlled study of preoperative oral carbohydrate loading versus fasting in patients undergoing colorectal surgery. Int J Colorectal Dis. 2019 Sep;34(9):1551-61. doi: 10.1007/s00384-019-03349-4. PMID: 31309323.
- Shen Y, Zhao X, Zhao H, et al. Clinical Application of Enteral Nutrition Combined with Microbial Preparation for Intestinal Preparation in Elderly Patients with Colorectal Cancer. Med Sci Monit. 2022 Mar 21;28:e935366. doi: 10.12659/MSM.935366. PMID: 35307727.
- Tesar M, Kozusnikova V, Martinek L, et al. Preoperative nutritional support for patients undergoing elective colorectal cancer surgery - does it really work? Biomed Pap Med Fac Univ Palacky Olomouc Czech Repub. 2022 Mar 1;01:01. doi: 10.5507/bp.2022.009. PMID: 35258042.

- Wang F, Hou MX, Wu XL, et al. Impact of enteral nutrition on postoperative immune function and nutritional status. Genet Mol Res. 2015 Jun 10;14(2):6065-72. doi: 10.4238/2015.June.8.4. PMID: 26125807.
- 44. Xu J, Zhong Y, Jing D, et al. Preoperative enteral immunonutrition improves postoperative outcome in patients with gastrointestinal cancer. World J Surg. 2006 Jul;30(7):1284-9. doi: 10.1007/s00268-005-0756-8. PMID: 16830214.
- 45. Zhao Q, Li Y, Yu B, et al. Effects of Preoperative Enteral Nutrition on Postoperative Recent Nutritional Status in Patients with Siewert II and III Adenocarcinoma of Esophagogastric Junction after Neoadjuvant Chemoradiotherapy. Nutr Cancer. 2018 Aug-Sep;70(6):895-903. doi: 10.1080/01635581.2018.1490780. PMID: 30273016.
- Celik JB, Gezginc K, Ozcelik K, et al. The role of immunonutrition in gynecologic oncologic surgery. Eur J Gynaecol Oncol. 2009;30(4):418-21. PMID: 19761135.
- Cereda E, Turri A, Klersy C, et al. Whey protein isolate supplementation improves body composition, muscle strength, and treatment tolerance in malnourished advanced cancer patients undergoing chemotherapy. Cancer Med. 2019 Nov;8(16):6923-32. doi: 10.1002/cam4.2517. PMID: 31568698.
- 48. Ghosh S, Dempsey G, Skelly R, et al. A double blind, randomised, placebo controlled, feasibility phase III clinical trial of perioperative immune-enhancing enteral nutrition in patients undergoing surgery for advanced head and neck cancer. e-SPEN Journal. 2012;7(3):e107-e14. doi: 10.1016/j.clnme.2011.12.001. PMID: 365552431.
- Haidari F, Abiri B, Iravani M, et al. Randomized Study Design to Test Effects of Vitamin D and Omega-3 Fatty Acid Supplementation as Adjuvant Therapy in Colorectal Cancer Patients. Methods Mol Biol. 2020;2138:337-50. doi: 10.1007/978-1-0716-0471-7_24. PMID: 32219761.

- 50. Healy LA, Ryan A, Doyle SL, et al. Does Prolonged Enteral Feeding With Supplemental Omega-3 Fatty Acids Impact on Recovery Post-esophagectomy: Results of a Randomized Double-Blind Trial. Ann Surg. 2017 Nov;266(5):720-8. doi: 10.1097/SLA.00000000002390. PMID: 28742713.
- 51. Jantharapattana K, Orapipatpong O. Efficacy of EPA-enriched supplement compared with standard formula on body weight changes in malnourished patients with head and neck cancer undergone surgery: a randomized study. Head Neck. 2020 Feb;42(2):188-97. doi: 10.1002/hed.25987. PMID: 31647147.
- 52. Jo S, Choi SH, Heo JS, et al. Missing effect of glutamine supplementation on the surgical outcome after pancreaticoduodenectomy for periampullary tumors: a prospective, randomized, double-blind, controlled clinical trial. World J Surg. 2006 Nov;30(11):1974-82; discussion 83-4. doi: 10.1007/s00268-005-0678-5. PMID: 16927064.
- 53. Oguz M, Kerem M, Bedirli A, et al. L-alanin-Lglutamine supplementation improves the outcome after colorectal surgery for cancer. Colorectal Dis. 2007 Jul;9(6):515-20. doi: 10.1111/j.1463-1318.2006.01174.x. PMID: 17573745.
- 54. Ryan AM, Reynolds JV, Healy L, et al. Enteral nutrition enriched with eicosapentaenoic acid (EPA) preserves lean body mass following esophageal cancer surgery: results of a double-blinded randomized controlled trial. Ann Surg. 2009 Mar;249(3):355-63. doi: 10.1097/SLA.0b013e31819a4789. PMID: 19247018.
- 55. Serrano PE, Parpia S, Simunovic M, et al. Perioperative optimization with nutritional supplements in patients undergoing gastrointestinal surgery for cancer: A randomized, placebo-controlled feasibility clinical trial. Surgery. 2022 May 20;20:20. doi: 10.1016/j.surg.2022.04.001. PMID: 35606184.
- 56. Sorensen LS, Thorlacius-Ussing O, Schmidt EB, et al. Randomized clinical trial of perioperative omega-3 fatty acid supplements in elective colorectal cancer surgery. Br J Surg. 2014 Jan;101(2):33-42. doi: 10.1002/bjs.9361. PMID: 24281905.

- Sorensen LS, Rasmussen SL, Calder PC, et al. Long-term outcomes after perioperative treatment with omega-3 fatty acid supplements in colorectal cancer. BJS Open. 2020 Aug;4(4):678-84. doi: 10.1002/bjs5.50295. PMID: 32391656.
- 58. Sultan J, Griffin SM, Di Franco F, et al. Randomized clinical trial of omega-3 fatty acid-supplemented enteral nutrition versus standard enteral nutrition in patients undergoing oesophagogastric cancer surgery. Br J Surg. 2012 Mar;99(3):346-55. doi: 10.1002/bjs.7799. PMID: 22237467.
- Tsuchiya T, Honda H, Oikawa M, et al. Oral administration of the amino acids cystine and theanine attenuates the adverse events of S-1 adjuvant chemotherapy in gastrointestinal cancer patients. Int J Clin Oncol. 2016 Dec;21(6):1085-90. doi: 10.1007/s10147-016-0996-7. PMID: 27306219.
- 60. Vidal-Casariego A, Calleja-Fernandez A, de Urbina-Gonzalez JJ, et al. Efficacy of glutamine in the prevention of acute radiation enteritis: a randomized controlled trial. JPEN J Parenter Enteral Nutr. 2014 Feb;38(2):205-13. doi: 10.1177/0148607113478191. PMID: 23471208.
- Yeğen SF, Kafadar MT, Gök MA. Comparison of Perioperative Standard and Immunomodulating Enteral Nutrition in Patients Received Major Abdominal Cancer Surgery: a Prospective, Randomized, Controlled Clinical Trial. Indian journal of surgery. 2020;82(5):828-34. doi: 10.1007/s12262-020-02114-0. PMID: CN-02418348.
- Braga M, Gianotti L, Nespoli L, et al. Nutritional approach in malnourished surgical patients: a prospective randomized study. Arch Surg. 2002 Feb;137(2):174-80. doi: 10.1001/archsurg.137.2.174. PMID: 11822956.
- 63. Brown T, Banks M, Hughes BGM, et al. Impact of early prophylactic feeding on long term tube dependency outcomes in patients with head and neck cancer. Oral Oncol. 2017 Sep;72:17-25. doi: 10.1016/j.oraloncology.2017.06.025. PMID: 28797454.

- 64. Ding D, Feng Y, Song B, et al. Effects of preoperative and postoperative enteral nutrition on postoperative nutritional status and immune function of gastric cancer patients. Turk J Gastroenterol. 2015 Mar;26(2):181-5. doi: 10.5152/tjg.2015.3993. PMID: 25835119.
- 65. Falewee MN, Schilf A, Boufflers E, et al. Reduced infections with perioperative immunonutrition in head and neck cancer: exploratory results of a multicenter, prospective, randomized, double-blind study. Clin Nutr. 2014 Oct;33(5):776-84. doi: 10.1016/j.clnu.2013.10.006. PMID: 24182765.
- 66. Gianotti L, Braga M, Nespoli L, et al. A randomized controlled trial of preoperative oral supplementation with a specialized diet in patients with gastrointestinal cancer. Gastroenterology. 2002 Jun;122(7):1763-70. doi: 10.1053/gast.2002.33587. PMID: 12055582.
- Miyata H, Yano M, Yasuda T, et al. Randomized study of clinical effect of enteral nutrition support during neoadjuvant chemotherapy on chemotherapy-related toxicity in patients with esophageal cancer. Clin Nutr. 2012 Jun;31(3):330-6. doi: 10.1016/j.clnu.2011.11.002. PMID: 22169459.
- 68. Mudge LA, Watson DI, Smithers BM, et al. Multicentre factorial randomized clinical trial of perioperative immunonutrition versus standard nutrition for patients undergoing surgical resection of oesophageal cancer. Br J Surg. 2018 Sep;105(10):1262-72. doi: 10.1002/bjs.10923. PMID: 29999517.
- Wong TX, Wong WX, Chen ST, et al. Effects of Perioperative Oral Nutrition Supplementation in Malaysian Patients Undergoing Elective Surgery for Breast and Colorectal Cancers-A Randomised Controlled Trial. Nutrients. 2022 Jan 30;14(3):30. doi: 10.3390/nu14030615. PMID: 35276977.

- 70. Wu S, You D, Lu L, et al. Effect of enteral nutrition support on the curative effect and immune system in patients with rectal cancer during fast track surgery. International journal of clinical and experimental medicine. 2020;13(8):6065-73. PMID: 2004989500.
- 71. Brown TE, Banks MD, Hughes BGM, et al. Randomised controlled trial of early prophylactic feeding vs standard care in patients with head and neck cancer. Br J Cancer. 2017 Jun 27;117(1):15-24. doi: 10.1038/bjc.2017.138. PMID: 28535154.
- 72. Bozzetti F, Gavazzi C, Miceli R, et al. Perioperative total parenteral nutrition in malnourished, gastrointestinal cancer patients: a randomized, clinical trial. JPEN J Parenter Enteral Nutr. 2000 Jan-Feb;24(1):7-14. doi: 10.1177/014860710002400107. PMID: 10638466.
- Chen H, Pan D, Li L. The effects of multi-oil fat emulsion on older patients with gastric cancer. Biomedical Research (India). 2017;28(1):4270-6. PMID: 616782903.
- 74. Feng J, Xu R, Li K, et al. Effects of preoperative oral carbohydrate administration combined with postoperative early oral intake in elderly patients undergoing hepatectomy with acute-phase inflammation and subjective symptom burden: A prospective randomized controlled study. Asian J Surg. 2022 Jan;45(1):386-95. doi: 10.1016/j.asjsur.2021.06.042. PMID: 34362624.
- 75. Ida S, Hiki N, Cho H, et al. Randomized clinical trial comparing standard diet with perioperative oral immunonutrition in total gastrectomy for gastric cancer. Br J Surg. 2017 Mar;104(4):377-83. doi: 10.1002/bjs.10417. PMID: 28072447.
- 76. Aoyama T, Yoshikawa T, Ida S, et al. Effects of perioperative Eicosapentaenoic acidenriched oral nutritional supplement on lean body mass after total gastrectomy for gastric cancer. J Cancer. 2019;10(5):1070-6. doi: 10.7150/jca.29632. PMID: 30854113.

- 77. Aoyama T, Yoshikawa T, Ida S, et al. Effects of perioperative eicosapentaenoic acid-enriched oral nutritional supplement on the long-term oncological outcomes after total gastrectomy for gastric cancer. Oncology letters. 2022;23(5). doi: 10.3892/ol.2022.13272. PMID: CN-02385995.
- Kong SH, Lee HJ, Na JR, et al. Effect of perioperative oral nutritional supplementation in malnourished patients who undergo gastrectomy: A prospective randomized trial. Surgery. 2018 Dec;164(6):1263-70. doi: 10.1016/j.surg.2018.05.017. PMID: 30055788.
- 79. Lidder P, Thomas S, Fleming S, et al. A randomized placebo controlled trial of preoperative carbohydrate drinks and early postoperative nutritional supplement drinks in colorectal surgery. Colorectal Dis. 2013 Jun;15(6):737-45. doi: 10.1111/codi.12130. PMID: 23406311.
- Moya P, Miranda E, Soriano-Irigaray L, et al. Perioperative immunonutrition in normonourished patients undergoing laparoscopic colorectal resection. Surg Endosc. 2016 Nov;30(11):4946-53. doi: 10.1007/s00464-016-4836-7. PMID: 26936601.
- 81. Poon RT, Yu WC, Fan ST, et al. Long-term oral branched chain amino acids in patients undergoing chemoembolization for hepatocellular carcinoma: a randomized trial. Aliment Pharmacol Ther. 2004 Apr 1;19(7):779-88. doi: 10.1111/j.1365-2036.2004.01920.x. PMID: 15043519.
- 82. Ritch CR, Cookson MS, Clark PE, et al. Perioperative Oral Nutrition Supplementation Reduces Prevalence of Sarcopenia following Radical Cystectomy: Results of a Prospective Randomized Controlled Trial. J Urol. 2019 Mar;201(3):470-7. doi: 10.1016/j.juro.2018.10.010. PMID: 30359680.
- Sanchez-Guillen L, Soriano-Irigaray L, Lopez-Rodriguez-Arias F, et al. Effect of Early Peripheral Parenteral Nutrition Support in an Enhanced Recovery Program for Colorectal Cancer Surgery: A Randomized Open Trial. J Clin Med. 2021 Aug 18;10(16):18. doi: 10.3390/jcm10163647. PMID: 34441942.

- 84. Lopez-Rodriguez-Arias F, Sanchez-Guillen L, Lillo-Garcia C, et al. Assessment of Body Composition as an Indicator of Early Peripheral Parenteral Nutrition Therapy in Patients Undergoing Colorectal Cancer Surgery in an Enhanced Recovery Program. Nutrients. 2021 Sep 18;13(9):18. doi: 10.3390/nu13093245. PMID: 34579122.
- 85. Sittitrai P, Ruenmarkkaew D, Booyaprapa S, et al. Effect of a perioperative immune-enhancing diet in clean-contaminated head and neck cancer surgery: A randomized controlled trial. Int J Surg. 2021 Sep;93:106051. doi: 10.1016/j.ijsu.2021.106051. PMID: 34371175.
- 86. Wu GH, Liu ZH, Wu ZH, et al. Perioperative artificial nutrition in malnourished gastrointestinal cancer patients. World J Gastroenterol. 2006 Apr 21;12(15):2441-4. doi: 10.3748/wjg.v12.i15.2441. PMID: 16688841.
- 87. Yan X, Liu L, Zhang Y, et al. Perioperative Enteral Nutrition Improves Postoperative Recovery for Patients with Primary Liver Cancer: A Randomized Controlled Clinical Trial. Nutr Cancer. 2021;73(10):1924-32. doi: 10.1080/01635581.2020.1814824. PMID: 32875913.
- 88. Baldwin C, Spiro A, McGough C, et al. Simple nutritional intervention in patients with advanced cancers of the gastrointestinal tract, non-small cell lung cancers or mesothelioma and weight loss receiving chemotherapy: a randomised controlled trial. J Hum Nutr Diet. 2011 Oct;24(5):431-40. doi: 10.1111/j.1365-277X.2011.01189.x. PMID: 21733143.
- Bourdel-Marchasson I, Blanc-Bisson C, Doussau A, et al. Nutritional advice in older patients at risk of malnutrition during treatment for chemotherapy: a two-year randomized controlled trial. PLoS One. 2014;9(9):e108687. doi: 10.1371/journal.pone.0108687. PMID: 25265392.

- 90. Britton B, Baker AL, Wolfenden L, et al. Eating As Treatment (EAT): A Stepped-Wedge, Randomized Controlled Trial of a Health Behavior Change Intervention Provided by Dietitians to Improve Nutrition in Patients With Head and Neck Cancer Undergoing Radiation Therapy (TROG 12.03). Int J Radiat Oncol Biol Phys. 2019 Feb 1;103(2):353-62. doi: 10.1016/j.ijrobp.2018.09.027. PMID: 30296472.
- 91. Forslund M, Ottenblad A, Ginman C, et al. Effects of a nutrition intervention on acute and late bowel symptoms and health-related quality of life up to 24 months post radiotherapy in patients with prostate cancer: a multicentre randomised controlled trial. Support Care Cancer. 2020 Jul;28(7):3331-42. doi: 10.1007/s00520-019-05182-5. PMID: 31758324.
- 92. Isenring EA, Capra S, Bauer JD. Nutrition intervention is beneficial in oncology outpatients receiving radiotherapy to the gastrointestinal or head and neck area. Br J Cancer. 2004 Aug 2;91(3):447-52. doi: 10.1038/sj.bjc.6601962. PMID: 15226773.
- 93. Loser A, Abel J, Kutz LM, et al. Head and neck cancer patients under (chemo-)radiotherapy undergoing nutritional intervention: Results from the prospective randomized HEADNUT-trial. Radiother Oncol. 2021 Jun;159:82-90. doi: 10.1016/j.radonc.2021.03.019. PMID: 33766702.
- 94. Movahed S, Seilanian Toussi M, Pahlavani N, et al. Effects of medical nutrition therapy compared with general nutritional advice on nutritional status and nutrition-related complications in esophageal cancer patients receiving concurrent chemoradiation: A randomized controlled trial. Mediterranean Journal of Nutrition and Metabolism. 2020;13(3):265-76. doi: 10.3233/mnm-200424. PMID: 632903213.
- 95. Orell H, Schwab U, Saarilahti K, et al. Nutritional Counseling for Head and Neck Cancer Patients Undergoing (Chemo) Radiotherapy-A Prospective Randomized Trial. Front Nutr. 2019;6:22. doi: 10.3389/fnut.2019.00022. PMID: 30937304.

- 96. Pettersson A, Johansson B, Persson C, et al. Effects of a dietary intervention on acute gastrointestinal side effects and other aspects of health-related quality of life: a randomized controlled trial in prostate cancer patients undergoing radiotherapy. Radiother Oncol. 2012 Jun;103(3):333-40. doi: 10.1016/j.radonc.2012.04.006. PMID: 22633817.
- 97. Qiu Y, You J, Wang K, et al. Effect of wholecourse nutrition management on patients with esophageal cancer undergoing concurrent chemoradiotherapy: A randomized control trial. Nutrition. 2020 Jan;69:110558. doi: 10.1016/j.nut.2019.110558. PMID: 31526964.
- 98. Ravasco P, Monteiro-Grillo I, Camilo M. Individualized nutrition intervention is of major benefit to colorectal cancer patients: long-term follow-up of a randomized controlled trial of nutritional therapy. Am J Clin Nutr. 2012 Dec;96(6):1346-53. doi: 10.3945/ajcn.111.018838. PMID: 23134880.
- 99. Regueme SC, Echeverria I, Moneger N, et al. Protein intake, weight loss, dietary intervention, and worsening of quality of life in older patients during chemotherapy for cancer. Support Care Cancer. 2021 Feb;29(2):687-96. doi: 10.1007/s00520-020-05528-4. PMID: 32435967.
- 100. Tu MY, Chien TW, Lin HP, et al. Effects of an intervention on nutrition consultation for cancer patients. Eur J Cancer Care (Engl). 2013 May;22(3):370-6. doi: 10.1111/ecc.12040. PMID: 23320428.
- 101. Um MH, Choi MY, Lee SM, et al. Intensive nutritional counseling improves PG-SGA scores and nutritional symptoms during and after radiotherapy in Korean cancer patients. Support Care Cancer. 2014 Nov;22(11):2997-3005. doi: 10.1007/s00520-014-2304-2. PMID: 24906838.

- 102. van der Werf A, Langius JAE, Beeker A, et al. The effect of nutritional counseling on muscle mass and treatment outcome in patients with metastatic colorectal cancer undergoing chemotherapy: A randomized controlled trial. Clin Nutr. 2020 Oct;39(10):3005-13. doi: 10.1016/j.clnu.2020.01.009. PMID: 32037284.
- 103. Zhang Z, Zhu Y, Zhang L, et al. Nutritional education and counseling program for adult cancer patients during radiotherapy: a cluster-randomized clinical trial. Support Care Cancer. 2022 Apr;30(4):3279-89. doi: 10.1007/s00520-021-06704-w. PMID: 34984549.
- 104. Cheng M, Zhang S, Ning C, et al. Omega-3 Fatty Acids Supplementation Improve Nutritional Status and Inflammatory Response in Patients With Lung Cancer: A Randomized Clinical Trial. Front Nutr. 2021;8:686752. doi: 10.3389/fnut.2021.686752. PMID: 34395492.
- 105. Chitapanarux I, Traisathit P, Chitapanarux T, et al. Arginine, glutamine, and fish oil supplementation in cancer patients treated with concurrent chemoradiotherapy: A randomized control study. Curr Probl Cancer. 2020 Feb;44(1):100482. doi: 10.1016/j.currproblcancer.2019.05.005. PMID: 31146957.
- 106. da Gama Torres HO, Vilela EG, da Cunha AS, et al. Efficacy of glutamine-supplemented parenteral nutrition on short-term survival following allo-SCT: a randomized study. Bone Marrow Transplant. 2008 Jun;41(12):1021-7. doi: 10.1038/bmt.2008.27. PMID: 18317456.
- 107. de Luis DA, Izaola O, Cuellar L, et al. Randomized clinical trial with an enteral arginine-enhanced formula in early postsurgical head and neck cancer patients. Eur J Clin Nutr. 2004 Nov;58(11):1505-8. doi: 10.1038/sj.ejcn.1601999. PMID: 15138461.

- 108. de Luis DA, Izaola O, Aller R, et al. A randomized clinical trial with oral Immunonutrition (omega3-enhanced formula vs. arginine-enhanced formula) in ambulatory head and neck cancer patients. Ann Nutr Metab. 2005 Mar-Apr;49(2):95-9. doi: 10.1159/000084742. PMID: 15802904.
- 109. de Luis DA, Izaola O, Cuellar L, et al. Clinical and biochemical outcomes after a randomized trial with a high dose of enteral arginine formula in postsurgical head and neck cancer patients. Eur J Clin Nutr. 2007 Feb;61(2):200-4. doi: 10.1038/sj.ejcn.1602515. PMID: 16929239.
- 110. de Luis DA, Izaola O, Aller R, et al. A randomized clinical trial with two omega 3 fatty acid enhanced oral supplements in head and neck cancer ambulatory patients. Eur Rev Med Pharmacol Sci. 2008 May-Jun;12(3):177-81. PMID: 18700689.
- 111. De Luis DA, Izaola O, Cuellar L, et al. High dose of arginine enhanced enteral nutrition in postsurgical head and neck cancer patients. A randomized clinical trial. Eur Rev Med Pharmacol Sci. 2009 Jul-Aug;13(4):279-83. PMID: 19694342.
- 112. De Luis DA, Izaola O, Cuellar L, et al. A randomized double-blind clinical trial with two different doses of arginine enhanced enteral nutrition in postsurgical cancer patients. Eur Rev Med Pharmacol Sci. 2010 Nov;14(11):941-5. PMID: 21284343.
- 113. De Luis DA, Izaola O, Terroba MC, et al. Effect of three different doses of arginine enhanced enteral nutrition on nutritional status and outcomes in well nourished postsurgical cancer patients: a randomized single blinded prospective trial. Eur Rev Med Pharmacol Sci. 2015;19(6):950-5. PMID: 25855918.
- 114. Farreras N, Artigas V, Cardona D, et al. Effect of early postoperative enteral immunonutrition on wound healing in patients undergoing surgery for gastric cancer. Clin Nutr. 2005 Feb;24(1):55-65. doi: 10.1016/j.clnu.2004.07.002. PMID: 15681102.

- 115. Fietkau R, Lewitzki V, Kuhnt T, et al. A disease-specific enteral nutrition formula improves nutritional status and functional performance in patients with head and neck and esophageal cancer undergoing chemoradiotherapy: results of a randomized, controlled, multicenter trial. Cancer. 2013 Sep 15;119(18):3343-53. doi: 10.1002/cncr.28197. PMID: 23765693.
- 116. Golkhalkhali B, Rajandram R, Paliany AS, et al. Strain-specific probiotic (microbial cell preparation) and omega-3 fatty acid in modulating quality of life and inflammatory markers in colorectal cancer patients: a randomized controlled trial. Asia Pac J Clin Oncol. 2018 Jun;14(3):179-91. doi: 10.1111/ajco.12758. PMID: 28857425.
- 117. Iwase S, Kawaguchi T, Yotsumoto D, et al. Efficacy and safety of an amino acid jelly containing coenzyme Q10 and L-carnitine in controlling fatigue in breast cancer patients receiving chemotherapy: a multiinstitutional, randomized, exploratory trial (JORTC-CAM01). Support Care Cancer. 2016 Feb;24(2):637-46. doi: 10.1007/s00520-015-2824-4. PMID: 26105516.
- 118. Jiang ZM, Wilmore DW, Wang XR, et al. Randomized clinical trial of intravenous soybean oil alone versus soybean oil plus fish oil emulsion after gastrointestinal cancer surgery. Br J Surg. 2010 Jun;97(6):804-9. doi: 10.1002/bjs.6999. PMID: 20473991.
- Klek S, Kulig J, Szczepanik AM, et al. The clinical value of parenteral immunonutrition in surgical patients. Acta Chir Belg. 2005 Apr;105(2):175-9. PMID: 15906909.
- 120. Lobo DN, Williams RN, Welch NT, et al. Early postoperative jejunostomy feeding with an immune modulating diet in patients undergoing resectional surgery for upper gastrointestinal cancer: a prospective, randomized, controlled, double-blind study. Clin Nutr. 2006 Oct;25(5):716-26. doi: 10.1016/j.clnu.2006.04.007. PMID: 16777271.

- 121. Adiamah A, Rollins KE, Kapeleris A, et al. Postoperative arginine-enriched immune modulating nutrition: Long-term survival results from a randomised clinical trial in patients with oesophagogastric and pancreaticobiliary cancer. Clin Nutr. 2021 Nov;40(11):5482-5. doi: 10.1016/j.clnu.2021.09.040. PMID: 34656029.
- 122. Lu CY, Shih YL, Sun LC, et al. The inflammatory modulation effect of glutamine-enriched total parenteral nutrition in postoperative gastrointestinal cancer patients. Am Surg. 2011 Jan;77(1):59-64. PMID: 21396307.
- 123. Matsuda Y, Habu D, Lee S, et al. Enteral Diet Enriched with omega-3 Fatty Acid Improves Oxygenation After Thoracic Esophagectomy for Cancer: A Randomized Controlled Trial. World J Surg. 2017 Jun;41(6):1584-94. doi: 10.1007/s00268-017-3893-y. PMID: 28138734.
- 124. Miyata H, Yano M, Yasuda T, et al. Randomized study of the clinical effects of omega-3 fatty acid-containing enteral nutrition support during neoadjuvant chemotherapy on chemotherapy-related toxicity in patients with esophageal cancer. Nutrition. 2017 Jan;33:204-10. doi: 10.1016/j.nut.2016.07.004. PMID: 27644137.
- 125. Pathak S, Soni TP, Sharma LM, et al. A Randomized Controlled Trial to Evaluate the Role and Efficacy of Oral Glutamine in the Treatment of Chemo-radiotherapyinduced Oral Mucositis and Dysphagia in Patients with Oropharynx and Larynx Carcinoma. Cureus. 2019 Jun 7;11(6):e4855. doi: 10.7759/cureus.4855. PMID: 31410338.
- 126. Pottel L, Lycke M, Boterberg T, et al. Echium oil is not protective against weight loss in head and neck cancer patients undergoing curative radio(chemo)therapy: a randomised-controlled trial. BMC Complement Altern Med. 2014 Oct 7;14:382. doi: 10.1186/1472-6882-14-382. PMID: 25293388.

- 127. Sun LC, Shih YL, Lu CY, et al. Randomized, controlled study of branched chain amino acid-enriched total parenteral nutrition in malnourished patients with gastrointestinal cancer undergoing surgery. Am Surg. 2008 Mar;74(3):237-42. PMID: 18376691.
- 128. Takeshita S, Ichikawa T, Nakao K, et al. A snack enriched with oral branched-chain amino acids prevents a fall in albumin in patients with liver cirrhosis undergoing chemoembolization for hepatocellular carcinoma. Nutr Res. 2009 Feb;29(2):89-93. doi: 10.1016/j.nutres.2008.12.005. PMID: 19285598.
- 129. Tanca FM, Madeddu C, Macciò A, et al. New perspective on the nutritional approach to cancer-related anorexia/cachexia: preliminary results of a randomised phase III clinical trial with five different arms of treatment. Mediterranean Journal of Nutrition and Metabolism. 2009;2(1):29-36. doi: 10.1007/s12349-009-0041-y. PMID: 358288574.
- 130. Tumas J, Tumiene B, Jurkeviciene J, et al. Nutritional and immune impairments and their effects on outcomes in early pancreatic cancer patients undergoing pancreatoduodenectomy. Clin Nutr. 2020 Nov;39(11):3385-94. doi: 10.1016/j.clnu.2020.02.029. PMID: 32184025.
- 131. Wang J, Li Y, Qi Y. Effect of glutamineenriched nutritional support on intestinal mucosal barrier function, MMP-2, MMP-9 and immune function in patients with advanced gastric cancer during perioperative chemotherapy. Oncol Lett. 2017 Sep;14(3):3606-10. doi: 10.3892/ol.2017.6612. PMID: 28927119.
- 132. Wang WP, Yan XL, Ni YF, et al. Effects of lipid emulsions in parenteral nutrition of esophageal cancer surgical patients receiving enteral nutrition: a comparative analysis. Nutrients. 2013 Dec 27;6(1):111-23. doi: 10.3390/nu6010111. PMID: 24379010.
- 133. Wang X, Pan L, Zhang P, et al. Enteral nutrition improves clinical outcome and shortens hospital stay after cancer surgery. J Invest Surg. 2010 Dec;23(6):309-13. doi: 10.3109/08941939.2010.519428. PMID: 21208095.

- 134. Wu Z, Qin J, Pu L. Omega-3 fatty acid improves the clinical outcome of hepatectomized patients with hepatitis B virus (HBV)associated hepatocellular carcinoma. J Biomed Res. 2012 Nov;26(6):395-9. doi: 10.7555/JBR.26.20120058. PMID: 23554777.
- 135. Yang J, Zhang X, Li K, et al. Effects of EN combined with PN enriched with n-3 polyunsaturated fatty acids on immune related indicators and early rehabilitation of patients with gastric cancer: A randomized controlled trial. Clin Nutr. 2022 Apr 6;41(6):1163-70. doi: 10.1016/j.clnu.2022.03.018. PMID: 35500316.
- 136. Yeh KY, Wang HM, Chang JW, et al. Omega-3 fatty acid-, micronutrient-, and probioticenriched nutrition helps body weight stabilization in head and neck cancer cachexia. Oral Surg Oral Med Oral Pathol Oral Radiol. 2013 Jul;116(1):41-8. doi: 10.1016/j.0000.2013.01.015. PMID: 23562359.
- 137. Zhang B, Wei G, Li R, et al. n-3 fatty acid-based parenteral nutrition improves postoperative recovery for cirrhotic patients with liver cancer: A randomized controlled clinical trial. Clin Nutr. 2017 Oct;36(5):1239-44. doi: 10.1016/j.clnu.2016.08.002. PMID: 27614675.
- 138. Zhu MW, Tang DN, Hou J, et al. Impact of fish oil enriched total parenteral nutrition on elderly patients after colorectal cancer surgery. Chin Med J (Engl). 2012 Jan;125(2):178-81. doi: 10.3760/cma.j.issn.0366-6999.2012.02.003. PMID: 22340541.
- 139. Gardner A, Mattiuzzi G, Faderl S, et al. Randomized comparison of cooked and noncooked diets in patients undergoing remission induction therapy for acute myeloid leukemia. J Clin Oncol. 2008 Dec 10;26(35):5684-8. doi: 10.1200/JCO.2008.16.4681. PMID: 18955453.

- 140. Ingersoll GL, Wasilewski A, Haller M, et al. Effect of concord grape juice on chemotherapy-induced nausea and vomiting: results of a pilot study. Oncol Nurs Forum. 2010 Mar;37(2):213-21. doi: 10.1188/10.ONF.213-221. PMID: 20189927.
- 141. Jatoi A, Qin R, Satele D, et al. "Enjoy glass of wine before eating:" a randomized trial to test the orexigenic effects of this advice in advanced cancer patients. Support Care Cancer. 2016 Sep;24(9):3739-46. doi: 10.1007/s00520-016-3190-6. PMID: 27039205.
- 142. Khodabakhshi A, Akbari ME, Mirzaei HR, et al. Feasibility, Safety, and Beneficial Effects of MCT-Based Ketogenic Diet for Breast Cancer Treatment: A Randomized Controlled Trial Study. Nutr Cancer. 2020;72(4):627-34. doi: 10.1080/01635581.2019.1650942. PMID: 31496287.
- 143. Khodabakhshi A, Seyfried TN, Kalamian M, et al. Does a ketogenic diet have beneficial effects on quality of life, physical activity or biomarkers in patients with breast cancer: a randomized controlled clinical trial. Nutr J. 2020 Aug 22;19(1):87. doi: 10.1186/s12937-020-00596-y. PMID: 32828130.
- 144. Lugtenberg RT, de Groot S, Kaptein AA, et al. Quality of life and illness perceptions in patients with breast cancer using a fasting mimicking diet as an adjunct to neoadjuvant chemotherapy in the phase 2 DIRECT (BOOG 2013-14) trial. Breast Cancer Res Treat. 2021 Feb;185(3):741-58. doi: 10.1007/s10549-020-05991-x. PMID: 33179154.
- 145. Miyakawa A, Kodera S, Sakuma Y, et al. Effects of Early Initiation of Solid Versus Liquid Diet after Endoscopic Submucosal Dissection on Quality of Life and Postoperative Outcomes: A Prospective Pilot Randomized Controlled Trial. Digestion. 2019;100(3):160-9. doi: 10.1159/000494490. PMID: 30554216.

- 146. Voss M, Wagner M, von Mettenheim N, et al. ERGO2: A Prospective, Randomized Trial of Calorie-Restricted Ketogenic Diet and Fasting in Addition to Reirradiation for Malignant Glioma. Int J Radiat Oncol Biol Phys. 2020 Nov 15;108(4):987-95. doi: 10.1016/j.ijrobp.2020.06.021. PMID: 32619561.
- 147. Wedlake LJ, McGough C, Shaw C, et al. Clinical trial: Efficacy of a low or modified fat diet for the prevention of gastrointestinal toxicity in patients receiving radiotherapy treatment for pelvic malignancies. J Hum Nutr Diet. 2012 Jun;25(3):247-59. doi: 10.1111/j.1365-277X.2012.01248.x. PMID: 22515941.
- 148. Berkelmans GHK, Fransen LFC, Dolmans-Zwartjes ACP, et al. Direct Oral Feeding Following Minimally Invasive Esophagectomy (NUTRIENT II trial): An International, Multicenter, Open-label Randomized Controlled Trial. Ann Surg. 2020 Jan;271(1):41-7. doi: 10.1097/SLA.00000000003278. PMID: 31090563.
- 149. Boelens PG, Heesakkers FF, Luyer MD, et al. Reduction of postoperative ileus by early enteral nutrition in patients undergoing major rectal surgery: prospective, randomized, controlled trial. Ann Surg. 2014 Apr;259(4):649-55. doi: 10.1097/SLA.00000000000288. PMID: 24169163.
- 150. Bozzetti F, Braga M, Gianotti L, et al. Postoperative enteral versus parenteral nutrition in malnourished patients with gastrointestinal cancer: a randomised multicentre trial. The Lancet. 2001;358(9292):1487-92.
- 151. Braga M, Gianotti L, Gentilini O, et al. Early postoperative enteral nutrition improves gut oxygenation and reduces costs compared with total parenteral nutrition. Critical care medicine. 2001;29(2):242-8.
- 152. Dag A, Colak T, Turkmenoglu O, et al. A randomized controlled trial evaluating early versus traditional oral feeding after colorectal surgery. Clinics (Sao Paulo). 2011;66(12):2001-5. doi: 10.1590/s1807-59322011001200001. PMID: 22189721.

- 153. Feo CV, Romanini B, Sortini D, et al. Early oral feeding after colorectal resection: a randomized controlled study. ANZ J Surg. 2004 May;74(5):298-301. doi: 10.1111/j.1445-1433.2004.02985.x. PMID: 15144242.
- 154. Gao L, Zhao Z, Zhang L, et al. Effect of early oral feeding on gastrointestinal function recovery in postoperative gastric cancer patients: a prospective study. J BUON. 2019 Jan-Feb;24(1):194-200. PMID: 30941970.
- 155. Huang D, Sun Z, Huang J, et al. Early enteral nutrition in combination with parenteral nutrition in elderly patients after surgery due to gastrointestinal cancer. Int J Clin Exp Med. 2015;8(8):13937-45. PMID: 26550350.
- 156. Hyltander A, Bosaeus I, Svedlund J, et al. Supportive nutrition on recovery of metabolism, nutritional state, health-related quality of life, and exercise capacity after major surgery: a randomized study. Clin Gastroenterol Hepatol. 2005 May;3(5):466-74. doi: 10.1016/s1542-3565(05)00151-5. PMID: 15880316.
- 157. Kita R, Miyata H, Sugimura K, et al. Clinical effect of enteral nutrition support during neoadjuvant chemotherapy on the preservation of skeletal muscle mass in patients with esophageal cancer. Clin Nutr. 2021 Jun;40(6):4380-5. doi: 10.1016/j.clnu.2021.01.007. PMID: 33526287.
- 158. Kurbanalievich SD, Vladimirovich DV, Kabildina NA. Nutritional Support for Patients with Diseases of Hepatopancreotoduodenal Zone in the Early After the Operational Period. Open Access Macedonian Journal of Medical Sciences. 2020;8(B):769-74. doi: 10.3889/oamjms.2020.4717. PMID: 2005564409.
- 159. Li B, Liu HY, Guo SH, et al. The postoperative clinical outcomes and safety of early enteral nutrition in operated gastric cancer patients. J BUON. 2015 Mar-Apr;20(2):468-72. PMID: 26011337.
- 160. Liu C, Du Z, Lou C, et al. Enteral nutrition is superior to total parenteral nutrition for pancreatic cancer patients who underwent pancreaticoduodenectomy. Asia Pac J Clin Nutr. 2011;20(2):154-60. PMID: 21669582.

- 161. Luo Z, Wang J, Zhang Z, et al. Efficacy of Early Enteral Immunonutrition on Immune Function and Clinical Outcome for Postoperative Patients With Gastrointestinal Cancer. JPEN J Parenter Enteral Nutr. 2018 May;42(4):758-65. doi: 10.1177/0148607117715439. PMID: 28666095.
- 162. Ma BQ, Chen SY, Jiang ZB, et al. Effect of postoperative early enteral nutrition on clinical outcomes and immune function of cholangiocarcinoma patients with malignant obstructive jaundice. World J Gastroenterol. 2020 Dec 14;26(46):7405-15. doi: 10.3748/wjg.v26.i46.7405. PMID: 33362392.
- 163. Mahmoodzadeh H, Shoar S, Sirati F, et al. Early initiation of oral feeding following upper gastrointestinal tumor surgery: a randomized controlled trial. Surg Today. 2015 Feb;45(2):203-8. doi: 10.1007/s00595-014-0937-x. PMID: 24875466.
- 164. Minig L, Biffi R, Zanagnolo V, et al. Reduction of postoperative complication rate with the use of early oral feeding in gynecologic oncologic patients undergoing a major surgery: a randomized controlled trial. Ann Surg Oncol. 2009 Nov;16(11):3101-10. doi: 10.1245/s10434-009-0681-4. PMID: 19760046.
- 165. Perinel J, Mariette C, Dousset B, et al. Early Enteral Versus Total Parenteral Nutrition in Patients Undergoing Pancreaticoduodenectomy: A Randomized Multicenter Controlled Trial (Nutri-DPC). Ann Surg. 2016 Nov;264(5):731-7. doi: 10.1097/SLA.000000000001896. PMID: 27429039.
- 166. Roberts S, Miller J, Pineiro L, et al. Total parenteral nutrition vs oral diet in autologous hematopoietic cell transplant recipients. Bone Marrow Transplant. 2003 Oct;32(7):715-21. doi: 10.1038/sj.bmt.1704204. PMID: 13130320.
- 167. Ryu J, Nam BH, Jung YS. Clinical outcomes comparing parenteral and nasogastric tube nutrition after laryngeal and pharyngeal cancer surgery. Dysphagia. 2009 Dec;24(4):378-86. doi: 10.1007/s00455-009-9213-4. PMID: 19255706.

- 168. Sadasivan A, Faizal B, Kumar M. Nasogastric and percutaneous endoscopic gastrostomy tube use in advanced head and neck cancer patients: a comparative study. J Pain Palliat Care Pharmacother. 2012 Sep;26(3):226-32. doi: 10.3109/15360288.2012.702199. PMID: 22973911.
- 169. Seven H, Calis AB, Turgut S. A randomized controlled trial of early oral feeding in laryngectomized patients. Laryngoscope. 2003 Jun;113(6):1076-9. doi: 10.1097/00005537-200306000-00030. PMID: 12782826.
- 170. Sousa AA, Porcaro-Salles JM, Soares JM, et al. Does early oral feeding increase the likelihood of salivary fistula after total laryngectomy? J Laryngol Otol. 2014 Apr 15;128(4):1-7. doi: 10.1017/S0022215114000747. PMID: 24736040.
- 171. Sun HB, Li Y, Liu XB, et al. Early Oral Feeding Following McKeown Minimally Invasive Esophagectomy: An Open-label, Randomized, Controlled, Noninferiority Trial. Ann Surg. 2018 Mar;267(3):435-42. doi: 10.1097/SLA.00000000002304. PMID: 28549015.
- 172. Tao Z, Zhang Y, Zhu S, et al. A Prospective Randomized Trial Comparing Jejunostomy and Nasogastric Feeding in Minimally Invasive McKeown Esophagectomy. J Gastrointest Surg. 2020 Oct;24(10):2187-96. doi: 10.1007/s11605-019-04390-y. PMID: 31512101.
- 173. van Barneveld KW, Smeets BJ, Heesakkers FF, et al. Beneficial Effects of Early Enteral Nutrition After Major Rectal Surgery: A Possible Role for Conditionally Essential Amino Acids? Results of a Randomized Clinical Trial. Crit Care Med. 2016 Jun;44(6):e353-61. doi: 10.1097/CCM.00000000001640. PMID: 26937858.
- 174. Wang Q, Yang KL, Guo BY, et al. Safety of early oral feeding after total laparoscopic radical gastrectomy for gastric cancer (SOFTLY-1): a single-center randomized controlled trial. Cancer Manag Res. 2019;11:4839-46. doi: 10.2147/CMAR.S199552. PMID: 31239762.

- 175. Wang J, Zhao J, Zhang Y, et al. Early enteral nutrition and total parenteral nutrition on the nutritional status and blood glucose in patients with gastric cancer complicated with diabetes mellitus after radical gastrectomy. Exp Ther Med. 2018 Jul;16(1):321-7. doi: 10.3892/etm.2018.6168. PMID: 29896256.
- 176. Xiao-Bo Y, Qiang L, Xiong Q, et al. Efficacy of early postoperative enteral nutrition in supporting patients after esophagectomy. Minerva Chir. 2014 Feb;69(1):37-46. PMID: 24504222.
- 177. Xu B, Ma J, Chen X, et al. The effect of early enteral nutrition on the postoperative immune function and inflammatory indexes in patients with digestive tract cancers. International journal of clinical and experimental medicine. 2020;13(4):2541-7. PMID: 2004238521.
- 178. Zhang J, Si X, Li W, et al. Effect of peripherally inserted central catheter (PICC) parenteral nutrition on immune function and nutritional support after radical gastrectomy for gastric cancer. Pak J Pharm Sci. 2019 May;32(3 Special):1441-5. PMID: 31551229.
- 179. Bozzetti F, Santarpia L, Pironi L, et al. The prognosis of incurable cachectic cancer patients on home parenteral nutrition: a multi-centre observational study with prospective follow-up of 414 patients. Ann Oncol. 2014 Feb;25(2):487-93. doi: 10.1093/annonc/mdt549. PMID: 24406425.
- 180. Minig L, Biffi R, Zanagnolo V, et al. Early oral versus "traditional" postoperative feeding in gynecologic oncology patients undergoing intestinal resection: a randomized controlled trial. Ann Surg Oncol. 2009 Jun;16(6):1660-8. doi: 10.1245/s10434-009-0444-2. PMID: 19330379.
- 181. Akita H, Takahashi H, Asukai K, et al. The utility of nutritional supportive care with an eicosapentaenoic acid (EPA)-enriched nutrition agent during pre-operative chemoradiotherapy for pancreatic cancer: Prospective randomized control study. Clin Nutr ESPEN. 2019 Oct;33:148-53. doi: 10.1016/j.clnesp.2019.06.003. PMID: 31451252.

- 182. Baker J, Janda M, Graves N, et al. Quality of life after early enteral feeding versus standard care for proven or suspected advanced epithelial ovarian cancer: Results from a randomised trial. Gynecol Oncol. 2015 Jun;137(3):516-22. doi: 10.1016/j.ygyno.2015.03.048. PMID: 25827292.
- 183. Barlow R, Price P, Reid TD, et al. Prospective multicentre randomised controlled trial of early enteral nutrition for patients undergoing major upper gastrointestinal surgical resection. Clin Nutr. 2011 Oct;30(5):560-6. doi: 10.1016/j.clnu.2011.02.006. PMID: 21601319.
- 184. Bouleuc C, Anota A, Cornet C, et al. Impact on Health-Related Quality of Life of Parenteral Nutrition for Patients with Advanced Cancer Cachexia: Results from a Randomized Controlled Trial. Oncologist. 2020 May;25(5):e843-e51. doi: 10.1634/theoncologist.2019-0856. PMID: 32212354.
- 185. Cereda E, Cappello S, Colombo S, et al. Nutritional counseling with or without systematic use of oral nutritional supplements in head and neck cancer patients undergoing radiotherapy. Radiother Oncol. 2018 Jan;126(1):81-8. doi: 10.1016/j.radonc.2017.10.015. PMID: 29111172.
- 186. Chen X, Zhao G, Zhu L. Home enteral nutrition for postoperative elderly patients with esophageal cancer. Ann Palliat Med. 2021 Jan;10(1):278-84. doi: 10.21037/apm-20-2197. PMID: 33545764.
- 187. Chen T, Jiang W, He G. Effect of family enteral nutrition on nutritional status in elderly patients with esophageal carcinoma after minimally invasive radical surgery: a randomized trial. Ann Palliat Med. 2021 Jun;10(6):6760-7. doi: 10.21037/apm-21-1219. PMID: 34237976.
- 188. Chu L, Ren Y, Zhang L, et al. Evaluation of effects of nutritional risk assessment and enteral and parenteral nutritional interventions after esophageal cancer surgery. International journal of clinical and experimental medicine. 2018;11(5):5110-6. PMID: 622367981.

- 189. Deibert CM, Silva MV, RoyChoudhury A, et al. A Prospective Randomized Trial of the Effects of Early Enteral Feeding After Radical Cystectomy. Urology. 2016 Oct;96:69-73. doi: 10.1016/j.urology.2016.06.045. PMID: 27402372.
- 190. Faccio AA, Mattos C, Santos E, et al. Oral Nutritional Supplementation in Cancer Patients Who Were Receiving Chemo/Chemoradiation Therapy: A Multicenter, Randomized Phase II Study. Nutr Cancer. 2021;73(3):442-9. doi: 10.1080/01635581.2020.1758170. PMID: 32363940.
- 191. Gavazzi C, Colatruglio S, Valoriani F, et al. Impact of home enteral nutrition in malnourished patients with upper gastrointestinal cancer: A multicentre randomised clinical trial. Eur J Cancer. 2016 Sep;64:107-12. doi: 10.1016/j.ejca.2016.05.032. PMID: 27391922.
- 192. Huang S, Piao Y, Cao C, et al. A prospective randomized controlled trial on the value of prophylactic oral nutritional supplementation in locally advanced nasopharyngeal carcinoma patients receiving chemo-radiotherapy. Oral Oncol. 2020 Dec;111:105025. doi: 10.1016/j.oraloncology.2020.105025. PMID: 33032180.
- 193. Imamura H, Nishikawa K, Kishi K, et al. Effects of an Oral Elemental Nutritional Supplement on Post-gastrectomy Body Weight Loss in Gastric Cancer Patients: A Randomized Controlled Clinical Trial. Ann Surg Oncol. 2016 Sep;23(9):2928-35. doi: 10.1245/s10434-016-5221-4. PMID: 27084538.
- 194. Kimura Y, Nishikawa K, Kishi K, et al. Longterm effects of an oral elemental nutritional supplement on post-gastrectomy body weight loss in gastric cancer patients (KSES002). Ann Gastroenterol Surg. 2019 Nov;3(6):648-56. doi: 10.1002/ags3.12290. PMID: 31788653.

- 195. Jiang W, Ding H, Li W, et al. Benefits of Oral Nutritional Supplements in Patients with Locally Advanced Nasopharyngeal Cancer during Concurrent Chemoradiotherapy: An Exploratory Prospective Randomized Trial. Nutr Cancer. 2018 Nov-Dec;70(8):1299-307. doi: 10.1080/01635581.2018.1557222. PMID: 30633580.
- 196. Jin Y, Yong C, Ren K, et al. Effects of Post-Surgical Parenteral Nutrition on Patients with Gastric Cancer. Cell Physiol Biochem. 2018;49(4):1320-8. doi: 10.1159/000493410. PMID: 30205371.
- 197. Kanat O, Cubukcu E, Avci N, et al. Comparison of three different treatment modalities in the management of cancer cachexia. Tumori. 2013 Mar-Apr;99(2):229-33. doi: 10.1700/1283.14197. PMID: 23748819.
- 198. Katada C, Fukazawa S, Sugawara M, et al. Randomized study of prevention of gastrointestinal toxicities by nutritional support using an amino acid-rich elemental diet during chemotherapy in patients with esophageal cancer (KDOG 1101). Esophagus. 2021 Apr;18(2):296-305. doi: 10.1007/s10388-020-00787-w. PMID: 33009977.
- 199. Klek S, Kulig J, Sierzega M, et al. Standard and immunomodulating enteral nutrition in patients after extended gastrointestinal surgery--a prospective, randomized, controlled clinical trial. Clin Nutr. 2008 Aug;27(4):504-12. doi: 10.1016/j.clnu.2008.04.010. PMID: 18571296.
- 200. Klek S, Sierzega M, Szybinski P, et al. The immunomodulating enteral nutrition in malnourished surgical patients - a prospective, randomized, double-blind clinical trial. Clin Nutr. 2011 Jun;30(3):282-8. doi: 10.1016/j.clnu.2010.10.001. PMID: 21074910.
- 201. Klek S, Scislo L, Walewska E, et al. Enriched enteral nutrition may improve short-term survival in stage IV gastric cancer patients: A randomized, controlled trial. Nutrition. 2017 Apr;36:46-53. doi: 10.1016/j.nut.2016.03.016. PMID: 28336107.

- 202. Klek S, Sierzega M, Szybinski P, et al. Perioperative nutrition in malnourished surgical cancer patients - a prospective, randomized, controlled clinical trial. Clin Nutr. 2011 Dec;30(6):708-13. doi: 10.1016/j.clnu.2011.07.007. PMID: 21820770.
- 203. Klek S, Szybinski P, Szczepanek K. Perioperative immunonutrition in surgical cancer patients: a summary of a decade of research. World J Surg. 2014 Apr;38(4):803-12. doi: 10.1007/s00268-013-2323-z. PMID: 24178185.
- 204. Li B, Liu HY, Guo SH, et al. Impact of early postoperative enteral nutrition on clinical outcomes in patients with gastric cancer. Genet Mol Res. 2015 Jun 29;14(2):7136-41. doi: 10.4238/2015.June.29.7. PMID: 26125924.
- 205. Li C, Ni L, Liu C. Early enteral immunonutrition support protects the cellular and humoral immune functions of patients with pancreatic cancer after chemotherapy. International journal of clinical and experimental medicine. 2020;13(2):700-8. PMID: 2003914110.
- 206. Lyu J, Shi A, Li T, et al. Effects of Enteral Nutrition on Patients With Oesophageal Carcinoma Treated With Concurrent Chemoradiotherapy: A Prospective, Multicentre, Randomised, Controlled Study. Front Oncol. 2022;12:839516. doi: 10.3389/fonc.2022.839516. PMID: 35280748.
- 207. McGough C, Wedlake L, Baldwin C, et al. Clinical trial: normal diet vs. partial replacement with oral E028 formula for the prevention of gastrointestinal toxicity in cancer patients undergoing pelvic radiotherapy. Aliment Pharmacol Ther. 2008 Jun 1;27(11):1132-9. doi: 10.1111/j.1365-2036.2008.03665.x. PMID: 18315590.
- 208. Meng Q, Tan S, Jiang Y, et al. Post-discharge oral nutritional supplements with dietary advice in patients at nutritional risk after surgery for gastric cancer: A randomized clinical trial. Clin Nutr. 2021 Jan;40(1):40-6. doi: 10.1016/j.clnu.2020.04.043. PMID: 32563598.

- 209. Tan S, Meng Q, Jiang Y, et al. Impact of oral nutritional supplements in post-discharge patients at nutritional risk following colorectal cancer surgery: A randomised clinical trial. Clin Nutr. 2021 Jan;40(1):47-53. doi: 10.1016/j.clnu.2020.05.038. PMID: 32563599.
- 210. Miyazaki Y, Omori T, Fujitani K, et al. Oral nutritional supplements versus a regular diet alone for body weight loss after gastrectomy: a phase 3, multicenter, openlabel randomized controlled trial. Gastric Cancer. 2021 Sep;24(5):1150-9. doi: 10.1007/s10120-021-01188-3. PMID: 33835329.
- 211. Nie J, Su X, Wei L, et al. Early enteral nutrition support for colon carcinoma patients can improve immune function and promote physical recovery. Am J Transl Res. 2021;13(12):14102-8. PMID: 35035754.
- 212. Ohkura Y, Ueno M, Shindoh J, et al. Randomized controlled trial on efficacy of oligomeric formula (HINE E-GEL(R)) versus polymeric formula (MEIN(R)) enteral nutrition after esophagectomy for esophageal cancer with gastric tube reconstruction. Dis Esophagus. 2019 May 1;32(5):01. doi: 10.1093/dote/doy084. PMID: 30169605.
- 213. Okabayashi T, Iyoki M, Sugimoto T, et al. Oral supplementation with carbohydrate- and branched-chain amino acid-enriched nutrients improves postoperative quality of life in patients undergoing hepatic resection. Amino Acids. 2011 Apr;40(4):1213-20. doi: 10.1007/s00726-010-0748-3. PMID: 20852905.
- 214. Ravasco P, Monteiro-Grillo I, Marques Vidal P, et al. Impact of nutrition on outcome: a prospective randomized controlled trial in patients with head and neck cancer undergoing radiotherapy. Head Neck. 2005 Aug;27(8):659-68. doi: 10.1002/hed.20221. PMID: 15920748.
- 215. Ravasco P, Monteiro-Grillo I, Vidal PM, et al. Dietary counseling improves patient outcomes: a prospective, randomized, controlled trial in colorectal cancer patients undergoing radiotherapy. J Clin Oncol. 2005 Mar 1;23(7):1431-8. doi: 10.1200/JCO.2005.02.054. PMID: 15684319.

- 216. Sanchez-Lara K, Turcott JG, Juarez-Hernandez E, et al. Effects of an oral nutritional supplement containing eicosapentaenoic acid on nutritional and clinical outcomes in patients with advanced non-small cell lung cancer: randomised trial. Clin Nutr. 2014 Dec;33(6):1017-23. doi: 10.1016/j.clnu.2014.03.006. PMID: 24746976.
- 217. Scislo L, Pach R, Nowak A, et al. The Impact of Postoperative Enteral Immunonutrition on Postoperative Complications and Survival in Gastric Cancer Patients - Randomized Clinical Trial. Nutr Cancer. 2018 Apr;70(3):453-9. doi: 10.1080/01635581.2018.1445770. PMID: 29533110.
- 218. Shimizu N, Oki E, Tanizawa Y, et al. Effect of early oral feeding on length of hospital stay following gastrectomy for gastric cancer: a Japanese multicenter, randomized controlled trial. Surg Today. 2018 Sep;48(9):865-74. doi: 10.1007/s00595-018-1665-4. PMID: 29721714.
- 219. Sim E, Kim JM, Lee SM, et al. The Effect of Omega-3 Enriched Oral Nutrition Supplement on Nutritional Indices and Quality of Life in Gastrointestinal Cancer Patients: A Randomized Clinical Trial. Asian Pac J Cancer Prev. 2022 Feb 1;23(2):485-94. doi: 10.31557/APJCP.2022.23.2.485. PMID: 35225460.
- 220. Vidal A, Arnold N, Vartolomei MD, et al. Oncological and functional outcomes of postoperative total parenteral nutrition after radical cystectomy in bladder cancer patients: A single-center randomized trial. Int J Urol. 2016 Dec;23(12):992-9. doi: 10.1111/iju.13228. PMID: 27770454.
- 221. Wang J, Wang L, Zhao M, et al. Effect of Early Enteral Nutrition Support Combined with Chemotherapy on Related Complications and Immune Function of Patients after Radical Gastrectomy. J Healthc Eng. 2022;2022:1531738. doi: 10.1155/2022/1531738. PMID: 35126900.

- 222. Wu W, Zhong M, Zhu DM, et al. Effect of Early Full-Calorie Nutrition Support Following Esophagectomy: A Randomized Controlled Trial. JPEN J Parenter Enteral Nutr. 2017 Sep;41(7):1146-54. doi: 10.1177/0148607116651509. PMID: 27208039.
- 223. Xie H, Chen X, Xu L, et al. A randomized controlled trial of oral nutritional supplementation versus standard diet following McKeown minimally invasive esophagectomy in patients with esophageal malignancy: a pilot study. Ann Transl Med. 2021 Nov;9(22):1674. doi: 10.21037/atm-21-5422. PMID: 34988183.
- 224. Yang L, Gao J, Zhou Y, et al. Effect of Oral Nutritional Supplements on Patients with Esophageal Cancer During Radiotherapy. Cancer Biother Radiopharm. 2020 Aug 20;20:20. doi: 10.1089/cbr.2020.3888. PMID: 32833549.
- 225. Yao R, Zhang T, Zhang J, et al. Effects of postoperative enteral nutrition combined with adjuvant radiotherapy on inflammatory response, nutrition, healing and prognosis in patients receiving radical surgery for esophageal carcinoma. J BUON. 2019 Jul-Aug;24(4):1673-8. PMID: 31646824.
- 226. Zhang Y, Liu L, Li D, et al. Effectiveness of Noninvasive Positive Pressure Ventilation Combined with Enteral Nutrition in the Treatment of Patients with Combined Respiratory Failure after Lung Cancer Surgery and Its Effect on Blood Gas Indexes. Emergency Medicine International Print. 2022;2022:1508082. doi: <u>https://dx.doi.org/10.1155/2022/1508082</u>. PMID: 35811605.
- 227. Zhao M, Li XG, Ma YY, et al. Application of enteral nutrition during perichemotherapy of acute non-lymphocytic leukemia. Journal of Chemical and Pharmaceutical Research. 2014;6(6):768-71. PMID: 602985290.
- 228. Zhu MW, Yang X, Xiu DR, et al. Effect of oral nutritional supplementation on the postdischarge nutritional status and quality of life of gastrointestinal cancer patients after surgery: a multi-center study. Asia Pac J Clin Nutr. 2019;28(3):450-6. doi: 10.6133/apjcn.201909_28(3).0004. PMID: 31464391.

- 229. Zietarska M, Krawczyk-Lipiec J, Kraj L, et al. Chemotherapy-Related Toxicity, Nutritional Status and Quality of Life in Precachectic Oncologic Patients with, or without, High Protein Nutritional Support. A Prospective, Randomized Study. Nutrients. 2017 Oct 11;9(10):11. doi: 10.3390/nu9101108. PMID: 29019951.
- 230. Meng L, Wei J, Ji R, et al. Effect of Early Nutrition Intervention on Advanced Nasopharyngeal Carcinoma Patients Receiving Chemoradiotherapy. J Cancer. 2019;10(16):3650-6. doi: 10.7150/jca.33475. PMID: 31333782.
- 231. Abdollahi R, Najafi S, Razmpoosh E, et al. The Effect of Dietary Intervention Along with Nutritional Education on Reducing the Gastrointestinal Side Effects Caused by Chemotherapy Among Women with Breast Cancer. Nutr Cancer. 2019;71(6):922-30. doi: 10.1080/01635581.2019.1590608. PMID: 30945949.
- 232. Demark-Wahnefried W, Case LD, Blackwell K, et al. Results of a diet/exercise feasibility trial to prevent adverse body composition change in breast cancer patients on adjuvant chemotherapy. Clin Breast Cancer. 2008 Feb;8(1):70-9. doi: 10.3816/CBC.2008.n.005. PMID: 18501061.
- 233. Lin JX, Chen XW, Chen ZH, et al. A multidisciplinary team approach for nutritional interventions conducted by specialist nurses in patients with advanced colorectal cancer undergoing chemotherapy: A clinical trial. Medicine (Baltimore). 2017 Jun;96(26):e7373. doi: 10.1097/MD.00000000007373. PMID: 28658162.
- 234. Poulsen GM, Pedersen LL, Osterlind K, et al. Randomized trial of the effects of individual nutritional counseling in cancer patients. Clin Nutr. 2014 Oct;33(5):749-53. doi: 10.1016/j.clnu.2013.10.019. PMID: 24269077.
- 235. Qin N, Jiang G, Zhang X, et al. The Effect of Nutrition Intervention With Oral Nutritional Supplements on Ovarian Cancer Patients Undergoing Chemotherapy. Front Nutr. 2021;8:685967. doi: 10.3389/fnut.2021.685967. PMID: 34249995.

- 236. Silander E, Nyman J, Bove M, et al. Impact of prophylactic percutaneous endoscopic gastrostomy on malnutrition and quality of life in patients with head and neck cancer: a randomized study. Head Neck. 2012 Jan;34(1):1-9. doi: 10.1002/hed.21700. PMID: 21374756.
- 237. Silander E, Jacobsson I, Berteus-Forslund H, et al. Energy intake and sources of nutritional support in patients with head and neck cancer--a randomised longitudinal study. Eur J Clin Nutr. 2013 Jan;67(1):47-52. doi: 10.1038/ejcn.2012.172. PMID: 23169469.
- 238. Skaarud KJ, Veierod MB, Lergenmuller S, et al. Body weight, body composition and survival after 1 year: follow-up of a nutritional intervention trial in allo-HSCT recipients. Bone Marrow Transplant. 2019 Dec;54(12):2102-9. doi: 10.1038/s41409-019-0638-6. PMID: 31455897.
- 239. Song G, Liu H. Effect of Hospital to Home nutrition management model on postoperative clinical outcomes of patients with laryngeal carcinoma. Oncol Lett. 2017 Oct;14(4):4059-64. doi: 10.3892/ol.2017.6709. PMID: 28943912.
- 240. Uster A, Ruefenacht U, Ruehlin M, et al. Influence of a nutritional intervention on dietary intake and quality of life in cancer patients: a randomized controlled trial. Nutrition. 2013 Nov-Dec;29(11-12):1342-9. doi: 10.1016/j.nut.2013.05.004. PMID: 24103511.
- 241. Xie FL, Wang YQ, Peng LF, et al. Beneficial Effect of Educational and Nutritional Intervention on the Nutritional Status and Compliance of Gastric Cancer Patients Undergoing Chemotherapy: A Randomized Trial. Nutr Cancer. 2017 Jul;69(5):762-71. doi: 10.1080/01635581.2017.1321131. PMID: 28524705.
- 242. Najafi S, Haghighat S, Raji Lahiji M, et al. Randomized Study of the Effect of Dietary Counseling During Adjuvant Chemotherapy on Chemotherapy Induced Nausea and Vomiting, and Quality of Life in Patients With Breast Cancer. Nutr Cancer. 2019;71(4):575-84. doi: 10.1080/01635581.2018.1527375. PMID: 30449171.

- 243. Ansari M, Porouhan P, Mohammadianpanah M, et al. Efficacy of Ginger in Control of Chemotherapy Induced Nausea and Vomiting in Breast Cancer Patients Receiving Doxorubicin-Based Chemotherapy. Asian Pac J Cancer Prev. 2016;17(8):3877-80. PMID: 27644633.
- 244. Vafa S, Zarrati M, Malakootinejad M, et al. Calorie restriction and synbiotics effect on quality of life and edema reduction in breast cancer-related lymphedema, a clinical trial. Breast. 2020 Dec;54:37-45. doi: 10.1016/j.breast.2020.08.008. PMID: 32898787.
- 245. Pettersson A, Nygren P, Persson C, et al. Effects of a dietary intervention on gastrointestinal symptoms after prostate cancer radiotherapy: long-term results from a randomized controlled trial. Radiother Oncol. 2014 Nov;113(2):240-7. doi: 10.1016/j.radonc.2014.11.025. PMID: 25467005.
- 246. Villarini A, Pasanisi P, Raimondi M, et al. Preventing weight gain during adjuvant chemotherapy for breast cancer: a dietary intervention study. Breast Cancer Res Treat. 2012 Sep;135(2):581-9. doi: 10.1007/s10549-012-2184-4. PMID: 22869285.
- 247. Harvie M, Pegington M, Howell SJ, et al. Randomised controlled trial of intermittent vs continuous energy restriction during chemotherapy for early breast cancer. Br J Cancer. 2022 May;126(8):1157-67. doi: 10.1038/s41416-021-01650-0. PMID: 34912072.
- 248. Cho AR, Hong KW, Kwon YJ, et al. Effects of Single Nucleotide Polymorphisms and Mediterranean Diet in Overweight or Obese Postmenopausal Women With Breast Cancer Receiving Adjuvant Hormone Therapy: A Pilot Randomized Controlled Trial. Frontiers in Nutrition. 2022;9:882717. doi:

https://dx.doi.org/10.3389/fnut.2022.882717 . PMID: 35845810.

249. Gianotti L, Braga M, Frei A, et al. Health care resources consumed to treat postoperative infections: cost saving by perioperative immunonutrition. Shock. 2000 Sep;14(3):325-30. doi: 10.1097/00024382-200014030-00015. PMID: 11028551.

- 250. Wang J, Yang M, Wang Q, et al. Comparison of Early Oral Feeding With Traditional Oral Feeding After Total Gastrectomy for Gastric Cancer: A Propensity Score Matching Analysis. Front Oncol. 2019;9:1194. doi: 10.3389/fonc.2019.01194. PMID: 31788451.
- 251. Braga M, Gianotti L. Preoperative immunonutrition: cost-benefit analysis. JPEN J Parenter Enteral Nutr. 2005 Jan-Feb;29(1 Suppl):S57-61. doi: 10.1177/01486071050290s1s57. PMID: 15709546.
- 252. Martin B, Cereda E, Caccialanza R, et al. Costeffectiveness analysis of oral nutritional supplements with nutritional counselling in head and neck cancer patients undergoing radiotherapy. Cost Eff Resour Alloc. 2021 Jun 15;19(1):35. doi: 10.1186/s12962-021-00291-7. PMID: 34130709.
- 253. Pattamatta M, Evers SM, Smeets BJ, et al. An economic evaluation of perioperative enteral nutrition in patients undergoing colorectal surgery (SANICS II study). Journal of Medical Economics. 2019;22(3):238-44.
- 254. Webb N, Fricke J, Hancock E, et al. The clinical and cost-effectiveness of supplemental parenteral nutrition in oncology. ESMO open. 2020;5(3):e000709.
- 255. Pimiento JM, Evans DC, Tyler R, et al. Value of nutrition support therapy in patients with gastrointestinal malignancies: a narrative review and health economic analysis of impact on clinical outcomes in the United States. J Gastrointest Oncol. 2021 Apr;12(2):864-73. doi: 10.21037/jgo-20-326. PMID: 34012673.
- 256. Elia M, Normand C, Norman K, et al. A systematic review of the cost and cost effectiveness of using standard oral nutritional supplements in the hospital setting. Clin Nutr. 2016 Apr;35(2):370-80. doi: 10.1016/j.clnu.2015.05.010. PMID: 26123475.
- 257. Schuetz P, Sulo S, Walzer S, et al. Cost savings associated with nutritional support in medical inpatients: an economic model based on data from a systematic review of randomised trials. BMJ Open. 2021 Jul 9;11(7):e046402. doi: 10.1136/bmjopen-2020-046402. PMID: 34244264.

- 258. Pradelli L, Klek S, Mayer K, et al. Cost-Effectiveness of Parenteral Nutrition Containing ω-3 Fatty Acids in Hospitalized Adult Patients From 5 European Countries and the US. JPEN J Parenter Enteral Nutr. 2021 Jul;45(5):999-1008. doi: 10.1002/jpen.1972. PMID: 32713007.
- 259. Tyler R, Barrocas A, Guenter P, et al. Value of Nutrition Support Therapy: Impact on Clinical and Economic Outcomes in the United States. JPEN J Parenter Enteral Nutr. 2020 Mar;44(3):395-406. doi: 10.1002/jpen.1768. PMID: 31994761.
- 260. DeSantis CE, Miller KD, Dale W, et al. Cancer statistics for adults aged 85 years and older, 2019. CA: A Cancer Journal for Clinicians. 2019;69(6):452-67. doi: https://doi.org/10.3322/caac.21577.
- 261. DPCPSI. 2020-2030 Strategic plan for NIH nutrition research. National Institutes of Health Office of Nutrition Research; 2022. <u>https://dpcpsi.nih.gov/onr/strategic-plan</u>. Accessed on April 25 2022.
- 262. Schulz KF, Altman DG, Moher D. CONSORT 2010 statement: updated guidelines for reporting parallel group randomized trials. Ann Intern Med. 2010 Jun 1;152(11):726-32. doi: 10.7326/0003-4819-152-11-201006010-00232. PMID: 20335313.

- Winkfield KM, Phillips JK, Joffe S, et al. Addressing financial barriers to patient participation in clinical trials: ASCO policy statement. J Clin Oncol. 2018;36(33):3331-9.
- 264. Trujillo EB, Shapiro AC, Stephens N, et al. Monitoring rates of malnutrition risk in outpatient cancer centers utilizing the malnutrition screening tool embedded into the electronic health record. Journal of the Academy of Nutrition and Dietetics. 2021;121(5):925-30.
- 265. Bozzetti F, Mariani L, Lo Vullo S, et al. The nutritional risk in oncology: a study of 1,453 cancer outpatients. Supportive care in cancer. 2012;20(8):1919-28.
- 266. Everett W, Badaracco C, McCauley S. From hospital to home: why nutrition counts. Health Affairs Blog. 2020 Jan 24. doi: 10.1377/hblog20200117.329745.
- 267. Tappenden KA, Quatrara B, Parkhurst ML, et al. Critical role of nutrition in improving quality of care: an interdisciplinary call to action to address adult hospital malnutrition. Journal of the Academy of Nutrition and Dietetics. 2013;113(9):1219-37.

Abbreviations and Acronyms

AE	adverse event	
AND	Academy of Nutrition and Dietetics	
AHRQ	Agency for Healthcare Research and Quality	
BCAA	branched-chain amino acid	
BMI	body mass index	
BW	body weight	
CFU	colony forming unit	
СНО	carbohydrate	
CINV	hemotherapy-induced nausea and vomiting	
CMF	cyclophosphamide, methotrexate, and fluorouracil	
CONSORT	CONsolidated Standards of Reporting Trials	
CQ	contextual question	
CTCAE	Common Terminology Criteria for Adverse Events	
DFS	disease free survival	
DHA	docosahexaenoic acid	
EAT	Eating As Treatment program	
EEN	early enteral nutrition	
EEIN	early enteral immunonutrition	
EN	enteral nutrition	
EORTC QLQ-C30	European Organization for Research and Treatment of Cancer, Quality of Life Questionnaire	
EPA	eicosapentaenoic acid	
EPC	Evidence-based Practice Center	
EPN	early parenteral nutrition	
EQ-5D	EuroQol-5D	
ER	emergency roomFA fatty acid	
FFM	fat free mass	
FM	fat mass	
GLA	gamma-linolenic acid	
HRQOL	health related quality of life	
ICU	intensive care unit	
IEN	immunoenteral nutrition	
IMEN	immunomodulating enteral nutrition	
IMPN	immunomodulating parenteral nutrition	
ITT	intention-to-treat	

IV	intravenous
KQ	Key Question
LCT	long chain triglyceride
LOS	length of hospital stay
MA	medroxyprogesterone acetate/megestrol acetate
MAC	mid-arm circumference
MAMC	mid-arm muscle circumference
MNA-SF	Mini Nutrition Assessment – Short Form
MST	malnutrition screening tool
MUST	malnutrition universal screening tool
NA	not available
NG	nasogastric
NI	nutrition intervention
NIDT	nutrition intervention during treatment
NIH	National Institutes of Health
NJT	nasojejunal tube
NRS	Nutrition Risk Screening
ONS	oral nutrition supplement
OS	overall survival
P2P	Pathways to Prevention
PEG	percutaneous endoscopic gastrostomy
PICOTS	population, intervention, comparator, outcome, timing and setting
PMID	PubMed Identification Number
PN	parenteral nutrition
PNI	pre-treatment nutrition intervention
POD	post-operative day
PRISMA	Preferred Items for Reporting in Systematic Reviews and Meta-Analyses
PRO	patient reported outcome
PROMIS	Patient-Reported Outcomes Measurement Information System
PUFA	polyunsaturated fat
QOL	quality of life
RCT	randomized controlled trial
RNA	ribonucleic acid
ROB	risk of bias
RT	radiotherapy
SF-36	short-form health survey
SIRS	systemic inflammatory response syndrome
SPN	standard parenteral nutrition

SSI	surgical site infection
TPN	total parenteral nutrition
TSF	triceps skin fold
UTD	unable to determine
WPI	whey protein isolate

Appendixes

Appendix A. Methods

- Appendix B. Studies Excluded at Full Text
- Appendix C. Evidence Tables for Chapter 5
- Appendix D. Evidence Tables for Chapter 6
- Appendix E. Evidence Tables for Chapter 7
- Appendix F. Evidence Tables for Chapter 8
- Appendix G. Evidence Tables for Chapter 9
- Appendix H. References

Appendix A. Methods

Search Strategy

Search Details and Sources

The search strategy was designed and conducted by an experienced systematic review/medical librarian with input from the investigators. We applied the following limits or filters to the database searches:

- Date. We considered a literature search starting in 2000 sufficient for the purpose of this review. Date limitation was done in PICO Portal.
- Language. Publications were excluded if they were written in a language other than English. This was due to resource constraints.
- Publication status. We searched for published studies.
- Humans or organisms. A filter was used to remove animal studies.
- Study Design. The search was limited to randomized controlled trials, controlled trials, and observational cohort studies. After review of the breadth of included studies, study designs were further limited to randomized controlled trials randomizing at least 50 participants (i.e., 25 individuals per arm) to identify the literature with the highest likelihood of having statistical power to detect an effect from a nutrition intervention. The updated search on July 22, 2022 was limited to randomized controlled trials.

We conducted a comprehensive literature search in May 2021 (updated July 2022). We searched the following databases:

- Ovid MEDLINE(R) and Epub Ahead of Print, In-Process, In-Data-Review & Other Non-Indexed Citations, Daily and Versions(R) <1946 to May 21, 2021> Date searched: May 20, 2021
- Cochrane Central Register of Controlled Trials (Wiley) Issue 4 of 12, April 2021 Date searched: May 21,2021
- Embase Classic+Embase <1947 to 2021 May 20> (Ovid) Date searched: May 20, 2021.

For the contextual question we conducted a literature search through July 2022 to more broadly understand the effectiveness of nutrition interventions from a cost-perspective, including evaluations that reported information on intervention costs as well as cost-effectivenes, cost-benefit and value analyses. We searched the following databases:

- References of relevant systematic reviews and meta analysis (line 35 of Ovid MEDLINE search strategy below)
- Web search engines/specific web sites. We searched for grey literature using Google. The first 200 Google results yielded by each search string were reviewed.
- Search of the Center for the Evaluation of Value and Risk in Health, Tufts Cost-Effectiveness Analysis Registry

Database Search Strategies

Updated Search: Ovid MEDLINE(R) ALL <1946 to July 22, 2022>

1 Nutrition Therapy/ or Diet Therapy/ or ((diet* or nutrition*) adj3 (counsel* or intervention* or support* or supplement* or therap*)).ti,ab. 1

2 Prebiotics/ or Probiotics/ or Synbiotics/ or (prebiotic* or probiotic* or symbiotic* or symbiotic*).ti,ab.

3 Enteral Nutrition/ or Nutritional Support/ or exp Parenteral Nutrition/ or ((enteral or gastrostomy or jejunostomy or oral or parenteral or tube) adj3 (feeding or nutrition*)).ti,ab.

4 Caloric Restriction/ or Diet, Reducing/ or (calori* restrict* diet* or intermittent fasting or fasting mimicking diet* or short-term fasting).ti,ab.

5 Diet, High-Protein/ or Diet, Ketogenic/ or Diet, Carbohydrate-Restricted/ or Diet, High-Protein Low-Carbohydrate/ or Diet, Mediterranean/ or (high-protein diet* or high-calorie diet* or ketogenic diet* or mediterranean diet*).ti,ab.

6 or/1-5

7 Brachytherapy/ or Chemoprevention/ or Chemoradiotherapy/ or Chemoradiotherapy, Adjuvant/ or Chemotherapy, Adjuvant/ or Consolidation Chemotherapy/ or exp Neoplasms/ or Radiotherapy/ or (cancer* or carcinoma* or chemoprevention or chemotherap* or chemoradiotherap* or leuk?emia* or melanoma* or myeloma* or neoplasm* or radiotherap* or radiation therap*).ti,ab.

8 6 and 7

9 control groups/ or double-blind method/ or placebo effect/ or random allocation/ or exp randomized controlled trial/ or single-blind method/

10 (control* adj3 (study or studies or trial* or group*)).ti,ab,hw,kf.

11 (random* or sham or placebo*).ti,ab,hw,kf.

12 ((singl* or doubl*) adj (blind* or dumm* or mask*)).ti,ab,hw,kf.

13 ((tripl* or trebl*) adj (blind* or dumm* or mask*)).ti,ab,hw,kf.

- 14 or/9-13
- 15 8 and 14
- 16 limit 15 to english language

17 limit 16 to dt=20210501-20220731

Database: Embase <1974 to 2022 Week 29>

1 ((diet* or nutrition*) adj3 (counsel* or intervention* or support* or supplement* or prebiotic* or probiotic* or symbiotic* or therap*)).ti,ab. or diet therapy/ or nutritional counseling/ or diet supplementation/

2 ((enteral or gastrostomy or jejunostomy or oral or parenteral or tube) adj3 (feeding or nutrition*)).ti,ab. or nutritional support/ or parenteral nutrition/ or enteric feeding/ 3 (malnutrition universal screening or malnutrition screening or nutrition* assessment or nutrition* risk screening).ti,ab.

4 (calori* restrict* diet* or intermittent fasting or fasting mimicking diet* or short-term fasting).ti,ab. or caloric restriction/ or intermittent fasting/ or exp low calorie diet/ 5 (high-protein diet* or high-calorie diet* or ketogenic diet or mediterranean diet).ti,ab. or exp

ketogenic diet/ or exp protein diet/ or Mediterranean diet/ or low carbohydrate diet/ $6 \ \text{or}/1\text{-}5$

7 (cancer* or carcinoma* or chemoprevention or chemotherap* or chemoradiotherap* or leuk?emia* or melanoma* or myeloma* or neoplasm* or radiotherap* or radiation therap*).ti,ab. or exp neoplasms subdivided by anatomical site/ or cancer chemotherapy/ or chemoprophylaxis/ or consolidation chemotherapy/ or cancer radiotherapy/ or brachytherapy/ or chemoradiotherapy/ or adjuvant chemoradiotherapy/

8 6 and 7

9 (rats or rat or rabbit or porcine or cow or cows or chicken* or horse or horses or mice or mouse or bovine or sheep or ovine or murinae or cats or cat or dog or dogs or rodent or swine).tw. 10 8 not 9

11 exp randomized controlled trial/ or controlled clinical trial/ or control group/ or double blind procedure/ or randomization/ or single blind procedure/ or placebo effect/

12 (control* adj3 (study or studies or trial* or group*)).ti,ab,kw.

13 (random* or sham or placebo*).ti,ab,kw.

14 ((quasiexperimental or quasi experimental) adj3 (study or studies or trial*)).ti,ab,kw.

15 or/11-14

16 10 and 15

17 limit 16 to dd=20220701-20220731

18 limit 17 to conference abstracts

19 17 not 18

Cochrane Central Register of Controlled Trials Date Run: 31/07/2022

Issue 7 of 12, July 2022

25 total including 8 clinical trials from <u>ct.gov</u>

ID Search Hits

#1 [mh ^"diet therapy"] or [mh ^"nutrition therapy"]

#2 (diet NEAR/3 (counsel or intervention or support or supplement or therapy)):ti,ab (Word variations have been searched)

#3 (nutrition NEAR/3 (counsel or intervention or support or supplement or therapy)):ti,ab (Word variations have been searched)

#4 [mh prebiotics] or [mh probiotics] or [mh symbiotics] or [mh symbiotics]

#5 (prebiotic? or probiotic? or symbiotic?):ti,ab

#6 [mh "nutritional support"] or [mh "enteral nutrition"] or [mh "parenteral nutrition"] #7 enteral:ti,ab or gastrostomy:ti,ab or jejunostomy:ti,ab or parenteral:ti,ab or (tube near/2 feeding):ti,ab

#8 [mh "caloric restriction"] or [mh "diet, reducing"]

#9 (calorie near/3 restrict):ti,ab or (intermittent near/2 fasting):ti,ab or (fasting near/2 mimicking):ti,ab or ("short term" near/2 fasting):ti,ab (Word variations have been searched) #10 [mh "Diet, High-Protein"] or [mh "diet, ketogenic"] or [mh "diet, carbohydrate-restricted"] or [mh "diet, high-protein low-carbohydrate"] or [mh "diet, mediterranean"]

#11 ("high protein" near/3 diet):ti,ab or ("high calorie" near/3 diet):ti,ab or (ketogenic near/3 diet):ti,ab or (Mediterranean near/3 diet):ti,ab (Word variations have been searched) #12 [OR #1-#11]

#13 [mh Neoplasms] or [mh chemoprevention] or [mh chemoradiotherapy] or [mh

chemoradiotherapy] or [mh "chemoradiotherapy, adjuvant"] or [mh "chemotherapy, adjuvant"] or [mh "consolidation chemotherapy"] or [mh ^radiotherapy] or [mh brachytherapy] #14 (cancer or carcinoma or chemoprevention or chemotherapy or chemoradiotherapy or leukemia or melanoma or myeloma or neoplasm or radiotherapy or "radiation therapy" or "radiation therapies"):ti,ab (Word variations have been searched) #15 #13 or #14 235400 #16 #12 and #15 in Trials 4965

2022-05-01 to 2022-07-31: 25 total including 8 clinical trials from clinicatrials.gov

Original Searches May 2021

Ovid MEDLINE(R) and Epub Ahead of Print, In-Process, In-Data-Review & Other Non-Indexed Citations, Daily and Versions(R) <1946 to May 21, 2021>

1 Nutrition Therapy/ or diet therapy/ or ((diet* or nutrition*) adj3 (counsel* or intervention* or support* or supplement* or therap*)).ti,ab.

2 Prebiotics/ or Probiotics/ or Synbiotics/ or (prebiotic* or probiotic* or symbiotic* or symbiotic*).ti,ab.

3 Enteral Nutrition/ or Nutritional Support/ or exp Parenteral Nutrition/ or ((enteral or gastrostomy or jejunostomy or oral or parenteral or tube) adj3 (feeding or nutrition*)).ti,ab.

4 Caloric Restriction/ or Diet, Reducing/ or (calori* restrict* diet* or intermittent fasting or fasting mimicking diet* or short-term fasting).ti,ab.

5 Diet, High-Protein/ or Diet, Ketogenic/ or Diet, Carbohydrate-Restricted/ or Diet, High-Protein Low-Carbohydrate/ or Diet, Mediterranean or (high-protein diet* or high-calorie diet* or ketogenic diet* or mediterranean diet*).ti,ab. or /

6 or/1-5

7 Brachytherapy/ or Chemoprevention/ or Chemoradiotherapy/ or Chemoradiotherapy, Adjuvant/ or Chemotherapy, Adjuvant/ or Consolidation Chemotherapy/ or exp Neoplasms/ or Radiotherapy/ or (cancer* or carcinoma* or chemoprevention or chemotherap* or chemoradiotherap* or leuk?emia* or melanoma* or myeloma* or neoplasm* or radiotherap* or radiation therap*).ti,ab.

8 Controlled Clinical Trial/ or Control Groups/ or Double-Blind Method/ or Placebo Effect/ or exp Randomized Controlled Trial/ or Random Allocation/ or Single-Blind Method/

- 9 (control* adj3 (study or studies or trial* or group*)).ti,ab,hw,kf.
- 10 (random* or sham or placebo*).ti,ab,hw,kf.
- 11 ((quasiexperimental or quasi experimental) adj3 (study or studies or trial*)).ti,ab,hw,kf.
- 12 (nonrandom* or non random* or quasi-random* or quasirandom*).ti,ab,hw,kf.
- 13 or/8-12
- 14 Epidemiologic Studies/
- 15 exp Case-Control Studies/
- 16 exp Cohort Studies/
- 17 Cross-Sectional Studies/
- 18 (epidemiologic adj (study or studies)).ti,ab.
- 19 case control*.ab,ti.
- 20 (cohort adj3 (study or studies)).ti,ab.
- 21 cross sectional.ab,ti.

- cohort analy*.ab,ti.
- 23 (follow up adj3 (study or studies)).ti,ab.
- 24 longitudinal.ti,ab.
- 25 retrospective*.ti,ab.
- 26 prospective*.ti,ab.
- 27 (observ* adj3 (study or studies)).ti,ab.
- 28 or/14-27
- 29 13 or 28
- 30 6 and 7 and 29

31 (rats or rat or rabbits or rabbit or porcine or cow or cows or chicken* or horse or horses or mice or mouse or bovine or sheep or ovine or murinae or cats or cat or dog or dogs or rodent or swine or pigs or pig).tw.

32 30 not 31

33 limit 32 to (comment or editorial or letter or news or newspaper article or personal narrative or preprint)

- 34 32 not 33
- 35 limit 34 to (meta analysis or "systematic review")
- 36 34 not 35
- 37 limit 36 to english language

Embase Classic+Embase <1947 to 2021 May 20> (Ovid)

1 diet therapy/ or nutritional counseling/ or ((diet* or nutrition*) adj3 (counsel* or intervention* or support* or supplement* or therap*)).ti,ab.

2 enteric feeding/ or nutritional support/ or parenteral nutrition/ or ((enteric or enteral or gastrostomy or jejunostomy or oral or parenteral or tube) adj3 (feeding or nutrition*)).ti,ab. 3 diet supplementation/ or prebiotic agent/ or probiotic agent/ or synbiotic agent/ or (prebiotic* or probiotic* or synbiotic*).ti,ab.

4 exp diet restriction/ or (calori* restrict* diet* or intermittent fasting or fasting mimicking diet* or short-term fasting).ti,ab.

5 low carbohydrate diet/ or exp ketogenic diet/ or exp protein diet/ or Mediterranean diet/ or (high-protein diet* or high-calorie diet* or ketogenic diet or mediterranean diet).ti,ab. 6 1 or 2 or 4 or 5

7 adjuvant chemoradiotherapy/ or brachytherapy/ or cancer chemotherapy/ or cancer radiotherapy/ or chemoprophylaxis/ or chemoradiotherapy/ or consolidation chemotherapy/ or exp malignant neoplasms/ or (cancer* or carcinoma* or chemoprevention or chemotherap* or chemoradiotherap* or leuk?emia* or melanoma* or myeloma* or neoplasm* or radiotherap* or radiotherap* or tumo?r*).ti,ab.

8 6 and 7

9 (rats or rat or rabbits or rabbit or porcine or cow or cows or chicken* or horse or horses or mice or mouse or bovine or sheep or ovine or murinae or cats or cat or dog or dogs or rodent or swine or pig or pigs).tw.

10 8 not 9

11 controlled clinical trial/ or control group/ or double blind procedure/ or placebo effect/ or exp randomized controlled trial/ or randomization/ or single blind procedure/

12 (control* adj3 (group* or study or studies or trial*)).ti,ab,kw.

13 (placebo or random* or sham).ti,ab,kw.

14 ((quasiexperimental or quasi experimental) adj3 (study or studies or trial*)).ti,ab,kw.

15 or/11-14

16 10 and 15

17 epidemiology/ or epidemiolog* study.ti,ab.

18 cross-sectional study/ or cross sectional.ti,ab,kw.

19 cohort analysis/

20 case control study/

21 observational study/

22 prospective study/

23 longitudinal study/

24 retrospective study/

25 case control*.ti,ab,kw.

26 (cohort adj3 (study or studies)).ti,ab,kw.

27 cross sectional.ti,ab,kw.

28 cohort analy*.ti,ab,kw.

29 (longitudinal or prospective* or retrospective*).ti,ab,kw.

30 (observ* adj3 (study or studies)).ti,ab,kw.

31 or/17-30

32 15 or 31

33 10 and 33

34 limit 34 to english language

35 limit 34 to (conference abstract or conference paper or "conference review" or editorial or letter or note or "review")

36 34 not 35

37 limit 36 to (meta analysis or "systematic review")

38 36 not 37

Cochrane Central Register of Controlled Trials (Wiley)

Issue 5 of 12, May 2021

ID Search Hits

#1 [mh ^"diet therapy"] or [mh ^"nutrition therapy"]

#2 (diet* NEAR/3 (counsel* or intervention* or support* or supplement* or therap*)):ti,ab

#3 (nutrition* NEAR/3 (counsel* or intervention* or support* or supplement* or therap*)):ti,ab

#4 [mh prebiotics] or [mh probiotics] or [mh symbiotics] or [mh symbiotics]

#5 (prebiotic* or probiotic* or symbiotic* or symbiotic*):ti,ab

#6 [mh "nutritional support"] or [mh "enteral nutrition"] or [mh "parenteral nutrition"]

#7 (enteral or gastrostomy or jejunostomy or parenteral or tube feeding):ti,ab

#8 [mh "caloric restriction"] or [mh "diet, reducing"]

#9 (calori* NEAR restrict* or intermittent NEAR fasting or fasting NEAR mimicking or short-term NEAR fasting):ti,ab

#10 [mh "Diet, High-Protein"] or [mh "diet, ketogenic"] or [mh "diet, carbohydrate-restricted"] or [mh "diet, high-protein low-carbohydrate"] or [mh "diet, mediterranean"]

#11 (high-protein diet* or high-calorie diet* or ketogenic diet or mediterranean diet):ti,ab #12 [OR #1-#11] #13 [mh brachytherapy] or [mh chemoprevention] or [mh chemoradiotherapy] or [mh chemoradiotherapy] or [mh "chemoradiotherapy, adjuvant"] or [mh "chemotherapy] or [mh "chemotherapy"] or [mh Neoplasms] or [mh ^radiotherapy] #14 (cancer* or carcinoma* or chemoprevention or chemotherap* or chemoradiotherap* or leuk?emia* or melanoma* or myeloma* or neoplasm* or radiotherap* or radiation therap*):ti,ab #15 #13 or #14

#16 #12 and #15 3647 (PubMed, Embase, CINAHL)

Tufts CEA Registry Search Terms

"diet therapy" or "nutrition therapy" or "counsel*" or "nutrition supplement*" or "enteral nutrition" or "parenteral nutrition" or "oral nutrition supplement" or "caloric restriction" or "diet" AND "cancer" or "carcinoma"

Risk of Bias Assessment Guide

INSTRUCTIONS: Review the methods of each trial and assess each risk of bias component as described in these instructions. You may need to have separate assessments for different outcomes (i.e., different measures; different time points may have different attrition rates).

Description/Guiding Questions	Notes
Systematic differences between baseline characteristics of the groups that arise from self-selection of treatments, physician-directed selection of treatments, or association of treatment assignments with demographic, clinical, or social characteristics.	"Good" Randomization: Detailed methodology would include providing method of randomization such as use of a random numbers table, or computer random number generator. Limited methodology would be the study saying simply saying they randomized in the methods or provided limited detail such as
 Did method of randomization create biased allocation to interventions (inadequate randomization)? 	 randomizing by a 2:1 ratio. "Poor"/No Randomization: Randomized based on week of the month of birthday or a non-randomized clinical trial, observational study.

Figure A.1. Selection bias assessment guidance



Table A.2. Detection bias

Description/Guiding Questions	Notes
Systematic differences in outcomes assessment among groups being compared, including systematic misclassification of the exposure or intervention, covariates, or outcomes because of variable definitions and timings, diagnostic thresholds, recall from memory	 Independent assessor is someone not directly involved in delivering the intervention
inadequate assessor blinding, and faulty measurement techniques. Erroneous statistical analysis might also affect the validity of effect estimates.	
 Were the outcome assessors blinded to the intervention ("outcome assessor blinded")? Was the timing of the outcome assessment similar in 	
all groups ("comparable timing outcomes assessment")?Was the scale used to measure outcomes validated, reliable?	
 Were outcomes measured in clinically meaningful ways? 	

Figure A.2. Detection bias assessment guidance



Table A.3. Performance bias

Description/Guiding Questions	Notes
Systematic differences in the care provided to participants and protocol deviation. Examples include contamination of the control group with the exposure or intervention, problems with fidelity to the intervention, unbalanced provision of additional interventions or co- interventions, difference in co-interventions, and inadequate blinding of providers and participants.	 Intention-to-Treat (ITT): Includes every subject according to randomized treatment assignment. Ignores noncompliance, protocol deviations, withdrawal, and anything that happens after randomization. Concurrent Intervention: Study participants are receiving another intervention (i.e., treatment) that is not part of the intervention being tested. Example: Participants are randomized to a physical activity intervention (or no intervention), but are also dieting. If the denominator at baseline and in the results are reported and both numbers match, we can assume ITT unless there is an explicit statement in the methods for another approach.
Figure A.3. Performance bias assessment guidance



Abbreviations: ITT=intention to treat; NA= not applicable; RCT=randomized controlled trial.

Table A.4. Fidelity to intervention

Description/Guiding Questions	Notes
We anticipate that care delivery studies will generally fall in the range of NIH Stage 3 to 4, with the possibility that one or a few may be carried out as quality improvement and thus Stage 5. Since the Stage Model is explicitly designed to balance, or trade off, internal and external validity, we will approach risk of bias assessment as a threshold requirement rather than a continuum.	 Information may appear in methods, results, or discussion sections.
 Look for reporting on intervention compliance, any data reported on consistency of intervention use, or any mechanisms used to ensure compliance (e.g., reminders, guides, manuals). 	

Table A.5. Fidelity	to intervention	assessment	guidance
---------------------	-----------------	------------	----------

Domain	Options	Rating
Fidelity to intervention	Yes (at least 70%)	Low
	Yes-adaptation planned/ replicable	Medium
	No-adaptation not planned	High
	Unclear/Not Reported	Medium

Table A.6. REPORTING Bias

Description/Guiding Questions	Notes
 Systematic differences between reported and unreported findings (e.g., differential reporting of outcomes or harms, incomplete reporting of study findings, potential for bias in reporting through source of funding). Was a select group of outcomes reported? 	 Compare results to methods section and/ or protocol. Check if some results are reported in a different publication.

Table A.7. Reporting bias assessment guidance

Domain	Options	Rating
All outcomes reported	Yes	Low
	No	Medium
	Not Reported	Medium

Table A.8. Attrition bias

Description/Guiding Questions	Notes
Systematic differences in the loss of participants from	Attrition assessment is dependent on overall study
the study and how they were accounted for in the results	duration (see flowchart)
(e.g., incomplete follow-up, differential attrition). Those	Report attrition rate in spreadsheet.
who drop out of the study or who are lost to follow-up	If a study reports outcomes at multiple intervals
may be systematically different from those who remain	(e.g., 6 months, 12 months, 18 months) assess
in the study. Attrition bias can potentially change the	attrition at the first relevant time point and the last
collective (group) characteristics of the relevant groups	time-point separately, you do not need to do every
and their observed outcomes in ways that affect study	time point.
results by confounding and spurious associations.	Analysis should be done with appropriate method
	(i.e. sensitivity analysis with various scenarios);
 Reasons for incomplete/missing data adequately 	last value forward would only be appropriate for
explained?	interventions that are supposed to improve the
 Do the author's attempt to address attrition in the 	outcomes (i.e. memory training that intends to
analysis?	improve memory).
	Look for withdrawals in the flow diagram that are
	not accounted for in results table.

Figure A.4. Attrition bias assessment guidance





Figure A.5. Overall risk of bias assessment guidance

Appendix B. Studies Excluded at Full Text

P=Population I=Intervention C=Comparison O=Outcome S=Study Design SS=Sample Size PT=Publication Type NE=Non English D=Duplicate OR=Other Reason R=Retracted

- Clinical Efficacy of FOLFOX Chemotherapy Combined with Nutritional Support in the Treatment of Gastrointestinal Malignancies. Anti-tumor pharmacy. 2018;8(4):635-8 and 52. doi: 10.3969/j.issn.2095-1264.2018.04.35. PMID: CN-01966940. PT
- Effect of postoperative diet nursing with patient involved on nutritional status of the patients with rectal cancer. Chinese journal of clinical nutrition. 2018;26(5):288-92. doi: 10.3760/cma.j.issn.1674-635X.2018.05.006. PMID: CN-01993733. PT
- The Effect of n-3 polyunsaturated fatty acid supplementation on the nutritional condition of the patients and on the achievement of the chemotherapy against advanced colorectal cancer. Therapeutic research. 2019;40(8):649-60. PMID: CN-02092344. PT
- Effect of nutrition intake method with adaptive viscosity on nutritional status of laryngeal cancer patients with dysphagia after surgery and radiotherapy. Chinese journal of clinical nutrition. 2019;27(2):113-7. doi: 10.3760/cma.j.issn.1674-635X.2019.02.009. PMID: CN-02003689. PT
- Effect of Targeted Nutritional Intervention on Intestinal Flora, Defecation Function, and Postoperative Complications in Patients Who Underwent Radical Resection of Rectal Carcinoma. Anti-tumor pharmacy. 2019;9(2):338-43. doi: 10.3969/j.issn.2095-1264.2019.02.34. PMID: CN-01993849. PT

- 6. Effects of oral nutritional supplements on body weight and life qualityin post-discharge patients with gastricor colorectal cancer: a prospective randomized controlled clinical trial. Chinese journal of clinical nutrition. 2019;27(5):271-5. doi: 10.3760/cma.j.issn.1674-635X.2019.05.002. PMID: CN-02147061. PT
- 7. Effects of whole- course MDT nutrition management on radiotherapy in elderly patients with esophageal cancer. Chinese journal of clinical oncology. 2020;47(1):29-33. doi: 10.3969/j.issn.1000-8179.2020.01.461. PMID: CN-02199723. PT
- Aarts MA, Okrainec A, Glicksman A, et al. Adoption of enhanced recovery after surgery (ERAS) strategies for colorectal surgery at academic teaching hospitals and impact on total length of hospital stay. Surgical Endoscopy. 2012;26(2):442-50. doi: 10.1007/s00464-011-1897-5. PMID: 51677040. I
- Abdelgadir MA, Mahadi SE, Nasr AO, et al. Role of jejunostomy feeding catheter as a model for nutritional support. International Journal Of Surgery. 2010;8(6):439-43. doi: 10.1016/j.ijsu.2010.05.006. PMID: 20538080. C
- Abdellatif AA, Shams MMK, Helmy AFH. A comparative Study between Tube Feeding versus Parenteral Nutrition in GIT Cancer Patients in ICU. QJM : monthly journal of the Association of Physicians. 2021;114(S). doi: 10.1093/qjmed/hcab086.002. PMID: CN-02346319. PT
- 11. Abe T, Hosoi T, Kawai R, et al. Perioperative enteral supplementation with glutamine, fiber, and oligosaccharide reduces early postoperative surgical stress following esophagectomy for esophageal cancer. Esophagus. 2019;16(1):63-70. doi: 10.1007/s10388-018-0630-z. PMID: 30030739. S
- Achiam MP, Nilsson M, Åkesson O, et al. Perioperative Strategies in Esophageal Cancer Resection in 12 Nordic University Hospitals [abstract]. Diseases of the Esophagus. 2022;35(Supplement_1):10-. doi: 10.1093/dote/doac015.104. PMID: CN-02421964. PT

- 13. Achilli P, Mazzola M, Bertoglio CL, et al. Preoperative immunonutrition in frail patients with colorectal cancer: an intervention to improve postoperative outcomes. International Journal of Colorectal Disease. 2020;35(1):19-27. doi: 10.1007/s00384-019-03438-4. PMID: 31754818. S
- 14. Adiamah A, Rollins KE, Kapeleris A, et al. The impact of arginine-containing postoperative immune modulating nutrition on survival in patients undergoing surgical resection for upper gastrointestinal cancer: the long-term follow-up of a randomized controlled trial. Clinical Nutrition ESPEN. 2021;46:S570-. doi: 10.1016/j.clnesp.2021.09.085. PMID: CN-02347928. PT
- 15. Adiamah A, Rollins KE, Kapeleris A, et al. Postoperative arginine-enriched immune modulating nutrition: Long-term survival results from a randomised clinical trial in patients with oesophagogastric and pancreaticobiliary cancer. Clin Nutr. 2021 Nov;40(11):5482-5. doi: 10.1016/j.clnu.2021.09.040. PMID: 34656029. D
- 16. Aeberhard C, Leuenberger M, Joray M, et al. Management of Home Parenteral Nutrition: A Prospective Multicenter Observational Study. Annals of Nutrition & Metabolism. 2015;67(4):210-7. doi: 10.1159/000440683. PMID: 26418158. P
- 17. Aiko S, Kumano I, Yamanaka N, et al. Effects of an immuno-enhanced diet containing antioxidants in esophageal cancer surgery following neoadjuvant therapy. Diseases of the Esophagus. 2012;25(2):137-45. doi: 10.1111/j.1442-2050.2011.01221.x. PMID: 21762279. SS
- 18. Aiko S, Yoshizumi Y, Matsuyama T, et al. Influences of thoracic duct blockage on early enteral nutrition for patients who underwent esophageal cancer surgery. Japanese journal of thoracic and cardiovascular surgery : official publication of the Japanese Association for Thoracic Surgery = Nihon Kyobu Geka Gakkai zasshi. 2003;51(7):263-71. doi: 10.1007/bf02719376. PMID: CN-00439717. SS

- 19. Aiko S, Yoshizumi Y, Sugiura Y, et al. Beneficial effects of immediate enteral nutrition after esophageal cancer surgery. Surgery Today. 2001;31(1):971-8. doi: 10.1007/s005950170005. PMID: 33141281. SS
- 20. Alberts DS, Martinez ME, Roe DJ, et al. Lack of effect of a high-fiber cereal supplement on the recurrence of colorectal adenomas. New England Journal of Medicine.
 2000;342(1):1156-62. doi: 10.1056/nejm200004203421602. PMID: 30211046. P
- 21. Alimena S, Falzone M, Feltmate CM, et al. Perioperative glycemic measures among non-fasting gynecologic oncology patients receiving carbohydrate loading in an enhanced recovery after surgery (ERAS) protocol. International Journal of Gynecological Cancer. 2020;30(4):533-40. doi: 10.1136/ijgc-2019-000991. PMID: 32107317. I
- 22. Alivizatos V, Athanasopoulos P, Makris N, et al. Early postoperative glutamine-supplemented parenteral nutrition versus enteral immunonutrition in cancer patients undergoing major gastrointestinal surgery. Journal of B.U.On. 2005;10(1):119-22. PMID: 17335142. SS
- 23. Aliyazicioglu T, Canturk NZ, Simsek T, et al. Effects of Standard and/or Glutamine Dipeptide and/or Omega-3 Fatty Ascid-Supplemented Parenteral Nutrition on Neutrophil Functions, Interleukin-8 Level and Length of Stay--a Double Blind,Controlled, Randomised Study. East African Medical Journal. 2013;90(2):59-66. PMID: 26866103. SS
- Allen-Winters S, Wakefield D, Gaudio E, et al. "Eat to Live"-Piloting a Culinary Medicine Program for Head & Neck Radiotherapy Patients. Supportive Care in Cancer. 2020;28(6):2949-57. doi: 10.1007/s00520-019-05180-7. PMID: 31768735. S
- Alshawa A, Cadena AP, Stephen B, et al. Effects of glutamine for prevention of radiationinduced esophagitis: a double-blind placebocontrolled trial. Investigational New Drugs. 2021;13:13. doi: 10.1007/s10637-021-01074-w. PMID: 33580845. SS

- 26. Al-Taan O, Stephenson JA, Spencer L, et al. Changes in plasma and erythrocyte omega-6 and omega-3 fatty acids in response to intravenous supply of omega-3 fatty acids in patients with hepatic colorectal metastases. Lipids in Health & Disease. 2013;12:64. doi: 10.1186/1476-511x-12-64. PMID: 23648075. O
- 27. Al-Temimi MH, Dyurgerova AM, Kidon M, et al. Feeding Jejunostomy Tube Placed during Esophagectomy: Is There an Effect on Postoperative Outcomes? Permanente Journal. 2019;23. doi: 10.7812/tpp/18.210. PMID: 31496496. P
- Alvarez-Sarrado E, Mingol Navarro F, R JR, et al. Feeding Jejunostomy after esophagectomy cannot be routinely recommended. Analysis of nutritional benefits and catheter-related complications. American Journal of Surgery. 2019;217(1):114-20. doi: 10.1016/j.amjsurg.2018.08.027. PMID: 30309617. I
- 29. Alvaro Sanz E, Abiles J, Garrido Siles M, et al. Evaluation of a protocol to detect malnutrition and provide nutritional care for cancer patients undergoing chemotherapy. Scientific Reports. 2020;10(1):21186. doi: 10.1038/s41598-020-78246-w. PMID: 33273641. S
- 30. Amano K, Maeda I, Ishiki H, et al. Effects of enteral nutrition and parenteral nutrition on survival in patients with advanced cancer cachexia: Analysis of a multicenter prospective cohort study. Clinical Nutrition. 2021;40(3):1168-75. doi: 10.1016/j.clnu.2020.07.027. PMID: 32771283. I
- 31. Amano K, Morita T, Baba M, et al. Effect of nutritional support on terminally ill patients with cancer in a palliative care unit. American Journal of Hospice & Palliative Medicine. 2013;30(7):730-3. doi: 10.1177/1049909112469273. PMID: 23242171. P
- Ambrosone CB, Zirpoli GR, Hutson AD, et al. Dietary Supplement Use During Chemotherapy and Survival Outcomes of Patients With Breast Cancer Enrolled in a Cooperative Group Clinical Trial (SWOG S0221). Journal of Clinical Oncology. 2020;38(8):804-14. doi: 10.1200/jco.19.01203. PMID: 31855498. I

- AmenaOmer S, Shyam H, Siddiqui S. Impacts of high protein supplementation on oncology patients. JPEN. Journal of parenteral and enteral nutrition. 2021;45(S):S123-S5. doi: 10.1002/jpen.2095. PMID: CN-02261924. PT
- 34. Anandavadivelan P, Wikman A, Malberg K, et al. Role of dietitian support in improving weight loss and nutrition impact symptoms after oesophageal cancer surgery. European Journal of Clinical Nutrition. 2021;14:14. doi: 10.1038/s41430-020-00830-0. PMID: 33446903. S
- 35. Andersen MR, Sweet E, Hager S, et al. Effects of Vitamin D Use on Health-Related Quality of Life of Breast Cancer Patients in Early Survivorship. Integrative Cancer Therapies. 2019;18:1534735418822056. doi: 10.1177/1534735418822056. PMID: 30616390. P
- 36. Andersen S, Staudacher H, Weber N, et al. Pilot study investigating the effect of enteral and parenteral nutrition on the gastrointestinal microbiome post-allogeneic transplantation. British Journal of Haematology. 2020;188(4):570-81. doi: 10.1111/bjh.16218. PMID: 31612475. O
- 37. Anelli L, Di Nardo A, Bonucci M. Integrative Treatment of Lung Cancer Patients: Observational Study of 57 Cases. Asian Journal of Oncology. 2021;7(2):64-75. doi: 10.1055/s-0040-1722380. PMID: 634814190. C
- 38. Ang SY, Lim ML, Ong HS, et al. A Descriptive Study of enteral tube feeding among adults in an acute care tertiary hospital-patient selection, characteristics and complications. Clinical Nutrition ESPEN. 2020;37:58-64. doi: 10.1016/j.clnesp.2020.03.021. PMID: 32359756. P
- Antonio M, Arnan Sangerman M, Domingo-Domenech E, et al. Impact of a Tailored Nutritional and Physical Exercise Programme on Efficacy and Functional Outcomes in Older Patients with Hematological Malignancies Classified By Frailty Profile. Blood. 2019;134:3424-. doi: 10.1182/blood-2019-129908. PMID: CN-02295839. x

- 40. Antoun S, Boige V, Ducreux M, et al. Protective effect of an enteral formula containing TGF-beta2 in the prevention of chemotherapy-induced diarrhoea: A pilot study. e-SPEN. 2009;4(6):e348-e50. doi: 10.1016/j.eclnm.2009.10.005. PMID: 50696376. S
- Anwander T, Berge S, Appel T, et al. Percutaneous endoscopic gastrostomy for long-term feeding of patients with oropharyngeal tumors. Nutrition & Cancer. 2004;50(1):40-5. PMID: 15572296. I
- 42. Aoyama T, Yoshikawa T, Ida S, et al. Effects of perioperative eicosapentaenoic acid-enriched oral nutritional supplement on the long-term oncological outcomes after total gastrectomy for gastric cancer. Oncology letters. 2022;23(5). doi: 10.3892/ol.2022.13272. PMID: CN-02385995. D
- 43. Aoyama T, Yoshikawa T, Ida S, et al. Effects of perioperative eicosapentaenoic acid-enriched oral nutritional supplement on the long-term oncological outcomes after total gastrectomy for gastric cancer. Oncol. 2022 May;23(5):151. doi: 10.3892/ol.2022.13272. PMID: 35836480. D
- 44. Aredes MA, da Camara AO, de Paula NS, et al. Efficacy of omega-3 supplementation on nutritional status, skeletal muscle, and chemoradiotherapy toxicity in cervical cancer patients: A randomized, triple-blind, clinical trial conducted in a middle-income country. Nutrition. 2019;67(-):110528. doi: 10.1016/j.nut.2019.06.009. PMID: 31445316. SS
- 45. Arif A, Bangash AN, Gul A, et al. Frequency of complications following trans hiatal esophagectomy for esophageal carcinoma with or without feeding jejunostomy. Journal of Postgraduate Medical Institute. 2018;32(1):70-5. PMID: 621404948. I
- 46. Arribas L, Hurtos L, Gonzalez-Tampan AR, et al. Relationship between body composition changes and EPA supplementation in patients diagnosed with locally advanced squamous cell carcinoma of the head and neck (la-scchn). Clinical Nutrition ESPEN. 2021;46:S723-S4. doi: 10.1016/j.clnesp.2021.09.509. PMID: CN-02347963. PT

- 47. Artene DV, Bordea CI, Blidaru A. Behavioral medicine solutions for overweight and obese breast cancer patients with sleep disturbances. Proceedings of the Nutrition Society. 2016;75(O):E18. doi: 10.1017/s0029665115004516. PMID: CN-01603094. PT
- Artene DV, Bordea CI, Blidaru A. Results of 1year Diet and Exercise Interventions for ER+/PR+/-/HER2- Breast Cancer Patients Correlated with Treatment Type. Chirurgia (Bucuresti). 2017;112(4):457-68. doi: 10.21614/chirurgia.112.4.457. PMID: 28862123. I
- Ashida R, Okamura Y, Wakabayashi-Nakao K, et al. The Impact of Preoperative Enteral Nutrition Enriched with Eicosapentaenoic Acid on Postoperative Hypercytokinemia after Pancreatoduodenectomy: The Results of a Double-Blinded Randomized Controlled Trial. Digestive Surgery. 2019;36(4):348-56. doi: 10.1159/000490110. PMID: 29886499. SS
- Asprer JM, Llido LO, Sinamban R, et al. Effect on immune indices of preoperative intravenous glutamine dipeptide supplementation in malnourished abdominal surgery patients in the preoperative and postoperative periods. Nutrition. 2009;25(9):920-5. doi: 10.1016/j.nut.2009.01.014. PMID: 355023346. P
- 51. Assenat E, Thezenas S, Flori N, et al. Prophylactic percutaneous endoscopic gastrostomy in patients with advanced head and neck tumors treated by combined chemoradiotherapy. Journal of Pain & Symptom Management. 2011;42(4):548-56. doi: 10.1016/j.jpainsymman.2011.01.009. PMID: 21477980. S
- 52. Augustus E, Granderson I, Rocke KD. The Impact of a Ketogenic Dietary Intervention on the Quality of Life of Stage II and III Cancer Patients: A Randomized Controlled Trial in the Caribbean. Nutrition & Cancer. 2020:1-11. doi: 10.1080/01635581.2020.1803930. PMID: 32791011. SS

- 53. Awasthi R, Minnella EM, Ferreira V, et al. Supervised exercise training with multimodal pre-habilitation leads to earlier functional recovery following colorectal cancer resection. Acta Anaesthesiologica Scandinavica. 2019;63(4):461-7. doi: 10.1111/aas.13292. PMID: 30411316. I
- 54. Aybar PES, Parpia S, Simunovic M, et al. Perioperative optimization with nutritional supplements in patients undergoing gastrointestinal surgery for cancer: a randomized, placebo controlled feasibility clinical trial. Journal of Clinical Oncology. 2022;40(4). doi: 10.1200/JCO.2022.40.4_suppl.482. PMID: CN-02373023. PT
- 55. Azman M, Mohd Yunus MR, Sulaiman S, et al. Enteral glutamine supplementation in surgical patients with head and neck malignancy: A randomized controlled trial. Head and Neck. 2015;37(1):1799-807. doi: 10.1002/hed.23839. PMID: 604955369. SS
- 56. Baguley BJ, Skinner TL, Jenkins DG, et al. Mediterranean-style dietary pattern improves cancer-related fatigue and quality of life in men with prostate cancer treated with androgen deprivation therapy: A pilot randomised control trial. Clinical Nutrition. 2021;40(1):245-54. doi: 10.1016/j.clnu.2020.05.016. PMID: 32534948. SS
- 57. Bajpai J, Chandrasekharan A, Simha V, et al. Osteosarcoma journey over two decades in India: Small steps, big changes. Pediatric Blood & Cancer. 2019;66(9):e27877. doi: 10.1002/pbc.27877. PMID: 31207015. C
- 58. Bajramagic S, Hodzic E, Mulabdic A, et al. Usage of Probiotics and its Clinical Significance at Surgically Treated Patients Sufferig from Colorectal Carcinoma. Medicinski Arhiv. 2019;73(5):316-20. doi: 10.5455/medarh.2019.73.316-320. PMID: 31819304. I
- 59. Bakker N, Cakir H, Doodeman HJ, et al. Eight years of experience with Enhanced Recovery After Surgery in patients with colon cancer: Impact of measures to improve adherence. Surgery. 2015;157(6):1130-6. doi: 10.1016/j.surg.2015.01.016. PMID: 25791027. I

- 60. Barajas-Galindo DE, Vidal-Casariego A, Pintorde la Maza B, et al. Postoperative enteral immunonutrition in head and neck cancer patients: Impact on clinical outcomes. Endocrinologia Diabetes y Nutricion. 2020;67(1):13-9. doi: 10.1016/j.endinu.2019.05.006. PMID: 31474502. S
- 61. Barak-Nahum A, Haim LB, Ginzburg K. When life gives you lemons: The effectiveness of culinary group intervention among cancer patients. Social Science & Medicine. 2016;166:1-8. doi: 10.1016/j.socscimed.2016.07.046. PMID: 27522112. P
- 62. Barber MD, McMillan DC, Preston T, et al. Metabolic response to feeding in weightlosing pancreatic cancer patients and its modulation by a fish-oil-enriched nutritional supplement. Clinical Science. 2000;98(4):389-99. doi: 10.1042/cs0980389. PMID: 30197852. S
- 63. Barsevick AM, Dudley W, Beck S, et al. A Randomized Clinical Trial of Energy Conservation for Patients with Cancer-Related Fatigue. Cancer. 2004;100(6):1302-10. doi: 10.1002/cncr.20111. PMID: 38325960. I
- 64. Baschnagel AM, Yadav S, Marina O, et al. Toxicities and costs of placing prophylactic and reactive percutaneous gastrostomy tubes in patients with locally advanced head and neck cancers treated with chemoradiotherapy. Head & Neck. 2014;36(8):1155-61. doi: 10.1002/hed.23426. PMID: 23852670. S
- 65. Bauer SR, Van Blarigan EL, Stampfer MJ, et al. Mediterranean diet after prostate cancer diagnosis and urinary and sexual functioning: The health professionals follow-up study. Prostate. 2018;78(3):202-12. doi: 10.1002/pros.23457. PMID: 29194691. P
- 66. Bauersfeld SP, Kessler CS, Wischnewsky M, et al. The effects of short-term fasting on quality of life and tolerance to chemotherapy in patients with breast and ovarian cancer: a randomized cross-over pilot study. BMC Cancer. 2018;18(1):476. doi: 10.1186/s12885-018-4353-2. PMID: 29699509. SS

- 67. Bazargani ST, Djaladat H, Ahmadi H, et al. Gastrointestinal Complications Following Radical Cystectomy Using Enhanced Recovery Protocol. European Urology Focus. 2018;4(6):889-94. doi: 10.1016/j.euf.2017.04.003. PMID: 28753885. I
- Beaver ME, Matheny KE, Roberts DB, et al. Predictors of weight loss during radiation therapy. Otolaryngology - Head & Neck Surgery. 2001;125(6):645-8. PMID: 11743469. C
- 69. Beck AK, Baker AL, Carter G, et al. Is fidelity to a complex behaviour change intervention associated with patient outcomes? Exploring the relationship between dietitian adherence and competence and the nutritional status of intervention patients in a successful steppedwedge randomised clinical trial of eating as treatment (EAT). Implementation Science. 2021;16(1):46. doi: 10.1186/s13012-021-01118-y. PMID: 33902652. I
- 70. Beck AK, Baker AL, Carter G, et al. Assessing adherence, competence and differentiation in a stepped-wedge randomised clinical trial of a complex behaviour change intervention. Nutrients. 2020;12(8):1-18. doi: 10.3390/nu12082332. PMID: 2004854231. O
- 71. Beckerson J, Szydlo RM, Hickson M, et al. Impact of route and adequacy of nutritional intake on outcomes of allogeneic haematopoietic cell transplantation for haematologic malignancies. Clinical Nutrition. 2019;38(2):738-44. doi: 10.1016/j.clnu.2018.03.008. PMID: 29650256. S
- Beltran Chaidez YL, Reyes Barretero DY, Flores Merino MV, et al. Effect of parenteral glutamine in patients with gastrointestinal cancer undergoing surgery. Nutricion Hospitalaria. 2019;36(1):5-12. doi: 10.20960/nh.1816. PMID: 30829529. S
- 73. Ben-Arye E, Keshet Y, Shahbar IM, et al. The kitchen as therapy: qualitative assessment of an integrative cuisine workshop for patients undergoing chemotherapy. Supportive Care in Cancer. 2016;24(4):1487-95. doi: 10.1007/s00520-015-2934-z. PMID: 26361759. S

- 74. Ben-David K, Kim T, Caban AM, et al. Pretherapy laparoscopic feeding jejunostomy is safe and effective in patients undergoing minimally invasive esophagectomy for cancer. Journal of Gastrointestinal Surgery. 2013;17(8):1352-8. doi: 10.1007/s11605-013-2231-4. PMID: 23709367. C
- 75. Benes P, Pytlík R, Chocenská E, et al. Parenteral glutamine does not improve the nutritional status in patients during high-dose chemotherapy and autologous peripheral stem cell transplantation. Vnitrni lekarstvi. 2002;48(1):1039-48. PMID: CN-00413177. NE
- 76. Benes P, Pytlik R, Klepetar J, et al. Impaired intestinal resorption caused by cytostatic treatment - Effect of parenteral glutamine and the preparatory regime. Ceska a slovenska gastroenterologie a hepatologie. 2002;56(5):190-5. PMID: CN-00443975. NE
- 77. Berea-Baltierra R, Rivas-Ruiz R, Vela-Martinez E, et al. Risk factors for subclavian vein thrombosis in cancer patients with total parenteral nutrition. Journal of Clinical Medicine Research. 2014;6(5):345-53. doi: 10.14740/jocmr1862w. PMID: 25110538. I
- Bertrand J, Siegler N, Murez T, et al. Impact of preoperative immunonutrition on morbidity following cystectomy for bladder cancer: a case-control pilot study. World Journal of Urology. 2014;32(1):233-7. doi: 10.1007/s00345-013-1229-6. PMID: 24362882. S
- Bhattacharjee A, Bahar I, Saikia A. Nutritional Assessment of Patients with Head and Neck Cancer in North-East India and Dietary Intervention. Indian Journal of Palliative Care. 2015;21(3):289-95. doi: 10.4103/0973-1075.164889. PMID: 26600696. S
- Bi Y, Yi M, Yu Z, et al. Covered metallic stent for the treatment of malignant esophageal fistula combined with stricture. BMC Gastroenterology. 2020;20(1):248. doi: 10.1186/s12876-020-01398-6. PMID: 32731861. I

- 81. Biffi R, Lotti M, Cenciarelli S, et al. Complications and long-term outcome of 80 oncology patients undergoing needle catherter jejunostomy placement for early postoperative enteral feeding. Clinical Nutrition. 2000;19(4):277-9. doi: 10.1054/clnu.2000.0108. PMID: 30686017. C
- 82. Bille SJ, Fjalstad BW, Clausen MB, et al. The Effect of Special Diets on Weight and Nutritional Intake in Hematological Cancer Patients: A Randomized Study. Nutrition and Cancer. 2018;70(6):874-8. doi: 10.1080/01635581.2018.1490446. PMID: 623609982. C
- 83. Bjorklund G. The Adjuvant Nutritional Intervention in Cancer (ANICA) Trial. Nutrition & Cancer. 2015;67(8):1355-8. doi: 10.1080/01635581.2015.1085582. PMID: 26473998. PT
- 84. Blakely AM, Ajmal S, Sargent RE, et al. Critical analysis of feeding jejunostomy following resection of upper gastrointestinal malignancies. World Journal of Gastrointestinal Surgery. 2017;9(2):53-60. doi: 10.4240/wjgs.v9.i2.53. PMID: 28289510. I
- Bosland MC, Enk E, Schmoll J, et al. Soy protein supplementation in men following radical prostatectomy: a 2-year randomized, placebo-controlled clinical trial. American Journal of Clinical Nutrition. 2021;113(4):821-31. doi: 10.1093/ajcn/nqaa390. PMID: 33564828. I
- 86. Boulton-Jones JR, Lewis J, Jobling JC, et al. Experience of post-pyloric feeding in seriously ill patients in clinical practice. Clinical Nutrition. 2004;23(1):35-41. PMID: 14757391. P
- 87. Bousquet-Dion G, Awasthi R, Loiselle SE, et al. Evaluation of supervised multimodal prehabilitation programme in cancer patients undergoing colorectal resection: a randomized control trial. Acta Oncologica. 2018;57(6):849-59. doi: 10.1080/0284186x.2017.1423180. PMID: 29327644. I

- Bower M, Jones W, Vessels B, et al. Nutritional support with endoluminal stenting during neoadjuvant therapy for esophageal malignancy. Annals of Surgical Oncology. 2009;16(1):3161-8. doi: 10.1245/s10434-009-0630-2. PMID: 19636630. S
- Bowrey DJ, Baker M, Halliday V, et al. A randomised controlled trial of six weeks of home enteral nutrition versus standard care after oesophagectomy or total gastrectomy for cancer: report on a pilot and feasibility study. Trials [Electronic Resource]. 2015;16:531. doi: 10.1186/s13063-015-1053-y. PMID: 26590903. SS
- 90. Bozec A, Benezery K, Chamorey E, et al. Nutritional status and feeding-tube placement in patients with locally advanced hypopharyngeal cancer included in an induction chemotherapy-based larynx preservation program. European Archives of Oto-Rhino-Laryngology. 2016;273(9):2681-7. doi: 10.1007/s00405-015-3785-4. PMID: 26395117. S
- 91. Bozzetti F. Survival of the starving cancer patient: a food for thought for oncologists. Eur J Surg Oncol. 2022 Jun 3;03:03. doi: 10.1016/j.ejso.2022.05.032. PMID: 35835631. I
- 92. Bozzetti F, Santarpia L, Pironi L, et al. The prognosis of incurable cachectic cancer patients on home parenteral nutrition: a multi-centre observational study with prospective follow-up of 414 patients. Annals of Oncology. 2014;25(2):487-93. doi: 10.1093/annonc/mdt549. PMID: 24406425. C
- 93. Braal CL, Hussaarts K, Seuren L, et al. Influence of green tea consumption on endoxifen steady-state concentration in breast cancer patients treated with tamoxifen. Breast Cancer Research & Treatment. 2020;184(1):107-13. doi: 10.1007/s10549-020-05829-6. PMID: 32803636. I
- 94. Brard L, Weitzen S, Strubel-Lagan SL, et al. The effect of total parenteral nutrition on the survival of terminally ill ovarian cancer patients. Gynecologic Oncology. 2006;103(1):176-80. PMID: 16564074. P

- 95. Braumann C, Guenther N, Wendling P, et al. Multimodal perioperative rehabilitation in elective conventional resection of colonic cancer: results from the German Multicenter Quality Assurance Program 'Fast-Track Colon II'. Digestive Surgery. 2009;26(2):123-9. doi: 10.1159/000206149. PMID: 19262064. I
- 96. Breitkreutz R, Tesdal K, Jentschura D, et al. Effects of a high-fat diet on body composition in cancer patients receiving chemotherapy: A randomized controlled study. Wiener Klinische Wochenschrift. 2005;117(1):685-92. doi: 10.1007/s00508-005-0455-3. PMID: 41719684. I
- 97. Brenkman HJF, Roelen SVS, Steenhagen E, et al. Postoperative complications and weight loss following jejunostomy tube feeding after total gastrectomy for advanced adenocarcinomas. Chinese Journal of Cancer Research. 2017;29(4):333-40. doi: 10.21147/j.issn.1000-9604.2017.04.06. PMID: 28947865. C
- 98. Britton B, Baker A, Clover K, et al. Heads Up: a pilot trial of a psychological intervention to improve nutrition in head and neck cancer patients undergoing radiotherapy. European Journal of Cancer Care. 2017;26(4). doi: 10.1111/ecc.12502. PMID: 27125571. C
- 99. Brown T, Banks M, Hughes BGM, et al. Tube feeding during treatment for head and neck cancer - Adherence and patient reported barriers. Oral Oncology. 2017;72:140-9. doi: 10.1016/j.oraloncology.2017.07.017. PMID: 28797450. S
- 100. Brown TE, Banks MD, Hughes BGM, et al. Comparison of Nutritional and Clinical Outcomes in Patients with Head and Neck Cancer Undergoing Chemoradiotherapy Utilizing Prophylactic versus Reactive Nutrition Support Approaches. Journal of the Academy of Nutrition & Dietetics. 2018;118(4):627-36. doi: 10.1016/j.jand.2016.10.013. PMID: 27986517. S
- 101. Bruce WR, Cirocco M, Giacca A, et al. A pilot randomised controlled trial to reduce colorectal cancer risk markers associated with B-vitamin deficiency, insulin resistance and colonic inflammation. British Journal of Cancer. 2005;93(6):639-46. doi: 10.1038/sj.bjc.6602770. PMID: 41486454. P

102. Bruera E, Hui D, Dalal S, et al. Parenteral hydration in patients with advanced cancer: a multicenter, double-blind, placebocontrolled randomized trial. Journal of Clinical Oncology. 2013;31(1):111-8. doi: 10.1200/jco.2012.44.6518. PMID: 23169523. I

103. Buijs N, van Bokhorst-de van der Schueren MA, Langius JA, et al. Perioperative arginine-supplemented nutrition in malnourished patients with head and neck cancer improves long-term survival. American Journal of Clinical Nutrition. 2010;92(5):1151-6. doi: 10.3945/ajcn.2010.29532. PMID: 20881073. SS

- 104. Bumrungpert A, Pavadhgul P, Nunthanawanich P, et al. Whey Protein Supplementation Improves Nutritional Status, Glutathione Levels, and Immune Function in Cancer Patients: A Randomized, Double-Blind Controlled Trial. Journal of Medicinal Food. 2018;21(6):612-6. doi: 10.1089/jmf.2017.4080. PMID: 29565716. SS
- 105. Buntzel J, Buntzel H, Micke O, et al. Nutritional support for head and neck cancer patients before irradiation - A pilot project for malnutrition risk group. Trace Elements and Electrolytes. 2014;31(1):1-5. doi: 10.5414/tex01310. PMID: 372178638. S
- 106. Burney RE, Bryner BS. Safety and long-term outcomes of percutaneous endoscopic gastrostomy in patients with head and neck cancer. Surgical Endoscopy. 2015;29(1):3685-9. doi: 10.1007/s00464-015-4126-9. PMID: 25740644. C
- 107. Byrd DA, Judd S, Flanders WD, et al. Associations of Novel Dietary and Lifestyle Inflammation Scores with Incident, Sporadic Colorectal Adenoma. Cancer Epidemiol Biomarkers Prev. 2020 Nov;29(11):2300-8. doi: 10.1158/1055-9965.EPI-20-0568. PMID: 32856603. ST

- 108. Caccialanza R, Cereda E, Klersy C, et al. Early intravenous administration of nutritional support (IVANS) in metastatic gastric cancer patients at nutritional risk, undergoing first-line chemotherapy: study protocol of a pragmatic, randomized, multicenter, clinical trial. Therapeutic Advances in Medical Oncology. 2020;12:1758835919890281. doi: 10.1177/1758835919890281. PMID: 32127922. PT
- 109. Caetano RS, Lima FF, Gomes EP, et al. Quality of Life of Patients After Treatment for Cancer in the Head and Neck Region: A Case-Control Study. Cureus. 2022 Jun;14(6):e25800. doi: 10.7759/cureus.25800. PMID: 35822149. ST
- 110. Cai J, Wang H, Zhou S, et al. Effect of Sijunzi Decoction and enteral nutrition on T-cell subsets and nutritional status in patients with gastric cancer after operation: a randomized controlled trial. Zhong xi yi jie he xue bao [Journal of Chinese integrative medicine]. 2008;6(1):37-40. doi: 10.3736/jcim20080108. PMID: CN-00731426. NE
- 111. Camargo P, Almeida JP, Landoni G, et al. Early vs. late parenteral nutrition in patients with cancer undergoing major gastrointestinal surgery: a randomized and clinical study. Intensive care medicine experimental. Conference: 31st european society of intensive care medicine annual congress, ESICM 2018. France. 2018;6(S). doi: 10.1186/s40635-018-0201-6. PMID: CN-01669829. PT
- 112. Cao J, Luo S, Liang L, et al. Effects of parenteral nutrition with and without GH on the GH/IGF-1 axis after hepatectomy in hepatocellular carcinoma with liver cirrhosis. Frontiers of medicine in China. 2007;1(3):287-93. doi: 10.1007/s11684-007-0055-x. PMID: 24573868. I
- 113. Cao S, Zhao G, Cui J, et al. Fast-track rehabilitation program and conventional care after esophagectomy: a retrospective controlled cohort study. Supportive Care in Cancer. 2013;21(3):707-14. doi: 10.1007/s00520-012-1570-0. PMID: 22933129. I

- 114. Cao Y, Han D, Yang S, et al. Effects of preoperative enteral immunonutrition for esophageal cancer patients treated with neoadjuvant chemoradiotherapy: protocol for a multicenter randomized controlled trial (point trial, pre-operative immunonutrition therapy). BMC Cancer. 2022 Jun 13;22(1):650. doi: 10.1186/s12885-022-09721-y. PMID: 35698100. PT
- 115. Carmody J, Olendzki B, Reed G, et al. A dietary intervention for recurrent prostate cancer after definitive primary treatment: results of a randomized pilot trial. Urology. 2008;72(6):1324-8. doi: 10.1016/j.urology.2008.01.015. PMID: 18400281. P
- 116. Carvalho TC, Cruz BC, Viana MS, et al. Effect of Nutritional Supplementation Enriched with Eicosapentaenoic Acid on Inflammatory Profile of Patients With Oral Cavity Cancer in Antineoplastic Pretreatment: A Controlled and Randomized Clinical Trial. Nutrition & Cancer. 2017;69(3):428-35. doi: 10.1080/01635581.2017.1274406. PMID: 28128983. O
- 117. Casas F, Leon C, Jovell E, et al. Adapted ice cream as a nutritional supplement in cancer patients: Impact on quality of life and nutritional status. Clinical and Translational Oncology. 2012;14(1):66-72. doi: 10.1007/s12094-012-0763-9. PMID: 364770784. S
- 118. Casas-Rodera P, Gomez-Candela C, Benitez S, et al. Immunoenhanced enteral nutrition formulas in head and neck cancer surgery: a prospective, randomized clinical trial. Nutricion Hospitalaria. 2008;23(2):105-10. PMID: 18449445. SS
- 119. Catarci M, Berlanda M, Grassi GB, et al. Pancreatic enzyme supplementation after gastrectomy for gastric cancer: a randomized controlled trial. Gastric Cancer. 2018;21(3):542-51. doi: 10.1007/s10120-017-0757-y. PMID: 28804801. I
- 120. Catho H, Guigard S, Toffart AC, et al. What are the barriers to the completion of a homebased rehabilitation programme for patients awaiting surgery for lung cancer: a prospective observational study. BMJ Open. 2021;11(2):e041907. doi: 10.1136/bmjopen-2020-041907. PMID: 33568371. I

- 121. Cerchietti LCA, Navigante AH, Castro MA. Effects of eicosapentaenoic and docosahexaenoic n-3 fatty acids from fish oil and preferential Cox-2 inhibition on systemic syndromes in patients with advanced lung cancer. Nutrition and Cancer. 2007;59(1):14-20. doi: 10.1080/01635580701365068. PMID: 47549776. I
- 122. Cetin T, Arpaci F, Dere Y, et al. Total parenteral nutrition delays platelet engraftment in patients who undergo autologous hematopoietic stem cell transplantation. Nutrition. 2002;18(7):599-603. doi: 10.1016/s0899-9007(02)00779-7. PMID: 34687098. S
- 123. Chabot JA, Tsai WY, Fine RL, et al. Pancreatic proteolytic enzyme therapy compared with gemcitabine-based chemotherapy for the treatment of pancreatic cancer. Journal of Clinical Oncology. 2010;28(1):2058-63. doi: 10.1200/jco.2009.22.8429. PMID: 19687327. I
- 124. Challine A, Rives-Lange C, Danoussou D, et al. Impact of Oral Immunonutrition on Postoperative Morbidity in Digestive Oncologic Surgery: A Nation-wide Cohort Study. Annals of Surgery. 2021;273(4):725-31. doi: 10.1097/sla.00000000003282. PMID: 30946082. S
- 125. Chambrier C, Garcia I, Bannier E, et al. Feeding the gut early after digestive surgery: Results of a nine-year experience. Clinical Nutrition. 2002;21(1):59-65. doi: 10.1054/clnu.2001.0504. PMID: 34257513. S
- 126. Champ CE, Palmer JD, Volek JS, et al. Targeting metabolism with a ketogenic diet during the treatment of glioblastoma multiforme. Journal of Neuro-Oncology. 2014;117(1):125-31. doi: 10.1007/s11060-014-1362-0. PMID: 24442482. I
- 127. Chan JM, Weinberg V, Magbanua MJ, et al. Nutritional supplements, COX-2 and IGF-1 expression in men on active surveillance for prostate cancer. Cancer Causes & Control. 2011;22(1):141-50. doi: 10.1007/s10552-010-9684-5. PMID: 21103921. P

- 128. Chang JH, Gosling T, Larsen J, et al. Prophylactic gastrostomy tubes for patients receiving radical radiotherapy for head and neck cancers: a retrospective review. Journal of Medical Imaging & Radiation Oncology. 2009;53(5):494-9. doi: 10.1111/j.1754-9485.2009.02103.x. PMID: 19788486. I
- 129. Chang PH, Yeh KY, Huang JS, et al. Chemoradiotherapy in elderly patients with advanced head and neck cancer under intensive nutritional support. Asia-Pacific Journal of Clinical Oncology. 2015;11(3):228-35. doi: 10.1111/ajco.12323. PMID: 25535674. C
- 130. Chao PC, Lin CF, Chuang HJ. Parenteral nutrition combined with enteral feeding improves the outcome of cancer patients. Asia Pacific Journal of Clinical Nutrition. 2017;26(6):1032-8. doi: 10.6133/apjcn.012017.06. PMID: 28917228. I
- 131. Chao PC, Lin FC. Improved nutritional support with immune-modulating formula in patients with head and neck and esophageal cancer undergoing radiochemotherapy: A retrospective clinical study. Asia Pacific Journal of Clinical Nutrition. 2020;29(3):462-8. doi: 10.6133/apjcn.202009_29(3).0003. PMID: 32990604. I
- 132. Chapman JS, Roddy E, Westhoff G, et al. Postoperative enteral immunonutrition for gynecologic oncology patients undergoing laparotomy decreases wound complications. Gynecologic Oncology. 2015;137(3):523-8. doi: 10.1016/j.ygyno.2015.04.003. PMID: 25888979. S
- 133. Chen B, Zhou Y, Yang P, et al. Clinical observation of preoperative administration of enteral nutrition support in gastric cancer patients at risk of malnutrition. Zhonghua wei chang wai ke za zhi [Chinese journal of gastrointestinal surgery]. 2013;16(1):1055-8. PMID: CN-01120115. NE
- 134. Chen CC, Chang TC, Wang MY, et al. Parenteral glutamine supplement has synergic effects in minimally invasive surgery of subtotal gastrectomy patients. Hepato-Gastroenterology. 2012;59(1):1776-9. doi: 10.5754/hge10270. PMID: 365886527. I

- 135. Chen CJ, Shih SC, Wang HY, et al. Clinical application of total parenteral nutrition in patients with peritoneal carcinomatosis. European Journal of Cancer Care. 2013;22(4):468-73. doi: 10.1111/ecc.12052. PMID: 23730735. P
- 136. Chen FM, Wang JY, Sun LC, et al. Efficacy of medium-chain triglycerides compared with long-chain triglycerides in total parenteral nutrition in patients with digestive tract cancer undergoing surgery. Kaohsiung Journal of Medical Sciences. 2005;21(1):487-94. doi: 10.1016/s1607-551x(09)70156-1. PMID: 41789664. SS
- 137. Chen H, Xia Y, Shi C, et al. Effects of perioperative probiotics administration on patients with colorectal cancer. Chinese journal of clinical nutrition. 2014;22(2):74-81. doi: 10.3760/cma.j.issn.1674-635X.2014.02.002. PMID: CN-00993213. I
- 138. Chen H, Zhang YL. Early enteral nutrition support in patients with gastric cancer after surgical treatment: clinical efficacy and nursing strategies. World Chinese Journal of Digestology. 2014;22(2):3475-8. doi: 10.11569/wcjd.v22.i23.3475. PMID: CN-01072129. NE
- 139. Chen J, Xu M, Zhang Y, et al. Effects of a stepwise, local patient-specific early oral feeding schedule after gastric cancer surgery: a single-center retrospective study from China. Scientific Reports. 2019;9(1):16539. doi: 10.1038/s41598-019-52629-0. PMID: 31719569. I
- 140. Chen JH, Jin HW. Effect of evidence-based nursing combined with nutritional intervention on serum levels of il-6, il-8, and tnf-α in patients after combined laparoscopic-endoscopic radical surgery for rectal cancer. World Chinese Journal of Digestology. 2018;26(1):1137-43. doi: 10.11569/wcjd.v26.i18.1137. PMID: CN-01618730. NE
- 141. Chen JH, Ye JN, Song W, et al. Application of enteral nutrition in preoperative bowel preparation for rectal cancer patients undergoing radical operation. Zhonghua wei chang wai ke za zhi [Chinese journal of gastrointestinal surgery]. 2013;16(1):1059-62. PMID: CN-01076968. NE

- 142. Chen K, Yao F, Chen X, et al. Effectiveness of telerehabilitation on short-term quality of life of patients after esophageal cancer surgery during COVID-19: a single-center, randomized, controlled study. 2021;12(4):1255-64. doi: 10.21037/jgo-21-385. PMID: 34532085. I
- 143. Chen Y, Zhong H, Feng J. Effects of high quality nursing combined with fat emulsion parenteral nutrition support on nutritional status and lipid metabolism of elderly colon cancer patients. Progress in Nutrition. 2021;23(1). doi: 10.23751/pn.v23i1.9831. PMID: 2011728585. I
- 144. Chen YH, Li SH, Chiu YC, et al. Comparative study of esophageal stent and feeding gastrostomy/jejunostomy for tracheoesophageal fistula caused by esophageal squamous cell carcinoma. PLoS ONE [Electronic Resource].
 2012;7(8):e42766. doi: 10.1371/journal.pone.0042766. PMID: 22912737. I
- 145. Chermesh I, Mashiach T, Amit A, et al. Home parenteral nutrition (HTPN) for incurable patients with cancer with gastrointestinal obstruction: do the benefits outweigh the risks? Medical Oncology. 2011;28(1):83-8. doi: 10.1007/s12032-010-9426-2. PMID: 20107935. P
- 146. Chestnut C, Smelser W, Dum T, et al. Glycemic impact of a diet and lifestyle intervention on diabetics and prediabetics during treatment for non-muscle invasive bladder cancer. Nutrition & Cancer. 2020;72(7):1219-24. doi: 10.1080/01635581.2019.1672761. PMID: 31588804. C
- 147. Chi JT, Lin PH, Tolstikov V, et al. The influence of low-carbohydrate diets on the metabolic response to androgen-deprivation therapy in prostate cancer. Prostate. 2021. doi: 10.1002/pros.24136. PMID: 2011406522. O
- 148. Chia-Hui Chen C, Yang YT, Lai IR, et al. Three Nurse-administered Protocols Reduce Nutritional Decline and Frailty in Older Gastrointestinal Surgery Patients: A Cluster Randomized Trial. Journal of the American Medical Directors Association. 2019;20(5):524-9.e3. doi: 10.1016/j.jamda.2018.09.016. PMID: 2001261758. I

- 149. Chlebowski RT, Aragaki AK, Anderson GL, et al. Association of Low-Fat Dietary Pattern With Breast Cancer Overall Survival: A Secondary Analysis of the Women's Health Initiative Randomized Clinical Trial. JAMA Oncology. 2018;4(1):e181212. doi: 10.1001/jamaoncol.2018.1212. PMID: 29800122. OR
- 150. Cho SW, Kim JH, Lee SM, et al. Effect of 8-week nutrition counseling to increase phytochemical rich fruit and vegetable consumption in korean breast cancer patients: a randomized controlled trial. Clinical Nutrition Research. 2014;3(1):39-47. doi: 10.7762/cnr.2014.3.1.39. PMID: 24527419. P
- 151. Choi AH, O'Leary MP, Merchant SJ, et al. Complications of Feeding Jejunostomy Tubes in Patients with Gastroesophageal Cancer. Journal of Gastrointestinal Surgery. 2017;21(2):259-65. doi: 10.1007/s11605-016-3297-6. PMID: 27785689. S
- 152. Chouhan J, Gupta R, Ensor J, et al. Retrospective analysis of systemic chemotherapy and total parenteral nutrition for the treatment of malignant small bowel obstruction. Cancer Medicine. 2016;5(2):239-47. doi: 10.1002/cam4.587. PMID: 26714799. C
- 153. Cipolla BG, Havouis R, Moulinoux JP. Polyamine reduced diet (PRD) nutrition therapy in hormone refractory prostate cancer patients. Biomedicine & Pharmacotherapy. 2010;64(5):363-8. doi: 10.1016/j.biopha.2009.09.022. PMID: 20106631. S
- 154. Cipolla BG, Miglianico L, Bligny D, et al. Effect of combination of a polyamine-free oral nutritional supplement and docetaxel in symptomatic, metastatic castration-resistant prostate cancer patients. BioMedicine (Netherlands). 2013;3(4):153-9. doi: 10.1016/j.biomed.2013.07.001. PMID: 52763017. C
- 155. Claudino MM, Lopes JR, Rodrigues VD, et al. Postoperative complication rate and survival of patients with gastric cancer undergoing immunonutrition: A retrospective study. Nutrition. 2020;70:110590. doi: 10.1016/j.nut.2019.110590. PMID: 31739174. I

- 156. Clavel S, Fortin B, Despres P, et al. Enteral feeding during chemoradiotherapy for advanced head-and-neck cancer: a singleinstitution experience using a reactive approach. International Journal of Radiation Oncology, Biology, Physics. 2011;79(3):763-9. doi: 10.1016/j.ijrobp.2009.12.032. PMID: 20510546. I
- 157. Coghlin Dickson TM, Wong RM, Negrin RS, et al. Effect of oral glutamine supplementation during bone marrow transplantation. Journal of Parenteral and Enteral Nutrition. 2000;24(2):61-6. doi: 10.1177/014860710002400261. PMID: 30691626. P
- 158. Cohen CW, Fontaine KR, Arend RC, et al. A Ketogenic Diet Reduces Central Obesity and Serum Insulin in Women with Ovarian or Endometrial Cancer. Journal of Nutrition. 2018;148(8):1253-60. doi: 10.1093/jn/nxy119. PMID: 30137481. SS
- 159. Cohen CW, Fontaine KR, Arend RC, et al. A Ketogenic Diet Is Acceptable in Women with Ovarian and Endometrial Cancer and Has No Adverse Effects on Blood Lipids: A Randomized, Controlled Trial. Nutrition & Cancer. 2020;72(4):584-94. doi: 10.1080/01635581.2019.1645864. PMID: 31352797. O
- 160. Cohen CW, Fontaine KR, Arend RC, et al. Favorable Effects of a Ketogenic Diet on Physical Function, Perceived Energy, and Food Cravings in Women with Ovarian or Endometrial Cancer: A Randomized, Controlled Trial. Nutrients. 2018;10(9):30. doi: 10.3390/nu10091187. PMID: 30200193. SS
- 161. Cong M, Li S, Liu X, et al. Effect of nutritional counseling combined with oral nutritional supplements on clinical outcome of esophageal cancer patients under radiotherapy treatment. Chinese journal of clinical nutrition. 2016;24(2):86-90. doi: 10.3760/cma.j.issn.1674-635X.2016.02.005. PMID: CN-01473907. NE

- 162. Cong M, Song C, Zou B, et al. Impact of glutamine, eicosapntemacnioc acid, branched-chain amino acid supplements on nutritional status and treatment compliance of esophageal cancer patients on concurrent chemoradiotherapy and gastric cancer patients on chemotherapy. National medical journal of china. 2015;95(1):766-9. doi: 10.3760/cma.j.issn.0376-2491.2015.10.011. PMID: CN-01098135. NE
- 163. Cong MH, Li SL, Cheng GW, et al. An Interdisciplinary Nutrition Support Team Improves Clinical and Hospitalized Outcomes of Esophageal Cancer Patients with Concurrent Chemoradiotherapy. Chinese Medical Journal. 2015;128(2):3003-7. doi: 10.4103/0366-6999.168963. PMID: 26608978. I
- 164. Cooper SC, Hulley CM, Grimley CE, et al. Perioperative peripheral parenteral nutrition for patients undergoing esophagectomy for cancer: a pilot study of safety, surgical, and nutritional outcomes. International Surgery. 2006;91(6):358-64. PMID: 17256437. SS
- 165. Correa P, Fontham ETH, Bravo JC, et al. Chemoprevention of gastric dysplasia: Randomized trial of antioxidant supplements and anti-Helicobacter pylori therapy. Journal of the National Cancer Institute. 2000;92(2):1881-8. doi: 10.1093/jnci/92.23.1881. PMID: 32036918. P
- 166. Corry J, Poon W, McPhee N, et al. Randomized study of percutaneous endoscopic gastrostomy versus nasogastric tubes for enteral feeding in head and neck cancer patients treated with (chemo)radiation. Journal of Medical Imaging & Radiation Oncology. 2008;52(5):503-10. doi: 10.1111/j.1440-1673.2008.02003.x. PMID: 19032398. I
- 167. Corry J, Poon W, McPhee N, et al. Prospective study of percutaneous endoscopic gastrostomy tubes versus nasogastric tubes for enteral feeding in patients with head and neck cancer undergoing (chemo)radiation. Head & Neck. 2009;31(7):867-76. doi: 10.1002/hed.21044. PMID: 19296528. I

- 168. Costa Fortes R, Carvalho Garbi Novaes MR. The effects of Agaricus sylvaticus fungi dietary supplementation on the metabolism and blood pressure of patients with colorectal cancer during post surgical phase. Nutricion Hospitalaria. 2011;26(1):176-86. PMID: 21519745. O
- 169. Costa Fortes R, Lacorte Recova V, Lima Melo A, et al. Life quality of postsurgical patients with colorectal cancer after supplemented diet with agaricus sylvaticus fungus. Nutricion Hospitalaria. 2010;25(4):586-96. PMID: 20694295. OR
- 170. Cotogni P, Monge T, Fadda M, et al. Bioelectrical impedance analysis for monitoring cancer patients receiving chemotherapy and home parenteral nutrition. BMC Cancer. 2018;18(1):990. doi: 10.1186/s12885-018-4904-6. PMID: 30332998. S
- 171. Cotogni P, Monge T, Passera R, et al. Clinical characteristics and predictive factors of survival of 761 cancer patients on home parenteral nutrition: A prospective, cohort study. Cancer Medicine. 2020;9(1):4686-98. doi: 10.1002/cam4.3064. PMID: 32412178. S
- 172. Cotogni P, Ossola M, Passera R, et al. Home parenteral nutrition versus artificial hydration in malnourished patients with cancer in palliative care: a prospective, cohort survival study. BMJ supportive & palliative care. 2020;21:21. doi: 10.1136/bmjspcare-2020-002343. PMID: 32826263. P
- 173. Cox S, Powell C, Carter B, et al. Role of nutritional status and intervention in oesophageal cancer treated with definitive chemoradiotherapy: outcomes from SCOPE1. British Journal of Cancer. 2016;115(2):172-7. doi: 10.1038/bjc.2016.129. PMID: 27328311. I
- 174. Critselis E, Panagiotakos DB, Machairas A, et al. Postoperative hypoproteinemia in cancer patients following extensive abdominal surgery despite parenteral nutritional support. Nutrition & Cancer. 2011;63(7):1021-8. doi: 10.1080/01635581.2011.606392. PMID: 21919648. I

- 175. Crombie JM, Ng S, Spurgin AL, et al. Swallowing outcomes and PEG dependence in head and neck cancer patients receiving definitive or adjuvant radiotherapy +/chemotherapy with a proactive PEG: a prospective study with long term follow up. Oral Oncology. 2015;51(6):622-8. doi: 10.1016/j.oraloncology.2015.03.006. PMID: 25865554. C
- 176. Cui H, Yang X, Tang D, et al. Effect of oral nutritional supplementation on nutritional status and quality of life in patients with gastric cancer after operation (23 cases RCT observations). Chinese journal of clinical nutrition. 2017;25(3):183-8. doi: 10.3760/cma.j.issn.1674-635X.2017.03.010. PMID: CN-01417387. NE
- 177. Culine S, Chambrier C, Tadmouri A, et al. Home parenteral nutrition improves quality of life and nutritional status in patients with cancer: a French observational multicentre study. Supportive Care in Cancer. 2014;22(7):1867-74. doi: 10.1007/s00520-014-2164-9. PMID: 24557011. C
- 178. Dalton BGA, Friedant AJ, Su S, et al. Benefits of Supplemental Jejunostomy Tube Feeding During Neoadjuvant Therapy in Patients with Locally Advanced, Potentially Resectable Esophageal Cancer. Journal of Laparoendoscopic & Advanced Surgical Techniques. Part A. 2017;27(1):1279-83. doi: 10.1089/lap.2017.0320. PMID: 28777676. S
- 179. Daneshmand S, Ahmadi H, Schuckman AK, et al. Enhanced recovery protocol after radical cystectomy for bladder cancer. Journal of Urology. 2014;192(1):50-5. doi: 10.1016/j.juro.2014.01.097. PMID: 24518775. I
- 180. Dann GC, Squires MH, 3rd, Postlewait LM, et al. An assessment of feeding jejunostomy tube placement at the time of resection for gastric adenocarcinoma: A seven-institution analysis of 837 patients from the U.S. gastric cancer collaborative. Journal of Surgical Oncology. 2015;112(2):195-202. doi: 10.1002/jso.23983. PMID: 26240027. I

- 181. Das Virgens IPA, de Carvalho ALM, Nagashima YG, et al. Is perioperative fasting associated with complications, length of hospital stay and mortality among gastric and colorectal cancer patients? A cohort study. Sao Paulo Medical Journal. 2020;138(5):407-13. doi: 10.1590/1516-3180.2020.0084.R1.30062020. PMID: 2005469440. C
- 182. Dashti S, Abdul Hamid H, Mohamad Saini S, et al. A randomised controlled trial on the effects of a structural education module among women with polycystic ovarian syndrome on nutrition and physical activity changes. BMC Womens Health. 2022 Jul 6;22(1):277. doi: 10.1186/s12905-022-01861-4. PMID: 35794564. P
- 183. Datta M, Taylor ML, Frizzell B. Dietary and serum lycopene levels in prostate cancer patients undergoing intensity-modulated radiation therapy. Journal of Medicinal Food. 2013;16(1):1131-7. doi: 10.1089/jmf.2012.0223. PMID: 24180552. SS
- 184. Davies SJ, West MA, Rahman SA, et al. Oesophageal cancer: The effect of early nutrition support on clinical outcomes. Clinical Nutrition ESPEN. 2021;42:117-23. doi: 10.1016/j.clnesp.2021.02.006. PMID: 33745564. S
- 185. Davis CH, Ikoma N, Mansfield PF, et al. Comparison of laparoscopy versus minilaparotomy for jejunostomy placement in patients with gastric adenocarcinoma. Surgical Endoscopy. 2020. doi: 10.1007/s00464-020-08155-6. PMID: 2007240014. S
- 186. Davoodi SH, Yousefinejad V, Ghaderi B, et al. Oral Propolis, Nutritional Status and Quality of Life with Chemotherapy for Breast Cancer: A Randomized, Double-Blind Clinical Trial. Nutr Cancer. 2022;74(6):2029-37. doi: 10.1080/01635581.2021.1988118. PMID: 34622721. I
- 187. Dawson ER, Morley SE, Robertson AG, et al. Increasing dietary supervision can reduce weight loss in oral cancer patients. Nutrition and Cancer. 2001;41(1):70-4. PMID: 34666223. I

- 188. de Carvalho CS, Silva TH, Andre JCS, et al. Preoperative Fasting Abbreviation with Whey Protein Reduces the Occurrence of Postoperative Complications in Patients With Head and Neck Cancer: A Randomized Clinical Trial. Nutrition in Clinical Practice. 2020. doi: 10.1002/ncp.10624. PMID: 2007690683. SS
- 189. de Groot S, Lugtenberg RT, Cohen D, et al. Fasting mimicking diet as an adjunct to neoadjuvant chemotherapy for breast cancer in the multicentre randomized phase 2 DIRECT trial. Nature communications. 2020;11(1):3083. doi: 10.1038/s41467-020-16138-3. PMID: CN-02139812.
- 190. de Groot S, Vreeswijk MP, Welters MJ, et al. The effects of short-term fasting on tolerance to (neo) adjuvant chemotherapy in HER2-negative breast cancer patients: a randomized pilot study. BMC Cancer. 2015;15:652. doi: 10.1186/s12885-015-1663-5. PMID: 26438237. SS
- 191. de Liz S, Vieira FGK, de Assis MAA, et al. Adherence to the WCRF/AICR for Women in Breast Cancer Adjuvant Treatment submitted to Educational Nutritional Intervention. Nutrition & Cancer. 2018;70(5):737-47. doi: 10.1080/01635581.2017.1380207. PMID: 29166141. S
- 192. de Luis DA, Aller R, Izaola O, et al. Postsurgery enteral nutrition in head and neck cancer patients. European Journal of Clinical Nutrition. 2002;56(1):1126-9. doi: 10.1038/sj.ejcn.1601458. PMID: 35414799. SS
- 193. de Luis DA, Izaola O, Cuellar L, et al. A randomized clinical trial with two doses of a omega 3 fatty acids oral and arginine enhanced formula in clinical and biochemical parameters of head and neck cancer ambulatory patients. European Review for Medical & Pharmacological Sciences. 2013;17(8):1090-4. PMID: 23661523. SS
- 194. de Man FM, van Eerden RAG, van Doorn GM, et al. Effects of Protein and Calorie Restriction on the Metabolism and Toxicity Profile of Irinotecan in Cancer Patients. Clinical Pharmacology and Therapeutics. 2021;109(5):1304-13. doi: 10.1002/cpt.2094. PMID: 2007513579. I

- 195. de Oliveira Faria S, Howell D, Vamondes Kulcsar MA, et al. Nutritional outcomes in head and neck cancer patients: is intensive nutritional care worth it? Cancer Treatment And Research Communications. 2020;25:100233. doi: 10.1016/j.ctarc.2020.100233. PMID: 33161323. C
- 196. De Waele E, Mattens S, Honore PM, et al. Nutrition therapy in cachectic cancer patients. The Tight Caloric Control (TiCaCo) pilot trial. Appetite. 2015;91:298-301. doi: 10.1016/j.appet.2015.04.049. PMID: 25912786. SS
- 197. Deepjyoti K, Bannoth S, Purkayastha J, et al. Nasojejunal Feeding Is Safe and Effective Alternative to Feeding Jejunostomy for Postoperative Enteral Nutrition in Gastric Cancer Patients. South Asian Journal of Cancer. 2020;9(2):70-3. doi: 10.1055/s-0040-1721218. PMID: 33354547. O
- 198. Deibert CM, Silva MV, RoyChoudhury A, et al. A Prospective Randomized Trial of the Effects of Early Enteral Feeding After Radical Cystectomy. Urology. 2016;96:69-73. doi: 10.1016/j.urology.2016.06.045. PMID: 612934963. C
- 199. Della Valle S, Colatruglio S, La Vela V, et al. Nutritional intervention in head and neck cancer patients during chemo-radiotherapy. Nutrition. 2018;51(-):95-7. doi: 10.1016/j.nut.2017.12.012. PMID: 29625408. S
- 200. Demark-Wahnefried W, Nix JW, Hunter GR, et al. Feasibility outcomes of a presurgical randomized controlled trial exploring the impact of caloric restriction and increased physical activity versus a wait-list control on tumor characteristics and circulating biomarkers in men electing prostatectomy for prostate cancer. BMC Cancer. 2016;16:61. doi: 10.1186/s12885-016-2075x. PMID: 26850040. I
- 201. Demark-Wahnefried W, Rais-Bahrami S, Desmond RA, et al. Presurgical weight loss affects tumour traits and circulating biomarkers in men with prostate cancer. British Journal of Cancer.
 2017;117(9):1303-13. doi: 10.1038/bjc.2017.303. PMID: 28881355. I

- 202. Demark-Wahnefried W, Rogers LQ, Gibson JT, et al. Randomized trial of weight loss in primary breast cancer: Impact on body composition, circulating biomarkers and tumor characteristics. International Journal of Cancer. 2020;146(1):2784-96. doi: 10.1002/ijc.32637. PMID: 31442303. I
- 203. DeMille D, Deming P, Lupinacci P, et al. The effect of the neutropenic diet in the outpatient setting: A pilot study. Oncology Nursing Forum. 2006;33(2):337-43. doi: 10.1188/onf.06.337-343. PMID: 43672487. C
- 204. Demiral S, Beyzadeoglu M, Sager O, et al. Evaluation of transforming growth factorbeta2 for radiation-induced diarrhea after pelvic radiotherapy. Tumori. 2015;101(5):474-7. doi: 10.5301/tj.5000328. PMID: 25983098. SS
- 205. Demirer S, Aydintu GS, Ustun C, et al. Comparison of the efficacy of medium chain triglycerides with long chain triglycerides in total parenteral nutrition in patients with hematologic malignancies undergoing peripheral blood stem cell transplantation. Clinical Nutrition. 2000;19(4):253-8. doi: 10.1054/clnu.2000.0101. PMID: 30682263. OR
- 206. Den E, Steer B, Quinn P, et al. Effect of an Evidence-Based Nutrition Care Pathway for Cancer Patients Undergoing Gastrointestinal and Pelvic Surgery. Nutrition & Cancer. 2020:1-8. doi: 10.1080/01635581.2020.1839517. PMID: 33138651. PT
- 207. Deng G, Lin H, Seidman A, et al. A phase I/II trial of a polysaccharide extract from Grifola frondosa (Maitake mushroom) in breast cancer patients: immunological effects. Journal of Cancer Research & Clinical Oncology. 2009;135(9):1215-21. doi: 10.1007/s00432-009-0562-z. PMID: 19253021. P
- 208. Deng S, Zou C. Observation on application effect of TCM diet intervention in improving malnutrition in patients with endstage liver cancer. Chinese nursing research. 2015;29(4):1444-7. doi: 10.3969/j.issn.1009-6493.2015.12.012. PMID: CN-02198286. NE

- 209. Desideri I, Francolini G, Becherini C, et al. Use of an alpha lipoic, methylsulfonylmethane and bromelain dietary supplement (Opera^R) for chemotherapy-induced peripheral neuropathy management, a prospective study. Medical Oncology. 2017;34(3):46. doi: 10.1007/s12032-017-0907-4. PMID: 28205185. S
- 210. Deutz NE, Safar A, Schutzler S, et al. Muscle protein synthesis in cancer patients can be stimulated with a specially formulated medical food. Clinical Nutrition. 2011;30(6):759-68. doi: 10.1016/j.clnu.2011.05.008. PMID: 21683485. SS
- 211. deVere White RW, Tsodikov A, Stapp EC, et al. Effects of a high dose, aglycone-rich soy extract on prostate-specific antigen and serum isoflavone concentrations in men with localized prostate cancer. Nutrition & Cancer. 2010;62(8):1036-43. doi: 10.1080/01635581.2010.492085. PMID: 21058191. P
- 212. Di Franco R, Calvanese M, Murino P, et al. Skin toxicity from external beam radiation therapy in breast cancer patients: protective effects of Resveratrol, Lycopene, Vitamin C and anthocianin (Ixor R). Radiation Oncology. 2012;7:12. doi: 10.1186/1748-717x-7-12. PMID: 22289566. I
- 213. Di Renzo L, Marchetti M, Cioccoloni G, et al. Role of phase angle in the evaluation of effect of an immuno-enhanced formula in post-surgical cancer patients: a randomized clinical trial. European Review for Medical & Pharmacological Sciences. 2019;23(3):1322-34. doi: 10.26355/eurrev_201902_17027. PMID: 30779100. SS
- 214. di Sebastiano P, Festa L, De Bonis A, et al. A modified fast-track program for pancreatic surgery: a prospective single-center experience. Langenbecks Archives of Surgery. 2011;396(3):345-51. doi: 10.1007/s00423-010-0707-1. PMID: 20703500. I

- 215. Diaz-Feijoo B, Agusti N, Sebio R, et al. A multimodal prehabilitation program for the reduction of post-operative complications after surgery in advanced ovarian cancer under an ERAS pathway: a randomized multicenter trial (SOPHIE). Int J Gynecol Cancer. 2022 Jul 6;06:06. doi: 10.1136/ijgc-2022-003652. PMID: 35793862. S
- 216. Dickson EA, Keeler BD, Ng O, et al. Preoperative intravenous iron therapy and survival after colorectal cancer surgery: long-term results from the IVICA randomised controlled trial. Colorectal Disease. 2020. doi: 10.1111/codi.15342. PMID: 32871616. I
- 217. Dijksterhuis WPM, Latenstein AEJ, Van Kleef JJ, et al. Cachexia and dietetic interventions in patients with esophagogastric cancer: A multicenter cohort study. JNCCN Journal of the National Comprehensive Cancer Network. 2021;19(2):144-52. doi: 10.6004/jnccn.2020.7615. PMID: 2011085388. I
- 218. Ding GP, Chen P, Yi ZB, et al. Roles of nutrition risk screening and preventive enteral nutritional support before radical resection of gastric cancer. Zhonghua wei chang wai ke za zhi [Chinese journal of gastrointestinal surgery]. 2009;12(2):141-4. PMID: CN-00733330. X
- 219. Ding H, Xu J, You J, et al. Effects of enteral nutrition support combined with enhanced recovery after surgery on the nutritional status, immune function, and prognosis of patients with esophageal cancer after Ivor-Lewis operation. Journal of Thoracic Disease. 2020;12(1):7337-45. doi: 10.21037/jtd-20-3410. PMID: 33447423. I
- 220. Ding H, Xu J, You J, et al. Effects of enteral nutrition support combined with enhanced recovery after surgery on the nutritional status, immune function, and prognosis of patients with esophageal cancer after Ivor-Lewis operation. J Thorac Dis. 2020 Dec;12(12):7337-45. doi: 10.21037/jtd-20-3410. PMID: 33447423. I
- 221. Ding Q, Chen W, Gu Y, et al. Accelerated rehabilitation combined with enteral nutrition in the management of lung cancer surgery patients. Asia Pacific Journal of Clinical Nutrition. 2020;29(2):274-9. doi: 10.6133/apjcn.202007_29(2).0010. PMID: 32674235. I

- 222. Dintinjana RD, Guina T, Krznaric Z, et al. Effects of nutritional support in patients with colorectal cancer during chemotherapy. Collegium Antropologicum. 2008;32(3):737-40. PMID: 18982745. S
- 223. Diver E, O'Connor O, Garrett L, et al. Modest benefit of total parenteral nutrition and chemotherapy after venting gastrostomy tube placement. Gynecologic Oncology. 2013;129(2):332-5. doi: 10.1016/j.ygyno.2013.02.002. PMID: 23402902. I
- 224. Djuric Z, Ellsworth J, Rapai M, et al. A diet and exercise intervention in women being treated for breast cancer. Cancer Research. 2010. PMID: CN-01984892. PT
- 225. Djuric Z, Ellsworth JS, Weldon AL, et al. A diet and exercise intervention during Chemotherapy for breast cancer. Open Obesity Journal. 2011;3:87-97. doi: 10.2174/1876823701103010087. PMID: 362084717. SS
- 226. Dobrila-Dintinjana R, Trivanovic D, Zelic M, et al. Nutritional support in patients with colorectal cancer during chemotherapy: Does it work? Hepato-Gastroenterology. 2013;60(1):475-80. doi: 10.5754/hge12710. PMID: 369214138. C
- 227. Doglietto GB, Pacelli F, Papa V, et al. Use of a nasojejunal tube after total gastrectomy: a multicentre prospective randomised trial. Chirurgia Italiana. 2004;56(6):761-8. PMID: CN-00502371. NE
- 228. Dong QT, Zhang XD, Yu Z. Integrated Chinese and Western medical treatment on postoperative fatigue syndrome in patients with gastric cancer. Zhongguo zhong xi yi jie he za zhi zhongguo zhongxiyi jiehe zazhi = chinese journal of integrated traditional and western medicine. 2010;30(1):1036-40. PMID: CN-00912548. NE
- 229. Donohoe CL, Healy LA, Fanning M, et al. Impact of supplemental home enteral feeding postesophagectomy on nutrition, body composition, quality of life, and patient satisfaction. Diseases of the Esophagus. 2017;30(9):1-9. doi: 10.1093/dote/dox063. PMID: 28859364. C

- 230. Doppalapudi R, Vundavalli S, Prabhat MP. Effect of probiotic bacteria on oral Candida in head- and neck-radiotherapy patients: A randomized clinical trial. Journal of Cancer Research & Therapeutics. 2020;16(3):470-7. doi: 10.4103/jcrt.JCRT_334_18. PMID: 32719253. I
- 231. Dou L, Wang X, Cao Y, et al. Relationship between Postoperative Recovery and Nutrition Risk Screened by NRS 2002 and Nutrition Support Status in Patients with Gastrointestinal Cancer. Nutrition & Cancer. 2020;72(1):33-40. doi: 10.1080/01635581.2019.1612927. PMID: 31079488. I
- 232. Dou S, Ding H, Jiang W, et al. Effect of oral supplements on the nutritional status of nasopharyngeal carcinoma patients undergoing concurrent chemotherapy: a randomized controlled Phase II trial. Journal of cancer research and therapeutics. 2020;16(7):1678-85. doi: 10.4103/jcrt.JCRT_273_20. PMID: CN-02258023. D
- 233. Dou S, Ding H, Jiang W, et al. Effect of oral supplements on the nutritional status of nasopharyngeal carcinoma patients undergoing concurrent chemotherapy: A randomized controlled Phase II trial. Journal of Cancer Research & Therapeutics. 2020;16(7):1678-85. doi: 10.4103/jcrt.JCRT_273_20. PMID: 33565516. SS
- 234. Dragan S, Ilina R, Ursoniu S, et al. Role of multi-component functional foods in the complex treatment of patients with advanced breast cancer. Revista medico-chirurgicala a Societat&x031C;ii de Medici si Naturalisti din Iasi. 2007;111(4):877-84. PMID: 351916069. SS
- 235. Drissi M, Cwieluch O, Lechner P, et al. Nutrition care in patients with cancer: A retrospective multicenter analysis of current practice - Indications for further studies? Clinical Nutrition. 2015;34(2):207-11. doi: 10.1016/j.clnu.2014.03.002. PMID: 24679553. I
- 236. Du N, Rao Z, Che G, et al. What is Result: short-term Medium Chain Triglyceride Diet Effective on

- Postoperative Outcome in Lung Cancer Surgery? A Prospective Randomized Study. Zhongguo fei ai za zhi [Chinese journal of lung cancer]. 2016;19(1):821-6. doi: 10.3779/j.issn.1009-3419.2016.12.04. PMID: CN-01342286. NE
- 237. Dzierzanowski T, Sobocki J. Survival of Patients with Multi-Level Malignant Bowel Obstruction on Total Parenteral Nutrition at Home. Nutrients. 2021;13(3):10. doi: 10.3390/nu13030889. PMID: 33801869. P
- 238. Eda K, Uzer K, Murat T, et al. The effects of enteral glutamine on radiotherapy induced dermatitis in breast cancer. Clinical nutrition (Edinburgh, Scotland). 2016;35(2):436-9. doi: 10.1016/j.clnu.2015.03.009. PMID: CN-01141996. SS
- 239. Ehresman J, Ahmed AK, Schilling A, et al. Preoperative Nutrition Consults Associated with Decreased Postoperative Complication Rate and Decreased Length of Hospital Stay After Spine Metastasis Surgery. World Neurosurgery. 2020;133:e173-e9. doi: 10.1016/j.wneu.2019.08.197. PMID: 31493604. P
- 240. El Hamamsy M, Bondok R, Shaheen S, et al. Safety and efficacy of adding intravenous Nacetylcysteine to parenteral L-alanyl-Lglutamine in hospitalized patients undergoing surgery of the colon: A randomized controlled trial. Annals of Saudi Medicine. 2019;39(4):251-7. doi: 10.5144/0256-4947.2019.251. PMID: 2002527704. P
- 241. Elshaer M, Gravante G, White J, et al. Routes of early enteral nutrition following oesophagectomy. Annals of the Royal College of Surgeons of England. 2016;98(7):461-7. doi: 10.1308/rcsann.2016.0198. PMID: 27388543. I
- 242. Engelen M, Safar AM, Bartter T, et al. High anabolic potential of essential amino acid mixtures in advanced nonsmall cell lung cancer. Annals of Oncology. 2015;26(9):1960-6. doi: 10.1093/annonc/mdv271. PMID: 26113648. O

- 243. Erdem NZ, Yasti AC, Atli M, et al. The effects of perioperative oral enteral support with glutamine-added elemental formulas in patients with gastrointestinal cancers. A prospective, randomized, clinical study. Nutrition Research. 2002;22(9):977-88. doi: 10.1016/s0271-5317(02)00407-4. PMID: 34977236. OR
- 244. Espeli V, Vergotte S, Dietrich PY, et al. Prolonged Versus Short-Duration Use of Nasogastric Tubes in Patients with Head and Neck Cancer During Radiotherapy Alone or Combined Chemoradiotherapy. Nutrition & Cancer. 2018;70(7):1069-74. doi: 10.1080/01635581.2018.1497670. PMID: 30273007. S
- 245. Eze N, Jefford JM, Wolf D, et al. PEG and RIG tube feeding in Head and Neck patients: A retrospective review of complications and outcome. Journal of Evaluation in Clinical Practice. 2007;13(5):817-9. doi: 10.1111/j.1365-2753.2006.00741.x. PMID: 47373595. S
- 246. Faber J, Uitdehaag MJ, Spaander M, et al. Improved body weight and performance status and reduced serum PGE2 levels after nutritional intervention with a specific medical food in newly diagnosed patients with esophageal cancer or adenocarcinoma of the gastro-esophageal junction. Journal of Cachexia, Sarcopenia and Muscle. 2015;6(1):32-44. doi: 10.1002/jcsm.12009. PMID: 26136410. SS
- 247. Falciglia GA, Whittle KM, Levin LS, et al. A clinical-based intervention improves diet in patients with head and neck cancer at risk for second primary cancer. Journal of the American Dietetic Association. 2005;105(1):1609-12. doi: 10.1016/j.jada.2005.07.009. PMID: 41393044. O
- 248. Farshi Radvar F, Mohammad-Zadeh M, Mahdavi R, et al. Effect of synbiotic supplementation on matrix metalloproteinase enzymes, quality of life and dietary intake and weight changes in rectal cancer patients undergoing neoadjuvant chemoradiotherapy. Mediterranean Journal of Nutrition and Metabolism. 2020;13(3):225-35. doi: 10.3233/mnm-200413. PMID: CN-02200630. I

- 249. Febvey-Combes O, Jobard E, Rossary A, et al. Effects of an Exercise and Nutritional Intervention on Circulating Biomarkers and Metabolomic Profiling During Adjuvant Treatment for Localized Breast Cancer: Results From the PASAPAS Feasibility Randomized Controlled Trial. Integrative Cancer Therapies. 2021;20:1534735420977666. doi: 10.1177/1534735420977666. PMID: 33655799. I
- 250. Federico A, Iodice P, Del Rio A, et al. Effects of selenium and zinc supplementation on nutritional status in patients with cancer of digestive tract. European Journal of Clinical Nutrition. 2001;55(4):293-7. doi: 10.1038/sj.ejcn.1601157. PMID: 32273129. S
- 251. Felekis D, Eleftheriadou A, Papadakos G, et al. Effect of perioperative immuno-enhanced enteral nutrition on inflammatory response, nutritional status, and outcomes in head and neck cancer patients undergoing major surgery. Nutrition & Cancer. 2010;62(8):1105-12. doi: 10.1080/01635581.2010.494336. PMID: 21058198. SS
- 252. Feng HQ, Dai L, Ma SH, et al. Impact of early enteral nutrition on the intestinal motility of patients after esophagectomy. Zhonghua wei chang wai ke za zhi [Chinese journal of gastrointestinal surgery]. 2012;15(9):957-9. PMID: CN-00979318. NE
- 253. Fenton JR, Bergeron EJ, Coello M, et al. Feeding jejunostomy tubes placed during esophagectomy: are they necessary? Annals of Thoracic Surgery. 2011;92(2):504-11; discussion 11-2. doi: 10.1016/j.athoracsur.2011.03.101. PMID: 21704294. S
- 254. Ferreira IB, Lima E, Canto PPL, et al. Oral Nutritional Supplementation Affects the Dietary Intake and Body Weight of Head and Neck Cancer Patients during (Chemo) Radiotherapy. Nutrients. 2020;12(9):20. doi: 10.3390/nu12092516. PMID: 32825254. C
- 255. Ferreira V, Minnella EM, Awasthi R, et al. Multimodal Prehabilitation for Lung Cancer Surgery: A Randomized Controlled Trial. Annals of Thoracic Surgery. 2020;12:12. doi: 10.1016/j.athoracsur.2020.11.022. PMID: 33321089. I

- 256. Findlay M, Rankin NM, Shaw T, et al. Best Evidence to Best Practice: Implementing an Innovative Model of Nutrition Care for Patients with Head and Neck Cancer Improves Outcomes. Nutrients. 2020;12(5):19. doi: 10.3390/nu12051465. PMID: 32438607. C
- 257. Finger AE. Head and neck cancer patients under (chemo-)radiotherapy undergoing nutritional intervention: results from the prospective randomized HEADNUT-trial. Annals of Oncology. 2021;32:S810-. doi: 10.1016/j.annonc.2021.08.1327. PMID: CN-02337593. PT
- 258. Finocchiaro C, Gervasio S, Agnello E, et al. Multicentric study on home parenteral nutrition in advanced cancer patients. Rivista Italiana di Nutrizione Parenterale ed Enterale. 2002;20(2):98-107. PMID: CN-01620935. NE
- 259. Flesch AT, Tonial ST, Contu PC, et al. Perioperative synbiotics administration decreases postoperative infections in patients with colorectal cancer: a randomized, double-blind clinical trial. Revista do Colegio Brasileiro de Cirurgioes. 2017;44(6):567-73. doi: 10.1590/0100-69912017006004. PMID: 29267553. I
- 260. Focht BC, Lucas AR, Grainger E, et al. Effects of a Group-Mediated Exercise and Dietary Intervention in the Treatment of Prostate Cancer Patients Undergoing Androgen Deprivation Therapy: Results From the IDEA-P Trial. Annals of Behavioral Medicine. 2018;52(5):412-28. doi: 10.1093/abm/kax002. PMID: 29684136. I
- 261. Focht BC, Lucas AR, Grainger E, et al. Effects of a Group-Mediated Cognitive Behavioral Lifestyle Intervention on Select Social Cognitive Outcomes in Prostate Cancer Patients Undergoing Androgen Deprivation Therapy. Integrative Cancer Therapies. 2019;18:1534735419893764. doi: 10.1177/1534735419893764. PMID: 31838879. I
- 262. Fonseca J, Santos CA, Brito J. Malnutrition and Clinical Outcome of 234 Head and Neck Cancer Patients who Underwent Percutaneous Endoscopic Gastrostomy. Nutrition & Cancer. 2016;68(4):589-97. doi: 10.1080/01635581.2016.1158297. PMID: 27144413. C

- 263. Foucaut AM, Morelle M, Kempf-Lepine AS, et al. Feasibility of an exercise and nutritional intervention for weight management during adjuvant treatment for localized breast cancer: the PASAPAS randomized controlled trial. Supportive Care in Cancer. 2019;27(9):3449-61. doi: 10.1007/s00520-019-4658-y. PMID: 30680617. I
- 264. Fradet S, Pelletier JF, Singbo N, et al. Effects of omega-3 fatty acids supplementation on perioperative blood loss and complications after radical prostatectomy. Clinical Nutrition ESPEN. 2022;47:221-6. doi: 10.1016/j.clnesp.2021.12.011. PMID: 35063205. I
- 265. Fraser S. Exercise and nutrition to treat adverse musculoskeletal effects of hormone therapy in prostate cancer. Asia-Pacific journal of clinical oncology. 2019;15(S):67-8. doi: 10.1111/ajco.13262. PMID: CN-02258186. PT
- 266. Freedland SJ, Allen J, Jarman A, et al. A Randomized controlled trial of a 6-month low-carbohydrate intervention on disease progression in men with recurrent prostate cancer: Carbohydrate and prostate study 2 (CAPS2). Clinical Cancer Research. 2020;26(1):3035-43. doi: 10.1158/1078-0432.Ccr-19-3873. PMID: 2006839435. P
- 267. Froghi F, Sanders G, Berrisford R, et al. A randomised trial of post-discharge enteral feeding following surgical resection of an upper gastrointestinal malignancy. Clinical Nutrition. 2017;36(6):1516-9. doi: 10.1016/j.clnu.2016.10.022. PMID: 27842926. SS
- 268. Fruge AD, Van der Pol W, Rogers LQ, et al. Fecal Akkermansia muciniphila Is Associated with Body Composition and Microbiota Diversity in Overweight and Obese Women with Breast Cancer Participating in a Presurgical Weight Loss Trial. Journal of the Academy of Nutrition & Dietetics. 2020;120(4):650-9. doi: 10.1016/j.jand.2018.08.164. PMID: 30420171. I
- 269. Fujita T, Daiko H, Nishimura M. Early enteral nutrition reduces the rate of life-threatening complications after thoracic esophagectomy in patients with esophageal cancer.
 European Surgical Research. 2012;48(2):79-84. doi: 10.1159/000336574. PMID: 22377820. OR

- 270. Fukuda T, Seto Y, Yamada K, et al. Can immune-enhancing nutrients reduce postoperative complications in patients undergoing esophageal surgery? Diseases of the Esophagus. 2008;21(8):708-11. doi: 10.1111/j.1442-2050.2008.00861.x. PMID: 18847452. S
- 271. Fundora Ramos MI, Maden LB, Casanova FO, et al. Oncoxin-Viusid may improve quality of life and survival in patients with hormone-refractory prostate cancer undergoing onco-specific treatments. Molecular & Clinical Oncology. 2021;14(1):5. doi: 10.3892/mco.2020.2167. PMID: 33235733. S
- 272. Furness K, Silvers MA, Savva J, et al. Longterm follow-up of the potential benefits of early nutritional intervention in adults with upper gastrointestinal cancer: a pilot randomised trial. Supportive Care in Cancer. 2017;25(1):3587-93. doi: 10.1007/s00520-017-3789-2. PMID: 28612158. SS
- 273. Furuta M, Yokota T, Tsushima T, et al. Comparison of enteral nutrition with total parenteral nutrition for patients with locally advanced unresectable esophageal cancer harboring dysphagia in definitive chemoradiotherapy. Japanese Journal of Clinical Oncology. 2019;49(1):910-8. doi: 10.1093/jjco/hyz089. PMID: 31219161. S
- 274. Gabor S, Renner H, Matzi V, et al. Early enteral feeding compared with parenteral nutrition after oesophageal or oesophagogastric resection and reconstruction. British Journal of Nutrition. 2005;93(4):509-13. doi: 10.1079/bjn20041383. PMID: 43530411. C
- 275. Gade J, Levring T, Hillingso J, et al. The Effect of Preoperative Oral Immunonutrition on Complications and Length of Hospital Stay After Elective Surgery for Pancreatic Cancer--A Randomized Controlled Trial. Nutrition & Cancer. 2016;68(2):225-33. doi: 10.1080/01635581.2016.1142586. PMID: 26943500. SS
- 276. Gao JL, Gao W, Dou ZX. Effect of early postoperative enteral nutrition on nutrition status and immune function in gastric cancer patients. World Chinese Journal of Digestology. 2015;23(2):3451-5. doi: 10.11569/wcjd.v23.i21.3451. PMID: CN-01103076. PT

- 277. Gao J-X, Lu Y-P. Modified Sanxiang Xiaopi decoction improves gastrointestinal dysfunction in patients after surgery for colon cancer. World Chinese Journal of Digestology. 2017;25(1):1605-9. doi: 10.11569/wcjd.v25.i17.1605. PMID: CN-01411664. NE
- 278. Garcia-Peris P, Velasco C, Hernandez M, et al. Effect of inulin and fructo-oligosaccharide on the prevention of acute radiation enteritis in patients with gynecological cancer and impact on quality-of-life: a randomized, double-blind, placebo-controlled trial. European Journal of Clinical Nutrition. 2016;70(2):170-4. doi: 10.1038/ejcn.2015.192. PMID: 26603881. I
- 279. Ge Y. Effect of Standardized Nutritional Intervention in Patients with Nasopharyngeal Carcinoma Receiving Radiotherapy Complicated with Diabetes Mellitus. Dis Markers. 2022;2022:6704347. doi: 10.1155/2022/6704347. PMID: 35756497. I
- 280. Gencer A, Ozdemir Y, Sucullu I, et al. The effects of enteral immunonutrient products and total parenteral nutrition in patients who underwent major abdominal surgery. Trakya universitesi tip fakultesi dergisi. 2010;27(4):404-10. doi: 10.5174/tutfd.2009.02426.1. PMID: CN-01016713. NE
- 281. Georgakis GV, Eisenberg DP, Piorkowski RJ, et al. Effect of early enteral tube feeding on patient outcome following pancreaticoduodenectomy. Connecticut Medicine. 2012;76(4):213-8. PMID: 22611720. C
- 282. Ghaffar F, Javed N, Alam K. Effects of dietary modifications and nutritional interventions on the health status of breast cancer patients. Journal of Medical Sciences (Peshawar). 2016;24(4):251-7. PMID: 614337173. S
- 283. Ghisoni E, Casalone V, Giannone G, et al. Role of Mediterranean diet in preventing platinum based gastrointestinal toxicity in gynecolocological malignancies: A single Institution experience. World Journal of Clinical Oncology. 2019;10(1):391-401. doi: 10.5306/wjco.v10.i12.391. PMID: 31890648. S

- 284. Giger U, Buchler M, Farhadi J, et al. Preoperative immunonutrition suppresses perioperative inflammatory response in patients with major abdominal surgery - A randomized controlled pilot study. Annals of Surgical Oncology. 2007;14(1):2798-806. doi: 10.1245/s10434-007-9407-7. PMID: 47460623. SS
- 285. Giles KH, Kubrak C, Baracos VE, et al. Recommended European Society of Parenteral and Enteral Nutrition protein and energy intakes and weight loss in patients with head and neck cancer. Head & Neck. 2016;38(8):1248-57. doi: 10.1002/hed.24427. PMID: 27028732. I
- 286. Gillis C, Fenton TR, Sajobi TT, et al. Trimodal prehabilitation for colorectal surgery attenuates post-surgical losses in lean body mass: A pooled analysis of randomized controlled trials. Clinical Nutrition. 2019;38(3):1053-60. doi: 10.1016/j.clnu.2018.06.982. PMID: 30025745. I
- 287. Gillis C, Loiselle SE, Fiore JF, Jr., et al. Prehabilitation with Whey Protein Supplementation on Perioperative Functional Exercise Capacity in Patients Undergoing Colorectal Resection for Cancer: A Pilot Double-Blinded Randomized Placebo-Controlled Trial. Journal of the Academy of Nutrition & Dietetics. 2016;116(5):802-12. doi: 10.1016/j.jand.2015.06.007. PMID: 26208743. SS
- 288. Gioxari A, Tzanos D, Kostara C, et al. Mediterranean Diet Implementation to Protect against Advanced Lung Cancer Index (ALI) Rise: Study Design and Preliminary Results of a Randomised Controlled Trial. International Journal of Environmental Research & Public Health [Electronic Resource]. 2021;18(7):01. doi: 10.3390/ijerph18073700. PMID: 33916252. SS
- 289. Giralt J, Regadera JP, Verges R, et al. Effects of probiotic Lactobacillus casei DN-114 001 in prevention of radiation-induced diarrhea: results from multicenter, randomized, placebo-controlled nutritional trial. International Journal of Radiation Oncology, Biology, Physics. 2008;71(4):1213-9. doi: 10.1016/j.ijrobp.2007.11.009. PMID: 18243569. I

- 290. Girke J, Seipt C, Markowski A, et al. Quality of Life and Nutrition Condition of Patients Improve Under Home Parenteral Nutrition: An Exploratory Study. Nutrition in Clinical Practice. 2016;31(5):659-65. doi: 10.1177/0884533616637949. PMID: 27165116. C
- 291. Gómez Candela C, Castillo R, de Cos AI, et al. Effects of parenteral glutamine in patients submitted to bone marrow transplantation. Nutricion Hospitalaria. 2006;21(1):13-21. PMID: CN-00563617. NE
- 292. Gómez Sánchez MB, García Talavera Espín NV, Monedero Saiz T, et al. Evaluation of perioperative nutritional therapy in patients with gastrointestinal tract neoplasms. Nutricion Hospitalaria. 2011;26(5):1073-80. doi: 10.1590/s0212-16112011000500023. PMID: CN-00814246. NE
- 293. Gómez Sánchez MB, García-Talavera Espín NV, Sánchez Álvarez C, et al. Perioperative nutritional support in patients with colorectal neoplasms. Nutricion Hospitalaria. 2010;25(5):797-805. PMID: CN-00786125. NE
- 294. Gómez-Candela C, Villarino Sanz M, Horrisberger A, et al. Efficacy evaluation of an oral powder supplement enriched with eicosapentaenoic acid in cancer patients. Nutricion Hospitalaria. 2011;26(6):1385-93. doi: 10.1590/s0212-16112011000600028. PMID: CN-00864987. NE
- 295. Goncalves Dias MC, de Fatima Nunes Marucci M, Nadalin W, et al. Nutritional intervention improves the caloric and proteic ingestion of head and neck cancer patients under radiotherapy. Nutricion Hospitalaria. 2005;20(5):320-5. PMID: 16229399. O
- 296. Gontero P, Marra G, Soria F, et al. A randomized double-blind placebo controlled phase I-II study on clinical and molecular effects of dietary supplements in men with precancerous prostatic lesions. Chemoprevention or "chemopromotion"? Prostate. 2015;75(1):1177-86. doi: 10.1002/pros.22999. PMID: 25893930. P
- 297. Goodrose-Flores C, Schedin A, Nelander J, et al. High-protein compared with standard parenteral nutrition in palliative cancer care. BMJ supportive & palliative care.
 2020;25:25. doi: 10.1136/bmjspcare-2019-002139. PMID: 32451328. I

- 298. Grainger EM, Schwartz SJ, Wang S, et al. A combination of tomato and soy products for men with recurring prostate cancer and rising prostate specific antigen. Nutrition & Cancer. 2008;60(2):145-54. doi: 10.1080/01635580701621338. PMID: 18444145. P
- 299. Grau T, Ruiz de Adana JC, Zubillaga S, et al. Randomized study of two different fat emulsions in total parenteral nutrition of malnourished surgical patients;effect of infectious morbidity and mortality. Nutricion Hospitalaria. 2003;18(3):159-66. PMID: CN-00439507. NE
- 300. Grilo A, Santos CA, Fonseca J. Percutaneous endoscopic gastrostomy for nutritional palliation of upper esophageal cancer unsuitable for esophageal stenting. Arquivos de Gastroenterologia. 2012;49(3):227-31. doi: 10.1590/s0004-28032012000300012. PMID: 365766791. C
- 301. Gu YH, Cai H, Li YP, et al. Effects of glutamine on patients undergoing postoperative chemotherapy for gastric cancer. Chinese journal of clinical nutrition. 2006;14(6):369-73. PMID: CN-01019262. NE
- 302. Guertin MH, Robitaille K, Pelletier JF, et al. Effects of concentrated long-chain omega-3 polyunsaturated fatty acid supplementation before radical prostatectomy on prostate cancer proliferation, inflammation, and quality of life: study protocol for a phase IIb, randomized, double-blind, placebocontrolled trial. BMC Cancer. 2018;18(1):64. doi: 10.1186/s12885-017-3979-9. PMID: 29321047. PT
- 303. Guidera AK, Kelly BN, Rigby P, et al. Early oral intake after reconstruction with a free flap for cancer of the oral cavity. British Journal of Oral & Maxillofacial Surgery. 2013;51(3):224-7. doi: 10.1016/j.bjoms.2012.06.005. PMID: 22776518. I
- 304. Guieze R, Lemal R, Cabrespine A, et al. Enteral versus parenteral nutritional support in allogeneic haematopoietic stem-cell transplantation. Clinical Nutrition. 2014;33(3):533-8. doi: 10.1016/j.clnu.2013.07.012. PMID: 23938114. I

- 305. Guilbaud T, Birnbaum DJ, Loubiere S, et al. Comparison of different feeding regimes after pancreatoduodenectomy - a retrospective cohort analysis. Nutrition Journal. 2017;16(1):42. doi: 10.1186/s12937-017-0265-2. PMID: 28676052. S
- 306. Gunerhan Y, Koksal N, Sahin UY, et al. Effect of preoperative immunonutrition and other nutrition models on cellular immune parameters. World Journal of Gastroenterology. 2009;15(4):467-72. doi: 10.3748/wjg.15.467. PMID: 354133707. I
- 307. Gunn GB, Mendoza TR, Garden AS, et al. Minocycline for symptom reduction during radiation therapy for head and neck cancer: a randomized clinical trial. Supportive Care in Cancer. 2020;28(1):261-9. doi: 10.1007/s00520-019-04791-4. PMID: 31037378. I
- 308. Guo JC, Li J, Hu Y, et al. The role of perioperative enteral and parenteral nutrition treatment in pancreatic cancer: a multicenter, prospective randomized controlled trial. Zhonghua wai ke za zhi [Chinese journal of surgery].
 2013;51(1):987-90. PMID: CN-01118963. NE
- 309. Guo YM. Effects of emotional intervention and Chinese medicated diet in improving the quality of life in patients with liver cancer after chemotherapy. Chinese Journal of Clinical Rehabilitation. 2005;9(1):30-1. PMID: CN-00569461. NE
- 310. Hackshaw-McGeagh LE, Penfold C, Shingler E, et al. Phase II randomised control feasibility trial of a nutrition and physical activity intervention after radical prostatectomy for prostate cancer. BMJ Open. 2019;9(1):e029480. doi: 10.1136/bmjopen-2019-029480. PMID: 31699723. SS
- 311. Hagiwara S, Mori T, Tuchiya H, et al. Multidisciplinary nutritional support for autologous hematopoietic stem cell transplantation: a cost-benefit analysis. Nutrition. 2011;27(1):1112-7. doi: 10.1016/j.nut.2010.11.010. PMID: 21482071. I

- 312. Hallay J, Kovacs G, Kiss Sz S, et al. Changes in the nutritional state and immune-serological parameters of esophagectomized patients fed jejunaly with glutamine-poor and glutaminerich nutriments. Hepato-Gastroenterology. 2002;49(4):1555-9. PMID: 12397734. O
- 313. Hallay J, Micskei C, Fulesdi B, et al. Use of three lumen catheter facilitates bowel movement after pancreato-duodenectomy. Hepato-Gastroenterology. 2008;55(8):1099-102. PMID: 18705337. I
- 314. Hamaguchi R, Ito T, Narui R, et al. Effects of Alkalization Therapy on Chemotherapy Outcomes in Advanced Pancreatic Cancer: A Retrospective Case-Control Study. In Vivo. 2020;34(5):2623-9. doi: 10.21873/invivo.12080. PMID: 32871792. I
- 315. Hamilton-Reeves JM, Bechtel MD, Hand LK, et al. Effects of Immunonutrition for Cystectomy on Immune Response and Infection Rates: A Pilot Randomized Controlled Clinical Trial. European Urology. 2016;69(3):389-92. doi: 10.1016/j.eururo.2015.11.019. PMID: 26654125. SS
- 316. Han DJ, Yang Z, Zhao N, et al. Nutritional status and interventions of nasopharyngeal carcinoma patients treated with radiotherapy. Chinese journal of cancer prevention and treatment. 2013;20(1):786-9. PMID: CN-00875686. NE
- 317. Han H, Pan M, Tao Y, et al. Early Enteral Nutrition is Associated with Faster Post-Esophageatomy Recovery in Chinese Esophageal Cancer Patients: A Retrospective Cohort Study. Nutrition & Cancer. 2018;70(2):221-8. doi: 10.1080/01635581.2018.1412477. PMID: 29313724. I
- 318. Hanai N, Terada H, Hirakawa H, et al. Prospective randomized investigation implementing immunonutritional therapy using a nutritional supplement with a high blend ratio of omega-3 fatty acids during the perioperative period for head and neck carcinomas. Japanese Journal of Clinical Oncology. 2018;48(4):356-61. doi: 10.1093/jjco/hyy008. PMID: 29420749. SS

- 319. Harvie MN, Campbell IT, Howell A, et al. Acceptability and tolerance of a low tyrosine and phenylalanine diet in patients with advanced cancer - A pilot study. Journal of Human Nutrition and Dietetics. 2002;15(3):193-202. doi: 10.1046/j.1365-277X.2002.00365.x. PMID: 34661897. C
- 320. Haseen F, Murray LJ, O'Neill RF, et al. A randomised controlled trial to evaluate the efficacy of a 6 month dietary and physical activity intervention for prostate cancer patients receiving androgen deprivation therapy. Trials [Electronic Resource].
 2010;11. doi: 10.1186/1745-6215-11-86. PMID: 51034131. I
- 321. Hasenberg T, Essenbreis M, Herold A, et al. Early supplementation of parenteral nutrition is capable of improving quality of life, chemotherapy-related toxicity and body composition in patients with advanced colorectal carcinoma undergoing palliative treatment: results from a prospective, randomized clinical trial. Colorectal Disease. 2010;12(1):e190-9. doi: 10.1111/j.1463-1318.2009.02111.x. PMID: 19895595. OR
- 322. Hashemi-Khah MS, Arbab-Soleimani N, Forghanifard MM, et al. An In Vivo Study of Lactobacillus rhamnosus (PTCC 1637) as a New Therapeutic Candidate in Esophageal Cancer. Biomed Res Int. 2022;2022:7607470. doi: 10.1155/2022/7607470. PMID: 35782061. I
- 323. Hata H, Yamaguchi T, Hasegawa S, et al. Oral and Parenteral Versus Parenteral Antibiotic Prophylaxis in Elective Laparoscopic Colorectal Surgery (JMTO PREV 07-01): a Phase 3, Multicenter, Open-label, Randomized Trial. Annals of Surgery. 2016;263(6):1085-91. doi: 10.1097/sla.000000000001581. PMID: CN-01167296. I
- 324. Hatao F, Chen KY, Wu JM, et al. Randomized controlled clinical trial assessing the effects of oral nutritional supplements in postoperative gastric cancer patients. Langenbecks Archives of Surgery. 2017;402(2):203-11. doi: 10.1007/s00423-016-1527-8. PMID: 27807617. SS

- 325. Hausmann J, Kubesch A, Muller von der Grun J, et al. Prophylactic percutaneous endoscopic gastrostomy in patients with head and neck cancer: Influence on nutritional status, utilisation rate and complications. International Journal of Clinical Practice. 2019;73(1):e13405. doi: 10.1111/ijcp.13405. PMID: 31408231. I
- 326. He YF, Chen J, Wang G, et al. Randomizedly controlled trial on application of parenteral alanyl-glutamine dipeptide in chemotherapy of advanced patients with esophageal and cardiac carcinoma. Chinese journal of cancer prevention and treatment. 2008;15(1):936-8. PMID: CN-00708056. NE
- 327. Hebert JR, Ebbeling CB, Olendzki BC, et al. Change in women's diet and body mass following intensive intervention for earlystage breast cancer. Journal of the American Dietetic Association. 2001;101(4):421-31. PMID: 11320947. P
- 328. Heller AR, Rossel T, Gottschlich B, et al. Omega-3 fatty acids improve liver and pancreas function in postoperative cancer patients. International Journal of Cancer. 2004;111(4):611-6. doi: 10.1002/ijc.20291. PMID: 39045330. SS
- 329. Helminen H, Raitanen M, Kellosalo J. Immunonutrition in elective gastrointestinal surgery patients. Scandinavian Journal of Surgery. 2007;96(1):46-50. doi: 10.1177/145749690709600109. PMID: 46493764. P
- 330. Henderson VP, Massion AO, Clemow L, et al. A randomized controlled trial of mindfulness-based stress reduction for women with early-stage breast cancer receiving radiotherapy. Integrative Cancer Therapies. 2013;12(5):404-13. doi: 10.1177/1534735412473640. PMID: 23362338. O
- 331. Heng MS, Barbon Gauro J, Yaxley A, et al. Does a neutropenic diet reduce adverse outcomes in patients undergoing chemotherapy? European Journal of Cancer Care. 2020;29(1):e13155. doi: 10.1111/ecc.13155. PMID: 31441568. I

- 332. Henning SM, Galet C, Gollapudi K, et al. Phase II prospective randomized trial of weight loss prior to radical prostatectomy. Prostate Cancer & Prostatic Diseases.
 2018;21(2):212-20. doi: 10.1038/s41391-017-0001-1. PMID: 29203893. I
- 333. Henriksen MG, Hansen HV, Hessov I. Early oral nutrition after elective colorectal surgery: influence of balanced analgesia and enforced mobilization. Nutrition. 2002;18(3):263-7. PMID: 11882401. I
- 334. Hertlein L, Zeder-Gos C, Furst S, et al. Perioperative oral immunonutrition in malnourished ovarian cancer patients assessed by the nutritional risk screening. Archives of Gynecology & Obstetrics. 2018;297(6):1533-8. doi: 10.1007/s00404-018-4759-8. PMID: 29623417. I
- 335. Heymach JV, Shackleford TJ, Tran HT, et al. Effect of low-fat diets on plasma levels of NF-kappaB-regulated inflammatory cytokines and angiogenic factors in men with prostate cancer. Cancer Prevention Research. 2011;4(1):1590-8. doi: 10.1158/1940-6207.Capr-10-0136. PMID: 21764858. O
- 336. Hirao M, Tsujinaka T, Takeno A, et al. Patientcontrolled dietary schedule improves clinical outcome after gastrectomy for gastric cancer. World Journal of Surgery. 2005;29(7):853-7. doi: 10.1007/s00268-005-7760-x. PMID: 41800714. S
- 337. Ho M, Ho JWC, Fong DYT, et al. Effects of dietary and physical activity interventions on generic and cancer-specific health-related quality of life, anxiety, and depression in colorectal cancer survivors: a randomized controlled trial. Journal of Cancer Survivorship. 2020;14(4):424-33. doi: 10.1007/s11764-020-00864-0. PMID: 32072434. P
- 338. Ho YW, Yeh KY, Hsueh SW, et al. Impact of early nutrition counseling in head and neck cancer patients with normal nutritional status. Supportive Care in Cancer. 2021;29(5):2777-85. doi: 10.1007/s00520-020-05804-3. PMID: 32995998. I

- 339. Hofto S, Abbott J, Jackson JE, et al. Investigating adherence to Australian nutritional care guidelines in patients with head and neck cancer. Cancers of the Head & Neck. 2018;3:6. doi: 10.1186/s41199-018-0033-9. PMID: 31093359. C
- 340. Homkham N, Muangwong P, Pisprasert V, et al. Dynamic changes in practical inflammation and immunity markers in cancer patients receiving immune-enhancing nutritional supplementation during concurrent chemoradiotherapy. Cancer Biomarkers: Section A of Disease Markers. 2021;32(3):281-91. doi: 10.3233/cbm-210086. PMID: 34151843. O
- 341. Hong L, Han Y, Zhang H, et al. Effect of early oral feeding on short-term outcome of patients receiving laparoscopic distal gastrectomy: a retrospective cohort study. International Journal Of Surgery. 2014;12(7):637-9. doi: 10.1016/j.ijsu.2014.05.062. PMID: 24859351. S
- 342. Hopanci Bicakli D, Ozkaya Akagunduz O, Meseri Dalak R, et al. The Effects of Compliance with Nutritional Counselling on Body Composition Parameters in Head and Neck Cancer Patients under Radiotherapy. Journal of Nutrition and Metabolism. 2017;2017:8631945. doi: 10.1155/2017/8631945. PMID: 28116152. S
- 343. Horvat M, Krebs B, Potrc S, et al. Preoperative synbiotic bowel conditioning for elective colorectal surgery. Wiener Klinische Wochenschrift. 2010;122:26-30. doi: 10.1007/s00508-010-1347-8. PMID: 20517667. I
- 344. Hotta T, Kobayashi Y, Taniguchi K, et al. Evaluation of postoperative nutritional state after hepatectomy for hepatocellular carcinoma. Hepato-Gastroenterology. 2003;50(5):1511-6. PMID: 14571775. S
- 345. Hsu TC, Su CF, Huang PC, et al. Comparison of tolerance and change of intragastric pH between early nasogastric and nasojejunal feeding following resection of colorectal cancer. Clinical nutrition (Edinburgh, Scotland). 2006;25(4):681-6. doi: 10.1016/j.clnu.2005.12.011. PMID: CN-00570542. S

- 346. Hu Y, Ma Y, Wang J, et al. Early enteral infusion of traditional Chinese medicine preparation can effectively promote the recovery of gastrointestinal function after esophageal cancer surgery. Journal of Thoracic Disease. 2011;3(4):249-54. doi: 10.3978/j.issn.2072-1439.2011.09.08. PMID: 22263099. O
- 347. Huang CH, Hsieh TC, Chang BS, et al. Enteral Access Potentially Endangers Esophageal Carcinoma Patients Under Multi-modality Therapy: A Population-based Study. Anticancer Research. 2019;39(4):2227-32. doi: 10.21873/anticanres.13338. PMID: 30952771. I
- 348. Huang CH, Wang TF, Wu YF, et al. Efficacy of Enteral Access in Patients with Esophageal Squamous Cell Carcinoma Under Neoadjuvant Therapy. Anticancer Research. 2018;38(1):6939-45. doi: 10.21873/anticanres.13072. PMID: 30504413. S
- 349. Huang H, Fang F, Jia Z, et al. Influences of Oral Administration of Probiotics on Posthepatectomy Recovery in Patients in Child-Pugh Grade. Comput. 2022;2022:2942982. doi: 10.1155/2022/2942982. PMID: 35844449. I
- 350. Huang K, Wu B, Ding X, et al. Postesophagectomy tube feeding: a retrospective comparison of jejunostomy and a novel gastrostomy feeding approach. PLoS ONE [Electronic Resource]. 2014;9(3):e89190. doi: 10.1371/journal.pone.0089190. PMID: 24658763. S
- 351. Huang S, Piao Y, Cao C, et al. A prospective randomized controlled trial on the value of prophylactic oral nutritional supplementation in locally advanced nasopharyngeal carcinoma patients receiving chemo-radiotherapy. Oral Oncology. 2020;111:105025. doi: 10.1016/j.oraloncology.2020.105025. PMID: 33032180. SS
- 352. Huang X, Yuan S, Chen Q, et al. The clinical significance of preoperative administration of enteral immunonutrition to patients with malignant gastrointestinal tumors. Pharmaceutical Care and Research. 2010;10(6):450-2. doi: 10.5428/pcar20100616. PMID: 361168134. NE

- 353. Huang ZJ, Chen BS, You J, et al. The clinical significance of preoperative enteral immune nutrition in patients with malignant gastrointestinal tumors. Sichuan da xue xue bao. Yi xue ban [Journal of Sichuan University. Medical science edition].
 2014;45(1):167-70. PMID: CN-01097828. NE
- 354. Huggins C, Hanna L, Furness K, et al. Effect of dietetic counselling delivered via telephone or mHealth on quality of life in people with upper gastrointestinal cancer: a 3-arm randomised controlled trial. Asia-Pacific journal of clinical oncology. 2021;17(S):184-. doi: 10.1111/ajco.13716. PMID: CN-02356651. PT
- 355. Hung YC, Bauer JD, Horsely P, et al. Telephone-delivered nutrition and exercise counselling after auto-SCT: A pilot, randomised controlled trial. Bone Marrow Transplantation. 2014;49(6):786-92. doi: 10.1038/bmt.2014.52. PMID: 53090508. I
- 356. Hur H, Kim SG, Shim JH, et al. Effect of early oral feeding after gastric cancer surgery: a result of randomized clinical trial. Surgery. 2011;149(4):561-8. doi: 10.1016/j.surg.2010.10.003. PMID: 21146844. I
- 357. Hur H, Si Y, Kang WK, et al. Effects of early oral feeding on surgical outcomes and recovery after curative surgery for gastric cancer: pilot study results. World Journal of Surgery. 2009;33(7):1454-8. doi: 10.1007/s00268-009-0009-3. PMID: 19399550. S
- 358. Hyeda A, Costa E. Economic analysis of costs with enteral and parenteral nutritional therapy according to disease and outcome. Einstein. 2017;15(2):192-9. doi: 10.1590/s1679-45082017gs4002. PMID: 28767918. P
- 359. Isenring E, Capra S, Bauer J, et al. The impact of nutrition support on body composition in cancer outpatients receiving radiotherapy. Acta Diabetologica. 2003;40:S162-4. doi: 10.1007/s00592-003-0054-6. PMID: CN-00559339. SS
- 360. Ishibe A, Ota M, Kanazawa A, et al. Nutritional management of anastomotic leakage after colorectal cancer surgery using elemental diet jelly. Hepato-Gastroenterology. 2015;62(1):30-3. PMID: 25911862. C

- 361. Ishikawa S, Kitabatake K, Edamatsu K, et al. Evaluation of a Semi-Solidifying Liquid Formula for Nasogastric Tube Feeding After Oral and Maxillofacial Surgery. Journal of Oral and Maxillofacial Surgery. 2020;78(4):663.e1-.e7. doi: 10.1016/j.joms.2019.11.025. PMID: 2004512371. P
- 362. Ishikawa Y, Yoshida H, Mamada Y, et al. Prospective randomized controlled study of short-term perioperative oral nutrition with branched chain amino acids in patients undergoing liver surgery. Hepato-Gastroenterology. 2010;57(9):583-90. PMID: 20698232. SS
- 363. Ishiki H, Iwase S, Gyoda Y, et al. Oral nutritional support can shorten the duration of parenteral hydration in end-of-life cancer patients: a randomized controlled trial. Nutrition & Cancer. 2015;67(1):105-11. doi: 10.1080/01635581.2015.976312. PMID: 25437180. P
- 364. Iwase R, Suzuki Y, Yamanouchi E, et al. Double percutaneous transesophageal gastrotubing for gastric cancer: a pilot study. Journal of Surgical Research. 2018;232:470-4. doi: 10.1016/j.jss.2018.05.041. PMID: 2000973558. C
- 365. Iyikesici MS. Long-Term Survival Outcomes of Metabolically Supported Chemotherapy with Gemcitabine-Based or FOLFIRINOX Regimen Combined with Ketogenic Diet, Hyperthermia, and Hyperbaric Oxygen Therapy in Metastatic Pancreatic Cancer. Complementary Medical Research. 2020;27(1):31-9. doi: 10.1159/000502135. PMID: 31527373. S
- 366. Iyikesici MS. Survival outcomes of metabolically supported chemotherapy combined with ketogenic diet, hyperthermia, and hyperbaric oxygen therapy in advanced gastric cancer. Nigerian Journal of Clinical Practice. 2020;23(5):734-40. doi: 10.4103/njcp.njcp_509_18. PMID: 32367884. S
- 367. Jacot W, Arnaud A, Jarlier M, et al. Brief Hospital Supervision of Exercise and Diet During Adjuvant Breast Cancer Therapy Is Not Enough to Relieve Fatigue: A Multicenter Randomized Controlled Trial. Nutrients. 2020;12(1):09. doi: 10.3390/nu12103081. PMID: 33050321. I

- 368. Jahn P, Renz P, Stukenkemper J, et al. Reduction of chemotherapy-induced anorexia, nausea, and emesis through a structured nursing intervention: a clusterrandomized multicenter trial. Supportive Care in Cancer. 2009;17(1):1543-52. doi: 10.1007/s00520-009-0698-z. PMID: 19629539. I
- 369. Jain R, Shaikh T, Yee JL, et al. Impact of Clinical Markers of Nutritional Status and Feeding Jejunostomy Use on Outcomes in Esophageal Cancer Patients Undergoing Neoadjuvant Chemoradiotherapy. Nutrients. 2020;12(1):17. doi: 10.3390/nu12103177. PMID: 33080840. I
- 370. Jang A, Jeong O. Early Postoperative Oral Feeding After Total Gastrectomy in Gastric Carcinoma Patients: A Retrospective Before-After Study Using Propensity Score Matching. Jpen: Journal of Parenteral & Enteral Nutrition. 2019;43(5):649-57. doi: 10.1002/jpen.1438. PMID: 30144113. S
- 371. Jarden M, Baadsgaard MT, Hovgaard DJ, et al. A randomized trial on the effect of a multimodal intervention on physical capacity, functional performance and quality of life in adult patients undergoing allogeneic SCT. Bone Marrow Transplantation. 2009;43(9):725-37. doi: 10.1038/bmt.2009.27. PMID: 19234513. I
- 372. Jeong O, Ryu SY, Jung MR, et al. The safety and feasibility of early postoperative oral nutrition on the first postoperative day after gastrectomy for gastric carcinoma. Gastric Cancer. 2014;17(2):324-31. doi: 10.1007/s10120-013-0275-5. PMID: 23771588. I
- 373. Jiang Q. Effect of Glutamine's nutrition support on the postoperative nutrition and immune function in malignant tumor of gynecology patients. Pakistan Journal of Pharmaceutical Sciences. 2014;27(4):1023-7. PMID: 25016261. O
- 374. Jiang YJ, Kong XJ, Cheng G, et al. Effects of reasonable preoperative nutrition on recoveries of gastrointestinal cancer patients. World Chinese Journal of Digestology. 2006;14(1):1928-32. PMID: CN-00613178. NE

- 375. Jiang YJ, Kong XJ, Tian ZB, et al. Comparative studies on application of combination of postoperative enteral nutrition and parenteral nutrition with total parenteral nutrition after gastrointestinal surgery. Chinese journal of clinical nutrition. 2006;14(1):18-21. PMID: CN-00622742. NE
- 376. Jin T, Li KX, Li PJ, et al. An evaluation of nutrition intervention during radiation therapy in patients with locoregionally advanced nasopharyngeal carcinoma. Oncotarget. 2017;8(4):83723-33. doi: 10.18632/oncotarget.19381. PMID: 29137377. S
- 377. Jochum SB, Ritz EM, Bhama AR, et al. Early feeding in colorectal surgery patients: safe and cost effective. International Journal of Colorectal Disease. 2020;35(3):465-9. doi: 10.1007/s00384-019-03500-1. PMID: 2004010853. I
- 378. Joliat GR, Martin D, Labgaa I, et al. Early enteral vs. oral nutrition after Whipple procedure: Study protocol for a multicentric randomized controlled trial (NUTRIWHI trial). Front Oncol. 2022;12:855784. doi: 10.3389/fonc.2022.855784. PMID: 35865476. PT
- 379. Kabarriti R, Bontempo A, Romano M, et al. The impact of dietary regimen compliance on outcomes for HNSCC patients treated with radiation therapy. Supportive Care in Cancer. 2018;26(9):3307-13. doi: 10.1007/s00520-018-4198-x. PMID: 29671062. I
- 380. Kadam A, Bendale Y, Birari-Gawande P. Addressing and targeting earnest condition of advance breast cancer-related anorexia and cachexia through Rasayana therapy. Journal of Cancer Research & Therapeutics. 2020;16(6):1210-4. doi: 10.4103/jcrt.JCRT_96_20. PMID: 33342775. C
- 381. Kadota H, Fukushima J, Nakashima T, et al. Comparison of salvage and planned pharyngolaryngectomy with jejunal transfer for hypopharyngeal carcinoma after chemoradiotherapy. Laryngoscope. 2010;120(6):1103-8. doi: 10.1002/lary.20887. PMID: 20513024. I

- 382. Kaegi-Braun N, Schuetz P, Mueller B, et al. Association of Nutritional Support With Clinical Outcomes in Malnourished Cancer Patients: A Population-Based Matched Cohort Study. Frontiers in Nutrition. 2020;7:603370. doi: 10.3389/fnut.2020.603370. PMID: 33777987. S
- 383. Kaidarova DR, Kopp MV, Pokrovsky VS, et al. Multicomponent nutritional supplement Oncoxin and its influence on quality of life and therapy toxicity in patients receiving adjuvant chemotherapy. Oncology Letters. 2019;18(5):5644-52. doi: 10.3892/ol.2019.10868. PMID: 31641390. S
- 384. Kakuta T, Kosugi SI, Ichikawa H, et al. Palliative interventions for patients with incurable locally advanced or metastatic thoracic esophageal carcinoma. Esophagus. 2019;16(3):278-84. doi: 10.1007/s10388-019-00665-0. PMID: 30949884. P
- 385. Kamei H, Hachisuka T, Nakao M, et al. Quick recovery of serum diamine oxidase activity in patients undergoing total gastrectomy by oral enteral nutrition. American Journal of Surgery. 2005;189(1):38-43. doi: 10.1016/j.amjsurg.2004.03.015. PMID: 40215502. SS
- 386. Kanekiyo S, Takeda S, Iida M, et al. Efficacy of perioperative immunonutrition in esophageal cancer patients undergoing esophagectomy. Nutrition. 2019;59:96-102. doi: 10.1016/j.nut.2018.08.006. PMID: 30468936. SS
- 387. Kang SH, Lee Y, Min SH, et al. Multimodal Enhanced Recovery After Surgery (ERAS) Program is the Optimal Perioperative Care in Patients Undergoing Totally Laparoscopic Distal Gastrectomy for Gastric Cancer: A Prospective, Randomized, Clinical Trial. Annals of Surgical Oncology. 2018;25(1):3231-8. doi: 10.1245/s10434-018-6625-0. PMID: 30051365. I
- 388. Kang WX, Li W, Huang SG, et al. Effects of nutritional intervention in head and neck cancer patients undergoing radiotherapy: A prospective randomized clinical trial. Molecular and Clinical Oncology. 2016;5(3):279-82. doi: 10.3892/mco.2016.943. PMID: 611401262. SS

- 389. Kapoor N, Naufahu J, Tewfik S, et al. A Prospective Randomized Controlled Trial to Study the Impact of a Nutrition-Sensitive Intervention on Adult Women With Cancer Cachexia Undergoing Palliative Care in India. Integrative Cancer Therapies. 2017;16(1):74-84. doi: 10.1177/1534735416651968. PMID: 27252077. SS
- 390. Kato K, Omatsu K, Okamoto S, et al. Early oral feeding is safe and useful after rectosigmoid resection with anastomosis during cytoreductive surgery for primary ovarian cancer. World Journal of Surgical Oncology. 2021;19(1):77. doi: 10.1186/s12957-021-02186-6. PMID: 33722264. I
- 391. Kazmierczak-Siedlecka K, Folwarski M, Ruszkowski J, et al. Effects of 4 weeks of Lactobacillus plantarum 299v supplementation on nutritional status, enteral nutrition tolerance, and quality of life in cancer patients receiving home enteral nutrition - a double-blind, randomized, and placebo-controlled trial. European Review for Medical & Pharmacological Sciences. 2020;24(1):9684-94. doi: 10.26355/eurrev_202009_23059. PMID: 33015813. I
- 392. Khodabakhshi A, Akbari ME, Mirzaei HR, et al. Effects of Ketogenic metabolic therapy on patients with breast cancer: A randomized controlled clinical trial. Clinical Nutrition. 2021;40(3):751-8. doi: 10.1016/j.clnu.2020.06.028. PMID: 2007125008. O
- 393. Kikuchi S, Takata N, Kuroda S, et al. Impact of Amino Acids Nutrition Following Gastrectomy in Gastric Cancer Patients. Anticancer Res. 2022 Jul;42(7):3637-43. doi: 10.21873/anticanres.15852. PMID: 35790296. PT
- 394. Kim H, Suh EE, Lee HJ, et al. The effects of patient participation-based dietary intervention on nutritional and functional status for patients with gastrectomy: a randomized controlled trial. Cancer Nursing. 2014;37(2):E10-20. doi: 10.1097/NCC.0b013e31829193c8. PMID: 23632471. SS

- 395. Kim HU, Chung JB, Kim CB. The comparison between early enteral nutrition and total parenteral nutrition after total gastrectomy in patients with gastric cancer: the randomized prospective study. Taehan Sohwagi Hakhoe chi [The Korean journal of gastroenterology]. 2012;59(6):407-13. doi: 10.4166/kjg.2012.59.6.407. PMID: CN-00879845. NE
- 396. Kim J, Min YW, Lee H, et al. Comparative Study of Esophageal Self-expandable Metallic Stent Insertion and Gastrostomy Feeding for Dysphagia Caused by Lung Cancer. Korean Journal of Gastroenterology/Taehan Sohwagi Hakhoe Chi. 2018;71(3):124-31. doi: 10.4166/kjg.2018.71.3.124. PMID: 29566473. 1
- 397. Kim JM, Hong SG, Song BS, et al. Efficacy of Cereal-based Oral Nutrition Supplement on Nutritional Status, Inflammatory Cytokine Secretion and Quality of Life in Cancer Patients Under Cancer Therapy. Journal of Cancer Prevention. 2020;25(1):55-63. doi: 10.15430/jcp.2020.25.1.55. PMID: 32266180. S
- 398. Kim SH, Lee SM, Jeung HC, et al. The Effect of Nutrition Intervention with Oral Nutritional Supplements on Pancreatic and Bile Duct Cancer Patients Undergoing Chemotherapy. Nutrients. 2019;11(5):22. doi: 10.3390/nu11051145. PMID: 31121926. SS
- 399. Kiss N, Isenring E, Gough K, et al. Early and Intensive Dietary Counseling in Lung Cancer Patients Receiving (Chemo)Radiotherapy-A Pilot Randomized Controlled Trial. Nutrition & Cancer. 2016;68(6):958-67. doi: 10.1080/01635581.2016.1188972. PMID: 27348253. SS
- 400. Kiss N, Seymour JF, Prince HM, et al. Challenges and outcomes of a randomized study of early nutrition support during autologous stem-cell transplantation. Current Oncology. 2014;21(2):e334-9. doi: 10.3747/co.21.1820. PMID: 24764716. SS
- 401. Kiss NK, Krishnasamy M, Loeliger J, et al. A dietitian-led clinic for patients receiving (chemo)radiotherapy for head and neck cancer. Supportive Care in Cancer. 2012;20(9):2111-20. doi: 10.1007/s00520-011-1321-7. PMID: 22086406. S

- 402. Kitagawa H, Namikawa T, Iwabu J, et al. Bowel obstruction associated with a feeding jejunostomy and its association to weight loss after thoracoscopic esophagectomy. BMC Gastroenterology. 2019;19(1):104. doi: 10.1186/s12876-019-1029-6. PMID: 31238878. I
- 403. Klement RJ, Champ CE, Kammerer U, et al. Impact of a ketogenic diet intervention during radiotherapy on body composition: III-final results of the KETOCOMP study for breast cancer patients. Breast Cancer Research. 2020;22(1):94. doi: 10.1186/s13058-020-01331-5. PMID: 32819413. S
- 404. Klement RJ, Koebrunner PS, Krage K, et al. Short-term effects of a Paleolithic lifestyle intervention in breast cancer patients undergoing radiotherapy: a pilot and feasibility study. Medical Oncology. 2021;38(1). doi: 10.1007/s12032-020-01443-0. PMID: 2007463132. I
- 405. Klement RJ, Schafer G, Sweeney RA. A ketogenic diet exerts beneficial effects on body composition of cancer patients during radiotherapy: An interim analysis of the KETOCOMP study. Journal of Traditional & Complementary Medicine. 2020;10(3):180-7. doi: 10.1016/j.jtcme.2019.03.007. PMID: 32670812. I
- 406. Klement RJ, Weigel MM, Sweeney RA. A ketogenic diet consumed during radiotherapy improves several aspects of quality of life and metabolic health in women with breast cancer. Clinical Nutrition. 2021;27:27. doi: 10.1016/j.clnu.2021.01.023. PMID: 33551218. I
- 407. Klevebro F, Johar A, Lagergren J, et al. Outcomes of nutritional jejunostomy in the curative treatment of esophageal cancer. Diseases of the Esophagus. 2019;32(7):01. doi: 10.1093/dote/doy113. PMID: 30496419. I
- 408. Knight SR, Qureshi AU, Drake TM, et al. The impact of preoperative oral nutrition supplementation on outcomes in patients undergoing gastrointestinal surgery for cancer in low- and middle-income countries: a systematic review and meta-analysis. Sci. 2022 Jul 21;12(1):12456. doi: 10.1038/s41598-022-16460-4. PMID: 35864290. PT
- 409. Kobayashi D, Ishigure K, Mochizuki Y, et al. Multi-institutional prospective feasibility study to explore tolerability and efficacy of oral nutritional supplements for patients with gastric cancer undergoing gastrectomy (CCOG1301). Gastric Cancer. 2017;20(4):718-27. doi: 10.1007/s10120-016-0668-3. PMID: 27885538. S
- 410. Kobayashi K, Kaneko J, Yamaguchi T, et al. Late-Evening Carbohydrate and Branched-Chain Amino Acid Snacks Improve the Nutritional Status of Patients Undergoing Hepatectomy Based on Bioelectrical Impedance Analysis of Body Composition. Gastrointestinal Tumors. 2019;6(3):81-91. doi: 10.1159/000501452. PMID: 31768352. I
- 411. Kobayashi K, Koyama Y, Kosugi S, et al. Is early enteral nutrition better for postoperative course in esophageal cancer patients? Nutrients. 2013;5(9):3461-9. doi: 10.3390/nu5093461. PMID: 24067386. I
- 412. Koemans WJ, Houwink A, van der Kaaij RT, et al. Perioperative Management of Gastric Cancer Patients Treated With (Sub)Total Gastrectomy, Cytoreductive Surgery, and Hyperthermic Intraperitoneal Chemotherapy (HIPEC): Lessons Learned. Annals of Surgical Oncology. 2021. doi: 10.1245/s10434-020-09465-8. PMID: 2007717668. I
- 413. Kono M, Wakisaka R, Kumai T, et al. Effects of early nutritional intervention by a nutritional support team for patients with head and neck cancer undergoing chemoradiotherapy or radiotherapy. Head and Neck. 2021;43(2):514-9. doi: 10.1002/hed.26502. PMID: 2006876105. S

- 414. Konosu M, Iwaya T, Kimura Y, et al. Peripheral vein infusions of amino acids facilitate recovery after esophagectomy for esophageal cancer: Retrospective cohort analysis. Annals of Medicine & Surgery. 2017;14:29-35. doi: 10.1016/j.amsu.2017.01.016. PMID: 28138387. I
- 415. Koppold-Liebscher D, Kessler CS, Steckhan N, et al. Short-term fasting accompanying chemotherapy as a supportive therapy in gynecological cancer: protocol for a multicenter randomized controlled clinical trial. Trials [Electronic Resource].
 2020;21(1):854. doi: 10.1186/s13063-020-04700-9. PMID: 33059765. PT
- 416. Koterazawa Y, Oshikiri T, Hasegawa H, et al. Routine placement of feeding jejunostomy tube during esophagectomy increases postoperative complications and does not improve postoperative malnutrition. Diseases of the Esophagus. 2020;33(1):16. doi: 10.1093/dote/doz021. PMID: 30997494. I
- 417. Kotzampassi K, Stavrou G, Damoraki G, et al. A Four-Probiotics Regimen Reduces Postoperative Complications After Colorectal Surgery: A Randomized, Double-Blind, Placebo-Controlled Study. World Journal of Surgery. 2015;39(1):2776-83. doi: 10.1007/s00268-015-3071-z. PMID: 25894405. I
- 418. Koyama Y, Moro K, Nakano M, et al. Intravenous Carnitine Administration in Addition to Parenteral Nutrition With Lipid Emulsion May Decrease the Inflammatory Reaction in Postoperative Surgical Patients. Journal of Clinical Medicine Research. 2017;9(1):831-7. doi: 10.14740/jocmr3113w. PMID: 28912919. SS
- 419. Kramer S, Newcomb M, Hessler J, et al. Prophylactic versus reactive PEG tube placement in head and neck cancer. Otolaryngology - Head and Neck Surgery (United States). 2014;150(3):407-12. doi: 10.1177/0194599813517081. PMID: 372497591. S

- 420. Kranse R, Dagnelie PC, Van Kemenade MC, et al. Dietary intervention in prostate cancer patients: PSA response in a randomized double-blind placebo-controlled study. International Journal of Cancer. 2005;113(5):835-40. doi: 10.1002/ijc.20653. PMID: 40101267. P
- 421. Kubota K, Kuroda J, Yoshida M, et al. Preoperative oral supplementation support in patients with esophageal cancer. Journal of Nutrition, Health & Aging. 2014;18(4):437-40. doi: 10.1007/s12603-014-0018-2. PMID: 24676327. I
- 422. Kufeldt J, Viehrig M, Schweikert D, et al. Treatment of malnutrition decreases complication rates and shortens the length of hospital stays in a radiation oncology department. Strahlentherapie und Onkologie. 2018;194(1):1049-59. doi: 10.1007/s00066-018-1360-9. PMID: 30182247. I
- 423. Kuroda H, Ushio A, Miyamoto Y, et al. Effects of branched-chain amino acid-enriched nutrient for patients with hepatocellular carcinoma following radiofrequency ablation: a one-year prospective trial. Journal of Gastroenterology & Hepatology. 2010;25(9):1550-5. doi: 10.1111/j.1440-1746.2010.06306.x. PMID: 20796154. I
- 424. Kutz LM, Abel J, Schweizer D, et al. Quality of life, HPV-status and phase angle predict survival in head and neck cancer patients under (chemo)radiotherapy undergoing nutritional intervention: Results from the prospective randomized HEADNUT-trial. Radiotherapy & Oncology. 2022;166:145-53. doi: 10.1016/j.radonc.2021.11.011. PMID: 34838889. PT
- 425. Lai JM, Wang ST, Li W, et al. Effects of ω-3 fish oil emulsion on inflammation and coagulation function of hepatocellular carcinoma patients after partial hepatectomy. Chinese journal of clinical nutrition. 2012;20(5):269-73. doi: 10.3760/cma.j.issn.1674-635X.2012.05.002. PMID: CN-01763539. O
- 426. Langmore S, Krisciunas GP, Miloro KV, et al. Does PEG use cause dysphagia in head and neck cancer patients? Dysphagia. 2012;27(2):251-9. doi: 10.1007/s00455-011-9360-2. PMID: 21850606. I

- 427. Laviano A, Calder PC, Schols A, et al. Safety and Tolerability of Targeted Medical Nutrition for Cachexia in Non-Small-Cell Lung Cancer: A Randomized, Double-Blind, Controlled Pilot Trial. Nutrition & Cancer. 2020;72(3):439-50. doi: 10.1080/01635581.2019.1634746. PMID: 31290697. SS
- 428. Lee H, Havrila C, Bravo V, et al. Effect of oral nutritional supplementation on weight loss and percutaneous endoscopic gastrostomy tube rates in patients treated with radiotherapy for oropharyngeal carcinoma. Supportive Care in Cancer. 2008;16(3):285-9. doi: 10.1007/s00520-007-0313-0. PMID: 351422801. S
- 429. Lee J, Kwon CH, Kim JM, et al. Effect of early enteral nutrition after hepatectomy in hepatocellular carcinoma patients. Korean Journal of Hepatobiliarypancreatic Surgery. 2012;16(4):129-33. doi: 10.14701/kjhbps.2012.16.4.129. PMID: 26388922. I
- 430. Lee MK, Yun YH, Park HA, et al. A Webbased self-management exercise and diet intervention for breast cancer survivors: pilot randomized controlled trial. International Journal of Nursing Studies. 2014;51(1):1557-67. doi: 10.1016/j.ijnurstu.2014.04.012. PMID: 24856854. P
- 431. Levine ME, Gillis MG, Koch SY, et al. Protein and ginger for the treatment of chemotherapy-induced delayed nausea. Journal of Alternative and Complementary Medicine. 2008;14(5):545-51. doi: 10.1089/acm.2007.0817. PMID: 351873389. I
- 432. Lewis SL, Brody R, Touger-Decker R, et al. Feeding tube use in patients with head and neck cancer. Head & Neck. 2014;36(1):1789-95. doi: 10.1002/hed.23538. PMID: 24478227. I
- 433. Li B, Li P, Gong F, et al. Comparative clinical study of nasojejunal and jejunostomy tube implants. Chinese journal of clinical oncology. 2014;41(1):1163-5. doi: 10.3969/j.issn.1000-8179.20141363. PMID: CN-01072258. PT

- 434. Li C, Bian L, Lian L. Nursing Effect and Prognosis of Perioperative Nutritional Support Therapy for Severe Malnutrition Patients with Colorectal Cancer. Anti-Tumor Pharmacy. 2021;11(2):240-3. doi: 10.3969/j.issn.2095-1264.2021.02.22. PMID: CN-02294081. NE
- 435. Li D, Sun CL, Kim H, et al. Geriatric Assessment-Driven Intervention (GAIN) on Chemotherapy-Related Toxic Effects in Older Adults With Cancer: A Randomized Clinical Trial. JAMA Oncology. 2021;7(1):e214158. doi: 10.1001/jamaoncol.2021.4158. PMID: 34591080. I
- 436. Li H, Feng S, Li Z, et al. Effects of overall enteral nutrition management overall on nutritional condition and life quality of nsclc patients treated by apatinib combined with chemoradiotherapy. International journal of clinical and experimental medicine. 2020;13(1):8782-9. PMID: 2005527884. S
- 437. Li H, Li H, Feng S, et al. Effects of overall enteral nutrition management overall on nutritional condition and life quality of nsclc patients treated by apatinib combined with chemoradiotherapy. International journal of clinical and experimental medicine. 2020;13(1):8782-9. PMID: CN-02288237. S
- 438. Li JM, Lai DN. Effect of compound branch chain amino acids injection (Aminic) on protein metabolism in patients with radical resection of cardiac carcinoma. Zhonghua wei chang wai ke za zhi [Chinese journal of gastrointestinal surgery]. 2005;8(2):137-40. PMID: CN-00721780. NE
- 439. Li K, Xu Y, Hu Y, et al. Effect of Enteral Immunonutrition on Immune, Inflammatory Markers and Nutritional Status in Gastric Cancer Patients Undergoing Gastrectomy: A Randomized Double-Blinded Controlled Trial. Journal of Investigative Surgery. 2020;33(1):950-9. doi: 10.1080/08941939.2019.1569736. PMID: 30885012. O
- 440. Li K, Yang J. The effect of different nutrition on the immune function of patients with colorectal cancer. Chinese journal of evidence-based medicine. 2009;9(1):1288-91. PMID: CN-00802967. NE

- 441. Li M, Tang L, Gao N, et al. Effects of enteral and parenteral nutrition support on nutritional status and adverse reactions of patients with lung cancer chemotherapy. Anti-Tumor Pharmacy. 2021;11(5):635-9. doi: 10.3969/j.issn.2095-1264.2021.05.20. PMID: CN-02342411. OR
- 442. Li NM, Liu F, Lv FY, et al. Influencing factors and interventional strategies for early enteral nutrition after gastric carcinoma surgery. Journal of Cancer Research & Therapeutics. 2016;12(2):689-92. doi: 10.4103/0973-1482.179179. PMID: 27461634. P
- 443. Li T, Lyu J, Zhu G, et al. Influence of enteral nutrition on nutritional status, treatment toxicities, and short-term outcomes in esophageal carcinoma patients treated with concurrent chemoradiotherapy: A prospective, multicenter, randomized controlled study. Annals of oncology : official journal of the European Society for Medical Oncology. 2018;29(S):viii207. doi: 10.1093/annonc/mdy282.004. PMID: 631202630. PT
- 444. Li Y, Liu Z, Liu G, et al. Impact on Short-Term Complications of Early Oral Feeding in Patients with Esophageal Cancer After Esophagectomy. Nutrition & Cancer. 2021;73(4):609-16. doi: 10.1080/01635581.2020.1769690. PMID: 32482102. I
- 445. Li Y, Xie H, Cai Z, et al. Effect of standardized process management of early enteral nutrition after operation on rapid recovery of elderly patients with gastric cancer. Chinese journal of clinical nutrition. 2019;27(6):361-6. doi: 10.3760/cma.j.issn.1674-635X.2019.06.006. PMID: CN-02095454. NE
- 446. Li YL, Ma L, Fan XQ. Clinical effects of early standardized enteral nutrition through a nasogastric tube combined with effective monitoring in patients with gastric cancer. World Chinese Journal of Digestology. 2015;23(2):3446-50. doi: 10.11569/wcjd.v23.i21.3446. PMID: CN-01084935. NE

- 447. Li Z, Aronson WJ, Arteaga JR, et al. Feasibility of a low-fat/high-fiber diet intervention with soy supplementation in prostate cancer patients after prostatectomy. European Journal of Clinical Nutrition. 2008;62(4):526-36. doi: 10.1038/sj.ejcn.1602743. PMID: 352117623. P
- 448. Liang B, Wang S, Ye YJ, et al. Impact of postoperative omega-3 fatty acidsupplemented parenteral nutrition on clinical outcomes and immunomodulationsi in colorectal cancer patients. World Journal of Gastroenterology. 2008;14(1):2434-9. doi: 10.3748/wjg.14.2434. PMID: 351877620. SS
- 449. Liang S, Xu L, Zhang D, et al. Effect of probiotics on small intestinal bacterial overgrowth in patients with gastric and colorectal cancer. Turkish Journal of Gastroenterology. 2016;27(3):227-32. doi: 10.5152/tjg.2016.15375. PMID: 27210778. I
- 450. Liang X, Ying H, Wang H, et al. Enhanced recovery care versus traditional care after laparoscopic liver resections: a randomized controlled trial. Surgical Endoscopy. 2018;32(6):2746-57. doi: 10.1007/s00464-017-5973-3. PMID: 29234943. I
- 451. Liang Y, Liu H, Nurse LZ, et al. Enhanced recovery after surgery for laparoscopic gastrectomy in gastric cancer: A prospective study. Medicine. 2021;100(7):e24267. doi: 10.1097/md.000000000024267. PMID: 33607765. I
- 452. Liao M, Xia Z, Huang P, et al. Early enteral feeding on esophageal cancer patients after esophageal resection and reconstruction. Annals of Palliative Medicine. 2020;9(3):816-23. doi: 10.21037/apm.2020.04.13. PMID: 32312065. SS
- 453. Liao Q, Zhao YP, Wang WB, et al. Perioperative nutrition support of the patients with pancreatic head cancer. Zhongguo yi xue ke xue yuan xue bao. Acta Academiae Medicinae Sinicae. 2005;27(5):579-82. PMID: CN-00575928. NE

- 454. Lidoriki I, Schizas D, Mylonas KS, et al. Oral Nutritional Supplementation Following Upper Gastrointestinal Cancer Surgery: A Prospective Analysis Exploring Potential Barriers to Compliance. Journal of the American College of Nutrition. 2020;39(7):650-6. doi: 10.1080/07315724.2020.1723453. PMID: 32017674. C
- 455. Liew AC, Peh KK, Tan BS, et al. Evaluation of chemotherapy-induced toxicity and healthrelated quality of life amongst early-stage breast cancer patients receiving Chinese herbal medicine in Malaysia. Supportive Care in Cancer. 2019;27(1):4515-24. doi: 10.1007/s00520-019-04724-1. PMID: 30911917. I
- 456. Ligthart-Melis GC, Weijs PJ, te Boveldt ND, et al. Dietician-delivered intensive nutritional support is associated with a decrease in severe postoperative complications after surgery in patients with esophageal cancer. Diseases of the Esophagus. 2013;26(6):587-93. doi: 10.1111/dote.12008. PMID: 23237356. SS
- 457. Lin CL, Perng CL, Chao Y, et al. Application of stent placement or nasojejunal feeding tube placement in patients with malignant gastric outlet obstruction: a retrospective series of 38 cases. Journal of the Chinese Medical Association: JCMA. 2012;75(1):624-9. doi: 10.1016/j.jcma.2012.08.013. PMID: 23245477. P
- 458. Lin DW, Neuhouser ML, Schenk JM, et al. Low-fat, low-glycemic load diet and gene expression in human prostate epithelium: A feasibility study of using cDNA microarrays to assess the response to dietary intervention in target tissues. Cancer Epidemiology Biomarkers and Prevention. 2007;16(1):2150-4. doi: 10.1158/1055-9965.Epi-07-0154. PMID: 47621193. SS
- 459. Lin HH, Lin YH, Hwang TZ, et al. The Effects of a Diet Education Program on Nutritional Status and Quality of Life in Oral Cancer Patients Who Underwent Surgery. Hu li za zhi [Journal of nursing]. 2020;67(1):33-43. doi: 10.6224/jn.202002_67(1).06. PMID: CN-02077757. NE

- 460. Lin ZH, Chuang VP, Soong TC, et al. Safety and effectiveness of percutaneous fluoroscopic gastrostomy in cancer patients. Journal of the Formosan Medical Association. 2000;99(3):206-12. PMID: 10820952. C
- 461. Lindman A, Rasmussen HB, Andersen NF. Food caregivers influence on nutritional intake among admitted haematological cancer patients - a prospective study. European Journal of Oncology Nursing. 2013;17(6):827-34. doi: 10.1016/j.ejon.2013.06.010. PMID: 24012191. S
- 462. Lindschinger M, Pamperl I, Anderhuber W, et al. Effect of an early percutaneous gastrostomy on the nutritional state, quality of life, anxiety and depression in patients with malignomas in the region of the larynx and pharynx. Aktuelle Ernahrungsmedizin. 2000;25(2):74-6. PMID: CN-00423981. NE
- 463. Link LB, Thompson SM, Bosland MC, et al. Adherence to a low-fat diet in men with prostate cancer. Urology. 2004;64(5):970-5. doi: 10.1016/j.urology.2004.06.040. PMID: 39469516. P
- 464. Linn YH, Thu KK, Win NHH. Effect of Probiotics for the Prevention of Acute Radiation-Induced Diarrhoea Among Cervical Cancer Patients: a Randomized Double-Blind Placebo-Controlled Study. Probiotics & Antimicrobial Proteins. 2019;11(2):638-47. doi: 10.1007/s12602-018-9408-9. PMID: 29550911. I
- 465. Lis CG, Cambron JA, Grutsch JF, et al. Self-reported quality of life in users and nonusers of dietary supplements in cancer. Supportive Care in Cancer. 2006;14(2):193-9. doi: 10.1007/s00520-005-0876-6. PMID: 43119617. I
- 466. Liu D, Lian H, Wang J, et al. Efficacy of early bundle therapy on moderate to severe aspiration pneumonia after cervical esophagogastrostomy for esophageal cancer. Minerva Medica. 2015;106(2):79-86. PMID: 25517501. I

- 467. Liu D, Shi G, Yin C, et al. Effect of Psychological Intervention Combined with Dietary Guidance on Quality of Life and Long-Term Efficacy of Bushen Quyu Decoction in Treatment of Patients with Advanced Ovarian Cancer. Evidence-Based Complementary & Alternative Medicine: eCAM. 2021;2021:1075513. doi: 10.1155/2021/1075513. PMID: 34733335. I
- 468. Liu FF, Wang LM, Rong WQ, et al. [Clinical effectiveness of postoperative nutritional support in patients undergoing hepatectomy for hepatocellular carcinoma].
 2018;40(1):787-92. doi: 10.3760/cma.j.issn.0253-3766.2018.10.012. PMID: 30392345. NE
- 469. Liu H, Ling W, Cao H. Effects of immuneenhanced enteral nutrition and parenteral nutrition on immune and nutritional function in elderly patients with gastric cancer after total gastrectomy. Journal of Shanghai Jiaotong University (Medical Science). 2011;31(7):1000-4. doi: 10.3969/j.issn.1674-8115.2011.07.029. PMID: 362231369. NE
- 470. Liu H, Ling W, Shen ZY, et al. Clinical application of immune-enhanced enteral nutrition in patients with advanced gastric cancer after total gastrectomy. Journal of Digestive Diseases. 2012;13(8):401-6. doi: 10.1111/j.1751-2980.2012.00596.x. PMID: 22788925. SS
- 471. Liu H, Wang B, Zhang J, et al. The effect of early postoperative enteral nutrition and parenteral nutrition in gastric cancer. Chinese journal of clinical oncology. 2014;41(1):1166-9. doi: 10.3969/j.issn.1000-8179.20141362. PMID: CN-01072257. NE
- 472. Liu J, Xia Q. Relationship of dietary fiber and early enteral nutrition with digestive complications after surgical treatment of gastric cancer. Zhonghua wei chang wai ke za zhi [Chinese journal of gastrointestinal surgery]. 2005;8(3):223-5. PMID: CN-00719359. NE
- 473. Liu JZ, Lan T, Zhang JS, et al. Use of postoperative enteral immunonutrition in malnutrition patients with gastrointestinal malignant tumor. Zhonghua wei chang wai ke za zhi [Chinese journal of gastrointestinal surgery]. 2011;14(1):799-802. PMID: CN-00919808. NE

- 474. Liu K, Ji S, Xu Y, et al. Safety, feasibility, and effect of an enhanced nutritional support pathway including extended preoperative and home enteral nutrition in patients undergoing enhanced recovery after esophagectomy: a pilot randomized clinical trial. Diseases of the Esophagus. 2020;33(2):05. doi: 10.1093/dote/doz030. PMID: 31329828. I
- 475. Liu MY, Tang HC, Hu SH, et al. Influence of preoperative peripheral parenteral nutrition with micronutrients after colorectal cancer patients. BioMed Research International. 2015;2015:535431. doi: 10.1155/2015/535431. PMID: 26000296. I
- 476. Liu Q, Yu YK, Wang DY, et al. Factors associated with the costs of hospitalization after esophagectomy: a retrospective observational study at a three-tertiary cancer hospital in China. Journal of Thoracic Disease. 2020;12(1):5970-9. doi: 10.21037/jtd-20-2770. PMID: 33209429. I
- 477. Liu SY, Chen XF, Wang F, et al. Study of different enteral nutrition formulation treatment after esophagectomy. Zhonghua wei chang wai ke za zhi [Chinese journal of gastrointestinal surgery]. 2012;15(5):473-5. PMID: CN-00979328. NE
- 478. Liu SY, Chen XF, Wang F, et al. Perioperative nutrition support for esophageal cancer complicated with diabetes mellitus. Zhonghua wei chang wai ke za zhi [Chinese journal of gastrointestinal surgery].
 2013;16(9):864-7. PMID: CN-01121382. NE
- 479. Liu Y, Jia Z, Dong L, et al. A randomized pilot study of atractylenolide I on gastric cancer cachexia patients. Evidence-Based Complementary & Alternative Medicine: eCAM. 2008;5(3):337-44. doi: 10.1093/ecam/nem031. PMID: 18830451. SS
- 480. Liu Y, Zhang X, Liu Z, et al. Effects of enteral nutrition rich in ω -3 polyunsaturated fatty acids on the nutritional status and quality of life of patients with gastric cancer. Chinese journal of clinical nutrition. 2020;28(3):151-7. doi: 10.3760/cma.j.cn115822-20200609-00145. PMID: CN-02265531. OR

- 481. Liu Z, Li C, Huang M, et al. Positive regulatory effects of perioperative probiotic treatment on postoperative liver complications after colorectal liver metastases surgery: a double-center and double-blind randomized clinical trial. BMC Gastroenterology. 2015;15:34. doi: 10.1186/s12876-015-0260-z. PMID: 25881090. I
- 482. Liu Z, Yu J. Effect of immune enhanced enteral nutrition on postoperative immune function and inflammatory responses in gastric cancer patients with radical gastrectomy. Chinese journal of cancer prevention and treatment. 2011;18(1):66-7+9. PMID: CN-00894136. O
- 483. Liu ZH, Huang MJ, Zhang XW, et al. The effects of perioperative probiotic treatment on serum zonulin concentration and subsequent postoperative infectious complications after colorectal cancer surgery: a double-center and double-blind randomized clinical trial. American Journal of Clinical Nutrition. 2013;97(1):117-26. doi: 10.3945/ajcn.112.040949. PMID: 23235200. I
- 484. Liu ZH, Su GQ, Zhang SY, et al. Study on early postoperative nutritional support in elderly patients with gastric cancer. Zhonghua wei chang wai ke za zhi [Chinese journal of gastrointestinal surgery]. 2013;16(1):1063-6. PMID: CN-01120114. NE
- 485. Llop-Talaveron JM, Badia-Tahull MB, Leiva-Badosa E, et al. Parenteral fish oil and liver function tests in hospitalized adult patients receiving parenteral nutrition: A propensity score-matched analysis. Clinical Nutrition. 2017;36(4):1082-8. doi: 10.1016/j.clnu.2016.06.027. PMID: 27435303. P
- 486. Lonbro S, Dalgas U, Primdahl H, et al. Feasibility and efficacy of progressive resistance training and dietary supplements in radiotherapy treated head and neck cancer patients--the DAHANCA 25A study. Acta Oncologica. 2013;52(2):310-8. doi: 10.3109/0284186x.2012.741325. PMID: 23190359. SS

- 487. Lopes LP, Menezes TM, Toledo DO, et al. Early Oral Feeding Post-Upper Gastrointestinal Tract Resection and Primary Anastomosis in Oncology. ABCD, Arquivos Brasileiros de Cirurgia Digestiva. 2018;31(1):e1359. doi: 10.1590/0102-672020180001e1359. PMID: 29947693. S
- 488. Lorimer PD, Motz BM, Watson M, et al. Enteral Feeding Access Has an Impact on Outcomes for Patients with Esophageal Cancer Undergoing Esophagectomy: An Analysis of SEER-Medicare. Annals of Surgical Oncology. 2019;26(5):1311-9. doi: 10.1245/s10434-019-07230-0. PMID: 30783851. S
- 489. Loser A, Abel J, Kutz L, et al. Nutritional intervention in head and neck cancer patients undergoing (chemo-)radiotherapy. Radiotherapy and oncology. 2021;161:S836-S7. doi: 10.1016/s0167-8140(21)07457-0. PMID: CN-02343221. PT
- 490. Lu CY, Chuang HY, Yu FJ, et al. Hypocaloric peripheral parenteral nutrition with lipid emulsion in postoperative gastrointestinal cancer patients. World Journal of Gastrointestinal Oncology. 2010;2(1):51-5. doi: 10.4251/wjgo.v2.i1.51. PMID: 21160817. S
- 491. Lu QL, Zheng K, Zhang P. Effect of preoperative enteral nutrition on postoperative infections and nutritional indices in esophageal cancer patients with esophageal stenosis. World Chinese Journal of Digestology. 2013;21(2):2434-9. doi: 10.11569/wcjd.v21.i24.2434. PMID: CN-00910473. NE
- 492. Lu YX, Wang YJ, Xie TY, et al. Effects of early oral feeding after radical total gastrectomy in gastric cancer patients. World Journal of Gastroenterology. 2020;26(3):5508-19. doi: 10.3748/wjg.v26.i36.5508. PMID: 33024401. S
- 493. Lundholm K, Daneryd P, Bosaeus I, et al. Palliative Nutritional Intervention in Addition to Cyclooxygenase and Erythropoietin Treatment for Patients with Malignant Disease: Effects on Survival, Metabolism, and Function: A Randomized Prospective Study. Cancer. 2004;100(9):1967-77. doi: 10.1002/cncr.20160. PMID: 38529575. P

- 494. Lundholm K, Korner U, Gunnebo L, et al. Insulin treatment in cancer cachexia: Effects on survival, metabolism, and physical functioning. Clinical Cancer Research. 2007;13(9):2699-706. doi: 10.1158/1078-0432.Ccr-06-2720. PMID: 46788038. I
- 495. Luvian-Morales J, Delgadillo-Gonzalez M, Flores-Cisneros L, et al. Effect of an antiinflammatory diet on cervical cancer patients. Clinical Nutrition ESPEN. 2021;46:S710-S1. doi: 10.1016/j.clnesp.2021.09.472. PMID: CN-02347927. PT
- 496. Ma C, Tsai H, Su W, et al. Combination of arginine, glutamine, and omega-3 fatty acid supplements for perioperative enteral nutrition in surgical patients with gastric adenocarcinoma or gastrointestinal stromal tumor (GIST): A prospective, randomized, double-blind study. Journal of Postgraduate Medicine. 2018;64(3):155-63. doi: 10.4103/jpgm.JPGM_693_17. PMID: 29848836. SS
- 497. Ma CJ, Huang CW, Yeh YS, et al. Supplemental home parenteral nutrition improved nutrition status with comparable quality of life in malnourished unresectable/metastatic gastric cancer receiving salvage chemotherapy. Supportive Care in Cancer. 2021;29(4):1977-88. doi: 10.1007/s00520-020-05687-4. PMID: 32827265. S
- 498. Ma CJ, Sun LC, Chen FM, et al. A double-blind randomized study comparing the efficacy and safety of a composite vs a conventional intravenous fat emulsion in postsurgical gastrointestinal tumor patients. Nutrition in Clinical Practice. 2012;27(3):410-5. doi: 10.1177/0884533611436115. PMID: 22460385. SS
- 499. Ma HW, Zhao JT, Zhao X. The Effect of Fennel Tea Drinking on Postoperative Gut Recovery after Gynecological Malignancies Operation. Sichuan da xue xue bao. Yi xue ban [Journal of Sichuan University. Medical science edition]. 2015;46(6):940-3. PMID: CN-01157622. NE

- 500. Ma L, Luo GY, Ren YF, et al. Concurrent chemoradiotherapy combined with enteral nutrition support: a radical treatment strategy for esophageal squamous cell carcinoma patients with malignant fistulae. Chinese Journal of Cancer. 2017;36(1):8. doi: 10.1186/s40880-016-0171-6. PMID: 28077159. S
- 501. Machon C, Thezenas S, Dupuy AM, et al. Immunonutrition before and during radiochemotherapy: improvement of inflammatory parameters in head and neck cancer patients. Supportive Care in Cancer. 2012;20(1):3129-35. doi: 10.1007/s00520-012-1444-5. PMID: 22453793. C
- 502. Mack LA, Kaklamanos IG, Livingstone AS, et al. Gastric decompression and enteral feeding through a double-lumen gastrojejunostomy tube improves outcomes after pancreaticoduodenectomy. Annals of Surgery. 2004;240(5):845-51. doi: 10.1097/01.sla.0000143299.72623.73. PMID: 39425778. SS
- 503. Macleod M, Steele RJC, O'Carroll RE, et al. Feasibility study to assess the delivery of a lifestyle intervention (TreatWELL) for patients with colorectal cancer undergoing potentially curative treatment. BMJ Open. 2018;8(6). doi: 10.1136/bmjopen-2017-021117. PMID: 623793587. I
- 504. Madeddu C, MacCio A, Astara G, et al. Open phase II study on efficacy and safety of an oral amino acid functional cluster supplementation in cancer cachexia. Mediterranean Journal of Nutrition and Metabolism. 2010;3(2):165-72. doi: 10.1007/s12349-010-0016-9. PMID: 50984478. C
- 505. Madhok BM, Yeluri S, Haigh K, et al. Parenteral nutrition for patients with advanced ovarian malignancy. Journal of Human Nutrition & Dietetics. 2011;24(2):187-91. doi: 10.1111/j.1365-277X.2010.01127.x. PMID: 21843153. I
- 506. Maev IV, Shaburov RI, Khenkina NA. Clinical assessment of the efficacy of early enteral nutrition with the help of "peptamen" nutriment in patients who underwent gastrectomy due to stomach cancer. Eksperimental'naia i klinicheskaia gastroenterologiia [Experimental & clinical gastroenterology]. 2003(3):100-6. PMID: CN-00471321. NE

- 507. Magne N, Marcy PY, Foa C, et al. Comparison between nasogastric tube feeding and percutaneous fluoroscopic gastrostomy in advanced head and neck cancer patients. European Archives of Oto-Rhino-Laryngology. 2001;258(2):89-92. doi: 10.1007/s004050000311. PMID: 32186485. S
- 508. Mahawongkajit P, Techagumpuch A. Gastrostomy in Patients with Previous Abdominal Surgery: A Comparative Study Between the Laparoscopy-Assisted Introducer Percutaneous Endoscopic Gastrostomy Versus Open Gastrostomy in Advanced Esophageal Cancer. Dysphagia. 2021;36(1):67-72. doi: 10.1007/s00455-020-10110-5. PMID: 2004659802. S
- 509. Mahawongkajit P, Techagumpuch A, Limpavitayaporn P, et al. Comparison of Introducer Percutaneous Endoscopic Gastrostomy with Open Gastrostomy in Advanced Esophageal Cancer Patients. Dysphagia. 2020;35(1):117-20. doi: 10.1007/s00455-019-10017-w. PMID: 31025103. I
- 510. Mahdavi R, Faramarzi E, Mohammad-Zadeh M, et al. Effects of conjugated linoleic acid supplementation on nutritional status, symptoms of eating problems and dietary intake in rectal cancer patients undergoing chemoradiotherapy. Current Topics in Nutraceutical Research. 2013;11(1):15-20. PMID: 370229417. SS
- 511. Mahendran R, Tewari M, Dixit VK, et al. Enhanced recovery after surgery protocol enhances early postoperative recovery after pancreaticoduodenectomy. Hepatobiliary & Pancreatic Diseases International. 2019;18(2):188-93. doi: 10.1016/j.hbpd.2018.12.005. PMID: 30573300. I
- 512. Makay O, Kaya T, Firat O, et al. omega-3 Fatty acids have no impact on serum lactate levels after major gastric cancer surgery. Jpen: Journal of Parenteral & Enteral Nutrition. 2011;35(4):488-92. doi: 10.1177/0148607110386611. PMID: 21700966. SS

- 513. Malá E, Vejražková E, Bielmeierová J, et al. Long Term Monitoring of Nutritional, Clinical Status and Quality of Life in Head and Neck Cancer Patients. Klinicka Onkologie. 2015;28(3):200-14. doi: 10.14735/amko2015200. PMID: CN-01078437. NE
- 514. Manasek V, Bezdek K, Foltys A, et al. The impact of high protein nutritional support on clinical outcomes and treatment costs of patients with colorectal cancer. Klinicka Onkologie. 2016;29(5):351-7. doi: 10.14735/amko2016351. PMID: 613452761. S
- 515. Manba N, Koyama Y, Kosugi S, et al. Is early enteral nutrition initiated within 24 hours better for the postoperative course in esophageal cancer surgery? Journal of Clinical Medicine Research. 2014;6(1):53-8. doi: 10.4021/jocmr1665w. PMID: 24400032. S
- 516. Manfredelli S, Delhorme JB, Venkatasamy A, et al. Could a Feeding Jejunostomy be Integrated into a Standardized Preoperative Management of Oeso-gastric Junction Adenocarcinoma? Annals of Surgical Oncology. 2017;24(1):3324-30. doi: 10.1245/s10434-017-5945-9. PMID: 28653159. S
- 517. Mangiante G, Rossi L, Carluccio S, et al. Influence of enteral nutrition on cytokine response in resective liver surgery. Chirurgia Italiana. 2002;54(5):613-9. PMID: CN-00412043. NE
- 518. Mansouri-Tehrani HS, Khorasgani MR, Roayaei M. Effects of Probiotics with or without Honey on Radiation-induced Diarrhea. International journal of radiation research. 2016;14(3):205-13. doi: 10.18869/acadpub.ijrr.14.3.205. PMID: CN-01288414. I
- 519. Mantovani G, Maccio A, Madeddu C, et al. A phase II study with antioxidants, both in the diet and supplemented, pharmaconutritional support, progestagen, and anticyclooxygenase-2 showing efficacy and safety in patients with cancer-related anorexia/cachexia and oxidative stress. Cancer Epidemiology Biomarkers and Prevention. 2006;15(5):1030-4. doi: 10.1158/1055-9965.Epi-05-0538. PMID: 43823535. S

- 520. Mantovani G, Maccio A, Madeddu C, et al. Randomized phase III clinical trial of five different arms of treatment for patients with cancer cachexia: interim results. Nutrition. 2008;24(4):305-13. doi: 10.1016/j.nut.2007.12.010. PMID: 18262758. SS
- 521. Manzanares Campillo MDC, Martín Fernández J, Amo Salas M, et al. A randomized controlled trial of preoperative oral immunonutrition in patients undergoing surgery for colorectal cancer: hospital stay and health care costs. Cirugia y Cirujanos. 2017;85(5):393-400. doi: 10.1016/j.circir.2016.10.029. PMID: CN-01704650. NE
- 522. Manzanares MDC, Martin J, Amo-Salas M, et al. Reduction of postoperative morbidity in programmed colorectal cancer: Preoperative oral immunonutrition. Revista Chilena de Cirugia. 2017;69(5):389-96. doi: 10.1016/j.rchic.2017.04.001. PMID: 615889841. NE
- 523. Mao Q-Q, Liu Y. Impact of early enteral nutrition on intestinal mucosal barrier and nutrition status in advanced esophageal cancer patients undergoing synchronous chemoradiotherapy. World Chinese Journal of Digestology. 2019;27(2):101-6. doi: 10.11569/wcjd.v27.i2.101. PMID: CN-01932613. NE
- 524. Mareschal J, Weber K, Rigoli P, et al. The ADAPP trial: a two-year longitudinal multidisciplinary intervention study for prostate cancer frail patients on androgen deprivation associated to curative radiotherapy. Acta Oncologica. 2017;56(4):569-74. doi: 10.1080/0284186x.2016.1273545. PMID: 614072407. I
- 525. Margari N, Lymperopoulos N, Noula M, et al. Indications and results of dietary intervention in patients with cancer. Review of Clinical Pharmacology and Pharmacokinetics, International Edition. 2002;16(2):157-62. PMID: 34984298. O
- 526. Margolis M, Alexander P, Trachiotis GD, et al. Percutaneous Endoscopic Gastrostomy before Multimodality Therapy in Patients with Esophageal Cancer. Annals of Thoracic Surgery. 2003;76(5):1694-8. doi: 10.1016/s0003-4975(02)04890-7. PMID: 37378340. I

- 527. Mari GM, Costanzi A, Maggioni D, et al. Fasttrack versus standard care in laparoscopic high anterior resection: A prospective randomized-controlled trial. Surgical Laparoscopy, Endoscopy and Percutaneous Techniques. 2014;24(2):118-21. doi: 10.1097/SLE.0b013e3182a50e3a. PMID: 372762492. I
- 528. Martinez-Sanchez MA, Nunez-Sanchez MA, Balaguer-Roman A, et al. Gut Microbiome Modification through Dietary Intervention in Patients with Colorectal Cancer: Protocol for a Prospective, Interventional, Controlled, Randomized Clinical Trial in Patients with Scheduled Surgical Intervention for CRC. J Clin Med. 2022 Jun 22;11(13):22. doi: 10.3390/jcm11133613. PMID: 35806897. S
- 529. Martin-McGill KJ, Marson AG, Tudur Smith C, et al. Ketogenic diets as an adjuvant therapy for glioblastoma (KEATING): a randomized, mixed methods, feasibility study. Journal of Neuro-Oncology. 2020;147(1):213-27. doi: 10.1007/s11060-020-03417-8. PMID: 32036576. SS
- 530. Martos-Benitez FD, Gutierrez-Noyola A, Garcia AS, et al. Program of Intestinal Rehabilitation and Early Postoperative Enteral Nutrition: A Prospective Cohort Study. ABCD, Arquivos Brasileiros de Cirurgia Digestiva. 2018;31(3):e1387. doi: 10.1590/0102-672020180001e1387. PMID: 30133679. I
- 531. Marx C, Rasmussen T, Jakobsen DH, et al. The effect of accelerated rehabilitation on recovery after surgery for ovarian malignancy. Acta Obstetricia et Gynecologica Scandinavica. 2006;85(4):488-92. PMID: 16612713. I
- 532. Matsuyama J, Imamura H, Fukui J, et al. Effects of an oral elemental nutrition supplement in gastric cancer patients with adjuvant S-1 chemotherapy after gastrectomy: a phase II study (OGSG1108). Annals of surgical oncology. Conference: 70th annual cancer symposium of the society of surgical oncology, SSO 2017. United states. 2017;24(1):S176. doi: 10.1245/s10434-017-5785-7. PMID: CN-01408315. C

- 533. Mays AC, Bartels HG, Wistermayer PR, et al. Potential for health care cost savings with preoperative gastrostomy tube placement in the head and neck cancer population. Head & Neck. 2018;40(1):111-9. doi: 10.1002/hed.24992. PMID: 29131450. I
- 534. Mays AC, Worley M, Ackall F, et al. The association between gastrostomy tube placement, poor post-operative outcomes, and hospital re-admissions in head and neck cancer patients. Surgical Oncology. 2015;24(3):248-57. doi: 10.1016/j.suronc.2015.08.005. PMID: 26321115. S
- 535. McAllister P, MacIver C, Wales C, et al. Gastrostomy insertion in head and neck cancer patients: a 3 year review of insertion method and complication rates. British Journal of Oral & Maxillofacial Surgery. 2013;51(8):714-8. doi: 10.1016/j.bjoms.2013.07.005. PMID: 23954134. I
- 536. Medina JE, Khafif A. Early oral feeding following total laryngectomy. Laryngoscope. 2001;111(3):368-72. doi: 10.1097/00005537-200103000-00002. PMID: 32200075. S
- 537. Medina-Jimenez AK, Monroy-Torres R. Repurposing Individualized Nutritional Intervention as a Therapeutic Component to Prevent the Adverse Effects of Radiotherapy in Patients With Cervical Cancer. Frontiers in Oncology. 2020;10:595351. doi: 10.3389/fonc.2020.595351. PMID: 33364195. C
- 538. Mehrara BJ, Chunilal A, Bui D, et al. Timing of percutaneous endoscopic gastrostomy tube placement after cervical esophageal reconstruction with free jejunal transfer. Annals of Plastic Surgery. 2004;52(6):578-80. PMID: 15166987. C
- 539. Mekhail TM, Adelstein DJ, Rybicki LA, et al. Enteral nutrition during the treatment of head and neck carcinoma: Is a percutaneous endoscopic gastrostomy tube preferable to a nasogastric tube? Cancer. 2001;91(9):1785-90. doi: 10.1002/1097-0142(20010501)91:9<1785::Aidcncr1197>3.0.Co. PMID: 32377965. S

- 540. Mendivil AA, Rettenmaier MA, Abaid LN, et al. The impact of total parenteral nutrition on postoperative recovery in patients treated for advanced stage ovarian cancer. Archives of Gynecology & Obstetrics. 2017;295(2):439-44. doi: 10.1007/s00404-016-4227-2. PMID: 27832350. S
- 541. Meng L, Wei J, Ji R, et al. Effect of early nutrition intervention on advanced nasopharyngeal carcinoma patients receiving chemoradiotherapy. Journal of Cancer. 2019;10(1):3650-6. doi: 10.7150/jca.33475. PMID: 628861261. S
- 542. Mercuri A, Lim Joon D, Wada M, et al. The effect of an intensive nutritional program on daily set-up variations and radiotherapy planning margins of head and neck cancer patients. Journal of Medical Imaging & Radiation Oncology. 2009;53(5):500-5. doi: 10.1111/j.1754-9485.2009.02105.x. PMID: 19788487. S
- 543. Meyer F, Bairati I, Jobin E, et al. Acute adverse effects of radiation therapy and local recurrence in relation to dietary and plasma beta carotene and alpha tocopherol in head and neck cancer patients. Nutrition and Cancer. 2007;59(1):29-35. doi: 10.1080/01635580701397590. PMID: 47549778. I
- 544. Meyer F, Liu G, Douville P, et al. Dietary vitamin D intake and serum 25-hydroxyvitamin D level in relation to disease outcomes in head and neck cancer patients. International Journal of Cancer. 2011;128(7):1741-6. doi: 10.1002/ijc.25496. PMID: 20533282. I
- 545. Mi L, Zhong B, Zhang DL, et al. Effect of early oral enteral nutrition on clinical outcomes after gastric cancer surgery. Zhonghua wei chang wai ke za zhi [Chinese journal of gastrointestinal surgery]. 2012;15(5):464-7. PMID: CN-00979330. NE
- 546. Migita K, Matsumoto S, Wakatsuki K, et al. Effect of Oral Nutritional Supplementation on the Prognostic Nutritional Index in Gastric Cancer Patients. Nutrition & Cancer. 2020:1-8. doi: 10.1080/01635581.2020.1826990. PMID: 32996343. C

- 547. Minig L, Biffi R, Zanagnolo V, et al. Early oral versus "traditional" postoperative feeding in gynecologic oncology patients undergoing intestinal resection: a randomized controlled trial. Annals of Surgical Oncology. 2009;16(6):1660-8. doi: 10.1245/s10434-009-0444-2. PMID: 19330379. SS
- 548. Mitchell S, Williams JP, Bhatti H, et al. A retrospective matched cohort study evaluating the effects of percutaneous endoscopic gastrostomy feeding tubes on nutritional status and survival in patients with advanced gastroesophageal malignancies undergoing systemic anticancer therapy. PLoS ONE [Electronic Resource]. 2017;12(1):e0188628. doi: 10.1371/journal.pone.0188628. PMID: 29186164. I
- 549. Miyachi T, Oyama A, Tsuchiya T, et al. Perioperative oral administration of cystine and theanine enhances recovery after distal gastrectomy: A prospective randomized trial. Journal of Parenteral and Enteral Nutrition. 2013;37(3):384-91. doi: 10.1177/0148607112458798. PMID: 368796322. O
- 550. Mizuta M, Endo I, Yamamoto S, et al. Perioperative supplementation with bifidobacteria improves postoperative nutritional recovery, inflammatory response, and fecal microbiota in patients undergoing colorectal surgery: a prospective, randomized clinical trial. Bmfh. 2016;35(2):77-87. doi: 10.12938/bmfh.2015-017. PMID: 27200261. I
- 551. Mohamed Abd Elaziz L, Salah T, Gharib F. The role of neoadjuvant FLOT chemotherapy with and without omega 3 in locally advanced gastric carcinoma. Journal of B.U.On. 2020;25(6):2672-7. PMID: 33455112. SS
- 552. Mohamed IM, Whiting J, Tan BH. Impact of regular enteral feeding via jejunostomy during neo-adjuvant chemotherapy on body composition in patients with oesophageal cancer. World Journal of Gastrointestinal Oncology. 2019;11(1):1182-92. doi: 10.4251/wjgo.v11.i12.1182. PMID: 31908723. C

- 553. Molassiotis A, Brown T, Cheng HL, et al. The effects of a family-centered psychosocialbased nutrition intervention in patients with advanced cancer: the PiCNIC2 pilot randomised controlled trial. Nutrition Journal. 2021;20(1):2. doi: 10.1186/s12937-020-00657-2. PMID: 33388075. SS
- 554. Moleiro J, Faias S, Fidalgo C, et al. Usefulness of Prophylactic Percutaneous Gastrostomy Placement in Patients with Head and Neck Cancer Treated with Chemoradiotherapy. Dysphagia. 2016;31(1):84-9. doi: 10.1007/s00455-015-9661-y. PMID: 26487063. C
- 555. Morales-Oyarvide V, Yuan C, Babic A, et al. Dietary Insulin Load and Cancer Recurrence and Survival in Patients With Stage III Colon Cancer: findings From CALGB 89803 (Alliance). Journal of the National Cancer Institute. 2019;111(2):170-9. doi: 10.1093/jnci/djy098. PMID: CN-02084063. I
- 556. Morato-Martinez M, Santurino C, Lopez-Plaza B, et al. A standardized, integral nutritional intervention and physical activity program reduces body weight in women newly diagnosed with breast cancer. Nutricion Hospitalaria. 2021;05:05. doi: 10.20960/nh.03409. PMID: 33813835. OR
- 557. Mori R, Matsuyama R, Taniguchi K, et al. Efficacy of prolonged elemental diet therapy after pancreaticoduodenectomy for pancreatic ductal adenocarcinoma: A pilot prospective randomized trial (UMIN000004108). Clinical Nutrition ESPEN. 2019;34:116-24. doi: 10.1016/j.clnesp.2019.07.017. PMID: 31677701. SS
- 558. Moro K, Koyama Y, Kosugi SI, et al. Low fatcontaining elemental formula is effective for postoperative recovery and potentially useful for preventing chyle leak during postoperative early enteral nutrition after esophagectomy. Clinical Nutrition. 2016;35(6):1423-8. doi: 10.1016/j.clnu.2016.03.018. PMID: 27071696. I

- 559. Moses AWG, Slater C, Preston T, et al. Reduced total energy expenditure and physical activity in cachectic patients with pancreatic cancer can be modulated by an energy and protein dense oral supplement enriched with n-3 fatty acids. British Journal of Cancer. 2004;90(5):996-1002. doi: 10.1038/sj.bjc.6601620. PMID: 38495818. O
- 560. Motoori M, Sugimura K, Tanaka K, et al. Comparison of synbiotics combined with enteral nutrition and prophylactic antibiotics as supportive care in patients with esophageal cancer undergoing neoadjuvant chemotherapy: A multicenter randomized study. Clinical Nutrition. 2022;41(5):1112-21. doi: 10.1016/j.clnu.2022.03.023. PMID: 35413573. I
- 561. Motoori M, Yano M, Miyata H, et al. Randomized study of the effect of synbiotics during neoadjuvant chemotherapy on adverse events in esophageal cancer patients. Clinical Nutrition. 2017;36(1):93-9. doi: 10.1016/j.clnu.2015.11.008. PMID: 26644166. SS
- 562. Mudarra Garcia N, Naranjo Pena I, Olivares Pizarro SP, et al. Pre-Surgical Nutrition Support Reduces the Incidence of Surgical Wound Complications in Oncological Patients. Nutrition & Cancer. 2020;72(5):801-7. doi: 10.1080/01635581.2019.1653473. PMID: 31433266. S
- 563. Mueller MH, Vandenbussche K, Pelliccia M, et al. Enteral Nutrition Support Reduces the Necessity of Total Parenteral Nutrition to Reach Patient-Specific Caloric Goals Postpancreaticoduodenectomy. Southern Medical Journal. 2015;108(1):748-53. doi: 10.14423/smj.00000000000385. PMID: 26630897. S
- 564. Mundi MS, Velapati S, Kuchkuntla AR, et al. Reduction in Healthcare Utilization With Transition to Peptide-Based Diets in Intolerant Home Enteral Nutrition Patients. Nutrition in Clinical Practice. 2020;35(3):487-94. doi: 10.1002/ncp.10477. PMID: 2004385360. P
- 565. Munshi A, Pandey MB, Durga T, et al. Weight loss during radiotherapy for head and neck malignancies: what factors impact it? Nutrition & Cancer. 2003;47(2):136-40. PMID: 15087265. ST

- 566. Nagano T, Fujita H, Tanaka T, et al. Randomized controlled trial comparing antioxidant-enriched enteral nutrition with immune-enhancing enteral nutrition after esophagectomy for cancer: a pilot study. Surgery Today. 2013;43(1):1240-9. doi: 10.1007/s00595-012-0424-1. PMID: 23224142. SS
- 567. Nagata K, Tsujimoto H, Nagata H, et al. Nutritional benefit of laparoscopic jejunostomy during neoadjuvant chemotherapy for obstructing esophageal cancer. Molecular & Clinical Oncology. 2019;11(6):612-6. doi: 10.3892/mco.2019.1938. PMID: 31692945. ST
- 568. Nagata S, Fukuzawa K, Iwashita Y, et al. Comparison of enteral nutrition with combined enteral and parenteral nutrition in post-pancreaticoduodenectomy patients: a pilot study. Nutrition Journal. 2009;8:24. doi: 10.1186/1475-2891-8-24. PMID: 19519910. SS
- 569. Nakajima H, Yokoyama Y, Inoue T, et al. Clinical Benefit of Preoperative Exercise and Nutritional Therapy for Patients Undergoing Hepato-Pancreato-Biliary Surgeries for Malignancy. Annals of Surgical Oncology. 2019;26(1):264-72. doi: 10.1245/s10434-018-6943-2. PMID: 30367303. I
- 570. Nakajima N. Differential Diagnosis of Cachexia and Refractory Cachexia and the Impact of Appropriate Nutritional Intervention for Cachexia on Survival in Terminal Cancer Patients. Nutrients. 2021;13(3):12. doi: 10.3390/nu13030915. PMID: 33808957. ST
- 571. Nakamura K, Kariyazono H, Komokata T, et al. Influence of preoperative administration of omega-3 fatty acid-enriched supplement on inflammatory and immune responses in patients undergoing major surgery for cancer. Nutrition. 2005;21(6):639-49. doi: 10.1016/j.nut.2005.03.001. PMID: 40733556. SS
- 572. Nakamura M, Iwahashi M, Takifuji K, et al. Optimal dose of preoperative enteral immunonutrition for patients with esophageal cancer. Surgery Today. 2009;39(1):855-60. doi: 10.1007/s00595-009-3967-z. PMID: 19784723. SS

- 573. Nascimento M, Aguilar-Nascimento JE, Caporossi C, et al. Efficacy of synbiotics to reduce acute radiation proctitis symptoms and improve quality of life: a randomized, double-blind, placebo-controlled pilot trial. International Journal of Radiation Oncology, Biology, Physics. 2014;90(2):289-95. doi: 10.1016/j.ijrobp.2014.05.049. PMID: 25304789. I
- 574. Nascimento M, Caporossi C, Eduardo Aguilar-Nascimento J, et al. Efficacy of Synbiotics to Reduce Symptoms and Rectal Inflammatory Response in Acute Radiation Proctitis: A Randomized, Double-Blind, Placebo-Controlled Pilot Trial. Nutrition & Cancer. 2020;72(4):602-9. doi: 10.1080/01635581.2019.1647254. PMID: 31364875. SS
- 575. Navez J, Hubert C, Dokmak S, et al. Early Versus Late Oral Refeeding After Pancreaticoduodenectomy for Malignancy: a Comparative Belgian-French Study in Two Tertiary Centers. Journal of Gastrointestinal Surgery. 2020;24(7):1597-604. doi: 10.1007/s11605-019-04316-8. PMID: 31325133. I
- 576. Newlands RSN, Ntessalen M, Clark J, et al. Pilot randomised controlled trial of Weight Watchers R referral with or without dietitian-led group support for weight loss in women treated for breast cancer: the BRIGHT (BReast cancer weIGHT loss) trial. Pilot & Feasibility Studies. 2019;5:24. doi: 10.1186/s40814-019-0405-x. PMID: 30805199. P
- 577. Nguyen LT, Dang AK, Duong PT, et al. Nutrition intervention is beneficial to the quality of life of patients with gastrointestinal cancer undergoing chemotherapy in Vietnam. Cancer Medicine. 2021;10(5):1668-80. doi: 10.1002/cam4.3766. PMID: 33550719. S
- 578. Nishikawa H, Osaki Y, Inuzuka T, et al. Branched-chain amino acid treatment before transcatheter arterial chemoembolization for hepatocellular carcinoma. World Journal of Gastroenterology. 2012;18(1):1379-84. doi: 10.3748/wjg.v18.i12.1379. PMID: 364529601. O

- 579. Nishino T, Yoshida T, Goto M, et al. The effects of the herbal medicine Daikenchuto (TJ-100) after esophageal cancer resection, open-label, randomized controlled trial. Esophagus. 2018;15(2):75-82. doi: 10.1007/s10388-017-0601-9. PMID: 619906620. SS
- 580. Niu WB, Li ZY, Zhou CX, et al. Clinical effects of early enteral nutrition in patients after laparoscopic surgery for colorectal cancer. World Chinese Journal of Digestology. 2015;23(5):857-61. doi: 10.11569/wcjd.v23.i5.857. PMID: CN-01077279. NE
- 581. Noblett SE, Watson DS, Huong H, et al. Preoperative oral carbohydrate loading in colorectal surgery: a randomized controlled trial. Colorectal Disease. 2006;8(7):563-9. doi: 10.1111/j.1463-1318.2006.00965.x. PMID: CN-00571745. I
- 582. Nomura E, Lee SW, Kawai M, et al. Comparison between early enteral feeding with a transnasal tube and parenteral nutrition after total gastrectomy for gastric cancer. Hepato-Gastroenterology. 2015;62(1):536-9. doi: 10.5754/hge14917. PMID: 604064105. S
- 583. Norman K, Stubler D, Baier P, et al. Effects of creatine supplementation on nutritional status, muscle function and quality of life in patients with colorectal cancer-A double blind randomised controlled trial. Clinical Nutrition. 2006;25(4):596-605. doi: 10.1016/j.clnu.2006.01.014. PMID: 44096728. SS
- 584. Nugent B, Parker MJ, McIntyre IA. Nasogastric tube feeding and percutaneous endoscopic gastrostomy tube feeding in patients with head and neck cancer. Journal of Human Nutrition & Dietetics. 2010;23(3):277-84. doi: 10.1111/j.1365-277X.2010.01047.x. PMID: 20337841. I
- 585. Obling SR, Wilson BV, Pfeiffer P, et al. Home parenteral nutrition increases fat free mass in patients with incurable gastrointestinal cancer. Results of a randomized controlled trial. Clinical Nutrition. 2019;38(1):182-90. doi: 10.1016/j.clnu.2017.12.011. PMID: 29305245. SS

- 586. Odelli C, Burgess D, Bateman L, et al. Nutrition support improves patient outcomes, treatment tolerance and admission characteristics in oesophageal cancer. Clinical Oncology. 2005;17(8):639-45. doi: 10.1016/j.clon.2005.03.015. PMID: 41700330. C
- 587. Oguz M, Kurukahvevioglu O, Salman B, et al. A subjective global assessment of nutritional status: A study of 1400 surgical patients. Gazi Medical Journal. 2005;16(2):66-9. PMID: 43028319. NE
- 588. Oh SY, Jun HJ, Park SJ, et al. A randomized phase II study to assess the effectiveness of fluid therapy or intensive nutritional support on survival in patients with advanced cancer who cannot be nourished via enteral route. Journal of Palliative Medicine. 2014;17(1):1266-70. doi: 10.1089/jpm.2014.0082. PMID: 24984081. P
- 589. Ohkura Y, Haruta S, Tanaka T, et al. Effectiveness of postoperative elemental diet (Elental R) in elderly patients after gastrectomy. World Journal of Surgical Oncology. 2016;14(1):268. PMID: 27756322. I
- 590. Ojo O. Balloon gastrostomy tubes for long-term feeding in the community. British Journal of Nursing. 2011;20(1):34-8. doi: 10.12968/bjon.2011.20.1.34. PMID: 361282184. P
- 591. Ok JH, Lee H, Chung HY, et al. The Potential Use of a Ketogenic Diet in Pancreatobiliary Cancer Patients After Pancreatectomy. Anticancer Research. 2018;38(1):6519-27. doi: 10.21873/anticanres.13017. PMID: 30396981. I
- 592. Okabayashi T, Nishimori I, Sugimoto T, et al. The benefit of the supplementation of perioperative branched-chain amino acids in patients with surgical management for hepatocellular carcinoma: A preliminary study. Digestive Diseases and Sciences. 2008;53(1):204-9. doi: 10.1007/s10620-007-9844-y. PMID: 350265671. S

- 593. Okabayashi T, Nishimori I, Sugimoto T, et al. Effects of branched-chain amino acidsenriched nutrient support for patients undergoing liver resection for hepatocellular carcinoma. Journal of Gastroenterology & Hepatology. 2008;23(1):1869-73. doi: 10.1111/j.1440-1746.2008.05504.x. PMID: 18717761. I
- 594. Okabayashi T, Nishimori I, Yamashita K, et al. Preoperative oral supplementation with carbohydrate and branched-chain amino acid-enriched nutrient improves insulin resistance in patients undergoing a hepatectomy: a randomized clinical trial using an artificial pancreas. Amino Acids. 2010;38(3):901-7. doi: 10.1007/s00726-009-0297-9. PMID: 19399583. I
- 595. Okada T, Nakajima Y, Nishikage T, et al. A prospective study of nutritional supplementation for preventing oral mucositis in cancer patients receiving chemotherapy. Asia Pacific Journal of Clinical Nutrition. 2017;26(1):42-8. doi: 10.6133/apjcn.112015.03. PMID: 28049260. SS
- 596. Okamoto K, Fukatsu K, Hashiguchi Y, et al. Lack of preoperative enteral nutrition reduces gut-associated lymphoid cell numbers in colon cancer patients: a possible mechanism underlying increased postoperative infectious complications during parenteral nutrition. Annals of Surgery. 2013;258(6):1059-64. doi: 10.1097/SLA.0b013e31827a0e05. PMID: 23187750. I
- 597. Okugawa Y, Shirai Y, Toiyama Y, et al. Clinical burden of modified glasgow prognostic scale in colorectal cancer. Anticancer Research. 2018;38(3):1599-610. doi: 10.21873/anticanres.12390. PMID: 621166545. I
- 598. Onar P, Yildiz BD, Yildiz EA, et al. Olive oilbased fat emulsion versus soy oil-based fat emulsion in abdominal oncologic surgery. Nutrition in Clinical Practice. 2011;26(1):61-5. doi: 10.1177/0884533610392920. PMID: 21266699. SS

- 599. O'Neill RF, Haseen F, Murray LJ, et al. A randomised controlled trial to evaluate the efficacy of a 6-month dietary and physical activity intervention for patients receiving androgen deprivation therapy for prostate cancer. Journal of Cancer Survivorship. 2015;9(3):431-40. doi: 10.1007/s11764-014-0417-8. PMID: 25916660. I
- 600. Oozeer NB, Corsar K, Glore RJ, et al. The impact of enteral feeding route on patientreported long term swallowing outcome after chemoradiation for head and neck cancer. Oral Oncology. 2011;47(1):980-3. doi: 10.1016/j.oraloncology.2011.07.011. PMID: 21856212. P
- 601. Orlemann T, Reljic D, Zenker B, et al. A Novel Mobile Phone App (OncoFood) to Record and Optimize the Dietary Behavior of Oncologic Patients: Pilot Study. JMIR Cancer. 2018;4(2):e10703. doi: 10.2196/10703. PMID: 30459139. I
- 602. Ostadrahimi A, Ziaei JE, Esfahani A, et al. Effect of beta glucan on white blood cell counts and serum levels of IL-4 and IL-12 in women with breast cancer undergoing chemotherapy: a randomized double-blind placebo-controlled clinical trial. Asian Pacific journal of cancer prevention : APJCP. 2014;15(1):5733-9. PMID: 605040598. SS
- 603. Osterlund P, Ruotsalainen T, Korpela R, et al. Lactobacillus supplementation for diarrhoea related to chemotherapy of colorectal cancer: a randomised study. British Journal of Cancer. 2007;97(8):1028-34. doi: 10.1038/sj.bjc.6603990. PMID: CN-00619645. I
- 604. Owen PJ, Daly RM, Livingston PM, et al. Efficacy of a multi-component exercise programme and nutritional supplementation on musculoskeletal health in men treated with androgen deprivation therapy for prostate cancer (IMPACT): study protocol of a randomised controlled trial. Trials [Electronic Resource]. 2017;18(1):451. doi: 10.1186/s13063-017-2185-z. PMID: 28974267. PT

- 605. Oya H, Koike M, Iwata N, et al. Feeding duodenostomy decreases the incidence of mechanical obstruction after radical esophageal cancer surgery. World Journal of Surgery. 2015;39(5):1105-10. doi: 10.1007/s00268-015-2952-5. PMID: 25665669. I
- 606. Paccagnella A, Morello M, Da Mosto MC, et al. Early nutritional intervention improves treatment tolerance and outcomes in head and neck cancer patients undergoing concurrent chemoradiotherapy. Supportive Care in Cancer. 2010;18(7):837-45. doi: 10.1007/s00520-009-0717-0. PMID: 19727846. ST
- 607. Page RD, Oo AY, Russell GN, et al. Intravenous hydration versus naso-jejunal enteral feeding after esophagectomy: A randomised study. European Journal of Cardio-Thoracic Surgery. 2002;22(5):666-72. doi: 10.1016/s1010-7940(02)00489-x. PMID: 35300106. SS
- 608. Palma-Milla S, Lopez-Plaza B, Santamaria B, et al. New, Immunomodulatory, Oral Nutrition Formula for Use Prior to Surgery in Patients With Head and Neck Cancer: An Exploratory Study. Jpen: Journal of Parenteral & Enteral Nutrition. 2018;42(2):371-9. doi: 10.1177/0148607116676839. PMID: 29443404. SS
- 609. Pan H, Cai S, Ji J, et al. The impact of nutritional status, nutritional risk, and nutritional treatment on clinical outcome of 2248 hospitalized cancer patients: a multicenter, prospective cohort study in Chinese teaching hospitals. Nutrition & Cancer. 2013;65(1):62-70. doi: 10.1080/01635581.2013.741752. PMID: 23368914. X
- 610. Pan H, Hu X, Yu Z, et al. Use of a fast-track surgery protocol on patients undergoing minimally invasive oesophagectomy: preliminary results. Interactive Cardiovascular & Thoracic Surgery. 2014;19(3):441-7. doi: 10.1093/icvts/ivu172. PMID: 24916581. I

- 611. Pan K, Aragaki AK, Michael Y, et al. Longterm dietary intervention influence on physical activity in the Women's Health Initiative Dietary Modification randomized trial. Breast Cancer Res Treat. 2022 Jul 12;12:12. doi: 10.1007/s10549-022-06655-8. PMID: 35821536. P
- 612. Pang KH, Groves R, Venugopal S, et al. Prospective Implementation of Enhanced Recovery After Surgery Protocols to Radical Cystectomy. European Urology. 2018;73(3):363-71. doi: 10.1016/j.eururo.2017.07.031. PMID: 617739103. I
- 613. Papakostas P, Tsaousi G, Stavrou G, et al. Percutaneous endoscopic gastrostomy feeding of locally advanced oro-pharygolaryngeal cancer patients: Blenderized or commercial food? Oral Oncology. 2017;74:135-41. doi: 10.1016/j.oraloncology.2017.10.001. PMID: 29103742. I
- 614. Papapietro K, Díaz E, Csendes A, et al. Early enteral nutrition in cancer patients subjected to a total gastrectomy. Revista medica de Chile. 2002;130(1):1125-30. PMID: CN-00412329. NE
- 615. Paramo Zunzunegui J, Alonso Garcia M, Martin Cruz B, et al. Impact of the implementation of a preoperative nutritional program for colorectal surgery patients. Revista Espanola de Enfermedades Digestivas. 2020;112(1):909-14. doi: 10.17235/reed.2020.6572/2019. PMID: 33054293. NE
- 616. Park JS, Chung HK, Hwang HK, et al. Postoperative nutritional effects of early enteral feeding compared with total parental nutrition in pancreaticoduodectomy patients: a prosepective, randomized study. Journal of Korean Medical Science. 2012;27(3):261-7. doi: 10.3346/jkms.2012.27.3.261. PMID: 22379336. SS
- 617. Park KO, Choi-Kwon S. Effects of individualized nutritional education programs on the level of nutrient intake and nutritional status of colorectal cancer patients undergoing palliative chemotherapy. Journal of Korean Academy of Nursing. 2012;42(6):799-809. doi: 10.4040/jkan.2012.42.6.799. PMID: 368633774. NE

- 618. Park YE, Park SJ, Park Y, et al. Impact and outcomes of nutritional support team intervention in patients with gastrointestinal disease in the intensive care unit. Medicine (United States). 2017;96(4). doi: 10.1097/md.000000000008776. PMID: 620016995. P
- 619. Parlak E, Atalay BG. The effects of protein support with various content on nutrition status and clinical outcomes in elderly malnourished cancer patients. Progress in Nutrition. 2020;22(4). doi: 10.23751/pn.v22i4.8492. PMID: 2011524367. S
- 620. Parry BM, Milne JM, Yadegarfar G, et al. Dramatic dietary fat reduction is feasible for breast cancer patients: Results of the randomised study, WINS (UK) - stage 1. European Journal of Surgical Oncology. 2011;37(1):848-55. doi: 10.1016/j.ejso.2011.07.010. PMID: 21868187. P
- 621. Parsons JK, Marshall JR, Nelson H. Does altering diet affect progression of prostate cancer? The MEAL study. Bulletin of the American College of Surgeons. 2013;98(1):57-9. PMID: 563030301. PT
- 622. Parsons JK, Newman V, Mohler JL, et al. The Men's Eating and Living (MEAL) study: a Cancer and Leukemia Group B pilot trial of dietary intervention for the treatment of prostate cancer. Urology. 2008;72(3):633-7. doi: 10.1016/j.urology.2007.11.050. PMID: 18280560. O
- 623. Parsons JK, Newman VA, Mohler JL, et al. Dietary modification in patients with prostate cancer on active surveillance: a randomized, multicentre feasibility study. BJU International. 2008;101(1):1227-31. doi: 10.1111/j.1464-410X.2007.07365.x. PMID: 18218061. O
- 624. Parsons JK, Newman VA, Mohler JL, et al. Dietary intervention after definitive therapy for localized prostate cancer: results from a pilot study. Canadian Journal of Urology. 2009;16(3):4648-54. PMID: 19497171. P

- 625. Parsons JK, Pierce JP, Mohler J, et al. Men's Eating and Living (MEAL) study (CALGB 70807 [Alliance]): recruitment feasibility and baseline demographics of a randomized trial of diet in men on active surveillance for prostate cancer. BJU International. 2018;121(4):534-9. doi: 10.1111/bju.13890. PMID: 28437029. P
- 626. Parsons JK, Pierce JP, Natarajan L, et al. A randomized pilot trial of dietary modification for the chemoprevention of noninvasive bladder cancer: the dietary intervention in bladder cancer study. Cancer Prevention Research. 2013;6(9):971-8. doi: 10.1158/1940-6207.Capr-13-0050. PMID: 23867158. P
- 627. Pastore CA, Orlandi SP, Gonzalez MC. Introduction of an omega-3 enriched oral supplementation for cancer patients close to the first chemotherapy: may it be a factor for poor compliance? Nutrition & Cancer. 2014;66(8):1285-92. doi: 10.1080/01635581.2014.956253. PMID: 25329228. SS
- 628. Patel SH, Kooby DA, Staley CA, 3rd, et al. An assessment of feeding jejunostomy tube placement at the time of resection for gastric adenocarcinoma. Journal of Surgical Oncology. 2013;107(7):728-34. doi: 10.1002/jso.23324. PMID: 23450704. I
- 629. Paternostro R, Trauner M. Current treatment of non-alcoholic fatty liver disease. J Intern Med. 2022 Aug;292(2):190-204. doi: 10.1111/joim.13531. PMID: 35796150. P
- 630. Patursson P, Moller G, Muhic A, et al. N-3 fatty acid EPA supplementation in cancer patients receiving abdominal radiotherapy A randomised controlled trial. Clinical Nutrition ESPEN. 2021. doi: 10.1016/j.clnesp.2021.03.001. PMID: 2011529630. SS
- 631. Paxton RJ, Garcia-Prieto C, Berglund M, et al. A randomized parallel-group dietary study for stages II-IV ovarian cancer survivors. Gynecologic Oncology. 2012;124(3):410-6. doi: 10.1016/j.ygyno.2011.11.031. PMID: 22119991. P

- 632. Pedrazzoli P, Caccialanza R, Cotogni P, et al. The Advantages of Clinical Nutrition Use in Oncologic Patients in Italy: Real World Insights. Healthcare. 2020;8(2):06. doi: 10.3390/healthcare8020125. PMID: 32384639. I
- 633. Peerawong T, Phungrassami T, Pruegsanusak K, et al. Comparison of treatment compliance and nutritional outcomes among patients with nasopharyngeal carcinoma with and without percutaneous endoscopic gastrostomy during chemoradiation. Asian Pacific Journal of Cancer Prevention. 2012;13(1):5805-9. doi: 10.7314/apjcp.2012.13.11.5805. PMID: 368370150. S
- 634. Peñalva A, San Martín A, Rosselló J, et al. Oral nutritional supplementation in hematologic patients. Nutricion Hospitalaria.
 2009;24(1):10-6. PMID: CN-00686026. NE
- 635. Peng CB, Li WZ, Xu R, et al. Effects of Early Enteral Immunonutrition on Postoperative Immune Function and Rehabilitation of Patients with Gastric Cancer and Nutritional Risk. Sichuan da xue xue bao. Yi xue ban [Journal of Sichuan University. Medical science edition]. 2017;48(3):427-30. PMID: CN-01944665. NE
- 636. Peng YL, Gong QF, Wand ZQ. The prospective study on application of parenteral nutrition with alanyl-glutamine dipeptide in chemotherapy of gastrointestinal neoplasms patients. Ai zheng [Chinese journal of cancer]. 2006;25(8):1044-7. PMID: CN-00697284. NE
- 637. Pereira NC, Turrini RNT, Poveda VB. Perioperative fasting time among cancer patients submitted to gastrointestinal surgeries. Revista da Escola de Enfermagem da U S P. 2017;51:e03228. doi: 10.1590/s1980-220x2016036203228. PMID: 627741354. I
- 638. Perez Cruz E, Reyes Marin A, Asbun Bojalil J, et al. Effectiveness of Immunonutrition on Inflammatory Markers in Patients with Cancer; Randomized Clinical Trial. Nutricion Hospitalaria. 2015;32(4):1676-82. doi: 10.3305/nh.2015.32.4.9507. PMID: 26545535. SS

- 639. Perrier L, Foucaut AM, Morelle M, et al. Costeffectiveness of an exercise and nutritional intervention versus usual nutritional care during adjuvant treatment for localized breast cancer: the PASAPAS randomized controlled trial. Supportive Care in Cancer. 2020;28(6):2829-42. doi: 10.1007/s00520-019-05078-4. PMID: 31729566. C
- 640. Persson CR, Johansson BBK, Sjoden PO, et al. A randomized study of nutritional support in patients with colorectal and gastric cancer. Nutrition and Cancer. 2002;42(1):48-58. doi: 10.1207/s15327914nc421_7. PMID: 35007006. SS
- 641. Petrelli NJ, Cheng C, Driscoll D, et al. Early postoperative oral feeding after colectomy: An analysis of factors that may predict failure. Annals of Surgical Oncology. 2001;8(1):796-800. doi: 10.1245/aso.2001.8.10.796. PMID: 34003328. C
- 642. Pexe-Machado PA, de Oliveira BD, Dock-Nascimento DB, et al. Shrinking preoperative fast time with maltodextrin and protein hydrolysate in gastrointestinal resections due to cancer. Nutrition. 2013;29(7):1054-9. doi: 10.1016/j.nut.2013.02.003. PMID: 369112525. SS
- 643. Pi RX, Wu Q, Xu XL. Effect of intestinal flora adjustment after interventional therapy for liver cancer on liver function and endotoxin. Journal of practical oncology. 2014;29(5):454-6. PMID: CN-01022907. I
- 644. Pierce JP, Faerber S, Wright FA, et al. A randomized trial of the effect of a plantbased dietary pattern on additional breast cancer events and survival: The Women's Healthy Eating and Living (WHEL) Study. Controlled Clinical Trials. 2002;23(6):728-56. doi: 10.1016/s0197-2456(02)00241-6. PMID: 36005481. P
- 645. Pinto E, Nardi MT, Marchi R, et al. QOLEC2: a randomized controlled trial on nutritional and respiratory counseling after esophagectomy for cancer. Supportive Care in Cancer. 2021;29(2):1025-33. doi: 10.1007/s00520-020-05573-z. PMID: 32572611. SS

- 646. Piquet MA, Ozsahin M, Larpin I, et al. Early nutritional intervention in oropharyngeal cancer patients undergoing radiotherapy. Supportive Care in Cancer. 2002;10(6):502-4. doi: 10.1007/s00520-002-0364-1. PMID: 36056276. S
- 647. Polakowski CB, Kato M, Preti VB, et al. Impact of the preoperative use of synbiotics in colorectal cancer patients: A prospective, randomized, double-blind, placebocontrolled study. Nutrition. 2019;58:40-6. doi: 10.1016/j.nut.2018.06.004. PMID: 30278428. I
- 648. Pothuri B, Montemarano M, Gerardi M, et al. Percutaneous endoscopic gastrostomy tube placement in patients with malignant bowel obstruction due to ovarian carcinoma. Gynecologic Oncology. 2005;96(2):330-4. PMID: 15661217. C
- 649. Pramyothin P, Manyanont S, Trakarnsanga A, et al. A prospective study comparing prophylactic gastrostomy to nutritional counselling with a therapeutic feeding tube if required in head and neck cancer patients undergoing chemoradiotherapy in Thai realworld practice. Journal of Human Nutrition & Dietetics. 2016;29(6):768-76. doi: 10.1111/jhn.12377. PMID: 27028666. I
- 650. Prasad KC, Sreedharan S, Dannana NK, et al. Early oral feeds in laryngectomized patients. Annals of Otology, Rhinology and Laryngology. 2006;115(6):433-8. doi: 10.1177/000348940611500606. PMID: 43865678. S
- 651. Pratt VC, Watanabe S, Bruera E, et al. Plasma and neutrophil fatty acid composition in advanced cancer patients and response to fish oil supplementation. British Journal of Cancer. 2002;87(1):1370-8. doi: 10.1038/sj.bjc.6600659. PMID: 36044310. C
- 652. Pu J, Ji WP, Long F, et al. Postoperative early enteral nutrition therapy with long peptide mixture for gastric cancer patients: a clinical control study. Academic journal of second military medical university. 2014;35(3):256-9. doi: 10.3724/SP.j.1008.2014.00256. PMID: CN-00984715. NE

- 653. Pytlik R, Benes P, Patorkova M, et al. Standardized parenteral alanyl-glutamine dipeptide supplementation is not beneficial in autologous tranplant patients: A randomized, double-blind, placebo controlled study. Bone Marrow Transplantation. 2002;30(1):953-61. doi: 10.1038/sj.bmt.1703759. PMID: 36097486. SS
- 654. Qin Y, Rivera RL, Zhang Y, et al. A randomized intervention of Supplemental Nutrition Assistance Program-Education did not improve dietary outcomes except for vitamin D among lower income women in Indiana. J Acad Nutr Diet. 2022 Jun 30;30:30. doi: 10.1016/j.jand.2022.06.030. PMID: 35781080. P
- 655. Qiu M, Zhou YX, Jin Y, et al. Nutrition support can bring survival benefit to high nutrition risk gastric cancer patients who received chemotherapy. Supportive Care in Cancer. 2015;23(7):1933-9. doi: 10.1007/s00520-014-2523-6. PMID: 25492636. C
- 656. Qiu Y, Zhu X, Wang W, et al. Nutrition support with glutamine dipeptide in patients undergoing liver transplantation. Transplantation Proceedings. 2009;41(1):4232-7. doi: 10.1016/j.transproceed.2009.08.076. PMID: 20005375. P
- 657. Quon H, Myers C, Lambert P, et al. Impact of feeding tubes on prospective functional outcomes in patients with locally advanced head and neck cancer undergoing radiation therapy. Practical Radiation Oncology. 2015;5(6):e567-73. doi: 10.1016/j.prro.2015.05.007. PMID: 26215583. S
- 658. Rabinovitch R, Grant B, Berkey BA, et al. Impact of nutrition support on treatment outcome in patients with locally advanced head and neck squamous cell cancer treated with definitive radiotherapy: A secondary analysis of RTOG trial 90-03. Head and Neck. 2006;28(4):287-96. doi: 10.1002/hed.20335. PMID: 43536740. S
- 659. Rajabi Mashhadi MT, Bagheri R, Ghayour-Mobarhan M, et al. Early Post Operative Enteral Versus Parenteral Feeding after Esophageal Cancer Surgery. Iranian journal of otorhinolaryngology. 2015;27(8):331-6. PMID: 26568935. SS

- 660. Raji Lahiji M, Sajadian A, Haghighat S, et al. Effectiveness of logotherapy and nutrition counseling on psychological status, quality of life, and dietary intake among breast cancer survivors with depressive disorder: a randomized clinical trial. Support Care Cancer. 2022 Jun 27. doi: 10.1007/s00520-022-07237-6. PMID: 35759049. P
- 661. Raveendran C, Yadev I, Sharan K, et al. Prophylactic percutaneous endoscopic gastrostomy tube feeding to prevent weight loss in head and neck cancer patients-a retrospective cohort study. Onkologia i Radioterapia. 2021;15(1):1-6. PMID: 2006094698. S
- 662. Raykher A, Correa L, Russo L, et al. The role of pretreatment percutaneous endoscopic gastrostomy in facilitating therapy of head and neck cancer and optimizing the body mass index of the obese patient. Jpen: Journal of Parenteral & Enteral Nutrition. 2009;33(4):404-10. doi: 10.1177/0148607108327525. PMID: 19520799. C
- 663. Read JA, Beale PJ, Volker DH, et al. Nutrition intervention using an eicosapentaenoic acid (EPA)-containing supplement in patients with advanced colorectal cancer. Effects on nutritional and inflammatory status: A phase II trial. Supportive Care in Cancer. 2007;15(3):301-7. doi: 10.1007/s00520-006-0153-3. PMID: 46295216. C
- 664. Reeves JG, Suriawinata AA, Ng DP, et al. Short-term preoperative diet modification reduces steatosis and blood loss in patients undergoing liver resection. Surgery. 2013;154(5):1031-7. doi: 10.1016/j.surg.2013.04.012. PMID: 23809869. S
- 665. Reiter M, Gerken M, Lindberg-Scharf P, et al. Health services research in colorectal cancer: a quasi-experimental interventional pilot study on in- and outpatient oncology. Journal of Cancer Research & Clinical Oncology. 2021;147(6):1789-802. doi: 10.1007/s00432-020-03454-w. PMID: 33373026. S

- 666. Reitschuler-Cross EB, Arnold B. ACP Journal Club. Parenteral hydration did not improve dehydration or quality of life in advanced cancer. Ann Intern Med. 2013 Mar 19;158(6):JC10. doi: 10.7326/0003-4819-158-6-201303190-02010. PMID: 23552960. S
- 667. Richter E, Denecke A, Klapdor S, et al. Parenteral nutrition support for patients with pancreatic cancer--improvement of the nutritional status and the therapeutic outcome. Anticancer Research. 2012;32(5):2111-8. PMID: 22593497. C
- 668. Riedinger CJ, Kimball KJ, Kilgore LC, et al. Water only fasting and its effect on chemotherapy administration in gynecologic malignancies. Gynecologic Oncology. 2020;159(3):799-803. doi: 10.1016/j.ygyno.2020.09.008. PMID: 32958269. SS
- 669. Rifatbegovic Z, Mesic D, Ljuca F, et al. Effect of probiotics on liver function after surgery resection for malignancy in the liver cirrhotic. Medicinski Arhiv. 2010;64(4):208-11. PMID: 21246916. I
- 670. Rimini M, Pecchi A, Prampolini F, et al. The Prognostic Role of Early Skeletal Muscle Mass Depletion in Multimodality Management of Patients with Advanced Gastric Cancer Treated with First Line Chemotherapy: A Pilot Experience from Modena Cancer Center. Journal of Clinical Medicine. 2021;10(8):15. doi: 10.3390/jcm10081705. PMID: 33921004. I
- 671. Riso S, Aluffi P, Brugnani M, et al. Postoperative enteral immunonutrition in head and neck cancer patients. Clinical Nutrition. 2000;19(6):407-12. doi: 10.1054/clnu.2000.0135. PMID: 30994510. P
- 672. Rivelsrud M, Paur I, Sygnestveit K, et al. Nutritional treatment is associated with longer survival in patients with pancreatic disease and concomitant risk of malnutrition. Clinical Nutrition. 2021;40(4):2128-37. doi: 10.1016/j.clnu.2020.09.037. PMID: 2008042707. I

- 673. Robinson LA, Tanvetyanon T, Grubbs D, et al. Preoperative nutrition-enhanced recovery after surgery protocol for thoracic neoplasms. Journal of Thoracic & Cardiovascular Surgery. 2020;25:25. doi: 10.1016/j.jtcvs.2020.06.016. PMID: 32713631. I
- 674. Robles LA, Shingler E, McGeagh L, et al. Attitudes and adherence to changes in nutrition and physical activity following surgery for prostate cancer: a qualitative study. BMJ Open. 2022 Jun 29;12(6):e055566. doi: 10.1136/bmjopen-2021-055566. PMID: 35768108. S
- 675. Roca-Rodriguez MM, Garcia-Almeida JM, Lupianez-Perez Y, et al. Effect of a specific supplement enriched with n-3 polyunsaturated fatty acids on markers of inflammation, oxidative stress and metabolic status of ear, nose and throat cancer patients. Oncology Reports. 2014;31(1):405-14. doi: 10.3892/or.2013.2806. PMID: 24154820. SS
- 676. Rock CL, Moskowitz A, Huizar B, et al. High vegetable and fruit diet intervention in premenopausal women with cervical intraepithelial neoplasia. Journal of the American Dietetic Association. 2001;101(1):1167-74. PMID: 11678487. P
- 677. Rock CL, Thomson C, Caan BJ, et al. Reduction in fat intake is not associated with weight loss in most women after breast cancer diagnosis: Evidence from a randomized controlled trial. Cancer. 2001;91(1):25-34. doi: 10.1002/1097-0142(20010101)91:1<25::Aidcncr4>3.0.Co. PMID: 32052610. P
- 678. Rodriguez-De Loera LH, Ortiz GG, Rivero Moragrega P, et al. Effect of symbiotic supplementation on fecal calprotectin levels and lactic acid bacteria, Bifidobacteria, Escherichia coli and Salmonella DNA in patients with cervical cancer. Nutricion Hospitalaria. 2018;35(6):1394-400. doi: 10.20960/nh.1762. PMID: 30525855. NE
- 679. Rotovnik Kozjek N, Kompan L, Zagar T, et al. Influence of enteral glutamine on inflammatory and hormonal response in patients with rectal cancer during preoperative radiochemotherapy. European Journal of Clinical Nutrition. 2017;71(5):671-3. doi: 10.1038/ejcn.2017.11. PMID: 28272402. O

- 680. Rowan NR, Johnson JT, Fratangelo CE, et al. Utility of a perioperative nutritional intervention on postoperative outcomes in high-risk head & neck cancer patients. Oral Oncology. 2016;54:42-6. doi: 10.1016/j.oraloncology.2016.01.006. PMID: 26803343. SS
- 681. Ruggeri E, Giannantonio M, Ostan R, et al. Choice of access route for artificial nutrition in cancer patients: 30 y of activity in a home palliative care setting. Nutrition. 2021;90:111264. doi: 10.1016/j.nut.2021.111264. PMID: 34004413. I
- 682. Rupnik E, Skerget M, Sever M, et al. Feasibility and safety of exercise training and nutritional support prior to haematopoietic stem cell transplantation in patients with haematologic malignancies. BMC Cancer. 2020;20(1):1142. doi: 10.1186/s12885-020-07637-z. PMID: 33234112. I
- 683. Rustom IK, Jebreel A, Tayyab M, et al. Percutaneous endoscopic, radiological and surgical gastrostomy tubes: a comparison study in head and neck cancer patients. Journal of Laryngology & Otology. 2006;120(6):463-6. PMID: 16772054. S
- 684. Ryan AM, Healy LA, Power DG, et al. Short-term nutritional implications of total gastrectomy for malignancy, and the impact of parenteral nutritional support. Clinical Nutrition. 2007;26(6):718-27. doi: 10.1016/j.clnu.2007.08.013. PMID: 350082499. I
- 685. Saeed SM, Fontaine JP, Dam AN, et al. Is Preoperative G-Tube Use Safe for Esophageal Cancer Patients? Journal of the American College of Nutrition. 2020;39(4):301-6. doi: 10.1080/07315724.2019.1646168. PMID: 31397638. I
- 686. Sagar RC, Kumar KVV, Ramachandra C, et al. Perioperative Artificial Enteral Nutrition in Malnourished Esophageal and Stomach Cancer Patients and Its Impact on Postoperative Complications. Indian Journal of Surgical Oncology. 2019;10(3):460-4. doi: 10.1007/s13193-019-00930-9. PMID: 31496591. S

- 687. Saito K, Nakajima Y, Kawada K, et al. Is a Central Venous Catheter Necessary for the Perioperative Management of Esophagectomy? A Prospective Randomized Pilot Study Comparing Two Different Perioperative Regimens. Digestive Surgery. 2016;33(6):478-87. doi: 10.1159/000446572. PMID: 27250846. SS
- 688. Salas S, Baumstarck-Barrau K, Alfonsi M, et al. Impact of the prophylactic gastrostomy for unresectable squamous cell head and neck carcinomas treated with radio-chemotherapy on quality of life: Prospective randomized trial. Radiotherapy & Oncology. 2009;93(3):503-9. doi: 10.1016/j.radonc.2009.05.016. PMID: 19524315. SS
- 689. Sandmael JA, Bye A, Solheim TS, et al. Physical rehabilitation in patients with head and neck cancer: Impact on health-related quality of life and suitability of a posttreatment program. Laryngoscope Investigative Otolaryngology. 2020;5(2):330-8. doi: 10.1002/lio2.368. PMID: 32337365. I
- 690. Sandmael JA, Bye A, Solheim TS, et al. Feasibility and preliminary effects of resistance training and nutritional supplements during versus after radiotherapy in patients with head and neck cancer: A pilot randomized trial. Cancer. 2017;123(2):4440-8. doi: 10.1002/cncr.30901. PMID: 28759113. I
- 691. Santos JG, Faria G, Cruz W, et al. Adjuvant effect of low-carbohydrate diet on outcomes of patients with recurrent glioblastoma under intranasal perillyl alcohol therapy. Surgical neurology international. 2020;11:389. doi: 10.25259/sni_445_2020. PMID: 33282452. P
- 692. Sanz-Paris A, Martinez-Trufero J, Lambea-Sorrosal J, et al. Clinical and Nutritional Effectiveness of a Nutritional Protocol with Oligomeric Enteral Nutrition in Patients with Oncology Treatment-Related Diarrhea. Nutrients. 2020;12(5):25. doi: 10.3390/nu12051534. PMID: 32466127. C

- 693. Sanz-Paris A, Martinez-Trufero J, Lambea-Sorrosal J, et al. Impact of an Oral Nutritional Protocol with Oligomeric Enteral Nutrition on the Quality of Life of Patients with Oncology Treatment-Related Diarrhea. Nutrients. 2020;13(1):29. doi: 10.3390/nu13010084. PMID: 33383949. C
- 694. Sasidharan BK, Ramadass B, Viswanathan PN, et al. A phase 2 randomized controlled trial of oral resistant starch supplements in the prevention of acute radiation proctitis in patients treated for cervical cancer. Journal of Cancer Research & Therapeutics. 2019;15(6):1383-91. doi: 10.4103/jcrt.JCRT_152_19. PMID: 31898677. I
- 695. Savikko J, Ilmakunnas M, Makisalo H, et al. Enhanced recovery protocol after liver resection. British Journal of Surgery. 2015;102(1):1526-32. doi: 10.1002/bjs.9912. PMID: 26331595. I
- 696. Scheid C, Hermann K, Kremer G, et al. Randomized, double-blind, controlled study of glycyl-glutamine-dipeptide in the parenteral nutrition of patients with acute leukemia undergoing intensive chemotherapy. Nutrition. 2004;20(3):249-54. doi: 10.1016/j.nut.2003.11.018. PMID: 38264378. SS
- 697. Scheier MF, Helgeson VS, Schulz R, et al. Interventions to enhance physical and psychological functioning among younger women who are ending nonhormonal adjuvant treatment for early-stage breast cancer. Journal of Clinical Oncology. 2005;23(1):4298-311. doi: 10.1200/jco.2005.05.362. PMID: 46207004. P
- 698. Scheier MF, Helgeson VS, Schulz R, et al. Moderators of interventions designed to enhance physical and psychological functioning among younger women with early-stage breast cancer. Journal of Clinical Oncology. 2007;25(3):5710-4. doi: 10.1200/jco.2007.11.7093. PMID: 351024716. O

- 699. Schiavon CC, Vieira FG, Ceccatto V, et al. Nutrition education intervention for women with breast cancer: effect on nutritional factors and oxidative stress. Journal of Nutrition Education & Behavior. 2015;47(1):2-9. doi: 10.1016/j.jneb.2014.09.005. PMID: 25528078. S
- 700. Schink K, Herrmann HJ, Schwappacher R, et al. Effects of whole-body electromyostimulation combined with individualized nutritional support on body composition in patients with advanced cancer: a controlled pilot trial. BMC Cancer. 2018;18(1):886. doi: 10.1186/s12885-018-4790-y. PMID: 30208857. I
- 701. Schink K, Reljic D, Herrmann HJ, et al. Whole-Body Electromyostimulation Combined With Individualized Nutritional Support Improves Body Composition in Patients With Hematological Malignancies - A Pilot Study. Frontiers in Physiology. 2018;9:1808. doi: 10.3389/fphys.2018.01808. PMID: 30618820. I
- 702. Schmidt H, Boese S, Bauer A, et al. Interdisciplinary care programme to improve self-management for cancer patients undergoing stem cell transplantation: a prospective non-randomised intervention study. European Journal of Cancer Care. 2017;26(4). doi: 10.1111/ecc.12458. PMID: 26857103. I
- 703. Schmidt M, Pfetzer N, Schwab M, et al. Effects of a ketogenic diet on the quality of life in 16 patients with advanced cancer: A pilot trial. Nutrition and Metabolism. 2011;8. doi: 10.1186/1743-7075-8-54. PMID: 362397702. C
- 704. Schmidt N, Moller G, Baeksgaard L, et al. Fish oil supplementation in cancer patients. Capsules or nutritional drink supplements? A controlled study of compliance. Clinical Nutrition ESPEN. 2020;35:63-8. doi: 10.1016/j.clnesp.2019.12.004. PMID: 31987122. S
- 705. Schneider S. Perioperative parenteral nutrition in malnourished patients with digestive system cancer. Gastroenterologie Clinique et Biologique. 2000;24(5):596-7. PMID: CN-01735717. NE

- 706. Schneider SM, Pouget I, Staccini P, et al. Quality of life in long-term home enteral nutrition patients. Clinical Nutrition. 2000;19(1):23-8. doi: 10.1054/clnu.1999.0068. PMID: 30150805. C
- 707. Schricker T, Meterissian S, Lattermann R, et al. Anticatabolic effects of avoiding preoperative fasting by intravenous hypocaloric nutrition a randomized clinical trial. Annals of Surgery. 2008;248(6):1051-9. doi: 10.1097/SLA.0b013e31818842d8. PMID: 354644044. O
- 708. Schricker T, Wykes L, Eberhart L, et al. Randomized clinical trial of the anabolic effect of hypocaloric parenteral nutrition after abdominal surgery. British Journal of Surgery. 2005;92(8):947-53. doi: 10.1002/bjs.5105. PMID: CN-00523356. O
- 709. Schroder FH, Roobol MJ, Boeve ER, et al. Randomized, double-blind, placebocontrolled crossover study in men with prostate cancer and rising PSA: Effectiveness of a dietary supplement. European Urology. 2005;48(6):922-31. doi: 10.1016/j.eururo.2005.08.005. PMID: 41641637. P
- 710. Scolapio JS, Spangler PR, Romano MM, et al. Prophylactic placement of gastrostomy feeding tubes before radiotherapy in patients with head and neck cancer: Is It Worthwhile? Journal of Clinical Gastroenterology. 2001;33(3):215-7. doi: 10.1097/00004836-200109000-00009. PMID: 32762891. C
- 711. Seguin P, Locher C, Boudjema K, et al. Effect of a Perioperative Nutritional Supplementation with Oral Impact R in Patients undergoing Hepatic Surgery for Liver Cancer: A Prospective, Placebo-Controlled, Randomized, Double-Blind Study. Nutrition & Cancer. 2016;68(3):464-72. doi: 10.1080/01635581.2016.1153670. PMID: 27007018. SS
- 712. Seguy D, Berthon C, Micol JB, et al. Enteral feeding and early outcomes of patients undergoing allogeneic stem cell transplantation following myeloablative conditioning. Transplantation. 2006;82(6):835-9. PMID: 17006332. S

- 713. Seike J, Tangoku A, Yuasa Y, et al. The effect of nutritional support on the immune function in the acute postoperative period after esophageal cancer surgery: Total parenteral nutrition versus enteral nutrition. Journal of Medical Investigation. 2011;58(1):75-82. doi: 10.2152/jmi.58.75. PMID: 361319395. S
- 714. Senesse P, Tadmouri A, Culine S, et al. A prospective observational study assessing home parenteral nutrition in patients with gastrointestinal cancer: benefits for quality of life. Journal of Pain & Symptom Management. 2015;49(2):183-91.e2. doi: 10.1016/j.jpainsymman.2014.05.016. PMID: 24945492. C
- 715. Senkal M, Haaker R, Deska T, et al. Early enteral gut feeding with conditionally indispensable pharmaconutrients is metabolically safe and is well tolerated in postoperative cancer patients - A pilot study. Clinical Nutrition. 2004;23(5):1193-8. doi: 10.1016/j.clnu.2004.03.010. PMID: 39243037. C
- 716. Senkal M, Haaker R, Linseisen J, et al. Preoperative oral supplementation with long-chain OMEGA-3 fatty acids beneficially alters phospholipid fatty acid patterns in liver, gut mucosa, and tumor tissue. Journal of Parenteral and Enteral Nutrition. 2005;29(4):236-40. doi: 10.1177/0148607105029004236. PMID: 44296713. SS
- 717. Serbanescu-Kele CM, Halmos GB, Wedman J, et al. Early feeding after total laryngectomy results in shorter hospital stay without increased risk of complications: a retrospective case-control study. Clinical Otolaryngology. 2015;40(6):587-92. doi: 10.1111/coa.12420. PMID: 25816718. S
- 718. Shah UA, Iyengar NM. Plant-Based and Ketogenic Diets As Diverging Paths to Address Cancer: A Review. JAMA Oncol. 2022 Jul 7;07:07. doi: 10.1001/jamaoncol.2022.1769. PMID: 35797039. S
- 719. Shang E, Weiss C, Post S, et al. The influence of early supplementation of parenteral nutrition on quality of life and body composition in patients with advanced cancer. Jpen: Journal of Parenteral & Enteral Nutrition. 2006;30(3):222-30. PMID: 16639069. RE

- 720. Shao F, Xin FZ, Yang CG, et al. The impact of microbial immune enteral nutrition on the patients with acute radiation enteritis in bowel function and immune status. Cell Biochemistry & Biophysics. 2014;69(2):357-61. doi: 10.1007/s12013-013-9807-1. PMID: 24366547. P
- 721. Shao F, Yang CG, Liu X, et al. Application of microbiological and immunological enteral nutrition in patients with gastrointestinal cancer complicated with diabetes mellitus. Zhonghua wei chang wai ke za zhi [Chinese journal of gastrointestinal surgery]. 2012;15(5):476-9. PMID: CN-00979327. NE
- 722. Sharifian HA, Najafi M, Khajavi M. Early oral feeding following total laryngectomy. Tanaffos. 2008;7(2):64-70. PMID: 354728618. I
- 723. Shaw C, Mortimer P, Judd PA. Randomized controlled trial comparing a low-fat diet with a weight-reduction diet in breast cancer-related lymphedema. Cancer. 2007;109(1):1949-56. doi: 10.1002/cncr.22638. PMID: 46744187. P
- 724. Shaw C, Mortimer P, Judd PA. A randomized controlled trial of weight reduction as a treatment for breast cancer-related lymphedema. Cancer. 2007;110(8):1868-74. doi: 10.1002/cncr.22994. PMID: CN-00611936. SS
- 725. Sheean PM, Kilkus JM, Liu D, et al. Incident hyperglycemia, parenteral nutrition administration and adverse outcomes in patients with myeloma admitted for initial auto-SCT. Bone Marrow Transplantation. 2013;48(8):1117-22. doi: 10.1038/bmt.2013.11. PMID: 23419432. S
- 726. Sheehan P, Denieffe S, Murphy NM, et al. Exercise is more effective than health education in reducing fatigue in fatigued cancer survivors. Supportive Care in Cancer. 2020;28(1):4953-62. doi: 10.1007/s00520-020-05328-w. PMID: 2004197816. P
- 727. Shen J, Dai S, Li Z, et al. Effect of Enteral Immunonutrition in Patients Undergoing Surgery for Gastrointestinal Cancer: An Updated Systematic Review and Meta-Analysis. Front. 2022;9:941975. doi: 10.3389/fnut.2022.941975. PMID: 35845793. S

- 728. Shen Y, Zhou Y, He T, et al. Effect of Preoperative Nutritional Risk Screening and Enteral Nutrition Support in Accelerated Recovery after Resection for Esophageal Cancer. Nutrition and Cancer. 2021;73(4):596-601. doi: 10.1080/01635581.2020.1764981. PMID: 2004969096. S
- 729. Sheng B, Chen W, Zhao L. Efficacy of perioperative enteral nutrition in management of hepatocellular carcinoma with cirrhosis. World Chinese Journal of Digestology. 2013;21(2):2999-3003. doi: 10.11569/wcjd.v21.i28.2999. PMID: CN-00914124. NE
- 730. Shetiwy M, Fady T, Shahatto F, et al. Standardizing the Protocols for Enhanced Recovery From Colorectal Cancer Surgery: Are We a Step Closer to Ideal Recovery? Annals of Coloproctology. 2017;33(3):86-92. doi: 10.3393/ac.2017.33.3.86. PMID: 28761868. I
- 731. Shibata Y. Efficacy of perioperative immunonutritional support with immunemodulating nutrients for postoperative infection-related complications, such as surgical site infection, in elective gastrointestinal cancer surgery. Journal of Clinical Oncology. 2011;29(1). PMID: CN-00830802. PT
- 732. Shintani Y, Ikeda N, Matsumoto T, et al. Nutritional status of patients undergoing chemoradiotherapy for lung cancer. Asian Cardiovascular & Thoracic Annals. 2012;20(2):172-6. doi: 10.1177/0218492311435249. PMID: 22499965. O
- 733. Shiozawa S, Usui T, Kuhara K, et al. Impact of Branched-Chain Amino Acid-Enriched Nutrient on liver Cirrhosis with Hepatocellular Carcinoma Undergoing Transcatheter Arterial Chemoembolization in Barcelona Clinic Liver Cancer Stage B: A Prospective Study. Journal of Nippon Medical School = Nihon Ika Daigaku Zasshi. 2016;83(6):248-56. doi: 10.1272/jnms.83.248. PMID: 28133005. I

- 734. Shirai Y, Okugawa Y, Hishida A, et al. Fish oilenriched nutrition combined with systemic chemotherapy for gastrointestinal cancer patients with cancer cachexia. Scientific Reports. 2017;7(1):4826. doi: 10.1038/s41598-017-05278-0. PMID: 625716763. S
- 735. Shirakawa H, Kinoshita T, Gotohda N, et al. Compliance with and effects of preoperative immunonutrition in patients undergoing pancreaticoduodenectomy. Journal of Hepato-biliary-pancreatic Sciences. 2012;19(3):249-58. doi: 10.1007/s00534-011-0416-3. PMID: 21667052. S
- 736. Shoar S, Naderan M, Mahmoodzadeh H, et al. Early Oral Feeding After Surgery for Upper Gastrointestinal Malignancies: A Prospective Cohort Study. Oman Medical Journal. 2016;31(3):182-7. doi: 10.5001/omj.2016.36. PMID: 27162588. I
- 737. Shumsky A, Bilan E, Sanz E, et al. Oncoxin nutritional supplement in the management of chemotherapy- and/or radiotherapyassociated oral mucositis. Molecular & Clinical Oncology. 2019;10(4):463-8. doi: 10.3892/mco.2019.1809. PMID: 30931119. SS
- 738. Shusterman M, Brar G, Klute K, et al. Phase II randomized controlled trial (RCT) of medical intensive nutrition therapy (MINT) to improve chemotherapy (CT) tolerability in malnourished patients with solid tumor malignancies. Journal of Clinical Oncology. 2020;38(1). doi: 10.1200/JCO.2020.38.15suppl.12090. PMID: CN-02355897. PT
- 739. Siddiqui AA, Glynn C, Loren D, et al. Selfexpanding plastic esophageal stents versus jejunostomy tubes for the maintenance of nutrition during neoadjuvant chemoradiation therapy in patients with esophageal cancer: a retrospective study. Diseases of the Esophagus. 2009;22(3):216-22. doi: 10.1111/j.1442-2050.2008.00905.x. PMID: 19207544. I
- 740. Sieron HL, Eberle F, Gress TM, et al. Safety of Prophylactic Gastrostomy Tube Placement and Gastrostomy Tube Usage in Patients Treated by Radio(chemo)therapy for Head and Neck Cancer. Anticancer Research. 2020;40(2):1167-73. doi: 10.21873/anticanres.14059. PMID: 32014970. C

- 741. Sierzega M, Choruz R, Pietruszka S, et al. Feasibility and outcomes of early oral feeding after total gastrectomy for cancer. Journal of Gastrointestinal Surgery. 2015;19(3):473-9. doi: 10.1007/s11605-014-2720-0. PMID: 25519083. C
- 742. Silander E, Nyman J, Bove M, et al. The use of prophylactic percutaneous endoscopic gastrostomy and early enteral feeding in patients with advanced head and neck cancer-A prospective longitudinal study. e-SPEN. 2010;5(4):e166-e72. doi: 10.1016/j.eclnm.2010.04.002. PMID: 50922617. C
- 743. Silvers MA, Savva J, Huggins CE, et al. Potential benefits of early nutritional intervention in adults with upper gastrointestinal cancer: a pilot randomised trial. Supportive Care in Cancer. 2014;22(1):3035-44. doi: 10.1007/s00520-014-2311-3. PMID: 24908429. SS
- 744. Slotwinski R, Olszewski WL, Krasnodebski IW, et al. Changes in interleukin-6 and cytokines antagonists serum concentrations in patients after pancreatic cancer surgery receiving nutritional support. Central-European Journal of Immunology. 2006;31(1):25-30. PMID: 44639174. C
- 745. Slotwinski R, Olszewski WL, Lech G, et al. Immunonutrition after major pancreatic surgery. Central-European Journal of Immunology. 2008;33(2):67-73. PMID: 351768886. O
- 746. Slotwinski R, Olszewski WL, Slotkowski M, et al. Can the interleukin-1 receptor antagonist (IL-1ra) be a marker of anti-inflammatory response to enteral immunonutrition in malnourished patients after pancreaticoduodenectomy? Journal of the Pancreas. 2007;8(6):759-69. PMID: 350200194. SS
- 747. Snegovoy AV, Larionova VB, Kononenko IB. Anorexia-cachexia syndrome in cancer patients: pathogenetic aspects and treatment options. Oncogematologiya. 2021;15(4):91-102. doi: 10.17650/1818-8346-2020-15-4-91-102. PMID: CN-02247874. NE

- 748. Soares JDP, Siqueira JM, Oliveira ICL, et al. A high-protein diet, not isolated BCAA, is associated with skeletal muscle mass index in patients with gastrointestinal cancer. Nutrition. 2020;72:110698. doi: 10.1016/j.nut.2019.110698. PMID: 32007808. I
- 749. Sobani ZU, Ghaffar S, Ahmed BN. Comparison of outcomes of enteral feeding via nasogastric versus gastrostomy tubes in post operative patients with a principle diagnosis of squamous cell carcinoma of the oral cavity. JPMA - Journal of the Pakistan Medical Association. 2011;61(1):1042-5. PMID: 22356051. I
- 750. Solheim TS, Laird BJA, Balstad TR, et al. A randomized phase II feasibility trial of a multimodal intervention for the management of cachexia in lung and pancreatic cancer. Journal of Cachexia, Sarcopenia and Muscle. 2017;8(5):778-88. doi: 10.1002/jcsm.12201. PMID: 28614627. I
- 751. Sommacal HM, Bersch VP, Vitola SP, et al. Perioperative synbiotics decrease postoperative complications in periampullary neoplasms: a randomized, double-blind clinical trial. Nutrition & Cancer. 2015;67(3):457-62. doi: 10.1080/01635581.2015.1004734. PMID: 25803626. I
- 752. Song J, Jing S, Shi H. The clinical observation of early oral feeding following total laryngectomy. Lin chuang er bi yan hou ke za zhi [Journal of clinical otorhinolaryngology]. 2003;17(9):527-8. PMID: CN-00559408. NE
- 753. Song JH, Ko J, Min YW, et al. Comparison between Percutaneous Gastrostomy and Self-Expandable Metal Stent Insertion for the Treatment of Malignant Esophageal Obstruction, after Propensity Score Matching. Nutrients. 2020;12(9):10. doi: 10.3390/nu12092756. PMID: 32927691. I
- 754. Sorensen D, McCarthy M, Baumgartner B, et al. Perioperative immunonutrition in head and neck cancer. Laryngoscope. 2009;119(7):1358-64. doi: 10.1002/lary.20494. PMID: 19459146. SS

- 755. Sorensen LS, Thorlacius-Ussing O, Rasmussen HH, et al. Effects of perioperative supplementation with omega-3 fatty acids on leukotriene B4 and leukotriene B5 production by stimulated neutrophils in patients with colorectal cancer: a randomized, placebo-controlled intervention trial. Nutrients. 2014;6(1):4043-57. doi: 10.3390/nu6104043. PMID: 25268838. O
- 756. Sornsuvit C, Komindr S, Chuncharunee S, et al. Pilot study: Effects of parenteral glutamine dipeptide supplementation on neutrophil functions and prevention of chemotherapyinduced side-effects in acute myeloid leukaemia patients. Journal of International Medical Research. 2008;36(6):1383-91. doi: 10.1177/147323000803600628. PMID: 354483168. SS
- 757. Soto-Lugo JH, Souto-Del Bosque MA, Vazquez-Martinez CA. Effectiveness of nutritional intervention in reduction of gastrointestinal toxicity during external beam radiotherapy in women with gynecological tumors. Gaceta Mexicana de Oncologia. 2017;16(2):84-90. doi: 10.24875/j.gamo.17000016. PMID: 621289191. SS
- 758. Soto-Lugo JH, Souto-Del-Bosque MA, Vazquez Martinez CA. Effectiveness of a nutritional intervention in the reduction of gastrointestinal toxicity during teletherapy in women with gynaecological tumours. Revista Medica del Hospital General de Mexico. 2018;81(1):7-14. doi: 10.1016/j.hgmx.2017.03.005. PMID: 618014911. SS
- 759. Sousa AA, Porcaro-Salles JM, Soares JM, et al. Tolerance of early oral feeding in patients subjected to total laryngectomy. Head & Neck. 2016;38:E643-8. doi: 10.1002/hed.24063. PMID: 25832556. O
- 760. Souza APS, Silva LCD, Fayh APT. Nutritional Intervention Contributes to the Improvement of Symptoms Related to Quality of Life in Breast Cancer Patients Undergoing Neoadjuvant Chemotherapy: A Randomized Clinical Trial. Nutrients. 2021;13(2):10. doi: 10.3390/nu13020589. PMID: 33579050. SS

- 761. Sowerbutts AM, Lal S, Sremanakova J, et al. Palliative home parenteral nutrition in patients with ovarian cancer and malignant bowel obstruction: experiences of women and family caregivers. BMC Palliative Care. 2019;18(1):120. doi: 10.1186/s12904-019-0507-5. PMID: 31884962. I
- 762. Srinathan SK, Hamin T, Walter S, et al. Jejunostomy tube feeding in patients undergoing esophagectomy. Canadian Journal of Surgery. 2013;56(6):409-14. PMID: 24284149. C
- 763. Steensma DP, Sloan JA, Dakhil SR, et al. Phase III, randomized study of the effects of parenteral iron, oral iron, or no iron supplementation on the erythropoietic response to darbepoetin alfa for patients with chemotherapy-associated anemia. Journal of Clinical Oncology. 2011;29(1):97-105. doi: 10.1200/jco.2010.30.3644. PMID: CN-00771200. I
- 764. Storck LJ, Ruehlin M, Gaeumann S, et al. Effect of a leucine-rich supplement in combination with nutrition and physical exercise in advanced cancer patients: A randomized controlled intervention trial. Clinical Nutrition. 2020;39(1):3637-44. doi: 10.1016/j.clnu.2020.04.008. PMID: 32340904. I
- 765. St-Pierre J, Drummond K, Minella E, et al. Feasibility of multimdodal prehabilitation to enhance preoperative functional capacity of esophageal cancer patients during concurrent neoadjuvant chemotherapies - a pilot interventional study. European journal of surgical oncology. 2022;48(2):e38-. doi: 10.1016/j.ejso.2021.12.450. PMID: CN-02373824. PT
- 766. Su D, He Y, Chen L, et al. Nutrition counseling combined with head and neck rehabilitation exercises can enhance outcomes among nasopharyngeal carcinoma patients in southern China: a prospective study in an epidemic area. Annals of Palliative Medicine. 2020;9(3):1152-63. doi: 10.21037/apm-20-1053. PMID: 32498530. I
- 767. Suehiro T, Matsumata T, Shikada Y, et al. Accelerated rehabilitation with early postoperative oral feeding following gastrectomy. Hepato-Gastroenterology. 2004;51(6):1852-5. PMID: 15532842. S

- 768. Sugisawa N, Tokunaga M, Makuuchi R, et al. A phase II study of an enhanced recovery after surgery protocol in gastric cancer surgery. Gastric Cancer. 2016;19(3):961-7. doi: 10.1007/s10120-015-0528-6. PMID: 26260875. I
- 769. Sukaraphat N, Chewaskulyong B, Buranapin S. Dietary Counseling Outcomes in Locally Advanced Unresectable or Metastatic Cancer Patients Undergoing Chemotherapy. Journal of the Medical Association of Thailand. 2016;99(1):1283-90. PMID: 29952503. SS
- 770. Sun HB, Liu XB, Zhang RX, et al. Early oral feeding following thoracolaparoscopic oesophagectomy for oesophageal cancer. European Journal of Cardio-Thoracic Surgery. 2015;47(2):227-33. doi: 10.1093/ejcts/ezu168. PMID: 24743002. C
- 771. Sun QY, Zhou JY, Du LY. Effect of targeted nutrition intervention on defecation and postoperative complications in patients undergoing radical resection for rectal cancer. World Chinese Journal of Digestology. 2018;26(2):1729-34. doi: 10.11569/wcjd.v26.i29.1729. PMID: CN-01668638. NE
- 772. Sun XN, Guo Y, Zhu ML. Effects of probiotics on immune function in postoperative colorectal cancer patients receiving adjuvant chemotherapy. Journal of practical oncology. 2012;27(6):610-2. PMID: CN-00915282. NE
- 773. Sun YB, Li YL, Li WM, et al. Effect of appetite-conditioned reflex stimulation on early enteral nutrition tolerance after surgery. Acta Gastroenterologica Belgica. 2020;83(4):527-31. PMID: 33321007. I
- 774. Sun Z, Shenoi MM, Nussbaum DP, et al. Feeding jejunostomy tube placement during resection of gastric cancers. Journal of Surgical Research. 2016;200(1):189-94. doi: 10.1016/j.jss.2015.07.014. PMID: 26248478. S
- 775. Sun ZW, Jia J, Yang Y, et al. [Enteral nutrition support reduces toxicity of chemotherapy in patients with advanced or metastatic esophageal cancer]. 2020;52(2):261-8. PMID: 32306008. NE

- 776. Suslu N, Sefik Hosal A. Early oral feeding after total laryngectomy: Outcome of 602 patients in one cancer center. Auris, Nasus, Larynx. 2016;43(5):546-50. doi: 10.1016/j.anl.2016.01.004. PMID: 26908188. I
- 777. Suzuki D, Furukawa K, Kimura F, et al. Effects of perioperative immunonutrition on cell-mediated immunity, T helper type 1 (Th1)/Th2 differentiation, and Th17 response after pancreaticoduodenectomy. Surgery. 2010;148(3):573-81. doi: 10.1016/j.surg.2010.01.017. PMID: 20227099. SS
- 778. Svanfeldt M, Thorell A, Nygren J, et al. Postoperative parenteral nutrition while proactively minimizing insulin resistance. Nutrition (Burbank, Los Angeles County, Calif.). 2006;22(5):457-64. doi: 10.1016/j.nut.2005.06.013. PMID: CN-00562577. O
- 779. Sykorova A, Horacek J, Zak P, et al. A randomized, double blind comparative study of prophylactic parenteral nutritional support with or without glutamine in autologous stem cell transplantation for hematological malignancies -- three years' follow-up. Neoplasma. 2005;52(6):476-82. PMID: 16284692. SS
- Szefel J, Kruszewski WJ, Ciesielski M, et al. Lcarnitine and cancer cachexia. I. L-carnitine distribution and metabolic disorders in cancer cachexia. Oncology Reports. 2012;28(1):319-23. doi: 10.3892/or.2012.1804. PMID: 22562434. I
- 781. Szefel J, Kruszewski WJ, Szajewski M, et al. Medium-chain triglycerides/long-chain triglycerides versus long-chain triglycerides in treatment of cancer patients with major body mass loss. Survival in patients with refractory cachexia. Przeglad Gastroenterologiczny. 2016;11(3):181-6. doi: 10.5114/pg.2016.57734. PMID: 612228940. O
- 782. Tabrizi R, Hosseinpour S, Taghizadeh F. Feeding in Oral Cancer Patients After Massive Ablative Surgery: Percutaneous Endoscopic Gastrostomy or Nasogastric Tube. Journal of Craniofacial Surgery. 2016;27(4):1010-1. doi: 10.1097/scs.00000000002662. PMID: 27228377. I

- 783. Takada K, Shimokawa M, Takamori S, et al. Clinical impact of probiotics on the efficacy of anti-PD-1 monotherapy in patients with nonsmall cell lung cancer: A multicenter retrospective survival analysis study with inverse probability of treatment weighting. International Journal of Cancer. 2021;15:15. doi: 10.1002/ijc.33557. PMID: 33720422. I
- 784. Takahashi M, Kosaka N, Wakui E, et al. Role of intensive nutrition support and prophylactic percutaneous endoscopic gastrostomy during concomitant chemoradiotherapy for oropharyngeal cancer. International Journal of Clinical Oncology. 2018;23(6):1023-8. doi: 10.1007/s10147-018-1328-x. PMID: 30121869. I
- 785. Takahashi M, Takemoto N, Sano A, et al. Effectiveness of prophylactic percutaneous endoscopic gastrostomy on nutritional status and mucositis in oropharyngeal cancer patients undergoing concurrent chemoradiotherapy. Japanese Journal of Head and Neck Cancer. 2012;38(3):336-42. doi: 10.5981/jjhnc.38.336. PMID: 366024819. NE
- 786. Takeda H, Nishikawa H, Iguchi E, et al. Effect of treatment with branched-chain amino acids during sorafenib therapy for unresectable hepatocellular carcinoma. Hepatology Research. 2014;44(3):302-12. doi: 10.1111/hepr.12125. PMID: 52569229. S
- 787. Takesue T, Takeuchi H, Ogura M, et al. A Prospective Randomized Trial of Enteral Nutrition After Thoracoscopic Esophagectomy for Esophageal Cancer. Annals of Surgical Oncology. 2015;22:S802-9. doi: 10.1245/s10434-015-4767-x. PMID: 26219242. SS
- 788. Talvas J, Garrait G, Goncalves-Mendes N, et al. Immunonutrition stimulates immune functions and antioxidant defense capacities of leukocytes in radiochemotherapy-treated head & neck and esophageal cancer patients: A double-blind randomized clinical trial. Clinical Nutrition. 2015;34(5):810-7. doi: 10.1016/j.clnu.2014.12.002. PMID: 25575640. O

- 789. Tan TC, Ye CN, Fang Q, et al. Effect of different nutritional support modes on humoral immunity and outcomes after esophagectomy. Chinese journal of clinical nutrition. 2011;19(6):372-6. doi: 10.3760/cma.j.issn.1674-635X.2011.06.005. PMID: CN-00903825. NE
- 790. Tanaka K, Yano M, Motoori M, et al. Impact of perioperative administration of synbiotics in patients with esophageal cancer undergoing esophagectomy: a prospective randomized controlled trial. Surgery. 2012;152(5):832-42. doi: 10.1016/j.surg.2012.02.021. PMID: 22503510. I
- 791. Tanaka N, Takeda K, Kawasaki Y, et al. Early intensive nutrition intervention with dietary counseling and oral nutrition supplement prevents weight loss in patients with advanced lung cancer receiving chemotherapy: A clinical prospective study. Yonago Acta Medica. 2018;61(4):204-12. doi: 10.33160/yam.2018.12.003. PMID: 2001542398. C
- 792. Tang J, Xu W. Effect of postoperative enteral and parenteral nutrition on the immunity and inflammatory reaction of esophageal cancer patients. Journal of practical oncology. 2019;34(2):155-9. doi: 10.13267/j.cnki.syzlzz.2019.02.012. PMID: CN-01959288. NE
- 793. Taniguchi H, Sasaki T, Fujita H, et al. Effects of goal-directed fluid therapy on enhanced postoperative recovery: An interventional comparative observational study with a historical control group on oesophagectomy combined with ERAS program. Clinical Nutrition ESPEN. 2018;23:184-93. doi: 10.1016/j.clnesp.2017.10.002. PMID: 29460796. C
- 794. Tesauro M, Guida AM, Siragusa L, et al. Preoperative immunonutrition vs. Standard dietary advice in normo-nourished patients undergoing fast-track laparoscopic colorectal surgery. Journal of Clinical Medicine. 2021;10(3):1-11. doi: 10.3390/jcm10030413. PMID: 2005861146. S

- 795. Tessier W, Piessen G, Briez N, et al. Percutaneous radiological gastrostomy in esophageal cancer patients: a feasible and safe access for nutritional support during multimodal therapy. Surgical Endoscopy. 2013;27(2):633-41. doi: 10.1007/s00464-012-2506-y. PMID: 22956002. S
- 796. Thodiyil PA, El-Masry NS, Peake H, et al. Ttube jejunostomy feeding after pancreatic surgery: a safe adjunct. Asian Journal of Surgery. 2004;27(2):80-4. PMID: 15140657. C
- 797. Thomson CA, Crane TE, Miller A, et al. A randomized trial of diet and physical activity in women treated for stage II-IV ovarian cancer: Rationale and design of the Lifestyle Intervention for Ovarian Cancer Enhanced Survival (LIVES): An NRG Oncology/Gynecologic Oncology Group (GOG-225) Study. Contemporary Clinical Trials. 2016;49:181-9. doi: 10.1016/j.cct.2016.07.005. PMID: 27394382. I
- 798. Tomaszek SC, Cassivi SD, Allen MS, et al. An alternative postoperative pathway reduces length of hospitalisation following oesophagectomy. European Journal of Cardio-Thoracic Surgery. 2010;37(4):807-13. doi: 10.1016/j.ejcts.2009.09.034. PMID: 19900819. I
- 799. Torricelli P, Antonelli F, Ferorelli P, et al. Oral nutritional supplement prevents weight loss and reduces side effects in patients in advanced lung cancer chemotherapy. Amino Acids. 2020;52(3):445-51. doi: 10.1007/s00726-020-02822-7. PMID: 32034492. S
- 800. Toure A, Chambrier C, Vanhems P, et al. Propensity score analysis confirms the independent effect of parenteral nutrition on the risk of central venous catheter-related bloodstream infection in oncological patients. Clinical Nutrition. 2013;32(6):1050-4. doi: 10.1016/j.clnu.2012.12.006. PMID: 23313357. S

- 801. Toyomasu Y, Mochiki E, Yanai M, et al. A prospective pilot study of an elemental nutritional supplement for prevention of oral mucositis during S-1 adjuvant chemotherapy for gastric cancer. Surgical Oncology. 2019;29:97-101. doi: 10.1016/j.suronc.2019.04.003. PMID: 31196501. SS
- 802. Trabal J, Leyes P, Forga M, et al. Potential usefulness of an EPA-enriched nutritional supplement on chemotherapy tolerability in cancer patients without overt malnutrition. Nutricion Hospitalaria. 2010;25(5):736-40. PMID: 21336429. SS
- 803. Trachootham D, Songkaew W, Hongsachum B, et al. Nutri-jelly may improve quality of life and decrease tube feeding demand in head and neck cancer patients. Supportive Care in Cancer. 2015;23(5):1421-30. doi: 10.1007/s00520-014-2488-5. PMID: 25370890. S
- 804. Trestini I, Carbognin L, Sperduti I, et al. Prognostic impact of early nutritional support in patients affected by locally advanced and metastatic pancreatic ductal adenocarcinoma undergoing chemotherapy. European Journal of Clinical Nutrition. 2018;72(5):772-9. doi: 10.1038/s41430-018-0155-5. PMID: 29581564. C
- 805. Trinidad Ruiz G, Luengo Pérez LM, Marcos García M, et al. Value of nutritional support in patients with pharingocutaneous fistula. Acta Otorrinolaringologica Espanola. 2005;56(1):25-30. doi: 10.1016/s0001-6519(05)78566-6. PMID: CN-00513652. NE
- 806. Tsuda M, Ishiguro H, Toriguchi N, et al. Overnight fasting before lapatinib administration to breast cancer patients leads to reduced toxicity compared with nighttime dosing: a retrospective cohort study from a randomized clinical trial. Cancer Medicine. 2020;9(2):9246-55. doi: 10.1002/cam4.3528. PMID: 2007045183. S
- 807. Tu H, Dinney CP, Ye Y, et al. Is folic acid safe for non-muscle-invasive bladder cancer patients? An evidence-based cohort study. American Journal of Clinical Nutrition. 2018;107(2):208-16. doi: 10.1093/ajcn/nqx019. PMID: 29529165. I

- 808. Turnock A, Calder PC, West AL, et al. Perioperative immunonutrition in wellnourished patients undergoing surgery for head and neck cancer: evaluation of inflammatory and immunologic outcomes. Nutrients. 2013;5(4):1186-99. doi: 10.3390/nu5041186. PMID: 23571650. SS
- 809. Usami M, Miyoshi M, Kanbara Y, et al. Effects of perioperative synbiotic treatment on infectious complications, intestinal integrity, and fecal flora and organic acids in hepatic surgery with or without cirrhosis. Jpen: Journal of Parenteral & Enteral Nutrition. 2011;35(3):317-28. doi: 10.1177/0148607110379813. PMID: 21527594. I
- 810. Uster A, Ruehlin M, Mey S, et al. Effects of nutrition and physical exercise intervention in palliative cancer patients: A randomized controlled trial. Clinical Nutrition. 2018;37(4):1202-9. doi: 10.1016/j.clnu.2017.05.027. PMID: 28651827. I
- 811. Vaithiswaran V, Srinivasan K, Kadambari D. Effect of early enteral feeding after upper gastrointestinal surgery. Tropical Gastroenterology. 2008;29(2):91-4. PMID: 18972768. P
- 812. Valadares F, Garbi Novaes MR, Canete R. Effect of Agaricus sylvaticus supplementation on nutritional status and adverse events of chemotherapy of breast cancer: a randomized, placebo-controlled, double-blind clinical trial. Indian Journal of Pharmacology. 2013;45(3):217-22. doi: 10.4103/0253-7613.111894. PMID: 23833361. SS
- 813. Valentini V, Marazzi F, Bossola M, et al. Nutritional counselling and oral nutritional supplements in head and neck cancer patients undergoing chemoradiotherapy. Journal of Human Nutrition & Dietetics. 2012;25(3):201-8. doi: 10.1111/j.1365-277X.2011.01220.x. PMID: 22257023. C
- 814. Van Blarigan EL, Zhang S, Ou FS, et al. Association of Diet Quality With Survival Among People With Metastatic Colorectal Cancer in the Cancer and Leukemia B and Southwest Oncology Group 80405 Trial. JAMA Network Open. 2020;3(1):e2023500. doi: 10.1001/jamanetworkopen.2020.23500. PMID: 33125497. I

- 815. Van Bokhorst-De Van Der Schueren MAE, Langendoen SI, Vondeling H, et al. Perioperative enteral nutrition and quality of life of severely malnourished head and neck cancer patients: A randomized clinical trial. Clinical Nutrition. 2000;19(6):437-44. doi: 10.1054/clnu.2000.0148. PMID: 30994514. SS
- 816. Van Bokhorst-De Van Der Schueren MAE, Quak JJ, Von Blomberg-Van Der Flier BME, et al. Effect of perioperative nutrition, with and without arginine supplementation, on nutritional status, immune function, postoperative morbidity, and survival in severely malnourished head and neck cancer patients. American Journal of Clinical Nutrition. 2001;73(2):323-32. doi: 10.1093/ajcn/73.2.323. PMID: 32109863. SS
- 817. van den Berg MG, Kalf JG, Hendriks JC, et al. Normalcy of food intake in patients with head and neck cancer supported by combined dietary counseling and swallowing therapy: A randomized clinical trial. Head & Neck. 2016;38:E198-206. doi: 10.1002/hed.23970. PMID: 25533021. P
- 818. van den Berg MG, Rasmussen-Conrad EL, Wei KH, et al. Comparison of the effect of individual dietary counselling and of standard nutritional care on weight loss in patients with head and neck cancer undergoing radiotherapy. British Journal of Nutrition. 2010;104(6):872-7. doi: 10.1017/s0007114510001315. PMID: 20441684. S
- 819. van der Meij BS, Langius JA, Smit EF, et al. Oral nutritional supplements containing (n-3) polyunsaturated fatty acids affect the nutritional status of patients with stage III non-small cell lung cancer during multimodality treatment. Journal of Nutrition. 2010;140(1):1774-80. doi: 10.3945/jn.110.121202. PMID: 20739445. SS
- 820. van der Meij BS, Langius JA, Spreeuwenberg MD, et al. Oral nutritional supplements containing n-3 polyunsaturated fatty acids affect quality of life and functional status in lung cancer patients during multimodality treatment: an RCT. European Journal of Clinical Nutrition. 2012;66(3):399-404. doi: 10.1038/ejcn.2011.214. PMID: 22234041. SS

- 821. van der Werf A, Blauwhoff-Buskermolen S, Langius JA, et al. The effect of individualized nutritional counseling on muscle mass and treatment outcome in patients with metastatic colorectal cancer undergoing chemotherapy: a randomized controlled trial protocol. BMC Cancer. 2015;15:98. doi: 10.1186/s12885-015-1092-5. PMID: 25884881. PT
- 822. Van Dyck E, Macken EJ, Roth B, et al. Safety of pull-type and introducer percutaneous endoscopic gastrostomy tubes in oncology patients: a retrospective analysis. BMC Gastroenterology. 2011;11:23. doi: 10.1186/1471-230x-11-23. PMID: 21410958. S
- 823. van Lieshout R, Tick LW, Dieleman JP, et al. Changes in body weight and serum liver tests associated with parenteral nutrition compared with no parenteral nutrition in patients with acute myeloid leukemia during remission induction treatment. Supportive Care in Cancer. 2020;28(9):4381-93. doi: 10.1007/s00520-019-05251-9. PMID: 31916008. S
- 824. van Rooijen SJ, Molenaar CJL, Schep G, et al. Making Patients Fit for Surgery: Introducing a Four Pillar Multimodal Prehabilitation Program in Colorectal Cancer. American Journal of Physical Medicine & Rehabilitation. 2019;98(1):888-96. doi: 10.1097/phm.000000000001221. PMID: 31090551. I
- 825. van Stijn MFM, Soeters MR, van Leeuwen PAM, et al. Effects of a Carbohydrate-, Glutamine-, and Antioxidant-Enriched Oral Nutrition Supplement on Major Surgery-Induced Insulin Resistance: A Randomized Pilot Study. Jpen: Journal of Parenteral & Enteral Nutrition. 2018;42(4):719-29. doi: 10.1177/0148607117711691. PMID: 28541810. O
- 826. van Tiel FH, Harbers MM, Terporten PHW, et al. Normal hospital and low-bacterial diet in patients with cytopenia after intensive chemotherapy for hematological malignancy: A study of safety. Annals of Oncology. 2007;18(6):1080-4. doi: 10.1093/annonc/mdm082. PMID: 47050500. SS

- 827. Vashi P, Popiel B, Lammersfeld C, et al. Outcomes of systematic nutritional assessment and medical nutrition therapy in pancreatic cancer. Pancreas. 2015;44(5):750-5. doi: 10.1097/mpa.00000000000336. PMID: 25872172. I
- 828. Vashi PG, Gupta D, Lammersfeld CA, et al. The relationship between baseline nutritional status with subsequent parenteral nutrition and clinical outcomes in cancer patients undergoing hyperthermic intraperitoneal chemotherapy. Nutrition Journal. 2013;12:118. doi: 10.1186/1475-2891-12-118. PMID: 23941331. S
- 829. Vasson MP, Talvas J, Perche O, et al. Immunonutrition improves functional capacities in head and neck and esophageal cancer patients undergoing radiochemotherapy: a randomized clinical trial. Clinical Nutrition. 2014;33(2):204-10. doi: 10.1016/j.clnu.2013.06.008. PMID: 23849811. SS
- 830. Vidhya C, Phoebe D, Dhina C, et al. Percutaneous endoscopic gastrostomy (PEG) versus radiologically inserted gastrostomy (RIG): A comparison of outcomes at an Australian teaching hospital. Clinical Nutrition ESPEN. 2018;23:136-40. doi: 10.1016/j.clnesp.2017.10.014. PMID: 29460789. S
- 831. Vilar Gomez E, Sanchez Rodriguez Y, Torres Gonzalez A, et al. Viusid, a nutritional supplement, increases survival and reduces disease progression in HCV-related decompensated cirrhosis: a randomised and controlled trial. BMJ Open. 2011;1(2):e000140. doi: 10.1136/bmjopen-2011-000140. PMID: 22021873. P
- 832. Villar-Taibo R, Calleja-Fernandez A, Vidal-Casariego A, et al. A short nutritional intervention in a cohort of hematological inpatients improves energy and protein intake and stabilizes nutritional status. Nutricion Hospitalaria. 2016;33(6):1347-53. doi: 10.20960/nh.794. PMID: 28000464. C
- 833. Volling P, Singelmann H, Ebeling O. Incidence of salivary fistulas in relation to timing of oral nutrition after laryngectomy. HNO. 2001;49(4):276-82. doi: 10.1007/s001060050746. PMID: CN-00456140. X

- 834. Volpe S, Marvaso G, Alterio D, et al. Nutritional Intervention for Nonsurgical Head and Neck Cancer Patients Treated with Radiation Therapy: Results from a Prospective Stepped-Wedge Clinical Protocol. Nutrition & Cancer. 2018;70(7):1051-9. doi: 10.1080/01635581.2018.1497187. PMID: 30273004. C
- 835. von Grundherr J, Koch B, Grimm D, et al. Impact of taste and smell training on taste disorders during chemotherapy - TASTE trial. Cancer management and research. 2019;11:4493-504. doi: 10.2147/cmar.S188903. PMID: 31191011. I
- 836. Voskuilen CS, van de Putte EEF, der Hulst JB, et al. Short-term outcome after cystectomy: comparison of early oral feeding in an enhanced recovery protocol and feeding using Bengmark nasojejunal tube. World Journal of Urology. 2018;36(2):221-9. doi: 10.1007/s00345-017-2133-2. PMID: 29167985. I
- 837. Voss M, Wenger KJ, von Mettenheim N, et al. Short-term fasting in glioma patients: analysis of diet diaries and metabolic parameters of the ERGO2 trial. European Journal of Nutrition. 2022;61(1):477-87. doi: 10.1007/s00394-021-02666-1. PMID: 34487222. SS
- 838. Vyawahare MA, Shirodkar M, Gharat A, et al. A comparative observational study of early versus delayed feeding after percutaneous endoscopic gastrostomy. Indian Journal of Gastroenterology. 2013;32(6):366-8. doi: 10.1007/s12664-013-0348-8. PMID: 23949988. S
- 839. Wang D, Zhang H, Zhang Y, et al. Effects of omega-3 polyunsaturated fatty acids on postoperative inflammatory reaction and clinical efficacy. Zhonghua wei chang wai ke za zhi [Chinese journal of gastrointestinal surgery]. 2015;18(7):651-5. PMID: CN-01131812. NE
- 840. Wang D, Zhong B, Zhao P, et al. A randomized control study of early oral enteral nutrition after colorectal cancer operation. Zhonghua wei chang wai ke za zhi [Chinese journal of gastrointestinal surgery]. 2014;17(1):977-80. PMID: CN-01112898. NE

- 841. Wang H, Hao Q, Wang M, et al. Esophagojejunostomy after laparoscopic total gastrectomy by OrVilTM or hemidouble stapling technique. World Journal of Gastroenterology. 2015;21(2):8858-67. doi: 10.3748/wjg.v21.i29.8943. PMID: 605744732. S
- 842. Wang HX, Li JP. Effects of modified bazhen decoction in assistant with enteral nutrition on the growth hormone, the nutritional state, and the immune function in patients with gastric cancer after operation. Zhongguo zhong xi yi jie he za zhi zhongguo zhongxiyi jiehe zazhi = chinese journal of integrated traditional and western medicine. 2011;31(1):1317-21. PMID: CN-00919611. NE
- 843. Wang HX, Xia Y, Shao SY. Influence of enteral nutrition during the preoperative and postoperative periods on postoperative nutritional status and immunologic function in patients with gastric cancer. Journal of xi'an jiaotong university (medical sciences). 2011;32(3):375-8. PMID: CN-00894419. NE
- 844. Wang J, Yang M, Wang Q, et al. Comparison of Early Oral Feeding With Traditional Oral Feeding After Total Gastrectomy for Gastric Cancer: A Propensity Score Matching Analysis. Frontiers in Oncology. 2019;9:1194. doi: 10.3389/fonc.2019.01194. PMID: 31788451. S
- 845. Wang R, Cai H, Li Y, et al. Impact Exerted by Nutritional Risk Screening on Clinical Outcome of Patients with Esophageal Cancer. BioMed Research International. 2018;2018:7894084. doi: 10.1155/2018/7894084. PMID: 29780831. S
- 846. Wang SM, Taylor PR, Fan JH, et al. Effects of Nutrition Intervention on Total and Cancer Mortality: 25-Year Post-trial Follow-up of the 5.25-Year Linxian Nutrition Intervention Trial. Journal of the National Cancer Institute. 2018;110(1):1229-38. doi: 10.1093/jnci/djy043. PMID: 29617851. P
- 847. Wang WK, Tu CY, Shao CX, et al. Impact of enhanced recovery after surgery on postoperative rehabilitation, inflammation, and immunity in gastric carcinoma patients: a randomized clinical trial. Brazilian Journal of Medical & Biological Research. 2019;52(5):e8265. doi: 10.1590/1414-431x20198265. PMID: 31116313. I

- 848. Wang WP, Yan XL, Ni YF, et al. Effects of lipid emulsions in parenteral nutrition of esophageal cancer surgical patients receiving enteral nutrition: a comparative analysis. Nutrients. 2013 Dec 27;6(1):111-23. doi: 10.3390/nu6010111. PMID: 24379010. D
- 849. Wang XY, Huang WN, Wang LY, et al. Effects of glutamine on plasma endotoxin and immune function in postperative patients with laryngeal carcinoma. Chinese journal of clinical nutrition. 2006;14(4):222-6. PMID: CN-00613230. NE
- 850. Wang XY, Li N, Gu J, et al. The effects of the formula of amino acids enriched BCAA on nutritional support in traumatic patients. World Journal of Gastroenterology. 2003;9(3):599-602. doi: 10.3748/wjg.v9.i3.599. PMID: 36372134. P
- 851. Wang Z, Chen JX, Wang PZ. Effect of compound branch chain amino acid injection on nutritional support in patients after radical resection for colorectal cancer. Zhonghua wei chang wai ke za zhi [Chinese journal of gastrointestinal surgery]. 2006;9(5):399-401. PMID: CN-00721257. NE
- 852. Wang ZD, Peng JS, Chen S, et al. Effects of perioperative enteral immunonutrition on nutritional status, immunity and inflammatory response of elderly patients. Zhonghua yi xue za zhi. 2006;86(2):1410-3. PMID: CN-00585760. NE
- 853. Wang ZH, Zhong B, Xiang JY, et al. Effect of early oral enteral nutrition on clinical outcomes after colorectal cancer surgery. Zhonghua wei chang wai ke za zhi [Chinese journal of gastrointestinal surgery]. 2013;16(8):735-8. PMID: CN-01121818. NE
- 854. Wei J, Wu J, Meng L, et al. Effects of early nutritional intervention on oral mucositis in patients with radiotherapy for head and neck cancer. Qjm. 2020;113(1):37-42. doi: 10.1093/qjmed/hcz222. PMID: 31432089. S

- 855. Wei J, Wu J, Wang H, et al. A Bioadhesive Barrier-Forming Oral Liquid Gel Improved Oral Mucositis and Nutritional Status in Patients With Head and Neck Cancers Undergoing Radiotherapy: A Retrospective Single Center Study. Frontiers in Oncology. 2021;11:617392. doi: 10.3389/fonc.2021.617392. PMID: 33692954. I
- 856. Wei Z, Wang W, Chen J, et al. A prospective, randomized, controlled study of omega-3 fish oil fat emulsion-based parenteral nutrition for patients following surgical resection of gastric tumors. Nutrition Journal. 2014;13:25. doi: 10.1186/1475-2891-13-25. PMID: 24655407. SS
- 857. Weijs TJ, van Eden HWJ, Ruurda JP, et al. Routine jejunostomy tube feeding following esophagectomy. Journal of Thoracic Disease. 2017;9(S):S851-S60. doi: 10.21037/jtd.2017.06.73. PMID: 617532529. C
- 858. Wendel M, Rossel T, Bergmann S, et al. Impact of total parenteral nutrition including omega-3 fatty acids on the regulation of plasma lipoproteins and glycemic control after major abdominal surgery. e-SPEN. 2007;2(5):e103-e10. doi: 10.1016/j.eclnm.2007.06.002. PMID: 47489005. O
- 859. Wenger KJ, Wagner M, Harter PN, et al. Maintenance of energy homeostasis during calorically restricted ketogenic diet and fastingmr-spectroscopic insights from the ergo2 trial. Cancers. 2020;12(1):1-16. doi: 10.3390/cancers12123549. PMID: 2005519701. O
- 860. Werner K, Kullenberg de Gaudry D, Taylor LA, et al. Dietary supplementation with n-3fatty acids in patients with pancreatic cancer and cachexia: marine phospholipids versus fish oil - a randomized controlled doubleblind trial. Lipids in Health & Disease. 2017;16(1):104. doi: 10.1186/s12944-017-0495-5. PMID: 28578704. SS
- 861. Wesselink E, Kok DE, Bours MJL, et al. Vitamin D, magnesium, calcium, and their interaction in relation to colorectal cancer recurrence and all-cause mortality. American Journal of Clinical Nutrition. 2020;111(5):1007-17. doi: 10.1093/ajcn/nqaa049. PMID: 32190892. I

- 862. White KL, Henson CC, Hann M, et al. Randomised clinical trial of a gastrointestinal care bundle to reduce symptoms in patients with pelvic cancer undergoing chemoradiotherapy. BMJ Open Gastroenterology. 2020;7(1):08. doi: 10.1136/bmjgast-2020-000432. PMID: 32771983. SS
- 863. Wierdak M, Surmiak M, Milian-Ciesielska K, et al. Immunonutrition Changes Inflammatory Response in Colorectal Cancer: Results from a Pilot Randomized Clinical Trial. Cancers. 2021;13(6):22. doi: 10.3390/cancers13061444. PMID: 33809994. O
- 864. Williams GF, Teo MT, Sen M, et al. Enteral feeding outcomes after chemoradiotherapy for oropharynx cancer: a role for a prophylactic gastrostomy? Oral Oncology. 2012;48(5):434-40. doi: 10.1016/j.oraloncology.2011.11.022. PMID: 22209648. I
- 865. Winters BL, Mitchell DC, Smiciklas-Wright H, et al. Dietary patterns in women treated for breast cancer who successfully reduce fat intake: The women's intervention nutrition study (WINS). Journal of the American Dietetic Association. 2004;104(4):551-9. doi: 10.1016/j.jada.2004.01.012. PMID: 38443376. P
- 866. Wright JL, Plymate S, D'Oria-Cameron A, et al. A study of caloric restriction versus standard diet in overweight men with newly diagnosed prostate cancer: a randomized controlled trial. Prostate. 2013;73(1):1345-51. doi: 10.1002/pros.22682. PMID: 23775525. SS
- 867. Wu GH, Zhang YW, Pan HT, et al. A randomized controlled trial of postoperative artificial nutrition in malnourished patients with gastrointestinal cancer. Zhonghua wei chang wai ke za zhi [Chinese journal of gastrointestinal surgery]. 2007;10(6):546-9. PMID: CN-00729498. NE
- 868. Wu G-Z, Gao S-Q, Mao G-J, et al. Clinical efficacy of different nutritional methods in patients after laparoscopic radical surgery for distal gastric cancer. World Chinese Journal of Digestology. 2020;28(1):898-903. doi: 10.11569/wcjd.v28.i18.898. PMID: CN-02236015. NE

- 869. Wu MH, Lin MT, Chen WJ. Effect of perioperative parenteral nutritional support for gastric cancer patients undergoing gastrectomy. Hepato-Gastroenterology. 2008;55(8):799-802. PMID: 18613458. S
- 870. Wu PR, Xu L, Zhang ZM. Comparative study of postoperative early enteral nutrition and parenteral nutrition in esophageal carcinoma. Zhonghua wei chang wai ke za zhi [Chinese journal of gastrointestinal surgery]. 2006;9(4):320-2. PMID: CN-00723149. NE
- 871. Wu Q, Yu JC, Kang WM, et al. Short-term effects of supplementary feeding with enteral nutrition via jejunostomy catheter on post-gastrectomy gastric cancer patients. Chinese Medical Journal. 2011;124(2):3297-301. doi: 10.3760/cma.j.issn.0366-6999.2011.20.017. PMID: 362794505. C
- 872. Wu XN, Cui Y, Yang J, et al. Clinical observation of enteral nutrition in postoperative patient with laryngeal squamous cell carcinoma. Chinese journal of clinical nutrition. 2007;15(3):152-4. PMID: CN-01019254. NE
- 873. Wu Z, Wu M, Wang Q, et al. Home enteral nutrition after minimally invasive esophagectomy can improve quality of life and reduce the risk of malnutrition. Asia Pacific Journal of Clinical Nutrition. 2018;27(1):129-36. doi: 10.6133/apjcn.032017.22. PMID: 29222890. I
- 874. Wuryanti S, Andrijono, Susworo, et al. The Effect of High Poly Unsaturated Fatty Acid (PUFA) Dietary Supplementation on Inflammatory Status of Patients with Advanced Cervical Cancer on Radiation Treatment. Acta Medica Indonesiana. 2015;47(1):45-9. PMID: 25948767. O
- 875. Xie H, Lu Q, Wang H, et al. Effects of probiotics combined with enteral nutrition on immune function and inflammatory response in postoperative patients with gastric cancer. Journal of B.U.On. 2018;23(3):678-83. PMID: 30003737. I
- 876. Xin Y, Cai H, Wu L, et al. The Effect of Immunonutrition on the Postoperative Complications in Thymoma with Myasthenia Gravis. Mediators of Inflammation. 2016;2016:8781740. PMID: 27956763. S

- 877. Xu HB, Huang HP. Clinical application of early enteral nutrition support after operation of esophagus carcinoma. Tumor. 2007;27(1):832-4. PMID: CN-00641139. X
- 878. Xu LN, Xu YY, Li GP, et al. Effect of Postoperative omega-3 Fatty Acid Immunonutritional Therapy on NK Cell Gene Methylation in Elderly Patients with Gastric Cancer. Current Medical Science. 2022;42(2):373-8. doi: 10.1007/s11596-022-2567-7. PMID: 35467300. O
- 879. Xu Q, Xu P, Cen Y, et al. Effects of preoperative oral administration of glucose solution combined with postoperative probiotics on inflammation and intestinal barrier function in patients after colorectal cancer surgery. Oncology Letters. 2019;18(1):694-8. doi: 10.3892/ol.2019.10336. PMID: 31289543. I
- 880. Xu R, Xiao S, Ding Z, et al. Does early postoperative enteral ecoimmunonutrition enhance intestinal function in gastric cancer? Asia Pacific Journal of Clinical Nutrition. 2020;29(3):469-75. doi: 10.6133/apjcn.202009_29(3).0004. PMID: 32990605. I
- 881. Xu Y, Gao L-Y, Xu D. Effect of individualized nutrition nursing program on postoperative recovery in rectal cancer patients undergoing anterior resection. World Chinese Journal of Digestology. 2017;25(2):2214-9. doi: 10.11569/wcjd.v25.i24.2214. PMID: CN-01445443. NE
- 882. Yamaguchi T, Morita T, Shinjo T, et al. Effect of parenteral hydration therapy based on the Japanese national clinical guideline on quality of life, discomfort, and symptom intensity in patients with advanced cancer. Journal of Pain & Symptom Management. 2012;43(6):1001-12. doi: 10.1016/j.jpainsymman.2011.06.028. PMID: 22651946. I
- 883. Yamazaki T, Enokida T, Wakasugi T, et al. Impact of prophylactic percutaneous endoscopic gastrostomy tube placement on treatment tolerance in head and neck cancer patients treated with cetuximab plus radiation. Japanese Journal of Clinical Oncology. 2016;46(9):825-31. doi: 10.1093/jjco/hyw079. PMID: 27317736. S

- 884. Yang CW, Lin HH, Hsieh TY, et al. Palliative enteral feeding for patients with malignant esophageal obstruction: a retrospective study. BMC Palliative Care. 2015;14:58. doi: 10.1186/s12904-015-0056-5. PMID: 26542798. S
- 885. Yang DJ, He WL, Wang L, et al. Effect of postoperative early enteral nutrition on the recovery of humoral immune function in patients with colorectal carcinoma undergoing elective resection. Zhonghua wei chang wai ke za zhi [Chinese journal of gastrointestinal surgery]. 2013;16(1):1051-4. PMID: CN-01120116. NE
- 886. Yang J, Zhang Q, Wang X. Role of nutritional support for postoperative recovery of respiratory function in patients with primary lung cancer. Oncology Letters. 2018;16(5):5978-82. doi: 10.3892/ol.2018.9348. PMID: 30333868. S
- 887. Yang X, Zhu MW, Xiu DR, et al. Effect of an oral nutritional supplementation on nutritional status and quality of life in patients with colorectal cancer and postoperative adjuvant chemotherapy: a multi-center prospective randomized control trial. Zhonghua wei chang wai ke za zhi [Chinese journal of gastrointestinal surgery]. 2020;23(6):566-71. doi: 10.3760/cma.j.cn.441530-20190724-00287. PMID: CN-02123819. NE
- 888. Yang Y, Xia Y, Chen H, et al. The effect of perioperative probiotics treatment for colorectal cancer: short-term outcomes of a randomized controlled trial. Oncotarget. 2016;7(7):8432-40. doi: 10.18632/oncotarget.7045. PMID: 26824990. I
- 889. Yang YC, Lee MS, Cheng HL, et al. More Frequent Nutrition Counseling Limits Weight Loss and Improves Energy Intake During Oncology Management: A Longitudinal Inpatient Study in Taiwan. Nutrition & Cancer. 2019;71(3):452-60. doi: 10.1080/01635581.2018.1516791. PMID: 30463443. C
- 890. Yanni A, Dequanter D, Lechien JR, et al. Malnutrition in head and neck cancer patients: Impacts and indications of a prophylactic percutaneous endoscopic gastrostomy. European annals of otorhinolaryngology, head & neck diseases. 2019;136(3):S27-S33. doi: 10.1016/j.anorl.2019.01.001. PMID: 30846293. S
- 891. Yao H, Bian X, Mao L, et al. Preoperative Enteral Nutritional Support in Patients Undergoing Hepatectomy for Hepatocellular Carcinoma: A Strengthening the Reporting of Observational Studies in Epidemiology Article. Medicine. 2015;94(4):e2006. doi: 10.1097/md.000000000002006. PMID: 26579806. S
- 892. Ye X, He D, Zhao J, et al. Application value of nursing intervention combined with early nutritional support in preventive stoma reversion of low rectal cancer. Oncology Letters. 2019;17(4):3777-82. doi: 10.3892/ol.2019.10055. PMID: 30930985. S
- 893. Yeh CN, Lee HL, Liu YY, et al. The role of parenteral glutamine supplement for surgical patient perioperatively: result of a single center, prospective and controlled study. Langenbecks Archives of Surgery. 2008;393(6):849-55. doi: 10.1007/s00423-008-0405-4. PMID: 18712409. S
- 894. Yexi Z, Man G, Lin L, et al. Assessment of therapeutic, nutritional index and radiation damage index effects of fucoidan in patients with thoracic tumor. Latin American Journal of Pharmacy. 2020;39(1):2403-9. PMID: 2005542565. O
- 895. Yi HC, Ibrahim Z, Abu Zaid Z, et al. Impact of Enhanced Recovery after Surgery with Preoperative Whey Protein-Infused Carbohydrate Loading and Postoperative Early Oral Feeding among Surgical Gynecologic Cancer Patients: An Open-Labelled Randomized Controlled Trial. Nutrients. 2020;12(1):20. doi: 10.3390/nu12010264. PMID: 31968595. I
- 896. Yildiz SY, Yazicioglu MB, Tiryaki C, et al. The effect of enteral immunonutrition in upper gastrointestinalsurgery for cancer: a prospective study. Turkish Journal of Medical Sciences. 2016;46(2):393-400. doi: 10.3906/sag-1411-102. PMID: 27511501. SS

- 897. Yin S, Hu SL, Shen G, et al. The effect of amino acid nutritional support on serum tryptophan and melatonin in lung cancer patients receiving chemotherapy. Zhonghua zhong liu za zhi [Chinese journal of oncology]. 2006;28(1):840-3. PMID: CN-00627158. NE
- 898. Yixun Z, Haiyi L, Yaoping L, et al. Effect of early enteral nutrition on postoperative recovery in patients with colon cancer. Cancer research and clinic. 2014;26(7):470-2. doi: 10.3760/cma.j.issn.1006-9801.2014.07.011. PMID: CN-01107193. X
- 899. Yoshii R, Yokoyama J, Ohba S, et al. Impact of EPA nutritional approach on cachexic patients with advanced hypopharyngeal cancer treated by induction chemotherapy. Head and Neck Oncology. 2014;6(2). PMID: 600781265. S
- 900. Yu FJ, Shih HY, Wu CY, et al. Enteral nutrition and quality of life in patients undergoing chemoradiotherapy for esophageal carcinoma: a comparison of nasogastric tube, esophageal stent, and ostomy tube feeding. Gastrointestinal Endoscopy. 2018;88(1):21-31.e4. doi: 10.1016/j.gie.2017.11.030. PMID: 29225081. I
- 901. Yu HM, Tang CW, Feng WM, et al. Early Enteral Nutrition Versus Parenteral Nutrition After Resection of Esophageal Cancer: a Retrospective Analysis. Indian Journal of Surgery. 2017;79(1):13-8. doi: 10.1007/s12262-015-1420-7. PMID: 28331260. S
- 902. Yu HZ, Long X, Liu CM, et al. Impact of enteral nutrition or parenteral nutrition in post-operative colorectal cancer patients on viscera organ functions and "passing wind" time. Chinese journal of clinical nutrition. 2009;17(5):268-70. doi: 10.3760/cma.j.issn.1674-635X.2009.05.003. PMID: CN-00889453. NE
- 903. Yu J, Xiao G, Zhou Y, et al. Impact of perioperative enteral immunonutrition in patients with gastrointestinal cancer undergoing elective surgery: a randomized controlled trial. Clinical Nutrition ESPEN. 2021;46:S776-. doi: 10.1016/j.clnesp.2021.09.659. PMID: CN-02347925. PT

- 904. Yuce Sari S, Yazici G, Yuce D, et al. The effect of glutamine and arginine-enriched nutritional support on quality of life in head and neck cancer patients treated with IMRT. Clinical Nutrition ESPEN. 2016;16:30-5. doi: 10.1016/j.clnesp.2016.08.003. PMID: 28531452. SS
- 905. Yun BK, Song M, Hwang HK, et al. Potential Nutritional and Metabolomic Advantages of High Fat Oral Supplementation in Pancreatectomized Pancreaticobiliary Cancer Patients. Nutrients. 2019;11(4):20. doi: 10.3390/nu11040893. PMID: 31010058. S
- 906. Zajac JF, Storman D, Swierz MJ, et al. Are systematic reviews addressing nutrition for cancer prevention trustworthy? A systematic survey of quality and risk of bias. Nutr Rev. 2022 May 9;80(6):1558-67. doi: 10.1093/nutrit/nuab093. PMID: 34921318.
- 907. Zaorsky NG, Churilla TM, Ruth K, et al. Men's health supplement use and outcomes in men receiving definitive intensity-modulated radiation therapy for localized prostate cancer1-3. American Journal of Clinical Nutrition. 2016;104(6):1583-93. doi: 10.3945/ajcn.115.119958. PMID: 613567321. I
- 908. Zdenkowski N, Radvan G, Pugliese L, et al. Treatment of pancreatic insufficiency using pancreatic extract in patients with advanced pancreatic cancer: a pilot study (PICNIC). Supportive Care in Cancer. 2017;25(6):1963-71. doi: 10.1007/s00520-017-3602-2. PMID: 614318521. I
- 909. Zemanova M, Novak F, Vitek P, et al. Outcomes of patients with oesophageal cancer treated with preoperative chemoradiotherapy, followed by tumor resection: influence of nutritional factors. Journal of B.U.On. 2012;17(2):310-6. PMID: 22740211. I
- 910. Zeng J, Hu J, Chen Q, et al. Home enteral nutrition's effects on nutritional status and quality of life after esophagectomy. Asia Pacific Journal of Clinical Nutrition. 2017;26(5):804-10. doi: 10.6133/apjcn.112016.07. PMID: 28802289. S

- 911. Zeng X, Yang S-W, Yang H-Q, et al. Effect of bifid triple viable combined with enteral nutrition support on gastrointestinal function and nutritional indexes in patients with gastric cancer after operation. World Chinese Journal of Digestology. 2020;28(1):410-6. doi: 10.11569/wcjd.v28.i11.410. PMID: CN-02147866. NE
- 912. Zhan WH, Jiang ZM, Tang Y, et al. Impact of hypocaloric and hypo-nitrogen parenteral nutrition on clinical outcome in postoperative patients: a multi-center randomized controlled trial of 120 cases. Zhonghua yi xue za zhi. 2007;87(2):1729-33. PMID: CN-00647242. NE
- 913. Zhang C, Gong L, Wu W, et al. Association between low-fat enteral nutrition after esophagectomy and a lower incidence of chyle leakage: A call for more and better evidence. Journal of International Medical Research. 2020;48(5):300060520926370. doi: 10.1177/0300060520926370. PMID: 32468882. S
- 914. Zhang CH, Li N, Wang XY, et al. Influence of Lipoplus fat emulsion on postoperative nutritional status and early inflammatory response in patients with gastrointestinal malignancies. Zhonghua wei chang wai ke za zhi [Chinese journal of gastrointestinal surgery]. 2012;15(5):448-51. PMID: CN-00979331. NE
- 915. Zhang J, Wang S, He Y, et al. Supporting role of the esophagus stent insertion and peripheral intravenous nutrition during palliative chemotherapy in advanced esophageal squamous cell carcinoma. Cancer research and clinic. 2016;28(1):743-7. doi: 10.3760/cma.j.issn.1006-9801.2016.11.006. PMID: CN-01339815. X
- 916. Zhang JW, Du P, Chen DW, et al. Effect of viable Bifidobacterium supplement on the immune status and inflammatory response in patients undergoing resection for colorectal cancer. Zhonghua wei chang wai ke za zhi [Chinese journal of gastrointestinal surgery]. 2010;13(1):40-3. PMID: CN-00753176. I

- 917. Zhang JW, Du P, Gao J, et al. Preoperative probiotics decrease postoperative infectious complications of colorectal cancer. American Journal of the Medical Sciences. 2012;343(3):199-205. doi: 10.1097/MAJ.0b013e31823aace6. PMID: 22197980. I
- 918. Zhang L, Qi X, Zhao H. Roles of clinical pharmacists in nutritional therapy for gastric cancer patients treated with neoadjuvant chemotherapy. Journal of practical oncology. 2016;31(6):537-40. PMID: CN-01327861. NE
- 919. Zhao B, Wang YX, Liu XY, et al. Comparison of effectiveness, safety, and costs of standardized and customized parenteral nutrition support among gastric cancer patients after gastrectomy: a retrospective cohort study. Asia Pacific Journal of Clinical Nutrition. 2018;27(4):818-22. doi: 10.6133/apjcn.012018.03. PMID: 30045426. S
- 920. Zhao CH, He B, Yang YF, et al. Dietary therapy of qi-yin-reinforcing porridge for the alleviation of chemotherapy related symptoms of gastrointestinal tumors: a single-case randomized controlled study. Chinese Journal of Integrative Medicine. 2013;19(6):418-23. doi: 10.1007/s11655-013-1329-y. PMID: 23784467. I
- 921. Zhao F, Chen Y, Li J, et al. Enteral nutrition in combination with parenteral nutrition support improves liver function and immune cell and inflammatory cytokine levels after hepatectomy procedures in elderly patients with liver cancer. International journal of clinical and experimental medicine. 2019;12(7):8502-11. PMID: 2002286562. O
- 922. Zhao G, Cao S, Zhang K, et al. Effect of early enteral nutrition on immune response and clinical outcomes after esophageal cancer surgery. Zhonghua wei chang wai ke za zhi [Chinese journal of gastrointestinal surgery]. 2014;17(4):356-60. PMID: CN-01076934. NE
- 923. Zhao H, Zhao H, Wang Y, et al. Randomized clinical trial of arginine-supplemented enteral nutrition versus standard enteral nutrition in patients undergoing gastric cancer surgery. Journal of Cancer Research & Clinical Oncology. 2013;139(9):1465-70. doi: 10.1007/s00432-013-1466-5. PMID: 23812551. RE

- 924. Zhao Q, Li Y, Yu B, et al. Effect of postoperative precision nutrition therapy on postoperative recovery for advanced gastric cancer after neoadjuvant chemotherapy. Zhonghua zhong liu za zhi [Chinese journal of oncology]. 2018;40(2):127-32. doi: 10.3760/cma.j.issn.0253-3766.2018.02.009. PMID: CN-01453937. NE
- 925. Zheng J, Tabung FK, Zhang J, et al. Post-cancer diagnosis dietary inflammatory potential is associated with survival among women diagnosed with colorectal cancer in the Women's Health Initiative. European Journal of Nutrition. 2020;59(3):965-77. doi: 10.1007/s00394-019-01956-z. PMID: 30955051. I
- 926. Zheng R, Rios-Diaz AJ, Liem S, et al. Is the placement of jejunostomy tubes in patients with esophageal cancer undergoing esophagectomy associated with increased inpatient healthcare utilization? An analysis of the National Readmissions Database. American Journal of Surgery. 2021;221(1):141-8. doi: 10.1016/j.amjsurg.2020.06.028. PMID: 32828519. S
- 927. Zhong HJ, Ying JE, Ma SL. Effect of Supportan on nutritional status and immune function of late-staged gastric cancer patients undergoing chemotherapy. Zhonghua wei chang wai ke za zhi [Chinese journal of gastrointestinal surgery]. 2006;9(5):405-8. PMID: CN-00721256. NE
- 928. Zhong ZQ, Song MM, Bai RX, et al. Clinical effects of enteral nutrition-assisted preoperative bowel preparation. Chinese journal of clinical nutrition. 2006;14(1):11-4. PMID: CN-00622744. NE
- 929. Zhu X, Liu D, Zong M, et al. Effect of swallowing training combined with nutritional intervention on the nutritional status and quality of life of laryngeal cancer patients with dysphagia after operation and radiotherapy. J Oral Rehabil. 2022 Jul;49(7):729-33. doi: 10.1111/joor.13328. PMID: 35352383. I

- 930. Zhu X, Wu Y, Qiu Y, et al. Effects of omega-3 fish oil lipid emulsion combined with parenteral nutrition on patients undergoing liver transplantation. Jpen: Journal of Parenteral & Enteral Nutrition. 2013;37(1):68-74. doi: 10.1177/0148607112440120. PMID: 22421017. P
- 931. Zhu XH, Wu YF, Qiu YD, et al. Liverprotecting effects of omega-3 fish oil lipid emulsion in liver transplantation. World Journal of Gastroenterology. 2012;18(4):6141-7. doi: 10.3748/wjg.v18.i42.6141. PMID: 23155344. P
- 932. Zhuang W, Wu H, Liu H, et al. Utility of feeding jejunostomy in patients with esophageal cancer undergoing esophagectomy with a high risk of anastomotic leakage. Journal of Gastrointestinal Oncology. 2021;12(2):433-45. doi: 10.21037/jgo-21-133. PMID: 2011937168. S

- 933. Zong L, Li H, Li S. Effects of neoadjuvant chemotherapy combined with enteral nutrition on perioperative immunity, inflammation and intestinal flora in gastric cancer patients. Journal of B.U.On. 2019;24(3):1113-9. PMID: 31424669. O
- 934. Zorn S, Ehret J, Schauble R, et al. Impact of modified short-term fasting and its combination with a fasting supportive diet during chemotherapy on the incidence and severity of chemotherapy-induced toxicities in cancer patients - a controlled cross-over pilot study. BMC Cancer. 2020;20(1):578. doi: 10.1186/s12885-020-07041-7. PMID: 32571329. S
- 935. Zorn S, Raynor A, Urbain P, et al. Impact of short-term modified fasting and the combination with a fasting supportive diet during chemotherapy on the incidence and severity of chemotherapy-induced toxicities in cancer patients - a randomised controlled cross-over pilot study (MOFAX). Aktuelle Ernahrungsmedizin. 2018;43(3):244-5. doi: 10.1055/s-0038-1647234. PMID: CN-02327042. SS

Appendix C. Evidence Tables for Chapter 5

Dietary Supplements

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Dominant Cancer Treatment Type Limited to Malnourished Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Dechaphunkul, 2022 ¹ (35007812) Asia Key Question 1, 3 Not conducted	110 52 years 21% female Race NR	Head & Neck 76% Stage IV Chemotherapy No Other (Nutritional Risk Index)	Immunonutrition with FA's, arginine, fiber & nucleotides, 3x/day for 5 days before each chemotherapy session. Outpatient Nutritionist Oral	Standard enteral formula, 3x/day for 5 days before each chemotherapy	Weight or body composition changes Symptoms Survival
de Miranda Torrinhas, 2013 ² (23398953) Other Key Question 1 High	63 58 years 37% female Race NR	Gastrointestinal % Stage IV disease NR Surgery alone No Other metric	Supplements (fish oil- based lipid emulsions) Inpatient Other Provider Parenteral	Lipid emulsion rich in medium-chain triglycerides	Adverse events Length of stay
Feijo, 2019 ³ (30710885) Other Key Question 1 High	68 58 years 35% female Race NR	Gastrointestinal 7% Stage IV disease Chemotherapy alone No Multiple tools	Single Supplement (received supplementation of formula enriched with omega-3 fractionated in 2 steps of 200 mL/d each) Outpatient Dietitian/nutritionist Oral	Received standard supplementation without omega-3	Weight or body composition changes

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Dominant Cancer Treatment Type Limited to Malnourished Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Kaya, 2016 ⁴ (26782276) Europe Key Question 1 Not conducted	58 58 years 6% female Race NR	Other % Stage IV disease NR Surgery alone No Screening tool NR	Supplements (immune modulating formulae (enriched with arginine, omega-3 fatty acids and nucleotides) for ten days preoperatively) Setting NR Provider NR Route of Intervention NR	Normal diet without any additional nutritional product	Changes in nutritional status
Lende, 2019 ^s (31703648) Europe Key Question 1 Not conducted	80 Age NR 100% female Race NR	Other 0% Stage IV disease Surgery alone No Screening tool NR	Supplements (400 ml Nutrica preop (12% CHO, 2% glucose, 10% polysaccharides) 18 hr before breast cancer surgery) Inpatient Physician Oral	Standard fasting preoperative protocol with unlimited access to drinking water	Adverse events Survival

Abbreviations: KQ = Key Question; PMID = PubMed Identification Number; NR = not reported; CHO = carbohydrate. *For select studies only. †Reports median age when mean is not available.

Nutrition Support Including Oral Nutrition Supplements

Table C.2.	Characteristics of included studies: use of nutrition support	including oral nutrition	supplements prior to cancer treatment
(KQ1)		-	

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Burden, 2017 ⁶ (28052576) Europe Key Question 1 Medium	101 70 years 33% female Race NR	Gastrointestinal 23% Stage IV disease Surgery alone Yes Other tool	Nutrition (Oral supplementation (Fortisip Compact) at a dose of 250 mL daily a minimum of 5 days pre- operatively) Multiple settings Dietitian/nutritionist Oral	The control group received the same leaflet and therapy with nutritionist and boxes the same size/weight but filled with water	Weight or body composition changes Changes in nutritional status Adverse events Survival Length of stay
Burden, 2011 ⁷ (21699587) Europe Key Question 1 Medium	125 65 years 38% female	Gastrointestinal 5% Stage IV disease Surgery alone No Other tool	Nutrition (400 mL of an oral supplementary drink daily and dietary advice) Inpatient Dietitian/nutritionist Oral	Received dietary advice only	Adverse events
Chen X, 2021 (33752148) ⁸ Asia Key Question 1,3 Medium	139 59 years 38% female Race NR	Gastrointestinal 0 Stage IV Surgery (post-chemo- radiotherapy) No Screening Tool NR	Oral Nutrition (1000 ml of 10% glucose solution 10 hr. before surgery + 500 ml of same 2-3 hr. before surgery) Setting NR Provider NR Oral	500 ml of 10% glucose solution 2-3 hr. before surgery	Adverse events Length of stay Readmissions or Emergency room visits Symptoms
Hamamoto, 2018 ⁹ (29915986) Asia Key Question 1 High	64 70 years 48% female Race NR	Gastrointestinal 5% Stage IV disease Surgery alone No Screening tool NR	Nutrition therapy (Carbohydrate-rich beverage night before surgery and before anesthesia) Inpatient Provider NR Oral	Clear drinking water 2 h prior to anesthesia	Weight or body composition changes Adverse events Length of stay

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
He F, 2022 ¹⁰ (35406085) Asia Key Question 1, 3 Low	67 62 years 29% female Race/ethnicity NR	Gastrointestinal Stage NR Surgery No NRS-2002	Preoperative ONS 500 ml. x 7 days (plus NJ feedings day 1-5 post- surgery) Outpatient Dietitian Oral	Preoperative dietary advice (plus NJ feedings day 1-5 post- surgery)	Adverse events Readmissions or emergency room visits Symptoms
Kabata, 2015 ¹¹ (25091056) Europe Key Question 1 High	102 64 years 49% female Race NR	Gastrointestinal % Stage IV disease NR Surgery alone No Other metric	Nutrition (oral nutrition supplement (Nutridrink) 14 days pre-operatively) Outpatient Provider NR Oral	Regular diet	Weight or body Composition changes Adverse events
Kruger, 2016 ¹² (27861546) Europe Key Question 1, 3 High	100 64 years 57% female Race NR	Gastrointestinal % Stage IV disease NR Surgery alone No NRS-2002	Nutrition (in-hospital peripheral intravenous nutrition on fasting days (1000 ml peripheral intravenous nutrition, 700 kcal)) Inpatient Multiple providers Parenteral	1000 ml isotonic electrolyte solution	Weight or body composition changes Quality of life
Lee, 2021 ¹³ (34353994) Asia Key Question 1 Medium	161 65.3 years 34% female Race NR	Gastrointestinal 7% Stage IV disease Surgery alone No Other tool	Nutrition (400 mL/day of immunonutrient- enriched ONS (Newcare Omega®), which contained high protein levels, arginine, and ω -3 fatty acids, in addition to the normal diet for 7 consecutive days prior to surgery) Inpatient Provider NR Oral	Patients in the control group had a normal diet without taking placebo.	Weight or body composition changes Adverse events Length of stay

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Martin, 2017 ¹⁴ (28162818) North America Key Question 1 High	71 61 years 55% female Non-Hispanic White	Gastrointestinal % Stage IV disease NR Surgery alone No Other tool	Nutrition (Preoperative immunonutrition three containers a day five days prior to surgery) Inpatient Provider NR Oral	No supplemental preoperative nutrition	Changes in nutritional Status Adverse events Length of stay
Rizvanovic, 2019 ¹⁵ (31309323) Europe Key Question 1, 3 Low	50 61 years 46% female Race NR	Gastrointestinal % Stage IV disease NR Surgery alone No NRS-2002	Nutrition (400 mL of carb drink 22 h before surgery and 200 mL 2 h before anaesthesia) Inpatient Provider NR Oral	Fasted for 8 hours before surgery	Length of stay Quality of life Symptoms
Shen Y, 2022 ¹⁶ (35307727) Asia Key Question 1 High	160 66 years 23% female Race NR	Gastrointestinal 0% Stage IV disease Surgery alone No Screening Tool NR	Nutrition (oral enteral nutrition for the three days prior to surgery along with bifidobacteria capsules 3x/day) Inpatient Provider NR Oral	Standard intestinal preparation	Adverse events Length of Stay
Tesar, 2022 ¹⁷ (35258042) Europe Key Question 1, 3 Medium	120 65 years 34% female Race NR	Gastrointestinal 0% Stage IV Surgery No NRS-2002	ONS (oral nutritional supplements 2x a day for 7 days before surgery) Inpatient Provider NR Oral	No oral nutritional supplement	Weight or body composition changes Adverse events Length of Stay Functional status
Wang, 2015 ¹⁸ (26125807) Asia Key Question 1 High	200 Age NR 35% female Race NR	Gastrointestinal % Stage IV disease NR Surgery alone No Screening tool NR	Nutrition therapy (preoperative enteral nutrition starting 1 week before surgery) Inpatient Provider NR Enteral	EN starting early after surgery	Changes in nutritional status Adverse events

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Xu, 2006 ¹⁹ (16830214) Asia Key Question 1 Medium	60 59 years 40% female Race NR	Gastrointestinal % Stage IV disease NR Surgery alone No Screening tool NR	Nutrition (Enteral Impact® for 7 days preoperatively) Inpatient Provider NR Enteral	Conventional diet preoperatively	Adverse events Length of stay
Zhao, 2018 ²⁰ (30273016) Asia Key Question 1 High	66 62 years 14% female Race NR	Gastrointestinal % Stage IV disease NR Multiple therapies No NRS-2002	Nutrition (500 ml oral nutrition support (Nutrison Fiber) seven days preoperatively) Other Provider NR Oral	Routine preop diet	Weight or body Composition changes Changes in nutritional status Length of Stay

Abbreviations: KQ = Key Question; PMID = PubMed Identification Number; NR = not reported; NRS = nutrition risk screening; EN = enteral nutrition; *For select studies only.

†Reports median age when mean is not available.

Risk of Bias and Outcomes Assessment

Table C.3. Risk of bias assessment: use of dietary supplements prior to cancer treatment (KQ1)

Author, Year, PMID	Outcome Timing	Selection Bias	Detection Bias	Performance Bias	Fidelity Bias	Reporting Bias	Attrition %	Overall Rating
de Miranda Torrinhas, 2013 ² (23398953)	Inpatient	Low	Low	High	Medium	Low	High (25%)	High
Feijo, 2019 ³ (30710885)	30 days	Low	Medium	High	Medium	Low	Medium (18%)	High

Abbreviations: PMID = PubMed Identification Number; KQ = Key Question

Author, Year,	Outcome	Selection	Detection	Performance	Fidelity Bias	Reporting	Attrition %	Overall
PMID Dumlar 004766	Timing	Blas	Blas	Bias	NA - aliana	Bias	L (00/)	Rating
(28052576)	30 a	Medium	Mealum	Medium	Medium	LOW	LOW (9%)	Medium
Burden, 2011 ⁷ (21699587)	Inpatient, 30 d	Low	Medium	Medium	Low	Low	Low (7%)	Medium
Chen X, 2021 ⁸ (33752148)	Inpatient, 30 d	Low	Medium	Medium	Medium (NR)	Low	Low (2%)	Medium
Hamamoto, 2018 ⁹⁹ (29915986)	X	Medium	High	X	x	X	X	High
He F 2022 ¹⁰ (35406085)	Inpatient, 30 d	Low	Low	Medium	Medium (NR)	Low	Low (2%)	Low
Kabata, 2015 ¹¹ (25091056)	30 days	Low	High	High	Х	Х	Х	High
Kruger, 2016 ¹² (27861546)	Х	Medium	High	X	Х	Х	Х	High
Lee, 2021 ¹³ (34353994)	30 days post- surgery	Low	Low	High	Medium	Low	Low (9%)	Medium
Martin, 2017 ¹⁴ (28162818)	X	High	X	X	X	X	X	High
Rizvanovic, 2019 ¹⁵ (31309323)	POD 2	Low	Low	Medium	Medium	Low	Low (0)	Low
Shen Y, 2022 ¹⁶ (35307727)	Inpatient	Low	Medium	High	Medium	Low	Low (0%)	High
Tesar 2022 ¹⁷ (35258042)	Inpatient, 30d	Medium	Medium	Medium	Medium	Low	Low (2%)	Medium
Wang, 2015 ¹⁸ (26125807)	X	Medium	High	X	X	X	X	High
Xu, 2006 ¹⁹ (16830214)	Inpatient	Medium	Medium	Medium	Medium	Low	Low (0)	Medium
Zhao, 2018 ²⁰ (30273016)	Inpatient	Low	Medium	High	x	X	X	High

Table C.4.1. Risk of bias assessment: use of nutrit	on support includin	a oral nutrition supplements	prior to cancer treatment (K	(Q1)

Note: X indicates domain not assessed due to already determined high risk of bias. **Abbreviations:** PMID = PubMed Identification; Number; d = days; KQ= Key Question

Author, Year, PMID, RoB Timing	N	Treatment	Control	Weight/ Body Comp.	Changes in Nutr. Status	Adverse Events	Readmissions / Emergency Room Visits	Surv ival	LOS	Treat. Toleran ce	QOL	Sympto ms	Functi onal Status
Burden, 2017 ⁶⁶ (28052576) Medium 30 d	101	Oral supplement ation with dietary advice	Same dietary leaflet and therapy with nutritionist and boxes the same	∱weight loss	\leftrightarrow	†surgical site infections ↔chest infections ↔complicat ions	NA	\leftrightarrow	\leftrightarrow	NA	-	-	
Burden, 2011 ⁷ (21699587) Medium Inpatient, 30 d	125	Oral supplement ary drink daily and dietary advice	Received dietary advice only	NA	NA	↔complicat ions	NA	NA	NA	-	-	-	-
Chen X, 2021 ⁸ (33752148) Medium Inpatient, 30 d	139	Double- dose oral CHO drink before gastrectomy	Single- dose oral CHO drink before gastrectom y	NA	NA	\leftrightarrow	 ↔ readmission ↔ reoperation 	NA	\leftrightarrow	-	-	-	-
He F, 2022 ¹⁰ (35406085) Low Inpatient, 30 d	67	Preoperativ e ONS 500 ml. x 7 days	Dietary advice	NA	NA	\leftrightarrow	↔ readmission	NA	NA	-	-	-	_

Table C.4.2. Outcomes assessment: use of nutrition support including oral nutrition supplements prior to cancer treatment (KQ1)

Author, Year, PMID, RoB Timing	N	Treatment	Control	Weight/ Body Comp.	Changes in Nutr. Status	Adverse Events	Readmissions / Emergency Room Visits	Surv ival	LOS	Treat. Toleran ce	QOL	Sympto ms	Functi onal Status
Lee, 2021 ¹³ (34353994) Medium 30 days post-surgery	161	Oral nutrition supplement for 7 consecutive days prior to surgery	Normal diet without taking placebo	 ↔weigh t/body composition changes ↑weight recover y after discharg e 	NA	↔infectious and total complicatio ns	NA	NA	\$	-	-	-	-
Rizvanovic, 2019 ¹⁵¹⁵¹⁵ (31309323) Low POD 2	50	Preop. Oral CHO supplement	Preop. Fasting	NA	NA	NA	NA	NA	Ť	-	-	-	-
Tesar, 2022 ¹⁷ (35258042) Medium Inpatient, 30d	120	oral nutritional supplement s 2x a day for 7 days before surgery	No oral nutrition supplemen t	↔ muscle, water or fat weight	NA	↔ Clavien- Dindo complicatio ns	NA	NA	↔ ICU, hospi tal stay	-	-	-	-
Xu, 2006 ¹⁹¹⁹¹⁹ (16830214) Medium	60	Enteral Impact® for 7 days preoperative ly	Conventio nal diet preoperati vely	NA	NA	↑noninfecti ous complicatio ns	NA	NA	Î	-	-	-	-

Abbreviations: PMID = PubMed Identification Number; ROB=risk of bias; CHO=carbohydrate; NA=not assessed

-: not applicable

↑: Intervention group had a statistically significantly better outcome than comparison group (e.g. fewer Aes, shorter LOS than comparison group)
↓: Intervention group had a statistically significantly worse outcome than comparison group (e.g. more Aes, longer LOS)

 \leftrightarrow : No statistically significant difference between groups

Appendix D. Evidence Tables for Chapter 6

Dietary Supplements

Table D.1. Characteristics of included studies: use of dietary supplements prior to and during cancer treatment (spans KQ 1 and 2)

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Celik, 2009 ²¹ (19761135) Europe Key Question 1 & 2 Not conducted	50 63 years 100% female Race NR	Other % Stage IV disease NR Surgery alone No Other tool	Supplements (immune- enhancing enteral nutrition) Inpatient Provider NR Oral	Standard, oral enteral nutrition formula	Adverse events Survival Length of stay
Cereda, 2019 ²² (31568698) Europe Key Question 1 & 2, 3 Not conducted	166 65 years 40% female Race NR	Multiple cancers 81% Stage IV disease Chemotherapy alone Yes Other metric	Single Supplement (WPI supplementation) Outpatient Dietitian/nutritionist Oral	Nutritional counseling alone	Weight or body composition changes Treatment tolerance Quality of life
Ghosh, 2012 ²³ PMID NA Europe Key Question 1 & 2 Not conducted	60 61 years 12% female Race NR	Head & Neck 30% Stage IV disease Surgery alone No Screening Tool NR	Supplements (IMPACT ONS – added arginine, nucleotides, & omega- 3-FAs) Inpatient Other Enteral	ONS with same calories & nitrogen (no supplements) same dose, timing & duration	Adverse events Survival Length of stay
Haidari, 2020 ²⁴ (32219761) Asia Key Question 1 & 2 High	81 58 years 43% female Race NR	Gastrointestinal 0% Stage IV disease Chemotherapy alone No Screening Tool NR	Supplements (added omega-3-FAs) Outpatient Provider NR Oral	Placebo capsule for omega-3 fatty acids twice/day plus a placebo vitamin D once/week started 1 week before first chemotherapy appt and going for 8 weeks	Weight or body composition changes

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Healy, 2017 ²⁵ (28742713) Europe Key Question 1 & 2 High	268 62 years % female NR Race NR	Gastrointestinal % Stage IV disease NR Surgery alone No Other Tool	Single Supplement (Prosure ONS) Multiple settings Multiple providers Enteral	Standard ONS (Ensure Plus)	Weight or body composition changes Changes in nutritional status Adverse events
Jantharapattana, 2020 ²⁶ (31647147) Asia Key Question 1 & 2 Not conducted	65 57 years 18% female Race NR	Head & Neck 0% Stage IV disease Surgery alone Yes MST	Supplement (Ensure like product – Prosure) Setting NR Provider NR Oral	Conventional isocaloric supplement (Blendera)	Weight or body composition changes Adverse events
Jo, 2006 ²⁷ (16927064) Asia Key Question 1 & 2 Medium	60 57 years 52% female Race NR	Gastrointestinal % Stage IV disease NR Surgery alone No Other tool	Single Supplement (added glutamine) Inpatient Provider NR Parenteral	TPN amino acid solution (1.3 g/kg/day amino acids) – Preop day 2-5 and postop day (not incl operation day; 7 days total)	Weight or body composition changes Adverse events Survival Length of stay
Oguz, 2007 ²⁸ (17573745) Other Key Question 1 & 2 High	109 54 years 35% female Race NR	Gastrointestinal % Stage IV disease NR Surgery alone No Other tool	Single Supplement (added glutamine) Inpatient Multiple providers Enteral	Ensure oral (1000 ml/d x 5d preop, 500 ml/day x 2d postop then 1000ml/d 3+ more days postop)	Changes in nutritional status Adverse events Survival
Ryan, 2009 ²⁹ (19247018) Europe Key Question 1 & 2 High	53 64 years 9% female Race NR	Gastrointestinal % Stage IV disease NR Surgery alone No Other Tool	Single Supplements (added triglycerides/fatty acids) Setting NR Dietitian/nutritionist Enteral	Iso-caloric iso- nitrogenous standard nutritional feed	Weight or body Composition changes Adverse events Survival
Serrano, 2022 ³⁰ (35606184) Canada Key Question 1 & 2, 3 Medium	71 65 years 41% female Race NR	Gastrointestinal 69% Stage IV disease Surgery No MUST	Supplements (protein supplementation rich in arginine and omega-6 and carbohydrate loading) Other Nurse Oral	One placebo to each supplement	Adverse events Readmissions or Emergency room visits Survival Length of Stay Quality of life

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Sorensen, 2014 ³¹ (24281905) Sorenson, 2020 ³² (32391656) Europe Key Question 1 & 2 Medium	148 71 years 46% female Race NR	Gastrointestinal 10% Stage IV disease Surgery alone No NRS-2002	Single Supplement (n-3 FA-enriched ONS)) Outpatient Multiple providers Oral	Standard isocaloric and isonitrogenous ONS for 7 days before & after surgery	Adverse events Readmissions or Emergency room visits Survival Length of stay
Sultan, 2012 ³³ (22237467) Europe Key Question 1 & 2 Medium	195 64 years 26% female Race NR	Gastrointestinal % Stage IV disease NR Surgery alone No Screening tool NR	Nutrition therapy (immunoenhancing diet – Oxepa) Inpatient Provider NR Enteral	No preoperative nutritional support; enteral Osmolite for 7 days after surgery	Weight or body composition changes Adverse events Survival Length of stay
Tsuchiya, 2016 ³⁴ (27306219) Asia Key Question 1 & 2, 3 High	70 63 years 40% female Race NR	Gastrointestinal % Stage IV disease NR Multiple therapies No Screening Tool NR	Supplements (added amino acids) Outpatient Provider NR Oral	No supplement	Adverse events Treatment tolerance
Vidal-Casariego, 2014 ³⁵ (23471208) Europe Key Question 1 & 2 Not conducted	69 67 years 35% female Race NR	Multiple cancers 7% Stage IV disease Radiation alone No Other Tool	Single Supplements (added glutamine) Outpatient Other Oral	30 mg/d oral casein (placebo) 3 days before RT until the completion of RT	Weight or body composition changes Changes in nutritional status Adverse events
Yegen, 2020 ³⁶ PMID NA Europe Key Question 1 & 2 High	78 62.9 42% female Race NR	Gastrointestinal 0% Stage IV disease Surgery No Multiple tools	Supplements (ONS with arginine, glutamine, Omega-3 fatty acid, and RNA. (3x/day) from 7 d before surgery to 30 days after (NJ tube if needed)) Inpatient Provider NR Enteral	200 ml ONS (Resource 2.0 with fiber), 3x/day from 7 d before surgery to 30 days after (NJ tube if needed	Weight or body composition changes Adverse events Survival Length of Stay

Abbreviations: KQ = Key Question; PMID = PubMed Identification Number; NR = not reported; NA = not available; WPI = whey protein isolate; ONS = oral nutritional supplement; FA = fatty acid; EPA = eicosapentaenoic acid; MST = malnutrition screening tool; TPN = total parenteral nutrition; RT = radiotherapy.

*For select studies only.

†Reports median age when mean is not available.

Route or Timing of Nutrition Interventions

Table D.2. Characte	eristics of included studies: re	oute or timing of nutrition	interventions delivered	prior to and during cancer treatment
(spans KQ 1 and 2)	J. Contraction of the second se	-		

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Braga, 2002 ³⁷ (11822956) Europe Key Question 1 & 2 Not conducted	150 65 years 44% female Race NR	Gastrointestinal % Stage IV disease NR Surgery alone Yes Other metric	Nutrition therapy (oral diet enriched with arginine, w-3 fatty acids, and RNA) Inpatient Provider NR Other	Postoperative enteral feeding with a standard diet within 12 hours of surgery	Adverse events Length of stay
Brown, 2017 ³⁸ (28797454) Other Key Question 1 & 2 3 Not conducted	131 60 years 12% female Race NR	Head & Neck 31% Stage IV disease Multiple therapies Limited to malnourished NR Screening tool NR	Nutrition (early feeding intervention via prophylactic gastrostomy tube) Multiple settings Dietitian/nutritionist Enteral	Usual care which commenced feeding when clinically indicated	Weight or body composition changes Changes in nutritional status Readmissions or emergency room visits Length of stay Treatment tolerance Quality of life
Ding, 2015 ³⁹ (25835119) Asia Key Question 1 & 2 Not conducted	106 Age NR 34% female Race NR	Gastrointestinal % Stage IV disease NR Surgery alone No Screening tool NR	Nutrition therapy (preoperative EN for 1 week) Inpatient Provider NR Enteral	Early postoperative EN	Weight or body composition changes
Falewee, 2014 ⁴⁰ (24182765) Europe Key Question 1 & 2 Not conducted	298 59 years 16% female Race NR	Head & Neck % Stage IV disease NR Surgery alone No Multiple tools	Nutrition therapy (perioperative IMPACT) Inpatient Provider NR Other	Oral or enteral, 7 days before surgery and 7-15 days postop; Perioperative formula of IMPACT without immune nutrients = "reference diet"	Adverse events Length of stay

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Gianotti, 2002 ⁴¹ (12055582) Europe Key Question 1 & 2 Not conducted	305 64 years 45% female Race NR	Gastrointestinal % Stage IV disease NR Surgery alone No Screening tool NR	Nutrition therapy (IV supplementation) Multiple settings Provider NR Oral	No supplementation preop, oral supplementation postop	Adverse events Length of stay
Miyata, 2012 ⁴² (22169459) Asia Key Question 1 & 2 Not conducted	91 63 years 24% female Race NR	Gastrointestinal 38% Stage IV disease Multiple therapies No Screening tool NR	Nutrition therapy (omega 3-FAs rich nutritional supplements – enteral if unable to take orally) Setting NR Provider NR Oral	Nutrition support at 600kcal/day (parental if unable to take orally)	Weight or body composition changes Adverse events
Mudge, 2018 ⁴³ (1596002) Other Key Question 1& 2, 3 Not conducted	278 64 years 19% female Race NR	Gastrointestinal % Stage IV disease NR Surgery alone No Other tool	Nutrition therapy (immunonutrition (pre-, peri-, and post-op)) Inpatient Physician Other	Perioperative standard nutrition	Adverse events Survival Length of stay Quality of life
Wong TX, 2022 ⁴⁴ (35276977) Asia Key Question 1 & 2, 3 Not conducted	91 60 years 76% female Race NR	Multiple Cancers % Stage IV disease NR Surgery Yes Screening tool NR	Group SS: ONS (milk- based formula fortified with micronutrients)for 5 to 14 days preoperatively and postoperatively until discharge Group SS-E: ONS for 5 to 14 days pre- operatively and for 90- days post-discharge Multiple settings Provider NR Oral	ONS postoperatively until discharge	Weight or body composition changes Functional status

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Wu, 2020 ⁴⁵ PMID NA Asia Key Question 1 & 2 Not conducted	76 60 years 32% female Race NR	Gastrointestinal % Stage IV disease NR Surgery alone Yes Other tool	Nutrition therapy (enteral support preoperative, Nutrison Fibre postoperative) Inpatient Provider NR Enteral	Normal diet before surgery, 250-500 mL of compound amino acid injection and 400 mL of Glucose Solution, PN	Changes in nutritional status Adverse events

Abbreviations: KQ = Key Question; PMID = PubMed Identification Number; NR = not reported; NA = not available; FA = fatty acid; EN = enteral nutrition; PN = parenteral nutrition.

*For select studies only.

†Reports median age when mean is not available.

Nutrition Support Including Oral Nutrition Supplements

Table D.3. Characteristics of included studies: use of nutrition support including oral nutrition supplements prior to and during or	cancer
treatment (spans KQ1 and 2)	

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Bozzetti, 2000 ⁴⁶ (10638466) Europe Key Question 1 & 2 Medium	90 Age NR 50% female Race NR	Gastrointestinal % Stage IV disease NR Surgery alone Yes Other metric	Nutrition therapy (special TPN) Inpatient Provider NR Parenteral	Standard central IV feeding postop	Adverse events Survival Length of stay
Chen, 2017 ⁴⁷ PMID NA Asia Key Question 1 & 2 Medium	120 71 years 39% female Race NR	Gastrointestinal 0% Stage IV disease Surgery alone Yes NRS-2002	Nutrition (multi-oil fat emulsion PN) Inpatient Provider NR Parenteral	No preop nutritional support	Adverse events Length of stay

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Feng, 2022 ⁴⁸ (34362624) Asia Key Question 1 & 2, 3 Low	126 70 years 37% female Race NR	Gastrointestinal 0% Stage IV Surgery No NR	Preoperative oral CHO 200 ml. 2-3 hr. before surgery & postoperative early oral feeding (POD1=special clear liquid but no TPN; POD2=normal diet) Hospital inpatient Provider NR Oral	Fasted pre-surgery (> 8 hrs.) + TPN x 24 hr. post-surgery; advance diet as tolerated Hospital inpatient Provider NR Oral & TPN	Adverse events Readmissions/Emergency Room Visits Symptoms
Ida, 2017 ⁴⁹ (28072447) Aoyama, 2019 ⁵⁰ (30854113) Aoyama, 2022 (35836480) ⁵¹ Asia Key Question 1 & 2 High	126 65 years 34% female Race NR	Gastrointestinal 0% Stage IV disease Surgery alone No Screening Tool NR	Supplement (Ensure like product – EPA) Setting NR Provider NR Oral	Standard diet	Weight or body composition changes Adverse events
Kong, 2018 ⁵² (30055788) Asia Key Question 1 & 2, 3 High	144 Age NR 37% female Race NR	Gastrointestinal 7% Stage IV disease Surgery alone Yes Other tool	Nutrition (standard oral nutritional supplement (Ensure powder) for 2 weeks before gastrectomy and for 4 weeks postoperatively in malnourished patients) Multiple settings Dietitian/nutritionist Oral	Standard meals designed for post- gastrectomy patients	Weight or body composition changes Adverse events Quality of life
Lidder, 2013 ⁵³ (23406311) Europe Key Question 1 & 2 Low	120 70 years 47% female Race NR	Gastrointestinal % Stage IV disease NR Surgery alone No Screening tool NR	Nutrition therapy (Nutricia PreOp solution + carbohydrate supplement) Setting NR Provider NR Oral	Placebo drink given 2 hours preop equal volume of flavored water with artificial sweetener postop	Adverse events Length of Stay

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Moya, 2016 ⁵⁴ (26936601) Europe Key Question 1 & 2 Medium	128 69 years 53% female Race NR	Gastrointestinal % Stage IV disease NR Surgery alone No Screening tool NR	Nutrition (immune- enhancing oral supplement) Inpatient Provider NR Oral	Dietary advice	Adverse events Readmissions or emergency room visits Survival Length of stay
Poon, 2004 ⁵⁵ (15043519) Asia Key Question 1 & 2, 3 Not conducted	88 59 years 7% female Race NR	Other 0% Stage IV disease Chemotherapy alone No Screening tool NR	Nutrition (nutritional supplementation with BCAAs twice a day in the morning and evening in addition to the usual diet) Inpatient Provider NR Oral	Standard diet	Weight or body composition changes Adverse events Survival Functional status
Ritch, 2019 ⁵⁶ (30359680) North America Key Question 1 & 2 Not conducted	61 68 years 12% female Non-Hispanic White	Other % Stage IV disease NR Surgery alone No Screening tool NR	Nutrition (Supplement – Ensure) Multiple settings Dietitian/nutritionist Oral	Twice daily servings of multivitamins for 3-4 weeks before surgery and 4 weeks after surgery	Weight or body Composition changes Adverse events Readmissions or Emergency room visits Survival Length of stay
Sanchez-Guillen, 2021 ⁵⁷ (34441942) Lopez-Rodriguez-Arias, 2021 ⁵⁸ (34579122) Europe Key Question 1 & 2 Medium	170 69.7 years 39% female Race NR	Gastrointestinal Stage IV disease NR Surgery No Other Tool (ESPEN)	Peripheral PN Peri- Olimel N4E for 4 days (1 day before surgery + 3 days after) in addition to an ERAS protocol Inpatient Provider NR Parenteral	Conventional fluid therapy 1 day before surgery; post-surgery diet per ERAS protocol	Adverse events Length of stay Survival
Sittirai, 2021 ⁵⁹ (34371175) Asia Key Question 1 & 2, 3 Not conducted	126 57 years 34% female Race NR	Head & Neck 87% Stage IV Surgery No Screening tool NR	ONS (Immune- enhancing diet containing arginine, glutamine, and fish oil- derived fatty acids) Inpatient Provider NR Oral	Hospital-prepared blenderized diet	Adverse events Length of stay Symptoms

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Wu, 2006 ⁶⁰ (16688841) Asia Key Question 1 & 2 High	512 57 years 36% female Race NR	Gastrointestinal % Stage IV disease NR Surgery alone Yes Other tool	Nutrition therapy (EN or PN preop and postop) Inpatient Provider NR Route of Administration NR	Standard oral hospital diet preop and PN in postop	Adverse events Survival Length of stay
Yan, 2021 ⁶¹ (32875913) Asia Key Question 1 & 2 High	200 56 years 16% female Race NR	Gastrointestinal % Stage IV disease NR Surgery alone No NRS-2002	Nutrition therapy (perioperative protein enriched EN) Inpatient Provider NR Enteral	Regular food intake (low-fat diet pre- surgery, liquid/semi- liquid diet for 7 days post-surgery)	Length of stay

Abbreviations: KQ = Key Question; PMID = PubMed Identification Number; NR = not reported; TPN = total parenteral nutrition; NRS = nutritional risk screening; PN = parenteral nutrition; ONS = oral nutritional supplement; EN = enteral nutrition; BCAA = branched-chain amino acid.

*For select studies only.

†Reports median age when mean is not available.

Multi-Component Interventions

Table D.4. Characteristics of included studies: use of multi-component interventions prior to and during cancer treatment (s	pans KQ1
and 2)	-

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Baldwin, 2011 ⁶² (21733143) Europe Key Question 1 & 2, 3 Not conducted	358 66 years 37% female Race NR	Multiple cancers % Stage IV disease NR Chemotherapy alone Yes Screening tool NR	Multi-modality ((1) dietary advice alone; (2) nutritional supplement alone; and (3) dietary and nutritional supplement) Outpatient Dietitian/nutritionist Oral	No intervention	Survival Quality of life

Abbreviations: KQ = Key Question; PMID = PubMed Identification Number; NR = not reported

Risk of Bias and Outcomes Assessment

Table D.5. Risk of bias assessment: use of dietary supplements prior to and during cancer treatment (spans KQ 1 and 2)

Author, Year, PMID	Outcome Timing	Selection Bias	Detection Bias	Performance Bias	Fidelity Bias	Reporting Bias	Attrition %	Overall Rating
Haidari, 2020 ²⁴ (32219761)	8 wks	Medium	Low	High	Х	Х	Х	High
Healy, 2017 ²⁵ (28742713)	6 months	Low	Low	High	Low	Low	High (29%)	High
Jo, 2006 ²⁷ (16927064)	Inpatient	Low	Low	High	Medium	Low	Low (0)	Medium
Oguz, 2007 ²⁸ (17573745)	Inpatient	Medium	High	High	Х	Х	Х	High
Ryan, 2009 ²⁹ (19247018)	21 days	Low	High	Х	Х	Х	Х	High
Serrano, 2022 ³⁰ (35606184)	4 wks, 12 weeks post- surgery	Low	Low	Medium	High	Low	Low (9%)	Medium

Author, Year, PMID	Outcome Timing	Selection Bias	Detection Bias	Performance Bias	Fidelity Bias	Reporting Bias	Attrition %	Overall Rating
Sorenson 2014 ³¹ (24281905) Sorensen 2020 ³² (32391656)	30 days; 3, yr, 5 yr	Medium	Low	Low	Medium	Low	Medium 13%	Medium
Sultan 2012 ³³ (22237467)	Inpatient	Low	Low	Low	Low	Medium	Low (0)	Medium
Tsuchiya, 2016 ³⁴ (27306219)	28 days	Low	Medium	High	x	X	X	High
Yegen, 2020 ³⁶ PMID NA	Inpatient (post- hospital non- lab outcomes NR)	Medium	Medium	High	Medium	Low	Low (0%)	High

Note: X indicates domain not assessed due to previously determined high risk of bias **Abbreviations**: PMID = PubMed Identification Number; yr = year; KQ = Key Question

Table D.5.1. Outcomes assessment: use of dietary supplement	s prior to and during cance	r treatment (low and mediu	m ROB studies),
spans KQ 1 and 2			

Author, year, PMID, RoB Timing	N	Treatment	Control	Weight / Body Comp.	Change s in Nutr.	Adverse Events	Readmission s/Emergency Room Visits	Surviv al	LOS	Treat. Toleran ce	QOL	Sym pto ms	Functional Status
			1		Status								
Jo 2006 ²⁷ (16927064) Medium Inpatient	60	Glutamine 0.2 g/kg/d in TPN postop	TPN (presumed, NR)	↑MAC ↔TSF, MAMC	NA	↔ by POD 7	NA	\leftrightarrow	\leftrightarrow	-	-	-	-
Serrano, 2022 ³⁰ 35606184 Medium 4 wks, 12 weeks post- surgery	71	Protein supplementation rich in arginine and omega-6 and carbohydrate loading	One placebo to each supplement	NA	NA	↔ Infectious complicat ions	↔ Unplanned ER visits ↔ readmissions	↔ 90-day mortalit y	↔	-	-	-	-
Sorenson 2014 ³¹ (24281905) Sorensen 2020 ³² (32391656) Medium 30 days	148	Omega-3 FA enriched ONS	Standard isocaloric ONS	NA	NA	\leftrightarrow	Readmissions ↔	\leftrightarrow	↔	-	-	-	-

Author, year, PMID, RoB Timing	Ν	Treatment	Control	Weight / Body Comp.	Change s in Nutr. Status	Adverse Events	Readmission s/Emergency Room Visits	Surviv al	LOS	Treat. Toleran ce	QOL	Sym pto ms	Functional Status
Sultan 2012 ³³ (22237467) Medium Inpatient	195	1. IEN with omega 3 FAs 7d preop & postop 2. Standard EN for 7d preop & postop	Standard EN postop only	\leftrightarrow	NA	↔ count of infectious AEs ↔ % of patients. infectious AEs ↔ other AFs	NA	\leftrightarrow	\leftrightarrow	-	-	-	-

Abbreviations: PMID = PubMed Identification Number; ROB=risk of bias; BMI: body mass index; FFM: fat free mass; AE: adverse events; LOS: length of stay; NA=not assessed; QOL: quality of life; PRO: patient reported outcome; TPN: total parenteral nutrition; FA: fatty acid: ONS: oral nutritional supplement: EN: enteral nutrition: POD: post-operative day; IEN=immunoenteral nutrition; NR: Not reported; MAC: mid-arm circumference; TSF: triceps skin fold; MAMC: mid-arm muscle circumference; KQ = Key Question; NA=not assessed.

-: Not applicable

↑: Intervention group had a statistically significantly better outcome than comparison group (e.g. fewer AEs, shorter LOS than comparison group)

↓: Intervention group had a statistically significantly worse outcome than comparison group (e.g. more AEs, longer LOS)

 \leftrightarrow : No statistically significant difference between groups

Table D.6.1. Risk of bias	s assessment: use of nutrition support	including oral nutrition su	upplements prior to and du	ring cancer treatment
(spans KQ 1 and 2)		-		-

Author, Year, PMID	Outcome Timing	Selection Bias	Detection Bias	Performance Bias	Fidelity Bias	Reporting Bias	Attrition %	Overall Rating
Bozzetti, 2000 ⁴⁶ (10638466)	10 d preop + inpatient postop	Low	Medium	Medium	Medium	Low	Low (0)	Medium
Chen, 2017 ⁴⁷ (616782903)	Inpatient	Low	Low	Low	Medium	Medium	Low (0)	Medium
Feng, 2022 ⁴⁸ (34362624)	Inpatient, 30 d	Low	Low	Medium	Medium	Low	Low (0)	Low
lda, 2017 ⁴⁹ (28072447) Aoyama, 2019 ⁵⁰ (30854113) Aoyama, 2022 ⁵¹ (35836480)	3 months	Low	Medium	High	X	X	X	High
Kong, 2018 ⁵² (30055788)	30 days	Low	Medium	High	X	X	Х	High
Lidder 2013 ⁵³ (23406311)	Inpatient, 30 days	Low	Low	Low	Medium	Low	Low (0)	Low
Moya 2016 ⁵⁴ (26936601)	7 d preop, inpatient, 30d	Low	Low	Medium	Medium	Low	Low 8%	Medium

Author, Year, PMID	Outcome Timing	Selection Bias	Detection Bias	Performance Bias	Fidelity Bias	Reporting Bias	Attrition %	Overall Rating
Sanchez-Guillen, 2021 ⁵⁷ (34441942) Lopez- Rodriguez-Arias, 2021 ⁵⁸ (34579122)	90 days post- surgery	Low	Medium	Medium	Medium	Low	Low (7%)	Medium
Yan, 2021 ⁶¹ (32875913)	Inpatient	Medium	High	Х	Х	Х	Х	High
Wu, 2006 ⁶⁰ (16688841)	Inpatient	Medium	High	Х	Х	X	Х	High

Note: X indicated domain not assessed due to previously determined high risk of bias. **Abbreviations:** PMID = PubMed Identification Number; d = days; preop = preoperative; KQ = Key Question

Table D.6.2. Outcomes assessment: use of nutrition su	pport including oral nutritie	ion supplements prior to and	during cancer treatment
(low and medium ROB Studies; spans KQ 1 and 2)			

Author, Year, PMID, RoB Timing	N	Treatment	Control	Weight/ Body Comp.	Changes in Nutr. Status	Adverse Events	Readmissions/ Emergency Room Visits	Surviv al	LOS	Treat. Tolerance	QOL	Symptoms	Functional Status
Bozzetti, 2000 ⁴⁶ (10638466) Medium Inpatient	317	Total parenteral nutrition for 10 days perioperatively and 9 days postoperatively	Standard central IV feeding postop	NA	NA	↑ complicat ions/noni nfectious complicai tons	NA	Î	Ļ	-	-	-	-
Chen 2017 ⁴⁷ (616782903) Medium Inpatient 3d, 7d	120	 Preop multi-oil fat emulsion PN Intralipid PN 	No nutritional support	NA	NA	↑ Complica tions	NA	NA	\leftrightarrow	-	-	-	-
Feng, 2022 ⁴⁸ (34362624) KQ 1&2, 3 Low Inpatient, 30 d	126	Preoperative oral CHO 200 ml 2-3 hr. before surgery & postoperative early oral feeding	Fasted pre- surgery (> 8 hrs.) + TPN x 24 hr. post- surgery	NA	NA	↔ complicat ions	↑hospitalization	NA	NA	-	-	-	-

Author, Year, PMID, RoB Timing	N	Treatment	Control	Weight/ Body Comp.	Changes in Nutr. Status	Adverse Events	Readmissions/ Emergency Room Visits	Surviv al	LOS	Treat. Tolerance	QOL	Symptoms	Functional Status
Lidder, 2013 ⁵³ (23406311) Low Inpatient, 30d	120	1.Preop+postop CHO drink 2. Preop CHO drink 3. Postop CHO drink	Placebo drink preop & postop	NA	NA	↑ preop+ postop vs control in complicat ions at 30 days	NA	NA	\leftrightarrow	-	-	-	-
Moya, 2016 ⁵⁴ (26936601) Medium Inpatient 30d	128	IEN x 7 d preop + 5 d postop	Dietary advice	NA	NA	↑ wound infection ↔ other AEs	\leftrightarrow	↔ (0)	\leftrightarrow	-	-	-	-
Sanchez- Guillen, 2021 ⁵⁷ (34441942) Lopez- Rodriguez- Arias, 2021 ⁵⁸ (34579122)	170	Peripheral parenteral nutrition 4 days (the day before the scheduled surgery and 3 days after surgery	Convention al fluid therapy	NA	NA	↔ overall AEs ↑ major AEs ↑ minor AEs ↔ SSI	NA	UTD	\leftrightarrow	-	-	-	-
Medium													
90 days post- surgery													

Abbreviations: PMID = PubMed Identification Number; ROB=risk of bias; BMI: body mass index; FFM: fat free mass; AE: adverse events; LOS: length of stay; QOL: quality of life; PRO: patient reported outcome; PN: parenteral nutrition; IEN=immunoenteral nutrition; UTD: unable to determine; KQ = Key Question; NA=not assessed

-: Not applicable

1: Intervention group had a statistically significantly better outcome than comparison group (e.g. fewer Aes, shorter LOS than comparison group)

: Intervention group had a statistically significantly worse outcome than comparison group (e.g. more Aes, longer LOS)

↔: No statistically significant difference between groups

Appendix E. Evidence Tables for Chapter 7

Nutritional Counseling

Table E.1. Characteristics of included studies: use of nutritional counseling during cancer treatment (KQ2)

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Bourdel-Marchasson, 2014 ⁶³ (25265392) Europe Key Question 2 Not conducted	341 78 years 49% female Race NR	Multiple cancers 7% Stage IV disease Chemotherapy alone Yes MNA-SF	Counseling only (dietary advice) Outpatient Dietitian/nutritionist Not applicable	Routine nutritional care	Weight or body composition changes Readmissions or emergency room visits Survival
Britton, 2019 ⁶⁴ (30296472) Other Key Question 2, 3 Not conducted	307 58 years 21% female Other	Head & Neck 65% Stage IV disease Radiation alone No Screening tool NR	Counseling only (Eating As Treatment (EAT) dietary advice) Outpatient Dietitian/nutritionist Not applicable	Treatment as usual	Weight or body composition changes Changes in nutritional status Treatment tolerance Quality of life
Forslund, 2020 ⁶⁵ (31758324) Europe Key Question 2, 3 Not conducted	180 67 years 0% female Race NR	Other 4% Stage IV disease Radiation alone No Screening tool NR	Counseling only (advised to increase soluble fiber and decrease lactose in diet) Outpatient Dietitian/nutritionist Not applicable	Standard care	Adverse events Quality of life
Isenring, 2004 ⁶⁶ (15226773) Other Key Question 2, 3 Not conducted	60 62 years 15% female Race NR	Multiple cancers % Stage IV disease NR Multiple therapies No Other tool	Counseling only (intensive nutritional counseling for up to 3 months) Outpatient Dietitian/nutritionist Not applicable	Less intensive diet education by nurses	Weight or body composition changes Changes in nutritional Status Quality of life Functional status

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Loser, 2021 ⁶⁷ (33766702) Europe Key Question 2 Not conducted	69 Age NR 28% female Race NR	Head & Neck 33% Stage IV disease Multiple therapies No Multiple tools	Counseling only (individualized nutritional counseling every two weeks) Outpatient Multiple providers Not applicable	Did not receive any nutritional counseling, only interim medical visits by a physician	Weight or body composition changes Changes in nutritional status Adverse events Survival
Movahed, 2020 ⁶⁸ PMID NA Other Key Question 2, 3 Not conducted	100 67 years 49% female Race NR	Gastrointestinal % Stage IV disease NR Multiple therapies No Other tool	Counseling only (individualized dietary plan and nutrition education) Setting NR Other provider Not applicable	General dietary advice at the beginning of the treatment	Weight or body composition changes Changes in nutritional status Symptoms Functional status
Orell, 2019 ⁶⁹ (30937304) Europe Key Question 2 Not conducted	58 60 years % female NR Race NR	Head & Neck 64% Stage IV disease Multiple therapies No Other tool	Counseling only (intensive nutritional counseling) Outpatient Dietitian/nutritionist Not applicable	Baseline nutritional counseling, that included one dietetic consultation before chemoradiotherapy	Weight or body composition changes Changes in nutritional status Adverse events Survival
Pettersson, 2012 ⁷⁰ (22633817) Region NR Key Question 2, 3 Not conducted	130 66 years 0% female Race NR	Other % Stage IV disease NR Radiation alone No Other tool	Counseling only (advised to increase soluble fiber and decrease lactose in diet) Setting NR Dietitian/nutritionist Not applicable	Standard care (advised to continue normal diet)	Adverse events Symptoms Functional status
Qiu, 2020 ⁷¹ (31526964) Asia Key Question 2, 3 Not conducted	85 67 years 36% female Race NR	Gastrointestinal % Stage IV disease NR Multiple therapies No Multiple tools	Counseling only (whole- course nutrition management) Inpatient Multiple providers Not applicable	The control group was treated with general nutritional supplementation	Weight or body composition changes Adverse events Length of stay Quality of life Symptoms

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Ravasco, 2012 ⁷² (23134880) Europe Key Question 2, 3 Not conducted	89 64 years 46% female Race NR	Gastrointestinal % Stage IV disease NR Radiation alone No Other tool	Counseling only (individualized nutritional counseling and education about regular foods weekly for 6 weeks) Outpatient Provider NR Not applicable	Group 2: dietary supplements and consumed usual diet of regular foods Group 3: consumed usual diet of regular foods	Weight or body composition changes Changes in nutritional status Adverse events Survival Quality of life Symptoms
Regueme, 2021 ⁷³ (32435967) Europe Key Question 2, 3 Not conducted	283 77 years 50% female Race NR	Multiple cancers % Stage IV disease NR Chemotherapy alone Yes Other tool	Counseling only (usual care + support to increase intake) Outpatient Dietitian/nutritionist Not applicable	Usual care: received nutritional care routinely used in the cancer treatment setting	Weight or body composition changes Quality of life Symptoms Functional status
Tu, 2013 ⁷⁴ (23320428) Asia Key Question 2 Not conducted	537 64 years % female NR Race NR	Multiple cancers 21% Stage IV disease Cancer treatment type NR No Multiple tools	Counseling only (30 minutes of nutrition counseling and meal planning during first week post-discharge) Outpatient Dietitian/nutritionist Not applicable	No nutrition consulting or meal planning	Weight or body composition changes Changes in nutritional status
Um, 2014 ⁷⁵ (24906838) Asia Key Questions 2, 3 Not conducted	87 60 years 36% female Race NR	Multiple cancers % Stage IV disease NR Radiation alone Yes Other tool	Counseling only (at least 3 sessions of individualized dietary counseling concurrent with radiation) Outpatient Dietitian/nutritionist Not applicable	One routine 20-min education session by a dietitian within 4 days of starting RT	Weight or body composition changes Changes in nutritional status Survival Quality of life

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
van der Werf, 2020 ⁷⁶ (32037284) Europe Key Questions 2, 3 Not conducted	107 65 years 37% female Race NR	Gastrointestinal 100% Stage IV disease Chemotherapy alone No Screening tool NR	Counseling only (individualized nutritional counseling) Outpatient Dietitian/nutritionist Not applicable	Usual care	Weight or body composition changes Survival Treatment tolerance Quality of life Functional status
Zhang Z, 2022 ⁷⁷ (34984549) Asia Key Question 2,3 Not conducted	468 60 years 42% female Race NR	Multiple Cancers 37% Stage IV disease Radiation Therapy No Multiple tools	Nutrition education (guideline-based nutrition education counseling that was tailored and dynamic) Inpatient Nurse Not applicable	No nutrition education	Changes in nutritional status Treatment tolerance

Abbreviations: KQ = Key Question; PMID = PubMed Identification Number; NR = not reported; NA = not available; MNA-SF = mini nutritional assessment – short form. *For select studies only

†Reports median age when mean is not available

Dietary Supplements

Table E.2. Characteristics of included studies: use of dietary supplements during cancer treatment (KQ2)

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Cheng, 2021 ⁷⁸ (34395492) Asia Key Question 2 Not conducted	60 64 years 45% female Race NR	Other (Lung) 24% Stage IV disease Chemotherapy No NRS 2002	Single Supplement (Omega-3 fatty acid supplement (EPA 1.6 g/day + DHA 0.8 g/day))x 12 weeks Setting NR Provider NR Oral	Placebo gel capsule x 12 weeks	Weight or body composition changes

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Chitapanarux, 2020 ⁷⁹ (31146957) Asia Key Question 2 Not conducted	88 56 years 44% female Race NR	Multiple Cancers % Stage IV Disease NR Multiple therapies No Screening tool NR	Supplements (added arginine, glutamine, and fish oil supplements) Outpatient Nurse Oral	Regular diet	Readmissions or emergency room visits Survival
da Gama Torres, 2008 ⁸⁰ (18317456) Other Key Question 2 Not conducted	53 37 years 53% female Race NR	Other % Stage IV disease NR Multiple therapies No Screening tool NR	Single Supplements (glutamine supplemented PN) Inpatient Other Parenteral	Standard PN for 7+ days	Adverse events Survival Length of stay
de Luis, 2004 ⁸¹ (15138461) Europe Key Question 2 High	90 61 years 93% female Race NR	Head & Neck 61% Stage IV disease Surgery alone No Screening tool NR	Single Supplements (enteral diet supplement with arginine and fiber) Inpatient Provider NR Enteral	Enteral diet supplement with fiber, 12 hr postop to 10+ days	Weight or body composition changes Adverse events Length of stay
de Luis, 2005 ⁸² (15802904) Europe Key Question 2 High	73 62 years 7% female Race NR	Head & Neck 58% Stage IV disease Surgery alone No Screening tool NR	Single Supplements (2 cans/day fatty acid- enhanced oral supplement) Setting NR Provider NR Oral	2 cans/d arginine- enhanced supplement x 12 wks	Weight or body composition changes Changes in nutritional status Adverse events Survival
de Luis, 2007 ⁸³ (16929239) Europe Key Question 2 Medium	72 62 years 90% female Race NR	Head & Neck 75% Stage IV disease Surgery alone No Other metric	Single Supplements (arginine enhanced enteral supplement via NG tube) Inpatient Provider NR Enteral	Enteral/NG formula without added arginine	Weight or body composition changes Adverse events Length of stay
de Luis, 2008 ⁸⁴ (18700689) Europe Key Question 2 Medium	65 63 years 9% female Race NR	Head and Neck 65% Stage IV disease Surgery alone Yes Other metric	Supplements (Supplement with a high ratio of omega- 3 and omega-6) Outpatient Dietitian/nutritionist Enteral	Supplement with a low ratio of omega-3 and omega-6	Weight or body composition changes Adverse events

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
de Luis, 2009 ⁸⁵ (31496287) Europe Key Question 2 Medium	72 62 years 20% female Race NR	Head & Neck 74% Stage IV disease Surgery alone No Screening tool NR	Supplements (enteral diet with arginine) Outpatient Provider NR Enteral	Isocaloric, isonitrogenous enteral formula	Weight or body composition changes Adverse events Length of stay
de Luis, 2010 ⁸⁶ (21284343) Europe Key Question 2 Medium	115 62 years 22% female Race NR	Head & Neck 71% Stage IV disease Surgery alone No Screening tool NR	Single Supplements (enteral diet with high arginine dose) Outpatient Provider NR Enteral	Enteral diet supplements with medium dose of arginine	Weight or body composition changes Adverse events Length of stay
de Luis, 2015 ⁸⁷ (25855918) Europe Key Question 2 High	84 62 years 25% female Race NR	Head & Neck 45% Stage IV disease Surgery alone No Screening tool NR	Supplements (One group with medium dose of arginine, another treatment group with high dose of arginine) Outpatient Provider NR Enteral	Enteral diet with a low dose of arginine (5.8 g per day)	Weight or body composition changes Adverse events Length of stay
Farreras, 2005 ⁸⁸ (15681102) Europe Key Question 2 Medium	66 68 years 47% female Race NR	Gastrointestinal % Stage IV disease NR Surgery alone No Screening tool NR	Supplements (Enteral formula with arginine, omega-3 FAS and RNA (Impact)) Inpatient Multiple providers Enteral	Control formula same timing & route (Isosource) 12 hr postop x 7 d (no other food)	Adverse events Survival Length of stay
Fietkau, 2013 ⁸⁹ (23765693) Europe Key Question 2, 3 Not conducted	111 56 years 16% female Race NR	Multiple cancers % Stage IV disease NR Multiple therapies No Other tool	Single Supplement (Disease-specific enteral formula (Supportan)) Outpatient Physician Enteral	500 mL of the enteral standard nutrition Fresubin energy fibre	Weight or body composition changes Changes in nutritional Status Quality of life Functional status
Golkhalkhali, 2018 ⁹⁰ (28857425) Asia Key Question 2, 3 Medium	140 Age NR % female NR Asian	Gastrointestinal % Stage IV disease NR Chemotherapy alone No Screening tool NR	Single Supplement (Microbial cell prep and omega-3 fatty acids) Inpatient Provider NR Oral	Placebo	Weight or body compositions changes Quality of life Symptoms

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Iwase, 2016 ⁹¹ (26105516) Asia Key Question 2, 3 Not conducted	59 Age NR 100% female Race NR	Other 14% Stage IV disease Chemotherapy alone No Screening tool NR	Supplements (Amino acid jelly (Inner Power) oral diet supplement) Setting NR Provider NR Oral	Regular care	Adverse events Quality of life Symptoms
Jiang, 2010 ⁹² (20473991) Asia Key Question 2 Low	206 Age NR 35% female Race NR	Gastrointestinal % Stage IV disease NR Surgery alone No Screening tool NR	Single Supplements (Soybean plus fish oil emulsion) Inpatient Nurse Parenteral	Soybean oil only (PN via central catheter)	Weight or body Composition changes Adverse events Length of stay
Klek, 2005 ⁹³ (15906909) Europe Key Question 2 High	105 61 years 43% female Race NR	Gastrointestinal 0% Stage IV disease Surgery alone No Other metric	Supplements (One group PN and glutamine, another treatment group PN and Omega 3-FA (Omegaven)) Setting NR Provider NR Enteral	Standard PN	Weight or body composition changes Changes in nutritional Status Length of stay
Lobo, 2006 ⁹⁴ (16777271) Adiamah, 2021 ⁹⁵ (34656029) Europe Key Question 2 Low	120 66 years 23% female Race NR	Gastrointestinal 0% Stage IV Disease Surgery alone No Screening tool NR	Supplements (Higher arginine, glutamine, cysteine and Omega-3-FAs (Stresson)) Inpatient Provider NR Enteral	Standard isocaloric, isonitrogenous diet	Adverse events Survival Length of stay
Lu, 2011 ⁹⁶ (21396307) Asia Key Question 2 Medium	50 67 years 32% female Race NR	Gastrointestinal 16% stage IV disease Surgery alone No Screening tool NR	Single Supplements (Glutamine enriched TPN) Inpatient Physician Parenteral	Standard TPN with 1.5 gm/kg/day of amino acids solution	Adverse events
Matsuda, 2017 ⁹⁷ (28138734) Asia Key Question 2 High	87 64 years 25% female Race NR	Gastrointestinal % Stage IV disease NR Multiple therapies No Screening tool NR	Single Supplements (Diet enriched with EPA, GLA and antioxidants) Inpatient Provider NR Enteral	Standard isocaloric, isonitrogenous diet	Weight or body composition changes Adverse events Survival Length of stay

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Miyata, 2017 ⁹⁸ (27644137) Asia Key Question 2 Medium	61 65 years 15% female Race NR	Gastrointestinal 23% Stage IV disease Chemotherapy alone No Screening tool NR	Single Supplements (Omega-3 rich EN support) Setting NR Physician Enteral	Omega-3 poor EN support	Weight or body composition changes Adverse events
Pathak, 2019 ⁹⁹ (31410338) Asia Key Question 2 Medium	60 Age NR 10% female Race NR	Head & Neck % Stage IV disease NR Multiple therapies No Screening tool NR	Single Supplements (glutamine supplementation) Setting NR Provider NR Oral	No glutamine	Weight or body composition changes Adverse events Readmissions or emergency room visits
Pottel, 2014 ¹⁰⁰ (25293388) Europe Key Question 2 Medium	85 61 years 16% female Race NR	Head & Neck % Stage IV Disease NR Multiple therapies No Other tool	Supplements (echium oil supplementation twice daily for 7 weeks) Setting NR Multiple providers Oral	Received a placebo, containing the same volume of n-3 PUFA deficient sunflower oil (Sunflower Oil High Oleic)	Weight or body composition changes Changes in nutritional Status Adverse events
Sun, 2008 ¹⁰¹ (18376691) Asia Key Question 2 Medium	64 65 years 31% female Race NR	Gastrointestinal 0% Stage IV disease Surgery alone Yes Other tool	Single Supplement (BCAA enriched TPN) Inpatient Provider NR Parenteral	Standard TPN from POD 1-6	Weight or body composition changes Adverse events Survival
Takeshita, 2009 ¹⁰² (19285598) Asia Key Question 2 Medium	58 70 years 29% female Race NR	Gastrointestinal % Stage IV disease NR Chemotherapy alone No Screening tool NR	Single Supplements (snack with 50 g of BCAA for 14 days) Outpatient Provider NR Oral	No BCAA snack	Weight or body composition changes Survival Length of stay

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Tanca, 2009 ¹⁰³ PMID NA Europe Key Question 2, 3 Not conducted	475 62 years 40% female Race NR	Multiple cancers 95% Stage IV disease Cancer treatment type NR Yes Multiple tools	Supplements (5 treatment groups: 1) Progestational agent, 2) EPA, 3) L- carnitine, 4) Thalidomide, 5) Progestational agent plus pharmacologic nutritional support and L- carnitine and thalidomide) Other Provider NR Oral	No placebo or control	Weight or body composition changes Quality of life Symptoms Functional status
Tumas, 2020 ¹⁰⁴ (32184025) Europe Key Question 2 Not conducted	70 63 years 47% female Race NR	Other 0% Stage IV disease Surgery alone No NRS-2002	Supplements (Supplemented with arginine (Cubitan Nutricia) for 5 days) Outpatient Provider NR Oral	Unclear	Changes in nutritional Status Adverse events
Wang, 2017 ¹⁰⁵ (28927119) Asia Key Question 2, 3 High	94 59 years 37% female Race NR	Gastrointestinal 100% Stage IV disease Multiple therapies No Screening tool NR	Single Supplements (glutamine enriched nutritional support) Inpatient Provider NR Parenteral	Basic nutritional support	Adverse events Quality of life
Wang, 2013 ¹⁰⁶ (24379010) Asia Key Question 2 Low	94 58 years 24% female Race NR	Gastrointestinal 0% Stage IV disease Surgery alone No NRS-2002	Single Supplements (EN with PN containing olive-oil based lipid emulsion) Inpatient Provider NR Parenteral	EN with medium/long chain triglyceride emulsion based PN after tumor resection for ≥7 days	Adverse events Survival Length of stay
Wang, 2010 ¹⁰⁷ (21208095) Asia Key Question 2, 3 Medium	229 56 years 36% female Race NR	Gastrointestinal % Stage IV disease NR Surgery alone No Screening tool NR	Supplements (EN enriched with medium-chain triglycerides and protein) Inpatient Multiple providers Enteral	Isocaloric EN for ≥5 days started within 48- 72 h post-surgery	Adverse events Length of stay Symptoms
Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
---	--	--	---	---	---
Wu, 2012 ¹⁰⁸ (23554777) Asia Key Question 2 High	63 52 years 21% female Race NR	Gastrointestinal % Stage IV disease NR Surgery alone No Screening tool NR	Single Supplements (TPN with omega-3 supplement) Inpatient Provider NR Parenteral	TPN with soybean oil (not omega 3) for >5 days	Adverse events Survival Length of stay
Yang, 2022 ¹⁰⁹ (35500316) Asia Key Question 2 Medium	120 59 years 32% female Race/Ethnicity NR	Gastrointestinal 0% Stage IV disease Surgery No Screening tool NR	Nutrition support: EN (NJ tube started POD 2) & (PN with n-3 fatty acids POD 1- 5) Inpatient Provider NR EN + PN	Nutrition support: EN (NJ tube started POD 2) & PN	Adverse events Readmissions/Emergency room visits Length of stay Survival
Yeh, 2013 ¹¹⁰ (23562359) Asia Key Question 2 High	68 54 years 1% female Race NR	Head & Neck 79% Stage IV diseaseRadiation alone Yes Screening tool NR	Single Supplements (Nutritional supplement including omega-3 fatty acids, glutamine, selenium for 3 months) Outpatient Multiple providers Oral	Isocaloric supplementation for 3 mo. (2 mo. during treatment, 1 mo. After)	Weight or body composition changes
Zhang, 2017 ¹¹¹ (27614675) Asia Key Question 2 Not conducted	320 49 years 27% female Race NR	Other 11% Stage IV Disease Surgery alone Yes NRS-2002	Single Supplements (PN supplemented with omega- 3 fatty acids) Inpatient Provider NR Parenteral	Standard TPN starting on POD 1 for 5 consecutive days	Adverse events Survival Length of stay
Zhu, 2012 ¹¹² (22340541) Asia Key Question 2 Low	57 70 years 42% female Race NR	Gastrointestinal % Stage IV disease NR Surgery alone No Screening tool NR	Single Supplements (PN with fish oil and soybean oil) Inpatient Provider NR Parenteral	PN POD 1-8 (1.2 g/kg soybean oil)	Adverse events Length of stay

Abbreviations: KQ = Key Question; PMID = PubMed Identification Number; NR = not reported; FA = fatty acid; PN = parenteral nutrition; EN = enteral nutrition; TPN = total parenteral nutrition; POD = post-operative day; EPA = eicosapentaenoic acid; NRS = nutritional risk screening; BCAA = branched-chain amino acid; PUFA = polyunsaturated fat; GLA = c-linolenic acid; MNA-SF = Mini Nutritional Assessment – Short Form; NG = nasogastric.

*For select studies only.

†Reports median age when mean is not available.

Special Diets

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Gardner, 2008 ¹¹³ (18955453) North America Key Question 2, 3 Not conducted	153 64 years % female NR Race NR	Other % Stage IV disease NR Chemotherapy alone No Screening tool NR	Special Diets (diet with raw fruits and vegetables) Inpatient Provider NR Oral	Diet without raw fruits & vegetables	Weight or body composition changes Adverse events Survival Symptoms
Ingersoll, 2010 ¹¹⁴ (20189927) North America Key Question 2, 3 Not conducted	77 54 years 81% female Race NR	Multiple cancers % Stage IV disease NR Chemotherapy alone No Screening tool NR	Special Diets (standard medical CINV management + grape juice) Outpatient Provider NR Oral	Standard medical CINV management	Adverse events Quality of life Symptoms
Jatoi, 2016 ¹¹⁵ (27039205) North America Key Question 2, 3 Not conducted	118 Age NR 52% female Race NR	Other 100% Stage IV disease Multiple therapies No Screening tool NR	Special Diets (white wine twice daily) Outpatient Provider NR Oral	Nutritional supplement like boost or ensure	Adverse events Quality of life Functional status
Khodabakhshi, 2020 ¹¹⁶ (314996287) Khodabakhshi, 2020 ¹¹⁷ (32828130) Other Key Question 2, 3 Not conducted	80 Age NR 100% female Race NR	Other 26% Stage IV disease Chemotherapy alone No Screening tool NR	Special Diets (ketogenic) Outpatient Dietitian/nutritionist Oral	Standard diet	Weight or body composition changes Quality of life
Lugtenberg, 2021 ¹¹⁸ (33179154) Europe Key Question 2, 3 Not conducted	129 50 years 100% female Race NR	Other % Stage IV disease NR Chemotherapy alone No Screening tool NR	Special Diets (calorie content declined for 3 days before and the day of treatment) Setting NR Physician Other	Regular diet	Weight or body composition changes Quality of life Symptoms Functional status

 Table E.3. Characteristics of included studies: use of special diets during cancer treatment (KQ2)

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Miyakawa, 2019 ¹¹⁹ (30554216) Asia Key Question 2, 3 Not conducted	100 71 years 66% female Race NR	Gastrointestinal % Stage IV disease NR Surgery alone No Screening tool NR	Special Diets (rice porridge) Inpatient Provider NR Oral	Liquid diet with gradual transition to solid food	Length of stay Quality of life Symptoms
Voss, 2020 ¹²⁰ (32619561) Europe Key Question 2, 3 Not conducted	50 57 years % female NR Race NR	Other % Stage IV disease NR Radiation alone No Screening tool NR	Special Diets (3 days of ketogenic diet, 3 days of intermittent fasting, then 3 days of ketogenic diet) Outpatient Dietitian/nutritionist Oral	Calorically unrestricted diet	Survival Symptoms
Wedlake, 2012 ¹²¹ (1605807) Europe Key Question 2, 3 Not conducted	117 65 years 32% female Race NR	Multiple cancers % Stage IV disease NR Radiation alone No Screening tool NR	Special Diets (Low fat and moderate fat intake) Outpatient Provider NR Oral	Normal fat (LCT – 40% of total energy)	Weight or body composition changes Quality of life Symptoms Functional status

Abbreviations: KQ = key question; PMID = PubMed Identification Number; NR = not reported; CINV = hemotherapy-indKKuced nausea and vomiting; LCT = long chain triglycerides. *For select studies only. †Reports median age when mean is not available.

Route or Timing of Nutritional Interventions

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Berkelmans, 2020 ¹²² (31090563) Europe Key Question 2 Medium	148 65 years 14% female Race NR	Gastrointestinal 0% Stage IV disease Surgery alone No Screening tool NR	Nutrition therapy (started directly with a liquid oral diet starting on the day of surgery) Inpatient Physician Oral	Delay in start of oral intake; only allowed to drink clear liquids; received tube feeding	Weight or body composition changes Changes in nutritional status Adverse events Readmissions or emergency room visits Survival Length of Stay
Boelens, 2014 ¹²³ (24169163) Europe Key Question 2 High	123 65 years 32% female Race NR	Gastrointestinal % Stage IV disease NR Multiple therapies No Screening tool NR	Nutrition therapy (EEN) Inpatient Physician Other	EPN	Adverse events Length of stay
Bozzetti, 2001 ¹²⁴ (11705560) Europe Key Question 2 Low	317 65 years 42% female Race NR	Gastrointestinal % Stage IV disease NR Surgery alone Yes Other metric	Nutrition therapy (enteral nutrition postop) Inpatient Provider NR Enteral	Parenteral nutrition postop	Adverse events Survival Length of stay
Braga, 2001 ¹²⁵ (11246300) Europe Key Question 2 Medium	257 64 years 46% female Race NR	Gastrointestinal % Stage IV disease NR Surgery alone No Other tool	Nutrition therapy (EEN postop until oral intake was GE 800 kcal/day) Inpatient Provider NR Enteral	TPN postop until oral intake was GE 800 kcal/cay	Adverse events Survival Length of stay
Dag, 2011 ¹²⁶ (22189721) Asia Key Question 2 High	199 62 years 43% female Race NR	Gastrointestinal % Stage IV disease NR Surgery alone No Screening tool NR	Nutrition therapy (early oral feeding) Inpatient Provider NR Oral	Fasted until first flatus or stools	Adverse events Length of stay

Table E.4. Characteristics of included studies: route or timing of nutritional interventions during cancer treatment (KQ2)

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Feo, 2004 ¹²⁷ (15144242) Europe Key Question 2, 3 Medium	100 68 years % female NR Race NR	Gastrointestinal 0% Stage IV disease Surgery alone No Screening tool NR	Nutrition therapy (clear oral liquids) Inpatient Provider NR Oral	Fasting until passage of flatus then NG diet	Adverse events Survival Length of stay Symptoms
Gao, 2019 ¹²⁸ (30941970) Asia Key Question 2 High	198 55 years 38% female Race NR	Gastrointestinal % Stage IV disease NR Surgery alone No Screening tool NR	Nutrition therapy (oral fluid diet started on the POD 2) Inpatient Provider NR Oral	Nasogastric tube 30 min before surgery until the recovery of gastrointestinal function	Adverse events
Huang, 2015 ¹²⁹ (1594390) Asia Key Question 2 High	105 Age NR 35% female Race NR	Gastrointestinal % Stage IV disease NR Surgery alone No MNA	Nutrition therapy (EEN) Setting NR Provider NR Other	TPN and EN + PN	Changes in nutritional status Adverse events Length of stay
Hyltander, 2005 ¹³⁰ (15880316) Europe Key Question 2 High	126 62 years 34% female Race NR	Gastrointestinal % Stage IV disease NR Surgery alone No Screening tool NR	(PN and EN) Multiple settings Multiple providers Other	Standard oral solution	Weight or body composition changes Changes in nutritional status Adverse events Survival
Kita, 2021 ¹³¹ (33526287) Asia Key Question 2 High	91 Age NR 22% female Race NR	Gastrointestinal 35% Stage IV disease Chemotherapy alone No Screening tool NR	Nutrition therapy (EN) (added triglycerides/fatty acids) Setting NR Provider NR Oral	Parenteral nutrition	Weight or body composition changes Adverse events

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Kurbanalievich, 2020 ¹³² PMID NA Asia Key Question 2 Medium	80 Age NR 45% female Race NR	Gastrointestinal % Stage IV disease NR Cancer treatment type NR Limited to malnourished NR Multiple tools	Nutrition therapy (EN from POD 2) Inpatient Provider NR Enteral	Isolated parenteral nutrition from POD 1	Weight or body composition changes
Li, 2015 ¹³³ (26011337) Asia Key Question 2 High	400 Age NR 46% female Race NR	Gastrointestinal % Stage IV disease NR Surgery alone No Screening tool NR	Nutrition therapy (EEN for 7 days) Inpatient Provider NR Enteral	Postoperative parenteral nutrition for 7 days	Changes in nutritional status Length of stay
Liu, 2011 ¹³⁴ (21669582) Asia Key Question 2 Not conducted	58 Age NR 43% female Race NR	Other % Stage IV disease NR Surgery alone No Screening tool NR	Nutrition therapy (EN at least 6 days postop) Inpatient Provider NR Enteral	TPN formulas for 7 days postop, until the TPN was completely replaced by oral intake	Changes in nutritional status Adverse events Length of stay
Luo, 2018 ¹³⁵ (28666095) Asia Key Question 2 High	78 62 years 21% female Race NR	Gastrointestinal % Stage IV disease NR Surgery alone No Screening tool NR	Nutrition therapy (EEIN) Inpatient Provider NR Other	Parenteral nutrition	Weight or body composition changes Adverse events Survival Length of stay
Ma, 2020 ¹³⁶ (33362392) Asia Key Question 2 Medium	60 59 years 41% female Race NR	Gastrointestinal % Stage IV disease NR Surgery alone Yes NRS-2002	Nutrition therapy (EEN) Inpatient Physician Enteral	Postoperative total parenteral nutrition	Adverse events Length of stay
Mahmoodzadeh, 2015 ¹³⁷ (24875466) Asia Key Question 2 High	109 65 years 47% female Race NR	Gastrointestinal 4% Stage IV disease Multiple therapies No Screening tool NR	Nutrition therapy (early oral feeding) Inpatient Physician Oral	Late oral feeding (kept nil per os)	Adverse events Readmissions or emergency room visits Length of stay

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Minig, 2009 ¹³⁸ (19760046) Europe Key Question 2 Not conducted	143 56 years 100% female Race NR	Multiple cancers % Stage IV disease NR Surgery alone No Other tool	Nutrition therapy (early oral feeding) Inpatient Physician Oral	Traditional oral feeding	Adverse events Length of stay
Perinel, 2016 ¹³⁹ (27429039) Europe Key Question 2 High	204 65 years 39% female Race NR	Gastrointestinal % Stage IV disease NR Surgery alone No Other tool	Nutrition therapy (EEN POD 1) Inpatient Provider NR Enteral	TPN starting on POD 1 and continued until oral intake reached 60% of nutritional requirements	Adverse events Survival Length of stay
Roberts, 2003 ¹⁴⁰ (13130320) North America Key Question 2, 3 Not conducted	55 44 years 100% female Race NR	Other 58% Stage IV disease Multiple therapies No Multiple tools	Nutrition therapy (TPN) Inpatient Provider NR Parenteral	Oral diet post-surgery with IV fluids	Weight or body composition changes Adverse events Length of stay Quality of life
Ryu, 2009 ¹⁴¹ (19255706) Asia Key Question 2 Not conducted	81 64 years 11% female Race NR	Head & Neck 31% Stage IV disease Surgery alone No Screening tool NR	Nutrition therapy (nasogastric tube nutrition) Inpatient Dietitian/nutritionist Enteral	Total parenteral nutrition commenced 24h after surgery	Weight or body composition changes Adverse events Length of stay
Sadasivan, 2012 ¹⁴² (22973911) Asia Key Question 2, 3 Not conducted	100 Age NR % female NR Race NR	Head & Neck 50% Stage IV Disease Multiple therapies No Screening tool NR	Nutrition therapy (percutaneous endoscopic gastrostomy) Setting NR Multiple providers Enteral	Nasogastric tube administration	Weight or body composition changes Adverse events Treatment tolerance
Seven, 2003 ¹⁴³ (12782826) Asia Key Question 2 Not conducted	67 56 years 8% female Race NR	Head & Neck % Stage IV disease NR Surgery alone No Screening tool NR	Nutrition therapy (clear liquid diet began POD 1) Inpatient Provider NR Oral	Tracheoesophageal puncture feeding, oral began POD 7	Adverse events Length of stay

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Sousa, 2014 ¹⁴⁴ (24736040) Other Key Question 2 Not conducted	89 63 years 8% female Race NR	Head & Neck 97% Stage IV diseaseSurgery alone No Other tool	Nutrition therapy (early oral feeding beginning 24 hr after postop) Inpatient Provider NR Oral	Late oral feeding (began postop day 7)	Adverse events
Sun, 2018 ¹⁴⁵ (28549015) Asia Key Question 2, 3 Medium	280 63 years 30% female Race NR	Gastrointestinal 0% Stage IV disease Surgery alone No NRS-2002	Nutrition therapy (early oral feeding starting on POD 1) Inpatient Multiple providers Oral	Late oral feeding starting on POD 7	Weight or body composition changes Adverse events Survival Length of stay Quality of life Symptoms
Tao, 2020 ¹⁴⁶ (31512101) Asia Key Question 2, 3 Medium	120 64 years 52% female Race NR	Gastrointestinal 11% Stage IV diseaseSurgery alone No Screening tool NR	Nutrition therapy (jejunostomy feeding starting on POD 1 until 3 wks past discharge) Inpatient Provider NR Enteral	Nasogastric feeding starting on POD 1 until discharge	Weight or body composition changes Adverse events Survival Quality of life
van Barneveld, 2016 ¹⁴⁷ (26937858) Europe Key Question 2 High	123 64 years 32% female Race NR	Gastrointestinal % Stage IV disease NR Surgery alone No MUST	Nutrition therapy (EEN) Inpatient Physician Enteral	EPN (jugular vein catheter) followed by parenteral nutrition	Adverse events Length of stay
Wang, 2019 ¹⁴⁸ (31239762) Asia Key Question 2 Medium	100 54 years 29% female Race NR	Gastrointestinal 0% Stage IV disease Surgery alone No NRS-2002	Nutrition therapy (early oral feeding POD1) Inpatient Physician Oral	Delayed oral feeding starting on POD 4	Adverse events Survivall
Wang, 2018 ¹⁴⁹ (29896256) Asia Key Question 2 Medium	129 48 years 49% female Race NR	Gastrointestinal 0% Stage IV disease Surgery alone No Screening tool NR	Nutrition therapy (EEN within 24 h after operation) Inpatient Provider NR Other	Total parenteral nutrition	Weight or body compostion changes Adverse events Length of stay

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Xiao-Bo, 2014 ¹⁵⁰ (24504222) Asia Key Question 2 Medium	120 64 years 30% female Race NR	Gastrointestinal % Stage IV disease NR Surgery alone Yes Other tool	Nutrition therapy (EN for 7 days starting on POD 1) Inpatient Provider NR Enteral	PN for 7 days postoperatively	Adverse events Length of stay
Xu, 2020 ¹⁵¹ PMID NA Asia Key Question 2 High	108 53 years 44% female Race NR	Gastrointestinal 0% Stage IV disease Surgery alone No Screening tool NR	Nutrition therapy (EN 2 days post op) Inpatient Provider NR Enteral	PN postop, half total amount on the first day and full amount on the next day	Changes in nutritional status
Zhang, 2019 ¹⁵² (31551229) Asia Key Question 2 High	140 67 years 36% female Race NR	Gastrointestinal 0% Stage IV disease Surgery alone No Screening tool NR	Nutrition therapy (PN for 7 days) Inpatient Provider NR Parenteral	Central venous catheter support for 7 days	Weight or body composition changes Adverse events Length of stay

Abbreviations: KQ = Key Question; PMID = PubMed Identification Number; NR = not reported; NA = not available; NRS = nutrition risk screening; MUST = malnutrition universal screening tool; EEN = early enteral nutrition; EPN = early parenteral nutrition; TPN = total parenteral nutrition; EN = enteral nutrition; PN = parenteral nutrition; EEIN = early enteral immunonutrition; POD = post operative day.

*For select studies only.

†Reports median age when mean is not available.

Nutrition Support Including Oral Nutrition Supplements

Study (PMID) Region Key Question RoB Rating* Akita, 2019 ¹⁵³ (314451252) Asia Key Question 2 Not conducted	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity 62 67 years 56% female Race NR	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool Other % Stage IV disease NR Multiple therapies Yes Screening tool NR	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention Nutrition (2 bottles (440ml) per day (560 kcal/day) of EPA-enriched nutritional supplement) Outpatient Provider NR Oral	Comparison Description Received nutritional guidance three times - immediately before, at 3 weeks and immediately after radiotherapy	Outcomes Weight or body composition changes Adverse events
Baker, 2015 ¹⁵⁴ (25827292) Other Key Question 2, 3 Not conducted	109 63 years 100% female Race NR	Other 10% Stage IV disease Multiple therapies Yes Other tool	Nutrition (intraoperative nasojejunal tube placement and enteral feeding until adequate oral intake could be maintained) Inpatient Multiple providers Enteral	Postoperative diet as tolerated	Adverse events Length of stay Quality of life
Barlow, 2011 ¹⁵⁵ (21601319) Europe Key Question 2 Medium	121 64 years 32% female Race NR	Gastrointestinal 3% Stage IV disease Multiple therapies No Screening tool NR	Nutrition (early enteral nutrition was delivered via a needle catheter jejunostomy) Inpatient Physician Oral	Kept nil by mouth	Adverse events Readmissions or Emergency room visits Length of stay

Table E.5. Characteristics of included studies: use of nutrition support including oral nutrition supplements during cancer treatment (KQ2)

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Bouleuc, 2020 ¹⁵⁶ (32212354) Europe Key Question 2, 3 Not conducted	148 66 years 55% female Race NR	Multiple cancers % Stage IV disease NR Multiple therapies Yes Screening tool NR	Nutrition (parenteral nutrition was administered by central venous route using industrial ternary preparations and systematic daily addition of polyvitamins, trace elements, and electrolytes (sodium, potassium, vitamin K, magnesium, phosphorus), adapted as required) Inpatient Dietitian/nutritionist Parenteral	Oral feeding	Weight or body composition changes Adverse events Survival Quality of life
Cereda, 2018 ¹⁵⁷ (29111172) Europe Key Question 2, 3 Not conducted	159 65 years 28% female Race NR	Head & Neck 29% Stage IV disease Multiple therapies No Screening tool NR	Nutrition (2 bottles/day 250 mL of a ready-to-use energy-dense, high-protein, omega-3 enriched oral formula plus counseling) Outpatient Dietitian/nutritionist Oral	Counseling only	Weight or body composition changes Treatment tolerance Quality of life
Chen, 2021 ¹⁵⁸ (33545764) Asia Key Question 2 Not conducted	60 70 years 13% female Race NR	Other Stage IV disease NR Multiple therapies No Other tool	Nutrition (orally administered enteral nutrition (Ensure) with regular diet Outpatient Provider NR Oral	Regular diet only	Changes in nutritional status
Chen T, 2021 ¹⁵⁹ (34237976) Asia Key Question 2, 3 High	106 68 years 21% female Race/ethnicity NR	Gastrointestinal 0% Stage IV disease Surgery No NRS 2002	Post-hospital enteral nutritional emulsion via J- tube: 200-500 ml x 3- 6x/day Outpatient (Home) Patient-delivered but directed by doctor, nurse & nutritionist J-tube x 1 month Enteral	Post-hospital gradual transition to soft to regular diet after 1 month (J tube in place x 1 mo, not used at home)	Weight or body composition changes Changes in nutritional status Symptoms

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Chu, 2018 ¹⁶⁰ (622367981) Asia Key Question 2 Medium	98 63 years 43% female Race NR	Gastrointestinal 0% Stage IV disease Surgery alone No NRS-2002	Nutrition (enteral and parenteral nutritional interventions) Inpatient Provider NR Other	Only received routine parenteral nutritional intervention	Changes in nutritional status Adverse events Length of stay
Deibert, 2016 ¹⁶¹ (27402372) North America Key Question 2 Not conducted	102 Age NR 18% female Other	Other 1% Stage IV Disease Surgery alone No Screening tool NR	Nutrition (clear liquid diet POD 1, access to regular diet POD 2 & beyond) Inpatient Provider NR Oral	Standard postoperative care not part of a specific fast-track pathway, including nothing by mouth status until either flatus or a bowel movement, whichever occurred first	Adverse events Readmissions or emergency room visits Survival Length of stay
Faccio, 2021 ¹⁶² (32363940) Other Key Question 2, 3 Not conducted	85 59 years 60% female Race NR	Multiple Cancers % Stage IV disease NR Chemotherapy alone No Screening tool NR	Nutrition (nutritional supplement with counseling) Outpatient Dietitian/nutritionist Oral	Nutritional counseling alone	Weight or body composition changes Adverse events Quality of life
Gavazzi, 2016 ¹⁶³ (27391922) Europe Key Question 2, 3 High	79 Age NR 38% female Race NR	Gastrointestinal 0% Stage IV disease Multiple therapies Yes NRS-2002	Nutrition (home enteral nutrition) Inpatient Multiple providers Enteral	Specific nutritional indications including total energy and protein requirements were provided	Weight or body composition changes Treatment tolerance Quality of life
Huang, 2020 ¹⁶⁴ (33032180) Asia Key Question 2, 3 Not conducted	114 50 years 27% female Asian	Head & Neck 0% Stage IV disease Multiple therapies No NRS-2002	Nutrition (prophylactic oral nutritional supplements) Other Dietitian/nutritionist Oral	Regular diet	Weight or body composition changes Changes in nutritional status Adverse events Survival Treatment tolerance Quality of life Symptoms

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes	
Imamura, 2016 ¹⁶⁵ (27084538) Kimura, 2019 ¹⁶⁶ (31788653) Asia Key Question 2 High	112 Age NR 29% female Race NR	Gastrointestinal 1% Stage IV disease Surgery alone No Screening tool NR	Nutrition (oral elemental diet post-surgery 6-8 weeks) Outpatient Provider NR Oral	Regular diet	Weight or body composition changes Adverse events	
Jiang, 2018 ¹⁶⁷ (30633580) Asia Key Question 2, 3 Not conducted	100 Age NR 31% female Race NR	Head & Neck 57% Stage IV disease Multiple therapies No Other tool	Nutrition (oral nutritional supplement) Outpatient Multiple providers Oral	No oral nutritional supplement	Weight or body Composition changes Changes in nutritional Status Adverse events Quality of life	
Jin, 2018 ¹⁶⁸ (30205371) Asia Key Question 2, 3 High	80 Age NR 26% female Race NR	Gastrointestinal % Stage IV disease NR Surgery alone No Other metric	Nutrition (PN day 1 POD to 4-8 POD) Inpatient Provider NR Parenteral	Isotonic electrolyte solution	Changes in nutritional Status Quality of life	
Kanat, 2013 ¹⁶⁹ (23748819) Europe Key Question 2, 3 Not conducted	69 60 years 14% female Race NR	Multiple Cancers 70% Stage IV disease Multiple therapies No Screening tool NR	Nutrition ((1) megesterol acetate (MA) plus meloxicam; (2) MA plus meloxicam plus oral eicosapentaenoic acid (EPA)-enriched nutritional supplement or (3) meloxicam plus oral EPA- enriched nutritional supplement Outpatient Provider NR Oral	*No control (three treatment groups)	Weight or body composition changes Adverse events Quality of life	
Katada, 2021 ¹⁷⁰ (33009977) Asia Key Question 2 High	80 Age NR 15% female Race NR	Gastrointestinal % Stage IV Disease NR Multiple therapies No Screening tool NR	Nutrition (elemental supplementary group received ELENTAL® (160 g/day) orally 9 weeks after the start of chemotherapy Setting NR Multiple providers Oral	Non-supplemental group	Changes in nutritional status Adverse events	

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes			
Klek, 2008 ¹⁷¹ (18571296) Klek, 2011 ¹⁷² (21074910) Klek, 2017 ¹⁷³ (28336107) Europe Key Question 2 Low	196 61 years 37% female Race NR	Gastrointestinal 0% Stage IV disease Surgery alone No Screening tool NR	Nutrition (Enteral feeding was commenced 6 h after operation or the first 12 h, followed by infusion of Peptisorb or Reconvan until 7th day) Setting NR Provider NR Enteral	Oligopeptic, isocaloric diet	Adverse events Survival			
Klek, 2011 ¹⁷⁴ (21820770) Klek, 2014 ¹⁷⁵ (24178182) Europe Key Question 2 Medium	167 61 years 45% female Race NR	Gastrointestinal 0% Stage IV disease Surgery alone Yes Other metric	Nutrition (immunomodulating enteral group received Stresson; immunomodulating parenteral group received Omegaven) Inpatient Provider NR Oral	Standard enteral group oligopeptic diet Peptisorb; standard parenteral group	Adverse events Survival Length of stay			
Li, 2015 ¹⁷⁶ (26125924) Asia Key Question 2 High	300 Age NR 50% female Race NR	Gastrointestinal 0% Stage IV disease Surgery alone No Screening tool NR	Nutrition (500 mL of fractionated oral enteral nutrition) Inpatient Provider NR Oral	Traditional perioperative treatment nasoenteral feeding tube and postoperative IV)	Changes in nutritional status Adverse events Length of stay			
Li C, 2020 ¹⁷⁷ PMID NA Asia Key Question 2 Medium	78 62 years 45% female Race NR	Gastrointestinal 9% Stage IV disease Chemotherapy alone No Screening tool NR	Nutrition (enteral nutrition beginning on first day of treatment for 1 course of chemo) Inpatient Provider NR Enteral	Conventional nutrition support	Adverse events Survival			
Lyu, 2022 ¹⁷⁸ (35280748) Asia Key Question 2, 3 High	222 Age NR 21% female Race NR	Gastrointestinal 0% Stage IV disease Multiple therapies No Multiple tools	Nutrition (enteral nutrition with ONS (Nutrison (Nutricia)) Outpatient Nutrition support team (clinicians, nutritionists, pharmacologists and nutrition nurses) Oral	Unsystematic nutrition based on general eating conditions, blood tests & treatment toxicities	Weight or body composition changes Adverse events Survival Treatment Tolerance			

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes		
McGough, 2008 ¹⁷⁹ (18315590) Europe Key Question 2 Not conducted	50 60 years 58% female Race NR	Multiple cancers % Stage IV disease NR Radiation alone No Screening tool NR	Nutrition (patients in the intervention group were asked to replace one meal per day, equivalent to 33% of total caloric requirements, with elemental diet. A selection of E028 Extra ready to drink 250 mL cartons and E028 Extra flavoured powder sachets were provided) Setting NR Provider NR Oral	No intervention (ie, habitual diet during radiotherapy treatment)	Adverse events		
Meng, 2021 ¹⁸⁰ (32563598) Tan, 2021 ¹⁸¹ (32563599)	353 60 years 32% female Race NR	Gastrointestinal 7% Stage IV disease Surgery alone Yes	Nutrition (Oral nutritional supplements with dietary advice) Other	Dietary advice alone	Weight or body composition change		
Asia Key Question 2, 3 Low		NRS-2002	Oral		Symptoms		
Miyazaki, 2021 ¹⁸² (33835329) Asia Key Question 2 Medium	1003 67 years 35% female Race NR	Gastrointestinal 3% Stage IV disease Surgery alone No Screening tool NR	Nutrition (oral nutritional supplement) Inpatient Physician Oral	Regular diet without oral nutritional supplement	Weight or body composition changes Adverse events		
Nie, 2021 ¹⁸³ (35035754) Asia Key Question 2 High	97 59 years 42% female Race/ethnicity NR	Gastrointestinal 0% Stage IV disease Surgery No Screening tool NR	Oral enteral nutrition started 6-24 hour after surgery: 250 mL 5% glucose saline orally; EN POD 1 to 30 mL/kg/d Inpatient Provider NR Enteral	Parenteral nutrition (same calorie totals as Intervention group) Timing NR	Weight or body composition changes Adverse events Length of stay		

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes	
Ohkura, 2019 ¹⁸⁴ (30169605) Asia Key Question 2 High	67 Age NR 15% female Race NR	Gastrointestinal 15% Stage IV disease Surgery alone No Multiple tools	Nutrition (6 days of postoperative supplementation of the oligomeric formula) Inpatient Physician Enteral	6 days of postoperative supplementation of the polymeric formula	Weight or body composition changes Changes in nutritional status Adverse events Length of stay	
Okabayashi, 2011 ¹⁸⁵ (20852905) Asia Key Question 2, 3 High	96 67 years 70% female Race NR	Gastrointestinal % Stage IV disease NR Surgery alone No Screening tool NR	Nutrition (carbohydrate- and BCAA-enriched softpowder nutrient mixture) Inpatient Multiple providers Oral	Conventional diet with no supplementation	Weight or body composition changes Quality of life	
Ravasco, 2005 ¹⁸⁶ (15920748) Ravasco, 2005 ¹⁸⁷ (15684319) Europe Key Question 2, 3 Not conducted	75 Age NR % female NR Race NR	Head & Neck % Stage IV disease NR Radiation alone No Other tool	Nutrition ((1) diet plus regular food; (2) regular diet plus oral nutritional supplement) Outpatient Dietitian/nutritionist Oral	Regular food	Weight or body composition changes Changes in nutritional status Treatment tolerance Quality of life Symptoms	
Sanchez-Lara, 2014 ¹⁸⁸ (24746976) Other Key Question 2, 3 Not conducted	92 60 years 53% female Race NR	Other 62% Stage IV disease Chemotherapy alone No Other tool	Nutrition (oral nutritional supplement EPA) Setting NR Provider NR Oral	Standardized diets of 1400, 1600, 1800, 2000, or 2200kcal.	Weight or body composition changes Adverse events Survival Quality of life Symptoms Functional status	
Scislo, 2018 ¹⁸⁹ (29533110) Europe Key Question 2 High	98 63 years 28% female Race NR	Gastrointestinal 23% Stage IV disease Surgery alone No Other metric	Nutrition (postoperative immunomodulating enteral nutrition) Inpatient Provider NR Enteral	Standard enteral nutrition	Adverse events Survival	
Shimizu, 2018 ¹⁹⁰ (29721714) Asia Key Question 2 High	187 66 years 37% female Race NR	Gastrointestinal 0% Stage IV disease Surgery alone No Screening tool NR	Nutrition (early oral feeding with iEAT (POD 1-3)) Inpatient Provider NR Oral	Conventional nutritional management	Adverse events Readmissions or emergency room visits Length of stay	

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes	
Sim, 2022 ¹⁹¹ (35225460) Asia Key Question 2,3 High	58 64 years 20% female Race NR	Gastrointestinal 63% Stage IV disease Chemotherapy No Other tool	ONS twice a day (400ml, 400kcal) along with nutrition counseling and education Outpatient Multiple providers Oral	Nutrition counseling and education along with a weekly call from a trained dietitian	Changes in nutritional status Quality of Life Symptoms	
Vidal, 2016 ¹⁹² (27770454) Europe Key Question 2, 3 Not conducted	157 66 years 34% female Race NR	Other 8% Stage IV disease Surgery alone No Screening tool NR	Nutrition (TPN (Nutriflex special 70/240) for 5 days starting within 24 h after surgery) Outpatient Provider NR Parenteral	Oral nutrition alone started on day of surgery	Weight or body composition changes Survival Quality of life Functional status	
Wang J, 2022 (35126900) ¹⁹³ Asia Key Question 2, 3 High	80 43 years 43% female Race NR	Gastrointestinal 0% Stage IV disease Multiple therapies No Screening Tool NR	Enteral nutrition (Early enteral nutrition via nasointestinal tube 12-24 hours after surgery continuously for 7 days post surgery) Inpatient Provider NR Enteral	No enteral nutrition	Adverse events Length of stay Functional status	
Wu, 2017 ¹⁹⁴ (27208039) Asia Key Question 2, 3 Medium	73 56 years 32% female Race NR	Gastrointestinal 0% Stage IV disease Surgery alone No Screening tool NR	Nutrition (EN + PN for ≥7 days; EN started within 24 hr postsurgery) Inpatient Provider NR Parenteral	EN alone for ≥7 days started within 24 hr postsurgery	Weight or body composition changes Adverse events Length of stay Quality of life	
Xie, 2021 ¹⁹⁵ (34988183) Asia Key Question 2, 3 Medium	77 62 years 19% female Race NR	Gastrointestinal 0% Stage IV disease Surgery No Screening tool NR	Intervention type (300 mL oral nutritional supplement in addition to standard diet) Other Provider NR Oral	Regular diet alone	Weight or body composition changes Adverse events Readmissions/emergency room visits Survival Length of Stay Quality of life	

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Yang, 2020 ¹⁹⁶ (32833549) Asia Key Question 2, 3 High	120 64 years 23% female Race NR	Gastrointestinal 61% Stage IV disease Radiation alone No NRS-2002	Nutrition (nutritional counseling and dietary advice + oral nutritional supplement) Outpatient Multiple providers Oral	Nutritional counseling and dietary advice	Weight or body composition changes Changes in nutritional status Adverse events Symptoms
Yao, 2019 ¹⁹⁷ (31646824) Asia Key Question 2 Medium	114 65 years 40% female Race NR	Gastrointestinal 0% Stage IV disease Multiple therapies No Screening tool NR	Nutrition (early enteral nutrition support) Inpatient Multiple providers Enteral	Routine nutritional support	Adverse events Length of stay
Zhang Y, 2022 ¹⁹⁸ (35811605) Asia Key Question 2 Not conducted	82 69 years 37% female Race NR	Other (Lung) % Stage IV disease NR Surgery No MNA	Nutrition (Enteral nutrition support: a nasogastric tube was placed and whole protein enteral nutrition was infused with an initial amount of 750 kcal/d at 38°C via a nutrition pump at a controlled rate of 70– 150ml/h.) Inpatient Provider NR Enteral	Parenteral nutrition	Weight or body composition changes Changes in nutritional status Adverse Events Survival
Zhao, 2014 ¹⁹⁹ PMID NA Asia Key Question 2 Not conducted	64 33 years 22% female Race NR	Other % Stage IV Disease NR Chemotherapy alone No NRS-2002	Nutrition (enteral nutrition support) Inpatient Provider NR Enteral	Common hospital diets	Weight or body composition changes
Zhu, 2019 ²⁰⁰ (31464391) Asia Key Question 2, 3 Low	140 60 years 32% female Race NR	Gastrointestinal % Stage IV disease NR Surgery alone Yes NRS-2002	Nutrition (500 kcal/d total of ONS (over 3 feedings) from discharge to 90 days + nutrition education) Outpatient Physician Oral	Nutrition education	Weight or body composition changes Adverse events Quality of life Functional status

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes			
Zietarska, 2017 ²⁰¹ (29019951) Europe Key Question 2, 3 High	95 64 years 48% female Race NR	Gastrointestinal 32% Stage IV disease Chemotherapy alone Yes Multiple tools	Nutrition (high-energy, high-protein ONS, 250 ml/day x 3 mo.) Other Multiple providers Oral	Without nutritional support (ie, ONS)	Weight or body composition changes Changes in nutritional Status Adverse events Treatment tolerance Quality of life Symptoms Functional status			

Abbreviations: KQ = Key Question; PMID = PubMed Identification Number; NR = not reported; NA = not available; NRS = nutritional risk screening; POD = post operative day; PN = parenteral nutrition; EPA = eicosapentaenoic acid; TPN = total parenteral nutrition; EN = enteral nutrition; ONS = oral nutritional supplement; BCAA = branched-chain amino acid

*For select studies only.

†Reports median age when mean is not available.

Multi-Component Interventions

Study (PMID) Sample Size Cancer Type Intervention Type Comparison Outcomes Stage IV Disease (%) Region Age (Mean)† Intervention Delivery Description **Key Question** Sex (% Female) **Cancer Treatment** Setting Provider Delivering **RoB Rating*** Race/Ethnicity Type Limited to Intervention Malnourished (Yes or Route of Intervention No) Screening Tool Abdollahi, 2019²⁰² Multi-modality (face to 150 Other Received a pamphlet Adverse events (30945949) 46 years % Stage IV disease NR face nutritional education with dietary 100% female Other Chemotherapy alone Key Question 2 Race NR Yes recommendation) Not conducted Outpatient Screening tool NR Dietitian/nutritionist Oral 90 Multi-modality (calcium Weight or body Demark-Wahnefried. Other Calcium rich with 2008²⁰³ 42 vears 0% Stage IV disease rich diet with exercise) exercise and high fruit composition changes 100% female Chemotherapy alone and vegetable, low-fat Quality of life (18501061) Outpatient North America Non-Hispanic White No Dietitian/nutritionist diet Screening tool NR Oral Key Question 2, 3 Not conducted

Table E.6. Characteristics of included studies: use of multi-component interventions during cancer treatment (KQ2)

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes		
Lin, 2017 ²⁰⁴ (28658162) Asia Key Question 2 Not conducted	110 Age NR 36% female Asian	Gastrointestinal 44% Stage IV disease Chemotherapy alone No NRS-2002	Multi-modality (individual recipes plus enteral or parenteral nutrition) Multiple settings Nurse Other	Nutrition screening, nutritional guidance, diet mission, and arranged mealtimes	Weight or body composition changes		
Poulsen, 2014 ²⁰⁵ (24269077) Europe Key Question 2, 3 Not conducted	61 66 years 57% female Race NR	Multiple cancers % Stage IV disease NR Multiple therapies No NRS-2002	Multi-modality (nutritional counseling plus high-protein nutrition supplement) Outpatient Dietitian/nutritionist Oral	The control group was nutritionally instructed by the nurses with the possibility to call for a dietitian if needed	Weight or body composition changes Quality of life Symptoms		
Qin, 2021 ²⁰⁶ (34249995) Asia Key Question 2 Not conducted	60 54 years 100% female Race NR	Other (Ovarian) 12% Stage IV disease Chemotherapy No Other tool	Nutritional health education including designing healthy recipes and prevention for common nutritional problems along with ONS (250 ml (Ensure® three times a day) Setting NR Professional nutritionist Oral	Nutrition education	Changes in nutritional status		
Silander, 2012 ²⁰⁷ (21374756) Silander, 2013 ²⁰⁸ (23169469) Europe Key Question 2, 3 Not conducted	134 62 years 32% female Race NR	Head & Neck 74% Stage IV disease Chemotherapy alone Yes Screening tool NR	Multi-modality (nutritional counseling plus percutaneous endoscopic gastrostomy feeding Outpatient Dietitian/nutritionist Enteral	Nutritional support according to clinical praxis, which could include nutritional advice and enteral tube feeding	Weight or body composition changes Changes in nutritional status Adverse events Survival Length of stay Quality of life Symptoms Functional status		

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Skaarud, 2019 ²⁰⁹ (31455897) Europe Key Question 2 Not conducted	117 43 years 38% female Race NR	Other % Stage IV disease NR Multiple therapies No Screening tool NR	Multi-modality (patients received routine hospital food and were encouraged to take energy-enriched and lactose-reduced snacks and oral supplements daily. A nasoenteric tube was inserted within 5 days after transplantation) Multiple settings Provider NR Other	Routine hospital procedure for nutritional support	Weight or body composition changes Survival
Song, 2017 ²¹⁰ (28943912) Asia Key Question 2, 3 Not conducted	86 56 years 48% female Race NR	Head & Neck % Stage IV disease NR Surgery alone No Other tool	Multi-modality (preoperative nutritional support therapy, postoperative nutritional therapy during hospitalization, and nutrition therapy outside the hospital) Multiple settings Multiple providers Other	Routine nutritional support therapy after surgery	Changes in nutritional status Adverse events Readmissions or emergency room visits Survival Length of stay Quality of life
Uster, 2013 ²¹¹ (24103511) Europe Key Question 2, 3 Not conducted	58 65 years 21% female Race NR	Multiple Cancers % Stage IV disease NR Cancer Treatment Type NR Yes NRS-2002	Multi-modality (standardized individual nutritional therapy, including counseling by a dietitian, food fortification, and ONS if required) Outpatient Dietitian/nutritionist Oral	Standard care	Weight or body Composition changes Changes in nutritional Status Quality of life Functional status

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Xie, 2017 ²¹² (28524705) Asia Key Question 2 Not conducted	144 55 years 41% female Asian	Gastrointestinal 10% Stage IV disease Multiple therapies No Screening Tool NR	Multi-modality (intensive individualized nutritional and educational interventions during the entire course of chemotherapy) Multiple settings Multiple providers Other	Basic nutrition care and health education during hospitalization	Weight or body composition changes Changes in nutritional status

Abbreviations: KQ = Key Question; PMID = PubMed Identification Number; NR = not reported; ONS = oral nutritional supplement.

*For select studies only.

†Reports median age when mean is not available.

Risk of Bias and Outcome Assessment

Detection Performance Author, Year, Outcome Selection Fidelity Bias Reporting Attrition % Overall PMID Timing Bias Bias Bias Bias Rating de Luis, 200481 Х Х Inpatient Medium High Х Х High (15138461) de Luis. 2005⁸² 12 weeks Medium High Х Х Х Х High (15802904)De Luis 200783 Medium Low (0) Inpatient Medium Medium Low Low Medium (16929239)De Luis 200884 3 mo. Low Medium Medium Low Low Low (0) Medium (18700689)de Luis 200985 Inpatient Medium Low Low Medium Low Low (0) Medium (19694342) De Luis, 201086 Inpatient Medium Low Low Medium Low Low (0) Medium (21284343) de Luis. 201587 Inpatient Medium Low Low Low High Low (0) High (25855918) Farreras 200588 Low 9% Inpatient Medium Low Low Medium Low Medium (15681102) Golkhalkhali 2 mo, 6 mo Medium Low (attrition Medium Medium Medium Medium Low 201890 NR) (28857425)

Table E.7.1. Risk of bias assessment: use of dietary supplements during cancer treatment (KQ2)

Author, Year, PMID	Outcome Timing	Selection Bias	Detection Bias	Performance Bias	Fidelity Bias	Reporting Bias	Attrition %	Overall Rating
Jiang Z, 2010 ⁹² 20473991	8d postop	Low	Low	Medium	Low	Low	Low 1%	Low
Klek, 2005 ⁹³ (15906909)	Inpatient	Medium	High	Х	Х	X	X	High
Lobo 2006 ⁹⁴ (16777271)	11 wks; 4 mo mortality	Low	Low	Low	Low	Low	Medium 10%	Low
Lu, 2011 ⁹⁶ (213963072)	Inpatient	Medium	Medium	Medium	Medium	Low	Low	Medium
Matsuda, 2017 ⁹⁷ (28138734)	Inpatient, 28d	Medium	High	X	X	X	X	High
Miyata, 2017 ⁹⁸ (27644137)	Inpatient	Medium	Medium	Low	Medium	Low	Low	Medium
Pathak 2019 ⁹⁹ (31410338)	7 weeks	Low	Medium	Medium	Medium	Low	Low 7%	Medium
Pottel 2014 ¹⁰⁰ (25293388)	7 weeks	Low	Medium	Low	Low	Medium	Medium (17%)	Medium
Sun, L 2008 ¹⁰¹ (18376691)	Inpatient	Medium	Low	Medium	Medium	Low	Low (0)	Medium
Takeshita 2009 ¹⁰² (19285598)	2 weeks	Medium	Medium	Medium	Medium	Low	Low (0)	Medium
Wang, 2013 ¹⁰⁶ 24379010	Inpatient	Low	Low	Medium	Medium	Low	Low (0)	Low
Wang, 2017 ¹⁰⁵ (28927119)	Inpatient	Low	Low	Medium	Medium	Low	Low (0)	High
Wang, 2010 ¹⁰⁷ (21208095)	Inpatient	Low	Low	Low	High	Low	Low 4%	Medium
Wu, 2012 ¹⁰⁸ (23554777)	6 months	Low	Medium	High	Medium	Medium	Low (0)	High
Yang, 2022 ^{109,} ¹⁹⁶ (35500316)	Inpatient, 30 d	Low	Medium	Low	Medium	Low	Low (2%)	Medium
Yeh, 2013 ¹¹⁰ (23562359)	8 weeks	Low	Medium	High	Medium	Low	Low (0)	High
Zhu M, 2012 ¹¹² 22340541	Inpatient	Low	Low	Medium	Low	Low	Low 5%	Low

Note: X indicates domain not assessed due to previously determined high risk of bias. **Abbreviations:** PMID = PubMed Identification Number; mo = months; wks = weeks; POD = postoperative day; KQ=Key Question

Author, Year, RoB Timing	N	Treatment	Control	Weight / Body Comp.	Change s in Nutr. Status	Adverse Events	Readmission s/Emergency Room Visits	Surviv al	LOS	Treatme nt Toleran ce	QoL	Sym pto ms	Functional Status
de Luis 2007 ⁸³ (16929239) Medium Inpatient	72	Postop arginine enhanced EN (17 g/day)	Postop standard EN	\leftrightarrow	NA	↑ fistula ↔ other AEs	NA	NA	\leftrightarrow	-	-	-	-
De Luis 2008 ⁸⁴ (18700689) Medium 3 mos	65	Postop oral omega 3 supplement with high ratio omega3 /omega6	Postop oral omega 3 supplement with low ratio omega3/ omega6	\leftrightarrow	NA	\leftrightarrow	NA	NA	NA	_	-	-	-
de Luis 2009 ⁸⁵ (19694342) Medium Inpatient	72	Postop arginine enhanced EN (20 g/day)	Postop standard EN	\leftrightarrow	NA	↑ Fistula ↔ other AEs	NA	NA	ſ	-	-	-	-
De Luis, 2010 ⁸⁶ (21284343) Medium Postop day 10	115	Postop high arginine EN (20 g/day)	Postop medium arginine EN (12.3 g/day)	\leftrightarrow	NA	↑ Fistula ↔ other AEs	NA	NA	\leftrightarrow	-	-	-	-
Farreras 2005 ⁸⁸ (15681102) Medium	66	Early postop IEN with arg., omega 3 FAs & RNA	EN	NA	NA	↑ any AE ↑ wound- related	NA	\leftrightarrow	↑	-	-	-	-
Golkhalkhali 2018 ⁹⁰ (28857425) Medium 2mo, 6 mo	140	Probiotic (4 wks) + 2 g. omega 3 FAs (8 wks)	Placebo	↔ 2 mo ↔ 6 mo	NA	NA	NA	NA	NA	-	-	-	-
Jiang Z, 2010 ⁹² 20473991 Low Inpatient	200	PN with soybean oil + fish oil POD 1-8	PN with soybean oil POD 1-8	\leftrightarrow	NA	↑ SIRS ↔ other	NA	NA	↑	-	-	-	-
Lobo 2006 ⁹⁴ (16777271) Low 11wks, 4 mo	120	IEN (J-tube) x 10-15d postop	Other EN x 10-15d postop	NA	NA	\leftrightarrow infectious \leftrightarrow non- infectious \leftrightarrow J-tube	NA	\leftrightarrow	\leftrightarrow	-	-	-	-

Author, Year, RoB Timing	N	Treatment	Control	Weight / Body Comp.	Change s in Nutr. Status	Adverse Events	Readmission s/Emergency Room Visits	Surviv al	LOS	Treatme nt Toleran ce	QoL	Sym pto ms	Functional Status
Lu, 2011 ⁹⁶ (213963072) Medium Inpatient	50	Glutamine- enriched TPN x 7d postop.	Standard TPN x 7d psotop	NA	NA	↑ ↔ infectious AEs	NA	NA	NA	-	-	-	-
Miyata, 2017 ⁹⁸ (27644137) Medium Inpatient	61	Chemo ONS with FA's	Chemo Standard ONS	↔ Wt day 12 ↑ EPA levels day 8	NA	↑ Stomatitis ↔ other	NA	NA	NA	-	-	-	-
Pathak 2019 ⁹⁹ (31410338) Medium 7 weeks	60	10 gm oral glutamine 2 hr before chemo- radiation, 5d/wk x 7 wks	No glutamine x 7 wks	↑	NA	↑ oral mucositis ↑ dysphagi a	Î	NA	NA	-	-	-	-
Pottel 2014 ¹⁰⁰ (25293388) Medium 4 weeks	91	7.5 mL Echium oil 2x/day during radio(chemo)ther apy	7.5 mL Sunflower oil 2x/day during radio(chemo) therapy	↑ FFM ↔ other	\leftrightarrow	\leftrightarrow	NA	NA	NA	-	-	-	-
Sun, L 2008 ¹⁰¹ (18376691) Medium Inpatient	64	BCAA-enriched (30%) TPN postop	Standard TPN (24% BCAA)	↔ POD 7	NA	↑ any AE ↔ individual AEs	NA	\leftrightarrow	NA	-	-	-	-
Takeshita 2009 ¹⁰² (19285598) Medium 2 weeks	56	BCAA snack x 15 d (1d preop+14d) post chemo- embolization	"control group"	\leftrightarrow	NA	NA	NA	\leftrightarrow	↔	-	-	-	-
Wang, 2010 ¹⁰⁷ (21208095) Medium Inpatient	229	Triglyceride & protein-enriched EN	NA	NA	NA	↔ infections	NA	NA	↑	-	-	-	-
Wang, 2013 ¹⁰⁶ 24379010 Low Inpatient	94	EN + PN with olive oil-based lipids ≥ 7 d postop	EN + PN with triglycerides ≥ 7 d postop	NA	NA	↔ infectious ↔ any	NA	\leftrightarrow	↔ ↔ ICU	-	-	-	-
Yang, 2022 ¹⁰⁹ (35500316) Medium Inpatient, 30 d	120	EN & PN with n-3 fatty acids POD 1-5	EN & PN	NA	NA	↑postoper ative complicat ions	eadmissions	\leftrightarrow	\leftrightarrow	-	-	-	-

Author, Year, RoB Timing	Ν	Treatment	Control	Weight / Body Comp.	Change s in Nutr. Status	Adverse Events	Readmission s/Emergency Room Visits	Surviv al	LOS	Treatme nt Toleran ce	QoL :	Sym pto ms	Functional Status
Zhu M, 2012 ¹¹² 22340541 Low Inpatient	57	0.2 g/kg/d fish oil + 1 g/kg/d soybean oil in TPN x 7d postop	TPN x 7d postop	NA	NA	↔ infectious † SIRS ↔ serious AEs (none)	NA	NA	Ť	-		-	-

Abbreviations: PMID = PubMed Identification Number; ROB=risk of bias; BMI: body mass index; FFM: fat free mass; AE: adverse events; LOS: length of stay; QOL: quality of life; PN: parenteral nutrition; IEN=immunoenteral nutrition; UTD: unable to determine; abd. sx = abdominal symptoms; EN: enteral nutrition: TPN: total parenteral nutrition: ICU: intensive care unit: SIRS: systemic inflammatory response syndrome; d: day: BCAA=Branched chain amino acid; KQ= Key Question; NA=not assessed.

↑: Intervention group had a statistically significantly better outcome than comparison group (e.g. fewer AEs, shorter LOS than comparison group).

1: Intervention group had a statistically significantly worse outcome than comparison group (e.g. more AEs, longer LOS).

 \leftrightarrow : No statistically significant difference between groups.

Table E.8.1, Risk of bias assessment: route or timing of nutritional interventions during cancer treatment (KQ2)

Author, Year, PMID	Outcome Timing	Selection Bias	Detection Bias	Performance Bias	Fidelity Bias	Reporting Bias	Attrition %	Overall Rating
Berkelmans 2020 ¹²² (31090563)	Inpatient, 30 d, 3 mo.	Low	Medium	Medium	Medium	Low	Medium 11%	Medium
Boelens, 2014 ¹²³ (24169163)	Inpatient	Low	High	X	Х	Х	X	High
Bozzetti, 2001 ¹²⁴ (11705560)	Inpatient	Low	Low	Medium	Low	Low	Low 2%	Low
Braga, 2001 ¹²⁵ (11246300)	Inpatient	Low	Medium	Medium	Low	Medium	Low 3%	Medium
Dag, 2011 ¹²⁶ (22189721)	Inpatient	Low	High	X	Х	Х	X	High
Feo 2004 ¹²⁷ (15144242)	Inpatient	Low	Medium	Medium	Low	Low	Low (0)	Medium
Gao, 2019 ¹²⁸ (30941970)	Inpatient	Medium	High	X	Х	Х	X	High
Huang, 2015 ¹²⁹ (1594390)	Inpatient	Medium	High	X	Х	Х	X	High
Hyltander, 2005 ¹³⁰ (15880316)	1 year	Low	High	X	X	Х	X	High
Kita, 2021 ¹³¹ (33526287)	Inpatient	High	X	X	X	Х	X	High
Kurbanalievich, 2020 ¹³² (2005564409)	Inpatient	Medium	Medium	Medium	Medium	Low	Low (0)	Medium

Author, Year, PMID	Outcome Timing	Selection Bias	Detection Bias	Performance Bias	Fidelity Bias	Reporting Bias	Attrition %	Overall Rating
Li, 2015 ¹³³ (26011337)	Inpatient	Medium	Medium	High	Х	Х	Х	High
Ma, 2020 ¹³⁶ (33362392)	Inpatient	Low	Medium	Medium	Medium	Medium	Low 3%	Medium
Luo, 2018 ¹³⁵ (28666095)	Inpatient, 20 months	Low	Medium	High	X	X	Х	High
Mahmoodzadeh, 2015 ¹³⁷ (24875466)	Inpatient	Medium	High	X	X	X	X	High
Perinel, 2016 ¹³⁹ (27429039)	Inpatient, 30d	Low	Medium	Medium	High	X	Х	High
Sun, 2018 ¹⁴⁵ (28549015)	Inpatient, readmissions, HRQoL 24 wk	Low	Medium	Medium	Medium	Low	Low (0)	Medium
Tao 2020 ¹⁴⁶ (31512101)	Inpatient, 1 mo., 3 mo.	Medium	Medium	Medium	Low	Medium	Low (0)	Medium
van Barneveld, 2016 ¹⁴⁷ (26937858)	Inpatient	Low	High	X	X	X	X	High
Wang 2019 ¹⁴⁸ (931239762)	30 days	Low	Medium	Medium	Medium	Low	Low 1%	Medium
Wang, 2018 ¹⁴⁹ (29896256)	Inpatient	Medium	Medium	Medium	Medium	Medium	Low (0)	Medium
Xiao-Bo 2014 ¹⁵⁰ (924504222)	Inpatient	Medium	Medium	Medium	Medium	Low	Low (0)	Medium
Xu, 2020 ¹⁵¹ PMID NA	Inpatient	Medium	Medium	High	Х	X	Х	High
Zhang, 2019 ¹⁵² (31551229)	Inpatient	Low	High	X	X	X	X	High

Note: X indicates domain not assessed due to previously determined high risk of bias. Abbreviations: PMID = PubMed Identification Number; HRQoL = health related quality of life; NA = not available; mo = months; wk = weeks; KQ= Key Question.

Table E.8.2.	Outcomes assessment:	route or timing of nutritiona	al interventions during cance	er treatment (low and me	dium ROB studies),
KQ2		-	-		

Author, Year, PMID RoB Timing	N	Treatment	Control	Weight/ Body Comp.	Changes in Nutr. Status	Adverse Events	Readmissions/ Emergency Room Visits	Survival	LOS	Treat. Tolerance	QoL	Symptoms	Function al Status
Berkelmans 2020 ¹²² (31090563) Medium Inpatient, 30d; 90d (mortality)	148	Postop early oral feeding (POD1)	tube feeding x 5d (while NPO)	↔ 1 mo ↔ 3 mo	↓ kcal	↔ pneumonia ↑ chyle leak ↔ other AEs	↔ readmit	↔ 90d	\leftrightarrow	-	-	-	-

Author, Year, PMID RoB Timing	N	Treatment	Control	Weight/ Body Comp.	Changes in Nutr. Status	Adverse Events	Readmissions/ Emergency Room Visits	Survival	LOS	Treat. Tolerance	QoL	Symptoms	Function al Status
Bozzetti, 2001 ¹²⁴ (11705560) Low Inpatient	317	Postop EN in malnourish ed	Postop PN in malnourish ed	NA	NA	↑ any on list ↑ infectious	NA	UTD	1	-	-	-	-
Braga, 2001 ¹²⁵ (11246300) Medium Inpatient	257	Postop TPN	Postop early EN	NA	NA	 ↔ any ↔ infectious ↔ noninfect. 	NA	\leftrightarrow	\leftrightarrow	-	-	-	-
Feo 2004 ¹²⁷ (15144242) Medium Inpatient	100	No NG tube, clear liq POD 1, advance to soft diet	NG tube to flatus, liquid diet to soft diet	NA	NA	↔ complications	NA	↔ (0 death)	\leftrightarrow	-	-	-	-
Kurbanalievic h, 2020 ¹³² PMID NA Medium Inpatient	80	EN from POD 2	Isolated parenteral nutrition from POD 1	\leftrightarrow	NA	NA	NA	NA	NA	-	-	-	-
Ma, 2020 ¹³⁶ (33362392) Medium Inpatient	60	Postop early EN + PN	Postop TPN	NA	NA	↔ SSI ↔ bile leak	NA	NA	1	-	-	-	-
Sun, 2018 ¹⁴⁵ (28549015) Medium Inpatient, 6m	280	Early oral feeding with liquid foods on post-op day 1	Nasoentera I nutrition post-op	\leftrightarrow	NA	↔ complications	NA	\leftrightarrow	Ť	-	-	-	-
Tao 2020 ¹⁴⁶ (31512101) Medium Inpatient, 3mo	120	Postop EN via J-tube	Postop EN via NG tube	↑	NA	 ↔ overall complications ↓ intestinal obstruction 	NA	↔ DFS, OS	NA	-	-	-	-
Wang 2019 ¹⁴⁸ (31239762) Medium Inpatient	100	Early oral feeding postop	Delayed oral feeding postop	NA	NA	↔ complications		NA	Î↑	-	-	-	-

Author, Year, PMID RoB Timing	N	Treatment	Control	Weight/ Body Comp.	Changes in Nutr. Status	Adverse Events	Readmissions/ Emergency Room Visits	Survival	LOS	Treat. Tolerance	QoL	Symptoms	Function al Status
Wang, 2018 ¹⁴⁹ (29896256) Medium Inpatient	129	Postop EN	Postop TPN	↑ POD4 ↔ POD8	NA	↑complications	NA	NA	<u>↑</u>	-	-	-	_
Xiao-Bo 2014 ¹⁵⁰ (24504222)	120	Enteral nutrition, 1000 mL, 4 18 kJ/ml	Parenteral nutrition mixture	NA	NA	↑ bacterial, fungal infection	NA	NA	\leftrightarrow	-	-	-	-

Abbreviations: PMID = PubMed Identification Number; ROB=risk of bias; NA = not available; BMI: body mass index; FFM: fat free mass; AE: adverse events; LOS: length of stay; QOL: quality of life; PRO: patient reported outcome; POD: post operative day: DFS: disease free survival: OS: overall survival: EN: enteral nutrition: TPN: total parenteral nutrition: NG: nasogastric; sx = abdominal symptoms; SSI: surgical site infection; UTD: unable to determine; EORTC QLQ-C30=European Organization for Research and Treatment of Cancer, Quality of Life Questionnaire; SF-36 = short-form health survey; -: Not applicable.

Author, Year, PMID	Outcome Timing	Selection Bias	Detection Bias	Performance Bias	Fidelity Bias	Reporting Bias	Attrition %	Overall Rating
Barlow, 2011 ¹⁵⁵ (21601319)	Inpatient, 6 & 12 wks.	Low	Medium	Low	Low	Low	Low (0)	Medium
Chen T, 2021 ¹⁵⁹ (34237976)	1 month post- hospital discharge after surgery	Low	High	X	X	X	Х	High
Chu, 2018 ¹⁶⁰ (622367981)	Inpatient	Low	Medium	Medium	Medium	Low	Low (0)	Medium
Gavazzi, 2016 ¹⁶³ (27391922)	6 months	Low	High	Х	Х	Х	X	High
Imamura, 2016 ¹⁶⁵ (27084538) Kimura, 2019 ¹⁶⁶ (31788653)	8 weeks	Low	High	X	x	X	X	High
Jin, 2018 ¹⁶⁸ (30205371)	Inpatient	Medium	Medium	High	X	Х	X	High
Katada, 2021 ¹⁷⁰ (33009977)	2 months	Low	High	Х	Х	X	X	High
Klek, 2011 ¹⁷⁴ (21820770) Klek, 2014 ¹⁷⁵ (24178182)	Inpatient	Low	Medium	Medium	Low	Low	Low 4%	Medium

Table E.9.1. Risk of bias assessment: use of nutrition support includ	ng oral nutrition supplements during cancer treatment (KQ2
---	--

Author, Year,	Outcome Timing	Selection Bias	Detection Bias	Performance Bias	Fidelity Bias	Reporting Bias	Attrition %	Overall Rating
Klek, 2008 ¹⁷¹	Inpatient	Low	Low	Low	Medium	Low	Low 7%	Low
(18571296)								
Klek, 2011 ¹⁷²								
(21074910)								
Klek, 2017 ¹⁷³								
(28336107)								
Li C, 2020 ¹⁷⁷ (2003914110)	3 yrs (survival)	Medium	Medium	Medium	Medium	Low	Low	Medium
Li, 2015 ¹⁷⁶ (26125924)	Inpatient	Low	High	Х	Х	X	Х	High
Lyu, 2022 ¹⁷⁸ (35280748)	After treatment completion (2- 3 cycles) & up to 4 yrs	Low	Medium	High	Medium	Low	Medium (19%)	High
Meng, 2021 ¹⁸⁰ (32563598) Tan 2021 ¹⁸¹ (32563599)	3 mo	Low	Low	Medium	Medium	Low	Low 5%	Low
Miyazaki 2021 ¹⁸² (33835329)	1, 3, 6, 12 mo.	Low	Low	Medium	Medium	Low	Low 12%	Medium
Nie, 2021 ¹⁸³ (35035754)	NR	Medium	Medium	High	Medium	Low	Low (0%)	High
Okabayashi, 2011 ¹⁸⁵ (20852905)	1 year	Medium	Medium	High	X	X	X	High
Ohkura, 2019 ¹⁸⁴ (30169605)	Inpatient, 6 m	Low	High	High	Х	Х	Х	High
Scislo, 2018 ¹⁸⁹ (29533110)	Inpatient, 1 yr	Low	High	Х	Х	Х	Х	High
Shimizu, 2018 ¹⁹⁰ (29721714)	Inpatient	Low	Medium	High	Х	Х	Х	High
Sim, 2022 ¹⁹¹ (35225460)	Weeks 0, 4, 8	Medium	Medium	High	Medium	Low	High (31%)	High
Wang J, 2022 ¹⁹³ (35126900)	Inpatient	Low	High	Х	Х	Х	Х	High
Wu, 2017 ¹⁹⁴ (27208039)	Inpatient; 90 d QOL	Low	Medium	Medium	Medium	Low	Medium 13%	Medium
Xie, 2021 ¹⁹⁵ (334988183)	Inpatient, 1 mo., 3 mo., 6 mo.	Low	Medium	Medium	Low	Low	Medium (14%)	Medium
Yang, 2020 ¹⁹⁶ (32833549)	3 days	Low	Medium	High	X	X	X	High
Yao 2019 ¹⁹⁷ (31646824)	Inpatient	Low	Medium	Medium	Medium	Low	Low (0)	Medium

Author, Year, PMID	Outcome Timing	Selection Bias	Detection Bias	Performance Bias	Fidelity Bias	Reporting Bias	Attrition %	Overall Rating
Zhu 2019 ²⁰⁰ (31464391)	30, 60, 90 days	Low	Low	Medium	Midium	Low	Medium 19%	Low
Žietarska, 2017 ²⁰¹ (29019951)	12 weeks	Low	Low	Medium	Low	Medium	High (24%)	High

Note: X indicated domain not assessed due to previously determined high risk of bias. **Abbreviations:** PMID = PubMed Identification Number; d = days; wks = weeks; mo = months; yr = years; KQ= Key Question.

Table E.9.2. Outcomes	s assessment: u	se of nutrition supp	ort including ora	I nutrition supp	plements during cancer	treatment (low and
medium ROB studies)	, KQ2					

Author, Year, RoB	N	Treatment	Control	Weight/ Body	Changes in Nutr.	Adverse Events	Readmissions/ Emergency	Survival	LOS	Treatment Tolerance	QOL	Symptoms	Functional Status
Barlow, 2011 ¹⁵⁵ (21601319) Medium Inpatient, 12w	121	Early EN within 12h of surgery (20 kcal/h)	IV hydration	NA	NA	↑ infective complicat ions	↔	UTD	1	-	-	-	-
Chu, 2018 ¹⁶⁰ (62236798 1) Medium Inpatient	98	Postop EN+PN	Postop routine PN	NA	Î	↑ complicat ions	NA	NA	↑	-	-	-	-
Klek, 2011 ¹⁷⁴ (21820770) Klek, 2014 ¹⁷⁵ (24178182) Medium Inpatient	167	1. IMEN; 2. SPN 3. IMPN 24h after surgery for at least 7d	Standard EN	NA	NA	↔ Infectious complicat ions	NA	↔ 30d	\leftrightarrow	-	-	-	-
Klek, 2008 ¹⁷¹ (18571296) Klek, 2011 ¹⁷² (21074910) Klek, 2017 ¹⁷³ (28336107) Low Inpatient	196	IMEN	Standard EN	NA	NA	↔ complicat ions	NA	\leftrightarrow	NA	-		-	-

Author, Year, RoB Timing	Ν	Treatment	Control	Weight/ Body Comp.	Changes in Nutr. Status	Adverse Events	Readmissions/ Emergency Room Visits	Survival	LOS	Treatment Tolerance	QOL	Symptoms	Functional Status
Li C, 2020 ¹⁷⁷ (PMID NA) Medium 3yrs	78	Post- chemo early IEN	Conventi onal nutrition support	NA	NA	1	NA	↑ 3 yr	NA	-	-	-	-
Meng, 2021 ¹⁸⁰ (32563598) Tan 2021 ¹⁸¹ (32563599) Low	353	Oral nutritional supplement s with dietary advice	Dietary advice alone	↑	NA	NA	NA	NA	NA	-	-	-	-
Miyazaki 2021 ¹⁸² (33835329) Medium 1 yr	100 3	Postop 400 ml ONS x 3 mo. (+ regular diet)	No ONS postop (+ regular diet)	↑ 3 mo. ↑ 6 mo. ↔ 1 yr.	NA	↔ any surgical AE	NA	NA	NA	-	-	-	-
Wu, 2017 ¹⁹⁴ (27208039) Medium Inpatient	80	Post-op EN+PN via NJT/ j-tube	Post-op EN via NJT/ j- tube	↑ FFM ↑ BW ↔ FM	NA	↔ postoper ative complicat ions	NA	NA	\leftrightarrow	-	-	-	-
Xie, 2021 ¹⁹⁵ (33498818 3) Medium Inpatient, 1 mo., 3 mo., 6 mo.	77	300 mL oral nutritional supplement in addition to standard diet	Regular diet alone	↑body weight loss	NA	 ↔ Anastom otic leak, pneumon ia, pleural effusion, atelectasi s, 41espirat or failure, myocardi al arrhythmi a, heart failure, any complicat ion 	↔ readmissions	↔	↔ Hos pital, ICU stay	-	-		-

Author, Year, RoB Timing	Ν	Treatment	Control	Weight/ Body Comp.	Changes in Nutr. Status	Adverse Events	Readmissions/ Emergency Room Visits	Survival	LOS	Treatment Tolerance	QOL	Symptoms	Functional Status
Yao, 2019 ¹⁹⁷ (31646824) Medium Inpatient	114	Early EN postop via NG tube	Routine oral nutrition support	NA	NA	↑ total count	NA	NA	Ţ	-	-	-	-
Zhu, 2019 ²⁰⁰ (31464391) Low 30-90 days	140	500 kcal ONS 3x/d x 90d + dietary advice from doctor at discharge	Dietary advice from doctor at hospital discharg e	 ↔ 30d weight/ BMI ↑ 60d weight/ BMI ↑ 90d weight/ BMI ↔ FFM 	NA	↔ (none)	NA	NA	NA	-	-	-	-

Abbreviations: PMID = PubMed Identification Number; ROB=risk of bias; NA = not available BMI: body mass index; FFM: fat free mass; AE: adverse events; LOS: length of stay; QOL: quality of life; PRO: patient reported outcome; d; day: EN: enteral nutrition: NG: nasogastric: ONS: oral nutritional supplement: PN: parenteral nutrition: IMEN: immunomodulating enteral nutrition: SPN: standard parenteral nutrition: IMPN: immunomodulating parenteral nutrition: IEN: immunomodulating enteral nutrition: EQ-5D: EuroQol-5D: UTD: unable to determine; KQ = Key Question; FM = fat mass; NJT = nasojejunal tube; NA=not assessed. -: Not applicable.

↑: Intervention group had a statistically significantly better outcome than comparison group (e.g. fewer Aes, shorter LOS than comparison group).

: Intervention group had a statistically significantly worse outcome than comparison group (e.g. more Aes, longer LOS).

 \leftrightarrow : No statistically significant difference between groups.

Appendix F. Evidence Tables for Chapter 8

Nutritional Counseling

Table F.1. Characteristics of included studies: use of nutritional counseling for cancer and treatment related symptoms (KQ3)

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Britton, 2019 ⁶⁴ (30296472) Other Key Question 2, 3 Not conducted	307 58 years 21% female Other	Head & Neck 65% Stage IV disease Radiation alone No Screening tool NR	Counseling only (Eating As Treatment (EAT) dietary advice) Outpatient Dietitian/nutritionist Not applicable	Treatment as usual	Weight or body composition changes Changes in nutritional status Treatment tolerance Quality of life
Forslund, 2020 ⁶⁵ (31758324) Europe Key Question 2, 3 Not conducted	180 67 years 0% female Race NR	Other 4% Stage IV disease Radiation alone No Screening tool NR	Counseling only (advised to increase soluble fiber and decrease lactose in diet) Outpatient Dietitian/nutritionist Not applicable	Standard care	Adverse events Quality of life
Isenring, 2004 ⁶⁶ (15226773) Other Key Question 2, 3 Not conducted	60 62 years 15% female Race NR	Multiple cancers % Stage IV disease NR Multiple therapies No Other tool	Counseling only (intensive nutritional counseling for up to 3 months) Outpatient Dietitian/nutritionist Not applicable	Less intensive diet education by nurses	Weight or body composition changes Changes in nutritional Status Quality of life Functional status
Movahed, 2020 ⁶⁸ PMID NA Other Key Question 2, 3 Not conducted	100 67 years 49% female Race NR	Gastrointestinal % Stage IV disease NR Multiple therapies No Other tool	Counseling only (individualized dietary plan and nutrition education) Setting NR Other provider Not applicable	General dietary advice at the beginning of the treatment	Weight or body composition changes Changes in nutritional status Symptoms Functional status

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Najafi, 2019 ²¹³ (30449171) Asia Key Question 3 Not conducted	150 58 years 100% female Race NR	Other % Stage IV disease NR Chemotherapy alone No Screening tool NR	Counseling only (Personalized diet and nutrition education before each chemotherapy session) Setting NR Dietitian/nutritionist Not applicable	Usual diet + regular chemotherapy drug regimen with no pamphlet, nutritional education, or dietary intervention	Quality of life Symptoms Functional status
Pettersson, 2012 ⁷⁰ (22633817) Region NR Key Question 2, 3 Not conducted	130 66 years 0% female Race NR	Other % Stage IV disease NR Radiation alone No Other tool	Counseling only (advised to increase soluble fiber and decrease lactose in diet) Setting NR Dietitian/nutritionist Not applicable	Standard care (advised to continue normal diet)	Adverse events Symptoms Functional status
Qiu, 2020 ⁷¹ (31526964) Asia Key Question 2, 3 Not conducted	85 67 years 36% female Race NR	Gastrointestinal % Stage IV disease NR Multiple therapies No Multiple tools	Counseling only (whole- course nutrition management) Inpatient Multiple providers Not applicable	The control group was treated with general nutritional supplementation	Weight or body composition changes Adverse events Length of stay Quality of life Symptoms
Ravasco, 2012 ⁷² (23134880) Europe Key Question 2, 3 Not conducted	89 64 years 46% female Race NR	Gastrointestinal % Stage IV disease NR Radiation alone No Other tool	Counseling only (individualized nutritional counseling and education about regular foods weekly for 6 weeks) Outpatient Provider NR Not applicable	Group 2: dietary supplements and consumed usual diet of regular foods Group 3: consumed usual diet of regular foods	Weight or body composition changes Changes in nutritional status Adverse events Survival Quality of life Symptoms
Regueme, 2021 ⁷³ (32435967) Europe Key Question 2, 3 Not conducted	283 77 years 50% female Race NR	Multiple cancers % Stage IV disease NR Chemotherapy alone Yes Other tool	Counseling only (usual care + support to increase intake) Outpatient Dietitian/nutritionist Not applicable	Usual care: received nutritional care routinely used in the cancer treatment setting	Weight or body composition changes Quality of life Symptoms Functional status

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Um, 2014 ⁷⁵ (24906838) Asia Key Questions 2, 3 Not conducted	87 60 years 36% female Race NR	Multiple cancers % Stage IV disease NR Radiation alone Yes Other tool	Counseling only (at least 3 sessions of individualized dietary counseling concurrent with radiation) Outpatient Dietitian/nutritionist Not applicable	One routine 20-min education session by a dietitian within 4 days of starting RT	Weight or body composition changes Changes in nutritional status Survival Quality of life
van der Werf, 2020 ⁷⁶ (32037284) Europe Key Questions 2, 3 Not conducted	107 65 years 37% Race NR	Gastrointestinal 100% Stage IV disease Chemotherapy alone No Screening tool NR	Counseling only (individualized nutritional counseling) Outpatient Dietitian/nutritionist Not applicable	Usual care	Weight or body composition changes Survival Treatment tolerance Quality of life Functional status
Zhang Z, 2022 ⁷⁷ (34984549) Asia Key Question 2, 3 Not conducted	468 60 years 42% female Race NR	Multiple Cancers 37% Stage IV disease Radiation Therapy No Multiple tools	Nutrition education (guideline-based nutrition education counseling that was tailored and dynamic) Inpatient Nurse Not applicable	No nutrition education	Changes in nutritional status Treatment tolerance

Abbreviations: KQ = Key Question; PMID = PubMed Identification Number; NR = not reported; RT = radiotherapy. *For select studies only. †Reports median age when mean is not available.
Dietary Supplements

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Cereda, 2019 ²² (31568698) Europe Key Question 1 & 2, 3 Not conducted	166 65 years 40% female Race NR	Multiple cancers 81% Stage IV disease Chemotherapy alone Yes Other metric	Single Supplement (WPI supplementation) Outpatient Dietitian/nutritionist Oral	Nutritional counseling alone	Weight of body composition changes Treatment tolerance Quality of life
Dechaphunkul, 2022 ¹ 35007812 Asia Key Question 1, 3 Not conducted	110 52 years 21% female Race NR	Head & Neck 76% Stage IV disease Chemotherapy No Other (Nutritional Risk Index)	Immunonutrition with FA's, arginine, fiber & nucleotides, 3x/day for 5 days before each chemotherapy session. Outpatient Dietitian/nutritionist Oral	Standard enteral formula, 3x/day for 5 days before each chemotherapy	Weight or body composition changes Symptoms Survival
Fietkau, 2013 ⁸⁹ (23765693) Europe Key Question 2, 3 Not conducted	111 56 years 16% female Race NR	Multiple cancers % Stage IV disease NR Multiple therapies No Other tool	Single Supplement (Disease-specific enteral formula (Supportan)) Outpatient Physician Enteral	500 mL of the enteral standard nutrition Fresubin energy fibre	Weight or body composition changes Changes in nutritional Status Quality of life Functional status
Golkhalkhali, 2018 ⁹⁰ (28857425) Asia Key Question 2, 3 Medium	140 Age NR % female NR Asian	Gastrointestinal % Stage IV disease NR Chemotherapy alone No Screening tool NR	Single Supplement (Microbial cell prep and omega-3 fatty acids) Inpatient Provider NR Oral	Placebo	Weight or body composition changes Quality of life Symptoms
Iwase, 2016 ⁹¹ (26105516) Asia Key Question 2, 3 Not conducted	59 Age NR 100% female Race NR	Other 14% Stage IV disease Chemotherapy alone No Screening tool NR	Supplements (Amino acid jelly (Inner Power) oral diet supplement) Setting NR Provider NR Oral	Regular care	Adverse events Quality of life Symptoms

Table F.2. Characteristics of included studies: use of dietary supplements for cancer and treatment related symptoms (KQ3)

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Serrano, 2022 ³⁰ (35606184) Canada Key Question 1 & 2, 3 Medium	71 65 years 41% female Race NR	Gastrointestinal 69% Stage IV disease Surgery No MUST	Supplements (protein supplementation rich in arginine and omega-6 and carbohydrate loading) Other Nurse Oral	One placebo to each supplement	Adverse events Readmissions or Emergency room visits Survival Length of Stay Quality of life
Tanca, 2009 ¹⁰³ PMID NA Europe Key Question 2, 3 Not conducted	475 62 years 40% female Race NR	Multiple cancers 95% Stage IV disease Cancer treatment type NR Yes Multiple tools	Supplements (5 treatment groups: 1) Progestational agent, 2) EPA, 3) L-carnitine, 4) Thalidomide, 5) Progestational agent plus pharmacologic nutritional support and L-carnitine and thalidomide) Other Provider NR Oral	No placebo or control	Weight or body composition changes Quality of life Symptoms Functional status
Tsuchiya, 2016 ³⁴ (27306219) Asia Key Question 1 & 2, 3 High	70 63 years 40% female Race NR	Gastrointestinal % Stage IV disease NR Multiple therapies No Screening Tool NR	Supplements (added amino acids) Outpatient Provider NR Oral	No supplement	Adverse events Treatment tolerance
Wang, 2017 ¹⁰⁵ (28927119) Asia Key Question 2, 3 High	94 59 years 37% female Race NR	Gastrointestinal 100% Stage IV disease Multiple therapies No Screening tool NR	Single Supplements (glutamine enriched nutritional support) Inpatient Provider NR Parenteral	Basic nutritional support	Adverse events Quality of life

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Wang, 2010 ¹⁰⁷ (21208095) Asia Key Question 2, 3 Medium	229 56 years 36% female Race NR	Gastrointestinal % Stage IV disease NR Surgery alone No Screening tool NR	Supplements (EN enriched with medium- chain triglycerides and protein) Inpatient Multiple providers Enteral	Isocaloric EN for ≥5 days started within 48- 72 h post-surgery	Adverse events Length of stay Symptoms

Abbreviations: KQ = Key Question; PMID = PubMed Identification Number; NR = not reported; NA = not available; EN = enteral nutrition; WPI = whey protein isolate; *For select studies only.

†Reports median age when mean is not available.

Special Diets

Table F.3. Characteristics of included studies: use of special diets for cancer and treatment related symptoms (KQ3)

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Ansari, 2016 ²¹⁴ (27644633) Asia Key Question 3 Not conducted	100+ 49 years 100% female Race NR	Other % Stage IV disease NR Chemotherapy alone No Screening tool NR	Single Supplement (Ginger capsules) Outpatient Other Oral	Placebo, 2 capsules every 12 hours for 3 days	Symptoms
Gardner, 2008 ¹¹³ (18955453) North America Key Question 2, 3 Not conducted	153 64 years % female NR Race NR	Other % Stage IV disease NR Chemotherapy alone No Screening tool NR	Special Diets (diet with raw fruits and vegetables) Inpatient Provider NR Oral	Diet without raw fruits & vegetables	Weight or body composition changes Adverse events Survival Symptoms

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Ingersoll, 2010 ¹¹⁴ (20189927) North America Key Question 2, 3 Not conducted	77 54 years 81% female Race NR	Multiple cancers % Stage IV disease NR Chemotherapy alone No Screening tool NR	Special Diets (standard medical CINV management + grape juice) Outpatient Provider NR Oral	Standard medical CINV management	Adverse events Quality of life Symptoms
Jatoi, 2016 ¹¹⁵ (27039205) North America Key Question 2, 3 Not conducted	118 Age NR 52% female Race NR	Other 100% Stage IV disease Multiple therapies No Screening tool NR	Special Diets (white wine twice daily) Outpatient Provider NR Oral	Nutritional supplement like boost or ensure	Adverse events Quality of life Functional status
Khodabakhshi, 2020 ¹¹⁶ (314996287) Khodabakhshi, 2020 ¹¹⁷ (32828130) Other Key Question 2, 3 Not conducted	80 Age NR 100% female Race NR	Other 26% Stage IV disease Chemotherapy alone No Screening tool NR	Special Diets (ketogenic) Outpatient Dietitian/nutritionist Oral	Standard diet	Weight or body composition changes Quality of life
Lugtenberg, 2021 ¹¹⁸ (33179154) Europe Key Question 2, 3 Not conducted	129 50 years 100% female Race NR	Other % Stage IV disease NR Chemotherapy alone No Screening tool NR	Special Diets (calorie content declined for 3 days before and the day of treatment) Setting NR Physician Other	Regular diet	Weight or body composition changes Quality of life Symptoms Functional status
Miyakawa, 2019 ¹¹⁹ (30554216) Asia Key Question 2, 3 Not conducted	100 71 years 66% female Race NR	Gastrointestinal % Stage IV disease NR Surgery alone No Screening tool NR	Special Diets (rice porridge) Inpatient Provider NR Oral	Liquid diet with gradual transition to solid food	Length of stay Quality of life Symptoms
Vafa, 2020 ²¹⁵ (32898787) Asia Key Question 3, 4 Not conducted	135 53 years 100% female Race NR	Other 0% Stage IV disease Multiple therapies No Screening tool NR	Special Diets (Calorie restricted diet + CFU symbiotic supplement) Outpatient Provider NR Oral	Calorie-restricted diet + placebo	Weight of body composition changes Quality of life

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Voss, 2020 ¹²⁰ (32619561) Europe Key Question 2, 3 Not conducted	50 57 years % female NR Race NR	Other % Stage IV disease NR Radiation alone No Screening tool NR	Special Diets (3 days of ketogenic diet, 3 days of intermittent fasting, then 3 days of ketogenic diet) Outpatient Dietitian/nutritionist Oral	Calorically unrestricted diet	Survival Symptoms
Wedlake, 2012 ¹²¹ (1605807) Europe Key Question 2, 3 Not conducted	100+ 65 years 32% female Race NR	Multiple cancers % Stage IV disease NR Radiation alone No Screening tool NR	Special Diets (Low fat and moderate fat intake) Outpatient Provider NR Oral	Normal fat (LCT - 40% of total energy)	Weight or body composition changes Quality of life Symptoms Functional status

Abbreviations: KQ = Key Question; PMID = PubMed Identification Number; NR = not reported; CINV = hemotherapy-induced nausea and vomiting; LCT = long chain triglycerides.

*For select studies only.

†Reports median age when mean is not available.

Route or Timing of Nutritional Interventions

Table F.4. Characteristics of included s	tudies: route or timing of n	tritional interventions fo	or cancer and treatmer	t related symptoms
(KQ3)				

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Brown, 2017 ³⁸ (28797454) Other Key Question 1 & 2, 3 Not conducted	131 60 years 12% female Race NR	Head & Neck 31% Stage IV disease Multiple therapies Limited to Malnourished NR Screening tool NR	Nutrition therapy (early feeding intervention via prophylactic gastrostomy tube) Multiple settings Dietitian/nutritionist Enteral	Usual care which commenced feeding when clinically indicated	Weight or body composition changes Changes in nutritional status Readmissions or emergency room visits Length of stay Treatment tolerance Quality of life
Feo, 2004 ¹²⁷ (15144242) Europe Key Question 2, 3 Not conducted	100 68 years % female NR Race NR	Gastrointestinal 0% Stage IV disease Surgery alone No Screening tool NR	Nutritiontherapy (clear oral liquids) Inpatient Provider NR Oral	Fasting until passage of flatus then NG diet	Adverse events Length of stay Quality of life
Mudge, 2018 ⁴³ (1596002) Other Key Question 1 & 2, 3 Not conducted	278 64 years 19% female Race NR	Gastrointestinal % Stage IV disease NR Surgery alone No Other tool	Nutrition therapy (immunonutrition (pre-, peri-, and post-op)) Inpatient Physician Other	Perioperative standard nutrition	Adverse events Survival Length of stay Quality of life
Roberts, 2003 ¹⁴⁰ (13130320) North America Key Question 2, 3 Not conducted	55 44 years 100% female Race NR	Other 58% Stage IV disease Multiple therapies No Multiple tools	Nutrition therapy (TPN) Inpatient Provider NR Parenteral	Oral diet post-surgery with IV fluids	Weight or body composition changes Adverse events Length of stay Quality of life

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Sadasivan, 2012 ¹⁴² (22973911) Asia Key Question 2, 3 Not conducted	100 Age NR % female NR Race NR	Head & Neck 50% Stage IV disease Multiple therapies No Screening tool NR	Nutrition therapy (percutaneous endoscopic gastrostomy) Setting NR Multiple providers Enteral	Nasogastric tube administration	Weight or body composition changes Adverse events Treatment tolerance
Sun, 2018 ¹⁴⁵ (28549015) Asia Key Question 2, 3 Not conducted	280 63 years 30% female Race NR	Gastrointestinal 0% Stage IV disease Surgery alone No NRS-2002	Nutrition therapy (early oral feeding starting on POD 1) Inpatient Multiple providers Oral	Late Oral Feeding starting on POD 7	Adverse events Readmissions or emergency room visits Survival Length of stay Quality of life Symptoms
Tao, 2020 ¹⁴⁶ (31512101) Asia Key Question 2, 3 Not conducted	120 64 years 52% female Race NR	Gastrointestinal 11% Stage IV disease Surgery alone No Screening tool NR	Nutrition therapy (jejunostomy feeding starting on POD 1 until 3 wks past discharge) Inpatient Provider NR Enteral	Nasogastric feeding starting on POD 1 until discharge	Weight or body composition changes Changes in nutritional status Adverse events Survival Length of stay Quality of life
Wong TX, 2022 ⁴⁴ (35276977) Asia Key Question 1 & 2, 3 Not conducted	91 60 years 76% female Race NR	Multiple Cancers % Stage IV disease NR Surgery Yes Screening tool NR	Group SS: ONS (milk- based formula fortified with micronutrients)for 5 to 14 days preoperatively and postoperatively until discharge Group SS-E: ONS for 5 to 14 days pre- operatively and for 90- days post-discharge Multiple settings Provider NR	ONS postoperatively until discharge	Weight or body composition changes Functional status

Abbreviations: KQ = Key Question; PMID = PubMed Identification Number; NR = not reported; TPN = total parenteral nutrition; NG = nasogastric; POD = post-operative day; NRS = nutritional risk screening. *For select studies only. †Reports median age when mean is not available.

Nutrition Support Including Oral Nutrition Supplements

Table F.5. Characteristics of included studies: use of nutrition support including oral nutrition support for cancer and treatment related symptoms (KQ3)

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Baker, 2015 ¹⁵⁴ (25827292) Other Key Question 2, 3 Not conducted	109 63 years 100% female Race NR	Other 10% Stage IV diseaseMultiple therapies Yes Other tool	Nutrition (intraoperative nasojejunal tube placement and enteral feeding until adequate oral intake could be maintained) Inpatient Multiple providers Enteral	Postoperative diet as tolerated	Adverse events Length of stay Quality of life
Bouleuc, 2020 ¹⁵⁶ (32212354) Europe Key Question 2, 3 Not conducted	148 66 years 55% female Race NR	Multiple cancers % Stage IV disease NR Multiple therapies Yes Screening tool NR	Nutrition (parenteral nutrition was administered by central venous route using industrial ternary preparations and systematic daily addition of polyvitamins, trace elements, and electrolytes (sodium, potassium, vitamin K, magnesium, phosphorus), adapted as required) Inpatient Dietitian/nutritionist Parenteral	Oral feeding	Weight or body composition changes Adverse events Survival Quality of life

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Cereda, 2018 ¹⁵⁷ (29111172) Europe Key Question 2, 3 Not conducted	159 65 years 28% female Race NR	Head & Neck 29% Stage IV disease Multiple therapies No Screening tool NR	Nutrition (2 bottles/day 250 mL of a ready-to- use energy-dense, high-protein, omega-3 enriched oral formula plus counseling) Outpatient Dietitian/nutritionist Oral	Counseling only	Weight or body composition changes Treatment tolerance Quality of life
Chen T, 2021 ¹⁵⁹ (34237976) Asia Key Question 2, 3 High	106 68 years 21% female Race NR	Gastrointestinal 0% Stage IV Surgery No NRS-2002	Post-hospital enteral nutritional emulsion via J-tube: 200-500 ml x 3- 6x/day Outpatient (Home) Patient-delivered but directed by doctor, nurse & nutritionist J-tube x 1 month Enteral	Post-hospital gradual transition to soft to regular diet after 1 month (J tube in place x 1 mo, not used at home)	Weight or body composition changes Changes in nutritional status Symptoms
Chen X, 2021 ⁸ (33752148) Asia Key Question 1, 3 Medium	139 59 years 38% female Race NR	Gastrointestinal 0% Stage IV Surgery (post-chemo- radiotherapy) No Screening Tool NR	Oral Nutrition (1000 ml of 10% glucose solution 10 hr. before surgery + 500 ml of same 2-3 hr. before surgery) Setting NR Provider NR Oral	500 ml of 10% glucose solution 2-3 hr. before surgery	Adverse events Length of stay Symptoms Readmissions or Emergency room visits
Faccio, 2021 ¹⁶² (32363940) Other Key Question 2, 3 Not conducted	85 59 years 60% female Race NR	Multiple cancers % Stage IV disease NR Chemotherapy alone No Screening tool NR	Nutrition (nutritional supplement with counseling) Outpatient Dietitian/nutritionist Oral	Nutritional counseling alone	Weight or body composition changes Adverse events Quality of life

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Feng, 2022 ⁴⁸ (34362624) Asia Key Question 1 & 2, 3 Low	126 70 years 37% female Race NR	Gastrointestinal % Stage IV NR Surgery No Screening tool NR	Preoperative oral CHO 200 ml. 2-3 hr. before surgery & postoperative early oral feeding (POD1=special clear liquid but no TPN; POD2=normal diet) Hospital inpatient Provider NR Oral	Fasted pre-surgery (> 8 hrs.) + TPN x 24 hr. post-surgery; advance diet as tolerated Hospital inpatient Provider NR Oral & TPN	Adverse events Readmissions or Emergency room visits Symptoms
Gavazzi, 2016 ¹⁶³ (27391922) Europe Key Question 2, 3 High	79 Age NR 38% female Race NR	Gastrointestinal 0% Stage IV disease Multiple therapies Yes NRS-2002	Nutrition (home enteral nutrition) Inpatient Multiple providers Enteral	Specific nutritional indications including total energy and protein requirements were provided	Weight or body composition changes Treatment tolerance Quality of life Functional Status
He F, 2022 ¹⁰ (35406085) Asia Key Question 1, 3 Low	67 62 years 29% female Race NR	Gastrointestinal % Stage IV disease NR Surgery No NRS-2002	Preoperative ONS 500 ml. x 7 days (plus NJ feedings day 1-5 post- surgery) Outpatient Dietitian Oral	Preoperative dietary advice (plus NJ feedings day 1-5 post- surgery)	Adverse events Readmissions or emergency room visits Symptoms
Huang, 2020 ¹⁶⁴ (33032180) Asia Key Question 2, 3 Not conducted	114 50 years 27% female Asian	Head & Neck 0% Stage IV disease Multiple therapies No NRS-2002	Nutrition (prophylactic oral nutritional supplements) Other Dietitian/nutritionist Oral	Regular diet	Weight or body composition changes Changes in nutritional status Adverse events Survival Treatment tolerance Quality of life Symptoms

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Jiang, 2018 ¹⁶⁷ (30633580) Asia Key Question 2, 3 Not conducted	100 Age NR 31% female Race NR	Head & Neck 57% Stage IV disease Multiple therapies No Other tool	Nutrition (oral nutritional supplement) Outpatient Multiple providers Oral	No oral nutritional supplement	Weight or body composition changes Changes in nutritional status Adverse events Quality of life
Jin, 2018 ¹⁶⁸ (30205371) Asia Key Question 2, 3 High	80 Age NR 26% female Race NR	Gastrointestinal % Stage IV Disease NR Surgery alone No Other metric	Nutrition (PN day 1 POD to 4-8 POD) Inpatient Provider NR Parenteral	Isotonic electrolyte solution	Changes in nutritional Status Quality of life
Kanat, 2013 ¹⁶⁹ (23748819) Europe Key Question 2, 3 Not conducted	69 60 years 14% female Race NR	Multiple cancers 70% Stage IV disease Multiple therapies No Screening tool NR	Nutrition ((1) megesterol acetate (MA) plus meloxicam; (2) MA plus meloxicam plus oral eicosapentaenoic acid (EPA)-enriched nutritional supplement or (3) meloxicam plus oral EPA-enriched nutritional supplement Outpatient Provider NR Oral	*No control (three treatment groups)	Weight or body Composition changes Adverse events Quality of life
Kong, 2018 ⁵² (30055788) Asia Key Question 1 & 2, 3 High	144 Age NR 37% female Race NR	Gastrointestinal 7% Stage IV disease Surgery alone Yes Other tool	Nutrition (standard oral nutritional supplement (Ensure powder) for 2 weeks before gastrectomy and for 4 weeks postoperatively in malnourished patients) Multiple settings Dietitian/nutritionist Oral	Standard meals designed for post- gastrectomy patients	Weight or body composition changes Adverse events Quality of life

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Kruger, 2016 ¹² (27861546) Europe Key Question 1, 3 High	100 64 years 57% female Race NR	Gastrointestinal % Stage IV disease NR Surgery alone No NRS-2002	Nutrition (in-hospital peripheral intravenous nutrition on fasting days (1000 ml peripheral intravenous nutrition, 700 kcal)) Inpatient Multiple providers Parenteral	1000 ml isotonic electrolyte solution	Weight or body composition changes Quality of life
Lyu, 2022 ¹⁷⁸ (35280748) Asia Key Question 2, 3 High	222 Age NR 21% female Race NR	Gastrointestinal 0% Stage IV Multiple therapies No Multiple tools	Nutrition (enteral nutrition with ONS (Nutrison (Nutricia)) Outpatient Nutrition support team (clinicians, nutritionists, pharmacologists and nutrition nurses) Oral	Unsystematic nutrition based on general eating conditions, blood tests & treatment toxicities	Weight or body composition changes Adverse events Survival Treatment Tolerance
Meng, 2021 ¹⁸⁰ (32563598) Tan, 2021 ¹⁸¹ (32563599) Asia Key Question 2, 3 Low	353 60 years 32% female Race NR	Gastrointestinal 7% Stage IV disease Surgery alone Yes NRS-2002	Nutrition (Oral nutritional supplements with dietary advice) Other Provider NR Oral	Dietary advice alone	Weight or body composition changes Treatment tolerance Symptoms
Okabayashi, 2011 ¹⁸⁵ (20852905) Asia Key Question 2,3 High	96 67 years 70% female Race NR	Gastrointestinal % Stage IV disease NR Surgery alone No Screening tool NR	Nutrition (carbohydrate- and BCAA-enriched softpowder nutrient mixture) Inpatient Multiple providers Oral	Conventional diet with no supplementation	Weight or body composition changes Quality of life

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Poon, 2004 ⁵⁵ (15043519) Asia Key Question 1 & 2, 3 Not conducted	88 59 years 7% female Race NR	Other 0% Stage IV disease Chemotherapy alone No Screening tool NR	Nutrition (nutritional supplementation with BCAAs twice a day in the morning and evening in addition to the usual diet) Inpatient Provider NR Oral	Standard diet	Weight or body composition changes Adverse events Survival Functional status
Ravasco, 2005 ¹⁸⁶ (15920748) Ravasco, 2005 ¹⁸⁷ (15684319) Europe Key Question 2, 3 Not conducted	75 Age NR % female NR Race NR	Head & Neck % Stage IV disease NR Radiation alone No Other tool	Nutrition ((1) diet plus regular food; (2) regular diet plus oral nutritional supplement) Outpatient Dietitian/nutritionist Oral	Regular food	Weight or body composition changes Changes in nutritional status Treatment tolerance Quality of life Symptoms
Rizvanovic, 2019 ¹⁵ (31309323) Europe Key Question 1, 3 Low	50 61 years 46% female Race NR	Gastrointestinal % Stage IV disease NR Surgery alone No NRS-2002	Nutrition (400 mL of carb drink 22 h before surgery and 200 mL 2 h before anaesthesia) Inpatient Provider NR Oral	Fasted for 8 hours before surgery	Length of Stay Symptoms Functional Status
Sanchez-Lara, 2014 ¹⁸⁸ (24746976) Other Key Question 2, 3 Not conducted	92 60 years 53% female Race NR	Other 62% Stage IV Disease Chemotherapy alone No Other tool	Nutrition (oral nutritional supplement EPA) Setting NR Provider NR Oral	Standardized diets of 1400, 1600, 1800, 2000, or 2200kcal.	Weight or body composition changes Adverse events Survival Quality of life Symptoms Functional status
Sim, 2022 ¹⁹¹ (35225460) Asia Key Question 2,3 High	58 64 years 20% female Race NR	Gastrointestinal 63% Stage IV disease Chemotherapy No Other tool	ONS twice a day (400ml, 400kcal) along with nutrition counseling and education Outpatient Multiple providers Oral	Nutrition counseling and education along with a weekly call from a trained dietitian	Changes in nutritional status Quality of Life Symptoms

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Sittirai, 2021 ⁵⁹ (34371175) Asia Key Question 1 & 2, 3 Not conducted	126 57 years 34% female Race NR	Head & Neck 87% Stage IV disease Surgery No Screening tool NR	ONS (Immune- enhancing diet containing arginine, glutamine, and fish oil- derived fatty acids) Inpatient Provider NR Oral	Hospital-prepared blenderized diet	Adverse events Length of stay Symptoms
Tesar, 2022 ¹⁷ (35258042) Europe Key Question 1, 3 Medium	120 65 years 34% female Race NR	Gastrointestinal 0% Stage IV disease Surgery No NRS-2002	ONS (oral nutritional supplements 2x a day for 7 days before surgery) Inpatient Provider NR Oral	No oral nutritional supplement	Weight or body composition changes Adverse events Length of Stay Functional status
Vidal, 2016 ³⁵ (27770454) Europe Key Question 2, 3 Not conducted	157 66 years 34% female Race NR	Other 8% Stage IV disease Surgery alone No Screening tool NR	Nutrition (TPN (Nutriflex special 70/240) for 5 days starting within 24 h after surgery) Outpatient Provider NR Parenteral	Oral nutrition alone started on day of surgery	Weight or body composition changes Survival Quality of life Functional status
Wang J, 2022 ¹⁹³ (35126900) Asia Key Question 2, 3 High	80 43 years 43% female Race NR	Gastrointestinal 0% Stage IV disease Multiple therapies No Screening Tool NR	Enteral nutrition (Early enteral nutrition via nasointestinal tube 12- 24 hours after surgery continuously for 7 days post-surgery) Inpatient Provider NR Enteral	No enteral nutrition	Adverse events Length of stay Functional status
Wu, 2017 ¹⁹⁴ (27208039) Asia Key Question 2, 3 Medium	73 56 years 32% female Race NR	Gastrointestinal 0% Stage IV disease Surgery alone No Screening tool NR	Nutrition (EN + PN for ≥7 days; EN started within 24 hr postsurgery) Inpatient Provider NR Parenteral	EN alone for ≥7 days started within 24 hr postsurgery	Weight or body composition changes Adverse events Length of stay Quality of life

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Xie, 2021 ¹⁹⁵ (34988183) Asia Key Question 2, 3 Medium	77 62 years 19% female Race NR	Gastrointestinal 0% Stage IV disease Surgery No Screening tool NR	Intervention type (300 mL oral nutritional supplement in addition to standard diet) Other Provider NR Oral	Regular diet alone	Weight or body composition changes Adverse events Readmissions or emergency room visits Survival Length of Stay Quality of life
Yang, 2020 ¹⁹⁶ (32833549) Asia Key Question 2, 3 High	120 64 years 23% female Race NR	Gastrointestinal 61% Stage IV diseaseRadiation alone No NRS-2002	Nutrition (nutritional counseling and dietary advice + oral nutritional supplement) Outpatient Multiple providers Oral	Nutritional counseling and dietary advice	Weight or body composition changes Changes in nutritional Status Adverse events Symptoms
Zhu, 2019 ²⁰⁰ (31464391) Asia Key Question 2, 3 Low	140 60 years 32% female Race NR	Gastrointestinal % Stage IV disease NR Surgery alone Yes NRS-2002	Nutrition (500 kcal/d total of ONS (over 3 feedings) from discharge to 90 days + nutrition education) Outpatient Physician Oral	Nutrition education	Weight or body composition changes Adverse events Quality of life Functional status
Zietarska, 2017 ²⁰¹ (29019951) Europe Key Question 2, 3 High	95 64 years 48% female Race NR	Gastrointestinal 32% Stage IV disease Chemotherapy alone Yes Multiple tools	Nutrition (high-energy, high-protein ONS, 250 ml/day x 3 mo.) Other Multiple providers Oral	Without nutritional support (ie, ONS)	Weight or body composition changes Changes in nutritional status Adverse events Treatment tolerance Quality of life Symptoms Functional status

Abbreviations: KQ = Key Question; PMID = PubMed Identification Number; NR = not reported; NRS = nutritional risk screening; BCAA = branched-chain amino acid, ONS = oral nutrition supplement. *For select studies only. †Reports median age when mean is not available.

Multi-Component Interventions

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Baldwin, 2011 ⁶² (21733143) Europe Key Question 1 & 2, 3 Not conducted	358 66 years 37% female Race NR	Multiple cancers % Stage IV disease NR Chemotherapy alone Yes Screening tool NR	Multi-modality ((1) dietary advice alone; (2) nutritional supplement alone; and (3) dietary and nutritional supplement) Outpatient Dietitian/nutritionist Oral	No intervention	Survival Quality of life
Demark-Wahnefried, 2008 ²⁰³ (18501061) North America Key Question 2, 3 Not conducted	90 42 years 100% female Non-Hispanic White	Other 0% Stage IV disease Chemotherapy alone No Screening tool NR	Multi-modality (calcium rich diet with exercise) Outpatient Dietitian/nutritionist Oral	Calcium rich with exercise and high fruit and vegetable, low-fat diet	Weight or body composition changes Quality of life
Pettersson, 2014 ²¹⁶ (25467005) Europe Key Question 3 Not conducted	130 66 years 0% female Race NR	Other % Stage IV disease NR Radiation alone No Other tool	Multi-modality (advised to avoid foods high in insoluble dietary fiber and lactose during 26- month period) Outpatient Dietitian/nutritionist Oral	Usual diet, but were able to receive counseling	Symptoms
Poulsen, 2014 ²⁰⁵ (24269077) Europe Key Question 2, 3 Not conducted	61 66 years 57% female Race NR	Multiple cancers % Stage IV disease NR Multiple therapies No NRS-2002	Multi-modality (nutritional counseling plus high-protein nutrition supplement) Outpatient Dietitian/nutritionist Oral	The control group was nutritionally instructed by the nurses with the possibility to call for a dietitian if needed	Weight or body composition changes Quality of life Symptoms

Table F.6. Characteristics of included studies: use of multi-component interventions for cancer and treatment related symptoms (KQ3)

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Silander, 2012 ²⁰⁷ (21374756) Silander, 2013 ²⁰⁸ (23169469) Europe Key Question 2, 3 Not conducted	134 62 years 32% female Race NR	Head & Neck 74% Stage IV disease Chemotherapy alone Yes Screening tool NR	Multi-modality (nutritional counseling plus percutaneous endoscopic gastrostomy feeding Outpatient Dietitian/nutritionist Enteral	Nutritional support according to clinical praxis, which could include nutritional advice and enteral tube feeding	Weight or body composition changes Changes in nutritional status Adverse events Survival Length of stay Quality of life Symptoms Functional status
Song, 2017 ²¹⁰ (28943912) Asia Key Question 2, 3 Not conducted	86 56 years 48% female Race NR	Head & Neck % Stage IV disease NR Surgery alone No Other tool	Multi-modality (preoperative nutritional support therapy, postoperative nutritional therapy during hospitalization, and nutrition therapy outside the hospital) Multiple settings Multiple providers Other	Routine nutritional support therapy after surgery	Changes in nutritional status Adverse events Readmissions or emergency room visits Survival Length of stay Quality of life
Uster, 2013 ²¹¹ (24103511) Europe Key Question 2, 3 Not conducted	58 65 years 21% female Race NR	Multiple cancers % Stage IV disease NR Cancer treatment type NR Yes NRS-2002	Multi-modality (standardized individual nutritional therapy, including counseling by a dietitian, food fortification, and ONS if required) Outpatient Dietitian/nutritionist Oral	Standard care	Weight or body composition changes Changes in nutritional status Quality of life Functional status

Abbreviations: KQ = Key Question; PMID = PubMed Identification Number; NR = not reported. *For select studies only. †Reports median age when mean is not available.

Risk of Bias and Outcomes

Author, Year, PMID	Outcome Timing	Selection Bias	Detection Bias	Performance Bias	Fidelity Bias	Reporting Bias	Attrition %	Overall Rating
Golkhalkhali, 2018 ⁹⁰ (928857425)	2 mo, 6 mo	Medium	Medium	Medium	Medium	Low	Low (0)	Medium
Serrano, 2022 ³⁰ (35606184)	4 wks, 12 weeks post- surgery	Low	Low	Medium	High	Low	Low (9%)	Medium
Tsuchiya, 2016 ³⁴ (27306219)	28 days	Low	Medium	High	X	X	X	High
Wang, 2017 ¹⁰⁵ (28927119)	6 months	Medium	Medium	High	Medium	Low	Low (0)	High
Wang, 2010 ¹⁰⁷ (21208095)	Inpatient	Low	Low	Low	High	Low	Low 4%	Medium

Table F.7.1. Risk of bias assessment: use of dietary supplements for cancer and treatment related symptoms (KQ3)

Note: X indicates domain not assessed due to previously determined high risk of bias.

Abbreviations: PMID = PubMed Identification Number; NR = not reported; KQ = Key Question.

Table F.7.2. Outcomes a	ssessment: use of dietary	supplements for cancer	and treatment related	symptoms (KQ3) (low and m	nedium
ROB studies)	-				

Author, Year, RoB Timing	N	Treatment	Control	Weight/ Body Comp.	Changes in Nutr. Status	Adverse Events	Readmissions/ Emergency Room Visits	Survival	LO S	Treat ment Tolera nce	QO L	Sympto ms	Functio nal Status
Golkhalkhal i, 2018 ⁹⁰ (28857425) Medium 2 mo, 6 mo	140	Probiotic (4 wks) + 2 g. omega 3 Fas (8 wks)	Placebo	-	-	-	-	-	-	NA	↑ 2 mo ↑ 6 mo	Î	NA
Serrano, 2022 ³⁰ (35606184) Medium 4 wks, 12 weeks post- surgery	71	Protein supplement ation rich in arginine and omega-6 and carbohydrat e loading	One placebo to each supplem ent	-	-	-	-	-	-	NA	↔ FA CT- G	NA	NA

Author, Year, RoB Timing	N	Treatment	Control	Weight/ Body Comp.	Changes in Nutr. Status	Adverse Events	Readmissions/ Emergency Room Visits	Survival	LO S	Treat ment Tolera nce	QO L	Sympto ms	Functio nal Status
Wang, 2010 ¹⁰⁷ (21208095) Medium Inpatient	229	Triglyceride & protein- enriched EN	Standard EN	-	-	-	-	-	-	NA	NA	\leftrightarrow	NA

Abbreviations: PMID = PubMed Identification Number; ROB=risk of bias; BMI: body mass index; FFM: fat free mass; AE: adverse events; LOS: length of stay; QOL: quality of life; PRO: patient reported outcome; FA: fatty acid: wks: weeks: mo: month: EN: enteral nutrition; EPA: eicosapentaenoic acid; CHO: carbohydrate; ONS= oral nutrition supplement; KQ = Key Question

1: Intervention group had a statistically significantly better outcome than comparison group (e.g. fewer AEs, shorter LOS than comparison group).

: Intervention group had a statistically significantly worse outcome than comparison group (e.g. more AEs, longer LOS).

 \leftrightarrow : No statistically significant difference between groups.

Table F.8.1. Risk	of bias assessme	nt: use of nutrit	tion support in	cluding oral r	nutrition sup	plements for can	cer and treatme	nt related
symptoms (KQ3)								

Author, Year,	Outcome	Selection Bias	Detection	Performance Bias	Fidelity Bias	Reporting Bias	Attrition %	Overall Pating
Chen T, 2021 ¹⁵⁹ (34237976)	1 month post- hospital discharge after surgery	Low	High	X	X	X	x	High
Chen X, 2021 ⁸ (33752148)	Inpatient, 30 d	Low	Medium	Medium	Medium (NR)	Low	Low (2%)	Medium
Feng, 2022 ⁴⁸ (34362624)	Inpatient, 30 d	Low	Low	Medium	Medium	Low	Low (0)	Low
Gavazzi 2016 ¹⁶³ (27391922)	6 months	Low	High	Х	X	X	Х	High
He F, 2022 ¹⁰ (35406085)	Inpatient, 30 d	Low	Low	Medium	Medium (NR)	Low	Low (2%)	Low
Jin, 2018 ¹⁶⁸ (30205371)	Inpatient	Medium	Medium	High	X	X	Х	High
Kruger, 2016 ¹² (27861546)	Х	Medium	High	Х	X	X	Х	High
Kong, 2018 ⁵² (30055788)	30 days	Low	Medium	High	X	X	Х	High
Lyu, 2022 ¹⁷⁸ (35280748)	After treatment completion (2- 3 cycles) & up to 4 yrs	Low	Medium	High	Medium	Low	Medium (19%)	High
Meng, 2021 ¹⁸⁰ (32563598) Tan, 2021 ¹⁸¹ (32563599)	3 mo	Low	Low	Medium	Medium	Low	Low 5%	Low

Author, Year, PMID	Outcome Timing	Selection Bias	Detection Bias	Performance Bias	Fidelity Bias	Reporting Bias	Attrition %	Overall Rating
Okabayashi, 2011 ¹⁸⁵ (20852905)	1 year	Medium	Medium	High	X	X	X	High
Rizvanovic, 2019 ¹⁵ (31309323)	POD 2	Low	Low	Medium	Medium	Low	Low (0)	Low
Sim, 2022 ¹⁹¹ (35225460)	Weeks 0, 4, 8	Medium	Medium	High	Medium	Low	High (31%)	High
Tesar, 2022 ¹⁷ (35258042)	Inpatient, 30d	Medium	Medium	Medium	Medium	Low	Low (2%)	Medium
Wang J, 2022 ¹⁹³ (35126900)	Inpatient	Low	High	Х	X	X	X	High
Wu, 2017 ¹⁹⁴ (27208039)	Inpatient; 90 d QOL	Low	Medium	Medium	Medium	Low	Medium 13%	Medium
Xie, 2021 ¹⁹⁵ (334988183)	Inpatient, 1 mo., 3 mo., 6 mo.	Low	Medium	Medium	Low	Low	Medium (14%)	Medium
Yang, 2020 ¹⁹⁶ (32833549)	3 days	Low	Medium	High	X	X	X	High
Zhu, 2019 ²⁰⁰ (31464391)	30, 60, 90 days	LowMedium	Low	Medium	MediumLow	Low	Medium 19%	Low
Zietarska, 2017 ²⁰¹ (29019951)	12 weeks	Low	Low	Medium	Low	Medium	High (24%)	High

Note: X indicates domain not assessed due to previously determined high risk of bias. Abbreviations: PMID = PubMed Identification Number; POD=postoperative day; QOL = quality of life; mo = months; d = days; KQ = Key Question.

Table F.8.2. Outcomes assessment: use of nutrit	ion support including oral nutrition supplements for cancer and treatment related
symptoms (KQ3) (low and medium ROB studies)	

Author, Year, RoB Timing	N	Treatment	Control	Weight/ Body Comp.	Changes in Nutr. Status	Adverse Events	Readmissions/ Emergency Room Visits	Survival	LOS	Treat. Toler ance	QoL	Symptoms	Functional Status
Chen X, 2021 ⁸ (3375214 8) Medium Inpatient, 30 d	139	Double-dose oral CHO drink before gastrectomy	Single-dose oral CHO drink before gastrectomy	-	-	-	-	-	-	NA	NA	\leftrightarrow	NA

Author, Year, RoB Timing	N	Treatment	Control	Weight/ Body Comp.	Changes in Nutr. Status	Adverse Events	Readmissions/ Emergency Room Visits	Survival	LOS	Treat. Toler ance	QoL	Symptoms	Functional Status
Feng, 2022 ⁴⁸ (3436262 4) Low Inpatient, 30 d	126	Preoperative oral CHO 200 ml 2-3 hr. before surgery & postoperativ e early oral feeding	Fasted pre- surgery (> 8 hrs.) + TPN x 24 hr. post-surgery	-	-	-	-	-	-	NA	NA	↑ nausea	NA
He F, 2022 ¹⁰ (3540608 5) Low Inpatient, 30 d	67	Preoperative ONS 500 ml. x 7 days	Dietary advice	-	-	-	-	-	-	NA	NA	\leftrightarrow	NA
Meng, 2021 ¹⁸⁰ (3256359 8) Tan 2021 ¹⁸¹ (3256359 9) Low	353	Oral nutritional supplements with dietary advice	Dietary advice alone	-	-	-	-	-	-	↑ Chem other apy modiv icatio ns, delay s, dose reduc tions, termi nation s	NA	↑ fatigue, appetite loss	NA
Rizvanov ic, 2019 ¹⁵ (3130932 3) Low POD 2	50	Preop. oral CHO supplement	Preop. fasting	-	-	-	-	-	-	NA	NA	↑ nausea ↔ vomiting ↔ pain	1

Author, Year, RoB Timing	N	Treatment	Control	Weight/ Body Comp.	Changes in Nutr. Status	Adverse Events	Readmissions/ Emergency Room Visits	Survival	LOS	Treat. Toler ance	QoL	Symptoms	Functional Status
Tesar, 2022 ¹⁷ (3525804 2) Medium Inpatient, 30d	120	oral nutritional supplements 2x a day for 7 days before surgery	No oral nutrition supplement	-	-	-	-	-	-	NA	NA	NA	↔ Barthel Index
Wu, 2017 ¹⁹⁴ (2720803 9) Medium Inpatient	80	Post-op EN+PN via NJT/ j-tube	Post-op EN via NJT/ j-tube	-	-	-	-	-	-	NA	UTD	NA	NA
Xie, 2021 ¹⁹⁵ (3349881 83) Medium Inpatient , 1 mo., 3 mo., 6 mo.	77	300 mL oral nutritional supplement in addition to standard diet	Regular diet alone	-	-	-	-	-	-	NA	↔ EOR TC QLQ -C30	NA	NA
Zhu, 2019 ²⁰⁰ (3146439 1) Low 30-90 days	140	500 kcal ONS 3x/d x 90d + dietary advice from doctor at discharge	Dietary advice from doctor at hospital discharge	-	-	-	-	-	-	NA	EQ- 5D ↔ 30d ↔ 60d ↔ 90d	NA	↔ grip

Abbreviations: PMID = PubMed Identification Number; ROB=risk of bias; POD = postoperative day; AE = adverse events; BMI: body mass index; FFM: fat free mass; FM = fat mass; LOS: length of stay; QOL: quality of life; PRO: patient reported outcome; EN: enteral nutrition: PN: parenteral nutrition: ONS: oral nutritional supplement: EQ-5D: EuroQoL-5D: UTD: unable to determine: CHO: carbohydrade: BW: body weight; d: day; NJT = nasojejunal tube; NA=not assessed.

-: Not applicable

↑: Intervention group had a statistically significantly better outcome than comparison group (e.g. fewer AEs, shorter LOS than comparison group).

: Intervention group had a statistically significantly worse outcome than comparison group (e.g. more AEs, longer LOS).

 \leftrightarrow : No statistically significant difference between groups.

Appendix G. Evidence Tables for Chapter 9

Special Diets

Table G.1. Characteristic	of included studies: s	pecial diets ((KQ4)
---------------------------	------------------------	----------------	-------

Study (PMID) Region Key Question RoB Rating*	Sample Size Age (Mean)† Sex (% Female) Race/Ethnicity	Cancer Type Stage IV Disease (%) Cancer Treatment Type Limited to Malnourished (Yes or No) Screening Tool	Intervention Type Intervention Delivery Setting Provider Delivering Intervention Route of Intervention	Comparison Description	Outcomes
Cho, 2022 ²¹⁷ (35845810) Asia Key Question 4 Not conducted	78 60 years 100% females Race/ethnicity NR	Other 0% Stage IV Chemotherapy No Screening tool NR	Mediterranean diet + advice (1500 kcal/d) Other/Home (delivered 10 meals/week) Nutritionist Oral	Dietary advice (1500 kcal/d) Home	Weight or body composition changes
Harvie, 2022 ²¹⁸ (34912072) Europe Key Question 4 Not conducted	172 52 years 100% female 91% white	Other 0% Stage IV Chemotherapy No Screening Tool NR	Intermittent 25% energy restriction (Mediterranean diet 5d/wk. + low energy, low CHO diet 2d/wk.) + Physical activity program Outpatient Dietitian Oral	Continuous 25% energy restriction (Mediterranean diet) + physical activity program	Weight or body composition changes Symptoms Treatment tolerance
Vafa, 2020 ²¹⁵ (32898787) Asia Key Question 3, 4 Not conducted	135 53 years 100% female Race NR	Other 0% Stage IV disease Multiple therapies No Screening tool NR	Special Diets (Calorie restricted diet + CFU symbiotic supplement) Outpatient Not reported Oral	placebo	Weight of body composition changes Quality of life
Villarini, 2012 ²¹⁹ (22896285) Europe Key Question 4 Not conducted	94 51 years 100% female Race NR	Other 0% Stage IV disease Multiple therapies No Screening tool NR	Special diets (Mediterranean diet) Outpatient Multiple providers Oral	General recommendations for the prevention of cancer and a baseline brief kitchen course	Weight or body composition changes

Abbreviations: KQ = Key Question; PMID = PubMed Identification Number; NR = not reported; CFU = colony forming unit.

*For select studies only.

†Reports median age when mean is not available.

Appendix H. References for Appendixes C–G

- Dechaphunkul T, Arundon T, Raungkhajon P, et al. Benefits of immunonutrition in patients with head and neck cancer receiving chemoradiation: A phase II randomized, double-blind study. Clin Nutr. 2022 Feb;41(2):433-40. doi: 10.1016/j.clnu.2021.12.035. PMID: 35007812.
- de Miranda Torrinhas RS, Santana R, Garcia T, et al. Parenteral fish oil as a pharmacological agent to modulate post-operative immune response: a randomized, double-blind, and controlled clinical trial in patients with gastrointestinal cancer. Clin Nutr. 2013 Aug;32(4):503-10. doi: 10.1016/j.clnu.2012.12.008. PMID: 23398953.
- Feijo PM, Rodrigues VD, Viana MS, et al. Effects of omega-3 supplementation on the nutritional status, immune, and inflammatory profiles of gastric cancer patients: A randomized controlled trial. Nutrition. 2019 May;61:125-31. doi: 10.1016/j.nut.2018.11.014. PMID: 30710885.
- Kaya SO, Akcam TI, Ceylan KC, et al. Is preoperative protein-rich nutrition effective on postoperative outcome in non-small cell lung cancer surgery? A prospective randomized study. J Cardiothorac Surg. 2016 Jan 19;11:14. doi: 10.1186/s13019-016-0407-1. PMID: 26782276.
- Lende TH, Austdal M, Varhaugvik AE, et al. Influence of pre-operative oral carbohydrate loading vs. standard fasting on tumor proliferation and clinical outcome in breast cancer patients horizontal line a randomized trial. BMC Cancer. 2019 Nov 8;19(1):1076. doi: 10.1186/s12885-019-6275-z. PMID: 31703648.
- Burden ST, Gibson DJ, Lal S, et al. Pre-operative oral nutritional supplementation with dietary advice versus dietary advice alone in weight-losing patients with colorectal cancer: single-blind randomized controlled trial. J Cachexia Sarcopenia Muscle. 2017 Jun;8(3):437-46. doi: 10.1002/jcsm.12170. PMID: 28052576.
- Burden ST, Hill J, Shaffer JL, et al. An unblinded randomised controlled trial of preoperative oral supplements in colorectal cancer patients. J Hum Nutr Diet. 2011 Oct;24(5):441-8. doi: 10.1111/j.1365-277X.2011.01188.x. PMID: 21699587.

- Chen X, Li K, Yang K, et al. Effects of preoperative oral single-dose and double-dose carbohydrates on insulin resistance in patients undergoing gastrectomy:a prospective randomized controlled trial. Clin Nutr. 2021 Apr;40(4):1596-603. doi: 10.1016/j.clnu.2021.03.002. PMID: 33752148.
- Hamamoto H, Yamamoto M, Masubuchi S, et al. The impact of preoperative carbohydrate loading on intraoperative body temperature: a randomized controlled clinical trial. Surg Endosc. 2018 Nov;32(11):4393-401. doi: 10.1007/s00464-018-6273-2. PMID: 29915986.
- He FJ, Wang MJ, Yang K, et al. Effects of Preoperative Oral Nutritional Supplements on Improving Postoperative Early Enteral Feeding Intolerance and Short-Term Prognosis for Gastric Cancer: A Prospective, Single-Center, Single-Blind, Randomized Controlled Trial. Nutrients. 2022 Apr 1;14(7):01. doi: 10.3390/nu14071472. PMID: 35406085.
- Kabata P, Jastrzebski T, Kakol M, et al. Preoperative nutritional support in cancer patients with no clinical signs of malnutrition--prospective randomized controlled trial. Support Care Cancer. 2015 Feb;23(2):365-70. doi: 10.1007/s00520-014-2363-4. PMID: 25091056.
- Kruger J, Meffert PJ, Vogt LJ, et al. Early Parenteral Nutrition in Patients with Biliopancreatic Mass Lesions, a Prospective, Randomized Intervention Trial. PLoS One. 2016;11(11):e0166513. doi: 10.1371/journal.pone.0166513. PMID: 27861546.
- Lee SY, Lee J, Park HM, et al. Impact of Preoperative Immunonutrition on the Outcomes of Colon Cancer Surgery: Results from a Randomized Controlled Trial. Ann Surg. 2021 Aug 4;04:04. doi: 10.1097/SLA.000000000005140. PMID: 34353994.
- Martin RC, 2nd, Agle S, Schlegel M, et al. Efficacy of preoperative immunonutrition in locally advanced pancreatic cancer undergoing irreversible electroporation (IRE). Eur J Surg Oncol. 2017 Apr;43(4):772-9. doi: 10.1016/j.ejso.2017.01.002. PMID: 28162818.

- Rizvanovic N, Nesek Adam V, Causevic S, et al. A randomised controlled study of preoperative oral carbohydrate loading versus fasting in patients undergoing colorectal surgery. Int J Colorectal Dis. 2019 Sep;34(9):1551-61. doi: 10.1007/s00384-019-03349-4. PMID: 31309323.
- Shen Y, Zhao X, Zhao H, et al. Clinical Application of Enteral Nutrition Combined with Microbial Preparation for Intestinal Preparation in Elderly Patients with Colorectal Cancer. Med Sci Monit. 2022 Mar 21;28:e935366. doi: 10.12659/MSM.935366. PMID: 35307727.
- Tesar M, Kozusnikova V, Martinek L, et al. Preoperative nutritional support for patients undergoing elective colorectal cancer surgery does it really work? Biomed Pap Med Fac Univ Palacky Olomouc Czech Repub. 2022 Mar 1;01:01. doi: 10.5507/bp.2022.009. PMID: 35258042.
- Wang F, Hou MX, Wu XL, et al. Impact of enteral nutrition on postoperative immune function and nutritional status. Genet Mol Res. 2015 Jun 10;14(2):6065-72. doi: 10.4238/2015.June.8.4. PMID: 26125807.
- Xu J, Zhong Y, Jing D, et al. Preoperative enteral immunonutrition improves postoperative outcome in patients with gastrointestinal cancer. World J Surg. 2006 Jul;30(7):1284-9. doi: 10.1007/s00268-005-0756-8. PMID: 16830214.
- Zhao Q, Li Y, Yu B, et al. Effects of Preoperative Enteral Nutrition on Postoperative Recent Nutritional Status in Patients with Siewert II and III Adenocarcinoma of Esophagogastric Junction after Neoadjuvant Chemoradiotherapy. Nutr Cancer. 2018 Aug-Sep;70(6):895-903. doi: 10.1080/01635581.2018.1490780. PMID: 30273016.
- Celik JB, Gezginc K, Ozcelik K, et al. The role of immunonutrition in gynecologic oncologic surgery. Eur J Gynaecol Oncol. 2009;30(4):418-21. PMID: 19761135.
- Cereda E, Turri A, Klersy C, et al. Whey protein isolate supplementation improves body composition, muscle strength, and treatment tolerance in malnourished advanced cancer patients undergoing chemotherapy. Cancer Med. 2019 Nov;8(16):6923-32. doi: 10.1002/cam4.2517. PMID: 31568698.

- Ghosh S, Dempsey G, Skelly R, et al. A double blind, randomised, placebo controlled, feasibility phase III clinical trial of peri-operative immuneenhancing enteral nutrition in patients undergoing surgery for advanced head and neck cancer. e-SPEN Journal. 2012;7(3):e107-e14. doi: 10.1016/j.clnme.2011.12.001. PMID: 365552431.
- Haidari F, Abiri B, Iravani M, et al. Randomized Study Design to Test Effects of Vitamin D and Omega-3 Fatty Acid Supplementation as Adjuvant Therapy in Colorectal Cancer Patients. Methods Mol Biol. 2020;2138:337-50. doi: 10.1007/978-1-0716-0471-7_24. PMID: 32219761.
- 25. Healy LA, Ryan A, Doyle SL, et al. Does Prolonged Enteral Feeding With Supplemental Omega-3 Fatty Acids Impact on Recovery Postesophagectomy: Results of a Randomized Double-Blind Trial. Ann Surg. 2017 Nov;266(5):720-8. doi: 10.1097/SLA.00000000002390. PMID: 28742713.
- 26. Jantharapattana K, Orapipatpong O. Efficacy of EPAenriched supplement compared with standard formula on body weight changes in malnourished patients with head and neck cancer undergone surgery: a randomized study. Head Neck. 2020 Feb;42(2):188-97. doi: 10.1002/hed.25987. PMID: 31647147.
- Jo S, Choi SH, Heo JS, et al. Missing effect of glutamine supplementation on the surgical outcome after pancreaticoduodenectomy for periampullary tumors: a prospective, randomized, double-blind, controlled clinical trial. World J Surg. 2006 Nov;30(11):1974-82; discussion 83-4. doi: 10.1007/s00268-005-0678-5. PMID: 16927064.
- Oguz M, Kerem M, Bedirli A, et al. L-alanin-Lglutamine supplementation improves the outcome after colorectal surgery for cancer. Colorectal Dis. 2007 Jul;9(6):515-20. doi: 10.1111/j.1463-1318.2006.01174.x. PMID: 17573745.
- Ryan AM, Reynolds JV, Healy L, et al. Enteral nutrition enriched with eicosapentaenoic acid (EPA) preserves lean body mass following esophageal cancer surgery: results of a doubleblinded randomized controlled trial. Ann Surg. 2009 Mar;249(3):355-63. doi: 10.1097/SLA.0b013e31819a4789. PMID: 19247018.

- Serrano PE, Parpia S, Simunovic M, et al. Perioperative optimization with nutritional supplements in patients undergoing gastrointestinal surgery for cancer: A randomized, placebo-controlled feasibility clinical trial. Surgery. 2022 May 20;20:20. doi: 10.1016/j.surg.2022.04.001. PMID: 35606184.
- Sorensen LS, Thorlacius-Ussing O, Schmidt EB, et al. Randomized clinical trial of perioperative omega-3 fatty acid supplements in elective colorectal cancer surgery. Br J Surg. 2014 Jan;101(2):33-42. doi: 10.1002/bjs.9361. PMID: 24281905.
- Sorensen LS, Rasmussen SL, Calder PC, et al. Longterm outcomes after perioperative treatment with omega-3 fatty acid supplements in colorectal cancer. BJS Open. 2020 Aug;4(4):678-84. doi: 10.1002/bjs5.50295. PMID: 32391656.
- 33. Sultan J, Griffin SM, Di Franco F, et al. Randomized clinical trial of omega-3 fatty acid-supplemented enteral nutrition versus standard enteral nutrition in patients undergoing oesophagogastric cancer surgery. Br J Surg. 2012 Mar;99(3):346-55. doi: 10.1002/bjs.7799. PMID: 22237467.
- 34. Tsuchiya T, Honda H, Oikawa M, et al. Oral administration of the amino acids cystine and theanine attenuates the adverse events of S-1 adjuvant chemotherapy in gastrointestinal cancer patients. Int J Clin Oncol. 2016 Dec;21(6):1085-90. doi: 10.1007/s10147-016-0996-7. PMID: 27306219.
- 35. Vidal-Casariego A, Calleja-Fernandez A, de Urbina-Gonzalez JJ, et al. Efficacy of glutamine in the prevention of acute radiation enteritis: a randomized controlled trial. JPEN J Parenter Enteral Nutr. 2014 Feb;38(2):205-13. doi: 10.1177/0148607113478191. PMID: 23471208.
- 36. Yeğen SF, Kafadar MT, Gök MA. Comparison of Perioperative Standard and Immunomodulating Enteral Nutrition in Patients Received Major Abdominal Cancer Surgery: a Prospective, Randomized, Controlled Clinical Trial. Indian journal of surgery. 2020;82(5):828-34. doi: 10.1007/s12262-020-02114-0. PMID: CN-02418348.
- Braga M, Gianotti L, Nespoli L, et al. Nutritional approach in malnourished surgical patients: a prospective randomized study. Arch Surg. 2002 Feb;137(2):174-80. doi: 10.1001/archsurg.137.2.174. PMID: 11822956.

- Brown T, Banks M, Hughes BGM, et al. Impact of early prophylactic feeding on long term tube dependency outcomes in patients with head and neck cancer. Oral Oncol. 2017 Sep;72:17-25. doi: 10.1016/j.oraloncology.2017.06.025. PMID: 28797454.
- Ding D, Feng Y, Song B, et al. Effects of preoperative and postoperative enteral nutrition on postoperative nutritional status and immune function of gastric cancer patients. Turk J Gastroenterol. 2015 Mar;26(2):181-5. doi: 10.5152/tjg.2015.3993. PMID: 25835119.
- Falewee MN, Schilf A, Boufflers E, et al. Reduced infections with perioperative immunonutrition in head and neck cancer: exploratory results of a multicenter, prospective, randomized, doubleblind study. Clin Nutr. 2014 Oct;33(5):776-84. doi: 10.1016/j.clnu.2013.10.006. PMID: 24182765.
- Gianotti L, Braga M, Nespoli L, et al. A randomized controlled trial of preoperative oral supplementation with a specialized diet in patients with gastrointestinal cancer. Gastroenterology. 2002 Jun;122(7):1763-70. doi: 10.1053/gast.2002.33587. PMID: 12055582.
- Miyata H, Yano M, Yasuda T, et al. Randomized study of clinical effect of enteral nutrition support during neoadjuvant chemotherapy on chemotherapy-related toxicity in patients with esophageal cancer. Clin Nutr. 2012 Jun;31(3):330-6. doi: 10.1016/j.clnu.2011.11.002. PMID: 22169459.
- Mudge LA, Watson DI, Smithers BM, et al. Multicentre factorial randomized clinical trial of perioperative immunonutrition versus standard nutrition for patients undergoing surgical resection of oesophageal cancer. Br J Surg. 2018 Sep;105(10):1262-72. doi: 10.1002/bjs.10923. PMID: 29999517.
- 44. Wong TX, Wong WX, Chen ST, et al. Effects of Perioperative Oral Nutrition Supplementation in Malaysian Patients Undergoing Elective Surgery for Breast and Colorectal Cancers-A Randomised Controlled Trial. Nutrients. 2022 Jan 30;14(3):30. doi: 10.3390/nu14030615. PMID: 35276977.
- 45. Wu S, You D, Lu L, et al. Effect of enteral nutrition support on the curative effect and immune system in patients with rectal cancer during fast track surgery. International journal of clinical and experimental medicine. 2020;13(8):6065-73. PMID: 2004989500.

- Bozzetti F, Gavazzi C, Miceli R, et al. Perioperative total parenteral nutrition in malnourished, gastrointestinal cancer patients: a randomized, clinical trial. JPEN J Parenter Enteral Nutr. 2000 Jan-Feb;24(1):7-14. doi: 10.1177/014860710002400107. PMID: 10638466.
- Chen H, Pan D, Li L. The effects of multi-oil fat emulsion on older patients with gastric cancer. Biomedical Research (India). 2017;28(1):4270-6. PMID: 616782903.
- 48. Feng J, Xu R, Li K, et al. Effects of preoperative oral carbohydrate administration combined with postoperative early oral intake in elderly patients undergoing hepatectomy with acute-phase inflammation and subjective symptom burden: A prospective randomized controlled study. Asian J Surg. 2022 Jan;45(1):386-95. doi: 10.1016/j.asjsur.2021.06.042. PMID: 34362624.
- Ida S, Hiki N, Cho H, et al. Randomized clinical trial comparing standard diet with perioperative oral immunonutrition in total gastrectomy for gastric cancer. Br J Surg. 2017 Mar;104(4):377-83. doi: 10.1002/bjs.10417. PMID: 28072447.
- 50. Aoyama T, Yoshikawa T, Ida S, et al. Effects of perioperative Eicosapentaenoic acid-enriched oral nutritional supplement on lean body mass after total gastrectomy for gastric cancer. J Cancer. 2019;10(5):1070-6. doi: 10.7150/jca.29632. PMID: 30854113.
- 51. Aoyama T, Yoshikawa T, Ida S, et al. Effects of perioperative eicosapentaenoic acid-enriched oral nutritional supplement on the long-term oncological outcomes after total gastrectomy for gastric cancer. Oncology letters. 2022;23(5). doi: 10.3892/ol.2022.13272. PMID: CN-02385995.
- 52. Kong SH, Lee HJ, Na JR, et al. Effect of perioperative oral nutritional supplementation in malnourished patients who undergo gastrectomy: A prospective randomized trial. Surgery. 2018 Dec;164(6):1263-70. doi: 10.1016/j.surg.2018.05.017. PMID: 30055788.
- 53. Lidder P, Thomas S, Fleming S, et al. A randomized placebo controlled trial of preoperative carbohydrate drinks and early postoperative nutritional supplement drinks in colorectal surgery. Colorectal Dis. 2013 Jun;15(6):737-45. doi: 10.1111/codi.12130. PMID: 23406311.

- 54. Moya P, Miranda E, Soriano-Irigaray L, et al. Perioperative immunonutrition in normonourished patients undergoing laparoscopic colorectal resection. Surg Endosc. 2016 Nov;30(11):4946-53. doi: 10.1007/s00464-016-4836-7. PMID: 26936601.
- 55. Poon RT, Yu WC, Fan ST, et al. Long-term oral branched chain amino acids in patients undergoing chemoembolization for hepatocellular carcinoma: a randomized trial. Aliment Pharmacol Ther. 2004 Apr 1;19(7):779-88. doi: 10.1111/j.1365-2036.2004.01920.x. PMID: 15043519.
- 56. Ritch CR, Cookson MS, Clark PE, et al. Perioperative Oral Nutrition Supplementation Reduces Prevalence of Sarcopenia following Radical Cystectomy: Results of a Prospective Randomized Controlled Trial. J Urol. 2019 Mar;201(3):470-7. doi: 10.1016/j.juro.2018.10.010. PMID: 30359680.
- 57. Sanchez-Guillen L, Soriano-Irigaray L, Lopez-Rodriguez-Arias F, et al. Effect of Early Peripheral Parenteral Nutrition Support in an Enhanced Recovery Program for Colorectal Cancer Surgery: A Randomized Open Trial. J Clin Med. 2021 Aug 18;10(16):18. doi: 10.3390/jcm10163647. PMID: 34441942.
- 58. Lopez-Rodriguez-Arias F, Sanchez-Guillen L, Lillo-Garcia C, et al. Assessment of Body Composition as an Indicator of Early Peripheral Parenteral Nutrition Therapy in Patients Undergoing Colorectal Cancer Surgery in an Enhanced Recovery Program. Nutrients. 2021 Sep 18;13(9):18. doi: 10.3390/nu13093245. PMID: 34579122.
- 59. Sittitrai P, Ruenmarkkaew D, Booyaprapa S, et al. Effect of a perioperative immune-enhancing diet in clean-contaminated head and neck cancer surgery: A randomized controlled trial. Int J Surg. 2021 Sep;93:106051. doi: 10.1016/j.ijsu.2021.106051. PMID: 34371175.
- Wu GH, Liu ZH, Wu ZH, et al. Perioperative artificial nutrition in malnourished gastrointestinal cancer patients. World J Gastroenterol. 2006 Apr 21;12(15):2441-4. doi: 10.3748/wjg.v12.i15.2441. PMID: 16688841.

- Yan X, Liu L, Zhang Y, et al. Perioperative Enteral Nutrition Improves Postoperative Recovery for Patients with Primary Liver Cancer: A Randomized Controlled Clinical Trial. Nutr Cancer. 2021;73(10):1924-32. doi: 10.1080/01635581.2020.1814824. PMID: 32875913.
- 62. Baldwin C, Spiro A, McGough C, et al. Simple nutritional intervention in patients with advanced cancers of the gastrointestinal tract, non-small cell lung cancers or mesothelioma and weight loss receiving chemotherapy: a randomised controlled trial. J Hum Nutr Diet. 2011 Oct;24(5):431-40. doi: 10.1111/j.1365-277X.2011.01189.x. PMID: 21733143.
- Bourdel-Marchasson I, Blanc-Bisson C, Doussau A, et al. Nutritional advice in older patients at risk of malnutrition during treatment for chemotherapy: a two-year randomized controlled trial. PLoS One. 2014;9(9):e108687. doi: 10.1371/journal.pone.0108687. PMID: 25265392.
- 64. Britton B, Baker AL, Wolfenden L, et al. Eating As Treatment (EAT): A Stepped-Wedge, Randomized Controlled Trial of a Health Behavior Change Intervention Provided by Dietitians to Improve Nutrition in Patients With Head and Neck Cancer Undergoing Radiation Therapy (TROG 12.03). Int J Radiat Oncol Biol Phys. 2019 Feb 1;103(2):353-62. doi: 10.1016/j.ijrobp.2018.09.027. PMID: 30296472.
- 65. Forslund M, Ottenblad A, Ginman C, et al. Effects of a nutrition intervention on acute and late bowel symptoms and health-related quality of life up to 24 months post radiotherapy in patients with prostate cancer: a multicentre randomised controlled trial. Support Care Cancer. 2020 Jul;28(7):3331-42. doi: 10.1007/s00520-019-05182-5. PMID: 31758324.
- 66. Isenring EA, Capra S, Bauer JD. Nutrition intervention is beneficial in oncology outpatients receiving radiotherapy to the gastrointestinal or head and neck area. Br J Cancer. 2004 Aug 2;91(3):447-52. doi: 10.1038/sj.bjc.6601962. PMID: 15226773.
- 67. Loser A, Abel J, Kutz LM, et al. Head and neck cancer patients under (chemo-)radiotherapy undergoing nutritional intervention: Results from the prospective randomized HEADNUT-trial. Radiother Oncol. 2021 Jun;159:82-90. doi: 10.1016/j.radonc.2021.03.019. PMID: 33766702.

- 68. Movahed S, Seilanian Toussi M, Pahlavani N, et al. Effects of medical nutrition therapy compared with general nutritional advice on nutritional status and nutrition-related complications in esophageal cancer patients receiving concurrent chemoradiation: A randomized controlled trial. Mediterranean Journal of Nutrition and Metabolism. 2020;13(3):265-76. doi: 10.3233/mnm-200424. PMID: 632903213.
- Orell H, Schwab U, Saarilahti K, et al. Nutritional Counseling for Head and Neck Cancer Patients Undergoing (Chemo) Radiotherapy-A Prospective Randomized Trial. Front Nutr. 2019;6:22. doi: 10.3389/fnut.2019.00022. PMID: 30937304.
- Pettersson A, Johansson B, Persson C, et al. Effects of a dietary intervention on acute gastrointestinal side effects and other aspects of health-related quality of life: a randomized controlled trial in prostate cancer patients undergoing radiotherapy. Radiother Oncol. 2012 Jun;103(3):333-40. doi: 10.1016/j.radonc.2012.04.006. PMID: 22633817.
- Qiu Y, You J, Wang K, et al. Effect of whole-course nutrition management on patients with esophageal cancer undergoing concurrent chemoradiotherapy: A randomized control trial. Nutrition. 2020 Jan;69:110558. doi: 10.1016/j.nut.2019.110558. PMID: 31526964.
- Ravasco P, Monteiro-Grillo I, Camilo M. Individualized nutrition intervention is of major benefit to colorectal cancer patients: long-term follow-up of a randomized controlled trial of nutritional therapy. Am J Clin Nutr. 2012 Dec;96(6):1346-53. doi: 10.3945/ajcn.111.018838. PMID: 23134880.
- Regueme SC, Echeverria I, Moneger N, et al. Protein intake, weight loss, dietary intervention, and worsening of quality of life in older patients during chemotherapy for cancer. Support Care Cancer. 2021 Feb;29(2):687-96. doi: 10.1007/s00520-020-05528-4. PMID: 32435967.
- 74. Tu MY, Chien TW, Lin HP, et al. Effects of an intervention on nutrition consultation for cancer patients. Eur J Cancer Care (Engl). 2013 May;22(3):370-6. doi: 10.1111/ecc.12040. PMID: 23320428.

- 75. Um MH, Choi MY, Lee SM, et al. Intensive nutritional counseling improves PG-SGA scores and nutritional symptoms during and after radiotherapy in Korean cancer patients. Support Care Cancer. 2014 Nov;22(11):2997-3005. doi: 10.1007/s00520-014-2304-2. PMID: 24906838.
- 76. van der Werf A, Langius JAE, Beeker A, et al. The effect of nutritional counseling on muscle mass and treatment outcome in patients with metastatic colorectal cancer undergoing chemotherapy: A randomized controlled trial. Clin Nutr. 2020 Oct;39(10):3005-13. doi: 10.1016/j.clnu.2020.01.009. PMID: 32037284.
- 77. Zhang Z, Zhu Y, Zhang L, et al. Nutritional education and counseling program for adult cancer patients during radiotherapy: a cluster-randomized clinical trial. Support Care Cancer. 2022 Apr;30(4):3279-89. doi: 10.1007/s00520-021-06704-w. PMID: 34984549.
- Cheng M, Zhang S, Ning C, et al. Omega-3 Fatty Acids Supplementation Improve Nutritional Status and Inflammatory Response in Patients With Lung Cancer: A Randomized Clinical Trial. Front Nutr. 2021;8:686752. doi: 10.3389/fnut.2021.686752. PMID: 34395492.
- Chitapanarux I, Traisathit P, Chitapanarux T, et al. Arginine, glutamine, and fish oil supplementation in cancer patients treated with concurrent chemoradiotherapy: A randomized control study. Curr Probl Cancer. 2020 Feb;44(1):100482. doi: 10.1016/j.currproblcancer.2019.05.005. PMID: 31146957.
- 80. da Gama Torres HO, Vilela EG, da Cunha AS, et al. Efficacy of glutamine-supplemented parenteral nutrition on short-term survival following allo-SCT: a randomized study. Bone Marrow Transplant. 2008 Jun;41(12):1021-7. doi: 10.1038/bmt.2008.27. PMID: 18317456.
- 81. de Luis DA, Izaola O, Cuellar L, et al. Randomized clinical trial with an enteral arginine-enhanced formula in early postsurgical head and neck cancer patients. Eur J Clin Nutr. 2004 Nov;58(11):1505-8. doi: 10.1038/sj.ejcn.1601999. PMID: 15138461.
- de Luis DA, Izaola O, Aller R, et al. A randomized clinical trial with oral Immunonutrition (omega3-enhanced formula vs. arginineenhanced formula) in ambulatory head and neck cancer patients. Ann Nutr Metab. 2005 Mar-Apr;49(2):95-9. doi: 10.1159/000084742. PMID: 15802904.

- 83. de Luis DA, Izaola O, Cuellar L, et al. Clinical and biochemical outcomes after a randomized trial with a high dose of enteral arginine formula in postsurgical head and neck cancer patients. Eur J Clin Nutr. 2007 Feb;61(2):200-4. doi: 10.1038/sj.ejcn.1602515. PMID: 16929239.
- 84. de Luis DA, Izaola O, Aller R, et al. A randomized clinical trial with two omega 3 fatty acid enhanced oral supplements in head and neck cancer ambulatory patients. Eur Rev Med Pharmacol Sci. 2008 May-Jun;12(3):177-81. PMID: 18700689.
- 85. De Luis DA, Izaola O, Cuellar L, et al. High dose of arginine enhanced enteral nutrition in postsurgical head and neck cancer patients. A randomized clinical trial. Eur Rev Med Pharmacol Sci. 2009 Jul-Aug;13(4):279-83. PMID: 19694342.
- 86. De Luis DA, Izaola O, Cuellar L, et al. A randomized double-blind clinical trial with two different doses of arginine enhanced enteral nutrition in postsurgical cancer patients. Eur Rev Med Pharmacol Sci. 2010 Nov;14(11):941-5. PMID: 21284343.
- 87. De Luis DA, Izaola O, Terroba MC, et al. Effect of three different doses of arginine enhanced enteral nutrition on nutritional status and outcomes in well nourished postsurgical cancer patients: a randomized single blinded prospective trial. Eur Rev Med Pharmacol Sci. 2015;19(6):950-5. PMID: 25855918.
- Farreras N, Artigas V, Cardona D, et al. Effect of early postoperative enteral immunonutrition on wound healing in patients undergoing surgery for gastric cancer. Clin Nutr. 2005 Feb;24(1):55-65. doi: 10.1016/j.clnu.2004.07.002. PMID: 15681102.
- Fietkau R, Lewitzki V, Kuhnt T, et al. A diseasespecific enteral nutrition formula improves nutritional status and functional performance in patients with head and neck and esophageal cancer undergoing chemoradiotherapy: results of a randomized, controlled, multicenter trial. Cancer. 2013 Sep 15;119(18):3343-53. doi: 10.1002/cncr.28197. PMID: 23765693.

- 90. Golkhalkhali B, Rajandram R, Paliany AS, et al. Strain-specific probiotic (microbial cell preparation) and omega-3 fatty acid in modulating quality of life and inflammatory markers in colorectal cancer patients: a randomized controlled trial. Asia Pac J Clin Oncol. 2018 Jun;14(3):179-91. doi: 10.1111/ajco.12758. PMID: 28857425.
- 91. Iwase S, Kawaguchi T, Yotsumoto D, et al. Efficacy and safety of an amino acid jelly containing coenzyme Q10 and L-carnitine in controlling fatigue in breast cancer patients receiving chemotherapy: a multi-institutional, randomized, exploratory trial (JORTC-CAM01). Support Care Cancer. 2016 Feb;24(2):637-46. doi: 10.1007/s00520-015-2824-4. PMID: 26105516.
- 92. Jiang ZM, Wilmore DW, Wang XR, et al. Randomized clinical trial of intravenous soybean oil alone versus soybean oil plus fish oil emulsion after gastrointestinal cancer surgery. Br J Surg. 2010 Jun;97(6):804-9. doi: 10.1002/bjs.6999. PMID: 20473991.
- Klek S, Kulig J, Szczepanik AM, et al. The clinical value of parenteral immunonutrition in surgical patients. Acta Chir Belg. 2005 Apr;105(2):175-9. PMID: 15906909.
- 94. Lobo DN, Williams RN, Welch NT, et al. Early postoperative jejunostomy feeding with an immune modulating diet in patients undergoing resectional surgery for upper gastrointestinal cancer: a prospective, randomized, controlled, double-blind study. Clin Nutr. 2006 Oct;25(5):716-26. doi: 10.1016/j.clnu.2006.04.007. PMID: 16777271.
- 95. Adiamah A, Rollins KE, Kapeleris A, et al. Postoperative arginine-enriched immune modulating nutrition: Long-term survival results from a randomised clinical trial in patients with oesophagogastric and pancreaticobiliary cancer. Clin Nutr. 2021 Nov;40(11):5482-5. doi: 10.1016/j.clnu.2021.09.040. PMID: 34656029.
- 96. Lu CY, Shih YL, Sun LC, et al. The inflammatory modulation effect of glutamine-enriched total parenteral nutrition in postoperative gastrointestinal cancer patients. Am Surg. 2011 Jan;77(1):59-64. PMID: 21396307.
- 97. Matsuda Y, Habu D, Lee S, et al. Enteral Diet Enriched with omega-3 Fatty Acid Improves Oxygenation After Thoracic Esophagectomy for Cancer: A Randomized Controlled Trial. World J Surg. 2017 Jun;41(6):1584-94. doi: 10.1007/s00268-017-3893-y. PMID: 28138734.

- 98. Miyata H, Yano M, Yasuda T, et al. Randomized study of the clinical effects of omega-3 fatty acid-containing enteral nutrition support during neoadjuvant chemotherapy on chemotherapyrelated toxicity in patients with esophageal cancer. Nutrition. 2017 Jan;33:204-10. doi: 10.1016/j.nut.2016.07.004. PMID: 27644137.
- Pathak S, Soni TP, Sharma LM, et al. A Randomized Controlled Trial to Evaluate the Role and Efficacy of Oral Glutamine in the Treatment of Chemo-radiotherapy-induced Oral Mucositis and Dysphagia in Patients with Oropharynx and Larynx Carcinoma. Cureus. 2019 Jun 7;11(6):e4855. doi: 10.7759/cureus.4855. PMID: 31410338.
- 100. Pottel L, Lycke M, Boterberg T, et al. Echium oil is not protective against weight loss in head and neck cancer patients undergoing curative radio(chemo)therapy: a randomised-controlled trial. BMC Complement Altern Med. 2014 Oct 7;14:382. doi: 10.1186/1472-6882-14-382. PMID: 25293388.
- 101. Sun LC, Shih YL, Lu CY, et al. Randomized, controlled study of branched chain amino acidenriched total parenteral nutrition in malnourished patients with gastrointestinal cancer undergoing surgery. Am Surg. 2008 Mar;74(3):237-42. PMID: 18376691.
- 102. Takeshita S, Ichikawa T, Nakao K, et al. A snack enriched with oral branched-chain amino acids prevents a fall in albumin in patients with liver cirrhosis undergoing chemoembolization for hepatocellular carcinoma. Nutr Res. 2009 Feb;29(2):89-93. doi: 10.1016/j.nutres.2008.12.005. PMID: 19285598.
- 103. Tanca FM, Madeddu C, Macciò A, et al. New perspective on the nutritional approach to cancer-related anorexia/cachexia: preliminary results of a randomised phase III clinical trial with five different arms of treatment. Mediterranean Journal of Nutrition and Metabolism. 2009;2(1):29-36. doi: 10.1007/s12349-009-0041-y. PMID: 358288574.
- 104. Tumas J, Tumiene B, Jurkeviciene J, et al. Nutritional and immune impairments and their effects on outcomes in early pancreatic cancer patients undergoing pancreatoduodenectomy. Clin Nutr. 2020 Nov;39(11):3385-94. doi: 10.1016/j.clnu.2020.02.029. PMID: 32184025.

- 105. Wang J, Li Y, Qi Y. Effect of glutamine-enriched nutritional support on intestinal mucosal barrier function, MMP-2, MMP-9 and immune function in patients with advanced gastric cancer during perioperative chemotherapy. Oncol Lett. 2017 Sep;14(3):3606-10. doi: 10.3892/ol.2017.6612. PMID: 28927119.
- 106. Wang WP, Yan XL, Ni YF, et al. Effects of lipid emulsions in parenteral nutrition of esophageal cancer surgical patients receiving enteral nutrition: a comparative analysis. Nutrients. 2013 Dec 27;6(1):111-23. doi: 10.3390/nu6010111. PMID: 24379010.
- 107. Wang X, Pan L, Zhang P, et al. Enteral nutrition improves clinical outcome and shortens hospital stay after cancer surgery. J Invest Surg. 2010 Dec;23(6):309-13. doi: 10.3109/08941939.2010.519428. PMID: 21208095.
- 108. Wu Z, Qin J, Pu L. Omega-3 fatty acid improves the clinical outcome of hepatectomized patients with hepatitis B virus (HBV)-associated hepatocellular carcinoma. J Biomed Res. 2012 Nov;26(6):395-9. doi: 10.7555/JBR.26.20120058. PMID: 23554777.
- 109. Yang J, Zhang X, Li K, et al. Effects of EN combined with PN enriched with n-3 polyunsaturated fatty acids on immune related indicators and early rehabilitation of patients with gastric cancer: A randomized controlled trial. Clin Nutr. 2022 Apr 6;41(6):1163-70. doi: 10.1016/j.clnu.2022.03.018. PMID: 35500316.
- 110. Yeh KY, Wang HM, Chang JW, et al. Omega-3 fatty acid-, micronutrient-, and probiotic-enriched nutrition helps body weight stabilization in head and neck cancer cachexia. Oral Surg Oral Med Oral Pathol Oral Radiol. 2013 Jul;116(1):41-8. doi: 10.1016/j.oooo.2013.01.015. PMID: 23562359.
- 111. Zhang B, Wei G, Li R, et al. n-3 fatty acid-based parenteral nutrition improves postoperative recovery for cirrhotic patients with liver cancer: A randomized controlled clinical trial. Clin Nutr. 2017 Oct;36(5):1239-44. doi: 10.1016/j.clnu.2016.08.002. PMID: 27614675.
- 112. Zhu MW, Tang DN, Hou J, et al. Impact of fish oil enriched total parenteral nutrition on elderly patients after colorectal cancer surgery. Chin Med J (Engl). 2012 Jan;125(2):178-81. doi: 10.3760/cma.j.issn.0366-6999.2012.02.003. PMID: 22340541.

- 113. Gardner A, Mattiuzzi G, Faderl S, et al. Randomized comparison of cooked and noncooked diets in patients undergoing remission induction therapy for acute myeloid leukemia. J Clin Oncol. 2008 Dec 10;26(35):5684-8. doi: 10.1200/JCO.2008.16.4681. PMID: 18955453.
- 114. Ingersoll GL, Wasilewski A, Haller M, et al. Effect of concord grape juice on chemotherapy-induced nausea and vomiting: results of a pilot study. Oncol Nurs Forum. 2010 Mar;37(2):213-21. doi: 10.1188/10.ONF.213-221. PMID: 20189927.
- 115. Jatoi A, Qin R, Satele D, et al. "Enjoy glass of wine before eating:" a randomized trial to test the orexigenic effects of this advice in advanced cancer patients. Support Care Cancer. 2016 Sep;24(9):3739-46. doi: 10.1007/s00520-016-3190-6. PMID: 27039205.
- 116. Khodabakhshi A, Akbari ME, Mirzaei HR, et al. Feasibility, Safety, and Beneficial Effects of MCT-Based Ketogenic Diet for Breast Cancer Treatment: A Randomized Controlled Trial Study. Nutr Cancer. 2020;72(4):627-34. doi: 10.1080/01635581.2019.1650942. PMID: 31496287.
- 117. Khodabakhshi A, Seyfried TN, Kalamian M, et al. Does a ketogenic diet have beneficial effects on quality of life, physical activity or biomarkers in patients with breast cancer: a randomized controlled clinical trial. Nutr J. 2020 Aug 22;19(1):87. doi: 10.1186/s12937-020-00596-y. PMID: 32828130.
- 118. Lugtenberg RT, de Groot S, Kaptein AA, et al. Quality of life and illness perceptions in patients with breast cancer using a fasting mimicking diet as an adjunct to neoadjuvant chemotherapy in the phase 2 DIRECT (BOOG 2013-14) trial. Breast Cancer Res Treat. 2021 Feb;185(3):741-58. doi: 10.1007/s10549-020-05991-x. PMID: 33179154.
- 119. Miyakawa A, Kodera S, Sakuma Y, et al. Effects of Early Initiation of Solid Versus Liquid Diet after Endoscopic Submucosal Dissection on Quality of Life and Postoperative Outcomes: A Prospective Pilot Randomized Controlled Trial. Digestion. 2019;100(3):160-9. doi: 10.1159/000494490. PMID: 30554216.
- 120. Voss M, Wagner M, von Mettenheim N, et al. ERGO2: A Prospective, Randomized Trial of Calorie-Restricted Ketogenic Diet and Fasting in Addition to Reirradiation for Malignant Glioma. Int J Radiat Oncol Biol Phys. 2020 Nov 15;108(4):987-95. doi: 10.1016/j.ijrobp.2020.06.021. PMID: 32619561.

- 121. Wedlake LJ, McGough C, Shaw C, et al. Clinical trial: Efficacy of a low or modified fat diet for the prevention of gastrointestinal toxicity in patients receiving radiotherapy treatment for pelvic malignancies. J Hum Nutr Diet. 2012 Jun;25(3):247-59. doi: 10.1111/j.1365-277X.2012.01248.x. PMID: 22515941.
- 122. Berkelmans GHK, Fransen LFC, Dolmans-Zwartjes ACP, et al. Direct Oral Feeding Following Minimally Invasive Esophagectomy (NUTRIENT II trial): An International, Multicenter, Open-label Randomized Controlled Trial. Ann Surg. 2020 Jan;271(1):41-7. doi: 10.1097/SLA.00000000003278. PMID: 31090563.
- 123. Boelens PG, Heesakkers FF, Luyer MD, et al. Reduction of postoperative ileus by early enteral nutrition in patients undergoing major rectal surgery: prospective, randomized, controlled trial. Ann Surg. 2014 Apr;259(4):649-55. doi: 10.1097/SLA.000000000000288. PMID: 24169163.
- 124. Bozzetti F, Braga M, Gianotti L, et al. Postoperative enteral versus parenteral nutrition in malnourished patients with gastrointestinal cancer: a randomised multicentre trial. The Lancet. 2001;358(9292):1487-92.
- 125. Braga M, Gianotti L, Gentilini O, et al. Early postoperative enteral nutrition improves gut oxygenation and reduces costs compared with total parenteral nutrition. Critical care medicine. 2001;29(2):242-8.
- 126. Dag A, Colak T, Turkmenoglu O, et al. A randomized controlled trial evaluating early versus traditional oral feeding after colorectal surgery. Clinics (Sao Paulo). 2011;66(12):2001-5. doi: 10.1590/s1807-59322011001200001. PMID: 22189721.
- 127. Feo CV, Romanini B, Sortini D, et al. Early oral feeding after colorectal resection: a randomized controlled study. ANZ J Surg. 2004 May;74(5):298-301. doi: 10.1111/j.1445-1433.2004.02985.x. PMID: 15144242.
- 128. Gao L, Zhao Z, Zhang L, et al. Effect of early oral feeding on gastrointestinal function recovery in postoperative gastric cancer patients: a prospective study. J BUON. 2019 Jan-Feb;24(1):194-200. PMID: 30941970.

- 129. Huang D, Sun Z, Huang J, et al. Early enteral nutrition in combination with parenteral nutrition in elderly patients after surgery due to gastrointestinal cancer. Int J Clin Exp Med. 2015;8(8):13937-45. PMID: 26550350.
- 130. Hyltander A, Bosaeus I, Svedlund J, et al. Supportive nutrition on recovery of metabolism, nutritional state, health-related quality of life, and exercise capacity after major surgery: a randomized study. Clin Gastroenterol Hepatol. 2005 May;3(5):466-74. doi: 10.1016/s1542-3565(05)00151-5. PMID: 15880316.
- 131. Kita R, Miyata H, Sugimura K, et al. Clinical effect of enteral nutrition support during neoadjuvant chemotherapy on the preservation of skeletal muscle mass in patients with esophageal cancer. Clin Nutr. 2021 Jun;40(6):4380-5. doi: 10.1016/j.clnu.2021.01.007. PMID: 33526287.
- 132. Kurbanalievich SD, Vladimirovich DV, Kabildina NA. Nutritional Support for Patients with Diseases of Hepatopancreotoduodenal Zone in the Early After the Operational Period. Open Access Macedonian Journal of Medical Sciences. 2020;8(B):769-74. doi: 10.3889/oamjms.2020.4717. PMID: 2005564409.
- 133. Li B, Liu HY, Guo SH, et al. The postoperative clinical outcomes and safety of early enteral nutrition in operated gastric cancer patients. J BUON. 2015 Mar-Apr;20(2):468-72. PMID: 26011337.
- 134. Liu C, Du Z, Lou C, et al. Enteral nutrition is superior to total parenteral nutrition for pancreatic cancer patients who underwent pancreaticoduodenectomy. Asia Pac J Clin Nutr. 2011;20(2):154-60. PMID: 21669582.
- 135. Luo Z, Wang J, Zhang Z, et al. Efficacy of Early Enteral Immunonutrition on Immune Function and Clinical Outcome for Postoperative Patients With Gastrointestinal Cancer. JPEN J Parenter Enteral Nutr. 2018 May;42(4):758-65. doi: 10.1177/0148607117715439. PMID: 28666095.
- 136. Ma BQ, Chen SY, Jiang ZB, et al. Effect of postoperative early enteral nutrition on clinical outcomes and immune function of cholangiocarcinoma patients with malignant obstructive jaundice. World J Gastroenterol. 2020 Dec 14;26(46):7405-15. doi: 10.3748/wjg.v26.i46.7405. PMID: 33362392.

- 137. Mahmoodzadeh H, Shoar S, Sirati F, et al. Early initiation of oral feeding following upper gastrointestinal tumor surgery: a randomized controlled trial. Surg Today. 2015 Feb;45(2):203-8. doi: 10.1007/s00595-014-0937x. PMID: 24875466.
- 138. Minig L, Biffi R, Zanagnolo V, et al. Reduction of postoperative complication rate with the use of early oral feeding in gynecologic oncologic patients undergoing a major surgery: a randomized controlled trial. Ann Surg Oncol. 2009 Nov;16(11):3101-10. doi: 10.1245/s10434-009-0681-4. PMID: 19760046.
- Perinel J, Mariette C, Dousset B, et al. Early Enteral Versus Total Parenteral Nutrition in Patients Undergoing Pancreaticoduodenectomy: A Randomized Multicenter Controlled Trial (Nutri-DPC). Ann Surg. 2016 Nov;264(5):731-7. doi: 10.1097/SLA.000000000001896. PMID: 27429039.
- 140. Roberts S, Miller J, Pineiro L, et al. Total parenteral nutrition vs oral diet in autologous hematopoietic cell transplant recipients. Bone Marrow Transplant. 2003 Oct;32(7):715-21. doi: 10.1038/sj.bmt.1704204. PMID: 13130320.
- 141. Ryu J, Nam BH, Jung YS. Clinical outcomes comparing parenteral and nasogastric tube nutrition after laryngeal and pharyngeal cancer surgery. Dysphagia. 2009 Dec;24(4):378-86. doi: 10.1007/s00455-009-9213-4. PMID: 19255706.
- 142. Sadasivan A, Faizal B, Kumar M. Nasogastric and percutaneous endoscopic gastrostomy tube use in advanced head and neck cancer patients: a comparative study. J Pain Palliat Care Pharmacother. 2012 Sep;26(3):226-32. doi: 10.3109/15360288.2012.702199. PMID: 22973911.
- 143. Seven H, Calis AB, Turgut S. A randomized controlled trial of early oral feeding in laryngectomized patients. Laryngoscope. 2003 Jun;113(6):1076-9. doi: 10.1097/00005537-200306000-00030. PMID: 12782826.
- 144. Sousa AA, Porcaro-Salles JM, Soares JM, et al. Does early oral feeding increase the likelihood of salivary fistula after total laryngectomy? J Laryngol Otol. 2014 Apr 15;128(4):1-7. doi: 10.1017/S0022215114000747. PMID: 24736040.

- 145. Sun HB, Li Y, Liu XB, et al. Early Oral Feeding Following McKeown Minimally Invasive Esophagectomy: An Open-label, Randomized, Controlled, Noninferiority Trial. Ann Surg. 2018 Mar;267(3):435-42. doi: 10.1097/SLA.00000000002304. PMID: 28549015.
- 146. Tao Z, Zhang Y, Zhu S, et al. A Prospective Randomized Trial Comparing Jejunostomy and Nasogastric Feeding in Minimally Invasive McKeown Esophagectomy. J Gastrointest Surg. 2020 Oct;24(10):2187-96. doi: 10.1007/s11605-019-04390-y. PMID: 31512101.
- 147. van Barneveld KW, Smeets BJ, Heesakkers FF, et al. Beneficial Effects of Early Enteral Nutrition After Major Rectal Surgery: A Possible Role for Conditionally Essential Amino Acids? Results of a Randomized Clinical Trial. Crit Care Med. 2016 Jun;44(6):e353-61. doi: 10.1097/CCM.00000000001640. PMID: 26937858.
- 148. Wang Q, Yang KL, Guo BY, et al. Safety of early oral feeding after total laparoscopic radical gastrectomy for gastric cancer (SOFTLY-1): a single-center randomized controlled trial. Cancer Manag Res. 2019;11:4839-46. doi: 10.2147/CMAR.S199552. PMID: 31239762.
- 149. Wang J, Zhao J, Zhang Y, et al. Early enteral nutrition and total parenteral nutrition on the nutritional status and blood glucose in patients with gastric cancer complicated with diabetes mellitus after radical gastrectomy. Exp Ther Med. 2018 Jul;16(1):321-7. doi: 10.3892/etm.2018.6168. PMID: 29896256.
- Xiao-Bo Y, Qiang L, Xiong Q, et al. Efficacy of early postoperative enteral nutrition in supporting patients after esophagectomy. Minerva Chir. 2014 Feb;69(1):37-46. PMID: 24504222.
- 151. Xu B, Ma J, Chen X, et al. The effect of early enteral nutrition on the postoperative immune function and inflammatory indexes in patients with digestive tract cancers. International journal of clinical and experimental medicine. 2020;13(4):2541-7. PMID: 2004238521.
- 152. Zhang J, Si X, Li W, et al. Effect of peripherally inserted central catheter (PICC) parenteral nutrition on immune function and nutritional support after radical gastrectomy for gastric cancer. Pak J Pharm Sci. 2019 May;32(3 Special):1441-5. PMID: 31551229.

- 153. Akita H, Takahashi H, Asukai K, et al. The utility of nutritional supportive care with an eicosapentaenoic acid (EPA)-enriched nutrition agent during pre-operative chemoradiotherapy for pancreatic cancer: Prospective randomized control study. Clin Nutr ESPEN. 2019 Oct;33:148-53. doi: 10.1016/j.clnesp.2019.06.003. PMID: 31451252.
- 154. Baker J, Janda M, Graves N, et al. Quality of life after early enteral feeding versus standard care for proven or suspected advanced epithelial ovarian cancer: Results from a randomised trial. Gynecol Oncol. 2015 Jun;137(3):516-22. doi: 10.1016/j.ygyno.2015.03.048. PMID: 25827292.
- 155. Barlow R, Price P, Reid TD, et al. Prospective multicentre randomised controlled trial of early enteral nutrition for patients undergoing major upper gastrointestinal surgical resection. Clin Nutr. 2011 Oct;30(5):560-6. doi: 10.1016/j.clnu.2011.02.006. PMID: 21601319.
- 156. Bouleuc C, Anota A, Cornet C, et al. Impact on Health-Related Quality of Life of Parenteral Nutrition for Patients with Advanced Cancer Cachexia: Results from a Randomized Controlled Trial. Oncologist. 2020 May;25(5):e843-e51. doi: 10.1634/theoncologist.2019-0856. PMID: 32212354.
- 157. Cereda E, Cappello S, Colombo S, et al. Nutritional counseling with or without systematic use of oral nutritional supplements in head and neck cancer patients undergoing radiotherapy. Radiother Oncol. 2018 Jan;126(1):81-8. doi: 10.1016/j.radonc.2017.10.015. PMID: 29111172.
- 158. Chen X, Zhao G, Zhu L. Home enteral nutrition for postoperative elderly patients with esophageal cancer. Ann Palliat Med. 2021 Jan;10(1):278-84. doi: 10.21037/apm-20-2197. PMID: 33545764.
- 159. Chen T, Jiang W, He G. Effect of family enteral nutrition on nutritional status in elderly patients with esophageal carcinoma after minimally invasive radical surgery: a randomized trial. Ann Palliat Med. 2021 Jun;10(6):6760-7. doi: 10.21037/apm-21-1219. PMID: 34237976.
- 160. Chu L, Ren Y, Zhang L, et al. Evaluation of effects of nutritional risk assessment and enteral and parenteral nutritional interventions after esophageal cancer surgery. International journal of clinical and experimental medicine. 2018;11(5):5110-6. PMID: 622367981.

- 161. Deibert CM, Silva MV, RoyChoudhury A, et al. A Prospective Randomized Trial of the Effects of Early Enteral Feeding After Radical Cystectomy. Urology. 2016 Oct;96:69-73. doi: 10.1016/j.urology.2016.06.045. PMID: 27402372.
- 162. Faccio AA, Mattos C, Santos E, et al. Oral Nutritional Supplementation in Cancer Patients Who Were Receiving Chemo/Chemoradiation Therapy: A Multicenter, Randomized Phase II Study. Nutr Cancer. 2021;73(3):442-9. doi: 10.1080/01635581.2020.1758170. PMID: 32363940.
- 163. Gavazzi C, Colatruglio S, Valoriani F, et al. Impact of home enteral nutrition in malnourished patients with upper gastrointestinal cancer: A multicentre randomised clinical trial. Eur J Cancer. 2016 Sep;64:107-12. doi: 10.1016/j.ejca.2016.05.032. PMID: 27391922.
- 164. Huang S, Piao Y, Cao C, et al. A prospective randomized controlled trial on the value of prophylactic oral nutritional supplementation in locally advanced nasopharyngeal carcinoma patients receiving chemo-radiotherapy. Oral Oncol. 2020 Dec;111:105025. doi: 10.1016/j.oraloncology.2020.105025. PMID: 33032180.
- 165. Imamura H, Nishikawa K, Kishi K, et al. Effects of an Oral Elemental Nutritional Supplement on Post-gastrectomy Body Weight Loss in Gastric Cancer Patients: A Randomized Controlled Clinical Trial. Ann Surg Oncol. 2016 Sep;23(9):2928-35. doi: 10.1245/s10434-016-5221-4. PMID: 27084538.
- 166. Kimura Y, Nishikawa K, Kishi K, et al. Long-term effects of an oral elemental nutritional supplement on post-gastrectomy body weight loss in gastric cancer patients (KSES002). Ann Gastroenterol Surg. 2019 Nov;3(6):648-56. doi: 10.1002/ags3.12290. PMID: 31788653.
- 167. Jiang W, Ding H, Li W, et al. Benefits of Oral Nutritional Supplements in Patients with Locally Advanced Nasopharyngeal Cancer during Concurrent Chemoradiotherapy: An Exploratory Prospective Randomized Trial. Nutr Cancer. 2018 Nov-Dec;70(8):1299-307. doi: 10.1080/01635581.2018.1557222. PMID: 30633580.
- 168. Jin Y, Yong C, Ren K, et al. Effects of Post-Surgical Parenteral Nutrition on Patients with Gastric Cancer. Cell Physiol Biochem. 2018;49(4):1320-8. doi: 10.1159/000493410. PMID: 30205371.

- 169. Kanat O, Cubukcu E, Avci N, et al. Comparison of three different treatment modalities in the management of cancer cachexia. Tumori. 2013 Mar-Apr;99(2):229-33. doi: 10.1700/1283.14197. PMID: 23748819.
- 170. Katada C, Fukazawa S, Sugawara M, et al. Randomized study of prevention of gastrointestinal toxicities by nutritional support using an amino acid-rich elemental diet during chemotherapy in patients with esophageal cancer (KDOG 1101). Esophagus. 2021 Apr;18(2):296-305. doi: 10.1007/s10388-020-00787-w. PMID: 33009977.
- 171. Klek S, Kulig J, Sierzega M, et al. Standard and immunomodulating enteral nutrition in patients after extended gastrointestinal surgery--a prospective, randomized, controlled clinical trial. Clin Nutr. 2008 Aug;27(4):504-12. doi: 10.1016/j.clnu.2008.04.010. PMID: 18571296.
- 172. Klek S, Sierzega M, Szybinski P, et al. The immunomodulating enteral nutrition in malnourished surgical patients - a prospective, randomized, double-blind clinical trial. Clin Nutr. 2011 Jun;30(3):282-8. doi: 10.1016/j.clnu.2010.10.001. PMID: 21074910.
- 173. Klek S, Scislo L, Walewska E, et al. Enriched enteral nutrition may improve short-term survival in stage IV gastric cancer patients: A randomized, controlled trial. Nutrition. 2017 Apr;36:46-53. doi: 10.1016/j.nut.2016.03.016. PMID: 28336107.
- 174. Klek S, Sierzega M, Szybinski P, et al. Perioperative nutrition in malnourished surgical cancer patients a prospective, randomized, controlled clinical trial. Clin Nutr. 2011 Dec;30(6):708-13. doi: 10.1016/j.clnu.2011.07.007. PMID: 21820770.
- 175. Klek S, Szybinski P, Szczepanek K. Perioperative immunonutrition in surgical cancer patients: a summary of a decade of research. World J Surg. 2014 Apr;38(4):803-12. doi: 10.1007/s00268-013-2323-z. PMID: 24178185.
- 176. Li B, Liu HY, Guo SH, et al. Impact of early postoperative enteral nutrition on clinical outcomes in patients with gastric cancer. Genet Mol Res. 2015 Jun 29;14(2):7136-41. doi: 10.4238/2015.June.29.7. PMID: 26125924.
- 177. Li C, Ni L, Liu C. Early enteral immunonutrition support protects the cellular and humoral immune functions of patients with pancreatic cancer after chemotherapy. International journal of clinical and experimental medicine. 2020;13(2):700-8. PMID: 2003914110.

- 178. Lyu J, Shi A, Li T, et al. Effects of Enteral Nutrition on Patients With Oesophageal Carcinoma Treated With Concurrent Chemoradiotherapy: A Prospective, Multicentre, Randomised, Controlled Study. Front Oncol. 2022;12:839516. doi: 10.3389/fonc.2022.839516. PMID: 35280748.
- 179. McGough C, Wedlake L, Baldwin C, et al. Clinical trial: normal diet vs. partial replacement with oral E028 formula for the prevention of gastrointestinal toxicity in cancer patients undergoing pelvic radiotherapy. Aliment Pharmacol Ther. 2008 Jun 1;27(11):1132-9. doi: 10.1111/j.1365-2036.2008.03665.x. PMID: 18315590.
- 180. Meng Q, Tan S, Jiang Y, et al. Post-discharge oral nutritional supplements with dietary advice in patients at nutritional risk after surgery for gastric cancer: A randomized clinical trial. Clin Nutr. 2021 Jan;40(1):40-6. doi: 10.1016/j.clnu.2020.04.043. PMID: 32563598.
- 181. Tan S, Meng Q, Jiang Y, et al. Impact of oral nutritional supplements in post-discharge patients at nutritional risk following colorectal cancer surgery: A randomised clinical trial. Clin Nutr. 2021 Jan;40(1):47-53. doi: 10.1016/j.clnu.2020.05.038. PMID: 32563599.
- 182. Miyazaki Y, Omori T, Fujitani K, et al. Oral nutritional supplements versus a regular diet alone for body weight loss after gastrectomy: a phase 3, multicenter, open-label randomized controlled trial. Gastric Cancer. 2021 Sep;24(5):1150-9. doi: 10.1007/s10120-021-01188-3. PMID: 33835329.
- 183. Nie J, Su X, Wei L, et al. Early enteral nutrition support for colon carcinoma patients can improve immune function and promote physical recovery. Am J Transl Res. 2021;13(12):14102-8. PMID: 35035754.
- 184. Ohkura Y, Ueno M, Shindoh J, et al. Randomized controlled trial on efficacy of oligomeric formula (HINE E-GEL(R)) versus polymeric formula (MEIN(R)) enteral nutrition after esophagectomy for esophageal cancer with gastric tube reconstruction. Dis Esophagus. 2019 May 1;32(5):01. doi: 10.1093/dote/doy084. PMID: 30169605.

- 185. Okabayashi T, Iyoki M, Sugimoto T, et al. Oral supplementation with carbohydrate- and branched-chain amino acid-enriched nutrients improves postoperative quality of life in patients undergoing hepatic resection. Amino Acids. 2011 Apr;40(4):1213-20. doi: 10.1007/s00726-010-0748-3. PMID: 20852905.
- 186. Ravasco P, Monteiro-Grillo I, Marques Vidal P, et al. Impact of nutrition on outcome: a prospective randomized controlled trial in patients with head and neck cancer undergoing radiotherapy. Head Neck. 2005 Aug;27(8):659-68. doi: 10.1002/hed.20221. PMID: 15920748.
- 187. Ravasco P, Monteiro-Grillo I, Vidal PM, et al. Dietary counseling improves patient outcomes: a prospective, randomized, controlled trial in colorectal cancer patients undergoing radiotherapy. J Clin Oncol. 2005 Mar 1;23(7):1431-8. doi: 10.1200/JCO.2005.02.054. PMID: 15684319.
- 188. Sanchez-Lara K, Turcott JG, Juarez-Hernandez E, et al. Effects of an oral nutritional supplement containing eicosapentaenoic acid on nutritional and clinical outcomes in patients with advanced non-small cell lung cancer: randomised trial. Clin Nutr. 2014 Dec;33(6):1017-23. doi: 10.1016/j.clnu.2014.03.006. PMID: 24746976.
- 189. Scislo L, Pach R, Nowak A, et al. The Impact of Postoperative Enteral Immunonutrition on Postoperative Complications and Survival in Gastric Cancer Patients - Randomized Clinical Trial. Nutr Cancer. 2018 Apr;70(3):453-9. doi: 10.1080/01635581.2018.1445770. PMID: 29533110.
- 190. Shimizu N, Oki E, Tanizawa Y, et al. Effect of early oral feeding on length of hospital stay following gastrectomy for gastric cancer: a Japanese multicenter, randomized controlled trial. Surg Today. 2018 Sep;48(9):865-74. doi: 10.1007/s00595-018-1665-4. PMID: 29721714.
- 191. Sim E, Kim JM, Lee SM, et al. The Effect of Omega-3 Enriched Oral Nutrition Supplement on Nutritional Indices and Quality of Life in Gastrointestinal Cancer Patients: A Randomized Clinical Trial. Asian Pac J Cancer Prev. 2022 Feb 1;23(2):485-94. doi: 10.31557/APJCP.2022.23.2.485. PMID: 35225460.

- 192. Vidal A, Arnold N, Vartolomei MD, et al. Oncological and functional outcomes of postoperative total parenteral nutrition after radical cystectomy in bladder cancer patients: A single-center randomized trial. Int J Urol. 2016 Dec;23(12):992-9. doi: 10.1111/iju.13228. PMID: 27770454.
- 193. Wang J, Wang L, Zhao M, et al. Effect of Early Enteral Nutrition Support Combined with Chemotherapy on Related Complications and Immune Function of Patients after Radical Gastrectomy. J Healthc Eng. 2022;2022:1531738. doi: 10.1155/2022/1531738. PMID: 35126900.
- 194. Wu W, Zhong M, Zhu DM, et al. Effect of Early Full-Calorie Nutrition Support Following Esophagectomy: A Randomized Controlled Trial. JPEN J Parenter Enteral Nutr. 2017 Sep;41(7):1146-54. doi: 10.1177/0148607116651509. PMID: 27208039.
- 195. Xie H, Chen X, Xu L, et al. A randomized controlled trial of oral nutritional supplementation versus standard diet following McKeown minimally invasive esophagectomy in patients with esophageal malignancy: a pilot study. Ann Transl Med. 2021 Nov;9(22):1674. doi: 10.21037/atm-21-5422. PMID: 34988183.
- 196. Yang L, Gao J, Zhou Y, et al. Effect of Oral Nutritional Supplements on Patients with Esophageal Cancer During Radiotherapy. Cancer Biother Radiopharm. 2020 Aug 20;20:20. doi: 10.1089/cbr.2020.3888. PMID: 32833549.
- 197. Yao R, Zhang T, Zhang J, et al. Effects of postoperative enteral nutrition combined with adjuvant radiotherapy on inflammatory response, nutrition, healing and prognosis in patients receiving radical surgery for esophageal carcinoma. J BUON. 2019 Jul-Aug;24(4):1673-8. PMID: 31646824.
- 198. Zhang Y, Liu L, Li D, et al. Effectiveness of Noninvasive Positive Pressure Ventilation Combined with Enteral Nutrition in the Treatment of Patients with Combined Respiratory Failure after Lung Cancer Surgery and Its Effect on Blood Gas Indexes. Emergency Medicine International Print. 2022;2022:1508082. doi: <u>https://dx.doi.org/10.1155/2022/1508082</u>. PMID: 35811605.
- 199. Zhao M, Li XG, Ma YY, et al. Application of enteral nutrition during perichemotherapy of acute nonlymphocytic leukemia. Journal of Chemical and Pharmaceutical Research. 2014;6(6):768-71. PMID: 602985290.
- 200. Zhu MW, Yang X, Xiu DR, et al. Effect of oral nutritional supplementation on the post-discharge nutritional status and quality of life of gastrointestinal cancer patients after surgery: a multi-center study. Asia Pac J Clin Nutr. 2019;28(3):450-6. doi: 10.6133/apjcn.201909_28(3).0004. PMID: 31464391.
- 201. Zietarska M, Krawczyk-Lipiec J, Kraj L, et al. Chemotherapy-Related Toxicity, Nutritional Status and Quality of Life in Precachectic Oncologic Patients with, or without, High Protein Nutritional Support. A Prospective, Randomized Study. Nutrients. 2017 Oct 11;9(10):11. doi: 10.3390/nu9101108. PMID: 29019951.
- 202. Abdollahi R, Najafi S, Razmpoosh E, et al. The Effect of Dietary Intervention Along with Nutritional Education on Reducing the Gastrointestinal Side Effects Caused by Chemotherapy Among Women with Breast Cancer. Nutr Cancer. 2019;71(6):922-30. doi: 10.1080/01635581.2019.1590608. PMID: 30945949.
- 203. Demark-Wahnefried W, Case LD, Blackwell K, et al. Results of a diet/exercise feasibility trial to prevent adverse body composition change in breast cancer patients on adjuvant chemotherapy. Clin Breast Cancer. 2008 Feb;8(1):70-9. doi: 10.3816/CBC.2008.n.005. PMID: 18501061.
- 204. Lin JX, Chen XW, Chen ZH, et al. A multidisciplinary team approach for nutritional interventions conducted by specialist nurses in patients with advanced colorectal cancer undergoing chemotherapy: A clinical trial. Medicine (Baltimore). 2017 Jun;96(26):e7373. doi: 10.1097/MD.000000000007373. PMID: 28658162.
- 205. Poulsen GM, Pedersen LL, Osterlind K, et al. Randomized trial of the effects of individual nutritional counseling in cancer patients. Clin Nutr. 2014 Oct;33(5):749-53. doi: 10.1016/j.clnu.2013.10.019. PMID: 24269077.

- 206. Qin N, Jiang G, Zhang X, et al. The Effect of Nutrition Intervention With Oral Nutritional Supplements on Ovarian Cancer Patients Undergoing Chemotherapy. Front Nutr. 2021;8:685967. doi: 10.3389/fnut.2021.685967. PMID: 34249995.
- 207. Silander E, Nyman J, Bove M, et al. Impact of prophylactic percutaneous endoscopic gastrostomy on malnutrition and quality of life in patients with head and neck cancer: a randomized study. Head Neck. 2012 Jan;34(1):1-9. doi: 10.1002/hed.21700. PMID: 21374756.
- 208. Silander E, Jacobsson I, Berteus-Forslund H, et al. Energy intake and sources of nutritional support in patients with head and neck cancer--a randomised longitudinal study. Eur J Clin Nutr. 2013 Jan;67(1):47-52. doi: 10.1038/ejcn.2012.172. PMID: 23169469.
- 209. Skaarud KJ, Veierod MB, Lergenmuller S, et al. Body weight, body composition and survival after 1 year: follow-up of a nutritional intervention trial in allo-HSCT recipients. Bone Marrow Transplant. 2019 Dec;54(12):2102-9. doi: 10.1038/s41409-019-0638-6. PMID: 31455897.
- 210. Song G, Liu H. Effect of Hospital to Home nutrition management model on postoperative clinical outcomes of patients with laryngeal carcinoma. Oncol Lett. 2017 Oct;14(4):4059-64. doi: 10.3892/ol.2017.6709. PMID: 28943912.
- 211. Uster A, Ruefenacht U, Ruehlin M, et al. Influence of a nutritional intervention on dietary intake and quality of life in cancer patients: a randomized controlled trial. Nutrition. 2013 Nov-Dec;29(11-12):1342-9. doi: 10.1016/j.nut.2013.05.004. PMID: 24103511.
- 212. Xie FL, Wang YQ, Peng LF, et al. Beneficial Effect of Educational and Nutritional Intervention on the Nutritional Status and Compliance of Gastric Cancer Patients Undergoing Chemotherapy: A Randomized Trial. Nutr Cancer. 2017 Jul;69(5):762-71. doi: 10.1080/01635581.2017.1321131. PMID: 28524705.
- 213. Najafi S, Haghighat S, Raji Lahiji M, et al. Randomized Study of the Effect of Dietary Counseling During Adjuvant Chemotherapy on Chemotherapy Induced Nausea and Vomiting, and Quality of Life in Patients With Breast Cancer. Nutr Cancer. 2019;71(4):575-84. doi: 10.1080/01635581.2018.1527375. PMID: 30449171.

- 214. Ansari M, Porouhan P, Mohammadianpanah M, et al. Efficacy of Ginger in Control of Chemotherapy Induced Nausea and Vomiting in Breast Cancer Patients Receiving Doxorubicin-Based Chemotherapy. Asian Pac J Cancer Prev. 2016;17(8):3877-80. PMID: 27644633.
- 215. Vafa S, Zarrati M, Malakootinejad M, et al. Calorie restriction and synbiotics effect on quality of life and edema reduction in breast cancer-related lymphedema, a clinical trial. Breast. 2020 Dec;54:37-45. doi: 10.1016/j.breast.2020.08.008. PMID: 32898787.
- 216. Pettersson A, Nygren P, Persson C, et al. Effects of a dietary intervention on gastrointestinal symptoms after prostate cancer radiotherapy: long-term results from a randomized controlled trial. Radiother Oncol. 2014 Nov;113(2):240-7. doi: 10.1016/j.radonc.2014.11.025. PMID: 25467005.
- 217. Cho AR, Hong KW, Kwon YJ, et al. Effects of Single Nucleotide Polymorphisms and Mediterranean Diet in Overweight or Obese Postmenopausal Women With Breast Cancer Receiving Adjuvant Hormone Therapy: A Pilot Randomized Controlled Trial. Frontiers in Nutrition. 2022;9:882717. doi: <u>https://dx.doi.org/10.3389/fnut.2022.882717</u>. PMID: 35845810.
- 218. Harvie M, Pegington M, Howell SJ, et al. Randomised controlled trial of intermittent vs continuous energy restriction during chemotherapy for early breast cancer. Br J Cancer. 2022 May;126(8):1157-67. doi: 10.1038/s41416-021-01650-0. PMID: 34912072.
- 219. Villarini A, Pasanisi P, Raimondi M, et al. Preventing weight gain during adjuvant chemotherapy for breast cancer: a dietary intervention study. Breast Cancer Res Treat. 2012 Sep;135(2):581-9. doi: 10.1007/s10549-012-2184-4. PMID: 22869285.